NUREG-1242 Vol. 2, Pt. 1

# NRC Review of Electric Power Research Institute's Advanced Light Water Reactor Utility Requirements Document

Evolutionary Plant Designs Chapter 1

Project Number 669

**U.S. Nuclear Regulatory Commission** 

Office of Nuclear Reactor Regulation



9209240193 920831 PDR PROJ 669A PDR

#### AVAILABILITY NOTICE

Availability of Reference Materials Cited in NRC Publications.

Most documents cited in NRC publications will be available from one of the following sources:

- The NRC Public Document Room, 2120 L Street, NW., Lower Level, Washington, DC 20555
- The Superintendant of Documents, U.3. Government Printing Office, P.O. Box 37082, Washington, DC 20013-7082
- 3. The National Technical Information Service, Springfield, VA 22161

Although the listing that follows represents the majority of documents cited in NRC publications, it is not intended to be exhaustive.

Referenced documents available for inspection and copying for a fee from the NRC Public Document Room include NRC correspondence and internal NRC memoranda; NRC bulletins, circulars, information notices, inspection and investigation notices; licensee event reports; vendor reports and correspondence; Commission papers; and applicant and licensee documents and correspondence.

The following documents in the NUREG series are available for purchase from the GPO Sales Program: formal NRC staff and contractor reports, NRC-sponsored conference proceedings, international agreement reports, grant publications, and NRC booklets and brochures. Also available are regulatory guides, NRC regulations in the *Code of Federal Regulations*, and *Nuclear Regulatory Commission Issuances*.

Documents available from the National Technical Information Service include NUREG-series reports and technical reports prepared by other Federal agencies and reports prepared by the Atomic Energy Commission, forerunner agency to the Nuclear Regulatory Commission.

Documents available from public and special technical libraries include all open literature items, such as books, journal articles, and transactions. *Federal Register* notices, Federal and State legislation, and congressional reports can usually be obtained from these libraries.

Documents such as theses, dissertations, foreign reports to 2 transitions, and non-NRC conference proceedings are available for purchase from the organization spousoring the publication cited.

Single copies of NRC draft reports are available free, to the extent of supply, upon written request to the Office of Administration, Distribution and Mail Services Section, U.S. Nuclear Regulatory Commission, Washington, DC 20555.

Copies of industry codes and standards used in a substantive manner in the NRC regulatory process are maintained at the NRC Library, 7920 Norfolk Avenue, Bethesda, Maryland, for use by the public. Codes and standards are usually copyrighted and may be purchased from the criginating organization or, if they are American National Standards, from the American National Standards Institute, 1430 Broadway, New York, NY 10018.

NUREG-1242 Vol. 2, Pt. 1

# NRC Review of Electric Power Research Institute's Advanced Light Water Reactor Utility Requirements Document

Evolutionary Plant Designs Chapter 1

Project Number 669

Manuscript Completed: August 1992 Date Published: August 1992

Associate Directorate for Advanced Reactors and License Renewal Office of Nuclear Reactor Regulation U.S. Nuclear Regulatory Commission Washington, DC 20555



#### ABSTRACT

The Electric Power Research Institute (EPRI) is preparing a compendium of technical requirements, referred to as the "Advanced Light Water Reactor [ALWR] Utility Requirements Document," that is applicable to the design of an ALWR power plant. When completed, this document is intended to be a comprehensive statement of utility requirements for the design, construction, and performance of an ALWR power plant for the 1990s and beyond.

The Requirements Document consists of three volumes. Volume I, "ALWR Policy and Summary of Top-Tier Requirements," is a management-level synopsis of the Requirement. Document, including the design objectives and philosophy, the overal) physical configuration and features of a future nuclear plant design, and the steps necessary to take the proposed ALWR design criteria beyond the conceptual design state to a completed, functioning power plant. Volume II consists of 13 chapters and contains utility design requirements for an evolutionary nuclear power plant [approximately 1350 megawatts-electric (MWe)]. Volume III contains utility design requirements for nuclear plants for which passive features will be used in their designs (approximately 600 MWe).

The staff of the Office of Nuclear Reactor Regulation, U.S. Nuclear Regulatory Commission, has prepared Volumes 1 and 2 (Parts 1 and 2) of its safety evaluation report (SER) to document the results of its review of Volumes I and II of the Requirements Document. Volume 1, "NRC Review of Electric Power Research Institute's Advanced Light Water Reactor Utility Requirements Document - Program Summary," provides a discussion of the overall purpose and scope of the Requirements Document, the background of the staff's review, the review approach used by the staff, and a summary of the policy and technical issues raised by the staff during its review. Volume 2, "NRC Review of Electric Power Research Institute's Advanced Light Water Reactor Utility Requirements Document - Evolutionary Plant Designs," gives the results of the staff's review of the 13 chapters of the Requirements Document for evolutionary plant designs. Volume 3, "NRC Review of Electric Power Research Institute's Advanced Light Water Reactor Requirements Document - Passive Plant Designs," scheduled to be issued in September 1993, will give the results of the staff's review of the 13 chapters of the Requirements Document for passive plant designs. Preliminary drafts of Volumes 1 and 2 were forwarded to the Commission and the Advisory Committee on Reactor Safeguards (ACRS) on May 12. 1992.

In staff requirements memoranda (SRM), the Commission instructed the staff to provide an analysis detailing where the staff proposes departure from current regulations or where the staff is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs. The staff considers these to be policy issues. Appendix B to Chapter 1 of Volume 2 of this report gives the staff's regulatory analysis of those issues identified for the evolutionary plant designs. These issues have been addressed in Commission papers SECY-90-016, "Evolutionary Light Water Reactor Certification Issues and Their Relationship to Current Regulatory Requirements"; SECY-91-078, "Chapter 11 of the Electric Power Research Institute's Requirements Document and

EPRI Evolutionary Plant SER

۵

Additional Evolutionary Light Water Reactor Certification Issues"; and in draft Commission papers, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," and "Design Certification and Licensing Policy Issues Pertaining to Passive and Evolutionary Advanced Light Water Reactor Designs," that were issued on February 27 and July 6, 1992, respectively.

In SRM dated June 26, 1990, and April 1, 1991, the Commission provided its decisions on SECY-90-016 and SECY-91-078 as they apply to evolutionary designs. The Commission will be reviewing the basis for the approach that the staff is proposing for those issues discussed in the draft Commission papers of February 27 and July 6, 1992, and, accordingly, may at some future point in the review determine that such issues involve policy questions that the Commission may wish to consider. These issues are considered fundamental to agency decisions on the acceptability of the ALWR designs. The staff will ensure satisfactory implementation of Commission guidance regarding these matters during its review of individual applications for final design approval and design certification.

There are no open issues pertaining to the Requirements Document for evolusinary plant designs other than policy issues on which the staff has taken a contition, but for which the Commission has not had the opportunity to provide guidance. These issues are summarized in Section 4 of Volume 1 and discussed in detail in this report.

#### TABLE OF CONTENTS

Page

#### PART 1

#### CHAPTER 1, "OVERALL REQUIREMENTS"

1	INTR	ODUCTION	4 4 4 A	5	4 4 4		$i = \infty$			κ.	k - 1				4	×.	e 1		4	1.1-1
	1.1 1.2 1.3 1.4 1.5	Review C Scope an Policy I Outstand Vendor-	riteria d Struc ssues . ling Iss or Util	ture ues ity-5	of Ch	apter ic it	en.s		* * * * *	* * * *	  				* * *	* * * *			关 东 产 声 等	$1 \cdot 1 - 1 \\ 1 \cdot 1 - 1 \\ 1 \cdot 1 - 2 \\ 1 \cdot 1 - 3 \\ 1 \cdot 1 - 5$
2	SAFE	TY DESIGN	REQUIR	EMENT	rs.,			4	4					Α.,	i.					1.2-1
	2.1 2.2 2.3	Introduc Accident Core-Dam	tion . -Resist age-Pre	ance venti	Requi	remen quire	ts ment	t s	4 4 4	4 ) 4 ) 4 )		* * *	* * *	* *	- - 	, . , .		4		1.2-1 1.2-2 1.2-2
		2.3.1 G 2.3.2 L 2.3.3 S	eneral icensin afety-M	Requi g-Des largir	remen ign-B n-Basi	ts , asis s Req	Requ ui~e	uir eme	em	ent s	s		* *	1 1 1	× × ×			4	× + +	1.2-2 1.2-3 1.2-6
	2.4 2.5	Mitigati Analysis	on Requ Requir	ireme ement	ents . s and	Ácće	ptar	nce	Ċ	rit	er	ia		4	5 5	4 - 1 4 - 1		3		1.2-7 1.2-7
3	PERF	ORMANCE D	ESIGN R	EQUIF	REMENT	Ś	к. к		į,		c i			÷	ċ	à	ċ,	ų,	÷	1.3-1
	3.1 3.2 3.4 5.6 7.8 3.3 3.3 3.5 3.3 3.3 3.3 3.3 3.3 3.3 3.3	Introduc Plant Si Plant De Maneuver Event Re Core Per Radioact Occupati	tion . sign Li ing and sponse formanc ive Was onal Ra	fe . Resp Capat e . te Di diati	onse bilitv sposa on Ex	to Gr 1 posur	id {	)em	ian	ds										1.3-11.3-11.3-11.3-11.3-11.3-11.3-21.3-31.3-3
4	STRU	CTURAL DE	SIGN BA	SES		з ч.	÷ŝ.		i.	÷		÷.	÷	÷,	, i		Ċ,	í,		1.4-1
	4.1 4.2 4.3	Introduc Relation Classifi	tion , ships t cation	o Des Requi	ign-B remen	asis ts .	Éver	nts		•					*			•		1.4-1 1.4-1 1.4-1
		4.3.1 S 4.3.2 S	afety C eismic	lassi Class	ficat	ion tion			* 1					1			ģ		•	1.4-1

4.4	Codes a	and	Star	ndar	rdis	c.	÷			4	+	÷	×.	÷	÷		÷			×.	4	x	x	×.		1.	4-	5
	4.4.1	Maj	or (	)esi	ign Lect	an	d (	Con 1 S	sti	rua nda	ti	or	1 0	00	les		•	1		ŝ	1			ł	÷	1.	4-	5
	4.4.3	Reg	ulat	tory	P	os i	ti	ons		1		4	*	÷	-	*	ŝ	÷	4	ł,	ŝ		ŝ	•	*	î.	4-	6
4.5	Design	Loa	ds a	and	Cor	ndi	tii	on s												9			į,	-		1.	4-	9
	4.5.1	Int Nat	rodi ural	icti I Pł	ion nend	, ome	na			A 		4	3	•	*		1	1	4	1		1	4	× *	*	1.	4-4-	9 10
	4.5.3	Sit	e Pr	oxi	mit	V.	Mar	n-M	adi	a 1	łaż	ar	ds										1	4	*	1.	4-	16
	4.5.4	Pla	nt (	Der	at	ina	11	and	ē.													1		-	1.	1.	4-	16
	4.5.5	In-	Plar	nt i	laza	ard	\$			w.	÷	4	1	÷	ŝ	à	÷	X,	ŝ	÷	x	ł	8	÷	*	1.	4-	17
4.6	Load Co	ombi	nati	ions	i i			4				4		÷	÷	÷		÷			×.		÷	ź	4	3	4-	20
	4.6.1	Rui	Idie	ia s	and	i s	tra	int.	in	a e																1	4-	20
	4.6.2	Svs	tems	i ar	id i	au	in	nen	t			1								2			0		<u>.</u>	1.	4-	25
			e settis			. Ala	- Jes		÷	1		÷.	1			í						ĺ.						
4.7	Design	Met	hode	2010	IY .	*	8 - )	e 19		×	÷		1	×	×.	×	ł	*	×		×	4	×	÷	×.	1.	4 -	25
	4.7.1	Int	rodu	icti	on								1	÷.								i.		1		1.	4-	26
	4.7.2	Bui	ldir	1QS	and	1 5	tri	ict	uri	2.5								1								1.	4-	26
	4.7.3	Sys	tems	s ar	nd E	Equ	ip	nen	t	я.	4	×	8	÷	3	÷	÷.	÷.	÷	×	÷	÷	÷	ł,	8	1.	4-	29
4.8	Testing	g an	d Qı	ial i	fic	at	ioi	ţ	×	×	÷	÷	÷,		,×	ł	÷	÷	ж	÷	4	ж		÷.	×,	1	4-	32
	4.8.1	Sei	smic	ar	nd [	)yn	ami	ic	Qui	a]	ifi	03	ti	or	1 0	f	Me	ch	ar	iic	al	i a	ind	ł.				
		Ele	ctri	ical	E	iup	pme	ent			÷.,	à.	8	÷.	۰.	i.	÷.	÷.	÷		÷.	×	4	×	$\mathbf{x}_{i}^{(i)}$	1.	4-	33
	4.8.2	Env	iror	nmer	ital	Q	ua	lif	ici	ati	ion	0	f.	Mę	act	ar	ic	al	ê	til	1							сń.
		Ele	ctri	cal	E	ju i	pme	ent.	Υ.	×	A	н.	×,	i.	1	ł	÷	÷	×,	÷.	3	ł	×.	×	٢.	1.	4-	35
MATER	RIALS		1.1	1.7	÷.	+	ε		×	ż	a	£.	×	*	×	÷	5	×,	X	ł	ŝ	÷	÷	ŗ	1	1	5-	1
5.1	Introdu	ucti	on				1-	÷.,			1		. 1		1	ġ.	÷.	÷.	į.		d, i	÷		ġ.	38	1.	5-	1
5.2	General	RA	nuir	eme	int	÷.,	2.			1	0.	÷.	34	1	С.	2	0.	÷.	÷.,	Ξ.	£.,	3	1	2	÷.,	1	5-	î.
	warran wa	1 1.100	44.13			1				ĵ.		4	1	1			ŕ.					2				9		
	5.2.1	Res	pons	sibi	111	LV.	for	r M	ate	eri	al	s	Se	16	ect	10	n	ġ.		1	۰.	1	4	1	÷.	1.	5-	1
	5.2.2	Ide	ntif	Fica	itic	n	of	Ma	ter	ria	115	1	n	Cr	·it	ic	al	10	on	npc	ne	ent	s	5	10	1.	5-	1
	5.2.3	Cod	es a	and	Sta	and	are	15				1			11								η.	6	21	1.	5-	1
	5.2.4	Des	ian-	Bas	is	Co	ns	ide	rat	t i e	'n		01	0.			1	0.	0.		1	£.,	2	8	11	1	5-	1
	5.2.5	Haz	arde	aus	Mat	er	ial	Is									5							1		1	5	2
	5.2.6	Rev	IPW	of	IWS	2 F	XIN	ari	en	P		6							6							1	5-	2
	5.2.7	Mot	a114	IC N	late	ini.	ali								1						Ĩ.					1	5-	2
	5.2.8	Non	-Met	tall	ic	Ma	ter	ria	15		+	ŝ.			1	÷.	4.	Ş.	-	•	4	ŝ,	3	*	*	1.	5-	4
5.3	Materia	als	Sele	ecti	on						x	۰.	ě.	×	i,	i.	à			÷		÷	ł		i.	1.	5-	5
	5.3.1	Mat	eria	als	in	th	e f	Rea	cti	or	Co	101	an	it	Sy	st	en	n a	ind	1 6	le'	at	ed	ł				
		Sys	tems	s .	× 8		8.13	é i a	$\mathbf{x}_{i}$		÷.,	×.	*	÷.	4	k.	4			×	+	*		à.		1.	5-	5

# Page

		5.3.2 5.3.3 5.3.4 5.3.5	Mater Faste Therm Seals	ials ners al lr , Gas	in and sul sket	Fe A at	edwa dhes ion Pac	ate siv Ma cki	r, es te ng	St.	ea 1s ea	m, lar	an	d	C c	ind in	len	is a	te		iys	te	inis ,		1.	5- 5- 5-	13 13 14
			and C	lean	ing.	p	acka	àqi	ng	ora . à	nd	St	tor	au	i e	Ma	ite	rí	al	s	11	.10	, s.	1	1.	5-	15
		5.3.6	Elect	rical	Ma	te	ria	15	1		8	*	k.,	4	1	÷	16		k	*	÷.	÷	÷.	*	1.	5-	15
		5.3.7	Weld	Mater	"ial	5	4	۰.	κ.		. 4	×		4	£.,	×	*	1	1	х.	*	4	х,	1	1.	5-	15
	5.4	Process	Cont	rols	×	κ.		4			÷		8	÷	۰.	ŝ.	×.	a.	e i		÷	×.	×.	ŧ,	1.	5-	16
		5.4.1	Surfa	ce Co	ndi	ti	on			÷.,								4		,					1.	5-	16
		5.4.2	Fabri	catio	on C	on	trol	15	8		×	÷.		4		è.	5	4	i.	+			a.	4	1.	5-	16
		5.4.3	Exami	natio	on a	nd	Tes	sts		e . e	×	1	*	4	8	8	5	х.	ε.	•	Ξ.	÷	х.,	х.	1.	5-	17
		5.9.9	Shipp	ing a	ind	St	oraç	<u>je</u>	* .	e: - 4	- 4	. 4	×	8	1	3	×.	۰.	5	*	÷.	8.	÷	×	1	5- 2	10
		5.4.6	Flush	Hur	tro	1.00	nd I	av	in	e - 1	. *	1	*	1	1	×	÷	1	*	۴.	1	1	8	*	1	5-	18
			1 1 1 2 3 11			a	inu i	. a j	up.					8	1	Č.	•			1	×	÷.		÷.,			
	5.5	Enviror	imenta	1 Cor	idit	10	ns	*	*	×		×	*	*			٩,	4	۰.	ж.	×	×	*	×.	1.	5-	18
		5.5.1	Range	of f	rvi	ro	nmer	nta	1 (	Con	di	tic	ns	Ľ.	į.	s,	1	÷		÷	9	+	÷.	4	1.	5-	18
		5.5.2	Water	Cher	nist	ry	Des	sig	n I	Bas	is	÷	1	۰.	÷.	4	÷	a.	×.	ĸ	ч,	8	×	4	1.	5-	19
	5.6	Conclus	ion			2	÷ +	÷.			×	×	ĸ	8	÷	÷	÷			×.	ž.		×	×.	1.	5-	19
6	RELIA	ABILITY	AND A	VAIL	ABIL	IT	Ý,	÷.	,		÷				*	4	×.		÷	ŝ,		i.	4		1.	6-	1
	6.1 6.2 6.3	D-RAP ( Basic F D-RAP F	Goals Progra Perfor	and ( m Ele mance	)bje emen e St	ct ts an	ive: of dard	s th ds	ė I	Ď-Ř	AP	+	1.1	ж 4 2		4 A A	× * *	*	4 4 4			*	1. 1. 1.		1.1.1.	6- 6-	336
	0.4	System	Desig	n	14	6.7	8. 4	. *	8	÷	×	÷	1	ж	÷	÷.	х.,	Υ.,	÷	×	÷	×.	×.	χ.,	1.	6-	1
	0.0	Concius	sion	1.1		÷	4, 4.	8	8.7	÷	9	÷	۰.	÷.	÷	1	3	٤,	* ?	1	Υ.	1	×	٩.	4	0-1	5
7	CONST	TRUCTION	AND	CONST	rruc	TI	BIL	ITY		• •	×	÷	×.	×	ł	ķ.	£	÷		6		÷		Į.	1.	7-	1
8	OPERA	ABILITY	AND M	AINTA	AIN.	91	LIT	Ý	ŝ.		ŝ		÷	÷	ł,	ł	ł		ŧ,	ŝ,				ć.	1,	8-	1
	8.1	Introdu Provisi Minimiz	iction ions T	o Ent	hanc	e l	Open	rab	i1	ity	a	nd	Ma	in	ta	in	ab	iı	it	ý	i.				1.	8- 8-	1
	8.4	Facilit	v Rea	uirer	nent	5	0.0	10	1 31	onn	eı	*		2	1	1	1	1	÷.,	Ċ.	1	٩.	÷.		1	8-	4
	8.5	Provisi	ions f	or Re	apla	ice	ment	t o	fl	Maj	or	Ċe	omp	on	en	its	Ċ.	į.		h	Ľ.,				1.	8-	4
	8.6	Inspect	tion a	nd Te	esti	ng						1		4		-		į.				1			1.	8-	4
	8.7	Hazardo	ous an	d Top	cic	Ch	emic	cal	\$					1								į.			1.	8-	5
	8.8	Conclus	sion	1.	ί÷,	5	6		ŝ.		1		×	÷	÷	s.	÷	۶.		÷	я,		a.	*	1.	8-	5
9	QUAL	ITY ASSU	JRANCE			4	i a		è			ų,	,		ł			ł		÷.		i,	i.	ł,	1.	9-	1
10	LICE	NSING .	e e e	< 8.5	- 4	χ.	4 - 4	×	*		×		÷				×.						×		1.	10	-1

EPRI Evolutionary Plant SER

Page

1	1	DESIGN	PROCESS			ł.	а. л.						÷	i.											1.	11-1	
		11.1 11.2	Introduct	tion . gy Base	• •	*		4 	· · ·	*	2						X A		a A	9. A.	* *	•	4 4	1	1.	11-1	
		11.3	Design L	ife .	1.1.	-	÷ . +		× . *	*	*			4	•	*	× .	×	4	х.	8	*	4	3	1	11-1	5
		11.5	Standard	ization	atit		1	1		1	Ċ,	1	1	1	ţ.	1	1	1	а. 	ĵ.	ĵ.,	į.	÷.	×	î.	11-2	į.
		11.6	Specific	ALWR I	esig	in	Pro	ces	s R	leq	uiı	en	ien	ts		2	8			1	i.		4	1	1.	11-2	2
		11.7	Project	Informa	ition	1 11	etw	ork	1			4		4			ł.	4	٩	έ.		*		÷	1.	11-2	
		11.8	Design D	evelop	nent	P1	ari	14	4 4			*	÷	÷	×.	r.	÷	4	٩.	8	٩.	4	÷	4	1	11-1	š
		11.9	Configur	ation /	lanag	)em	ent	4	6.3		1	ч.	6.1	4	8	*	۴.	£.	*	*	3	A.		1	1	11-3	8 2
		11.10	Design I	ntegra	cion Neu 1	ìne	100	in a	i.i.	, na e		ч.	4	*	1	1	۰.	4	1	3	÷.,	1	1	1	1	11-1	6
		11.12	Informat	ion Ma	12001	tes	1 5	vet	4-m	CW S			1		1	1		8- -	Č.	1	1	Ľ.	÷.		1	11-1	4
		11.13	Design/C	onstru	ctio	n I	nte	ara	tic	'n		1	1		1		1	1	į.	1	1	÷.	1	1	- 1	11-	5
		11.14	Engineer	ing Fi	eld !	Ver	ifi	cat	ior	0 1	f	As-	Bi	111	t	Co	nd	it	10	ns			÷.		1	11-	5
		11.15	Conclusi	on	1.1							×	ч.	4	÷.	×.	ł.	8	х.	à.	÷	÷			.1	11-	6
	12	MECHA	NICAL EQU	IPMENT	DES	IGN	RE	QUI	REP	IEN	ITS	-			*	,		÷	ĸ	,	*	ł	ł	1	1	.12-	1
		12.1 12.2	Introduc Inservic	tion . e Test	ing	of	 Pun	ip s	and	i i	ai	ve	s	* *	1		1	*		н 3	*		1	x E	1	.12~ .12~	1
			12.2.1 12.2.2 12.2.3 12.2.4	Scope Valves Pumps Conclu	of t	he	Ins	erv	vic.	e 1	les	tii	ng	P1	roç	iri	1 1 1	×	* + * *			* * * *	年 年 注 法		1111	.12- .12- .12- .12-	2367
		12.3	Radiatio	on Prot	ecti	on	Cor	151	der	at	ion	s		3		÷		8		5	*		ī.		1	.12-	7
	13	CONCL	USION .		÷			- 4	x	÷		-4	×	×		4		j.	,			ļ		×	1	. 13-	1
			CHAPTER	1, APF	ENDI	X	Α,	"PR	A K	ΕY	AS	su	MP	ΤI	ON:	S	ANI	0	GR	OU	ND	RU	LE	S"			
	1	INTRU	DUCTION		4.4	4	×		×	×	×	×		,		X				4	ł			4	1	A.1-	1
		1.1 1.2 1.3 1.4 1.5 1.6	Review C Scope and Policy I Outstand Vendor- Overall	riteria d Struc ssues ing Is or Uti Scope	ture sues lity	- Sp Met	f A eci hod	ppe fic	ndi İt	x .em	A	to	ch	ap	te	r	1	te							1	A.1. A.1. A.1. A.1. A.1. A.1.	333356

1.7Definition of Core Damage1A.1-81.8Point-Estimate Quantification1A.1-91.9Uncertainty Treatment1A.1-91.10Form of Results1A.1-12

- 201			
- 2.2	100	2.	2014
87	38.1	6.3	6.0
	-	- 2	

2	PLANT MODELING	1A.2-1
	2.1 Model Structure 2.2 Initiating Evants 2.3 Success Criteria 2.4 Sequence Logical Identity 2.5 Quantification 2.6 Modeling of Dependencies 2.7 Interaction and Modeling of the Containment Systems 2.8 Common-Cause Failures 2.9 Human Interaction 2.10 Mission Time 2.11 Reliability Data	1A.2-1 1A.2-2 1A.2-2 1A.2-2 1A.2-2 1A.2-3 1A.2-4 1A.2-4 1A.2-4 1A.2-4 1A.2-4
3	EXTERNAL EVENTS	1A.3-1
	3.1 Identification of Initiating Events	1A.3-1
	3.3 Events That Will Require Quantitative Assessment for Each	1A.3-1
1	ALWR C	1A.3-4
4	CONTAINMENT ANALYSIS	1A.4-1
	<ul> <li>4.1 Core-Damage-Sequence Binning</li> <li>4.2 Containment System Analysis</li> <li>4.3 Containment Isolation</li> <li>4.4 Containment Bypass</li> <li>4.5 In-Plant Sequence Assessment</li> <li>4.6 Containment Event Analysis</li> <li>4.7 Source Term Definition</li> <li>4.8 Plant Release Categories</li> </ul>	1A.4-1 1A.4-1 1A.4-1 1A.4-2 1A.4-2 1A.4-3 1A.4-5 1A.4-5
5	OFFSITE CONSEQUENCES	1A.5-1
	5.1 Implementation of the Public-Safety Requirement	1A.5-1 1A.5-1
6	UNCERTAINTY AND SENSITIVITY ANALYSES	1A.6-1
	<ul> <li>6.1 Analysis of Systems and Sequences</li></ul>	1A.6-1 1A.6-4 1A.6-5
7	HUMAN RELIABILITY ANALYSIS	1A.7-1
	7.1 Scope and Objectives of HRA	1A.7-2 1A.7-3 1A.7-3

4

Page

8 CONCLUSION	1A.8-1
ANNEX A - RELIABILITY DATA BASE FOR ALWR PRAS	s
ANNEX B - ALWR REFERENCE SITE	1A.B-1
ANNEX C - COMPONENT SEISMIC CAPACITIES	1A.C-1

### CHAPTER 1, APPENDIX B, "LICENSING AND REGULATORY REQUIREMENTS AND GUIDANCE"

1	INTRO	DUCTION	-1
	1.1 1.2 1.3 1.4 1.5	Review Criteria	-1 -2 -3 -4
2	PLAN	OPTIMIZATION SUBJECTS	-1
	2.1	Issues Related to Chapter 1 of the Evolutionary Requirements Document	-1
		2.1.1 Operating-Basis Earthquake and Dynamic Analysis Methods	-1 -3
	2,2	Issues Related to Chapter 2 of the Evolutionary Requirements Document	-3
	2.3	Issues Related to Chapter 3 of the Evolutionary Requirements Document	2-4
		<ul> <li>2.3.1 BWR Main Steamline Isolation Valves and Leakage introl and Classification of Main Steamline of Boiling-Water Reactor</li> <li>2.3.2 Simplification of Postaccident Sampling System 18.3</li> </ul>	2-4 2-9
	2.4	Issues Related to Chapter 4 of the Evolutionary Requirements Document	2-10
		2.4.1 Reactor Pressure Vessel Level Instrumentation for PWRs 18.	2-10
	2.5	Issues Related to Chapter 5 of the Evolutionary Requirements Document	2-12
		2.5.1 Containment Leak Rate Testing	2-12

ġ,

9

ş

# Page

		2.5.2 2.5.3 2.5.4	Physically Based Source Term	18.2-12 18.2-17 18.2-17
3	GENE	RIC SAFE	ETY ISSUES	18.3-1
	3.1	Introdu	action	18.3-1
	3.2	Staff	Evaluations of Generic Issues	18.3-3
		3.2.1	I.D.3. Safety System Status Monitoring	18.3-3
		3.2.3	Research - On-Line Reactor Surveillance System II.E.4.3, Containment Design - Integrity Check II.H.2, Obtain Technical Data on the Conditions Inside	18.3-3 18.3-4
		3.2.5	the Three Mile Island, Unit 2, Containment Structure . 11.J.4.1, Revise Deficiency Reporting Requirements . A-29, Nuclear Power Plant Design for the Reduction of	18.3-4 18.3-5
		3.2.7	Vulnerability to Industrial Sabotage	18.3-5 18.3-7
		3.2.9	Plants A-47, Safety Implications of Control Systems	18.3-9 18.3-10
		3.2.11 3.2.12	Burns on Safety Equipment B-17, Criteria for Safety-Related Operator Actions B-32, Ice Effects on SafetRelated Water Supplies B-55, Improved Reliability of Target Rock Safety/Relief	18.3-10 18.3-11 18.3-13
		3.2.14	Valves B-56, Diesel Reliability B-61, Allowable Emergency Core Cooling System Equipment	18.3-14 18.3-15
		3.2.16	B-64, Decommissioning of Reactors C-8. Main Steamline Isolation Valve Leakage Control	18.3-16 18.3-16
		3.2.18 3.2.19 3.2.20	Systems 2. Failure of Protective Devices on Essential Equipment 15. Radiation Effects on Reactor Vessel Supports 23. Reactor Coolant Pump Seal Failures	18.3-17 18.3-17 18.3-18 18.3-19
		3.2.22	Recirculation 29, Bolting Degradation or Failure in Nuclear Power	1B,3-20
		3.2.23	Plants	18.3-20
		3.2.24	57, Effects of Fire Protection System Actuation on Safety-Related Equipment	18.3-24
		3.2.25	70, Power-Operated Relief Valve and Block Valve Reliability	1B.3-28
		3.2.26	73, Detached Thermal Sleeves	18.3-28

EPRI Evolutionary Plant SER

# Page

3.2.27	75, Generic Implications of ATWS (Anticipated Transient	ala a dala
	Without Scram) Events at the Salem Nuclear Plant	18.3-29
3.2.28	76. Instrumentation and Control Power Interactions	18.3-30
3.2.29	79. Unanalyzed Reactor Vessel Thermal Stress During	and a size
	Natural Convection Cooldown	1B.3-33
3.2.30	82. Beyond-Design-Basis Accidents in Spent Fuel Pools	18.3-35
3.2.31	83. Control Room Habitability	18.3-36
3.2.32	84. Combustion Engineering Power-Operated Relief Valves	18.3-39
3.2.33	87. Failure of High-Pressure Coolant Injection Steam-	
	line Without Isolation	18.3-39
3.2.34	\$4. Additional Low-Temperature Overpressure Protection	and a star
	for Light-Water Reactors	18.3-40
3.2.35	96. Residual Heat Removal Suction Valve Testing	18,3-42
3.2.36	99. Reactor Coolant System/Residual Heat Removal	
	Suction Line Valve Interlock in PWRs	18.3-42
3.2.37	101, BWR Water Level Redundancy	18.3-42
3.2.38	103, Design for Probable Maximum Precipitation	18.3-43
3.2.39	105, Interfacing Systems Loss-of-Coolant Accident	
	at LWRs	18.3-43
3.2.40	106 Piping and Use of Highly Combustible Gases in	
	Vital Areas	18.3-46
3.2.41	107, Generic Implications of Main Transformer Failures	18.3-48
3.2.42	110. Equipment Protective Devices on Engineered Safety	
	Features	1B.3-49
3.2.43	113, Dynamic Qualification Testing of Large-Bore	
	Hydraulic Snubbers	18.3-50
3.2.44	118, Tendon Anchorage Failure	18.3-51
3.2.45	120, On-Line Testability of Protection Systems	18.3-52
3.2.46	121, Hydrogen Control for Large, Dry PWR Containments	18.3-54
3.2.47	122.1.a, Davis-Besse Loss-of-All-Feedwater Event of	
	June 9, 1985: Short-Term Actions - Failure of isola-	in a be
	tion Valves in Closed Position	18.3-55
3.2.48	122.1.b, Davis-Besse Loss-of-All-Feedwater Event of	
	June 9, 1985: Short-Term Actions - Recovery of	10 0 00
1.	Auxiliary Feedwater	18.3-55
3.2.49	122.1.c, Davis-Besse Loss-of-All-Feedwater Event of	
	June 9, 1985: Short-Term Actions - Interruption of	10 2 22
	Auxiliary Feedwater Flow	18.3-55
3.2.50	122.2, Davis-Besse Loss-of-All-Feedwater Event of	
	June 9, 1985: Short-Term Actions - Initiating	10.0.00
1.1	Feed-and-Bleed	18.3-55
3.2.51	123, Deficiencies in the Regulations Governing Design-	
	Basis-Accident and Single-Failure Criteria Suggested	10 2 50
	by the Davis-Besse Event of June 9, 1985	18.3-50
3.2.52	124, Auxiliary Feedwater System Reliability	10.3-50
3.2.53	125.11.7, Davis-Besse Loss-of-All-Feedwater Event of	
	June 9, 1985: Long-Term Actions - Reevaluate Provision	
	To Automatically Isolate reedwater from Steam Generator	10 2 50
	During a Line Break	18.3-58

EPRI Evolutionary Plant SER

#### 3.2.54 125.11.11, Davis-Besse Loss-of-All Feedwater Event of June 9, 1985: Long-Term Actions - Recovery of Main Feedwater as Alternative to Auxiliary Feedwater . . . 18.3-58 3.2.55 127, Maintenance and Testing of Manual Valves in 18.3-59 18.3-59 3.2.55 128, Electrical Power Reliability . . . . . . . . . . . 3.2.57 130, Essential Service Water Pump Failures at Multi-18.3-61 3 2.58 132, Residual Heat Removal Pumps Inside Containment . 12.3-62 3.2.59 135, Steam Generator and Steamline Overfill . . . . 18.3-63 3.2.60 142, Leakage Through Electrical Isolators in Instrumentation Circuits . 3.2.61 143, Availability of Chilled Water Systems and Room 18.3-65 18.3-65 3.2.62 151, Reliability of Recirculation Pump Trip During an Anticipated Transient Without Scram . . . . . . 18.3-66 3.2.63 153, Loss of Essential Service Water in Light-Water Reactors 3.2.64 HF 4.4, Procedures - Guidelines for Upgrading Other 18.3-66 18.3-67 3.2.65 HF 5.1, Man-Machine Interface - Local Control Stations 18.3 - 683.2.66 HF 5.2. Man-Machine Interface - Review Criteria for Human Factors Aspects of Advanced Controls and 18.3-68 4 REGULATORY DEPARTURE ANALYSES 18.4-1 ANNEX A - DOCUMENTS RELATED TO SECY-90-016, "EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS, " JANUARY 12, 1990 . . . 1B.A-1 ANNEX B - DUC'L'ENTS RELATED TO SECY-91-078, "CHAPTER 11 OF THE ELECIKIC POWER RESEARCH INSTITUTE'S (EPRI'S) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES, " MARCH 25, 1991 . . . . . . . 1B.B-1 ANNEX C - DOCUMENTS RELATED TO THE DRAFT COMMISSION PAPER "ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS." FEBRUARY 27, 1992 ..... 1B.C-1 ANNEX D - DRAFT COMMISSION PAPER "DESIGN CERTIFICATION AND LICENSING POLICY ISSUES PERTAINING TO PASSIVE AND EVOLUTIONARY ADVANCED LIGHT WATER REACTOR DESIGNS," JULY 6, 1992 . . . 1B.D-1

Page

#### PART 2

#### CHAPTER 2, "POWER GENERATION SYSTEMS"

1	INTRODUCTION				×.	4	i. 1	×	4	÷	*		×	2.1-1
	<ol> <li>1.1 Review Criteria</li> <li>1.2 Scope and Structure of Chapter 2</li> <li>1.3 Policy Issues</li> <li>1.4 Outstanding Issues</li> <li>1.5 Vendor- or Utility-Specific Item</li> </ol>	* * *	* * * *			7 3 3 8 8 8	4 4 4 4 4 4 4 5 4 5 4	* * * *			医关注法 法		* * * * *	$\begin{array}{c} 2 \ , 1 \ - 1 \\ 2 \ , 1 \ - 1 \\ 2 \ , 1 \ - 2 \\ 2 \ , 1 \ - 2 \\ 2 \ , 1 \ - 2 \\ 2 \ , 1 \ - 2 \end{array}$
2	COMMON REQUIREMENTS	,	ć.		÷	,	i a	,		x				2.2-1
	2.1 General Requirements 2.2 Specific Requirements	4 4				4 4	4 - 4 4 - 4		4		х ) х , )	, .	*	2.2-1 2.2-1
	2.2.1 Valves . 2.2.2 Materials 2.2.3 Instrumentation and Contr						4 - 4 2 - 4 9 - 4	* * *	* *	4	4 1 4 1		2. 	2.2-1 2.2-2 2.2-2
3	MAIN/EXTRACTION STEAM SYSTEM		, .			a.	× +	÷	× .		ξ.,		÷	2.3-1
	<ul> <li>3.1 System Definition</li></ul>	化 来 清 清 清 年 天			法律师法律师法		****							$\begin{array}{c} 2 & .3 & -1 \\ 2 & .3 & -2 \\ 2 & .3 & -3 \\ 2 & .3 & -3 \\ 2 & .3 & -4 \\ 2 & .3 & -4 \\ 2 & .3 & -5 \end{array}$
4	FEEDWATER AND CONDENSATE SYSTEM	ł,		2	i.		ı.	÷.,	ų.	÷	Ę,	8	į.	2.4-1
	<ul> <li>4.1 System Definition</li></ul>													$\begin{array}{c} 2 \cdot 4 - 1 \\ 2 \cdot 4 - 1 \\ 2 \cdot 4 - 2 \\ 2 \cdot 4 - 2 \\ 2 \cdot 4 - 3 \\ 2 \cdot 4 - 5 \end{array}$
5	CHEMICAL ADDITION SYSTEM			6	į									2.5~1
	<ul> <li>5.1 System Definition</li></ul>						• •							2.5-1 2.5-1 2.5-1 2.5-1

Farm

	5.5 Instrumentati 5.6 Maintenance . 5.7 Conclusion .	on and	Cont	trol	5	1 1 1 1 1 1	*	4 4 4	4 - 4 4 - 4			4 . 4 3 . 3 4 . 4	* *		а н н н н н	*	*	2.5-2 2.5-2 2.5-2
6	CONDENSATE MAKEUP	PURIFI	CATIO	ON S	5YS1	EM	÷	×		×			×	÷			*	2.6-1
	<ul> <li>6.1 System Defini</li> <li>6.2 Performance R</li> <li>6.3 System Featur</li> <li>6.4 Component Fea</li> <li>6.5 Instrumentati</li> <li>6.6 Maintenance</li> <li>6.7 Conclusion</li> </ul>	tion . lequirent es . tures on and	Cont	s	5	* * * * * * *	* * * * * * *	* * * * * *	* * * *		* * * * * * *	* * * * * * *	******	* * * * * *	* * * * * *			2.6-1 2.6-1 2.6-2 2.6-2 2.6-3 2.6-4 2.6-4
7	AUXILIARY STEAM SY	STEM .					.+	×.			×			4				2.7-1
	<ul> <li>7.1 System Defini</li> <li>7.2 System Interf</li> <li>7.3 Performance F</li> <li>7.4 System Featur</li> <li>7.5 Component Feat</li> <li>7.6 Maintenance .</li> <li>7.7 Conclusion .</li> </ul>	tion . aces . lequiren es itures	ment:	5		* * * * * * * *	小学 阿米 子 资本		* * * *			* * * * * *	* * * * * * *	******				$\begin{array}{c} 2 $
8	CONCLUSION	A 4 4	× 4		1	k a	×	s.		с. х	÷	* *		, I		4		2.8-1
AP	PPENDIX A DEFINITIO	NS AND	ACRO	NYNC	15	4 A		×.				4		÷			a)	2.A-1
C	CHAPTER 3, "REACTOR	COOLAN	T SY	STEM	1 A I	ID R	EA	c to	RI	NON	-SA	FET	Y	AUX	11	AR	Y S	YSTEMS"
1	INTRODUCTION	* * *	* *	1		3 3		7			ł	4 . 4	4	4	•			3.1-1
	<ol> <li>1.1 Review Criter</li> <li>1.2 Scope and Str</li> <li>1.3 Policy Issues</li> <li>1.4 Outstanding I</li> <li>1.5 Vendor- or Ut</li> </ol>	ia nucture issues ility-	of Spec	Chap	oter	3 ems		$\frac{1}{2} + \frac{1}{2} + \frac{1}$			* * * * *				* *			3,1-1 3,1-1 3,1-2 3,1-2 3,1-2 3,1-2
2	REQUIREMENTS COMMO	N TO B	WRs /	AND	PWF	۱s.		i.	ć	. k	÷				ł	ł		3.2-1
	2.1 Definition 2.2 Common Requir 2.3 Conclusion	ements				* * * *						• •		•				3.2-1 3.2-1 3.2-3
3	PWR REACTOR COOLAN	T SYST	EM .	- A 3			÷.	Ŷ				÷ .						3.3-1
	3.1 System Defini 3.2 Performance	ition .	4 4 4 4		 		1		•			•	1	•	•			3.3-1 3.3-1

4	<pre>3.3 System Features</pre>	3.3-2 3.3-4 3.3-6 3.3-6 3.3-7 3.4-1
	<ul> <li>4.1 System Definition .</li> <li>4.2 Performance Requirements .</li> <li>4.3 System Features .</li> <li>4.4 Component Features .</li> <li>4.5 Instrumentation and Control .</li> <li>4.6 Maintenance .</li> <li>4.7 Conclusion .</li> </ul>	3.4-1 3.4-1 3.4-2 2.4-2 3.4-4 3.4-4 3.4-4
5	BWR REACTOR COOLANT SYSTEM (BWR NUCLEAR BOILER SYSTEM)         5.1       System Definition         5.2       Performance Requirements         5.3       System Features         5.4       Component Features         5.5       Conclusion	3.5-1 3.5-1 3.5-2 3.5-2 3.5-3 3.5-4
6	CHEMICAL AND VOLUME CONTROL SYSTEM (PWR AUXILIARY)	3.6-1
	6.1 System Definition	3.6-1 3.6-1 3.6-1 3.6-1 3.6-1 3.6-2 3.6-2 3.6-2
7	PROCESS SAMPLING SYSTEMS (BWR AND PWR)	3.7-1
	7.1System Definition7.2Performance Requirements7.3System Features7.4Component Features7.5Instrumentation and Control7.6Maintenance7.7Conclusion	$\begin{array}{c} 3  .  7 - 1 \\ 3  .  7 - 1 \\ 3  .  7 - 1 \\ 3  .  7 - 1 \\ 3  .  7 - 2 \\ 3  .  7 - 2 \\ 3  .  7 - 2 \\ 3  .  7 - 2 \end{array}$
8	REACTOR WATER CLEANUP SYSTEM (BWR AUXILIARY)	3.8-1
	<pre>8.1 System Definition</pre>	3.8-1 3.8-1 3.8-1 3.8-1 3.8-1

# 8.5 Instrumentation and Control 3.8-1 8.6 Conclusion 3.8-2 9 CONCLUSION 3.9-1 APPENDIX A DEFINITIONS AND ACRONYMS. 3.4-1

#### CHAPTER 4, "REACTOR SYSTEMS"

1	INTR	ODUCTION	x 3		÷	×.			,		4		,	4.1-1
	1.1 1.2 1.3 1.4 1.5	Review Criteria Scope and Structure of Chapter 4 Policy Issues Outstanding Issues Vendor- or Utility-Specific Items	, , , , , , , ,		* * * *	* * * *	* * *			* * * *	* * * * *	* * * * *	* * * * *	$\begin{array}{c} 4  ,  1 - 1 \\ 4  ,  1 - 1 \\ 4  ,  1 - 2 \\ 4  ,  1 - 2 \\ 4  ,  1 - 3 \end{array}$
2	REQU	IREMENTS COMMON TO BWRs AND PWRs	a - 4		÷.			ć i	1	×	×		ĸ	4.2-1
	2.1 2.2 2.3	Top-Level Requirements Performance Requirements Equipment Design Requirements	4 A 4 A		* *	* * *	4			* * *	* * *	* *		4.2-1 4.2-1 4.2-2
		2.3.1 Materials 2.3.2 Vessel and Internals Design		•	÷		a		i. X	e e	4 4	*	+ + +	4.2-2 4.2-3
3	BWR	REACTOR PRESSURE VESSEL AND INTERNALS			×	÷	i. K		,		÷	4		4.3-1
	3.1 3.2 3.3	Definition	к., к ң. ң. ң. ң. қ.							1.1.1				4.3-1 4.3-1 4.3-1
4	BWR	CORE AND FUEL	с. 4. д		×,	,	+ -		i,	i,	i.		,	4.4-1
	4.1 4.2 4.3	Definition	* *	*	н 14 14				. 4				1.8.8	4.4-1 4.4-1 4.4-2
5	BWR	CONTROL ROD DRIVE SYSTEM	x x		×.	į.	į.			÷		4		4.5-1
	5.1 5.2 5.3	Definition		*	1.1.1		•		*		* *		* * *	4.5-1 4.5-1 4.5-2
6	PWR	REACTOR PRESSURE VESSEL AND INTERNALS	i.		i.	ć	á							4.6-1
	6.1	Definition				*		4	4					4.6-1 4.6-1
EPI	RIEV	olutionary Plant SER xvii												

and sound

Page

Sin'

44

	6.3 Equipment Design Requirements			÷	,	4	×	÷	ł.		x	÷		e.	5.	ŝ	4.6-2
7	PWR CORE AND FUEL		÷.	ł,	k	i.			÷	x		×	ж	÷	4	s.	4.7-1
	<ul><li>7.1 Definition</li><li>7.2 Performance Requirements</li><li>7.3 Component Design Requirements</li></ul>	 н н ң	4 4 8			* * *	4	н ж	* *	4 4 4				4 4 4	* * *	4 4 4	4.7-1 4.7-1 4.7-2
8	PWR CONTROL ROD DRIVE SYSTEM	4		÷	,	÷	ī		÷	÷	×					ł.	4.8-1
	8.1 Definition	 * * *		3. × *	* * *				* * *	* * *	*	, , ,	* * *	* * *	* * *	* * *	4.8-1 4.8-1 4.8-2
9	CONCLUSION	 5	×		÷.	4			÷	×		÷				×	4.9-1
A	PPENDIX A DEFINITIONS AND ACRONYMS	3	4					×	.,				,i	.4			4.A-1

#### CHAPTER 5, "ENGINEERED SAFETY SYSTEMS"

1	NTRODUCTION	1	
	.1 Review Criteria	1 2 2 4	
2	OP-LEVEL REQUIREMENTS COMMON TO BWRs AND PWRs	-1	
	2.1ALWR Public Safety Goal5.22.2Station Blackout5.22.3Zirconium-Water Reaction and Hydrogen Generation5.22.4Decay Heat Calculations (American Nuclear Society (ANS) 5.1)5.22.5Fire Protection5.22.6Severe-Accident Analyses5.22.7Source-Term Issues5.2	-1-5-6-7-8-8	
3	ALWR CORE DAMAGE PREVENTION REQUIREMENTS	-1	
	3.1         Inservice Testing         5.3           3.2         Diesel Generator Start Time         5.3           3.3         Electric Valve Operators         5.3	-1 -2 -3	
4	BWR CORE DAMAGE PREVENTION REQUIREMENTS	-1	
	4.1 Elimination of Core Spray5.44.2 Anticipated Transients Without Scram5.4	-1	

j

•

.

Page

	4.3 4.4 4.5 4.6	Standby Safety Suppres Suppres	Liqu Class sion sion	id Co ifica Pool Pool	ntro tion Bypa Temp	l Sys of C ss Le erati	stem Conta Sakag ure-M	inπ je toni	ient tor	He	at S	R ys	emo	vva n	i	, Sy	st	, em ,	4 4 4		*	5555	4-1 4-2 4-3 4-4
5	PWR (	CORE DAM	IAGE P	REVEN	TION	REQU	JIREN	1EN7	ſS			×		3				,				5.	5-1
	5.1 5.2 5.3 5.4 5.5	Introdu Residua Emergen Safety Safety	iction 1 Hea cy Fe Injec Depre	t Rem edwat tion ssuri	oval er S Syst zati	Syst yster em . on ai	tem n . nd Ve	ent	Sy	ster			* * * * *	2 2 2 2 2	4 4 4 4	4. 4. 3. 4. 4.	* * * *	* 5 * 4			*	5.5.5	5-1 5-1 5-4 5-6
6	MITI	GATION R	EQUIR	EMENT	s.	x. +	÷ *		к. л.	÷			×		, k	į.	į.	÷	4		4	5.	6-1
	6.1 6.2 6.3	Introdu Contair Contair	action nment nment	Isola Leaka	ition age R	Sys ate	tem I Test	, Des ing	ign		 		1 1 1	4 4 4	1. 1. 1.	а 4 4	4. 4. 3.		4 4	1 1 1		555	6-1 6-1 6-3
		6.3.1 6.3.2 6.3.3	Conta Type Type	B Ain C Con	nt In r-Loc ntair	itegr k Te ment	ated sts Loc	Le al	ak Lea	Rat k R	e ati	fes e 1	t es	t.s		* * *	3 - 3 - 4		у э. х.	x . 9 3		555	6-4 6-5 6-6
	6.4 6.5	rission Combust	n Prod tible	duct I Gas d	Leaka Contr	ige C	ontr	0]	ж. н ж. н	*	 				k K	14 14	1	к	2	*	¥ k	55	6-7 6-7
		6.5.1 6.5.2 6.5.3 6.5.4	Meta Radii Iner Sove	l-Wati olyti ting/ re-Ac	er Re c Hyd Ignif cider	eacti iroge ters nt Ec	on a n Ge uipm	nd ner ient	Hyd ati Re	rog on qui	ien	C (	nt	en	tri	at	i o 1	ń				5555	.6-7 .6-9 .6-10 .6-11
	6.6	Severe	-Acci	dent	Requ	iremé	ents					e - 1	di.			×			.,	ų,		5	.6-11
		6.6.1	Cont	ainme tv/Pe	nt M. dest	argir al-D	l Vwol	1 0	ont	101	ira	÷i.	n		leb	-1		ļ	4		į.	5	.6-11
		6.6.3 6.6.4 6.6.5 6.6.6 6.6.7 6.6.8 6.6.9	Cool Cont Fiss RCS Equi Cont Seve Exte	abili ainme ion P Depre pment ainme re-Ac	ty nt H rodu ssur Sur nt M cide y In	eat I ct Co izat vival ixin nt M itia	Remove ontro ion ( bilit g Pro anage ted !	val Sapa ty ovi: emer Seve	abi sion at	lit: Ac	cid	len	ts									555555555	.6-12 .6-14 .6-15 .6-15 .6-16 .6-17 .6-17 .6-17
7	BWR	MITIGAT	ION/C	ONTAI	NMEN	T RE	QUIR	EMEI	NTS								1,				Ϊ,	5	. 7-1
	7.1 7.2 7.3	Introd Perfor Equipm	iuctic mance nent [	n . Requ Jesign	uirem n Req	ients Juire	ment	s .	3		1 + + +	* *						- 9 (- 9					5.7-1 5.7-1 5.7-3

.

Page

.....

8	PWR M	ITIC	ATION/C	ONTAI	NMENT	REQ	UIR	EMEN	IT S			ł					÷				ć		k.	5.8-1
	8.1	Prin	pary Con	tainm	ent .					,		÷								4	κ.	4	ć.	5.8-1
	8.2	Cont	tainment sion Pro	Spra duct	y Sys Remov	stem val a	ind	cont	re	i	ŝy	st	em	í.	4 ° 4	*	4 4	к.	•	4 	а. Д	*	s L	5.8-5
9	CONCL	USIC	ON									,	į.	,		į			4					5.9-1
AP	PENDI	( A	DEFINIT	IONS	AND /	ACRO	YMS						÷	į.	÷	÷	÷		,		÷	,	4	5.A-1
AP	PENDI	K B	GENERIC	SAFE	TY A	ND L		SIN	3	ISS	UE	s		ļ									×	5.B-1
AP	PENDI	A C	OPTIMI7	ATION	SUB	JECT	s .	x A			4		,		i.				÷			А,		5.C-1
AP	PENDI	x D	REGULAT	FORY D	EPAR	TURE	ANA	LYS	IS	÷	,			į,									į.	5.D-1

# CHAPTER 6, "BUILDING DESIGN AND ARRANGEMENT"

1	INTE	ODUCTION	1.1.1	4 4 4	4. 4		ę.	× 4.	×	× .)	÷	ę. j		ł	ж.		i.	ĸ.		÷,	8		6.1-1
	1.1 1.2 1.3 1.4 1.5	Review Scope a Policy Outstan Vendor-	Criter nd Str Issues ding I or Ut	ia ucture ssues ility-	of	Cha	pto c	er ( Iter	'ns	专业方法于	* * *	*	 		* * *	* * * * *	大 天 天 子	法法法法			* 2 * 4		$\begin{array}{c} 6  ,  1 - 1 \\ 6  ,  1 - 1 \\ 6  ,  1 - 2 \\ 6  ,  1 - 2 \\ 6  ,  1 - 3 \end{array}$
2	KEY	REQUIREM	IENTS .		×	e., +	÷	e. 4	÷		÷	×.	,		į,	k	х.,	Ì.	*	×	4	×,	6.2-1
	2.1 2.2 2.3 2.4	General Design Design Design	Desig for Co for Sa for Op	gn Requ onstruc afety . peratio	iren tion n a	ment n , nd I	is Mai	and  nte	Po	ili	cy		ta	ter	nen	ts ·		4 4 4 4	1	K.14. K.4			6.2-1 6.2-5 6.2-6 6.2-14
3	OVE	RALL SITE	E ARRAI	NGEMENT	Ş	* - x		і. ж. ж	А	ź.		ŝ.		ł,	, i	*	÷	×	×	к.	÷.	4	6.3-1
	3.1 3.2 3.3	Introdu Interfa Require	uction aces ements			+ + + +	3 2 3	1 8 8 8 8 8 8	-		*		4 4 4	4 4	* *			4.4		1.00			$     \begin{array}{r}       6.3-1 \\       6.3-1 \\       6.3-1 \\       6.3-1     \end{array} $
		3.3.1 3.3.2 3.3.3 3.3.4 3.3.5 3.3.6 3.3.7 3.3.8 3.3.8 3.3.8	Site Site Exter Servi Repai Utili Misce Const Trans	Drainag Layout nal Int ce Fac r Shop: ty Rou llaneo ructio portat	ge terf ilit s . ting us S n Fa	ace ies ite ite	s F it	acil	it	ie s	S												6.3-1 6.3-2 6.3-3 6.3-3 6.3-4 6.3-4 6.3-5 6.3-5 6.3-5

Page

	3.3.10	Tanks			х.		4		ł.	×	ł	A.		4		•		. )			× 1		$\sim$	+	6.3	-5	
POWER	GENERA	TION	COMPL	EX		2.						ε.	4			Â.	( )							k.	6.4	1-1	
4.1	introdu	rtin																							6.7	4-1	
4.2	Common	Requ	remen	nts				* *		ł	-		÷.	1	1	4				e.	×.	1		÷	6.4	4-1	
	4 2 1	Over	11 Å	erai	nna	mer	.+																		6	4-1	
	4.2.2	Build	lina	and	Ea	uit	ma	nt	Âr	re	ino	em	en	it	1	<u>.</u>	1					2		2	6.	4-2	
	4.2.3	Systi	em Su	ppo	rts							1									1		į.	*	6.	4-2	
	4.2.4	Pipi	ng Ar	ran	gem	ient	1			÷	κ.			4	8	ć.			i.	÷	1	ă.	6	ŝ.	6,	44	
	4.2.5	Heat	ing, :	Ven	til	at	ion	12	ind	1 4	lir	C	on	idi	ti	on	ing	3	(H)	VA	C)		÷.	۰.''	6.	4-6	
	4.2.6	Flec	trica	I A	rra	nge	me	nt	1	1	2	1.	1		÷.,	9	6	÷.,	e l	11	£	2	×.	4.1	6.	4-6	
	4.2.1	Inse	rvice	in	spe	ct	101	0	ons	110	ler	at	10	ns		÷	4	6	×.	ł	Υ.	х.	8	۰.,	0.	4-1 A 0	
	4.6.0	Ragi Doci	ation an of	20	nes	Day	10 4 4 4	50	161	0	ing		1	4	1	1	*	8.	٨.	×	х.,	*	×	×.	6	4-0	
	4 2 10	Cont	amina	tin	gn-	nai	i na	1	20	151	23	1.22	1		×.,	8	×	5	×	4	8		4	*	6	4-1	0
	4.2.11	Stru	ctura	ip	4551	on		1	21	٩.	2	Č.	1		÷.	1.	1			8. I	÷.,	2	1	1	6.	4-1	õ
	4.2.12	Cons	truct	ion	Re	au	ire	mei	nti		1	Ċ.	Ĵ.	1		1	2	2		÷.	P	1.	2		6.	4-1	3
	4.2.13	Cran	e Pat	h R	out	in	η.				į.	2							÷		ġ.		ġ.		6.	4-1	4
4.3	Primary	y Con	tainm	ent	St	ru	ctu	ire		1		1				j.	Ì				1			1	6.	4-1	4
	4.3.1	Defi	nitio	n,	à.		÷ 1	с. н	$\cdot$	×	÷	×.	×.	1.0	×.	÷	$_{A} \in$	÷ .	÷		÷		÷.	$\mathbf{x}^{+}$	6.	4-1	4
	4.3.2	Comm	on Re	qui	ren	nen	ts		. ÷.	3	$^{+}$	÷	×	÷	$\mathcal{A}_{\mathcal{A}}$	4	κ.,		κ.	÷.	÷	4	×.		6.	4-1	4
	4.3.3	BWR	Prima	ry	Cor	nta	inn	nen	t	St	1"11	cti	iri	e	×	×.	х.	۰.	£.	3	×	×	×		6.	4-1	17
	4.3.4	PWK	rrima	ry	001	nta	1.00	nen	t i	st	ru	C L I	in	e .	÷.	×,	٩.	х.	*	*	1	×,	*	×.	0.	4-1	3
4.4	BWR Re	actor	Buil	din	ig i	and	P	R	Au	хi	11	ar	V	Bu	110	lin	ġ	ł.		ł,	Į,	÷	¢.	÷	б.	4-2	21
	4.4.1	Intr	nduct	ion																					6	4-1	21
	4.4.2	BWR	React	or	Bu	i î d	in			1	÷,	ŝ.	1	1	ű.	č.	11	1	5	ð.	1	1	Ľ.		6.	4-	21
	4.4.3	PWR	Auxil	iar	y I	Bui	1d	ing	÷.	1		4	2	1	÷			È.	÷	÷	1	2	0		6.	4-	22
AE	Turbin	n Cor			er. Sandi	1.4.4																					
415	TUPDIN	6-061	eracc	11 1	sui	101	ng	- 1	2	4	×	1	18	1	1	1	2	*	۴.	1	1	1	×.	*	0.	4-1	23
	4.5.1	Defi	nitic	in a	and	Sc	op	e .	14	١.,	1	1	1							1		÷	5		6	4-	23
	4.5.2	Gene	eral f	legi	ir	eme	nt	s .		÷,	19	1	1	1		ŝ.	2	1	Ĵ.	Ū.	1				6	4-	23
	4.5.3	Equi	pment	L AI	rra	nge	me	nt	1		1	-2	Ε,	1	÷	1		1	÷.	÷	÷,	÷Q.	÷.	1	6	4	24
	4.5.4	BWR	Turb	ine	-Ge	ner	at	or	Bu	11	di	ng	1	n é			14	à.	÷.		×.		ŝ.		6	4-	24
	4.5.5	Cond	lurio	on .	e	×	х.	τ. 3	14	1	1	4		e s	1	ц,	×.	κ.	÷	÷	÷.	5	5		6	. 4 -	25
4.6	Other	Power	Gen	erat	tio	n (	om	ple	X	Fa	ci	11	ti	es	í,	à	i	i.	÷		ł.				6	. 4-	25
	1.2.2	D																									
	4.6.1	Def	initio	0n	• •	1	1	1	1			3	15	1	à	×	×		÷	×	×	X	×		6	.4-	25
	4.0.2	Pue	nan	0111 C.	ng	and	1 2	CON	ag	e	t a	C1	11	ty		. *	х	14	.*	.4	1				0	.4-	25
	4.0.3	Fradi	enste	rai	011	11.)	p.	* *					- 1		13	. 1		*		18	×	1	÷		0	. 4-	20
	4.0.4	Con	genc,	Com	1151	re	PO	wei	3	up	p l	У	r d	101	11	cy	1	*	1						0		27
	4.6.6	Toc	nica	1 0	unn	A	1	1. 1	100	. 1		. *	- 9			. 4	4			1					0	. 4-	20
	41010	100	100110-0	1 3	whh	011	6 6	CILI	1.01		1		- 18			1.1		4		. 6			. 1	-6	0	1 4	63

EPRI Evolutionary Plant SER xxi

0

١

-

-

5 CONCLU	USIC	N .	5. 8	1.1	i = i	÷ + +		4 9	4.1.19	×		< .	i s	+	×.	ξ.	8	4	×	χ.	6.5-1
APPENDIX	A	DEFI	(NIT)	ONS	AND	ACRON	IYMS	4 - 3			×.				×	,		×	÷	*	6.A-1
APPENDIX	В	GENE	RIC	SAFE	TYA	ND LI	CENS	ING	ISSU	ES		a.		à.			×	×	×		6.B-1
APPENDIX	C	BWR	CONS	STRUC	TION	PLAN	AND	ARR	ANGE	MEN	(Ť	SKI	ETC	HE S	1			4	e,	i.	6.C-1
APPENDIX	D	PHR	CONS	STRUC	TION	PLAN	AND	ARR	ANGE	MEN	IT	SKI	ETC	HES	1	4		×.	4	1	6.D-1
APPENDIX	Ε	MAII	TAI	ABIL	ITY.	EVALU	ATIO	ν.	÷.,			1							Å		6.E-1

# CHAPTER 7, "FUELING AND REFUELING SYSTEMS"

1	INTRO	DUCTIO	N	1 i 1	1.1			÷	÷	•			1.0	÷	×.	÷	÷.	×.	d.	4	÷	7.1-1
	1.1 1.2 1.3 1.4 1.5	Review Scope Policy Outstan Vendor	Criter and Str Issues nding I - or Ut	ia ucture ssues ility-	of C Speci	hap fic	ter İte	7 ms	* * * * *	X X X X	* * *				* * * * *	* * * * *		* * * * *	* * * * *	****	* * * * *	7.1-1 7.1-1 7.1-2 7.1-2 7.1-2 7.1-2
2	GENE	RAL REQ	UIREMEN	TS AND	POLI	CY.	STAT	EME	NT	Ş	÷						a.	ŝ,	d.		х.	7.2-1
3	SPEN	T FUEL	POOL .		x x	а . т.	÷ 4	×		×., (	x ji	i,	Ċ,	ġ	÷,	ŝ	÷		*	÷.	÷.	7.3-1
	3.1 3.2	Functi Evalua	ons and tion .	Key D	esigr	n Re	quir	eme	nt.	s	i K				Ì	i. X	÷.	i X	s' X-	en K	÷	7.3-1 7.3-2
		3.2.1 3.2.2 3.2.3 3.2.4 3.2.5	Qualit Accide Occupa Sabota Conclu	y Grou nt Ana tional ge Pro sion .	p Cla lysis Expo tecti	issi s - osur ion	fica Fuel Agai	tic Ha nst	n ind	li ef	ng ue	Ad	ic i	i dei Poi	nt ol	Dr	ai	ni	ng			7.3-2 7.3-3 7.3-4 7.3-5 7.3-5
4	FUEL	POOL C	OOLING	AND CL	EANUI	p sy	STEM		į	ł					į.		ł		÷		*	7.4-1
	4.1 4.2	Functi Evalua	ons and tion .	Opera	ting	Mod	es.	i,	*		s k				r x	i t		4 4			i.	7.4-1 7.4-2
5	NEW	FUEL ST	ORAGE F	ACILIT	Υ.,	3, 19		ź	k	ł	d.	,			k		ŝ,	÷	ł			7.5-1
6	CASK	RECEIV	ING AND	HANDL	ING	FACI	LITI	ES	×	ł		þ						ķ	÷			7.6-1
	6.1	Heavy	Loads C	onside	ratio	ons	5.	į,	¥,	į.	ł			( ) (	ş	k		Y	ć	ŝ	ł	7.6-1
		6.1.1 6.1.2	Key De Evalua	sign A tion .	equi	reme	nts		1		ţ					1			•			7.6-1 7.6-1

	6.2 6.3 6.4 6.5	Acci Sabo Radi Acci	der tag old der	nt-H je ( ogio nt /	Pre Con Cal Ana	ven sidi Coi lys	tio era nsio is	n S tio der	itra ins iati	iteg	ies	×	ž K K	14 14 14	1 1 1	12 14 15 14	* * *	3 3 7 1	*	3 4 3	4. 4. 4.	й. К. Т.	9 1 1 1		* * * *	* * * *	* *	7. 7. 7. 7.	6-2 6-2 6-3	
7	OTHER	REL	ATE	D S	SYS	TEM	S AI	ND	EQL	IPM	IENT											k	,			-		7.	7-1	
	7.1	Fue1	н	ind	lin	g S;	ysti	em		÷.,													i.	4	,			7,	7-1	
		7.1.	12	Kej Evi	y Di alu	esi ati	an I	Reg	iuir	eme	nts		e e				н к	4 4						4	5 5	a A	а а	7.	7 - 1 7 - 2	1
	7.2 7.3 7.4 7.5	Fuel Fuel Irra Reac	Tr Ha dia tor	nan: and atec n Di	sfe lin d Fi isa	r Sy g Ai uel sser	yste rea In: nbl;	em He spe y a	(P) ati cti ind	(R) ing ion Ser	and and vic	v R in	en ep g	ti ai Eq	ia r ui	ti Ec	on ui	s pm t	ýs en	te t	m ·	"主书书"					* * * *	7. 7. 7. 7.	7-3 7-4 7-5	
8	CONCL	USIC	N.	ć.				c - 4				i.	į.	*	į.	÷	÷			ł	*	4	1	÷	×	×.		7.	8-1	
API	PENDIX	A	DEF	IN	ITI	ONS	AN	) A	CRO	NYM	S		e.	1			i A					÷		a,	à.	4	×	7.	A-1	
API	PENDIX	В	GEN	VER	IC	SAF	ETY	AN	ID I	ICE	NSI	NG	1	SS	UE	S												7.	B-1	

#### CHAPTER 8, "PLANT COOLING WATER SYSTEMS"

1	INTRODUCTION	- 1
	1.1Review Criteria8.11.2Scope and Structure of Chapter 88.11.3Policy Issues8.11.4Outstanding Issues8.11.5Vendor- or Utility-Specific Items8.1	-1 -2 -3
2	SCOPE AND KEY FUNCTIONAL REQUIREMENTS	-1
3	COMMON REQUIREMENTS	-1
	3.1       Introduction       8.3         3.2       System Requirements       8.3         3.3       Equipment Requirements       8.3         3.4       Conclusion       8.3	-1 -3 -5
4	COMPONENT COOLING WATER SYSTEMS	-1
	4.1 Scope and Function	-1 -1 -2

EPRI Evolutionary Plant SER xxiii

\$

Page

5	SERVI	ICE W	ATER	SYST	EMS				ć.	×						,			k.		ŝ			1			8.5-1
	5.1 5.2 5.3	Defi Esse None	nitio ntia sseni	on an 1 Ser tial	d Sc vice Serv	ope Wa	ter Wat	Šy ter	st	en	te	< m	x e e	4 11 14	ж Х Х	*		4. 4. 4.	4 14 14		4. 4. 4.	* * *	ж ж	а ж. қ.	4 4 4		8.5-1 8.5-1 8.5-4
6	CIRC	ULATI	NG W	ATER	SYST	EM		1		,							4			ĸ	×	ч.	x.				8.6-1
7	HEAT	SINK	s .								4			i.	k.	4	л.		÷		×	4			х,	÷	8.7-1
	7.1 7.2 7.3	Defi Norm Ulti	niti Nal P mate	on an ower Heat	nd So Heat t Sin	cope t Si nk	ink		* * *			* * *	* * *	* *	4 4 y	* * *		* * *	*	4 + +	* * *		* * *		3 4 3	2 14 14	8.7-1 8.7-1 8.7-1
8	CHIL	LED W	ATER	SYS	Tuin\$	1		÷				ŝ	4	×.	ł	÷		4			×,		k	i.	×	×	8.8-1
	8.1 8.2	Esse	entia essen	1 Ch tial	ille Chi	d Wa 11e	ater d Wa	S) tei	yst r :	tei Sy	m ste	em.	i. Y	4	ar 9	е. 4	л 1	4		5	1.1		* 3	*	a A	÷ ×	8,8-1 8,8-1
9	FUEL	POOL		LING	AND	CL	EANU	IP (	SY:	ST	EM	4	x	×.	x	,	×		×	×		÷			÷	÷	8.9-1
10	CONC	LUSI	DN .					a.		ŝ	÷	×	×	÷	×			,	.4				×	×	ж	÷	8.10-1
A	PENDI	X A	DEFI	NITI	ONS	AND	ACF	ON	YM	S						×	.,				į						8.A-1
A	PPENDI	XB	GENE	RIC	SAFE	TY	AND	LI	CE	NS	IN	G	15	SU	ES								1				8.B-1

# CHAPTER 9, "SITE SUPPORT SYSTEMS"

1	INTRO	DUCTION	i e e		×	a. 4	$\tilde{c}_{i,2}$	× . 4	.*	÷	÷	ж÷.,	ĸ,	8. c	i. d		÷	X	×	ł,	9.1-1
	1.1 1.2 1.3 1.4 1.5	Review C Scope an Policy I Outstand Vendor-	riter d Stru ssues ing I or Ut	ia . uctur ssues ility	re of s	Chi	apte	r 9 tems	1 1 1 1 1 1	* * * * *	* * * * *	大 子 子									$9.1-1 \\ 9.1-1 \\ 9.1-2 \\ 9.1-2 \\ 9.3$
2	POLI	CY STATEM	ENTS	AND H	KEY F	REQU	IREM	ENTS		j,		÷,	×	÷	, i				k	÷	9.2-1
	2.1	Policy S Site Sup	itatem port	ents Syste	ems -	Co	mmon	Req	uii	ren	ner	its		ł	4 4	Ċ					9.2-1 9.2-1
		2.2.1 2.2.2 2.2.3 2.2.4	)perab System Safety Radiat	ilit Int -Rel ion	y and eract ated Expo	d Ma tion Pla sure	inta s nt E	inab quip	il me	ity nt	/ 			 	* * *	*					9.2-2 9.2-2 9.2-2 9.2-2 9.2-2 9.2-3

Page

3	FIRE	PROTECT	ION SYSTEM	MS .	× ×	έ,	s - 1	6 - 3	×	×	×.	ŝ	ł,	э.	d.	×	÷	ŝ.	×		×	e.	9.3-1
	3.1 3.2	System Perform	Definition nance Requ	n iremen	nts	х х	ж. ) 8. 3		x x	4	а х	i X X	A M	i K	*	e e	ж. Ж	4	*	•	1. 1	a je	9.3-3 9.3-3
		3.2.1 3.2.2	General Fire Hazan	rd Ana	iysi	ŝ			, ,	4.1	*	x x	ł	÷	ă X	A K	é K	к. К.	× ×	*	× +	ж. 9	9.3-3 9.3-4
	3.3	System	Features				χ.,	i i	×			4	÷	÷	*	κ.		÷				4	9.3-4
		3.3.1 3.3.2 3.3.3	Protection Component Extended H	n of R Repla Fire P	Redur Iceme Prote	nda ent ect	nt	Sa n C	fei	ty era	Di	ivi ,	51	on	\$	14-4-14	4 9 9	ж. ж.	4. 1. 1.	* *			9.3-4 9.3-5 9.3-6
	3.4	Compone	ent Feature	es .	8. 3	÷.	ж. 3		×	ĵ		į.	i.	*	ć	ie.	÷	÷.	ź	x	ł	i.	9.3-6
		3.4.1 3.4.2 3.4.3 3.4.4 3.4.5 3.4.6 3.4.7 3.4.8 3.4.7 3.4.8 3.4.9 3.4.10 3.4.11 3.4.12 3.4.13	Preaction Fire Pumps Water Supp Pressure M Halon and Cable Tray Portable I Fire Detec Control R Corrosive Human Fact Diesel Gen Seismical	Sprin s Mainte Carbo y Fire Exting ctors ctors fire tors herato ly Sen	nkler nanc n Di s uish Ager naiti	ier ier its ive	Pur ide s	np 9	ys										""""""""""""""""""""""""""""""		"""""""""""""""""""""""""""""""		9.3-6 9.3-7 9.3-7 9.3-7 9.3-8 9.3-8 9.3-8 9.3-8 9.3-8 9.3-9 9.3-9 9.3-9 9.3-9 9.3-10
4	ENVI	RONMENT	AL MONITOR	ING SY	STEN	1	¥ 3	÷.,	s,	÷	×,	è,	3	÷,	ł,	8	ŕ	ŝ,	÷		ł		9.4-1
	4.1 4.2	Scope a Perform	and Function nance Require	ons . iremen	nts Data				*					i.	*	•					* *		9.4-1 9.4-1
		4.2.2	Water Qua	lity [	)ata	1		н., н. 1, н. К.	÷.	ľ.	Ċ,	ļ	1	ŝ	•	Č.	i.		4. 4.	ł.	× *	4. 4	9.4-1
	4.3	System	Features			÷.		e ie	×	1	,	ŝ	į.		÷	į.	k.		ł,	i.	÷	i.	9.4-2
		4.3.1 4.3.2 4.3.3 4.3.4 4.3.5 4.3.6	Primary Me Supplement Siting . Control Le Weather Po System Poo	eteoro tary T ocatic rotect wer Sc	ologi lower ons tion burce	ica ·s	1	Tow	er														9.4-2 9.4-2 9.4-2 9.4-2 9.4-2 9.4-2 9.4-2
	4.4	Instru	nentation a	and Co	ontro	1		, is			i.		÷	•		4	*		•		•		9.4-2
		4.4.1	General R	equire	ement	s	1	1		*	4				ł.		•			•		*	9.4-2

#### EPRI Evolutionary Plant SER

		4.4.2	Gener	ic 1	nst	rum	ent	at	ior	ε.	κ.		k i	÷.,		×.		i.	x,	a.	i.	÷	a.	×	9.4-	3
		4.4.3	Instr	umen	itat	ion	Do	cu	men	ita	ti	on		6. 3	×	16	1	÷.	×		×.	÷.	4	х.	9.4-	3
		4.4.4	Testi	ng a	ind	Cal	ibr	at	ior	l	÷	*	8	с., к		*		Ŧ	×	4	×	÷.	3	н.	9.4-	8
		4.4.5	Instr	umen	itat	ion	Ra	ing	es	č.	×	۰.	4	1	1	1	*	٠	Χ.	*		×	1	÷.,	9.4	5
	4.5	Data Ma	nagem	ent									, i.,	1.		1			1						9.4-1	3
	4.6	Conclus	ion	÷			4		4	4	×				4	*	à.		÷	*	*	÷		a.	9.4-3	3
5	SITE	SECURIT	Y SYS	TEM				i,			į.							į.				÷.	÷	,	9.5-	1
	5.1	Scopp	nd Eu	net i	nne																				9.5-	1
	5.2	Perform	ance	Requ	ire	, men	ts	a a	i.	т. К	÷	i.	8 - 1 8 - 1			*	л. Д	4	* *-	e K	÷.	5	×.	÷,	9.5-1	2
		5.2.1	Prote	ctio	n S	tra	teo	ie	s	Ľ.															9.5-3	2
		5.2.2	Prote	ctio	in M	eth	odo	100	av	1	<u>.</u>	2	2.1		1	÷.	1.	1	ŝ.	2		ŝ.,	÷.	1	9.5-3	3
		5.2.3	Vital	Equ	ipm	ent	Ev	ali	uat	10	n				1	÷.	÷.	1	ł.	1	÷.	÷.,	21	i.	9.5-3	3
		5.2.0	Vital	Con	ipon	ent	La	YO	ut	5	η.					÷.	2					ú.		2	9.5-1	\$
		5.2.5	Physi	cal.	Pro	tec	tio	n I	Mea	su	re	s				2			Ĵ.	÷.			ŝ.		9.5-4	¢ -
		5.2.6	Strat	eqy	for	In	ope	ral	ble	V	it	al	Ed	iui	pm	ent		1	į.	÷.	1	2	1		9.5-4	4
		5.2	Prote	cted	Ar	eas	an	d	Bou	ind	ar	ie	s .					j.	£.,	21				÷.,	9.5-1	5
		5 .	Desig	n Ma	rai	ns							Ξ.		1.4	÷.			2	÷.					9.5-!	5
		5.2.9	Train	ing	Fac	111	tie	S				1					2	1	÷	4				1	9.5-6	5
		5.2.10	Acces	s Co	ntr	01	1.		÷.					1			1	1	÷.		91	÷.,	<u>.</u>	2	9.5-6	5
		5.2.11	Commu	nica	tio	ns	22		1		÷.					1	<u>.</u>	Q.,	1	2	1	4	1	21	9.5-6	5
		5.2.13	Power	Sou	irce		11									÷.	÷.					1			9.5-6	5
		5.2.13	Data	Mana	gem	ent	3	×	÷			4				×	x.	4	Ŷ,	S.	4	÷	×.		9.5-1	1
	5.3	Conclus	ion	y						,	÷		. ,		4	*	į.		į	ł,	×	÷.	,		9.5-3	7
6	DECO	NTAMINAT	ION S	YSTE	M (	FAC	ILI	TI	ES)		,			. *	÷	÷.		,	÷.	ć		÷	÷	į.	9.6-1	1
7	COMP	RESSED A	IR AN	ID GA	s s	YST	EMS		í.	8	e.					i.	ŝ,	ŝ,	÷	Į.	i.	į.	i,	į.	9.7-1	1
	7.1	Company	and A	1.00																					0 7 1	
	7 2	Compres	sed A	11 3	yst wet	em	ς÷.«	1.1	18	1	*	* :	5.13		÷.	. *	1	2	1	٩.	1	۲.	1		9.7-1	5
	1+6	compres	sed u	d5 3	yst	em	4	. *	1		÷.,	×.	1	6.8	1	1	1		*	1	٩.	1	٩,	1	9.1-1	
8	HEAT	ING, VEN	TILAT	ING,	AN	DA	IR	CO	NDI	TI	ON	IN	G S	SYS	TE	М	•	÷	÷		4	i.	ł.		9.8-1	1
	8.1	Scope a	nd Eu	ncti	ons																				0 8-1	í.
	8.2	Convision	Perfo	rman	ce	Req	uir	em	ent	s				H	1	4		ł	į.	1	į.	Ì.		į.	9.8-1	i
		111	100					Ú.P																		
		8.2.1	Gener	al R	lequ	ire	men	ts	1	٠.	4				- 4			÷.	۹.	۰.	4	*	+	*	9.8-1	1
		8.2.2	contr	010	omp	lex	HV	AC	23	st	em	S				. *	۰.	۰.	4		*	*		τ.	9.8-1	
		8.2.3	Unsit	e St	and	DY	AC	POI	wer	S	up	D1	y ł	ac	11	ity	1		+	+	á.	۰.	*	*	9.8-8	5
		8.2.4	Secur	ity	Bui	Idi	ng	. 4	1	:	*	1					*		.*	3	+	4	4	ж	9.8-9	2
		8.2.5	Fuel	Hand	llin	g/S	pen	ţ	Fue	1	Po	01	Ar	°e a	He	eat	11	9	ar	d						
			Venti	lati	ng	Sys	tem	1.6		5	+	.e					+	8	*		4	٩.,	٩.	5	9.8-1	10
		8.2.6	HVAC	Syst	ems	fo	rM	150	cel	la	ne	ou	s f	re	as					1	4.				9.8-1	11

-

	8.3		Pe	r	fo	rt	na	nc	e	R	eq	u	ir	eπ	er	nt	s	an	d	I	nt	er	fa	çe	ŝ	fo	r	٩W	Rs		×	ý.			,	×	9.	8-	12	
			8.	32	.1		I	nt	n	bd	uc	t	10	n		×					, V			i.,	1.	-					*	,		×.	×	×	9.	8-	12	
			8	3	.3		B	WR	1	2e	a. ac	t	or	B	u	11	di	na	10	Ve	'nť	11	at	10	n	SV	st	em	2 X	23	CH.	5	÷.,	1	1	ĵ.	9	8-	12	
			8	.3	.4		B	WR	-	St	ar	d	by	G	a	s	Tr	ea	t	me	nt	S	y s	te	m	1	4	×		κ.	į.	14	ŝ.	÷.		÷.	9	8-	13	
	8.4		Pi	er	fo	r	na	nc	e	R	ē¢	Įu	ir	en	iei	nt	s	ar	d	1	nt	er	fa	ce	s	fo	r	PW	Rs		ŝ	4	ł.	,	į.	*	9	8-	14	
			8	. 4	.1		I	nt	r	od	uc	a,	10	'n						i.				į.							÷			į,	×	1	9	8-	14	
			8	. 4	.2		2	WR		00	nt	a	ir	me	n	t	0.0	01	ļ	ng	8	nd		len	it	114	iti	on	S	ÿ i	ite	111:	÷	7		÷	9	.8-	14	
			5	. 4	- 3		A		U.	111	5	8	u1	10	11	ng	11	67	11	11	at	10	n.,	Sy	151	ten	1	1.	1	8	4	1	÷	÷	÷	×	9.0	8.	17	
			0	1	1.4		1	NIN.		nu.	×.	1	1.9	ir y		bu	1.1	a :	(III)	9	8.6	nı		al	-33	211	2)	51	en	ł.	Å.	η.	×.	*	2	5	2	. Q.ª	10	
)	LA:	R	A	10	RI	E	Ş	÷	i.	÷		į.	y.	х.		÷				*	+	+	÷	*	ŝ,	÷	3	×	×	×	ł	÷	×	×.	÷.	×.	9	. 9.	1	
0	CON	CI.	U	\$ I	ON	ł	ł.,		3					÷					h.,		4		ŝ.	÷		*	÷	4		×.		i.			į,	×,	9	. 10	)-1	
ĄP	PEND	11		Ă	.,	ć			×				,								÷		į,		÷	a.					÷.		÷		×.		9	. A.	1	
4p	PEND	11		R																																	0	8	1	
100				200								2		1		- A	- 3			18	10.		10		1.			- 6	18		1.0	1.4						1.50		

# CHAPTER 10, "MAN-MACHINE INTERFACE SYSTEMS"

1	INTR	ODUCTION	÷	÷	s, s	1	4	÷	e,	3	10.1-1
	1.1 1.2 1.3 1.4 1.5	Review Criteria Scope and Structu, of Chapter 10 Policy Issues Outstanding Issues Vendor- or Utility-Specific Items	1.4.4	* * * *	( ) ) ( ) ) ) ( ) ) ) (	* * * *	Contraction of the second		x 7 4 7 8		10.1-1 10.1-1 10.1-2 10.1-2 10.1-7
2	SCOP	E, OBJECTIVES, AND POLICY STATEMENTS		÷			į,	ŝ	į,	ł,	10.2-1
	2.1 2.2 2.3 2.4	Function and Scope				1. N. N. N. A.			* * * *		10.2-1 10.2-2 10.2-3 10.2-9
3	KEY	REQUIREMENTS		į,	ć.	à	ş	1	÷	ŝ	10.3-1
	3.1	M-MIS Design Process lequirements						i,		4	10.3-1
		<ul> <li>3.1.1 Overall Design Process Requirements</li> <li>3.1.2 M-MIS Design Organization and Plan</li> <li>3.1.3 Required Design Process Features</li> <li>3.1.4 Independent Review of Design Process</li> </ul>									10.3-1 10.3-2 10.3-4 10.3-8

EPRI Evolutionary Plant SER xxvii

-

8 · UA . 10

		rage
3.2	Proven Technology	10.3-9
	3.2.1 Criteria for Proven Technology	10.3-9 10.3-10
3.3 3.4	Cost	10.3 - 10 10.3 - 10
	3.4.1 Operator Workload 3.4.2 Operator Vigilance 3.4.3 Selection of Automatic or Manual Control 3.4.4 Selection of Remote or Local Control 3.4.5 Gperator Aids	10.3-10 10.3-10 10.3-11 10.3-11 10.3-11 10.3-11
3.5	Availability and Reliability	10.3-12
	3.5.1 Effects of Postulated M-MIS Failures	10.3-12 10.3-13 10.3-13 10.3-15
3.6	Testability Reovirements	10.3-16
	3.6.1 Continuous On-Line Testing	$\begin{array}{c} 10.3-16\\ 10.3-17\\ 10.3-17\\ 10.3-17\\ 10.3-18\\ 10.3-18\\ 10.3-18\\ 10.3-18\\ 10.3-18\\ 10.3-18\\ 10.3-19\\ 10.3-19\\ 10.3-19\\ 10.3-19\\ 10.3-19\\ 10.3-20\\ 10.3-20\\ 10.3-20\\ 10.3-20\\ \end{array}$
3.7	Maintainability	10.3-20
	3.7.1Maintenance Burden3.7.2Replacement of Equipment3.7.3Modular Replacement3.7.4Time To Detect and Repair a Failure3.7.5On-Line Calibration3.7.6On-Line Maintenance and Repair3.7.7Maintenance Human Factors	10.3-20 10.3-20 10.3-20 10.3-20 10.3-21 10.3-21 10.3-21 10.3-21

١

•

	3.8	Constru	ictibility .		÷ +			×	, ,				*				10.3-21
		3.8.1 3.8.2 3.8.3 3.8.4	Use of Proven Minimization Standardized Schedule for	Techni of Fiel Designs Constru	ques d Upe for iction	rati Cons	ons tru	cti	on	*	4 4 4 4 4 4			8 4 8			10.3-22 10.3-22 17.3-22 10.3-22
	3.9 3.10	Design Conclu	Flexibility sion	× + + + × + + + +	х 1 х х	a a a a	а. а а. а							4			10.3-22 10.3-22
4	CONTR	OL STA	TION REQUIREME	NTS				9						į.			10.4-1
	4.1	Contro	1 Station Desi	gn Proc	cess P	equ.	irem	ent	s								10.4-1
		4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6	Utilization of Control Stati Review of Cor Iteration of Definition of Documentation	f Funct on Conc ceptia functio Design of Fin	tions ceptua Desi ons, T n Prac nal De	and 1 D gns ask tic sig	Tas esig s, °	k.s n nd	De	 siç							10.4-1 10.4-1 10.4-1 10.4-2 10.4-2 10.4-2
	4.2 4.3	Operat Alarms	ing Crew	 		а с 1. л	• •	4	4 		•			•			10.4. 10.4-4
		4.3.1 4.3.2 4.3.3 4.3.4	General Alarn Selection of Alarm Proces Alarm Presen	n System Alarm sing . tation	m Requ Condit	ire ion	ment	s			* *					* * *	10.4-4 10.4-5 10.4-5 10.4-6
	4.4 4.5 4.6 4.7 4.8 4.9	Displa Contro Voice Arrang Contro Requir	ys ls Communication ement, Enviro l Panels ements for Sp	System nmcnt, ecific	s and Ec	uip	ment tat	lon									10.4-8 10.4-9 10.4-9 10.4-12 10.4-13 10.4-13
		4.9.1 4.9.2 4.9.3 4.9.4	Main Control Local Contro Rradius utd En	Room 1 Stati own Con sponse	ons . trol : Suppo	Stat	ion: aci	s . lit	ies	• •							10.4-13 10.4-15 10.4-16 10.4-18
	4.10	Concl	usion							Ξ.			. 1	í,	1		10.4-18
5	DATA	GATHER	RING, TRANSMIS	SION, A	ND PR	OCES	SIN	G R	EQU	IRE	MEN	ITS				÷	10.5-1
	5.2	Defin Genera	tion 1 Requirement	s	4 - 4 - 4 4 - 4 - 4					• •		• •					10.5-1 10.5-1
		5.2.1	Architecture Design Proce	ss Requ	iret							*					10.5-1 10.5-1

Page

à

4

١.

0

		5.2.3 P 5.2.4 R 5.2.5 S 5.2.6 S 5.2.7 C	erfo ar eliauil ignal Ti tandard ommunica	nce R ity a ransp izati ation	equin nd Av ort I on Pro	remen vaili Delaj toco	nts abil y . ls	ity	* * * *	* * * *	е. 				 			* * *	* * * *	10. 10. 10. 10. 10.	5-2 5-2 5-2 5-3 5-3
	5.3 5.4 5.5 5.6 5.7 5.8	Data Gat Data Tra Signal P Operator Hardware Conclusi	hering   nsmissi rocessi Aids . on	Requi	reme	nts  	· · ·	<ul> <li>4</li> <li>4</li> <li>4</li> <li>4</li> <li>4</li> <li>4</li> <li>4</li> <li>4</li> </ul>		* * * * * *	* * * * * *	· · ·	* * *	* * * * *	* * * *				* * * * *	10. 10. 10. 10. 10. 10.	5-3 5-4 5-4 5-5 5-5
5	COMM	ON SOFTWA	ARE, HAR	DWARE	, AN	D CO	NTRO	DL A	REQ	UIF	REM	ENT	rs		•			÷	÷	10.	6-1
	6.1	Common S	Software	Requ	irem	ients					*		,		4					10.	6-1
		6.1.1 6.1.2 6.1.3 6.1.4 6.1.5 6.1.6 6.1.7	Definiti Design P Sofc re Performa Verifica Availabi Maintain	on . Des nce l tion lity abil	ss ign Requi Tes and ity a	reme ting Reli	nts abi	nd lit	Qua y . abi	; 11 11	fic ty	at	ion						* * * * * * *	10. 10. 10. 10. 10. 10.	6-1 6-5 6-9 6-9 6-10 .6-12
	6.2	Common	Hardware	Req	uiren	nents	s .			,	×								1	10	6-12
		$\begin{array}{c} 6.2.1 \\ 6.2.2 \\ 6.2.3 \\ 6.2.4 \\ 2.5 \\ 6.2.6 \\ 6.2.7 \\ 6.2.8 \\ 6.2.9 \\ 6.2.10 \\ 6.2.11 \\ 6.2.12 \end{array}$	Definiti General Computer Switches Sensors Isolatic Valves Instrum Groundin Electric Cables, Field T	ion . Sys on De (Inst (Inst cal P Fibe ermin	tems vice rume ion Penet ropt	s . ntat and rati ics, n an	ion Cont ons and d Sp	and rol and Ra	i Co Po i Si aces		ro ro s	Sup	eat pl	tur						10 10 10 10 10 10 10 10 10 10 10	.6-12 .6-15 .6-15 .6-16 .6-17 .6-19 .6-21 .6-22 .6-23 .6-23 .6-24
	6.3	Cummon	Control	Syst	em R	equi	reme	ent	5	÷	•	•	÷		į,	÷	۰.	÷,	÷	10	.6-24
		$     \begin{array}{r}       6.3.1 \\       6.3.2 \\       6.3.3 \\       6.3.4 \\       6.3.5 \\       6.3.6 \\     \end{array} $	Definit Design Perform Availab Testabi Maintai	ion . Requi ance i <sup>7</sup> it lity nabi	Requ Requ /Ope and lity/	nts irem rabi Qual Serv	ent lit ifi	s . y . cat	ion lit	y		* * * * *								10 10 10 10 10	.6-24 .6-25 .6-25 .6-26 .6-27
	6.4	Conclu	sion .						÷							+				. 10	).6-27

.

Page

7	OVERA REQUI	LL PLAN REMENTS	T, REACTOR	AND	REACT	FOR C	001/	ANT	SY!	STE	MS .	M-	MIS	5					10.7-1	
	7.1	Purpose	and Scope	 			D1		, D										10.7-1	
	1.6	Reactor	Systems G	nus nu	-MIS	erall	11	ant,	, KI	eac	con		and	4					10 7-1	
	7.3	Overall	Plant M.M	IS IS	1-111.0	A . A	* *	1		*	*	1.1		1	1	1	1	×	10 7-2	
	7 4	Neutron	Monitorin	1 Suct	om M.	2TM-	3.0				*	e . e			*		*-	11	10.7-2	
	7.5	BWR Rod	Control S	vstem	M-MIS	2 2 2	* *			1		A . A			*				10.7-3	
	7.6	PWR Rod	Control S	vstem	M-MIS	ŝ	1.1							1	1	÷.,		1	10 7-3	
	7.7	BWR Rea	ctor Coola	nt Svs	tem M	M-MIS				1	1	11		1	1	1	1	Ĵ.,	10.7-3	
	7.8	PWR Rea	ctor Coola	nt Svs	tem M	M-MIS	- 0		11	1	÷.			1		÷.,	1.	÷.	10.7-4	
	7.9	WR Che	emical and	Volume	Cont	trol	Svs	tem	M-1	MIS		1		0.	1		2	1	18.7-4	
	7.10	Process	s Sampling	System	n M-M	IS .									3.	1	21	÷.,	10.7-4	
	7.11	PWR Bor	on Recycle	Syste	m M-I	MIS	2.2	11	11	1	2		- 1	2	1		9.	<u>.</u>	10.7-4	
	7.12	BWR Rea	actor Water	Clear	up S	vsten	1 M-	MIS						1	0		Ĉ.		10.7-4	
	7.13	PWR Ste	eam Generat	or sys	stem i	M-MIS	÷.,	-				1.		÷.		2	1	8	10.7-5	į.
	7.14	Reactor	r Coolant S	ystem	Leak	Dete	cti	on I	M-M	IS	21	1.4			6.	0.1			10.7-5	
	7.15	Conclus	sion							÷.,	÷.,	1		1	1	2	ĵ.		10.7-5	1
8	REACT	FOR PROT	TECTION AND	SAFET	TY SY	STEMS	6 M	MIS	RE	QUI	RE	MEN	TS		8.	Ť	÷	¥.	10.8-1	
	8.1 8.2	Purpose General	e and Scope 1 Requireme	nts fo	or Re	actor	Pr	ote	 cti	on	an	d S	af	etv		*	•		10.8-1	
		System	s Group M-M	IS.		3			1.1						•	÷	÷		10.8-1	
		8.2.1	Functions							1			ī.	ų,					10.8-1	1
		8.2.2	Boundaries	and	Inter	faces	š.,		÷.,	÷.,		1.0	- 14		÷.		ų,	1	10.8-2	ł.
		8.2.3	Common Con	trol	and M	on t:	rin	g S	tra	teg	jie	s f	for	Re	ead	tc	1			
			Protection	and !	Safet	y Sys	stem	S	÷			ξ.,		ж.	÷.	1		÷.	10.8-2	ł.
	8.3	Reacto	r Protectio	n Sys	tom.														10.8-2	2
		8.3.1	System Def	initi	on .			*					1			1.1		1	10.8-2	2
		8.3.2	Performanc	е.		1.1	÷.,					4	i	÷	1		1		10.8-3	3
		8.3.3	Configurat	ion		* 1	1.1											1	10.8-4	ŧ.
		8.3.4	Equipment	Requi	remen	ts.	4.14		×			х 1	e i q					÷	10.8-	5
	0.4	DUD D-							÷.											
	8.4	DWK Ke	actor Lore	Isola	tion	C001	ing	Sys	ten	1 M-	-M1	S							10.8-	5
	0.0	DWK MI	gn-Pressure	Inje	ction	ys	tem	M-M	115	$\sim$		A	6.19			+		$\sim$	10.8-	5
	0.0	DWK UE	cay Heat Ke	moval	Syst	em .	10.7		2.1			κ.	1	÷.			÷	-	10.8-0	5
	0.7	DWK ST	andby Liqui	alon	trol	Syst	em M	1-M1	5.	1	14	÷ .	e l'Is	14	14.	14	*	*	10.8-	5
	0.0	PWR Ke	sidual Heat	Kemo	val S	yster	m M-	MIS	1.0			4					×		10.8-	6
	0.9	PWK EM	ergency ree	nwate	r sys	teol	M-MI	S	+ +		Υ.	÷.)	•	i ie		1.00			10.8-	7
	0.10	PWK 5a	fety Inject	ion S	ystem	M-M	15 .				1	2.	4 . 9		×	÷	1	1	10.8-	1
	0.11	FWR Sa	Tety Depres	suriz	ation	and	Ver	nt S	yst	em	M-	-MI	5 .			÷.	÷	÷	10.8-	1
	0.12	Contal	nment Isola	tion	M-MIS		8. 9		÷		1	*	к <sup>1</sup> - к			×	. *	14	10.8-	1
	0.13	contal pup c	nment Syste	m m-m	15 .	1.1		÷	÷	- y	÷.		h., 1				÷		10.8-	1
	0.14	FWK LO	tible Cont	pray	Syste	em M-	MIS		8	1		4	81 - B	1	4		×.		10.8-	8
	0.10	compus	CIDIE Gas (	ontro	1 Sys	stem	M-M.	15		1.1	1								10.8-	8

....

а.

	8.16 8.17	Fissic Conclu	on F isic	Produ	ict I	eal	kage	e Co	ntri	01 5	iyst	em	M-1	MIS	*					•	•	4. 4	10.8-8 10.8-8
9	POWER	GENER	RATI	ION A	ND I	II AN	N TL	JRBI	NE-1	GENE	RAT	OR	SY	STE ,	MS	M	- M )	IS	*		×		10.9-1
	9.1	Purpos	se a	and S	cop								i ne		à	M =	in			,			10.9-1
	9.2 9.3 9.4 9.5 9.6 9.7 9.8 9.9 9.10	Turbin Main Main Main Main Main Main Main Main	al I ne-( and Turl Gene ater cal nsa iar; usi	Cener Extr bine erator Addi te Ma y Ste on	act Sys i Con i Con i tion i keu am	n System System	s fo yste M-M em M nsat yste tem	or P ems eam MIS M-MI te S em M Puri M-M	Gro Sys Syst M-MI fic NIS	em M satio	M-M M-M	IS Sys	tem	ат 	MI	S .		· · ·					10.9-1 10.9-2 10.9-2 10.9-2 10.9-2 10.9-2 10.9-2 10.9-2 10.9-3
10	AUXII REQUI	IARY IREMEN	AND TS	PLAN	IT S	UPPI	ORT	SEF	RVIC	ES S	SYS1	EM:	S M	-M]	S						*	•	10.10-1
	10.1	Purpo Gener	se al	and S Requi	Scop	e . ent	s f	or A	 Auxi	liai	ry a	ind	Pi	ant	. s	up	poi	rt			•		10.10-1
		Servi	ces	Syst	tems	Gr	oup	M-1	415	* ' '	×	*			ł	*	• ]	•	. *	1			10.10-1
		10.2.	1 2 3	Funct Bound Conti	tion dari col	s. es and	and Moi	 Int	 terf	ace: g Si	s . trat	tea	ies	, fo	r	Au	xi	112	ary	Å	*	*	10.10-1 10.10-2
		10.2. 10.2. 10.2.	4 5 6	and Inter Inder Fire	Supp grat pend Pro	ort ion enc tec	Sy and e and tio	ster d Co nd I n an	ns . bord Redu nd S	ina ndan ecun	tion ncy rity	Re	qui	ren	ner	its.							10.10-2 10.10-3 10.10-3 10.10-4
	10.3	Conc	lus	ion						÷		÷			÷	÷			Ċ,				10.10-4
11	CONC	LUSION							ι,					4	i,	÷	. 1					÷	10.11-1
AP	PENDI	XAD	EFI	NITI	ONS	0F	TER	MS /	AND	ACR	ONYI	4S		÷				ŝ		5	i,	÷	10.A-1
AP	PENDI	X B G	ENE	RIC	SAFE	ΤY	AND	LI	CENS	ING	IS	SUE	s.			į.					ŝ		10.B-1
AP	PENDI	ХСА	DVA	NCED	CON	TRO	LR	OOM	DES	IGN					ł								10.C-1
AP	PENDI	X D H D	UMA	N FA MENT	CTOR	S A	SSE	SSM	ENT	OF	EVO	LUT	ION	A?\	( F	REQ	UI	REP.	MEN.	ITS		•	10.0-1
AP	PENDI	XEO	PTI	MIZA	TION	SU	BJE	CT					2.							1	1		10.E-1

#### CHAPTER 11, "ELECTRIC POWER SYSTEMS"

1	INTRO	DUCTION	$\epsilon \to -\epsilon$	s x - x	1.1	×						+		< .	÷				•		11.1	-1
	1.1 1.2 1.3 1.4 1.5	Review ( Scope ar Policy ) Outstand Vendor-	Criteri nd Stru Issues ding Is or Uti	a cture  sues lity-	of (	hap ific	ter Ite	11 ems		* * * *				4 4 4	· · ·					4 4 4 4	11.1 11.1 11.1 11.1	-1 -1 -2 -2 1-3
2	GENE	RAL REQU	IREMENT	'S AND	POL	(CY	STAT	TEM	ENT	S							,		ł.		11.3	2-1
	2.1 2.2	Policy : General	Stateme Requir	ents . ement	s .	к к к к		с л к я	•	*		*	•	н н	• •	*	* *			к. ж	11. 11.	2 - 1 2 - 1
		2.2.1 2.2.2 2.2.3 2.2.4 2.2.5 2.2.6 2.2.7 2.2.8 2.2.9	Three-1 Securit Number Minimiz Eouipme Fire Pi Use of the Cor Emerger Therma Valve P	fier C y Sys of Sa tation ent . rotect Revis nmissi ncy Re l Over Motor	onceptems fety of ion ions on . spon load Oper	Div Div Clas to  to  Se F Dev ator	isio s 11 IEE aci	e c E S lit	omp tar ies rov	ida ida	ent	s N fo	ot	En	dor	se sti	d ł				11. 11. 11. 11. 11. 11. 11. 11.	2-3 2-4 2-5 2-5 2-6 2-6 2-8 2-9 2-10
	2.3	Conclus	ion .																		11.	2-10
3	OFFS	ITE POWE	R SYST	EM																	11.	3-1
	3.1 3.2	Functio Evaluat	inal De	script	ion.	• •		• . •		*			*	4	4. 10	 			•	•	11. 11.	3-1 3-1
		3.2.1 3.2.2	Use of Connec	Separ tion (	rate of th	Lowe	er V ffsi	olt te	age Tri	e S ans	wi mi	tch ssi	inton	S	ta	tio em	n to	ł	1	*	11,	3-1
			the Sa	fety (	Onsit	e Po	ower	Di	S.	rib	ut	ion	S.	yst	em			. *	ł	÷	11.	3-2
	3.3	Conclus	ion .			÷	• •	κ. 1		×	×	÷	×	÷	÷		ł	ķ	÷	×	11.	3-4
4	MEDI	UM-VOLTA	AGE AC	DISTR	IBUTI	ON	SYST	EM			÷.					, i s		,	j.	÷	11.	4-1
	4.1 4.2	Functio Evaluat	on and .	Descr	iptic	n				a A	•	, , , ,		1	а, а	, , , ,		•	,		11 11	4-1 4-1
		4.2.1 4.2.2	Lack o Loads Connec	f Alt	ernat	e P	ower y Bu	- S(	our Off	ce sit	fo	r N Pov	lon	-S	afe our	ty ce:	 S			*	11	4-1
			inroug	H N0.1	-Sare	sty	DUSE	:5	• •	. *	16	*		. *	. *		¥		4		11	4-4

EPRI Evolutionary Plant SER Accili

Page

-

		4.2.3 4.2.4 4.2.5	Sec Use Ele ity Pu	curit e of ectr y of	ty . Adji ical Rea	ust Fa	abl ult ir C	e S Ef	per fec	ed i cts t P	Mot or ump	or t	D he an	ci Ci d I	ves oas Rea	td	low or	n ( It	: Cap nte	at	, il nal			*	11.4-5 11.4-6
	4.3	Conclu	sio	n .									-									1			11.4-7
5	ONST	TE STAN	DRV	AC	POWE	p s		N V	CV1	TE	м														11.5-1
~	0451	IL STAR	001	nv.	TUNC	n 9	ior r		011					1									Ċ.		
	5.1 5.2	Functi Evalua	on i tio	and n .	Desc	rip	tio	n .	× +			4 4	a t	а А	έ . ε			*	х. х		к. к	8 -6	3 7	Ì	11.5-1 11.5-1
		5.2.1 5.2.2	Us Te Us	e of chni e of	the cal the	Co Spe Co	ombu ecif ombu	isti ica isti	on tic on	Tu on Tu	rbi Rec rbi	ne jui	re G	en mei ien	era nts era	ito ito	ors or	To To	Me	iat	tis t	fy.			11.5-1
		5.2.3 5.2.4 5.2.5 5.2.6 5.2.7	St Por Em Em Sa	atio wer wer erge erge fegu	n Bl Rati Rati ncy ncy ards	ack ng ng Die Die Co	out of of sel sel	the the Ge En der	pir Co D nen igin at	ng omb ies rat ne ion	Rec ust el or Aux	Ge LC Ge	re ine ine ine ine ine	me Tu ra /Li ry	nts rbi tor DOF Su	ine s s ipp	ieq	eni uei t	era nce Sys	ito is ite	ems			*. * * * *	11.5-2 11.5-2 11.5-4 11.5-6 11.5-7 11.5-11
	5.3	Conclu	isio	n.	+ +	÷	ί.		,		ł.	÷.				i a	1	1	÷	+	÷	÷			11.5-11
6	LOW-	VOLTAGE	AC	DIS	TRIB	UTI	ON	SYS	TE	М.,		i.		4	÷ ,			-1	ę			×			11.6-1
	6.1 6.2	Functi Evalua	on itio	and n an	Desc d Co	rip ncl	otic lusi	n Ion	•		х х	*	•	•			*	ł			ii ii	•			11.6-1 11.6-1
7	DC A	ND LOW-	VOL	TAGE	VIT	AL	AC	POW	IER	SU	PP	Y	SY	ST	EMS	ŝ.,		,	÷.	ł.	ŝ	x.	ł		11.7-1
	7.1 7.2	Functi Evalua	ions itio	and n .	Key	De	esig	n F	leq:	uir	eme	ent	s					1		;	-				11.7-1 11.7-3
		7.2.1	Lo Al	ss o lowe	f Po d Ou	wer tac	r to je 1	) a lime	DC	Bu for	s D(		af	et	y E		es.	ì	n' F	ii	NR		ł		11.7-3
		7.2.3	Ev Se	olut	iona ty.	ry	Pla	ant S.	Te	chn	ica	1	Sp	ec	ifi	ica	iti	on	s •	•		*	а		11.7-4 11.7-5
		1.6.4	Un	inte	rrup	tit	ple	Pov	ver	Su	pp	ie	s a t	et	y-1	(e i	at.	ed .			r.	i.	ł		11.7-5
	7.3	Conclu	isio	n.	÷						4	į,				į,		1	÷	1	ł.			1	11.7-6
8	NORM	IAL AND	EME	RGEN	ICY L	1Gł	TIM	IG			į,	*		*								÷			11.8-1
	8.1 8.2	Functi Evalua	ion atio	and n .	Desc	rip	otic	on .	- 	a' a à a	4	2 2					•	i k	-	÷,	*	*	*		11.8-1 11.8-2

a Paras

		8.2.	.1	Com	pari	sor	1 01	f t	he	AL	.WF	1.1	igl	hti	ing	S	ys	te	m	Re	qu	ir	en	er	its			
				Witi	h Cu	irre	ni	Li	ght	ti	ng.	Sy	st	em	De	si	gn			4	÷		4	+		,	4	11.8-2
		8.2	.4	Fme	nal roer	LIG	lic	ing	in	y S1	Len	1. 4		1	*	*	1	*	1	*	4	*	*	*	1	4	1	11.8-5
		8.2	. 4	Sec	urit	y l	igi	hti	ng	S	yst	tem		4	ł			i.	а. А.		i.	i.	1		*		a'	11.8-8
	8.3	Cond	clus	sion													÷										÷	11.8-9
9	ELEC	TRIC	AL I	PROT	ECTI	IVE	SY:	STE	MS							÷				,			,				*	11.9-1
	9.1	Fund	cti	ons	and	Des	scr	ipt	io	n				,	4	÷	÷.											11.9-1
	3.6	com	ciu	ston				1			1		1	1	÷.	1		1	1	1	*	1		1		1	*	11.9-2
10	CONC	LUSI	ON	4 . A		8		4		•									÷.	ų,			,		÷	×	÷	11.10-1
A	PENDI	ХА	DE	FINI	TIO	NS /	AND	A	RO	NYI	MS	١.	1						÷							ł	÷	11.A-1
A	PPENDI	ХB	GE	NERI	C SI	AFE	TY I	ANC	) L	ic	EN	S I N	G	15	SU	E S			×	, i							ł.	11.B-1
A	PPENDI	ХС	RE	GULA	TOR	Y DI	EPA	RTL	JRE	A	NAI	LYS	15			-			į.		1		1					11.0-1

# CHAPTER 12, "RADIOACTIVE WASTE PROCESSING SYSTEMS"

1	INTR	ODUCTION
	1.1 1.2 1.3 1.4 1.5	Review Criteria
2	POLI	CY STATEMENTS AND KEY REQUIREMENTS
	2.1 2.2	Policy Statements
		2.2.1Goals of Radioactive Releases and Waste Reduction12.2-42.2.2Source and Input Terms12.2-42.2.3Releases of Radioactive Materials12.2-72.2.4Personnel Radiation Exposure12.2-72.2.5Operating Conditions/Availability12.2-82.2.6Process Systems Operating Capacity12.2-82.2.7Seismic Design and Quality Group Classification12.2-82.2.8Control and Instrumentation12.2-82.2.9Process and Effluent Radiological Monitoring12.2-82.2.10Fire Protection12.2-10
	2.3	Conclusion
EF	PRI EN	volutionary Plant SER xxxv
# TABLE OF CONTENTS (Continued)

Page

di

4

3	GASEC	US RAD	IOACTIVE	WASTE	PROCE	SSING	G SY	STE	MS	ŝ								×		1	12.3	5-1	
	3.1 3.2 3.3	Functio Perform Evaluat	ons mance Req tion	uireme	ents	· · ·		• •	*		•	е х.	• •		*		*	* *	* * *		12. 12. 12.	3-1 3-2 3-3	
		3.3.1	Seal Ste	am .								4						.,			12.	3-3	
		3.3.2	Design .	Venti	latin	g, a	nd A	11r	0.0	nd	111	on	119		sys	, te	em.		x		12.	3-4	
		3.3.3	Installa Filters	tion c	of Hig	h-Ef	tici	end	су	Pa	rti	, cu	1 a '	te.	A1	r,			ž		12.	3-5	
		3.3.4 3.3.5 3.3.6	Hydroger Bypass ( Fire Pro	n Contr of Char otectio	rol . rcoal on .	Beds	* *	*		•	4 14 14		* * *	*							12. 12. 12.	3-6 3-7 3-7	
	3.4	Conclu	sion .								÷.			x							12.	3-7	
4	LIQU	ID RADI	OACTIVE V	NASTE	PROCES	SING	SY:	STE	м.			έ.									12.	4-1	
	4.1 4.2 4.3	Functi Perfor Conclu	ons mance Re ision .	quirem	ents	5 5 5 5 5 5	· · · · · · · · · · · · · · · · · · ·	* *	• •				*	ж. ж.					•		12. 12. 12.	4-1 4-1 4-3	1
5	SOLI	D RADIC	DACTIVE W	ASTE P	ROCESS	SING	SYS	TEM				,				8					12	5-1	l
	5.1 5.2	Defini Wet Sc	ition . olid Wast	 e Proc	 essing	g Sys	tem		*			•	•	÷.,			A.				12 12	5-1	1
		5.2.1	Functio System	ns . Requir	ement	s .	 	, ,	•		 	•	•	•	* *	, - , -	k N				12 12	5- 5-	1
	5.3	Dry So	olid Wast	e Proc	essin	g Sy:	stem					. *		÷		ł.					12	. 5 -	3
		5.3.1 5.3.2	Functio System	ns . Requir	 rement	s ,		× ×	4 4			4	.s .v	1	k X						12 12	. 5- . 5-	3 3
	5.4	Onsit	e Storage	Facil	lity .	• •		×	÷	÷					×,	ļ,		*			12	. 5-	4
		5.4.1 5.4.2	Functio	ons . nance F	Requir	emen	ts .		4. 1	i i					4 4	4. 4	ų L	, ,			12 12	. 5-	4
	5.5	Concl	usion .	e 1 1		· .			÷	,	÷,			÷	*		÷	4	i.		12	. 5-	5
6	CON	CLUSION	x .x > .x			÷			×	į.						i,		į.	į.	1	12	.6-	1
A	PPEND	IX A D	EFINITIO	NS AND	ACRON	IYMS	*	.,			*								κ.	*	12	. A-	1
P	PPEND	IX B S	UMMARY O	F DESI	GNS AN	D OP	ERA	TIN	G 1	TEC	HN.	IQU	JES	T	0 1	REI	DUC	E			12	2.B-	-1

## TABLE OF CONTENTS (Continued)

## CHAPTER 13, "MAIN TURBINE-GENERATOR SYSTEMS"

1	INTR	ODUCTIO	la da da da da			۰.		4		κ.	e ie		4	.4		×	4	13	.1-1
	1.1 1.2 1.3 1.4 1.5	Review Scope Policy Outstan Vendor	Criteria and Structure of Issues ding Issues . or Utility-Spe	Chapte	r 13 	* * *	· ·	* * * *		x ; ;				* * *	* * * *		4 4 4 8	13 13 13 13 13	1-1 1-1 1-2 1-2 1-3
2	POLI	CY STAT	EMENTS AND KEY F	EQUIREM	ENTS	1			÷			,	5	÷	÷	÷	×	13	2-1
	2.1 2.2 2.3	Policy Perform System	Statements mance and Operat and Equipment R	ional R lequirem	equin ents	eme	ent	s			 	- 4 - 4 - 4		×	* *	* * *	4 4 3	13 13 13	2-1 2-1 2-2
3	MAIN	TURBIN	SYSTEM			* *		÷	*	à.,				*			÷	13	3-1
	3.1	Main T	urbine												*	i	*	13	3-1
		3.1.1 3.1.2 3.1.3 3.1.4 3.1.5 3.1.6 3.1.7 3.1.8	Safety Classifi Inspection and Maintenance . Turbine Missile Rotor Disk . Performance Ver Turbine Exhaust Nozzle Block Al	cation Quality s ificati Boot ignment	Assu	irar	nce	* * * * * * *	* * * * * * *					* * * * * * * *				13 13 13 13 13 13 13 13	3-1 3-3 3-4 3-6 3-6 3-6 3-7
	3.2 3.3 3.4 3.5	Turbing Turbing Gland S Instrum	e Lube Oil Syste Control System Geal System Mentation	m 1 	4 4 4 4 5 5 4 4				* * * *	 		* * * *	* * * *					13 13 13 13	3-7 3-8 3-10 3-11
		3.5.1 3.5.2 3.5.3 3.5.4 3.5.5 3.5.6 3.5.7	Turbine Supervi Alarm-Initiatin Turbine/Reactor On-Line Diagnos Performance Ins Solid-State Dev Bearing Oil Dra	sory In g Devic Interf tic Ins trument ices in Flow	strum es . ace l trume atior	inst inst inta	tru tru	ion men cn	ta	tic	'n							13 13 13 13 13 13 13	3-11 3-12 3-12 3-12 3-12 3-12 3-12 3-13
4	MAIN	GENERA	OR SYSTEM		×			+	×.	j.		*		÷	ļ	×.	ł	13	4-1
	4.1	Main Ge	nerator									×			×		. ]	13	4-1
		4.1.1 4.1.2	Performance and Systems and Equ	Operat ipment	ional Requi	Re	equ nen	ire ts	me	nts		*		*		*	•	13. 13.	4-1 4-2

Page

## TABLE OF CONTENTS (Continued)

and the

	4.2	Excitation System	14		÷ .		* 0											13.4-2
	4.3	Stator Cooling Water System .	1	-	1	γ.	1.1	÷.,	4 ° 8	14	1	1		1	*	÷.		13.4-2
	4.4	Hydrogen Cooling System		1	41				i	1.							έ.	13.4-2
	4.5	Hydrogen Seal Oil System	- 4			×								1		×.		13.4-3
	4.6	Generator Hydrogen and Carbon	D	10)	(id	e	Sy:	st	em									13.4-3
	4.7	Generator Control Systems									÷	÷.			۰.	*	$\mathbf{r}^{\prime}$	13.4-3
	4.8	Instrumentation				۰.	÷		4		×	+	.*	×	4		5	13.4-3
5	CONC	USION					4				÷	×		÷			÷	13.5-1
AF	PENDI	A DEFINITIONS AND ACRONYMS		Ξ.		ï						1		Ϊ.				13.A-1

## LIST OF TABLES

1-1	Design-Basis Tornado Characterístics	1.4-36
1-2	BWR Water Chemistry Guidelines	1.5-21
1-3	PWR Water Chemistry Guidelines	1.5-22
3B.1	Generic Safety Issues Addressed in the DSERs	18.3-72
3B.2	Safety Issues Applicable to Future Plants That Have Not Been Prioritized	18.3-79
3B.3	Newly Identified Safety Issues	1B.3-81
48.1	Policy Issues for the Evolutionary Plant Designs	18.4-3

]•

## PREFACE

This safety evaluation report (SER) (Volume 2) documents the review by the U.S. Nuclear Regulatory Commission (NRC) staff of the 13 chapters of Volume II of the Electric Power Research Institute's (EPRI's) Advanced Light Water Reactor (ALWR) Utility Requirements Document (hereafter referred to as the "Evolutionary Requirements Document"). Volume 1, which contains the program summary of the NRC review of Volumes I, II, and III of the ALWR Utility Requirements Document, also contains the references cited and the abbreviations used in this SER.

Each chapter of the Evolutionary Requirements Document defines the ALWR Utility Steering Committee's requirements for the design of evolutionary plants. These requirements apply to boiling-water reactors (BWRs) and pressurized-water reactors (PWRs), which will be rated at approximately 1350 megawatts-electric.

The design criteria specified by EPRI are intended to ensure that EPRI's policy statements discussed in Volume I of the ALWR Utility Requirements Document are met. These policy statements are discussed in Section 1.3 of Volume 1 of this report. They include consideration of simplification, design margin, human factors, safety, regulatory stabilization, standardization, use of proven technology, maintainability, constructibility, quality assurance, economics, protection against sabotage, and environmental effects.

The format of each chapter of this SER follows that of the corresponding chapter of the Evolutionary Requirements Document as closely as possible. Unless otherwise noted, references to sections of the Evolutionary Requirements Document pertain to that chapter.

#### Outstanding Issues

During its review of the original version of the Evolutionary Requirements Document, the staff identified two types of issues for which additional information was required before the staff could reach a final conclusion. The staff considered these issues to be outstanding. These issues fell into one of two categories: (1) open issues that had to be resolved before the staff could complete its review of the Evolutionary Requirements Document or (2) confirmatory issues for which the staff would ensure that EPRI met its commitments to revise the Evolutionary Requirements Document.

There are no open issues remaining on the Requirements Document for evolutionary plant designs other than policy issues on which the staff has taken a position, but for which the Commission has not had the opportunity to provide guidance. To provide continuity of the review, both the open and confirmatory items identified in the DSERs and the remaining open policy issues are listed in Section 1.4 of each chapter.

## Vendor- or Utility-Specific Items

During its review of the Evolutionary Requirements Document, the staff identified items that were inadequately addressed by EPRI or were issues that could not be addressed generically. These items will have to be resolved during the staff's review of a vendor- or utility-specific application (i.e., an application for final design approval and design certification (FDA/DC) or a combined construction permit and operating license (combined license). They are listed in Section 1.5 of each chapter.

As discussed in Section 1.2 of Volume 1 of this report, the Requirements Document has no legal or regulatory status and is not intended to demonstrate complete compliance with the Commission's regulations, regulatory guidance, or policies. It is not intended to be used as a basis for supporting FDA/DC for a specific design, nor is it to be used to substitute for any portion of the staff's review of future applications for FDA/DC. Specifically, satisfactory resolution of the items identified in Sections 1.4 and 1.5 of each chapter for a vendor- or utility-specific application will not, by itself, support a finding that the application complies with the Commission's regulatory requirements. The staff will perform a complete licensing review of these applications using NUREG-0800, "Standard Review Plan [SRP] for the Review of Safety Analysis Reports for Nuclear Power Plants," and other appropriate Commission guidance. Satisfactory resolution of the open policy issues and vendor- or utility-specific items constitutes only one portion of the staff's review.

## Availability

Copies of this report are available for inspection at the NRC Public Document Room, 2120 L Street, N.W., Washington, DC 20555.

The NRC project managers for the staff's review of EPRI's ALWR Utility Requirements Document are J. H. Wilson and T. J. Kenyon. They may be contacted by calling (301) 504-1118 or by writing to: Associate Directorate for Advanced Reactors and License Renewal, U.S. Nuclear Regulatory Commission, Washington, DC 20555.

### CHAPTER 1, "OVERALL REC"IREMENTS"

## 1 INTRODUCTION

This chapter of the safety evaluation report (SER) documents the review by the staff of the U.S. Nuclear Regulatory Commission (NRC) of Chapter 1, "Overall Requirements," of Volume II of the Electric Power Research Institute's (EPRI's) Advanced Light Water Reactor [ALWR] Utility Requirements Document (hereafter referred to as the "Evolutionary Requirements Document") through Revision 3. Chapter 1 was prepared under the project direction of EPRI and the ALWR Utility Steering Committee, by ABB Combustion Engineering, Incorporated; MPR Associates, Incorporated; S. Levy, Incorporated; and TENERA, L.P.

On July 8, 1986, EPRI submitted Chapter 1 of the Evolutionary Requirements Document for staff review. By letters dated January 5 and March 18, 1987, the staff requested that EPRI supply additional information. EPRI provided the information in its response dated March 27, 1987.

On September 24. 1987, the staff issued its draft safety evaluation report (DSER) for Chapter 1 of the Evolutionary Requirements Document. On February 18, 1988, the staff issued a revision to its DSER. As a result of its review of the other chapters of the Evolutionary Requirements Document, the staff requested that EPRI provide additional information by letters dated March 25, 1988, and September 14, 1989. EPRI responded to these requests in its letters dated March 30 and December 22, 1989.

On July 12, 1990, and April 9, 1991, the staff and EPRI met with the Advisory Committee on Reactor Safeguards (ACRS) Subcommittee on Improved Light Water Reactors to discuss Chapter 1, the staff's corresponding DSER, the outstanding issues from the staff's review of Chapter 1, and EPRI's approach to resolving each issue.

On September 7, 1990, EPRI submitted Revision 1 of the Evolutionary Requirements Document. In that revision, EPRI significantly modified Chapter 1. Revisions 2, 3, and 4 of the Evolutionary Requirements Document were docketed on April 26 and November 15, 1991, and April 17, 1992, respectively. EPRI submitted additional information regarding Chapter 1 by letters dated May 13, July 2, and December 6 and 21, 1991, and January 9, February 3, and March 19, 1992.

## 1.1 Review Criteria

Section 1 of Volume 1 of this report describes the approach and review criteria used by the staff during its review of Chapter 1 of the Evolutionary Requirements Document.

#### 1.2 Scope and Structure of Chapter 1

Chapter 1 of the Evolutionary Requirements Document defines the ALWR Utility Steering Committee's overall requirements for the design of nuclear power plants that have evolved from the current generation of power plants.

The key topics addressed in the Chapter 1 review include EPRI-proposed design requirements pertaining to safety, performance, structural design, materials, reliability and availability, constructibility, operability and maintainability, quality assurance, and mechanical equipment. Additional topics include EPRI-proposed requirements for licensing and the design process.

## 1.3 Policy Issues

In a staff requirements memoranda (SRM) dated August 24, 1987, the Commission instructed the staff to provide an analysis detailing where the staff proposes departure from current regulations or where the staff is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs. The staff considers these to be policy issues. Appendix B to this chapter provides the staff's regulatory analysis of those issues identified for the evolutionary plant designs. These issues have been addressed in SECY-90-016, "Evolutionary Light Water Reactor Certification Issues and Their Relationship to Current Regulatory Requirements"; SECY-91-078, "Chapter 11 of the Electric Power Research Institute's Requirements Document and Additional Evolutionary Light Water Reactor Certification Issues"; and a draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," that was issued on February 27, 1992. In its SRM dated June 26, 1990, and August 15, 1991, the Commission provided its decisions or SECY-90-016 and SECY-91-078, respectively.

The February 27, 1992, draft Commission paper has been forwarded to the ACRS. The staff will include the views of the ACRS in the final paper and document its final positions before seeking Commission approval. Since the Commission has not reviewed the approaches to resolving these issues, they do not represent agency positions.

These policy issues are considered fundamental to agency decisions on the acceptability of the ALWR designs. The policy for the evolutionary plant designs are the following. They are also listed in Table 48.1 in Appendix B to Chapter 1 of this report.

Policy Issues for the Evolutionary Plant Designs

#### Policy Issue

- use of physically based source term
- anticipated transients without scram
- mid-loop operation
- station blackout
- fire protection
- intersystem loss-of-coolant accident
- hydrogen control
- core-concrete interaction ability to cool core debris
- high-pressure core melt ejection
- containment performance
- dedicated containment vent penetration
- equipment survivability
- elimination of operating-basis earthquake

EPRI Evolutionary Plant SER

1.1-2

- inservice testing of pumps and valves
- industry codes and standards
- electrical distribution
- seismic hazard curves
- leak before break
- classification of main steamline or boiling-water reactor
- tornado design basis
- containment bypass
- containment leak rate testing
- postaccident sampling system
- level of detail
- prototypiny
- inspections, tests, analyses, and acceptance criteria
- reliability assurance program
- site-specific risk assessments
- severe-accident mitigation design alternatives
- generic rulemaking related to FDA/DC

## 1.4 Outstanding Issues

The DSER for Chapter 1 of the Evolutionary Requirements Document contained the following outstanding issues:

## Open Issues

- plant site parameters (2.3.1 and 4.5.2)
- (2) classification of certain types of events (2.3.2)
- (3) station blackout classification (2.3.2)
- (4) EPRI ALWR public safety goal (2.3.3)
- (5) 60-year design life (3.3)
- (6) quality assurance program for certain seismic Category II items (4.3.1 and 4.3.2)
- (7) use of Uniform Building Code Zone 2A specification (4.3.2)
- (8) damping values in Code Case N-411 of American Society of Mechanical Engineers Boiler and Pressure Vessel Code (ASME Code) (4.4.3)
- (9) vibratory loads with significant high-frequency input deviation from Regulatory Guide (RG) 1.92 (4.4.3)
- (10) maximum ground water level (4.5.2)
- (11) tornado effects noncompliance with RG 1.76 (4.5.2)
- (12) BWR safety/relief valve loads (4.5.4 and 4.5.5)
- (13) leak before break (4.5.5)

- (14) in-plant hazards regarding remaining BWR suppression pool loads after demonstration of leak before break (4.6.1)
- (15) decoupling safe shutdown earthquake (SSE) from loss-of-coolant accident (4.6.1, 4.6.2, and Appendix B)
- (16) operating-basis earthquake/SSE relationship (4.6.1, and Appendix B)
- (17) seismic qualification of equipment (4.8.1)
- (18) selection of materials for reactor coolant pressure boundary piping compliance with NUREG-0313 (5.3.1)
- (19) construction program quality assurance (7)
- (20) reference to Institute of Electrical and Electronic Engineers (IEEE) P1023-1988 and EPRI-2350 for guidance regarding human factors engineering (8.2)
- (21) meaning of NRC approval of EPRI Requirements Document (10)

## Confirmatory Issues

- (1) sabotage protection (2)
- (2) initiating events (2.3.2)
- (3) seismic ductility factors and ductility limits (4.3.2)
- (4) structural codes and standards for structures, systems, and equipment (4.4, 4.4.1, and 4.4.2)
- (5) compliance with General Design Criterion 4 (4.5.5)
- (6) internal flooding (4.5.5)
- (7) seismic and dynamic qualification by experience (4.8.1)
- (8) seismically qualified anchorage (4.8.1)
- (9) hardness limits for martenritic stainless steel (5.3.1)
- (10) use of Alloy 600 (5.3.1)
- (11) allowance for carbon and low-alloy-steel corrosion (5.3.1)
- (12) failure mechanisms (6.2)
- (13) construction verification milestones (7)
- (14) inspectability and provisions for inservice testing (8.2 and 8.6)
- (15) preventive maintenance and inspections (8.2)
- (16) personnel qualification requirements (8.2)

EPRI Evolutionary Plant SER

1.1-4

- (18) operation problem areas (8.6)
- (19) quality assurance requirements (9)
- (20) quality problems during design and construction (9)
- (21) updating Appendix B cross-reference table (10)
- (22) cross-reference table of unresolved and generic safety issues (10)
- (23) use of life extension experience (11.3)
- (24) living probabilistic risk assessment (11.10)
- (25) Section XI of ASME Code (12.2)

The final disposition of each of these issues is discussed in greater detail in the appropriate section of this chapter, as indicated by the parenthetical notation following each issue. Most of the issues identified in the DSER for Chapter 1 have been resolved. However, during its review of the significantly modified version of Chapter 1 of the Evolutionary Requirements Document, the staff identified two open policy issues on which the staff has taken a position, but for which the Commission has not had the opportunity to provide guidance. The open issues, with references to appropriate sections of this chapter given in parentheses, are listed below. The designators in front of each issue provide a unique identifier for each issue. The letter "E" indicates that the issue applies to evolutionary plant designs. The first number designates the chapter in which it is identified. The letter "O" designates that it is an open issue. The final number is the sequential number assigned to it in the chapter.

## Open Issues

E.1.0-1 tornado wind speeds (4.5.2) E.1.0-2 leak before break (4.5.5)

## 1.5 Vendor- or Utility-Specific Items

During its review of Chapter 1 of the Evolutionary Requirements Document, the staff identified items that were inadequately addressed by EPRI or were issues that could not be addressed generically. These items will have to be resolved during the staff s review or a vendor- or utility-specific application (i.e., an application for final design approval and design certification (FDA/DC) or a combined construction permit and operating license (COL)).

These vendor- or utility-specific items, with references to appropriate sections of this chapter given in parentheses, are listed below. The designators in front of each issue provide a unique identifier for each issue. The letter "E" indicates that the issue applies to evolutionary plant designs. The first number designates the chapter in which it is identified. The letter "V" designates that it is a vendor- or utility-specific item. The final number is the sequential number assigned to it in the chapter.

- E.1.V-1 scope of mitigation features (2.1 and 2.4)
- E.1.V-2 implementation of design characteristics intended to enhance accident resistance (2.2)
- E.1.V-3 bounding analysis by standard site design parameters (2.3.1)
- E.1.V-4 selection of initiating events and their frequency categorization (2.3.2)
- E.1.V-5 acceptance criteria for transient and accident analysis (2.3.2)
- E.1.V-6 anticipated transient without scram response analysis (2.3.2)
- £ 1.V-7 acceptability of analytical codes and methodologies for safety analysis (2.5)
- E.1.V-8 60-year plant life (3.3, 4.8.2, 8.2, and 11.3)
- E.1.V-9 operation of PWR with a secured reactor coolant pump (3.5)
- E.1.V-10 defense-in-depth analysis (3.5)
- E.1.V-11 event response capability (3.5)
- E.1.V-12 fuel burnup requirements (3.6)
- E.1.V-13 extended operating life of control blades and control rod assemblies (3.6)
- E.1.V-14 safety classification (4.3.1)
- E.1.V-15 seismic qualification by experience (4.3.2 and 4.8.1)
- E.1.V-16 non-seismic building structures (4.3.2 and 4.7.2)
- E.1.V-17 structural design and construction codes (4.4 and 4.4.1)
- E.1.V-18 elimination of operating-basis earthquake from design (4.4.3, 4.7.3, and Appendix B)
- E.1.V-19 definition of support group (4.4.3)
- E.1.V-20 use of Appendix N of ASME Code, Section III (4.4.3 and 4.7.3)
- E.1.V-21 analysis of vibratory loads with significant high-frequency input (4.4.3)
- E.1.V-22 use of nonlinear analysis to account for gaps between pipes and piping supports (4.4.3)
- E.1.V-23 probabilistic approach for changing existing loads and/or loading combinations (4.5.1)
- E.1.V-24 recurrence interval for wind loadings (4.5.2)

EPRI Evolutionary Plant SER

1.1-6

- E.1.V-25 maximum ground water level (4.5.2)
- E.1.V-26 precipitation for roof design (4.5.2)
- E.1.V-27 snow loading (4.5.2)
- E.1.V-28 detailed quantification of soil parameters (4.5.2)
- E.1.V-29 minimum margin against liquefaction (4.5.2)
- E.1.V-30 external hazards evaluation (4.5.2)
- E.1.V-31 number of full-stress cycles (4.5.2 and 4.8.1)
- E.1.V-32 site-specific SSE (4.5.2)
- E.1.V 33 power spectrum density function of the time history (4.5.2)
- E.1. + external impact hazards (4.5.2)
- E.1.V-35 design temperature (4.5.2)
- E.1.V-38 protection against surface vehicle bombs (4.5.3)
- E.1.V-39 BWR safety/relief valve loads (4.5.4)
- E.1.V-40 NUREG-1061 methodology and acceptance criteria for leak before break (4.5.5)
- E.1.V-41 hydrodynamic loads from safety/relief valves (4.5.5)
- E.1.V-42 suppression pool dynamic loads (4.5.5)
- E.1.V-43 design against internal-missile generation (4.5.5)
- E.1.V-44 design of concrete containment (4.6.1)
- E.1.v-45 load combinations for seismic Category I buildings and structures (4.6.1)
- E.1.V-46 design of seismic Category I steel structures (4.6.1)
- E.1.V-47 combination of pipe rupture loads with seismic loads for seismic Category I structures (4. 1 and 4.6.1)
- E.1.V-48 combination of loss-of-coolant-accident and SSE loads (4.6.1)
- E.1.V-49 load combinations for safety-related portions of the plant (4.6.2)
- E.1.V-50 dynamic analysis techniques (4.7.2)
- E.1.V-51 methodology for generating design response spectra or time histories (4.7.2)
- E.1.V-52 structural damping values (4.7.2)

- E.1.V-53 masonry walls in Category I buildings (4.7.2)
- E.1.V-54 use of expansion anchor bolts compliance with Office of Inspection and Enforcement Bulletin 79-02 (4.7.2 and 4.7.3)
- E.1.V-55 stability of shell-type structures under compression (4.7.2)
- E.1.V-56 seismic evaluation and design of small-bore piping (4.7.3)
- E.1.V-57 use of ASME Code Case N-411 (4.7.3)
- E.1.V-58 use of ASME Code Cases N-411 and N-420 in same analysis (4.7.3)
- E.1.V-59 construction of core support structures (4.7.3)
- E.1.V-60 fatigue design curves (4.7.3)
- E.1.V-61 use of IEEE 323 (4.8.2)
- E.1.V-62 environmental qualification of mechanical and electrical equipment (4.8.2)
- E.1.V-63 use of zinc to reduce radiation fields (5.2.7)
- E.1.V-64 limits on nitrites, nitrates, and total halogens as chlorine (5.2.8)
- E.1.V-65 grinding controls for PWRs (5.3.1)
- E.1.V-66 effect of fabrication processes on intergranular stress corrosion cracking (5.3.1 and 5.3.1)
- E.1.V-67 hardness limits for stainless steel (5. .1)
- E.1.V-68 use of Alloy 600 and other alloys (5.3.1)
- E.1.V-69 allowance for carbon and low-alloy-steel corrosion (5.3.1)
- E.1.V-70 selection of seals, gaskets, and protective coatings (5.3.5)
- E.1.V-71 aging of cable insulation and other electrical materials (5.3.6)
- E.1.V-72 use of hydrogen water chemistry for the advanced BWR design (5.5.2)
- E.1.V-73 PWR water chemistry (5.5.2)
- E.1.V-74 submittal of operational reliability assurance program (O-RAP) (6)
- E.1.V-75 organizational description for reliability assurance program (6.1)
- E.1.V-76 analyses methods or models used in developing the reliability assurance program (6.2)
- E.1.V-77 reliability data bases (6.2)
- E.1.V-78 reliability, maintainability, and testability analyses (6.2)

- E.1.V-79 apportionment of contributions of structures, systems, and components to core damage frequency (6.3)
- E.1.V-80 priority of safety in accident recovery (6.3)
- E.1.V-81 relationship between safety and production availability (6.3)
- E.1.V-82 effect of limitations on refueling duration on plant safety (6.3)
- E.1.V-83 effect of planned outage duration on plant safety (6.5)
- E.1.V-84 effect of major outage duration on plant safety (6.3)
- E.1.V-85 inspection of construction activities (7 and 11.13)
- E.1.V-86 quality assurance for non-safety-related facilities and systems (7)
- E.1.V-87 installed operating-phase security system (7)
- E.1.V-88 reliability of modular construction (7)
- E.1.V-89 use of IEEE P1025 P1023-1988/D5 and EPRI-2360 for guidance regarding human factors engineering (8.2)
- E.1 V-90 inspection and verification of security locks robotically (8.3)
- E.1.V-91 quality assurance requirements for all equipment, structures, systems, facilities or software that have some safety importance or has one importance (9)
- E.1.V-92 compliance of FDA/DC applications with Commission's regulations and guidance (10)
- E.1.V-93 issue resolution for FDA/DC reviews (10)
- E.1.V-94 inspections, tests, analyses, and acceptance criteria (10)
- E.1.V-95 implementation of simplification objective (11.4)
- E.1.V-96 implementation of standardization objective (11.5)
- E.1.V-97 check valve testing methods (12.2.2)
- E.1.V-98 full-flow testing of check valves (12.2.2)
- E.1.V-99 qualification testing of active and nonactive motor-operated valves (MOVs) (12.2.2)
- E.1.V-100 technical concerns regarding MOVs (12.2.2)
- E.1.V-101 leak rate testing for individual containment isolation valves (12.2.2)
- E.1.V-102 instrumentation to determine net positive suction head during all modes of operation (12.2.3)

- E.1.V-103 testing of pump flow rate (12.2.3)
- E.1.V-104 frequency and extent of disassembly and inspection of safety-related pumps (12.2.3)

#### 2 SAFETY DESIGN REQUIREMENTS

In Section 2.2.F.7 of the DSER for Chapter 1, the staff identified a confirmatory issue regarding sabotage protection. EPRI had proposed to add failures caused by sabotage to its description of hazards for which the plant designer should consider spatial separation of systems and equipment as the way to protect against redundant trains being incapacitated by a single hazard. Because of the extensive revision of Chapter 1, the Evolutionary Requirements Document no longer includes this requirement. Because Section 73.55 of the Title 10 of the <u>Code of Federal Regulations</u> (10 CFR) does not require spatial separation of systems and equipment, the staff concludes that this DSER confirmatory issue is closed.

#### 2.1 Introduction

The EPRI ALWR safety design requirements consist of three levels of the defense-in-depth framework, that is, resistance to accidents, prevention of core damage, and mitigation of the consequences of accidents, and are separated into two types of requirements: licensing-design-basis (LDB) and safety-margin-basis (SMB) requirements. EPRI defines the LDB to be the set of events and associated boundary conditions and assumptions that must be analyzed to satisfy regulatory requirements. This analysis will be done using conservative or appropriately justified best-estimate, NRC-approved calculation methods and assumptions and will meet NRC-mandated acceptance criteria. The Evolutionary Requirements Document states that the EPRI-defined SMB contains design requirements that go beyond the minimum required by thr D CFR. EPRI states that these design requirements provide safety assurance beyond that provided by the regulatory requirements for investment protection and severe-accident protection.

In its letter dated April 24, 1991, the staff requested that EPRI identify additional hardware or other specific actions resulting from the SMB requirements in applicable sections of the Evolutionary Requirements Document. In its letter dated July 2, 1991, EPRI indicated that, to the extent practical. such additions of hardware or other specific actions resulting from the SMB requirements had been identified in specific sections of the Evolutionary Requirements Document. EPRI's response cocluded some examples that satisfied this concern.

Section 2.1 of Chapter 1 of the Evolutionary Recuirements Document states that safety design requirements consist of the three levels of the defense-in-depth framework. The defense-in-depth framework consists of requirements that increase the ALWR's resistance to an accident and improve its capability to prevent core damage or mitigate the consequences of an accident in the unlikely event it should progress beyond core damage.

It is unclear to the st. 'f that EPRI's mitigation requirements are intended to include design-basis accidents (DBAs). Therefore, the staff will evaluate this matter during its review of an individual application for FDA/DC to ensure that the mitigation features of the design include consideration of both DBAs and core-damage accidents. The staff discusses mitigation of accidents further in Section 2.4 of this chapter.

## 2.2 Accident-Resistance Requirements

Section 2.2 of Chapter 1 of the Evolutionary Requirements Document specifies the ALWR design characteristics that are intended to enhance accident resistance, such as emphasizing simplification, providing ample design margin, using the best available materials and water chemistry, using the best proven diagnostic monitoring techniques, and maintaining a negative overall power reactivity coefficient under all conditions. Improved design margin is attained by the use of a 15-percent fuel design margin; a larger reactor vessel, pressurizer, and steam generator secondary side; and sufficient margins to limiting conditions for operation and reactor trip setpoints.

The staff concludes that these design characteristic requirements are acceptable in principle. However, it will evaluate the acceptability of the implementation of each specific design characteristic during its review of an individual application for FDA/DC.

#### 2.3 Core-Damage-Prevention Requirements

## 2.3.1 General Requirements

The Evolutionary Requirements Document requires that the plant designer perform two types of analyses with respect to core-damage prevention (i.e., the LDB and the SMB analyses).

Section 2.3.1 of Chapter 1 specifies general requirements with regard to coredamage prevention. It requires the plant designer to identify a complete set of ALWR design-basis events and transients to be analyzed, taking credit for safety-grade equipment only; perform best-estimate analyses of the designbasis events to support the generation of plant operating procedures; and design the plant so as to (1) allow operators significant time to evaluate plant conditions; (2) minimize the potential for systems interactions; (3) require two separate and independent ac power connections to the grid to decrease the likelihood of a loss-of-offsite-power event; and (4) have the capability of achieving safe shutdown with safety-grade equipment only, assuming the most limiting single failure.

Because these requirements do not conflict with the Commission's regulations, the staff concludes that they are generally acceptable. However, EPRI's proposed implementation of these requirements is discussed throughout this chapter.

## Physical Security

Section 2.3.1.4 of Chapter 1 of the Evolutionary Requirements Document requires the plant to be so designed that the operator has significant time before taking any action needed to prevent core damage. In its letter dated March 1, 1991, the staff asked EPRI if this additional time might warrant, for these designs, relaxing the requirement of 10 CFR 73.55(d)(7)(ii) that the access control system be designed for rapid ingress to areas containing vital equipment. In its letter dated May 13, 1991, EPRI stated that the best interest of safety lies in maximizing the available time for diagnosing the failures, assembling personnel and equipment, and implementing recovery activities, and, therefore, the access control system should still provide for rapid emergency access to vital equipment. The staff concludes that this

section is not incompatible with NRC requirements related to physical security.

## Site Parameters

Among the ceneral requirements, Section 2.3.1.8 of Chapter 1 addresses plant siting. Table 1.2-6 lists the envelope of standard site design parameters, for which the requirements are discussed in greater detail in Section 4.5.2 of Chapter 1. These siting parameters are intended to cover most, but not all, potential sites for future ALWRs in the United States. As such, the Evolutionary Requirements Document requires that the plant designer review the conditions at the plant owner's site against the standard design siting parameters in order to assess the possible need for modifying any design parameter. Further, it requires that the final design parameters to be used for the particular site be approved in writing by the plant owner.

The results of the staff's review of the envelope of standard site design parameters are given in Section 4.5.2 of this charter. In the DSER for Chapter 1, the staff identified an open issue concerning worst-case site parameters. As for the final site design parameters to be used for any particular site, approval by the plant owner only is not sufficient. Approval by the NRC staff is also required. In addition, if one or more than one sitespecific design parameter exceeds the standard site design parameters at some potential nuclear plant site, the plant owner should conduct a plant-spec fic evaluation against these parameters and submit a detailed review to the staff for approval.

2.3.2 Licensing-Design-Basis Requirements

#### Selection of Transient/Accident Events

Section 2.3.2 and Table 1.2-1 of Chapter 1 of the Evolutionary Requirements Document specify a set of event initiators to be included in LDB analyses, and the corresponding frequency categories for these events. Section 2.3.2.2 specifies that the plant designer should review and identify any additional initiating events applicable to the specific advanced plant design considering its unique design features, but limits the types of events to those historically analyzed. It also specifies that the plant designer will document the basis for selecting the frequency category for each additional initiating event and will identify potential single equipment failures that could occur coincident with the initiating events. Section 2.3.2.3 specifies that these events by themselves cannot generate more serious incidents without other incidents occurring independently. Section 2.3.2.8 also specifies that acceptance criteria for fuel, reactor pressure boundary, and containment, and offsite dose consequence limits will be presented for those additional events identified.

The event frequency categorization in the Evolutionary Requirements Document is based on that of Regulatory Guide 1.70, "Standard Format and Content of Safety Analysis for Nuclear Power Plants," and the design requirements of American Nuclear Society (ANS) 18.2-1973, "Nuclear Safety Criteria for the Design of Stationary Pressurized Water Reactor Plants." These categories are: Condition II moderate-frequency events are those that may occur during a calendar year for a particular plant, Condition III infrequent events are those that may occur during the life of a particular plant, and Condition IV

limiting faults are those not expected to occur during the life of a plant but are postulated. NUREG-0800 (SRP) divides the events into anticipated operational occurrences (AOOs) and postulated accidents. Postulated a idents are limiting faults chosen as the design-basis accidents. A00s are defined in 10 CFR Part 50 as those conditions of normal operation and transients that are expected to occur one or more times during the life of a plant and, therefore, encompass the moderate-frequency and infrequent events. Chapter 15 of the SRP does not specify an infrequent-incident category but does specify specific acceptance criteria for those events that can be categorized as infrequent events. The event frequency categorization specified in the Evolutionary Requirements Document is, therefore, consistent with the Commission's current licensing approach.

In Section 3.3.A.3, Table 3-2, and Table 3-3 of the DSER for Chapter 1, the staff identified concerns regarding safety analysis events, event initiators, and frequency categorization. Because EPRI has extensively revised Chapter 1, Section 2.3.2 now addresses these topics. The staff concludes that the issues discussed in the DSER are superseded by the following discussion.

Section 2.3.2 of Chapter 1 of the Evolutionary Requirements Document states that the Plant Designer shall identify additional initiating events (other than those described in Table 1.2-1) applicable to the specific advanced plant design. The staff reviewed Section 2.3.2 assuming that this table was meant to be generally appropriate for ALWR designs, and it does appear that EPRI attempted to list all of those events that are typically analyzed for the current generation of plants. However, the staff concluies Table 1.2-1 was incomplete and that the following events should be add ...

Frequency Category Event Loss of all reactor coolant system flow MF (moderate frequency) Pressure regulatory failure MF MF Turbine trip without all bypass

In addition, because the Evolutionary Requirements Document does not present an actual design, the staff has not found sufficient justification to reclassify the following, events or accidents.

- Rod drop accident.
- Uncontrolled rod withdrawal at power-must continue to be reviewed as a moderate-frequency event. (Table 1.2, and 1.4.2)
- Inadvertent opening of a safety or relief valve--must continue to be reviewed as a moderate-frequency event. (Table 1.2-1, 1.4 and 6.1)

Selection of the appropriate set of plant events and, in some cases, their frequency classification are specific to the design of a plant. The staff concludes that each FDA/DC applicant must justify the selection of anticipated operational occurrences and accidents. Pending modifications to the Evolutionary Requirements Document as discussed above, the staff will evaluate the acceptability of the event initiators and associated frequency categorizations selected by the designer during its review of an individual application for FDA/DC. On the basis of the above discussion, the DSER issues are closed.

### Transient/Accident Acceptance Criteria

Table 1.2-2a of Chapter 1 specifies example reactor and fuel design limits and consequence analysis limits for moderate-frequency, infrequent, and limiting fault events. In Revision 3 of the Evolutionary Requirements Document, EPRI added requirements to maintain a coolable geometry and long-term cooling capability, as specified in 10 CFR 50.46, to Table 1.2-2a as part of the acceptance criteria for loss-of-coolant accidents (LOCAs).

Section 2.3.2.7 of Chapter 1 states that the plant designer will perform a consequence analysis for moderate-frequency and infrequent events with coincident single failures and specifies the acceptance criteria for these events, including limiting faults, as summarized in Table 1.2-2c. In Revision 3 of the Evolutionary Requirements Document, EPRI added a footnote to Table 1.2-2c to specify fuel cladding failure criteria for input to the radiological consequence analyses. That is, for the PWR, the fuel cladding failure criterion will be less than the 95/95 departure from nucleate boiling ratio limit, and for the BWR, it will be less than the minimum critical power ratio, except for (1) a LOCA event or (2) a fuel handling and cask drop event. For a LOCA, EPRI specifies that the vendor should use the source term as defined in Section 2.5.2 of Appendix B to Chapter 1, and for a fuel handling and cask drop event, the vendor should use the number of assemblies involved. These failure criteria are consistent with the SRP, except for the source term for LOCA consequence analysis. Table 1.2-2c also specifies limits based on 10 CFR Part 20 and Appendix I to 10 CFR Part 50 for moderate-frequency and infrequent events, and specifies that, for PWRs, the radiological consequences of infrequent events may exceed the guidelines of 10 CFR Part 20 but cannot be such that they interrupt or restrict public use of those areas beyond the exclusion areas. The staff concludes that the EPRI-proposed criteria are not specific enough to determine if they are consistent with the staff's review criteria. The plant designer should specify the exact acceptance criteria and identify deviations from those in the SRP, if any, and the bases for the deviations. The staff will address this matter during its review of an individual application for FDA/DC.

#### Anticipated Transients Without Scram Acceptance Criteria

Section 2.3.2.? of Chapter 1 specifies that analysis and acceptance criteria for events involving failures of multiple active components associated with anticipated transients without scram (ATWS) and station blackout will be in accordance with 10 CFR 50.62 and 10 CFR 50.63, respectively. However, 10 CFR 50.62 does not specify analysis and acceptance criteria, except for the prescriptive equipment design requirements that were based on analyses of the current generation of LWRs.

The staff concludes that EPRI has not provided adequate requirements in Chapter 1 for analyses for ATWS events. Therefore, the staff concludes that it will be necessary for designers of ALWRs to demonstrate that the bases for which the ATWS rule was developed remain valid for their proposed design, and, therefore, the ATWS rule requirements are still appropriate. Each evolutionary LWR designer must demonstrate that the new plants will not experience ATWS behavior unexpectably more severe than that experienced in current plants. The staff will evaluate this matter during its review of an individual application for FDA/DC.

In addition, 10 CFR 50.62 requires each BWP to have a standby liquid control system (SLCS) that is automatically initiated. Previously, Section 2.5.4 of Appendix B to Chapter 1 of the Evolutionary Requirements Document , quired the SLCS to be manually initiated. In its December 6, 1991, letter, EPRI stated that it had determined that automatic actuation of the SLCS was appropriate for evolutionary 'esigns and that it was modifying the Evolutionary Requirements Document to reflect that position. Revision 4 to the Evolutionary Requirements Document eliminated Section 2.5.4 of Appendix B to Chapter 1 and now required, in Chapter 5 Section 4.6.3.5.1 automatic initiation of SLCS. Therefore, this issue is closed. The staff also discusses this matter in the regulatory departure analysis in Appendix B of this chapter.

#### Station Blackout

In Revision 0 of the Evolutionary Requirements Document, station blackout was listed as a utility investment protection concern in Table 3-8. However, because of the importance of this issue, the staff concluded that it should be considered when the plant designer determines the most limiting event for analysis. In Section 2.3.2.2 of the revised Chapter 1, EPRI included a requirement that analysis and acceptance criteria for events involving failure of multiple active components associated with ATWS and station blackout be in accordance with 10 CFR 50.62 and 10 CFR 50.63, respectively. Also, in Table B.1-2 of Appendix B to Chapter 1, EPRI commits to meet the staff guidance in Regulatory Guide 1.155, "Station Blackout." The staff concludes that these provisions ensure that the plant designer will appropriately analyze the station blackout event in the + WR evolutionary plant designs. Therefore, this DSER open issue is closed. Additional staff evaluation of specific EPRI requirements pertaining to station 51 ckout in the ALWR is contained in Annex A of Appendix B to Chapter 1, Chapter 5, and Chapter 11 of this report.

#### 2.3.3 Safety-Margin-Basis Requirements

Section 2.3.3 of Chapter 1 of the Evolutionary Requirements Document specifies that the ALWR design will be such that no fuel damage is predicted to occur as a result of a postulated near-instantaneous pipe break with an area equivalent to up to 6 inches in diameter in the reactor coolant boundary. It also requires the plant designer to perform a best-estimate analysis.

Sections 2.3.3.2 and 2.3.3.3 of Chapter 1 state that the plant will be designed so that no fuel damage will occur for at least 2 hours after a sustained loss of feedwater with no operator action (PWR only) and the core will be able to withstand a loss of offsite and onsite ac power for at least & hours without fuel damage. The design criteria include requirements for an independent, safety-grade onsite ac power source and a non-safety-grade, alternate ac onsite power source. Section 2.3.3.5 requires that the operator have 30 minutes or more to act to prevent damage to equipment or plant conditions resulting in a significant outage due to an accident or transient.

Section 2.3.3.6 of Chapter 1 specifies that the probabilistic risk assessment (PRA), considering both internal and external events, will confirm a mean annual core damage frequency for the design of less than or equal to 1.0E-5 per reactor-year. The Evolutionary Requirements ocument also requires the plant designer to (1) fully define and document the technical basis of the PRA so that the plant owner can ensure that the reliability of risk-significant

systems, structures, and component is maintained; (2) develop the technical basis for a severe-accident management program to ensure the prevention and mitigation of core damage; and (3) use the plant-specific PRA and other relevant information to confirm that the plant design is compatible with the emergency procedure guidelines (LPGs) and severe-accident management program This is consistent with the requirement in 10 CFR 52.47 to perform a designspecific PRA in support of an application for FDA/DC. The use of the PRA for developing and confirming the severe-accident management program and EPGs is also consistent with the Commission's severe accident policy. In its staff requirements memorandum (SRM) dated June 26, 1990, the Commission approved the use of an overall me i frequency of a large release of radioactive materials to the environment from a reactor accident that is less than 1 in 1 million per year of reactor operation. Although the current regulations do not specify requirements in numerical terms of frequency of core ... mage, the Commission, in its June 15, 1990, SRM, pertaining to the implementation of the NRC's safety goals, stated that "a core damage probability of less than 1 in 10,000 per year of reactor operation appears to be a very useful subsidiary benchmark in making judgments about that portion of [the NRC's] regulations which are directed toward accident prevention."

The staff concludes that the EPRI-proposed core camage frequency of 1.0E-5 per reactor-year is more restrictive than the Commission's guidance and is, therefore, acceptable. However, it will not use this goal as a staff acceptance criteric.

## 2.4 Mitigation Requirements

Section 2.4 hapter 1 of the Evolutionary Requirements Document provides the design requirements for accident mitigation, including those necessary for licensing as well as those to provide protection ogainst severe accidents.

Section 2.4.1 of Chapter 1 provides the mitigation requirements for the licensing design basis, including requirements regarding the containment building and associated containment systems, site boundary dose criteria, source terms, and hydrogen control during degraded-core accidents. However, it does not include requirements for fission-product control or hydrogen control during design-basis LOCAs. The staff will evaluate this matter during its review of an individual application for FDA/DC to ensure that features for fission-product and hydrogen control during design-basis LOCAs are included in the design.

Section 2.4.2 of Chapter 1 provide: the mitigation requirements necessary to meet EPRI's safety margin basis, including design criter. for the containment building and associated systems, source terms, and hydrogen control.

## 2.5 Analysis Requirements and Acceptance Criteria

Section 2.5 of Chapter 1 of the Evolutionary Requirements Document specifies the EPRI-proposed requirements for the licensing-design-basis (LDB) and safety-margin-basis analyses. EPRI states that for the LDB analysis, NRCapproved methods will be used including (1) assumptions and limits that are based on actual physical conditions during the accident or transient being analyzed with conservatism constitent with regulatory requirements and (2) acceptance criteria in accordance with NRC requirements. Analysis techniques will be proven through previous use. If changes are made to

existing techniques or new techniques are proposed, they will be identified and justified. Section 2.5.2.5 states that for events considered in the LDB analysis, acceptance criteria will be in accordance with the NRC requirements. In addition, Section 2.1 requires that only safety-grade equipment be assumed available in the LDB accident analyses, except for a limited number of multiple-failure events such as ATWS and station blackout.

These LL8 requirements are consistent with NRC deterministic licensing analysis requirements. However, 10 CFR 52.47(b)(2)(i) requires that the performance of each safety feature and interdependent effects among the safety features of the ALWR designs be demonstrated and found acceptable by analysis, appropriate test programs, experience, or a combination thereof. It also requires sufficient data on the safety features of the design to assess the analytical tools used for safety analyses over a sufficient range of normal operating conditions and specified accident sequences, including equilibrium core conditions. The staff will evaluate the acceptability of the analytical codes and methodologies used for safety analyses as well as validation data, including test plans and facilities, during its review of an individual application for FDA/DC.

#### 3 PERFORMANCE DESIGN REQUIREMENTS

#### 3.1 Introduction

Section 3 of Chapter 1 of the Evolutionary Requirements Document provides design criteria that are not related primarily to safety or investment protection. This section includes EPRI proposed requirements related to plant size, life, maneuvering capability, event transient response, and event response times. Design criteria for the core, radioactive waste, and onsite radiation exposure are also provided. The staff reviewed only those items that fall under the purview of its safety review.

#### 3.2 Plant Size

EPRI states that the Evolutionary Requirements Document applies to designs extending up to 1350 MWa per unit.

## 3.3 Plant Design Life

Section 3.3 of Chapter 1 of the Evolutionary Requirements Document states that the plant will be designed to operate for 60 years without the need for an extended refurbishment outage. In addition, the plant will be designed to permit expeditious replacement of components because of obsolescence and failure over a lifetime of 60 years.

As stated in SECY-89-013, "Design Requirements Related to the Evolutionary Advanced Light Water Reactors," the staff will review the ALWR designs for a 60-year life rotwithstanding the fact that a 40-year license term limitation is specified in the Atomic Energy Act and NRC's regulations. It is the applicants' responsibility to identify the components and systems that are affected. The staff will address plant life during its review of an individual application for FDA/DC. These applications will have to provide information and programs to support design life and the staff's review of such issues as fatigue, corrosion, and thermal aging. This item is discussed further by the staff in Section 4.7.3 of this chapter.

### 3.4 Maneuvering and Response to Grid Demands

Section 3.4 of Chapter 1 of the Evolutionary Requirements Document provides requirements for load following, frequency control, grid breakup, and load rejection.

## 3.5 Event Response Capability

Section 3.5 of Chapter 1 of the Evolutionary Requirements Document specifies the requirements pertaining to the capabilities to cope with various events.

Section 3.5.1 of Chapter 1 specifies that the plant will be designed to be capable of starting from cold shutdown and going to hot standby at full pressure and temperature in 24 hours. Similarly, EPRI requires that the reactor be capable of shutdown from the reactor critical stage at full temperature and pressure to 140 °F in 24 hours. Section 3.5.2 of Chapter 1 requires the PWR plant to be capable of operating at reduced power with a secured coolant pump to enhance the availability of the plant and to reduce reactor trips. Appendix B to Chapter 1 indicates EPRI's commitment to comply with Generic Letter (GL) 86-09, "Technical Resolution of Generic Issue No. B-59 - (N-1) Loop Operation in BWRs and PWRs." GL 86-09 states that (N-1) loop operation is acceptable provided acceptable evaluation results are shown for certain plant-specific design characteristics, such as the impact of the down loop on instrumentation and control systems, human factors, operational systems, safety systems, status of valves, core-flow distribution, and potential for cold water reactivity insertion. Since these characteristics are highly dependent on the specific design of the plant, acceptability of operation with one secured reactor coolant pump is subject to plant-specific evaluation to address the concerns delineated in GL 86-09. The staff will evaluate that analysis during its review of an individual application for FDA/DC.

Section 3.5.3 of Chapter 1 requires that the plant responses to review trips, which are not complicated by failures beyond those that caused the trip, do not result (1) in the initiation of the emergency core cooling system, the primary system safety valve, or the emergency feedwater system (PWRs only) and (2) in the uncovering of the pressurizer heaters. Section 3.5.4 requires the plant to be capable of a turbine trip from 40 percent or less (BWR) and 100 percent or less (PWR) of the rated power without reactor trip and the lifting of the main steam safety valves. Section 3.5.5 requires that the loss of a running main feedwater or condensate pump while at full power not result in a reactor trip. Section 3.5.6 requires that rod insertions caused by single failures not result in a reactor trip and that the plant be capable of continued operation at reduced power. The staff considers these design objectives as important defense-in-depth goals. Each of these requirements is acceptable provided the designer performs design-specific analyses to demonstrate that the specified design limits are met. The staff will evaluate these analyses during its review of an individual application for FDA/DC.

Under the ading "Table 3-6" of the DSER for Chapter 1, the staff s ated that EPRI had is tified certain plant performance capabilities involving step and ramp power changes and inadvertent control rod insertion without a reactor trip. Although such activities may not impose significant challenges to fuel integrity, the staff requires that an analysis be provided to confirm that this is the case in specific plant designs. The staff will evaluate this matter during its review of an individual application for FDA/DC.

#### 3.6 Core Performance

Section 3.6 of Chapter 1 of the Evolutionary Requirements Document requires that the ALWR core be designed for up to a 24-month fuel cycle and that fuel mechanical designs have a peak bundle burnup of at least 45,000 and 55,000 megawatt-days per metric ton of uranium (MWD/MTU) for BWRs and PWRs, respectively. These minimum fuel burnup requirements are inconsistent with the EPRI-proposed requirements of 50,000 and 60,000 MWD/MTU specified in Sections 4.2.2.2 and 7.2.2.2 of Chapter 4 for BWRs and PWRs, respectively. In addition, although these values are inconsistent with each other, they are greater than NRC-approved fuel burnup levels. To support this high fuel burnup operation, each ALWR design application will need to include sufficient high fuel burnup data to demonstrate fuel integrity in the areas of fission gas release, cladding corrosion due to oxidation and hydriding, and reduction in cladding material strength.

In addition, Section 4.2.6.2 of Chapter 4 requires that the BWR control blades used for maximum core insertion be designed with a minimum exposure capability of 4.0E+21 neutrons/m<sup>2</sup> (nvt) with a target of 8.0E+21 nvt, and that the blades not used for maximum core insertion be designed for an operating life of 13 or 20 reactor full-power years (RFPYs), which may be selected by the plant owner. Section 7.2.3 of Chapter 4 requires the PWR control rod assemblies to be designed for a minimum operating lifetime of 15 RFPYs with an objective of 20 RFPYs. These requirements are beyond the operating experience data of the current LWRs. To support the desired extended operating life of the control blades and control rod assemblies, each ALWR design application will need to include sufficient performance data to demonstrate that irradiation effects, including material hardening, absorber depletion, and swelling, will not impair structural integrity.

#### 3.7 Radioactive Waste Disposal

Section 3.7 of Chapter 1 of the Evolutionary Requirements Document specifies the onsite storage capacity for high- and low-level radioactive waste. See Chapter 12 of this report for the staff's evaluation of EPRI's requirements for radioactive waste processing and handling.

#### 3.8 Occupational Radiation Exposure

In 1990, the average annual occupational exposure dose for U.S. nuclear power plants was approximately 340 person-rem. EPRI estimates that the ALWR can be operated so that this dose is less than 100 person-rem/year averaged over the life of the plant. To meet this ambitious goal, the Evolutionary Requirements Document lists several dose-reduction actions that should be implemented in the ALWR. EPRI states that these design changes will eliminate much of the dose incurred during nonroutine maintenance work. Improved chemistry control and selection of materials will result in reduced radiation fields, and the use of robotics will further reduce personnel exposures. These dose-reduction actions are in compliance with the guidelines of Regulatory Guide 8.8, "Information Relevant to Ensuring That Occupational Radiation Exposures at Nuclear Power Stations Will Be as Low as Is Reasonably Achievable," and are acceptable.

## 4 STRUCTURAL DESIGN BASES

#### 4.1 Introduction

Section 4 of Chapter 1 of the Evolutionary Requirements Document includes the general structural design criteria and requirements applicable to all buildings, structures, systems, and equipment. It also addresses the passive structural requirements and requirements to ensure active equipment functions. Requirements for a unified system of classifying the structures, systems, and equipment with respect to function and structural integrity are established. as are codes and standards and acceptance criteria. Design codes and load combinations, as well as the required measures to mitigate the effects of inplant hazards, are also established to ensure that equipment will function under specified dynamic and environmental conditions.

## 4.2 Relationships to Design-Basis Events

Section 4.2 of Chapter 1 of the Evolutionary Requirements Document states that the design-basis events specified in Section 2 of Chapter 1 will be used by the plant designer in implementing the design criteria in Section 4.

#### 4.3 Classification Requirements

Section 4.3 of Chapter 1 of the state Requirements Document provides general requirements to b the sector designer for safety and seismic classification of structure, set and a covingment in the plant.

## 4.3.1 Safety Classification

General Design Criterion (GDC) 1, "Quality Standards and Records," of 10 CFR Part 50, Appendix A, requires that nuclear power plant structures, systems, and components important to safety be designed, fabricated, erected, and tested to quality standards commensurate with the importance of the safety function to be performed. Regulatory Guide (RG) 1.26, "Quality Group Classification and Standards for Water-, Steam-, and Radioactive-Waste-Containing Components of Nuclear Power Plants," is the principal document used by the staff in its review of this subject. However, the cvolutionary Requirements Document proposes the use of American National Standards Institute/American Nuclear Society (ANSI/ANS) 51.1, "Nuclear Safoty Criteria for the Design of Stationary PWRs," and 52.1, "Nuclear Safety Criteria for the Design of Stationary BWRs," as an alternative way of complying with RG 1.26. As discussed in Section 4.3.A of the DSER for Chapter 1, the staff has not com-pletely endorsed these two industry standards. The standard safety analysis reports (SSARs) for Combustion Engineering. Inc.'s System 80+ and General Electric Company's advanced BWR (ABWR) Revolutionary plants also reference these standards. The staff concludes that it will complete its review of these two standard plant SSARs without these endorsements. Therefore, the staff's endorsement of these two standards is not necessary for the Evolutionary Requirements Document and any future evolutionary plant SSAR will be reviewed in a manner similar to the reviews being conducted for the System 80+ and the ABWR. Therefore, the staff will evaluate this matter during its review of an individual application for FDA/DC and this DSER open issue is closed.

#### 4.3.2 Seismic Classification

Section 4.3.2 of Chapter 1 of the Evolutionary Requirements Document states that each of the plant structures, systems, and components will be designated as seismic Category I (C-I), seismic Category II (C-II), or non-seismic (NS). GDC 2, "Pesign Bases for Protection Against Natural Phenomena," in part, requires that nuclear plant structures, systems, and components important to safety be designed to withstand the effects of earthquakes without loss of capability to perform their safety function. Such items are classified as seismic Category I. RG 1.29, "Seismic Design Classification," is the principal document used by the staff in its review of this subject. EPRI provides a commitment in Table B.1-2 of Appendix B to Chapter 1 to comply with RG 1.29 except for the optimization subject, BWR main steamline isolation valves and leakage control system. The staff's position on this cptimization subject relative to seismic classification is provided in Section 2.3.1 of Appendix B of this chapter.

Section 4.3.2 of Chapter 1 defines the seismic Category I items as including all structures, systems, and components whose safety class is SC-2 and SC-3 as well as spent fuel pool structures, including all fuel racks. EPRI states that the seismic Category II classification will be applied to all plant structures, systems, and equipment that will not perform a nuclear safety function and whose continued function will not be required, but whose structural failure or interaction could degrade the functioning of a seismic Category I structure, system, or component to an unacceptable safety level, or could result in incapacitating injury to occupants of the control room. Nonseismic structures are those that do not fall into the seismic Category I and II definitions specified in Sections 4.3.2.1 and 4.3.2.2.1 of Chapter 1 of the Evolutionary Requirements Document.

In its letter dated April 24, 1991, the staff requested EPRI to clarify the relationship between "seismic classification" and "safety classification" and confirm the use of the term "equipment" instead of the commonly used term "components." In its July 2, 1991, response, EPRI noted that no specific relationship exists between seismic and safety classifications except that all SC-1, SC-2, and SC-3 items need to be designed to meet the C-I requirements. In addition, EPRI confirmed that the term "equipment" is synonymous with "components." The staff concludes that these responses are acceptable.

1.4-2

The following table summarizes the staff's understanding of the ALWR seismic classifications and their relationship to safety classifications.

ALWR Seismic Classification	Current Seismic Classification	Safety Classification	Requirements
0-1	C-1	SC-1, 2, and 3	Functional capability & structural integrity
C-II	Non-C-I	NNS*	Structural integrity (failure of structures will not affect the functions of SC-1, 2, and 3 items)
NS	Non-C-1	NNS	None

\*NNS = non-nuclear safety.

## Physical Security Considerations

Section 4.3.2 of Chapter 1 of the Evolutionary Requirements Document states that seismic classification will be consistent with RG 1.29. In NRC Review Guideline 17, RG 1.29 is used as a reference for determining which equipment should be protected as vital equipment in the sense of 10 CFR 73.2. In its letter of May 13, 1991, EPRI stated that this linkage between equipment designated as seismic and equipment designated as vital continues to be appropriate for this design. The staff concludes that this is consistent with current staff guidelines and is, therefore, acceptable.

## Seismic Category I

Seismic Category I items include all structures, systems, and components (SSCs) designated as SC-1, SC-2, or SC-3. Seismic Category I SSCs will be designed to withstand the effects of a safe shutdown earthquake and to maintain the specified design functions.

## Seismic Category II

In Section 4.3.8 of the DSER for Chapter 1, the staff stated that EPRI had agreed that a quality assurance program in accordance with the applicable parts of Appendix B to 10 CFR Part 50 should be applied to seismic Category II items (i.e., structures, systems, and equipment) that will perform no safety function and whose continued operation is not required, but whose structural failure or interaction would degrade the functioning of a seismic Category I structure, system, or component to an unacceptable level. The staff also requested a specific commitment to Positions C.2 and C.4 of RG 1.29 for seismic Categorv II items. This commitment was added to the requirements portion of Section 4.3.2 in Revision 1 of the Evolutionary Requirements Document and is acceptable. Therefore, this DSER open issue is closed.

However, in the rationale portion of Section 4.3.2.2.1 of Chapter 1, EPRI states that extensive use of seismic qualification by experience, as addressed in Section 4.8, should expedite design efforts for seismic Category II structures, systems, and equipment. The staff's evaluation of the use of

seismic qualification by experience is provided in Section 4.8.1 of this chapter for equipment and Appendix B of this chapter for piping. In addition, the type of analyses required to satisfy Position C.2 of RG 1.29 is discussed below.

In its letter dated May 17, 1991, the staff requested that a requirement be added to Section 4.3.2.2 of Chapter 1 to state that if non-seismic Category 1 systems cannot be isolated from adjacent seismic Category I systems, the nonseismic system should be analyzed to the same criteria as those that are applicable to the seismic Category I system. This commitment would satisfy the guidelines in Item II.h of SRP Section 3.7.3, "Seismic Subsystem Analyand Item II.k of SRP Section 3.9.2, "Dynamic Testing and Analysis of \$15. Systems, Components, and Equipment." In its letter dated August 1, 1991, EPRI stated that since there was a commitment in Table B.1-2 of Appendix B to Chapter 1 to comply with both of the above SRP sections, it was not necessary to include them as a specific requirement in this section. On the basis of the commitment in Appendix B, the staff concludes that applicable non-seismic Category I structures and systems (seismic Category II) will be analyzed using the same type of dynamic seismic analysis methodology as that used for seismic Category I structures and systems. This commitment is consistent with SRP Sections 3.7.3 and 3.9.2 and is, therefore, acceptable.

Section 4.3.2.2.2 of Chapter 1 of the Evolutionary Requirements Document states that the plant designer will select appropriate ductility factors for the seismic design of C-II components to take credit for realistic amounts of energy dissipated in such items during a seismic event. In the DSER for Chapter 1, the staff stated that limits on the ductility factors should be correlated with research results and should be reviewed and approved by the NRC. As discussed in that DSER, EPRI agreed to revise the Evolutionary Requirements Document to require the use of appropriate "approved" ductility factors. Ine rationale portion of this section includes the statement: "The seismic ductility factors and ductility limits selected are anticipated to consider the results of research which is documented by the time of the ALWR, and to provide significant improvement over current practice." The staff concludes that this statement resolves its concern and is acceptable. Therefore, this DSER confirmatory issue is closed.

#### Non-Seismic

Revision 0 of Section 4.3.8.3 of Chapter 1 did not provide specific requirements concerning the building codes to be used. This was identified as an open issue in the DSER for Chapter 1. In Revision 3 of the Evolutionary Requirements Document, Section 4.3.2.3 of Chapter 1 requires that non-seismic (NS) building structures be designed to the Zone 2A specification in the Uniform Building Code (UBC) with an importance factor of 1.25 assigned to the structures. In its letter dated April 24, 1991, the staff questioned the use of Zone 2A, which according to the UBC seismic zone map is lower than the designation for many regions in the United States. In its July 2, 1991, response, EPRI noted that the UBC Zone 2A specification is intended solely to provide a high degree of investment protection for NS items. However, since many regions in the United States are designated as UBC seitmic Zone 2B or higher, the use of the Zone 2A specification in these zones may not be adequate for the design of NS items. However, as the staff has no regulatory requirements or criteria that apply to non-seismic structures, the staff's

review of an application for FDA/DC in this area will be limited to a determination that a particular structure is correctly characterized as non-seismic. On the basis of the above discussion, this DSER open issue is closed.

#### 4.4 Codes and Standards

Section 4.4 of Chapter 1 of the Evolutionary Requirements Document provides EPRI-proposed requirements relative to the applicability of major design and construction codes, industry standards, and regulatory positions to the ALWR evolutionary plant design. Tables 1.4-1 and 1.4-3 in Chapter 1 list major structural design and construction codes that are applicable to the ALWR. Table 1.4-2 lists the industry technical standards to be used.

In the DSER for Chapter 1, the staff stated that several of these standards had not been endorsed by the staff and should not be used as the basis for plant design and construction. In Revision 3 of the Evolutionary Requirements Document, EPRJ revised Section 4.4 to state that the use of applicable structural design and construction codes and industry standards that conflict with NRC positions will be resolved by the plant designer with the NRC and the resolution will be fully documented. The intent of this requirement is to ensure that the staff's review of applications for FDA/DC will be conducted using acceptance criteria that include the codes and standards most recently approved by the NRC. The staff concludes that this commitment is acceptable. Therefore, this portion of the DSER confirmatory issue is closed. The staff will evaluate this matter during its review of an individual poplication for FDA/DC or a combined license.

See the regulatory departure analysis in Appendix B of this chapter for additional staff discussion on the use of industry codes and standards.

## 4.4.1 Major Design and Construction Codes

In Revision 0 of the rationale portion of Section 4.4 of Chapter 1 of the Evolutionary Requirements Document, EPRI stated that it anticipated that the applicable edition of codes and standards will be that in effect approximately 42 months before start of construction. In the DSER for Chapter 1, the staff stated that this criterion deviated from the requirement of 10 CFR 50.55(a) that the edition and addenda of the American Society of Mechanical Engineers Boiler and Pressure Vessel Code (ASME Code) applied to the construction of components be determined by the provisions of Section NCA-1140 of ASME Code, Section III, incorporated by reference and rule. Paragraph NCA-1140(a)(2) states that, in no case, will the code edition and addenda dates established in the design specifications be earlier than 36 months before the date the plant construction permit is docketed. In response to this concern, EPRI deleted the 42-month criterion from the rationale portion in Revision 3 of the Evolutionary Requirements Document. In addition, in response to a concern discussed in the section entitled "Table 4-1" of the DSER for Chapter 1, EPRI added a note to Table 1.4-1 in Chapter 1 to require that all ASME Code, Section III items also satisfy the requirements of ASME Code, Section III, Divisions 1 and 2, Subsection NCA. This change complies with the 10 CFR 50.55(a) requirement and is acceptable. Therefore, this portion of the DSER confirmatory issue is closed.

In its letter dated May 17, 1991, the staff requested that EPRI revise Item g under "ASME Code" in Table 1.4-3 of Chapter 1 to add a provision that only those code cases that are approved or conditionally approved in RGs 1.84 ("Design and Fabrication Code Case Acceptability - ASME Section III, Division 1"), 1.85 ("Materials Code Case Acceptability - ASME Section III, Division 1"), or 1.147 ("Inservice Inspection Code Case Acceptability - ASME Section XI, Division 1") are available to the plant designer. In its letter dated August 1, 1991, EPRI stated that its modification of the Evolutionary Requirements Document specifying that applicable structural design and construction codes and industry standards that conflict with NRC positions will be resolved by the plant designer with the NRC (discussed in Section 4.4 above) should obviate the need for such a provision. Therefore, the staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

## 4.4.2 Industry Technical Standards

Section 4.4.2 of Chapter 1 of the Evolutionary Requirements Document references Table 1.4-2, which lists the industry technical standards that will be applicable to the ALWR evolutionary plant. In the section entitled "Table 4-2" of the DSER for Chapter 1, the staff stated that since some of the standards in this table, such as American Society of Civil Engineers (ASCE) 4-86, had not been endorsed by the NRC, EPRI had agreed to revise the introduction to Section 4.4 in Chapter 1 to address this concern. As discussed in Section 4.4 above, EPRI's response is acceptable and this DSER confirmatory issue is closed.

## 4.4.3 Regulatory Positions

In Section 4.4.3 and Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI identifies NRC regulatory guides and SRP sections that will be applicable to the structural design bases of the ALWR evolutionary plants. EPRI lists the applicable guidance in Table B.1-2 of Appendix B to Chapter 1. In Section 4.4.3, EPRI proposes several technical positions that, in its opinion, require exceptions to some of the regulatory positions. These issues are discussed below.

#### Implementation of Leak-Before-Break Criteria

This issue is discussed by the staff in Section 4.5.5 and Appendix B of this chapter.

#### Damping Values

Section 4.4.3.3.3 of Chapter 1 of the Evolutionary Requirements Document states that the plant designer will use approved realistic damping values in the analyses of buildings, structures, and equipment. The rationale portion of this section correctly states that RG 1.84 permits the use of damping values for piping systems in accordance with ASME Code Case N-411. The staff considers this section to be acceptable. However, the only staff-approved damping values for the design of buildings and structures are in RG 1.61, "Damping Valves for Seismic Design of Nuclear Power Plants," and RG 1.84 conditionally approves the use of Code Case N-411 for piping systems. This issue is also discussed by the staff in Section 4.7.3 of this chapter.

## Elimination of Operating-Basis Earthquake From Design

In its letter dated August 1, 1991, regarding the use of a single damping value for both the operating-basis earthquake (OBE) and the safe shutdown earthquake (SSE) in the analyses of buildings, structures, and equipment, EPRI stated that it had deleted the OPE as an earthquake to be considered in the design process in Revision 2 of Section 4.4.3.3.3 of Chapter 1 and Section 2.1.1 of Appendix B to Chapter 1.

In SECY-90-016 and the draft policy paper on evolutionary and passive plants dated February 27, 1992, the staff stated that the OBE should not control the design of safety-related systems. As a result, the staff is involved in the rulemaking process for Appendix A to 10 CFR Part 100 to decouple the OBE from the SSE in siting considerations. The staff is also evaluating the possibility of redefining the OBE in order to satisfy the OBE's function without explicitly analyzing responses. This change would diminish the role of the OBE in design by establishing a level that, if exceeded, would require that the plant be shut down for inspection activities. The staff agrees in principle with EPRI regarding the deletion of the OBE from plant design. However, certain issues related to the treatment of earthquake cycles for evaluations of piping and equipment fatigue, seismic anchor motion effects, criteria on postulated pipe-break locations, and design of concrete structures need to be adequately resolved as a direct consequence of eliminating the OBE from design. The elimination of the OBE from design would require all current OBE design-related checks to be performed for the SSE. The staff is developing alternatives with the industry to revise the codes and standards when design-related checks are based on the OBE. Resolution of these issues may result in staff recommendations for changes in applicable ASME Code, Section III rules. Therefore, the staff concludes that the elimination of the OBE from design is acceptable. However, the details of how current OBErelated design checks will be performed using the SSE will be resolved between industry and the staff through the appropriate code-related activities or supplemental regulatory guidance. The supplemental regulatory guidance could be in the form of revised SRP sections or the ITAAC (inspections, tests, analyses, and acceptance criteria). The elimination of the OBE from design would require an exemption from the current regulations until the final rulemaking pertaining to Appendix A to 10 CFR Part 100 is approved. In the interim, the specification of the OBE ground motion remains an option for ALWR FDA/DC applicants. The staff will evaluate the applications of those applicants that opt to eliminate the OBE from design in accordance with the forthcoming supplemental regulatory guidance mentioned above. This is also discussed by the staff in Appendix B to this chaoter.

## Use of the Independent Support Motion Response Spectrum Analysis Method

In Section 4.4.C(3) of the DSER for Chapter 1, the staff stated that it did not accept the damping values in ASME Code Case N-411 for analyses that use the independent support motion response spectrum methodology. This position is reflected in the conditional acceptance of Code Case N-411 in RG 1.84, Revision 25, dated May 1988. The staff is currently accepting the above analysis technique only if the independent support motion method is defined as discussed below. In Revision 3 of the Evolutionary Requirements Document, Section 4.4.3.3.4 of Chapter 1 FPRI states that the plant designer may use approved independent support motion response spectrum analyses techniques as a basis for seismic design and identifies this use as an exception to SRP Section 3.9.2. The staff's position regarding a definition of "approved the use" is that this method is only acceptable when used in accordance with information and recommendations in Sections 2.3 and 2.4 of NUREG-1061, "Report of the U.S. NRC Piping Review Committee," Volume 4. As a part of this position, a support group is defined by supports that have the same time history input. This usually means all supports located on the same floor (or portions of a floor) of a structure. The staff concludes that Sections 4.4.3.3.4 and 4.7.3.4 of Chapter 1 should be revised to provide this commitment. In the interim, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, the staff concludes that this issue is closed.

## Use of Spectral Shifting Analyses as an Alternative to Spectrum Broadening

Sections 4.4.3.3.5 and 4.7.3.3 of Chapter 1 and Section 2.1.1 of Appendix B to Chapter 1 state that the plant designer may use spectral shifting analyses in lieu of spectrum broadening and identifies this use as an exception to SRP Section 3.9.2 and RG 1.122, "Development of Floor Design Response Spectra for Seismic Design of Floor-Supported Equipment or Components." The spectral shifting procedures are described in ASME Code Case N-397. As stated by EPRI in the rationale portion of Section 4.4.3.3.5 of Chapter 1, the staff has conditionally accepted Code Case N-397 in RG 1.84. The rationale further correctly states that Code Case N-397 has been annulled and that its contents have been included in ASME Code, Section III, Subsection NCA, Appendix N. In its letter dated August 1, 1991, EPRI stated that including the above information in the rationale portion for the above sections of the Evolutionary Requirements Document was sufficient and that a change to the requirement portion of these sections was unnecessary. The staff does not agree that the plant designer will necessarily treat the rationale as a requirement. As discussed in Section 4.7.3 of this chapter, the staff has not endorsed Appendix N. Therefore, if the plant designer opts to use the spectral shifting procedures in either Code Case N-397 or Appendix N, the staff's position is that the condition in RG 1.84 will apply; that is, the staff will review the use of these procedures on a case-by-case basis. This position should be added to the requirement portion of Sections 4.4.3.3.5 and 4.7.3.3 of Chapter 1 in addition to Section 2.1.1 in Appendix B to Chapter 1. In the interim, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, the staff concludes .nat this issue is closed.

#### Analysis of Vibratory Loads With Significant High-Frequency Input

In Section 4.4.C(6) of the DSER for Chapter 1, the staff stated that in analyses of vibratory loads with high-frequency input, if the plant designer combines high-frequency results algebraically, this deviation from RG 1.92, "Combining Model Response and Spatial Components in Seismic Response Analysis," will have to be evaluated on a case-by-case basis. In Revision 3 of the Evolutionary Requirements Document, E°RI added a qualification to the rationale portion of Sections 4.4.3.3.6 and 4.7.3.3 of Chapter 1, and to Section 2.1.1 of Appendix B to Chapter 1, that indicates that for analyses of vibratory loads with high-frequency input, if high-frequency modal results are

combined by algebraic combination, the staff will review the methodology on a case-by-case basis. However, the staff does not agree that the plant designer will necessarily treat the rationale as a requirement. Therefore, this same qualification should be added to the requirement portion of Sections 4.4.3.3.6 and 4.7.3.3 in addition to Section 2.1.1 of Appendix B to Chapter 1. In the interim, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, the staff concludes that this DSER open issue is closed.

In its letter dated May 17, 1991, the staff requested that EPRI revise the requirement portion of Sections 4.4.3.3.6 and 4.7.3.12 of C apter 1 and Section 2.1.1.2 of Appendix B to Chapter 1 to require that / nonlinear analyses are used to account for gaps between pipes and piping supports subjected to vibratory loads with high-frequency input, such analyses must be submitted to the staff for review and approval before they are used. In its response dated August 1, 1991, EPRI stated that since this procedure was identified as an exception to SRP Section 3.9.2 in the requirement portion of the above sections, no further changes were required. The staff does not agree that merely identifying this procedure as an exception to the SRP is sufficient for a requirement. The staff position applies to the requirement portion of Sections 4.4.3.3.6 and 4.7.3.12 and to Section 2.1.1.2 of Appendix B to Chapter 1. Therefore, the staff will review individual applications for FDA/DC in accordance with the above position.

#### Seismic Qualification of Equipment Using Experience Data

This issue is discussed by the staff in Section 4.8.1 of this chapter.

## Design-Basis Tornado

The Evolutionary Requirements Document requires the use of ANSI/ANS 2.3 to define tornado effects based on an exceedance probability of 1.0E-6 per year. This represents an exception to RG 1.76, "Design Basis Tornado for Nuclear Power Plants." This issue is discussed by the staff in Section 4.5.2.5 and Appendix B of this chapter.

## 4.5 Design Loads and Conditions

## 4.5.1 Introduction

Section 4.5.1 of Chapter 1 of the Evolutionary Requirements Document provides general requirements for loads and conditions including natural phenomena, site proximity man-made hazards, plant operating loads, and in-plant hazards. Section 4.5.1.2 of Chapter 1 states that, on a case-by-case basis, the plant designer may, with the approval of the NRC, develop quantitative mechanistic design loads and combinations directly from design-basis events, using probabilistic methodology. The staff concludes that this is acceptable. However, as discussed in Section 4.5 of the DSER for Chapter 1, the staff is not accepting a probabilistic approach as a basis for changing existing loads and/or loading combinations, and the loading combinations recommended in SRP Sections 3.7, 3.8, and 3.9 remain valid. The staff will address this issue during its review of an individual application for FDA/DC.

#### 4.5.2 Natural Phenomena

Section 4.5.2 of Chapter 1 delineates requirements for considering the effects of the natural phenomena listed in Table 1.2-6 of Chapter 1, including wind loadings, hydrology, eology and foundation conditions, earthquakes, tornadoes, and volcanic activities. This table provides the envelope of plant design parameters associated with these natural phenomena.

#### Wind Loadings

Revision 0 of Chapter 1 of the Evolutionary Requirements Document specified basic wind speeds of 110 mph (based on a 50-year recurrence interval) and 130 mph (based on a 100-year recurrence interval) for non-safety-related and safety-related structures, respectively. In the DSER for Chapter 1, the staff found these criteria acceptable; however, in Revision 3, EPRI changed the accepted criteria. Revision 3 specifies the same basic wind speed of 110 mph (based on a 50-year recurrence interval); however, this basic wind speed is to be scaled by an importance factor (as defined in ANSI A58.1-1982, "Minimum Design Loadings for Buildings and Other Structures,") of 1.0 and 1.11 for non-safety-related and safety-related structures, respectively.

The use of importance factor 1.11 for adjusting the recurrence interval from 50 to 100 years is suitable for the design of safety-related structures because the use of an importance factor of 1.11 to calculate the wind speed for a 100-year recurrence interval is equivalent to the guidance in SRP Section 3.3.1, "Wind Loadings." For non-safety-related structures, the use of a 1.0 importance factor implies that a 50-year recurrence interval is suitable. However, the staff's interpretation of ANSI A58.1-1982 is that an extreme wind associated with a 50-year recurrence interval is suitable to calculate the wind speed only for Categories 1 and IV structures. Since nonsafety-related structures in an ALWR plant are more important than Category IV structures, the staff concludes that both safety-related structures and nonsafety-related structures whose failure could have an adverse impact on safety-related structures should be designed for an extreme wind associated with a 100-year recurrence interval. The importance factor of 1.0 is not acceptable for non-safety-related structures that are important to safety (e.g., turbine building). EPRI has not provided adequate justification for its position. Therefore, the staff will address this item during its review of an individual application for FDA/DC.

## Hydrology

Table 1.2-6 of Chapter 1 provides requirements for maximum ground water level, maximum flood level, and precipitation as follows:

 Maximum ground water level - The Evolutionary Requirements Document requires the maximum ground water level to be 2 feet below grade. This requirement is not acceptable; the maximum ground water level should be at grade. EPRI has not provided adequate justification for its position. Therefore, the staff will address this item during its review of an individual application for FDA/DC.
- Maximum flood (tsunami) level Revision 0 of the Evolutionary Requirements Document proposed that the maximum flood level be in the range of 1 foot below to 26 feet above grade. Although in the DSER for Chapter 1, the staff concluded that this design parameter was consistent with regulatory requirements, it recommended that the Evolutionary Requirements Document specify only the upper limit, which, assuming the design level includes static water level plus wind-induced waves, should be plant grade or lower. In response, EPRI changed Table 1.2-6 to specify the maximum flood level as 1 foot below plant grade. The staff concludes that this requirement is acceptable.
- Precipitation (for roof design) Revision 0 of the Evolutionary Requirements Document specified a maximum rainfall rate of 10 in./hour and a maximum snow load of 50 pounds per square foot (psf). In the DSER for Chapter 1, the staff stated that the rainfall rate of 10 in./hour was much too low, since the probable maximum precipitation (PMP) in a 5-minute interval over 1 mi<sup>2</sup> is 6.3 inches in the Great Lakes area and 6.2 inches along the Gulf Coast. Revision 1 specifies a higher 1-mi<sup>2</sup>, 1-hour PMP of 19.4 inches, together with a 1-mi<sup>2</sup>, 5-minute PMP of 6.2 inches. The 5-minute PMP value appears reasonable; however, it might exclude a number of sites in the Great Lakes area. The staff concludes that the plant designer should use the SRP guidelines and relevant RGs for developing an adequate structural and flood-prevention design basis for the PMP. The staff will address this item during its review of an individual application for FDA/DC. Therefore, this portion of the DSER open issue is closed.

In the DSER for Chapter 1, the staff stated that the 50-psf snow load might limit sites to below 38° north latitude in some regions, citing, for example, the 69-psf snow load and 72-psf snow plus ice load that were used in the design of the Beaver Valley plant. However, Revision 3 retained the 50-psf snow load, without addressing the staff's concern. The staff concludes that this snow load is unacceptable and that the plant designer should use the guidelines for snow load as specified in American Society of Civil Engineers 7-88 (formerly ANSI A58.1). The staff will address this item during its review of an individual application for FDA/DC. Therefore, this portion of the DSER open issue is closed.

# Geology and Foundation Conditions

Table 1.2-6 of Chapter 1 requires a minimum bearing capacity of >15 kips/ft<sup>2</sup> (ksf), a minimum shear wave velocity of >1000 ft/sec, and no liquefaction potential at the site-specific SSE level. In the DSER for Chapter 1, the staff concluded that these requirements were acceptable. However, in its letter dated April 24, 1991, the staff requested that EPRI clarify whether the specified minimum bearing capacity of >15 ksf was a static or a dynamic value. In its response dated July 2, 1991, EPRI stated that it was intended to be a static value. The staff finds that 15 ksf for a minimum static hearing capacity is sufficient for the expected demand from structural loading. Therefore, the staff concludes that this response is acceptable.

In its letter dated April 24, 1991, the staff indicated that Table 1.2-6 should give a range of soil properties to provide consistent guidance to the vendors of the standard plants and potential utilities. In its response dated

July 2, 1991, EPRI stated that the level of effort needed to quantify more specific soil parameters was beyond the scope of the Evolutionary Requirements Document, and that the ALWR objectives will be satisfied as long as the standard plant design will be suitable for a large range of foundation siting conditions that fall within the envelope of parameters of Table 1.2-6. The staff will address this during its review of an application for a combined license.

In its letter dated April 24, 1991, the staff requested that EPRI develop evaluation guidelines regarding the minimum margin against liquefaction. In its letter dated July 2, 1992, EPRI stated that the specific guidelines had not been developed for the Evolutionary Requirements Document and that a sitespecific evaluation must be performed when a plant is to be founded on a soil site. Consistent with the scope and level of technical details in the Evolutionary Requirements Document, the staff concludes that the guidelines for minimum margin against liquefaction potential may be addressed by the applicant for a combined license as a site-specific issue if the plant is to be founded on a soil site, or if any structures are to be founded on soil having a liquefaction potential at sites with multiple soil conditions. Such guidelines should include a detailed evaluation of the liquefaction potential (as described in SRP Section 2.5.4, "Stability of Subsurface Materials and Foundations"), and consequences of liquefaction, of all subsurface soils, including the settlement of foundations. These evaluations will be based on soil properties obtained by state-of-the-art laboratory and field tests and involve application of both deterministic and probabilistic procedures.

In its letter dated April 24, 1991, the staff requested that EPRI define the analyses or evaluation methods that will be used to evaluate hazards such as active faults, man-induced hazards, and soil stability. In its response dated July 2, 1991, EPRI noted that these issues were not applicable in the design of standard plants and should be considered in site-specific assessments, and that it anticipated that NRC-approved state-of-the-art analyses and evaluation methods will be used at that time. The staff will address this issue during its review of an application for a combined license.

# Earthquakes

Section 4.5.2.4 of Chapter 1 requires that the standardized seismic designs be based on the site parameter envelope of Table 1.2-6 in Chapter 1.

In Section 4.5.2.4 of Chapter 1, the operating-basis earthquake (OBE) was deleted for consideration in the design process. As discussed in Section 4.4.3 and Appendix B of this chapter, the staff is evaluating the effect of this change on current staff positions. This evaluation will address the requirement in Sections 4.5.2.4.4.1 and 4.8.1.1 of Chapter 1 that reduces the number of full-stress cycles for 1/2 safety shutdown earthquake (SSE) from 50 to 20. The results of this evaluation will be included in the supplemental regulatory guidance discussed in Section 4.4.3 of this chapter. The staff will review an individual application for FDA/DC in accordance with the supplemental guidance.

In Table 1.2.6 of Chapter 1, EPRI states that the ALWR plant design should be envelop an SSE ground motion having the following characteristics:

- peak ground acceleration (PGA) equal to 0.30g
- design response spectra in accordance with RG 1.60, "Design Response Spectra for Seismic Design of Nuclear Power Plants"
- generated artificial ground motion time history to envelop the design response spectra

Regarding the SSE, Note 10 of Table 1.2-6 requires that the free field motion be specified at plant grade. Section 4.5.2.4.2 of Chapter 1 requires that the SSE spectra input motion be deconvoluted to specific points for applications to structural analysis models taking soil-structure interaction (SSI) effects into account. The requirement in Section 4.5.2.4.2 implies that the free field motion will be placed at plant grade to facilitate the deconvolution analysis. In its letter dated April 24, 1991, the staff requested that EPRI justify specifying free field motion at plant grade because SRP Section 2.5.2, "Vibratory Ground Motion," Revision 2, accepts the specification of free field motion at plant grade only when sufficient data indicate that the sites consist of relatively uniform soil or rock with smooth variation of properties with depth. When there are insufficient recorded ground motion data, or when the site is composed of one or more than one relatively thin soil layer overlying a competent foundation material, the free field motion should be specified on an outcrop or a hypothetical outcrop in the free field. For SSI analyses, Revision 2 of SRP Section 3.7.2, "Seismic System Analysis," contains similar requirements regarding the location for specifying free field motion. In its letter dated July 2, 1991, EPRI stated that the analysis of the soil structure system should follow the guidance in NRC's regulatory guides and SRP Sections, including SRP Section. 3.7.1, "Seismic Design Parameters," 3.7.2, and 3.7.3, "Seismic Subsystem Analysis." EPRI indicated that assumptions for variations in the soil properties should be accommodated in the analysis. On the basis of this response, the staff concludes that the requirement to specify the free field ground motion at plant grade is acceptable.

Although the design-basis SSE of an RG 1.60 ("Design Response Spectra for Seismic Design of Nuclear Power Plan',") spectrum with a zero period acceleration of 0.30g is sufficient for most potential sites in the United States, it may not envelop the ground motion for sites near seismically active areas in the Eastern and Central United States or sites in the Western United States, in addition to those along the California coast. The staff has observed that earthquakes recorded in the Eastern United States possess more high-frequency (greater than 5 Hz) ground motion than those earthquakes whose records were used to develop the RG 1.60 response spectrum. This could limit the sites at which designs using 0.3g zero period RG 1.60 response spectrum could be located. The staff will review the site-specific SSE with respect to the design basis at the time of siting.

In Table 1.2-6, the criterion for the SSE ground sotion time history (time history) is that the response spectra obtained from the time history envelop the design response spectra. In accordance with SRP Section 3.7.1, the staff's position is that this criterion should also include the requirement that the power spectrum density (PSD) function of the time history envelop an approved target PSD function if a single time history is used. In addition,

SRP Section 3.7.1 specifies a different acceptance criterion if multiple time histories are used. The staff requires that the time history comply fully with the SRP and will address this matter during its review of an individual application for FDA/DC.

#### Tornado

In Section 4.4.C of the DSER for Chapter 1, the staff disagreed with EPRI's proposed use of ANSI/ANS 2.3, "Standard for Estimating Tornado and Extreme Wind Characteristics at Nuclear Power Sites," to define the tornado effects for the ALWR. In its letter dated March 25, 1988, the staff issued its interim position on this issue. In Revision 3 of Section 4.5.2.5 and Table 1.2-6 of Chapter 1 and Section 2.1.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI proposed that the maximum tornado wind speed of 260 mph and the tornado recurrence interval of 1 million years (tornado strike probability of 1.0E-6 per year) be used for the design-basis tornado. These parameters are based on ANSI/ANS 2.3. Section 2.1.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document lists tornado design for the evolutionary ALWR as a plant optimization subject. This subject is also discussed by the staff in the regulatory departure analysis in Appendix B of this chapter.

The current NRC regulatory position with regard to the design-basis tornado (DBT) is contained in two 1974 documents: WASH-1300, "Technical Basis for Interim Regional Tornado Criteria," and RG 1.76, "Design Basis Tornado for Nuclear Power Plants." WASH-1300 states that the probability of occurrence of a tornado that exceeds the DBT should be about 1.0E-7 per year per nuclear power plant, and the regulatory guide delineates the maximum wind speeds of 240 to 360 mph depending on the regions.

The staff has not endorsed ANSI/ANS 2.3. However, the regulatory positions in RG 1.76 were reevaluated by an NRC contractor using the considerable quantity of tornado data that are now available but were not when the regulatory guide was developed. The contractor's reevaluation is provided in NUREG/CR-4461, "Tornado ilimatology of the Contiguous United States," dated May 1986. At the heart of this study is the tornaco data tape, prepared by the National Severe Storm Forecast Center, that contains 30 years of data, 1954 through 1983. This data tape contains the data for the approximately 30,000 tornadoes that occurred during that period.

The contractor found that the tornado strike probabilities range from nearly 1.0E-7 per year for much of the Western United States to about 1.0E-3 per year in the Central United States. On the basis of discussions between the contractor and the staff, wind speed values associated with a tornado having a mean recurrence interval of 10E-7 per year were estimated to be about 200 mph for the United States west of the Rocky Mountains and 300 mph for the United States east of the Rocky Mountains.

In its December 6, 1991, letter, EPRI proposed that a maximum tornado wind speed of 300 mph and a tornado recurrence interval of 1.0E-7 per year tornado strike probability be used for the design-basis tornade in the design of the evolutionary ALWRs. In Revision 4 of Chapter 1, EPRI deleted the reference to the tornado recurrence interval in Table 1.2-6 from the Evolutionary Requirements Document.

The tornado design-basis requirements have been used in establishing structural requirements (minimum concrete wall thickness) for the protection of nuclear plant safety-related structures, systems, and components against the effects not covered explicitly in review guidance such as regulatory guides or the SRP. Specifically, some aviation (general aviation light aircraft) crashes, nearby explosions, and explosion debris or missiles have been reviewed and evaluated routinely by the staff by taking into account the tornado protection requirements. Hence, the staff's acceptance of the structural design will also necessitate a concurrent evaluation of their effect on the protection criteria for some external impact hazards, such as general aviation crashes or nearby explosions. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

In the draft Commission paper on evolutionary and passive plant policy issues dated February 27, 1992, the staff stated that it will accept the tornado design basis of 300 mph recently proposed by EPRI. Table 1.2 shows the DBT parameters that the staff considers acceptable. However, until the staff resolves this issue with the Commission, it considers this an open issue that must be resolved before it can complete its review of Chapter 1.

### Volcanic Activity

In its letter dated April 24, 1991, the staff requested that EPRI confirm that the ALWR site will not be located in areas subjected to the effects of volcanic activity near the site. In its July 2, 1991, response, EPRI confirmed that the standard plants will not be located in such areas and that site-specific design assessments will be necessary if a plant owner decides to locate a standard plant in such an area. The \_\_\_\_\_\_iff concludes that this response is acceptable.

### Design Temperatures

Design temperatures are not included or discussed in Section 4.5.2 of Chapter 1. However, in Revision O of the Evolutionary Requirements Document, Table 2-1 specified three categories of design temperatures: ambient, emergency cooling water inlet, and condenser cooling water inlet. The ambient temperature was expressed in terms of the maximum and minimum temperatures for both 1-percent exceedance probability and 0-percent exceedance probability. In the DSER for Chapter 1, the staff stated that it was not certain how the ambient temperature values will be used when they are derived from a probabilistic method and are associated with certain probab. ties of exceedance.

Revision 3 of Chapter 1 of the Evolutionary Requirements Document specifies the same probability-based ambient temperature values as those in Revision 0 and does not address the staff's concern. In Revision 3, the cooling water inlet temperature values specified in Revision O were changed. The new criterion requires that the site permit atmospheric heat rejection of cooling water system heat loads or be such as to provide cooling water at the flow rates and temperatures specified by the plant designer to achieve certain probability-based cooling performance limits. To review the safety-related water supply, the staff typically uses deterministic values based on worst 1-hour, 24-hour, and 30-day values of record. Therefore, the staff will use the deterministic approach to review an individual application for FDA/DC.

## 4.5.3 Site Proximity Man-Made Hazards

### Physical Security Considerations

The inherent resistance to sabotage of the current generation of LWRs is in part due to reinforced-concrete structures designed to tornado design criteria. Generic Letter 89-07, "Power Reactor Safeguards Contingency Planning for Surface Vehicle Bombs," contains a safeguards information addendum that includes generic standoff distances for protection against land vehicle bombs of reinforced-concrete walls constructed to existing tornado standards. That guidance is derived from NUREG/CR-2462, "Capacity of Nuclear Power Plant Structures To Resist Blast Loadings," which in turn relies on RG 1.76. The staff questioned whether the exception taken in Section 4.4.3.3.8 of Chapter 1 to RG 1.76 wo. I reduce the inherent resistance of an ALWR to sabotage or require differ . guidance on standoff distances for contingency planning. In its May 13, 1991, response, EPRI stated that the wall capacity (thickness and reinforcement) will likely be dictated by seismic shear load rather than tornado loading because of the 0.3g SSE requirement. The staff does not agree with this position. If minimum static capacities are less than those assumed in NUREG/CR-2462, the staff may need to revise its guidance on contingency planning for surface vehicle bombs. The staff will address this issue during its review of an application for a combined license.

### 4.5.4 Plant Operating Loads

Section 4.5.4 of Chapter 1 of the Evolutionary Requirements Document provides general requirements that state that the plant designer will consider and minimize the loads due to anticipated and extreme plant conditions resulting from the design-basis events listed in Section 2 of Chaster 1.

Anticipated plant conditions consist of normal operating loads such as live and deadweight loads, pressure, temperature effects, external reactions, and loads from expected plant transients such as plant heatup and cooldown, hot standby, and anticipated conditions during operations from zero to full power. Vibratory loads due to reciprocating and rotating equipment, pump pulsations, rapid opening and closing of valves, flow-induced vibrations, relief valve blowdown, cavitation, and any other fluid-system-induced loadings on equipment and structures will be minimized by system design, choice of equipment, and use of accumulators to dampen vibrations. Where vibrations cannot be avoided, energy absorbers, dashpots, or other such devices should be used. The use of seismic snubbers for vibration damping should be avoided. Potential waterhammer events such as turbine stop valve closure, safety/relief valve and fast valve actuations, and rupture disc loads will be considered in the design of the systems. The methodology discussed in Section 4.5.4.1.5 of Chapter 1, for defining BWR safety/relief valve loads to be used in the design of the suppression pool is still under review by the staff. The staff will evaluate this matter during its review of an individual application for FDA/DC.

Extreme plant condition loads include pressures, temperatures, forces, and vibrations due to transients and accidents not encountered in normal operation, but that might reasonably be anticipated to occur during the design life of the plant. Such loads include, but are not limited to, the following: (1) water hammer in subcooled water lines or flooded steam lines; (2) abnormal component operation, such as a relief salve stuck open or an isolation valve stuck closed downstream of a pump; and (3) low-flow thermal stratification in

feedwater lines. The plant designer will minimize the effects of such loads by designs that reduce or isolate the loads and by accounting for the loads early in the plant design.

The staff concludes that, with the exception noted above, the broad requirements in Section 4.5.4 of Chapter 1 relative to plant operating loads are consistent with current applicable staff guidelines and are acceptable.

# 4.5.5 In-Plant Hazards

Section 4.5.5 of Chapter 1 states that the plant designer will consider the potential structural effects of in-plant hazards due to the environmental conditions associated with plant operations under normal conditions and postulated accidents. These hazards are pipe rupture, internal flooding and fires, and the generation of missiles.

### Rupture of Piping

Section 4.5.5.1 of Chapter 1 provides general requirements to be used by the plant designer for determining postulated pipe-rupture locations and the dynamic effects associated with the postulated rupture of piping. The criteria of ANSI/ANS 58.2, "Design Basis for Protection of Light Water Nuclear Power Plants Against the Effects of Pipe Rupture," as supplemented by applicable regulatory positions, will be used for these evaluations. The dynamic effects that are to be evaluated include impact from whipping pipes, jet impingement loads, rapid subcompartment pressurization, hydraulic system interna? loads, and the motion of the equipment attached to components responding to these effects. Those portions of safety-related structures, systems, and components that could be affected by these dynamic effects will either be protected from such effects or be designed to withstand the resulting loads and still be able to perform the required safety function. Table B.1-2 of Appendix B to Chapter 1 commits to compliance with SRP Section 3.6.2, "Determination of Rupture Locations and Dynamic Effects Associated With the Postulated Rupture of Piping," with no exceptions. The staff has reviewed ANSI/ANS 58.2 and concludes that it is consistent with SRP Section 3.6.2. The staff further concludes that the commitment to ANSI/ANS 58.2. supplemented by applicable positions in SRP Section 3.6.2, provides acceptable criteria for determining postulated pipe-rupture locations and associated dynamic effects.

### Pipe Rupture Loads

The requirements regarding pipe rupture loads in Section 4.5.5 of Chapter 1 of the Evolutionary Requirements Document contain a statement that subsets of safety class structures, systems, and components shall be protected from the dynamic effects resulting from postulated ruptures or designed for the resulting loads. The dynamic effects that will be considered in the design include pipe whip, jet impingement, rapid subcompartment pressurization, hydraulic system internal loads, and the motion of attached equipment. It is the staff's understanding that this requirement is not applicable if the ALWR applicant has justified the use of the leak-before-break (LBB) concept. To be consistent with the implementation of the LBB technology under the recent broad-scope amendment of GDC 4 of Appendix A to 10 CFR Part 50, containment and emergency core cooling system (ECCS) functional and performance requirements are maintained. The staff's current interpretation of this rule change

is that the above local dynamic effects (pipe whip, jet impingement, etc.) can be excluded from the containment design basis. To retain high safety margins, the containment must continue to be designed to withstand all global loading and environmental effects up to and including the double-ended rupture of the largest pipe in the reactor coolant system.

# Leak Before Break

In Appendix A to the DSER for Chapter 1, the staff identified leak-beforebreak (LBB) considerations as an open issue. Compliance with General Design Criterion (GDC) 4 of Appendix A to 10 CFR Part 50 was listed as a confirmatory issue in Section 10 of it. DSER. Revision 3 of Section 4.5.5 of Chapter 1 of the Evolutionary Requirements Document specifies that, to the extent practicable, fracture mechanics and mechanistic analysis through the application of LBB criteria will be used to minimize the need to evaluate the effects of pipe rupture. The LBB approach is also addressed in Sections 4.4.3.3.1 and 4.5.5.1.3 of Chapter 1 of the Evolutionary Requirements Document.

EPRI is proposing to adopt for ALWRs the LBB approach when certain details of the piping design, materials properties, and stress conditions are known. The regulatory departure analysis in Appendix B of this chapter provides the background and staff position on this issue. In the draft Commission paper on evolutionary and passive plant policy issues dated February 27, 1992, the staff concludes that the application of the LBB approach to ALWRs for which FDA/DC is being sought under 10 CFR Part 52 is acceptable when appropriate bounding limits are established during the FDA/DC review phase using preliminary analyses results and are verified during the combined license phase by implementing the appropriate inspections, tests, analyses, criteria, and acceptance criteria (ITAAC) discussed herein. Since the Commission has not yet reviewed this approach to resolving the approach for resolving leak before break, it does not represent an agency position. Therefore, the staff regards this as a open issue that will be closed once the Commission approves this resolution or provides alternative guidance.

The staff has evaluated the Evolutionary Requirements Document for LBB applications in ALWRs and concludes that it is not completely acceptable. The limitations and acceptance criteria for LBB applications in ALWRs are the same as those established for currently operating nuclear power plants. Therefore, NUREG-1061, Volume 3, should be referenced by the individual FDA/DC or combined license applicant requesting approval of the LBB approach because it provides the NRC-approved LBB methodology and acceptance criteria. In the rationale portion of Section 4.5.5.2.4, EPRI states that its LBB methodology and acceptance criteria are the same as those in NUREG-1061, Volume 3. However, the staff has determined that they are not the same because the criteria in Section 4.5.5.2.4 are merely excerpts of the acceptance criteria in NUREG-1061. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

### BWR Suppression Pool Loads

In Appendix A to the DSER for Chapter 1, the staff stated that it had not completed its evaluation of the methodology proposed by EPRI for defining BWR safety/relief (SRV) valve loads to be used in the design of the suppression pool. The methodology proposed by EPRI to be imposed on the submerged structures within the suppression pool during SRV discharge did not

specifically reference staff-approved methods or propose verification of calculational techniques if new methods are used. For the Mark I, II and III BWR containment designs, a substantive test program was undertaken to develop a load definition report (LDR) that established both acceptable models as well as the basis for NRC acceptance of these models to calculate suppression pool hydrodynamic loads.

However, if either the methodology or the data base is used, the FDA/DC applicant must demonstrate applicability to the specific design. Parameters that could influence applicability include, but are not limited to, the following: quencher design details, piping layouts, submergence, piping diameter, and safety valve lift pressure. In addition to the physical layout, the response of the primary system is also a factor. This response is critical because it will determine the number of valves that will lift and the number that will reopen on the "second pop." Because of the necessary detail associated with this issue, the staff will conduct a specific design review of plant hydrodynamic loads and will evaluate the method of determining the SRV loads and the applicability of the data base used to support the model. Should the LDR be sufficiently different from the specific design, a new plant-specific experimental basis similar to the work done at the full-scale test facility would need to be developed before staff acceptance.

The staff concludes that EPRI has not justified its position for the use of a model to compute SRV loads. The staff will review individual applications for FDA/DC against the criteria in the SRP. On the basis of the above discussion, this DSER open issue is closed.

In Section 4.5.5.3.1 of Chapter 1, EPRI states that the models used to compute pool swell, condensation oscillation loads, and chugging loads are assumed applicable for calculating the containment boundary loads. The staff concludes that this is acceptable if the designer can demonstrate that the design is within the limits of the data base.

The Evolutionary Requirements Document does not address the methodology or data base to be used for calculating the hydrodynamic loads. Because the configuration of the ABWR vent and containment appears to be a combination of the Mark II and Mark III configurations, the methodology used by the plant designer must be justified. The plant designer must also show the applicability of the data base to the design. The following critical variables may have an effect on pool loads: vent diameter and vent pipe length, submergence, pool temperature limits, containment volume, and configuration. These variables need to be addressed by the plant designer. Because of the necessary detail associated with this issue, the staff will conduct a specific design review of plant hydrodynamic loads and will evaluate the method of determining the lossof-coolant-accident (LOCA) loads and the applicability of the data base used to support the model. Should the Mark III data base be sufficiently different from the specific design, a new plant-specific experimental basis similar to the work done to establish the generic methodology for establishing designbasis LOCA loads for the Mark II and III designs would need to be developed before staff acceptance.

The staff concludes that EPRI has not completely justified its position for the use of the ULD models to compute pool dynamic loads. The staff will evaluate this matter during its review of an individual application for FDA/DC.

The heat removal and mass replacement capability of the ECCS (flow rates, pressures, storage volumes) should continue to be designed to accommodate pipe ruptures up to and including the double-ended rupture of the largest pipe in the reactor coolant system, even when LBB is demonstrated.

# Internally Generated Missiles

In Table 8.1-2 of Appendix B to Chapter 1, EPRI commits to comply with the staff review guidance in SRP Section 3.5.1.1, "Internally Generated Missiles (Outside Containment)"; SRP Section 3.5.1.2, "Internally Generated Missiles (Inside Containment)"; and SRP Section 3.5.1.4, "Missiles Generated by Natural Phenomena." The staff concludes that this commitment is acceptable. However, Section 4.5.5.4.1 of Chapter 1 states that ANSI/ANS 58.1, "Plant Design Against Missiles," will be used for guidance in meeting the requirements pertaining to internal-missile generation. The staff has not endorsed ANSI/ANS 58.1. Therefore, if differences exist between the above SRP sections and ANSI/ANS 58.1, the SRP sections should be used. If a plant designer identifies and provides justification for the differences, the staff will review the justification on a case-by-case basis and address the issue during its review of an individual application for FDA/DC.

### Internal Flooding

In Section 2.2.F.7 of the DSER for Chapter 1, the staff recommended that EPRI ad information on protection against and mitigation of internal flooding. In Table B.1-2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI commits to comply with SRP Section 3.4.1, "Flood Protection." Section 4.5.5.5 of Chapter 1 states that the plant designer will identify those systems and equipment that must be protected from flooding and those that must be capable of normal operation under flooded conditions. The specific design conditions will be specified for systems and equipment required to operate when directly exposed to flooded conditions. EPRI states that internal flooding may be caused by such events as fire protection system operation or postulated breaks in tanks or piping. In addition, EPRI added information about internal flooding to Table 1.2-4 of Chapter 1 in Revision 4 of the Evolutionary Requirements Document. The staff concludes that the above commitment and the requirements in Section 4.5.5.5 and Table 1.2-4 provide the necessary requirements to protect structures and systems from the effects of internal flooding and are, therefore, acceptable.

# 4.6 Load Combinations

# 4.6.1 Buildings and Structures

Section 4.6.1 of Chapter 1 of the Evolutionary Requirements Document delineates the load combination requirements for the design of buildings and structures.

### Concrete and Steel Containments

Section 4.6.1.1 of Chapter 1 requires that the design of concrete containments satisfy the load combinations in ASME Code, Section III, Division 2, Subsection CC, and the design of steel containments follow SRP Section 3.8.2, "Steel Containment." The staff concludes that the specified load combinations for the steel containment design conform with SRP Section 3.8.2 and are

acceptable. Because SRP Section 3.8.1, "Concrete Containment," and RG 1.136, "Materials, Construction, and Testing of Concrete Containment," provide additional guidance on the use of ASME Code, Section III, Division 2, Subsection CC, for the design of concrete containments but are not referenced in the Evolutionary Requirements Document, in a letter dated April 24, 1991, the staff requested that EPRI confirm its position reg rding compliance. In its letter dated July 2, 1991, EPRI indicated that all ough it is impractical to list all applicable regulatory guides and SRP sections in individual paragraphs of the Evolutionary Requirements Document, "requirement of compliance for the ALWRs is shown in Table 8.1-1 of Appented to Chapter 1. To ensure that the plant designer will use proper additional regulatory guidance for the load combinations, the staff's position is that the concrete containment design will follow the guidelines of SRP Section 3.8.1 and RG 1.136. The staff will evaluate this issue during its review of an individual application for FDA/DC.

# Other Seismic Category I Buildings and Structures

Section 4.6.1.2 of Chapter 1 require the design of other seismic Category I reinforced-concrete and steel structures to satisfy the load combinations specified in American National Standards Institute/American Concrete Society (ANSI/ACI) 349, "Code Requirements for Nuclear Safety-Related Structures," and ANSI/American Institute of Steel Construction (AISC) N690. "Specification for the Design, Fabrication, and Erection of Steel Safety-Related Structures for Nuclear Power Plants," respectively. In its letter dated April 24, 1991, the staff noted that RG 1.142, "Safety-Related Concrete Structures for Nuclear Power Plants," and SRP Sections 3.8.3, "Concrete and Steel Internal Structures of Steel or Concrete Containments," and 3.8.4, "Other Seismic Category I Structures," provide additional guidance on the use of ANS1/ACI 349 for seismic Category I concrete buildings and that the NRC has not approved the use of ANSI/AISC N690 for seismic Category I steel structures. In its response of July 2, 1992, EPRI indicated that although it is impractical to list all applicable regulatory guides and SRP sections in individual paragraphs of the Evolutionary Requirements Document, the requirement of compliance for the ALWRs is shown in Table B.1-1 of Appendix B to Chapter 1. To ensure that the plant designer will use proper additional regulatory guidance for load combinations, the staff position is to require adherence to SKP Sections 3.8.3 and 3.8.4 and RG 1.142. The staff will evaluate compliance during its review of an individual application for FDA/DC.

EPRI proposes to use ANSI/AISC N690 for the design of seismic Category I steel structures. The acceptability of using this code is uncertain because it has not been reviewed and approved by the staff. Therefore, the staff will evaluate this issue during its review of an individual application for FDA/DC.

# Decoupling of Safe Shutdown Earthquake and Pipe Rupture Loads for Buildings and Structures

In Appendix A to the DSER for Chapter 1, the staff indicated that it was still evaluating EPRI's proposal to decouple the loads from a safe shutdown earthquake (SSE) and LOCA when the leak-before-break (LBB) approach is used. In addition, the staff raised a concern regarding protection of containment components and emergency core cooling system (ECCS) hardware against the dynamic effects of pipe ruptures for systems for which the LBB approach is used.

Section 4.6.1.3 of Chapter I of the Evolutionary Requirements Document specifies that when the LBB approach can be demonstrated to apply, SSE and pipe rupture loads will not be combined in the design of structures. In its letter dated April 24, 1991, the staff recommended use of the present practice of combining the loads for the design of the containment structure, structures inside the containment, and structures integral with the containment. In its response of July 2, 1991, EPRI maintained its original position as stated above.

In Section 4.6.1.1 of Chapter 1, EPRI requires using load combinations as specified in SRP Section 3.8.2 for steel containment structures and in ASME Code, Section III, Division 2, Subsection CC, for concrete containment structures. The only exception is that the Evolutionary Requirements Document excludes the operating-basis-earthquake (OBE) load terms from load combinations. The staff generally agrees with the approach to decouple the OBE from the SSE and possibly eliminate the OBE altogether from the design as discussed in Section 4.4.3 of this chapter.

In addition, in Section 4.6.1.3 of Chapter 1, EPRI proposes eliminating the combination of pipe ripture loads with seismic loads for buildings and structures when the LBB approach applies. SRP Section 3.8.2 and .51e CC-3230-1 of ASME Code, Section III, Division 2, Subsection CC, require that SSE and LOCA loads be combined for containment structural design. Furthermore, EPRI proposes in Section 4.6.1.7.1 of Chapter 1 to invoke a probability-based load combination with a cutoff frequency of 1.0E-6, on the basis of ANSI/ANS 2.12 and 51.1 for the design of the containment and other seismic Category I structures. In Section 4.6.2.3 of Chapter 1, EPRI specifies that the combination of LOCA and SSE loads for equipment and systems should explicitly be eliminated on the basis of the recommendations of NUREG-1061 and implementation of the LBB technology. EPRI states that the probability of a seismically induced LOCA is extremely low for PWR primary systems and for the BWR as well.

In the early 1960s, the double-ended guillotine break of reactor coolant loop piping was postulated for containment sizing and ECCS performance. Later, this pipe rupture load was combined with the earthquake loads and applied to the containment structural design and subsequently to the design of other plant features, including nuclear reactor piping and its support systems. Since the early 1970s, the NRC criteria for design and analysis of seismic Category I structures have been formulated with sufficient conservatism to ensure ample safety margins against premature failures. The margins that were built into the structures have served as one important element in the NRC staff's implementation of the defense-in-depth regulatory philosophy, and often provided key bases for the staff to justify and allow continued operation of existing plant structures whenever load demand for the structures had to be increased for a variety of reasons (e.g., increased hazards, design errors, or omissions and modifications).

The NRC promulgated a rule change to General Design Criterion 4 of 10 CFR Part 50, Appendix A, that allows the application of the LBB method to piping systems. The revision eliminates the need to design for the dynamic effects of postulated pipe breaks, including pipe whip restraints and jet impingement shields, and to design subcompartments for dynamic pressurization loads if these loads are not essential to the containment function. Also, when the LBB method is demonstrated deterministically applicable to a piping system, the combination of SSE loads and LOCA loads effectively becomes the SSE loads alone.

Although it has been considering modifications to SRP Sections 3.8.1 through 3.8.4, the staff found that there was an insufficient technical basis to extend decoupling to structures without resolving questions about the long-term, intermediate-term, and short-term pipe rupture effects of decoupling the pipe rupture from seismic loads.

The application of LBB technology eliminates the local dynamic effects of postulated pipe ruptures from the design basis. However, global effects still result from a source other than the postulated pipe rupture. Because the global effects from the postulated pipe rupture provide a convenient and conservative design envelope, and the NRC staff is not prepared at this time to propose alternative criteria for the containment, the load combinations indicated in the relevant sections of the SRP continue to be valid except for the combinations with the OBE. The staff will require consideration of certain aspects of the OBE in the design process to account for low-cycle fatigue and relative support motion. Although an adequate basis exists for the above-proposed decoupling of LOCA and SSE loads and the consequent deviation from the current NRC criteria for the dynamic effects for the mechanical design of components and their support, this deviation would result in a significant reduction in critical structural safety margins for ALWR containments vulnerable to severe-accident loadings. On the basis of its understanding of the design of all the existing containment structures, the staff concludes that the containment and other seismic Category I structures must continue to contain a concurrent application of LOCA and SSE loads. For the design of seismic Category I structures, elimination of load combinations with a probability of occurrence less than 1.0E-6 per reactor-year (for other than LOCA plus SSE) is acceptable in principle; however, the staff will only approve such an elimination when a specific design applicant justifies its position with specific examples of how and where the design is governed and a comparison of structural margins with and without the load combination in question.

In summary, the staff concludes that eliminating the combination of pipe rupture loads (global effects) with seismic loads for the containment and other seismic Category I structures is not acceptable. Furthermore, EPRI's proposal to decouple LOCA and SSE loads for equipment and systems is not acceptable at this time because of the insufficient technical bases to extend the decoupling to structures. The staff will evaluate this issue during its review of an individual application for FDA/DC.

# Treatment of Loads That Reduce the Effects of Other Loads

Section 4.6.1.4 of Chapter 1 requires that consideration be given to the treatment of loads that reduce the effects of other loads in the application of the design loading combination of seismic Category I buildings and structures. If any such load reduces the effects of other loads, the corresponding coefficient for that load will be taken as 0.9 if it can be demonstrated that the load is clways present or occurs simultaneously with the other load. Otherwise, the coefficient for the load will be taken as zero. The staff disagrees with this criterion. EPRI is unclear about the use of a coefficient of 0.9 or 0 for the context of a working stress design or load factor design.

EPRI Evol. Jonary Plant SER

The load combinations identified in SRP Section 3.8.3 represent current practice and are appropriate for the design to ensure adequate margins of safety. It is inappropriate to use a load coefficient of C.O. If the load is zero, its effects will be reflected in the design. The staff concludes that EPRI has no provided adequate requirements pertaining to the staff's position as documented in SRP Section 3.8.2 and requires that SRP Section 3.8.3 be used for considering design loading. The staff will evaluate this issue during its review of an individual application for FDA/DC.

## Foundation Design for Seismic Category I Buildings

Section 4.6.1.5 of Chapter 1 requires that the design of foundations for seismic Category I buildings and structures satisfy the minimum factors of safety, taking into account the effects of seismic soil-structure interaction with respect to sliding, overturning, and floatation. The Evolutionary Requirements Document references the minimum factors of safety that are specified in SRP Section 3.8.5, "Foundations." However, as stated in the staff's letter of April 24, 1991, the effects of normal and high ground water in the design of the embedded structure and foundation were not considered. In its response of July 2, 1991, EPRI committed to include such a requirement and implemented this commitment in Revision 2 of the Evolutionary Requirements Document. The staff concludes that the requirement for foundation design for seismic Category I buildings is acceptable.

# Hon-Seismic Buildings and Structures

Section 4.6.1.6 of Chapter 1 requires that non-seismic buildings, structures, and structural subsystems be designed to meet the load combinations specified in the Uniform Building Code (UBC). In its letter dated April 24, 1991, the staff expressed a concern about the use of the UBC seismic Zone 2A specification that it discusses in Section 3.2.3 of this chapter. However, as far as load combination is concerned, the use of the UBC for non-seismic structures is acceptable.

#### Probability-Based Mechanistic Design Loads

Section 4.6.1.7 of Chapter 1 allows the optional use of mechanic ic design loads and load combinations developed on a probabilistic basis if certain load combinations disproportionately control the design of plant structures without a rational basis. In its letter dated April 24, 1991, the staff requested that EPRI justify using the methodology in NUREG/CR-3876 "Probability Based Load Combination Criteria for Design of Concrete Containment Structures," in defining loads and load combinations without a corresponding definition of the reliability index and consistent acceptance criteria. The staff also requested that EPRI give the basis for using ANSI/ANS 51.1, 52.1, or 2.12 in defining loads and load combinations. To address these two concerns, EPRI revised the Evolutionary Requirements Document to require that probabilistic load combinations be identified and a suitable justification be prepared for NRC review and concurrence, if deemed warranted by the plant designer. Mean shile, the rationale p rtion in Revision 3 of this section also was revised to acknowledge that the use of probabilistic load combinations is a departure from EPRI's commitment to the load combinations of SRP Sections 3.8.1 through 3.8.5, as indicated in Table B.1-2 of Appendix B to the Evolutionary Requirements Document, and that the plant designer is encouraged to consider their use where warranted, but regulatory concurrence will be

required on a case-by-care basis. This revision in requirements is acceptable in light of the staff-proposed position documented in the draft Commission paper issued on February 27, 1992. This position allows, in principle, the elimination of load combinations (except an SSE plus LOCA) with a probability of occurrence less than 1.0E-6 per reactor-year. However, the staff will approve such an elimination only when a specific design applicant justifies its contention with specific examples of how and where the design is governed and a comparison of structural margins with and without the load combination in question. The staff will evaluate this issue during its review of an individual application for FDA/DC.

### 4.6.2 Systems and Equipment

In the section entitled "Appendix A to Chapter 1" in the D'CR - ' Chapter 1, the staff identified an open issue relative to the decoupl no is the SSE from the LOCA. As discussed below and in the DSER, the staff port in is that, for the design of all safety-related structures, systems, and equipment, including their supports, the loads resulting from dynamic events associated with the faulted condition (ASME Service Level D) should be combined with the LOCA loads. In its letter dated May 17, 1991, the staff requested that the rationale portion of Sections 4.6.2.3 and 4.6.2.5 and the load combinations in Tables 1.4-5 and 1.4-7 be revised to agree with the above staff position. In addition, a note should be added to Tables 1.4-5 and 1.4-7 to state that the method of combination of dynamic responses to loads is in accordance with NUREG-0484, "Methodology for Combining Dynamic Responses," Revision 1, dated May 1980. In its letter dated August 1, 1991, EPRI stated that the SSE and LOCA loads are not combined because each event is of very low probability and unrelated. The staff agrees that EPRI's position could be applicable only to piping systems in the majority of PWR plants. However, the staff's position is based on the requirements of GDC 2 of Appendix A to 10 CFR Part 50, which states that all structures, systems, and components be designed to withstand the effects of appropriate combinations of normal and accident conditions with natural phenomena. Historically, the staff has interpreted GDC 2 as requiring that the effects of the SSE and LOCA be combined for the design of all safetyrelated portions f the plant. Any change in this interpretation requires either an exemption from or a revision of GDC 2. Therefore, the staff position remains as stated above and as reflected in SRP Section 3.9.3. As a result, the staff will review an individual application for FDA/DC in accordance with this position.

The staff concludes that, with the exception noted above, the information in Section 4.6.2 of Chapter 1 relative to loading combinations for systems and equipment is consistent with SRP Section 3.9.3 and is, therefore, acceptable.

### 4.7 Design Methodology

Section 4.7 of Chapter 1 of the Evolutionary Requirements Document states that all analysis and design techniques for the ALWR will be in accordance with the industry codes and standards specified in Section 4.4 of Chapter 1. It further states that the plant designer will be required to (1) consider experience in existing LWR plants to identify design problems that have adversely affected construction costs, schedules, maintainability, or operability; (2) implement design methods and accepted advanced dynamic analysis techniques to minimize unnecessary conservatism in the plant design; and (3) develop a design approach that allows appropriate tolerance for

construction and erection problems and for potential deviations in layout and location. To implement these requirements, exceptions to several regulatory positions may be necessary. Some of these exceptions are discussed by the staff in Section 4.4 of this chapter. Others are discussed below.

### 4.7.1 Introduction

In its letter dated April 24, 1991, the staff requested specific examples of the accepted advanced dynamic analysis techniques referenced in Section 4.7.1.4 of Chapter 1 of the Evolutionary Requirements Document. In its letter dated July 2, 1991, EPRI noted that specific examples had been identified in Section 4.4.3 to Chapter 1 and these examples belong with the optimization subjects discussed in Appendix B to Chapter 1. The staff's evaluation of this issue can be found in Section 4.4.3 and Appendix B of this chapter.

# 4.7.2 Buildings and Structures

#### Containments

Section 4.7.2.1 of Charter 1 requires that concrete containments be designed, constructed, and test, in accordance with ASME Code, Section III, Division 2, Subsection CC, and stee) containments in accordance with ASME Code, Section III, Division 1, Subsection NE. These requirements are consistent with current LWR design practice and are acceptable.

# Other Category I Buildings and Structures

Section 4.7.2.2 of Chapter 1 requires that other Category 1 concrete and steel buildings and structures satisfy, respectively, ANSI/ACI 349 and ANSI/AISC N690 as supplemented by the AISC load and resistance factor detime method. For abnormal or extreme load combinations, the Evolutionary Requirements Document permits local yielding provided the yielding is contained and does not result in a collapse mechanism and the resulting ductility factor does not exceed that recommended in applicable regulatory positions.

In its letter dated April 24, 1991, the staff commented that the NRC has accepted the use of ductility factors for concrete and steel structures in design for impact and impulsive loadings only and not for other load combinations. In its response of July 2, 1991, EPRI acknowledged that a clarification to this requirement was necessary. Revision 3 of the Evolutionary Requirements Document states that local yielding is permissible for "impulsive and impactive" loads in the "abnormal/extreme environmental" load combinations. The staff concludes that this clarification is sufficient and that the design methodology specified for other Category I buildings and structures is acceptable.

### Dynamic Analysis Techniques

Section 4.7.2.3 of Chapter 1 requires that dynamic analysis techniques comply with American Society of Civil Engineers (ASCE) 4-86, as well as other applicable codes and standards, and be qualified and proven. In its letter dated April 24, 1991, the staff commented that the NRC has not accepted all analysis techniques in ASCE 4-86. In its response of July 2, 1991, EPRI stated at ASCE 4-86 is intended to supplement the overall criteria and methods...gy specified in regulatory guides and SRP sections, and that

Table B.1-1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document confirms EPRI's commitment to comply with the regulatory positions except for those analysis techniques associated with the optimization subjects. This response and the requirements in this section of Chapter 1 are not acceptable because ASCE 4-86 has not been reviewed and approved by the staff. Therefore, the staff concludes that the guidelines in the SRP and regulatory guides should be used for the plant analysis and design of future ALWRs. Plant designers proposing to use ASCE 4-86 should submit a request for staff review and approval on a case-by-case basis. The staff will evaluate this issue during its review of an individual application for FDA/DC.

# Timing of Seismic Analyses and Need for Confirmatory Analyses

Section 4.7.2.4 of Chapter 1 emphasizes the benefit of performing seismic analyses early in the design stage. It also requires that confirmatory analyses be performed after building structures and major equipment are in the final design stage if significant changes in the distributed mass and stiffness are introduced during the design. The staff concludes that this is a reasonable requirement and is acceriable.

# Generation of Design Response Spectra or Time Histories

Section 4.7.2.5 of Chapter 1 requires that the generation of design response spectra or time histories be based on methods that minimize unnecessary conservatism. Since the Evolutionary Requirements Ducument did not indicate the need for NRC approval of methods such as the spectrum-to-spectrum generation procedure, the staff requested that EPRI submit a complete discussion of the limitations and verification of such procedures. In its response of July 2, 1991, EPRI did not address the NRC concern, and, therefore, the requirements in Section 4.7.2.5 of Chapter 1 of the Evolutionary Requirements Document are not acceptable. The staff position is that all analysis methods used for the licensing design basis and safety margin basis must be approved by the NRC staff. The staff will evaluate this issue during its review of an individual application for FDA/DC.

## Structural Damping

Section 4.7.2.6 of Chapter 1 requires that structural damping values be based on confirmed test results, rather than conservative assumptions, whenever such data are available. In its letter dated April 24, 1991, the staff requested that EPRI submit the basis for using less conservative damping values. In its response of July 2, 1991, EPRI cited cable trays and hangers as examples for which extensive tests have been done by some utilities to generate more realistic damping values than those in RG 1.61. The staff position of using structural damping values for the design of structures is given in RG 1.61. However, the staff will not exclude the use of structural damping values based on test results. Therefore, it will evaluate the structural damping values during its review of an individual application for FDA/DC.

# Masonry Walls in Category I Buildings

Section 4.7.2.7 of Chapter 1 requires that masonry walls used as temporary or permanent partitions in Category I buildings be engineered as a substructure of the building, with consideration given to the effect they could have on safety-related items, and that they be designed according to the applicable

requirements of the Uniform Building Code (UBC). The use of the UBC for the design of masonry walls in Category I structures and the use of masonry walls in Category I buildings deviate from SRP guidelines. The staff concludes that the SRP guidelines should be used for the design of masonry walls located in the Category I structures of future ALWRs. In addition, it strongly discour-ages the use of masonry walls in Category I structures as load-carrying members or non-load-carrying members. The staff will waluate this issue against the criteria of Appendix A to SRP Section 3.8.4 during its review of an individual application for FDA/DC.

### Concrete Expansion Anchors

Section 4.7.2.8 of Chapter 1 specifies the use of the direct-bearing or undercut type of anchor bolts to ensure the ductile behavior of the bolt when high capacity is needed and the use of wedge and sleeve anchors for small loads. In its letter dated April 24, 1991, the staff questioned the use of expansion bolts for all safety-significant applications and encouraged qualification testing under field conditions. Where expansion anchors are used, the NRC requires the use of the conservative safety factors of Inspection and Enforcement Bulletin (IEB) 79-02, "Pipe Support Base Plate Designs Using Concrete Expansion Anchor Bolts," to account for uncertainty in field installation. In its response of July 2, 1991, EPRI acknowledged that its intent is to use expansion bolts only when necessary and that the expansion bolts will be of the undercut type (e.g., Maxibolts) in lieu o. the friction type. EPRI also noted that the conservative safety factors of IEB 79-02 are intended for friction-type expansion anchors and may not apply to Maxibolts. The response is not acceptable because the issue of uncertainty in field installation was not addressed and there is no assurance that the IEB 79-02 safety factors are not applicable to Maxibolts. Therefore, plant designers should submit to the staff the safety factors they propose to use for the capacity of the Maxibolts. The staff will evaluate this issue during its review of an individual application for FDA/DC.

### Stability of Shell-Type Structures Under Compression

Section 4.7.2.9 of Chapter 1 requires that the potential for global and local shell buckling be considered for shell-type structures under compression. In addition, it requires that, after appropriate consideration of the various uncertainties in materials, erection tolerances, and load description, a minimum factor of safety be maintained for all load combinations. In Revision 3 of the Evolutionary Requirements Document, the minimum factor of safety was based on ASME Code, Section III, Subsection NE, and supplemented by Code Case N-284.

Because Code Case N-284 provides lower safety factors against shell buckling than Subsection NE, the staff requested that EPRI alert the plant designer regarding the application of Code Case N-284 to asymmetric containments with large openings or provide specific conditions under which this code case can be used. In its response of July 2, 1991, EPRI stated that experienced professional containment vessel designers, working under the careful review of the utility owners and regulators, will not apply Code Case N-284 where its provisions do not apply. EPRI also stated that it was beyond the scope of the Evolutionary Requirements Document to explain the limitations of the specific code case. This is not acceptable to the staff. Subsection NE requirements

should be used for the evaluation of shell-type structures. As for Code Case N-284, the staff will evaluate its applicability during its review of an individual application for FDA/DC.

# Non-Seismic Structures

Section 4.7.2.10 of Chapter 1 requires that non-seismic structures, structural subsystems, and structural components be designed to the Zone 2A specifications of the Uniform Building Code. The use of Zone 2A is discussed by the staff in Sec. on 4.3.2.3 of this chapter.

### 4.7.3 Systems and Equipment

Section 4.7.3 of Chapter 1 provides discussions of design methodology requirements for systems and equipment that are more detailed than those in Section 4.4 of Chapter 1. The staff's evaluation of several of these requirements is given below.

# Use of ASME Code, Section III, Appendix N

Section 4.7.3.1 of Chapter 1 states that dynamic analysis techniques for safety class components will be in accordance with Appendix N of ASME Code, Section III. Appendix N is a nonmandatory appendix that is still evolving and does not currently agree with some staff positions. Therefore, it has not been endorsed by the staff, and the staff has no immediate plans to review it. In its letter dated May 17, 1991, the staff requested that EPki delete the reference to Appendix N and to reference applicable regulatory guides, SRP sections, or staff-approved ASME code cases in the requirement portion of Section 4.7.3.1. In its letter dated August 1, 1991, EPRI stated that only the rationale portion of Section 4.7.3.1 would be changed and that this change would only address the use of Code Case N-397. The issue of Code Case N-397 is discussed by the staff in Section 4.4.3 of this chapter. The use of Code Case N-397 is only one of several issues that are currently in Appendix N or are being proposed for future addenda to this document and that have not been endorsed by the staff. Some of these issues are damping values, use of the load coefficient method, use of the independent support motion response spectrum method of analysis, and the nonexceedance probability level in Subsection N-1725 of Appendix N. EPRI's response is not acceptable. The staff will evaluate this issue during its review of an individual application for FDA/DC in accordance with applicable SRP sections in lieu of Appendix N to ASME Code Section III. On the basis of the above discussion, this issue is closed.

# Use of ASME Code Cases N-411 and N-420 in the Same Analysis

Sections 4.7.3.2 and 4.7.3.11 of Chapter 1 allow the plant designer to use ASME Code, Section III, Code Cases N-411 and N-420 unconditionally. In its letter dated May 17, 1991, the staff requested that the following sentence be added to this requirement: "ASME Code Cases N-411 and N-420 may only be used in separate analyses and as further conditioned in RG 1.84." In its letter dated August 1, 1991, EPRI stated that since its intent to comply with RG 1.84 is indicated in Appendix B to Chapter 1 of the Evolutionary Requirements Document, this additional sentence is unnecessary. The staff's understanding of the Evolutionary Requirements Document is that a requirement in any section could override such a commitment in Appendix B. Therefore, this response is

unacceptable and the staff's position remains as stated in its letter of May 17, 1991. The staff will evaluate this issue during its review of an individual application for FDA/DC in accordance with the above position. On the basis of the above discussion, this issue is closed.

#### Use of Alternative Dynamic Analysis Methods

Sections 4.7.3.3 and 4.7.3.4 of Chapter 1 provide requirements that allow the plant designer to use dynamic analysis methods that have only been conditionally approved by the staff and are, therefore, not completely acceptable. These issues are discussed by the staff in Section 4.4.3 of this chapter and in the above section entitled "Use of ASME Code, Section III, Appendix N."

#### Requirement Applicable to Use of ASME Code Case N-411

In its letter dated May 17, 1991, the staff requested that EPRI revise Section 4.7.3.8 of Chapter 1 to clarify the use of a single damping value for both the OBE and the SSE. In its letter dated August 1, 1991, EPRI stated that the ALWR program had deleted the OBE from the design process and that the damping values in RG 1.61 for the SSE will be applicable to structures and systems except for piping, for which ASME Code Case N-411 is applicable. The elimination of the OBE is discussed by the staff in Section 4.4.3 of this chapter. The resolution of this issue may affect Section 4.7.3.8 of Chapter 1. The use of Cod- Case N-411, as stated in EPRI's response of August 1, 1991, is not completely acceptable. The staff requested that EPRI revise the requirement portion of Section 4.7.3.8 to include a requirement that Code Case N-411 be used only as conditioned by RG 1.84. In its response, EPRI stated that since Table B.1-2 of Appendix B to Chapter 1 indicates a commitment to comply with RG 1.84, this revision was unnecessary. The staff's understanding of the Evolutionary Requirements Document is that a requirement in any section could override such a commitment. Therefore, this portion of the response is unacceptable and the staff position remains as stated in its letter of May 17, 1991. The staff will evaluate this issue during its review of an individual application for FDA/DC, assuming that the requirement in Section 4.7.3.8 does not override the commitment to RG 1.84 in Appendix B to Chapter 1. On the basis of the above discussion, this issue is closed.

# Seismic Evaluation and Design of Small-Bore Piping

Revision 3 of Section 4.7.3.13 of Chapter 1 states that seismic Category I piping 2 inches and less in nominal diameter will be analyzed on the basis of the reference spectrum approach in NCIG-14 (EPRI NP-6628), "Procedure for Seismic Evaluation and Design of Small Bore Piping," dated April 1990, unless it encompasses in-line equipment with seismic qualification requirements or is connected to sensitive component nozzles. The staff is reviewing this document as a topical report, which was submitted to the staff by the Nuclear Management and Resources Council in its letter dated March 19, 1991. Pending completion of this review, the staff's position is that the methodology in EPRI NP-6628 is not acceptable. The staff will evaluate this matter during its review of an individual application for FDA/DC.

# Use of ASME Code, Section III, Subsection NF, and ANSI/AISC Standard N690

The first sentence in the requirement portion of Section 4.7.3.22 of Chapter 1 contains an acceptable commitment to the jurisdictional boundary rules of ASME

Code, Section III, Subsection NF. On the basis of EPRI's response of August 1, 1991, to a staff request for additional information dated May 17, 1991, it is the staff's understanding that the remainder of this section provides clarification only and does not permit the plant designer to take exception to this commitment. On the basis of this understanding, the staff concludes that the information in Section 4.7.3.22 is consistent with staff positions relative to jurisdictional boundary rules and is acceptable. For further clarification, the staff's position on this issue is discussed below.

The ongoing effort with regard to ASME Code, Section III referencing ANSI/AISC N690 in Subsection NF is not directly related to the issue of jurisdictional boundary. When this effort is complete and endorsed by the staff, all supports identified as falling under the jurisdiction of Subsection NF will be constructed to the rules of that subsection. Implicitly, the rules for the design portion of construction will then be in accordance with a modified version of ANSI/AISC N690.

### Rules for Construction of Core Support Structures

Section 4.7.3.23 of Chapter 1 states that core support structures will be designed to the criteria in ASME Code, Section III, Subsection NG. In its letter dated May 17, 1991, the staff requested that this requirement be revised to read: "Core support structures will be constructed to the criteria specified in ASME Code, Section III, Subsection NG, where 'construction' is as defined in ASME Code, Section III, NB/NC/ND-1100(a)." In its letter dated August 1, 1991, EPRI agreed with the staff's request, except that the requirement still contains the words "designed to" rather than "constructed to," which is not completely acceptable. During its reviews of individual applications for FDA/DC, the staff will require that core support structures be constructed to the rules of ASME Code, Section III, Subsection NG, where "construction" is either as defined above or as defined in ASME Code, Section III, Subsection NG-1110. On the basis of the above discussion, this issue is closed.

# Commitment to I&E Bulletin 79-02

In its letter dated May 17, 1991, the staff requested that a commitment be made in Section 4.7.3 of Chapter 1 of the Evolutionary Requirements Document that the applicable action items in I&E Bulletin 79-02, "Pipe Support Base Plate Designs Using Concrete Expansion Anchor Bolts," Revision 2, dated November 8, 1979, will be met for pipe support base plate designs using concrete expansion anchor bolts. In its response dated August 1, 1991, EPRI stated that since the only drilled-in anchors permitted for the ALWR are to be of the "undercut" type, the safety factors in Bulletin 79-02 should not be applicable. This response is not completely acceptable. The staff's position on this issue is as follows:

- In lieu of the safety factors in I&E Bulletin 79-02, EPRI should provide the factors that will be used in the design of the undercut type of expansion anchor bolt and the basis for these factors.
- Irrespective of the type of expansion anchor bolt that will be used, the staff requires a commitment to the action item in I&E Bulletin 79-02 relative to pipe support base plate flexibility.

.

The staff will evaluate this issue during its review of an individual application for FDA/DC. On the basis of the above discussion, this issue is closed.

### Use of ASME Fatigue Design Curves

Section 3.3 of Chapter 1 states that the plant design life for the ALWR will be 60 years. This proposed design life raises questions relative to the margins available in the current ASME fatigue design curves. These margins were established almost 30 years ago and were obtained from best-fit curves of fatigue test data by applying a factor of either 2 on stress or 20 on cycles, whichever was more conservative at each point. These factors were originally intended to cover such effects as environment, size effect, and scatter of data. However, on the basis of limited data currently available, the staff concludes that these margins may not be sufficient to account for variations in the original fatigue test data as a result of various environmental effects. In its letter dated May 17, 1991, the staff requested a commitment in Section 4.7.3 of Chapter 1 to consider such effects in the designs of applicable ASME Code Class 1 systems, components, and equipment. In its letter dated August 1, 1991, EPRI stated that if additional data or research results yield findings requiring changes to the current ASME fatigue design curves, the code consensus process will provide the proper vehicle to address such findings. The staff does not agree with all of the discussions in this response and concludes that the above commitment is not completely acceptable. The ASME Code curves may not be revised for many years. Therefore, the staff's position is that until these curves are revised, all ALWR applicants and all licensees applying for license renewal should propose appropriate fatigue design curves that will be reviewed by the staff. For the Evolutionary Requirements Document, a commitment to this position would be sufficient. Pending such a commitment, the staff will evaluate this issue during its review of an individual application for FDA/DC. On the basis of the above discussion, this issue is closed.

The staff is assessing all available data relative to environmental effects on fatigue for austenitic stainless steel and ferritic steel. The objective of this effort is to propose interim fatigue curves that may be used in lieu of the current ASME curves until ASME has acceptably addressed this concern. Pending staff implementation of these interim curves, the position stated above will be in effect.

### Conclusion

The staff concludes that, with the exceptions noted above, the requirements in Section 4.7.3 of Chapter 1 of the Evolutionary Requirements Document relative to design methodology for systems and equipment are consistent with applicable SRP sections, regulatory guides, and staff positions and are acceptable.

#### 4.8 Testing and Qualification

Section 4.8 of Chapter 1 of the Evolutionary Requirements Document provides requirements for the seismic, dynamic, and environmental qualification of mechanical and electrical equipment.

# 4.8.1 Seismic and Dynamic Qualification of Mechanical and Electrical Equipment

In Sections 4.4.C(7) and 4.7.A of the DSER for Chapter 1, the staff stated that EPRI had committed to revise these sections to state that the plant designer will make use of qualification by experience as permitted by governing codes and standards. This commitment was met in Revision 1 of Sec-tion 4.8.1 of Chapter 1. In addition to this commitment, Section 4.8.1 contains broad and generally acceptable requirements relative to qualification by testing, analysis, and combined testing and analysis. These gualification methods will be implemented by meeting the rules in Institute of Electrical and Electronic Engineers (IEEE) 344-1987, "Recommended Practices for Seismic Qualification of Class IE Equipment for Nuclear Power Generating Stations." This standard contains acceptable criteria with the exception of a portion of the rules relative to qualification using the seismic experience data base. In addition, Revision 2 of Table B.1-2 of Appendix B to Chapter 1 contains a commitment that EPRI will comply with RG 1.100. "Seismic Qualification of Electric and Mechanical Equipment for Nuclear Power Plants," Revision 2, which contains certain staff positions in addition to the rules of IEEE 344. However, Section 4.8.1 of Chapter 1 still contains unacceptable requirements and rationale relative to the use of the seismic experience data base. These issues are discussed below.

### Seismic Qualification of Equipment by Experience

The applicable portions of the NRC's regulations governing the seismic design of nuclear power plants require that systems, structures, and components important to safety be designed to withstand the effects of earthquakes and that they be seismically qualified to perform their safety functions. The regulations stipulate that seismic qualification of such equipment will be demonstrated by either a suitable dynamic analysis or a suitable qualification test. There are no explicit provisions in the regulations that accept experience data as a means of seismic qualification.

The NRC has not approved the use of the earthquake experience data base methodology as a seismic qualification method. Rather, it has concluded that the NRC-approved, experience-based methodology, when used in conjunction with appropriate restrictions and caveats, may be an acceptable means of verifying the seismic adequacy of certain equipment in certain operating nuclear power plants.

The extent to which the NRC has approved or endorsed the application of earthquake experience data for electrical and mechanical equipment in nuclear power plants has been restricted solely to the resolution of Unresolved Safety Issue (USI) A-46, "Verification of Seismic Adequacy of Mechanical and Electrical Equipment in Operating Reactors." In 1980, USI A-46 was formalized as a result of the safety concern that equipment needed to achieve and maintain safe shutdown in certain operating plants (i.e., those plants whose seismic licensing criteria were not reviewed for acceptability against IEEE 344-1975, RG 1.100, and SRP Section 3.10, "Seismic Qualification of Category I Instrumentation and Electrical Equipment") may not have been adequately qualified to ensure its survivability and functionality in the event of a safe shutdown carthquake. The staff discusses USI A-46 further in Section 3 of Appendix B of this chapter. The NRC determined that it was not feasible to require those older operating plants to meet the later licensing requirements, since direct application of those criteria to older plants could have required extensive, and probably impractical, modifications of those facilities. The NRC subsequently concluded that the use of earthquake experience data with appropriate restrictions and caveats, supplemented by some test results to verify the seismic capability of equipment within certain specified earthquake motion bounds, represented the most reasonable and cost-effective means of ensuring that the purpose of the NRC regulations related to seismic design is met for those plants. One of the programmatic restrictions includes the exclusion of the application of the earthquake experience in verifying the adequacy of structures and piping.

The NRC has since received requests from utilities and industry organizations proposing the use of the earthquake experience data base methodology as a qualification method for various applications, including its application to ALWRs by EPRI. However, the NRC has not approved or endorsed any of these initiatives. In Section 4.8.1.8 of Chapter 1, EPRI proposes to use the Seismic Qualification Utility Group (SQUG) data base as one means of seismic qualification. The staff concludes that this portion of Section 4.8.1.8 is not acceptable.

Current NRC guidance (RG 1.100, Revision 2) recognizes the use of experience data as a means of seismic qualification of equipment. However, the earthquake experience data base methodology described in IEEE 344-1987 - which, as stated in RG 1.100, Revision 2, is to be evaluated by the staff on a case-bycase basis - is different from the detailed criteria and approach in the SQUG Generic Implementation Procedure (GIP). The staff does not accept the GIP as a qualification procedure. Rather, it is a verification procedure and is intended to be used only at the older operating plants under the USI A-46 resolution. Since the staff does not accept the GIP as a qualification procedure, it is not applicable to newer operating reactors or future ALWR plants. The development of the GIP verification procedures and criteria was not necessarily based on the required elements of IEEE 344-1987 or staff requirements for newer operating reactors. Thus, a significant portion of the data base in the A-46 methodology is not applicable to future ALWRs.

Therefore, consistent with RG 1.100, Revision 2, the staff will evaluate the use of experience data during its review of an individual application for FDA/DC (see Section 3 of Chapter 2 of this report for a specific application of experience data). On the basis of the above discussion, these DSER open and confirmatory issues are closed.

### Number of Seismic Events Used in Equipment Qualification Programs

Section 4.8.1.1 of Chapter 1 repeats the requirement in Section 4.5.2.4.4.1 that reduces the number of full-stress cycles for 1/2 SSE from 50 to 20. This issue is discussed by the staff in Sections 4.4.3 and 4.5.2 of this chapter.

#### Criteria for Equipment Anchorage

In Section 4.7.A of the DSER for Chapter 1, the staff stated that EPRI had committed to revise the information in this section relative to anchorage criteria for equipment to be qualified. Section 4.8.1.3 of Chapter 1 contains a requirement that the plant designer specify the anchorage criteria for

equipment to be qualified or adopt the seismically qualified anchorage from the equipment manufacturer. The staff's interpretation of this requirement is that the plant designer will determine that the manufacturer's seismic qualification procedures are consistent with all of the ALWR qualification requirements approved by the staff. In addition, the plant designer will ensure that the "as-qualified" anchorage used in the qualification program corresponds to the "as-installed" anchorage actually used in the plant. Any differences between these two configurations will be resolved by the plant designer. The staff concludes that these requirements are consistent with current staff positions and are acceptable. Therefore, this DSER confirmatory issue is closed.

# Conclusion

The staff concludes that, with the exceptions noted above, the requirements in Section 4.8.1 of Chapter 1, supplemented by the staff's position on equipment seismic qualification by experience, provide reasonable assurance that an acceptable program for seismic and dynamic qualification of electrical and mechanical equipment will be implemented for the ALWR.

4.8.2 Environmental Qualification of Mechanical and Electrical Equipment

In Table B.1-2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI has committed to comply with SRP Section 3.11, "Environmental Qualification of Mechanical and Electrical Equipment". The staff concludes that this commitment is acceptable. Section 4.8.2.1 of Chapter 1 of the Evolutionary Requirements Document states that Class 1E electrical equipment will be environmentally qualified in accordance with 10 CFR 50.49, as outlined in IEEE 323, "Recommended Practices for Seismic Qualification of Class 1E Equipment for Nuclear Power Generating Stations." The staff finds that IEEE 323-1974 is acceptable as outlined in 10 CFR 50.49, but has not found IEEE 323-1983 acceptable. Where differences exist between IEEE 323 and 10 CFR 50.49, the designer must follow the NRC regulation or identify and justify the differences for the staff to review. The staff will evaluate this matter during its review of an individual application for FDA/DC.

In Table E 1-2 of Appendix B to Chapter 1, EPRI states that RG 1.89, "Qualification of Class 1E Equipment for Nuclear Power Plants," is an optimization subject. Further, Section 4.4.3.3.7 of Chapter 1 states that for environmental qualification of plant equipment, the ALWR source term will be a physically based source term in lieu of the source term specified in Atomic Energy Commission Technical Information Document (TID) 14844. As a result of reviewing the EPRI-proposed source term, the NRC staff is developing a revised source term. The staff position is that the source term should be based on the revised source term to be issued by the staff.

It should be noted that ALWRs will be designed for 60 years of operation, while the current plants are designed for 40 years. The staff concludes that the plant designer should ensure that plant equipment important to safety will be qualified for its intended service and will be able to perform its safety functions throughout its design life. The staff will address this issue during its review of an individual application for FDA/DC.

Section 4.8.2.4 of Chapter 1 states that qualification will be accomplished by physical test or by experience, demonstrating the equipment's similarity to

previously qualified equipment or to equipment that has been exposed to other more severe environments. The staff finds that the above statement can easily be misinterpreted and, therefore, needs to be clarified by stating that the method of qualification should be in accordance with 10 CFR 50.49(f). The staff will review the plant-specific designs against the requirements of 10 CFR 50.49(f), which requires that each item of electric equipment important to safety be qualified by one of the following methods:

- testing an identical item of equipment under identical conditions or under similar conditions with a supporting analysis to show that the equipment to be qualified is acceptable
- testing a similar item of equipment with a supporting analysis to show that the equipment to be qualified is acceptable
- experience with identical or similar equipment under similar conditions with a supporting analysis to show that the equipment to be qualified is acceptable
- analysis in combination with partial-type test data that support the analytical assumptions and conclusions

The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

Re- gion	Maximum wind speed (mph)	Rota- tional speed (mph)	Translat- ional speed (mph)	Radius of maximum rota- tional speed (feet)	Pres- sure drop (psi)	Rate of pres- sure drop (psi/ sec)
I	300	240	60	150	2.0	1.2
II	220	170	50	150	1.0	0.5
III	200	160	40	150	0.9	0.3

Table 1.1 Design-Basis Tornado Characteristics

# 5 MATERIALS

# 5.1 Introduction

Section 5 of Chapter 1 of the Evolutionary Requirements Document provides general guidance regarding the selection of materials and the methods that will be used to fabricate nuclear power plant components. The requirements are intended to prevent degradation of the materials from a large number of causes such as intergranular attack, stress corrosion cracking, crevice corrosion, thermal and mechanical fatigue, radiation embrittlement, welding failures, bolting failures, and casting flaws.

# 5.2 General Requirements

# 5.2.1 Responsibility for Materials Selection

Section 5.2.1 of Chapter 1 of the Evolutionary Requirements Document requires that the materials for the ALWR be specified by the plant designer and have been proven suitable in service. If the plant designer proposes to use unproven material, a written justification must be provided stating why the unproven material is being selected and the basis on which it is considered to be acceptable.

The staff considers these requirements acceptable because the design choices will be constrained to ensure the chosen materials are suitable for the intended service.

# 5.2.2 Identification of Materials in Critical Components

Section 5.2.2 of Chapter 1 requires that the materials for critical components, the criteria for selecting the materials, and any special requirements that may be required for the selected materials be identified.

The staff considers these requirements acceptable because they will ensure that safety-related plant components will be fabricated using materials that are compatible with the intended service.

# 5.2.3 Codes and Standards

Section 5.2.3 of Chapter 1 requires that the specified materials meet the requirements of the applicable design and construction codes and standards and that the plant designer consider the actual service condition to determine if more restrictive conditions are needed than those imposed by code specifications. The requirements of Regulatory Guide 1.85, "Materials Code Case Acceptability - ASME Section III, Division 1," also apply to ALWRs.

The staff considers these requirements acceptable because they comply with existing NRC regulations.

# 5.2.4 Design-Basis Consideration

Section 5.2.4 of Chapter 1 requires that materials used in the ALWR plant meet the special conditions imposed by the design bases incorporated in the Evolutionary Requirements Document. A minimum number of types and grades of

materials must be used, where practical, consistent with service conditions and design performance objectives. The design life requirements for the materials must incorporate all required environmental and service conditions, including off-normal conditions that affect the life of the component (e.g., composition of the cooling water, steam pressure, and radiation level). Preference must be given to designs that do not push material limits and that make use of conventional materials applied well within the limits for which successful experience has been obtained. In particular, high-strength bolts or fasteners will not be used where practical; rather, sufficiently robust designs will be used that do not require special, high-strength materials.

The staff considers these requirements acceptable. The use of a minimum number of types and grades of materials will simplify plant design. Operating experience has shown that all service conditions must be considered in the selection of materials.

# 5.2.5 Hazardous Materials

Section 5.2.5 of Chapter 1 requires that the use of materials that present a hazard to personnel or equipment (e.g., radioactivity, toxicity, flammability, or corrosivity) be limited to those applications where no satisfactory alternatives exist. If hazardous materials must be used, the plant designer will specify shipping, storage, handling, and usage requirements for the material to minimize the hazards and satisfy applicable regulations and practice. A building, meeting the requirements of the Environmental Protection Agency, will be provided for storage of hazardous and toxic wastes awaiting disposal.

The staff concludes that these requirements do not conflict with the Commission's regulations are, therefore, acceptable.

# 5.2.6 Review of LWR Experience

Section 5.2.6 of Chapter 1 requires that plant designers, when selecting ALWR materials, consider the lessons learned from the operation of existing plants and the measures necessary to prevent problems from recurring. The ALWR plant designer will conduct a specific review of LWR experience (both domestic and foreign) to identify significant materials problems and proven methods of resolving them. This review will supplement and update the review of experience that is inherent in the Evolutionary Requirements Document. For the review, the designer should fully use information available at the time the plant is designed from such sources as EPRI-sponsored research and studies and from government, academic, and industry sources. The problems identified in this review will be listed in a report to be submitted to the plant owner, and appropriate materials requirements will be incorporated into the design of the ALWR plant.

The staff considers these requirements acceptable. The incorporation of lessons learned from operation of existing plants will ensure that materials problems experienced in these plants will be prevented from recurring.

# 5.2.7 Metallic Materials

Section 5.2.7 of Chapter 1 requires that metallic materials in contact with reactor coolant be corrosion resistant, such as austenitic stainless steel or

carbon and low-alloy steels, with an adequate consideration of corrosion; resistant to detrimental forms of corrosion, such as intergranular attack, stress corrosion cracking, or contact corrosion between different materials; and restricted in cobalt content to as low a level as practical for all components that are made of stainless steel or nickel-based alloy and that have a large wetted surface area (e q., steam generator tubing, major piping, cladding). For such components fabricated with stainless steel or nickel-based alloys, the cobalt content will be restricted to 0.020 weight percent or less. Cobalt-based alloys will be avoided except in cases for which no proven alternative exists. The plant designer will identify to the plant owner all applications of cobalt-based alloys in wear applications and state the basis for their use. In addition, ferritic pressure boundary materials will be resistant to brittle fracture and will satisfy Appendix G to 10 CFR Part 50. The nil ductility temperature ( $RT_{NDT}$ ) of these materials will not exceed 10 °F.

Lead, antimony, cadmium, indium, mercury, zinc, bismuth, and tin metals and their alloys will not be allowed to come in contact with reactor coolant system (RCS) primary components or secondary components made of stainless steel or Inconel. For bearings in the secondary system, the plant designer will demonstrate that the bearing design is such that harmful amounts of material with a low melting point will not enter the feedwater to jeopardize stainless steel or Inconel components.

Zinc may be added to the coolant in BWRs in limited, controlled amounts. Copper alloys may be used for very limited, controlled applications in BWRs, such as the low-temperature pump bearings, where the material cannot enter the reactor coolant.

EPRI states that the plant designer will select materials for the reactor vessel support structures that are resistant to brittle fracture and experience a minimal shift in RT<sub>NDT</sub> as a result of neutron fluence for the support scheme used. The plant designer will provide an analysis of the reactor vessel support structure to demonstrate its acceptability for the maximum design plant life and capacity factor and to identify any operational constraints to avoid the potential for brittle fracture. Sufficient access will be provided to the reactor vessel support structure to enable monitoring of the material temperature and performing modifications or heat treatment of the structural materials, if required.

The staff considers these general requirements acceptable because only corrosion-resistant materials will be used in contact with the reactor coolant. Also, ferritic materials are required to satisfy the requirements of Appendix G to 10 CFR Part 50. Metals and alloys with low melting points will not be used in contact with the reactor coolant.

To reduce general radiation fields resulting from the presence of cobalt-60 in the oxide layer of the RCS piping, zinc may be added to the coolant in BWRs in limited, controlled amounts. Zinc injection reduces the radiation fields by replacing the cobalt with zinc in the piping oxide layer. One of the side effects of zinc injection is the creation of zinc-65, which increases piping dose rates and requires special consideration during radioactive waste disposal. EPRI is investigating a way to solve this problem by using a zinc

isotope depleted in zinc-64. The staff will review this issue again at the vendor application stage to determine what advances have been made in this area.

The staff finds the restrictions on the use of cobalt materials in contact with reactor coolant to be acceptable because activated cobalt-60 is a major source of occupational exposure in nuclear power plants.

## 5.2.8 Non-Metallic Materials

Section 5.2.8 of Chapter 1 requires that the impurity levels of non-metallic materials used within the nuclear steam supply system (NSSS) and associated systems be controlled to the limits specified below. These limits apply to specific locations within the plant.

For PWR RCS applications, non-metallic materials will meet the impurity limits specified below under any of the following conditions:

- The material is in contact with reactor coolant in service.
- The material is applied to surfaces that will be in contact with reactor coolant and the material is not completely removed before service.
- The material is in contact with stainless steel or nickel alloys on external surfaces of RCS components either by design or because it is not completely removed before service.
- The material is exposed to a radiation dose greater than 1E+5 rads during component life.

Impurity Limits

Chlorine	200 parts per million (ppm) maximum
Mercury (Hg)	1 ppm maximum
Arsenic (As)	2 ppm maximum
Lead (Pb)	10 ppm maximum
Sulfur (S)	200 ppm maximum
Zinc (Zn)	200 ppm maximum
Combined Hg, As, Pb, S, Zn	300 ppm max1mum

Examples of unacceptable non-metallic materials are polyvinylchloride (PVC), polytetrafluoroethylene (Teflon), fluorosilicones, and neoprene.

For PWR secondary system components where the non-metallic materials may contaminate the feedwater or are in the form of substances applied to clean stainless steel or nickel-based alloys used in secondary system components, the impurity limits are as specified below:

Total	chlorine plus	fluorine	500	ppm	maximum
Heavy	metals (total	Hg, As, Pb)	250	ppm	maximum
Total	sulfur		500	ppm	maximum

For the balance of PWR applications, non-metallic materials used inside the containment that can come in contact with steel or nickel-based alloys during fabrication, shipping, and storage will not contain more than the following:

Total chlorine Total fluorine Active sulfur Elements in metallic form with a low melting point

500 ppm maximum 300 ppm maximum 700 ppm maximum

500 ppm maximum total or 200 ppm maximum for any individual element

For BWR applications, non-metallic materials that are in contact with reactor coolant during plant operation, either as a result of design or because they are not completely removed after fabrication or installation, and that remain in direct contact with stainless steel or nickel-based alloys will meet the limits specified above for the balance of PWR applications.

The staff considers these requirements acceptable because non-metallic materials will be controlled to acceptable impurity levels. This will ensure that non-metallic materials do not adversely affect the corrosion resistance and ductility of metal components. However, EPRI should revise the Evolutionary Requirements Document to include limits on nitrites, nitrates, and total halogens as chlorine. In addition, a total limit on total chlorine + total sulfur + total nitrite + total nitrate expressed as mole-equivalents of chlorine should also be included. Pending such a revision, the staff will evaluate an individual application for FDA/DC or a combined license to ensure that such limits are imposed.

## 5.3 Materials Selection

5.3.1 Materials in the Reactor Coolant System and Related Systems

### Wrought Austenitic Stainless Steels

Section 5.3.1.1 of Chapter 1 of the Evolutionary Requirements Document specifies that the use of austenitic stainless steel will be subject to the following requirements:

- Materials will be in the solution heat-treated condition, or solution heat treated at a later step, regardless of service temperature. Heat treatment will be done at 1900 °F minimum metal temperature, followed by a gualified cooling process.
- Grain size and uniformity will be controlled in the material to ensure that adequate ultrasonic tests (UTs) can be performed where required. Requirements for attenuation may be used instead of controlling grain size and uniformity.
- Materials for use at temperatures above 200 °F in borated water or that are part of the RCS pressure boundary will be tested to verify freedom from sensitization.

- Cold-work control of materials for service above 200 °F in borated water or that are part of the RCS pressure boundary will meet the following requirements:
  - Hardness of austenitic stainless steel raw materials will not exceed 92 HRB.
  - Hardness will be controlled during fabrication by process control of bending, cold forming, straightening, or other similar operation.
- In PWRs, cold-worked austenitic stainless steel may be used for small parts (e.g., pins, fasteners, and sleeves) if no proven alternative is available; however, in those cases, the following specific requirements will be met:
  - All such applications of cold-worked stainless steel will be identified as "critical."
  - The technical justification of each application will be documented, including relevant laboratory and service experience.
  - The technical justification will include a description of the process controls that will be applied to ensure the material is subjected to the proper amount of cold work.
  - The following manual grinding controls are applicable to a BWR and are desirable but not required for a PWR:
    - Grinding performed before solution heat treatment requires no additional control except that after heavy grinding, light grinding or flapping will be required.
    - Grinding will be restricted except for fit-up, nondestructive testing, or to remove scratches or surface defects.
      - All grinding of austenitic stainless steel surfaces exposed to reactor water in service that are not subsequently solution heat treated will be performed in accordance with a written procedure. The procedure will involve finishing the ground surface with successively finer grit sizes to remove the bulk of cold-worked material, with the final grit size no coarser than #120 grit.
    - For all stainless steel welding materials including consumable inserts for components that operate above 200 °F, in borated water, or that are part of the RCS pressure boundary, the average ferrite content will be in the range of 5 FN (ferrite number) to 13 FN (ferrite content of undiluted weld deposits will be determined by magnetic measurements as specified in ASME Code, Section III, Division 1 (Winter 1976 or later).

The staff's evaluation regarding intergranular stress corrosion cracking of austenitic stainless steel is provided in Section 5.3.1.8 of this chapter.

In addition, the staff requires that the grinding controls also be applied to PWR applications. The staff will evaluate this issue during its review of an application for a combined license.

The staff concludes that, with the exception noted above, the implementation of these requirements will ensure that wrought austenitic stainless steel will perform in service as designed.

### Martensitic Stainless Steel

Section 5.3.1.2 of Chapter 1 requires that the use of wrought and cast martensitic stainless steel be subject to the following requirements:

- All components made from wrought and cast forms of 12 percent chromium martensitic stainless steel will be proper; heat treated for the intended service. Proper heat treatment consists of normalized and tempered or quenched and tempered. For applications in which stress corrosion cracking has been identified as a concern, the heat treatment will be controlled to limit the hardness. The hardness limits can vary depending on application. Hardness limits will be specified on the basis of existing industry experience and on relevant ASME codes and ASTM standards.
- Weld repairs on all wrought and cast parts will be followed by a full heat treatment. Localized post-weld heat treatment is not permitted.

In Section 5.3.A.2 of the DSER for Chapter 1, the staff determined that the ALWR requirements regarding hardness of martensitic and precipitation-hardened stainless steel were insufficient. The hardness of martensitic and precipitation-hardened stainless steel must be kept below that specified in the nuclear industry today. The hardness of 40 (equivalent Rockwell scale of C) is too high for any application; a hardness of 25 may even be too high for some applications. The harder the stainless steel, the lower the fracture toughness and the greater its susceptibility to stress corrosion cracking.

EPRI revised Sections 5.3.1.2.1 and 5.3.1.6.6 of Chapter 1 to state: "Hardness limits shall be specified on the basis of existing industry experience and on relevant ASME codes and ASTM standards." Therefore, an individual FDA/DC or combined license applicant will be required to specify hardness limits on the basis of Section III of the ASME Code and American Society for Testing and Materials (ASTM) standards. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license. On the basis of the above discussion, this DSER confirmatory issue is closed.

The staff concludes that the implementation of these requirements will ensure that martensiiic stainless steels will perform in service as designed.

### Nickel, Chromium, and Iron (Ni-Cr-Fe) Alloys

Section 5.3.1.3 of Chapter 1 requires for the PWR that Alloy 600 be restricted to applications outside the pressurizer and to applications for which lowcarbon stainless steel cannot be used. For PWR applications requiring the use of the special properties of Alloy 600 (e.g., resistance to chloride stress corrosion cracking and low thermal expansion coefficient), the application

must be justified. Justification should be based on need, the lack of a suitable alternative. evidence of successful service performance under the specific conditions anticipated, and a review of relevant laboratory experience. Ease of inspection and replacement also will be considered. Alloy 600 will be given a special thermal treatment to improve resistance to stress corrosion cracking. The use of Alloy 690 will be restricted to steam generator tube applications.

For BWR applications requiring the use of the special properties of Ni-Cr-Fe alloys (e.g., strength or low thermal expansion coefficient), the applications must be justified. Justification should be based on need, the lack of a suitable alternative, evidence of successful service performance under the specific conditions anticipated, and a review of relevant laboratory experience. Ease of inspection and replacement also will be considered.

Pickling of wetted surfaces for all Ni-Cr-Fe alloys is prohibited for both BWRs and PWRs.

In Section 5.3.A.3.b of the DSER for Chapter 1, the staff identified a concern regarding the use of Alloy 600 in future ALWRs. Section 5.3.1.3.1 of Chapter 1 of the Evolucionary Requirements Document specifies that Alloy 600 can be used in ALWRs. EPRI has revised Section 5.3.1.3 to place restrictions on the use of Alloy 600. EPRI states that the designer will not use Alloy 600 in the steam generators and pressurizers. For PWR applications, Alloy 600 will be given a special thermal treatment to improve resistance to stress corrosion cracking. However, in general, the staff considers Alloy 600 undesirable for use in the ALWR because of the stress corrosion cracking experienced in existing nuclear plants. It specifically discourages the use of Alloy 600 in steam generators and pressurizers. Therefore, it will require that the applicant for any standard design application identify the use of Alloy 600 and provide information concerning its use. Those applications will be reviewed and approved by the staff on a case-by-case basis. In addition, the use of other Ni-Cr-Fe alloys such as Alloy 690 or 800 should be considered in applications for which primary water stress corrosion cracking is a concern. These applications also will be reviewed on a case-by-case basis. On the basis of the above discussion, this DSER confirmatory issue is closed.

The staff concludes that, with the exception noted above, the implementation of these requirements will ensure that Ni-Cr-Fe alloys will perform in service as designed.

### Austenitic Stainless Steel Castings

Section 5.3.1.4 of Chapter 1 requires that austenitic stainless steel castings meet the following requirements:

- All castings will be solution heat treated after casting to a minimum of 1950 °F metal temperature followed by a qualified cooling process.
- For all austenitic stainless steel castings that operate in a water or steam environment at a temperature of 200 °F or greater, in borated water or that are part of the RCS pressure boundary, the ferrite content will be between 8 FN and 30 FN. The ferrite content will be determined using ASTM A800.

- Radiographic quality will be controlled by the applicable codes and standards.
- The filler metal used for weld repairs will be 308L, and the average ferrite content of the filler metal will be a minimum of 5 FN and a maximum of 13 FN. Measurements will be made on the as-deposited undiluted weld pads. For hard surfacing applications, loss of delta ferrite during subsequent solution heat treatment will be limited to an acceptable level that will be demonstrated by process gualification.
  - To minimize the detrimental effects of thermal aging in cast austemitic stainless steels, the following metallurgical factors will be controlled:
    - Ferrite content will not exceed 30 FN.
    - Molybdenum will not be used as an alloying element unless it can be demonstrated that sigma phase embrittlement will not affect design requirements.
    - Analysis and/or accelerated testing will be performed to ascertain if the toughness will not decrease below the limit required by design at the end of the component service life.
    - Weld joints between austenitic stainless steel castings will not be used if inservice ultrasonic inspection of the weld between the castings will be required.

The staff concludes that the implementation of these requirements will ensure that austenicic stainless steel castings will perform in service as designed.

### Carbon and Low-Alloy Steel Materials

.

Section 5.3.1.5.1 of Chapter 1 requires that allowance be made for corrosion (generil, pitting, and crevice) of any unprotected carbon steel materials exposed to a water environment. This allowance will be based on industry experience at the time the plant is designed. The review of experience will specifically include establishing a corrosion allowance and identifying its technical basis. In addition, the following requirements will be met for the material product forms listed below.

Pressure Vessel Steel

Section 5.3.1.5.2 of Chapter 1 specifies the following requirements for pressure vessel steel:

- SA533 Grade B Class 1 plate and SA508 Class 1, 2, and 3 forging materials will be used for primary coolant pressure boundary components.
- PWR pressure vessels and piping made of low-alloy or carbon steel in contact with the reactor coolant will be clad with austenitic stainless steel. The cladding will have a minimum of deposited ferrite between 5 FN and 13 FN.

For BWRs, unclad low-alloy or carbon steel may be used for such items as the reactor vessel head, nozzle areas, and piping systems. An appropriate allowance for corrosion will then be used. Ind the cleanup system will be sized appropriately to remove resultant corrosion products. If hydrogen water chemistry is used, the reactor vessel will be completely clad with austenitic stainless steel. In this case, technical justification of the type of cladding material and the cladding process must be provided on the basis of past experience and the specific environmental conditions.

# Seamless Pipe

Section 5.3.1.5.3 of Chapter 1 specifies that seamless plain carbon steel pipe be SA333-6. Wherever possible, seamless pipe is to be used instead of welded pipe for Class 1 systems inside the containment. The SA333-6 piping will be normalized or normalized and tempered.

#### Welded Pipe

Section 5.3.1.5.4 of Chapter 1 requires that welded plain carbon steel pipe be SA671-Grade CC70 or SA333-6. The impact properties of the finished pipe and weld metal will be 13 ft-1b minimum at -50 °F. The welded pipe will be normalized or normalized and tempered.

### Plate

Section 5.3.1.5.5 of Chapter 1 requires that plain carbon steel plate be SA516. The impact properties will be 13 ft-lb minimum at -50 °F.

#### Forgings

Section 5.3.1.5.6 of Chapter 1 requires that plain carbon steel forgings be SA350-Grade LF2, SA508 Class 1, or SA105. The forgings will be heat treated by normalizing or r malizing and tempering. Impact properties will meet the requirements for SA350-Grade LF2.

#### Castings

Section 5.3.1.5.7 of Chapter 1 requires that plain carbon steel castings be SA352-Grade LCB or SA216 WCB. Impact requirements for SA352-Grade LCB will be met. Cast iron will not be used in essertial functional parts of safety-related components such as valve yokes.

# Fittings

Section 5.3.1.5.8 of Chapter 1 requires that plain carbon steel fittings be SA420-Grade WPL-6.

# Evaluation

In Section 5.3.A.5 of the DSER for Chapter 1, the staff stated that the corrosion allowance for the carbon and low-alloy steel in the Evolutionary Requirements Document was inadequate because experience at operating nuclear plants has shown that the accepted standard corrosion allowance
is inadequate. The plant designer should consider all types of corrosion (i.e., microbiological, pitting, and crevice) in the design. EPRI added the following to Section 5.3.1.5 of Chapter 1: "The allowance shall be in conformance with current industry experience." The staff concludes that the above revision is unacceptable because conformance with current industry experience lacks specificity.

The staff concludes that the plant designer should control corrosion mechanisms, such as general, pitting, crevice, and microbiological corrosion, in the design of piping systems of primary and secondary systems. The general corrosion allowance should comply with the corrosion allowance specified in Section III of the ASME Code and ANSI/ASME B.31.1, "Power Piping." For the specific corrosion allowance (such as that for microbiological corrosion), the plant designer may use industry methodologies such as the EPRI computer code CHECMATE. However, the staff will have to review and approve any industry methodology before it is used in the ALWR design. The staff will evaluate this matter during its review of an individual application for FDA/DC. On the basis of the above discussion, this DSER confirmatory issue is closed.

The staff concludes that, with the exception noted above, the implementation of the requirements in Section 5.3.1.5 of Chapter 1 will ensure that carbon and low-alloy materials perform in service as designed. However, the impact properties of welded pipe and plate material also must meet the requirements of Table NB-2332(a)-1 of Section III of the ASME Code.

# Precipitation-Hardened Stainless Steel

Section 5.3.1.6 of Chapter 1 requires that precipitation-hardened stainless steel meet the following requirements:

- Precipitation-hardened stainless steel will be in accordance with the ASME requirements for SA564 Type 630 (17-4PH). No other precipitationhardened stainless steel is permitted unless specifically identified in the intended application and justified on the basis of need, lack of suitable alternatives, and good service experience in the specific application being considered.
- The material is to be used in the solution heat-treated and aged condition. The material will be heat treated to provide the required mechanical properties and resistance to stress corrosion cracking. The minimum aging temperature will be 1075 °F.
- The material will not be used if irrediation or elevated temperature will cause the material to be unsuitable for service.
- Welding is not permitted after final heat treatment. 1
- Any forming or bending of parts will be done before the aging heat treatment.
- Maximum hardness limits will be specified on the basis of existing industry experience and relevant ASME Codes and ASTM Standards.

EPRI Evolutionary Plant SER 1.5-11

The staff concludes that the implementation of these requirements will ensure that precipitation-hardered stainless steel will perform in service as designed.

# Ni-Cr-Fe Alloy X-750

Section 5.3.1.7 of Chapter 1 specifies that Ni-CR-Fe Alloy X-750 will be used in the ALWR onl in those applications where a lower strength material is impractical and for which a substantial base of successful experience exists. Where Inconel Alloy X-750 is used, the application will be in accordance with EPRI NP-6202, "Material Specification for Alloy X-750 in LWR Internal Components."

The staff considers these requirements acceptable because improperly heattreated Alloy X-750 is susceptible to cracking and EPRI NP-6202 provides the technical basis for the use of this alloy.

# Prevention of Intergranular Stress Corrosion Cracking of Austenitic Stainless Steel

Section 5.3.1.8 of Chapter 1 specifies that austenitic stainless steel must be resistant to intergranular stress corrosion cracking (IGSCC). All austenitic stainless steel that is in contact with BWR reactor coolant at a temperature above 200 °F during power operatio or that is part of the BWR RCS pressure boundary or that is in borated water at any temperature in a PWR and that is welded without subsequent plution heat treatment will mcet the following requirements:

- For PWR service, only low-carbon wrought austenitic stainless steel, which includes Types 304L, 316L, 304NG, 316NG, and modified 347, will be used.
- For BWR service, only low-carbon wrought austenitic stainless steel, Types 304NG, 316NG, and modified 347 with a maximum carbon content of 0.020 percent, will be used.
- These materials will be tested for resistance to sensitization in accordance with Section 5.4.2.4 of Chapter 1 of the Evolutionary Requirements Document.
- Incoming components, parts, raw materials, and heat-treated parts will be examined for excessive intergranular attack (IGA) unless a minimum of 0.030 inch of metal is removed from all as-received surfaces during fabrication.

In Section 5.3.A.1 of the DSER for Chapter 1, the staff stated that EPRI should revise the material selection for reactor coolant pressure boundary piping to prevent IGSCC of austenitic stainless steels. The staff has recommended that licensees and applicants follow Revision 2 of NUREG-0313 "Technical Report on Material Selection and Process Guidelines for BWR Coolant Pressure Boundary Piping," to prevent IGSCC in stainless steel. EPRI has revised Section 5.3.1.8 of Chapter 1 of the Evolutionary Requirements Document

to reference this document. In addition, in its letter dated February 3, 1992, EPRI revised Table B.1-2 of Appendix B to Chapter 1 to require the use of Revision 2 of NUREG-0313. The staff concludes that these revisions are acceptable, and, therefore, this DSER open issue is closed.

In addition, it is important that adequate field and shop fabrication processes be used to minimize the sensitization of materials to IGSCC. Therefore, the staff will evaluate this matter during its review of an individual application for a combined license.

The staff concludes that, with the exception noted above, the implementation of the requirements in Section 5.3.1.8 of Chapter 1 will ensure that austenitic stainless steel will be resistant to IGSCC in service.

5.3.2 Materials in Feedwater, Steam, and Condensate Systems

Section 5.3.2 of Chapter 1 imposes the following requirements on materials in feedwater, steam, and condensate systems:

- The use of copper alloys is prohibited for PWR components that will be in contact with feedwater, steam, or condensate. Copper alloys may be used in BWRs in certain applications for service conditions of 200 °F or less; however, all such applications will be identified by the plant designer, who will document the basis for their acceptability on the basis of current experience.
- The ALWR design will incorporate into the secondary system component design the material considerations identified in EPRI NP-2294, "Guide to Design of Secondary Systems and Their Comprnents To Minimize Oxygen-Induced Corrosion.'
- Corrosion/prosion-resistant materials will be used for all components exposed to wet steam or flashing liquid flow where significant erosion could occur. The degree of corrosion/erosion resistance c. the material will be consistent with the temperature, moisture content, and velocity of the wet steam to which the component is exposed. Plain carbon steel with no deliberate alloying additions other than carbon and manganese will be used for this application.

The staff concludes that the implementation of these requirements will ensure that materials in feedwater, steam, and condensate systems will perform as designed.

## 5.3.3 Fasteners and Adhesives

Section 5.3.3 of Chapter 1 imposes the following requirements on metallic fasteners and adhesives:

The materials for threaded fasteners used to maintain pressure boundary integrity in the reactor coolant and related systems and in the steam, feedwater, and condensate systems and the threaded fasteners used inside those systems and in pipe and component structural mountings for those

systems will be selected and specified by the plant designer on the basis of their previous satisfactory performance in similar applications. Similarity will be based on

- comparable temperature and environment, including radiation dose and abnormal conditions such as wetting by gasket leakage
  - comparable stresses, including primary, secondary, and peak stresses, and comparable design details that may affect the stresses (e.g., thread form, head configuration, fits, and tolerances)
  - comparable service cycles, including magnitude and frequency
  - comparable fabrication and installation (e.g., heat treating, plating or other surface treatments, thread-forming, head-forming, cleaning, lubricating, and preloading) comparable inspection during fabrication and installation and while in service
- The application of threaded fasteners will be in accordance with the requirements of Section 12.7 of Chapter 1 of the Evolutionary Requirements Document and with the guidelines of EPRI NP-6316, "Guidelines for Threaded-Fastener Application in Nuclear Power Plants."
- The lubricants to be used on all threaded fasteners that will maintain pressure boundary integrity in the reactor coolant and related systems and in the steam, feedwater, and condensate systems and the threaded fasteners used inside those systems and in pipe and component structural support for those systems will be completely specified by the plant designer in appropriate drawings and specifications. That is, field selection of thread lubricants will not be permitted. The thread lubricants will be selected on the basis of experience and test data that show they are effective, but will not cause or accelerate corrosion of the fastener. If leak sealants are used on threaded fasteners or can be in contact with the fastener in service, their selection will be based on satisfactory experience or test data. The plant designer will consider possible adverse interaction between sealants and lubricants.
  - Acceptable non-metallic adhesives are
    - silicone compounds for continuous service below 400 °F
    - polyether urethanes for continuous service below 200 °F

The staff concludes that the implementation of these requirements will ensure that metallic fasteners and adhesives will perform in service as designed.

5.3.4 Thermal Insulation Materials

Section 5.3.4 of Chapter 1 imposes the following requirements on thermal insulation materials:

- Metallic insulation (Type 304) or blanket insulation with metallic jackets will be required for piping and components where inservice inspection or possible contamination make non-metallic materials unsuitable.
- For the use of nor-metallic insulation of austenitic stainless steel materials, the guidance of Regulatory Guide 1.36, "Nonmetallic Thermal Insulation for Austenitic Stainless Steel," should be followed.

The staff concludes that these requirement are acceptable because metallic insulation has been proven to perform satisfactorily in service. Non-metallic insulation will be used in accordance with the guidance in Regulatory Guide 1.36.

5.3.5 Seals, Gaskets, Packing, Sealants, Paints and Protective Coatings, Lubric ts and Hydraulic Fluids, and Cleaning, Packaging, and Storage Mater<sup>4</sup>, s

Section 5.3.5 of Chapter 1 requires that the plant designer conduct a program for evaluating the effects of intended products on other ALWR components under normal and postaccident conditions. For each product evaluated, the designer will provide recommendations and limits for its use in the steam supply systems and other applications in the ALWR. The designer will rely on information from component vendors only when substantiated by operational experience.

The staff concludes that the implementation of these requirements will ensure that seals, gaskets, packing, scalants, paints, and protective coatings, lubricants and hydraulic fluids, and cleaning, packaging, and storage materials are selected on the basis of operational experience. However, the staff will require that the FDA/DC applicant specify the specific ANSI standard covering this subject. The staff will evaluate these components, as appropriate, during its review of an individual application for FDA/DC.

# 5.3.6 Electrical Materials

Section 5.3.6 of Chapter 1 requires that the plant designer review materials used in the plant's electrical systems, particularly those used in safetyrelated applications, for functional reliability during normal, abnormal, plant operation, and accident conditions. The fire-retardant characteristics of the materials used in the electrical systems will be addressed to minimize the probability of fire and the consequences should a fire occur.

The staff considers this requirement acceptable. However, it is not clear if it is sufficient to address such issues as aging of cable insulation and other electrical materials over the design life and full range of environmental conditions. The staff will evaluate this issue during its review of an individual application for FDA/DC.

# 5.3.7 Weld Materials

Section 5.3.7 of Chapter 1 imposes the following requirements on welding materials:

 The strength and toughness of the ferritic steel velds will be equivalent to that of the base metal.

EPRI Evolutionary Plant SER

The welding consumables will meet the requirements of the ASME (SFA) or the American Welding Society specification as appropriate for code or non-code construction.

These requirements are in accordance with Section III of the ASME Codes and are, therefore, acceptable.

#### 5.4 Process Controls

# 5.4.1 Surface Condition

Section 5.4.1 of Chapter I of the Evolutionary Requirements Document requires that fabrication and installation processes adversely affecting the surface condition or microstructure (e.g., forming, bending, welding, heat treating, and surface rinding) be controlled to ensure the product meets the engineering requirements. Cleanliness standards during fabrication and subsequent handling and storage will be adopted on the basis of practice and standards that are current when the ALWR is fabricated.

The staff concludes that this requirement is acceptable because installation processes will be controlled through all s ges, thus ensuring trouble-free operation.

### 5.4.2 Fabrication Controls

Section 5.4.2 of Chapter 1 imposes requirements related to the fabrication and welding of materials for general ALWR applications. As a minimum, this control will be applied to the fabrication steps given below for the materials and processes involved.

#### Ferritic Steels

Preheat and post-weld heat treatment of ferritic steels will be controlled in accordance with NRC Regulatory Guide 1.50, "Control of Preheat Temperature for Welfing of Low-Alloy Steel." Applications in areas of limited accessibility will equalified in accordance with Regulatory Guide 1.71, "Welder Qualificati for Areas of Limited Accessibility." Heat inputs in cladding will be in acc. ance with Regulatory Guide 1.43, "Control of Stainless Steel Weld Cladoing of Low-Alloy Steel Components."

#### Austenitic Stainlesr Steels

Austenitic stainless steel will be fabricated in accordance with Regulatory Guide 1.44, "Control of the Use of Sensitized Stainless Steel." Interpass temperatures will be controlled to improve resistance to IGSCC. Stainless steels will be welded in accordance with Regulatory Guide 1.31, "Control of Ferrite Content in Stainless Steel Weld Material," and applications in areas of mited accessibility will be qualified in accordance with Regulatory Guide 1.71.

#### Martensitic Stainless Steels

Martensitic stainless steels will be heat treated followed by proper tempering to prevent stress corrosion cracking (SCC) and hydrogen embrittlement.

# Precipitation-Hardenable Stainless Steels

Precipitation-hardenable stainless steels will be heat treated with age hardening and tempering to prevent SCC and hydrogen embrittlement.

# Nickel-Based Alloys

Nickel-based alloys will be properly heat treated, and surface contamination will be controlled to prevent intergranular penetration.

# Alloys in General

Copper penetration into base metal and detrimental chromium carbide precipitation will be eliminated to prevent IGSCC. Surface peening also will be controlled in PWR applications to prevent cracking. Deposition control welding or heat welding will be used to modify residual stresses in accordance with the guidelines of NUREG-0313, Revision 2.

#### Conclusion

The staff concludes that these requirements are acceptable because they comply with the guidelines of NUREG-0313.

5.4.3 Examination and Tests

Section 5.4.3 of Chapter 1 requires that material and components be tested during and following fabrication for compliance with service requirements. As a minimum:

- Pressure-retaining material will be tested for mechanical properties and chemical composition to ensure conformance with the material specification.
- Pressure-retaining material, including weldments, will be examined by code-required nondestructive methods including ultrasonic, radiographic, and magnetic or liquid penetrant examination.
- Material used for tensile and impact test specimens will be heat treated in accordance with the appropriate code.
- For PWR applications of austenitic stainless steel materials, the ASTM A 708 Strauss Test or the ASTM A262 Practice E, Modified Strauss Test, will be used to demonstrate freedom from sensitization in fabricated, unstabilized stainless steel. For austenitic stainless steel materials in BWR applications, the ASTM A262, Modified Practice A, or the electrochemical potentiokinetic reactivation test will be used.
- For joints that are not examined volumetrically and for which access is limited to 14 inches or less in two directions, the welder qualification procedure will provide for testing of the welder under simulated access conditions.

The staff concludes that these requirements are acceptable because they comply with Section III of the ASME Code and Regulatory Guides 1.44, and 1.71.

### 5.4.4 Shipping and Storage

Section 5.4.4 of Chapter 1 requires that all materials and components be suitably protected from damage as a result of environmental conditions in accordance with the requirements of ANSI N45.2.2, "Packaging, Shipping, Receiving, Storage, and Handling of Items for Nuclear Power Plants."

The staff concludes that this requirement is acceptable because it complies with ANSI N45.2.2 as endorsed by the staff.

#### 5.4.5 Installation

Section 5.4.5 of Chapter 1 requires that all materials and components be handled during plant construction in accordance with the housekeeping requirements of ANSI N45.2.3, "Housekeeping During the Construction Phase of Nuclear Power Plants."

The staff concludes that this requirement is acceptable because it complies with ANSI N45.2.3 as endorsed by the staff.

### 5.4.6 Flush, Hydro, and Layup

Section 5.4.6 of Chapter 1 requires that the requirements and recommendations of ANSI N45.2.1, "Cleaning of Fluid Systems and Associated Components During Construction Phase of Nuclear Power Plants," apply for onsite cleaning of materials and components, cleanliness control, and preoperational cleaning and layup of water-cooled nuclear, wer plant fluid systems.

The staff concludes that this requirement is acceptable because it complies with ANSI/N45.2.1 as endorsed by the staff.

5.5 Environmental Conditions

5.5.1 Range of Environmental Conditions

Section 5.5.1 of Chapter 1 of the Evolutionary Requirements Document requires that materials selected for use in the ALWR be compatible with the full range of environmental conditions that may be encountered over the plant life. These environmental conditions include temperature, humidity, radiation, chemistry of fluids or materials in contact with the material, and other external conditions that may affect the suitability of a material. The plant designer will document the environmental conditions used as the basis for selecting ALWR materials. These environmental conditions will be consistent with the specific requirements in other chapters of the Evolutionary Requirements Document.

The staff concludes that these requirements are acceptable because the selected materials will be compatible with the full range of environmental conditions throughout the plant life. In addition, the use of radiation-damage-resistant materials in high-radiation areas will maximize their service life, reduce the frequency of replacement, and thereby, reduce personnel radiation exposure.

# 5.5.2 Water Chemistry Design Basis

#### BWR Water Chemistry Design Basis

Section 5.5.2.1 of Chapter 1 requires that the water chemistry design basis for BWR plant systems be in accordance with EPRI NP-4947-SR, "BWR Hydrogen Water Chemistry (HWC) Guidelines," 1987 Revision, and its subsequent revi-sions, and as supplemented by the guidelines in Table 1.2 in this chapter. The specific HWC control values in EPRI's guidelines relative to recirculating piping (e.g., 230 millivolts) will apply to nozzles, components, and other nonreplaceable components in the reactor vessel lower plenum.

EPRI address a low use of HWC for the advanced BWR design. However, the use of HWC at r such as Hatch, Brunswick, and Duane Arnold has resulted in unexpected, g, operational and post-shutdown radiation levels in reactor coolant system piping. EPRI has acknowledged the potential drawbacks of using HWC and has stated that investigations are under way to identify a solution to some of the problems resulting from the use of HWC. A special evaluation will be made when considering carbon and low-alloy material for reactor coolant service with less than 10 ppm oxygen as a result of HWC. The evaluation will include erosion/corrosion, radiation buildup, and pitting at shutdown. The staff will evaluate the issue of HWC use during its review of an individual application for FDA/DC.

The staff concludes that these requirements are acceptable and their implementation will ensure that the BWR water "hemistry is compatible with the selected materials.

# PWR Water Chemistry Design Basis

Section 5.5.2.4 of Chapter 1 requires that the water chemistry design basis for PWR plant systems be in accordance with EPRI NP-5960, "PWR Primary Water Chemistry Guidelines," Revision 1, and its subsequent revisions; EPRI NP-6239, "PWR Secondary Water Chemistry Guidelines," Revision 2, and its subsequent revisions; and as supplemented by the guidelines in Table 1.3 in this chapter.

The staff concludes that these requirements are acceptable and their implementation will ensure that the PWR water chemistry is comp ible with the selected materials. However, the applicant for FDA/DC . a combined license should reference EPRI NP-7077, Revision 2, instead of EPRI NP-5960, Revision 1. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

# 5.6 Conclusion

The staff has reviewed the general materials, materials fabrication, and water chemistry requirements in Section 5 of Chapter 1 of the Evolutionary Requirements Document. These requirements supplement the regulatory guidance in the applicable Standard Review Plan Sections 4.5.1 ("Control Rod Drive Structural Materials"), 4.5.2 ("Reactor Internal and Core Supnort Materials"), 5.2.3 ("Reactor Coolant Pressure Boundary Materials"), 5.3.1 ("Reactor Vessel Materials"), 5.3.3 ("Reactor Vessel Integrity"), 5.4.2.1 ("Steam Generator Materials"), 6.1.1 ("Engineered Safety Features Materials"), and 10.3.6

("Steam and Feewater System Materials") and, therefore, are acceptable. The implementation of these requirements will ensure that the affected nuclear power plant components will perform in service as designed.

....

Parameter	RWCS	CTS1	MWSE	ICST
Water Quality Chloride (parts per	4.0	4.0	4.0	20.0
Sulfate (ppb) Conductivity at 25 °C(µS/cm) Silica (ppb as SiO <sub>2</sub> ) pH at 25 °C (minimum) (maximum)	4.0 0.075 10.0	4.0 0.075	4.0 0.095 10.0	20.0 0.3 20.0
	¢		6.5 7.5	6.2 8.0
Corrosion product (ppb) Insoluble iron Total copper		20.0 2.0	:	
All other metals	-	8.0		
Total	3.0	30.0	10.0	20.0
Dissolved oxygen (ppb) (minimum) (maximum)	2	20.0	1	:

Table 1.2 BWR Water Chemistry Guidelines

<u>Notes</u>: Reactor Water Cleanup System (RWCS) effluent during shutdowns and fuel pool/suppression pool cleanup system effluent follow the same guidelines as those for the demineralized water storage tank specified in EPRI NP-4947-SR.

RWCS - reactor water cleanup system during power operation CTSI - condensate treatment systems influent MWSE - makeup water systems effluent ICST - influent to condensate storage tank

2

Water quality parameter	MWST	MWSGs	
pH (minimum)		7.0	
(maximum)		7.5	
Conductivity at 25 °C (µS/cm)	0.2	0.1	
Sodium (parts per billion (ppb))		<3.0	
Silica (ppb)		10	
Oxygen (parts per million (ppm)) maximum	0.100		
Chloride (ppm) maximum	0.15		
Fluoride (ppm) maximum	0.15		
Suspended solids (ppm) maximum	1.0		
Roric acid (ppm)		1	
Lithium (nom)			
Culfur se culfato (nnm)			
Sullar as sullare (bbm)			

# Table 1.3 PWR Water Chemistry Guidelines

\*Concentration of solids is determined by filtration through a filter with a pore size of 0.45 micron.

# Notes:

i.

.

MWST - makeup water storage tank MWSGs - makeup water to steam generators

#### 6 RELIABILITY AND AVAILABILITY

Section 6 of Chapter 1 of the Evolutionary Requirements Document gives reliability and availability requirements for safety- and non-safety-related structures, systems, and components (SSCs). Section 6.1 is an introduction to this section. Section 6.2 contains the power production availability requirements and the methodologies, specific strategies, and system features aimed at enhancing reliability for non-safety-related SSCs. Section 6.3 states that quantitative reliability and availability requirements and an analysis process analogous to or integrated with the non-safety-related requirements would be satisfactory for safety-related SSCs. The staff reviewed Section 6 of Chapter 1 of the Evolutionary Requirements Document through Revision 3.

#### Review Criteria

The NRC identified the need for a safety-oriented reliability effort for the nuclear industry in Section II.C.4 of NUREG-0660, "NRC Action Plan Developed as a Result of the TMI-2 [Three Mile Island Unit 2] Accident," dated August 1980. Initial NRC research in the area of reliability assurance began in the early 1980s. The results of this research showed that an operational reliability program, based on a feedback process of monitoring performance. identifying problems, taking corrective actions, and verifying effectiveness of the actions, was needed and that other NRC initiatives (e.g., maintenance inspection, performance indicators, aging programs, and technical specification improvement) would address his need. The overall conclusion of this research was that an operational eliability program could be implemented most effectively in performance-based, nonprescriptive regulation, where NRC mandates the level of safety performance to be achieved. For example, licensees could be required to set availability/reliability targets for selected systems and to measure performance compared to the targets. The TMI task was closed out in October 1988 without further action because several NRC initiatives effectively subsumed the operational reliability program effort. The NRC initiatives that formed the basis for closing out this TMI task included efforts to (1) improve maintenance and better manage the effects of aging, (2) improve technical specifications, (3) develop and use plant performance indicators, and (4) develop an operational reliability program as an acceptable means of meeting the station blackout rule (10 CFR 50.63).

In NUREG-1070, "NRC Policy on Future Reactor Designs," dated 1985, the staff recommended the use of a systems reliability program to ensure that the reliability of components and systems important to safety would remain at a sufficient level. To ensure that reliability objectives will be met and to prevent degradation of reliability during operation, it was envisioned that the probabilistic risk assessment performed at the design stage would be used as a tool in making detailed design decisions affecting procurement, testing, and the formulation of operations and maintenance procedures.

In a few specific instances, the NRC is studying or has established reliability targets for systems and components. For example, SRP Section 10.4.9, "Auxiliary Feedwater System," requires that an acceptable auxiliary feedwater (AFW) system design have an unreliability in the range of 1.0E-4 to 1.0E-5 per demand. Generic Issue B-56, "Diesel Reliability," involves efforts to determine, monitor, and maintain emergency diesel generator reliability levels. Additional regulatory basis for key elements of a reliability assurance program (RAP) can be found in 10 CFR Part 50, Appendix A, and 10 CFR 50.65.

In SECY-89-13, "Design Requirements Related to the Evolutionary Advanced Light Water Reactors," dated January 19, 1989, the staff identified several issues for ALWRs that may go beyond present acceptance criteria defined in the SRP. The RAP, as discussed in SECY-89-13, involved the need for a program to ensure that the design reliability of safetv-significant systems, structures, and components is maintained over the life of a plant. In SECY-89-13, the staff informed the Commission that a RAP would be required for ALWR FDA/DC. In November 1989, potential applicants for FDA/DC were informed by letter that "the NRC staff was considering matters that went beyond the current Standard Review Plan...that [the NRC] expects these advanced reactor designs to embody." Reliability assurance was identified as one of these matters.

The RAP can be seen as a program that consists of two distinct parts: the first part, referred to as the "D-RAP (design RAP)", is the responsibility of the designer and applies to vendor submittals for FDA/DC; the second part, referred to as the "O-RAP (operational RAP)", is the responsibility of and applies to an applicant for a combined construction and operating license (combined license). At the design stage, the D-RAP involves a top-level program that defines the scope, conceptual framework, and essential elements of an effective RAP. The D-RAP also implements those aspects of the program that are applicable to the design process. In addition, the D-RAP identifies the relevant aspects of plant operation, maintenance, and performance monitoring for the risk-significant SSCs for the owner/operator's con-sideration in developing the site-specific O-RAP.

The staff's position on the RAP is that a designer's submittal for FDA/DC pursuant to 10 CFR Part 52 would include, in part, the framework for a RAP and would also implement those elements of the RAP that would be applicable during the design phase. In turn, the designer would provide the framework of a RAP to a combined license applicant. This applicant would augment the designer's RAP to reflect plant-specific information and implement those elements applicable during the construction and operations phases. The staff's evaluation was based on guidance contained in the supporting documentation for TMI Task II.C.4, "Reliability Engineering," and SECY-89-13.

### Scope of Evaluation

Section 6 of Chapter 1 addresses EPRI-proposed requirements for reliability and availability and provides the supporting rationale for each requirement. In its evaluation, the staff considered the requirements as well as the accompanying rationale of the Evolutionary Requirements Document through Revision 3. In addition, in its letter dated January 9, 1992, EPRI proposed significant modifications to Section 6 of Chapter 1. References to sections of the Evolutionary Requirements Document in this evaluation refer to those in the document as superseded by those proposed by EPRI in the January 9, 1992, letter. In making a safety determination, the staff applied the same acceptance criteria to both the requirements and the rationale. If an EPRIproposed requirement and its accompanying rationale did not conflict with existing NRC requirements, the staff found that requirement and rationale acceptable. The Evolutionary Requirements Document primarily addresses D-RAP requirements; only a limited number of requirements in the Evolutionary Requirements Document apply to O-RAP. The O-RAP requirements will need to be provided by an individual applicant for a combined license.

The staff limited its evaluation to only those requirements that could affect plant safety. For example, the staff did not evaluate the merits of economic considerations such as EPRI's plant availability or outage duration goals that are specified in the Evolutionary Requirements Document. It did, however, evaluate the general relationship between safety and economic requirements.

The staff's evaluation of EPRI's requirements resulted in two types of findings: (1) requirements that are acceptable as written and (2) those that are acceptable, but an applicant will be required to provide additional information or guidance. For the requirements of the second type, the staff has provided clarification regarding what additional information a plant designer or an applicant for a combined license should provide when making a submittal that references the Evolutionary Requirements Document.

# 6.1 D-RAP Goals and Objectives

Section 6.1 of Chapter 1 of the Evolutionary Requirements Document contains the D-RAP goals and objectives. In its letter dated October 10, 1991, the staff stated that a RAP should contain and define the basic framework (scope, purpose, and objective). In its letter dated January 9, 1992, EPRI stated that the scope of a D-RAP was defined in Section 6 of Chapter 1 of the Evolutionary Requirements Document and the D-RAP goals and objectives were defined in Section 6.1. The staff has reviewed EPRI's response and concludes that the definitions of D-RAP scope, purpose, and objective contained in Sections 6 and 6.1 do not conflict with any NRC requirements and are acceptable. The starf also concludes that, in addition to meeting the EPRI-proposed requirements in this section, a plant designer should provide the organizational description and specify organizational accountability for implementing the D-RAP. As an example, Section 3.1.2 of Chapter 10 of the Evolutionary Requirements Document specifies requirements that should ensure that the manmachine interface systems (M-MIS) design is coordinated and implemented properly. An application for FDA/DC should contain an organizational description and accountability for the RAP in similar depth and detail as that in Section 3.1.2 of Chapter 10 of the Evolutionary Requirements Document. The staff will evaluate this description during its review of an individual application for FDA/DC.

# 6.2 Basic Program Elements of the D-RAP

Section 6.2 of Chapter 1 of the Evolutionary Requirements Document describes the basic program elements required to be included in the D-RAP. In its letter dated October 10, 1991, the staff stated that a RAP should (1) contain and define the program elements and describe how the elements would be applied to the plant structures, systems, and equipment and (2) contain reliability engineering techniques used during the design phase to ensure that the overall design reliability and availability goals are met. In its letter dated January 9, 1992, EPRI proposed a revision to Section 6.2 that describes the requirements for the basic program elements and reliability engineering techniques. The staff's evaluation of this proposed revision is given below.

### Analysis Requirements

Section 6.2.1 of Chapter 1 requires an analysis that shows the adequacy of the plant system designs and the recommended maintenance activities, spare parts, surveillance tests, and test intervais needed to support the SSC reliability and availability assumptions of the probabilistic risk assessment (PRA). This analysis will be consistent with the PRA. Further, the reliability and availability analyses will be carried out as an integral part of the design process to influence the design options and allow appropriate cost/benefit tradeoffs during the design of the standard ALWR plant. The Evolutionary Requirements Document requires that this analysis be performed by the plant designer sufficiently ahead of procurement and construction to minimize the impact of potential design changes and ensure that SSC reliability assumptions are met. The staff concludes that these requirements do not conflict with existing NRC requirements and are acceptable. It also concludes that the EPRI requirements in this section should also apply, in addition to PRA methodologies, to deterministic and other methodologies used for making decisions about the adequacy of plant system designs. Furthermore, the staff encourages a vendor or combined license applicant to include references to the analytical methods or models that were used in performing the analyses required by this section in the top-level RAP program document. As an example, Section 3.5.4 of Chapter 10 of the Evolutionary Requirements Document describes analytical methods for use for the M-MIS design. The staff concludes that these or similar methods should be used for all reliability/maintainability analyses performed by the plant designer or combined license applicant. The staff will evaluate this issue during its review of an individual application for FDA/DC or a combined license.

# Application of Risk Significance

Section 6.2.2 of Chapter 1 requires the plant designer to ensure that the dominant modes of failure identified by the PRA are appropriately addressed in the plant design consistent with their risk significance. This section further states that the PRA methodology used by the plant designer will provide for importance-weighting of SSCs according to their risk significance and for identifying the dominant failure modes of these SSCs. The staff concludes that these requirements do not conflict with existing NRC requirements and are acceptable. It further concludes that, in addition to PRA methodology, deterministic or other methods may be used in establishing dominant failure modes and risk significance for the RAP. As discussed above, the staff will evaluate this issue during its review of an individual application for FDA/DC or a combined license.

### Nuclear Power Plant Reliability Data System Engineering Data Base

Section 6.2.3 of Chapter 1 requires the plant designer to supply the initial input information necessary to establish the nuclear power plant reliability data system (NPRDS) engineering data base for the plant. The staff concludes that the EPRI-proposed requirements do not conflict with existing NRC requirements and are therefore acceptable. However, this section limits the reliability data base to NPRDS. The staff does not want to preclude the use of other data bases that may have merit or that can be shown to be equivalent to current accepted practices. The staff would expect that, in addition to NPRDS data, other reliability data bases could also be considered by plant designers or combined license applicants if they provide the necessary reliability

EPRI Evolutionary Plant SER

information. An example may be the data base that results from meeting the requirements of Section 6.2.1.1 of Chapter 1. The staff will evaluate the data bases considered during its review of an individual application for FDA/DC or a combined license.

# Reliability Activity Task Descriptions

Section 6.2.4 of Chapter 1 requires the plant designer to provide the owner/operator with descriptions of recommended reliability activities. These activities will include those tasks necessary to maintain SSC environmental qualification, prevent predictable failures, or maintain functional reliability. The plant designer will also recommend appropriate condition monitoring parameters to be periodically evaluated and their acceptable limits to provide added assurance of operability. Condition monitoring recommendations will include provisions for detecting age-related degradation where appropriate. The staff concludes the requirements of this section are acceptable; however, additional guidance that references other areas of the Evolutionary Requirements Document would be beneficial to users of this document. As an example, reliability, maintainability, and testability analyses are described in Chapter 10, Sections 3.5.4, 3.6.1, and 3.6.2, of the Evolutionary Requirements Document. A plant designer should provide similar descriptions to a combined license applicant for ail reliability tasks.

In Section 6.2.8.4 of the DSER for Chapter 1, the staff expressed the concern that, as nuclear plants age, the accumulative effect of neutron radiation on the reactor pressure vessel beltline materials will lead to material embrittlement such that materials exhibit loss of fracture toughness and strength. The accumulative high temperature and thermal cycling of the primary coolant systems generate thermal fatigue to materials that causes loss of strength and fracture toughness.

As a result, EPRI added the following statement to Section 6.2.4.5 of Chapter 1 that lists known failure mechanisms: "Loss of strength and/or fracture resistance because of change(s) in metallurgical state of alloys resulting from exposure to high temperature, thermal cycling and/or high radiation." The staff concludes that this modification addresses the staff's concern and, therefore, this DSER confirmatory issue is closed.

### Control of PRA Design Assumptions

Section 6.2.5 of Chapter 1 requires the plant designer to provide a program for verifying that PRA assumptions remain within limits required to maintain reliability goals during the design and construction process. This program will apply to all participants in the design and construction process whose activities could affect the plant designer's PRA assumptions. The Evolutionary Requirements Document further states that the results of the verification program will be included as part of the final as-built design documentation. The staff concludes the requirements of this section do not conflict with existing NRC requirements and are acceptable.

#### 6.3 D-RAP Performance Standards

Section 6.3 of Chapter 1 contains the quantitative reliability and availability performance standards for use in the design phase. In its letter dated October 10, 1991, the staff stated that a RAP should contain overall reliability and availability design goals based on safety requirements that have core damage frequencies or probabilities associated with them. In its letter dated January 9, 1992, EPRI proposed to modify Section 6.3 of Chapter 1 to specify D-RAP performance standards. The staf?'s evaluation of this proposed revision is given below.

# Core Damage Frequency

Section 6.3.1 of Chapter 1 states that the plant designer will evaluate the mean annual core damage frequency of the design using PRA and will confirm that this frequency is less than or equal to 10E-5 event per reactor-year, including both internal and external events (excluding sabotage). In Section 2.3.3 of this chapter, the staff discusses EPRI's public safety goal and concludes that the use of this goal is acceptable. However, it will not use EPRI's goal as an acceptance criterion.

The staff concludes that the contributions of the SSCs to the core damage frequency should be apportioned for this, or any other, performance standard. Therefore, the applicant's submittal for FDA/DC should include apportionment of these contributions to the core damage frequency. The staff will evaluate the adequacy of the apportionment of the SSCs' contribution to core damage frequency during its review of an individual application for FDA/DC.

#### Inadvertent Depressurization

Section 6.3.2 of Chapter 1 states that non-safety-related active reactor coolant system (RCS) makeup capability and any other necessary measures will be provided so that the chance of inadvertent RCS depressurization can be demonstrated by reliability analysis to be less than 10 percent over the entire 60-year life of the plant. Further, recovery from inadvertent RCS depressurization will be rapid enough that the lifetime average design availability requirements can still be met assuming one inadvertent RCS depressurization during the 60-year plant life. The staff notes that referring to the active RCS makeup capability as non-safety related is inconsistent with risk-significant concepts in Section 6.2.2 of Chapter 1 of the Evolutionary Requirements Document. The staff concludes that these requirements do not conflict with existing NRC requirements and are, therefore, acceptable. However, in addition to these requirements, an FDA/DC applicant should explicitly state the priority of safety in a cident recovery. The staff will evaluate this matter during its review of an individual application for FDA/DC.

### Production Availability Requirement

Section 6.3.5 of Chapter 1 states the plant will be designed for an annual average production availability of more than 87 percent over its life. The staff's review of this requirement was limited to areas where economic considerations could potentially compromise plant safety. The staff concludes that this requirement does not conflict with existing NRC requirements and is therefore, acceptable. However, an FDA/DC or combined license applicant

EPRI Evolutionary Plant SER

should explicitly discuss the relationship between safety and production availability in a D-RAP or O-RAP when addressing this design requirement. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

# Refueling Duration Capability

Section 6.3.7 of Chapter 1 states that the plant will be designed so that the total duration of a no-problem refueling outage will be 17 days or less (breaker to breaker) assuming 24-hour productive days. The staff concludes that this requirement does not conflict with existing NRC requirements and is acceptable. However, a D-RAP or O-RAP submittal from an FDA/DC or combined license applicant should explicitly state that plant safety will not be compromised in attempting to satisfy this EPRI design requirement. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

# Planned Cutages

Section 6.3.8 of Chapter 1 states that the plant will be designed so that refueling and regular maintenance will be completed in an average of less than 25 days per year. An average of 25 days per year for refueling and plant maintenance corresponds to 50 days in a 24-month fuel cycle. The staff concludes that this requirement does not conflict with existing NRC requirements and is acceptable. However, a D-RAP or O-RAP submittal from an FDA/DC or combined license applicant should explicitly state that plant safety will not be compromised in attempting to satisfy this EPRI design requirement. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

### Major Outages

Section 6.3.9 of Chapter 1 states that the plant will be designed so that the frequency and duration of major outages will not exceed 180 days per 10 years. The staff concludes that this requirement does not conflict with existing NRC requirements and is, therefore, acceptable. However, a D-RAP or O-RAP submittal from an FDA/DC or combined license applicant should explicitly state that plant safety will not be compromised in attempting to satisfy this EPRI design requirement. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

#### 6.4 System Design

Section 6.4 of Chapter 1 of the Evolutionary Requirements Document describes the qualitative processes to be used in system design. In its letter dated October 10, 1991, the staff stated that a RAP should establish a set of reliability and availability goals at the system level to ensure that the reliability and availability goals for the overall design are met. In its letter dated January 9, 1992, EPRI proposed to modify Section 6.4 of Chapter 1 to specify general system design goals for a RAP.

### Shared Systems

Section 6.4.1 of Chapter 1 requires that, for multiple-unit plants on a single site, the number of shared systems will be limited to auxiliary support systems such as sewer, auxiliary steam, or site security. For any systems that are shared, the plant designer or combined license applicant will analyze the effect of any failure or any testing in that system that will affect the maintenance, ALARA (as low as is reasonably achievable) goals, availability, safety, or operability of other systems and the availability of each unit. The staff concludes that these requirements do not conflict with existing NRC requirements and are, therefore, acceptable.

#### Startup and Testing

Section 6.4.2 of Chapter 1 requires that the plant designer or combined license applicant review and optimize the startup testing program for initial startup and for startup following refueling/maintenance outages to support completion of required testing with a minimum impact on the availability of the plant. As a basis for this review and optimization, the plant designer will review existing LWR startup test programs and any available reports on optimization of these test programs. The staff concludes that these requirements do not conflict with existing NRC requirements and are, therefore, acceptable.

#### Failure Mechanisms

Section 6.4.3 of Chapter 1 states that the plant designer will ensure that the systems, equipment, and plant design will account for failure mechanisms shown to have a significant effect on downtime. This section further states the ALWR will be designed so that knowr failure mechanisms will not prevent the ALWR from achieving its design lif. r meeting the availability and evert frequencies as described in Section 6.3 of Chapter 1. The staff concludes that these requirements do not conflict with existing NRC requirements and are, therefore, acceptable.

# Specific Syst ) Design Features

Section 6.4.4 of Chapter 1 states that the plant designer will incorporate design features to support high reliability of safety-significant SSCs and high plant availability. The staff concludes that this requirement does not conflict with existing NRC requirements and is, therefore, acceptable.

#### Minimum Number of Components

Section 6.4.5 of Chapter 1 states that the plant designer will reduce the number of active components required to meet the intended function of operability and maintainability. The staff concludes that this requirement does not conflict with existing NRC requirements and is, therefore, acceptable.

#### 6.5 Conclusion

The staff's overall conclusion is that the requirements in Section 6 of Chapter 1 of the Evolutionary Requirements Document are directed at the plant designer and the D-RAP and very few requirements are directed toward the

procurement, construction, and operational phases, which are the responsibility of the combined license applicant. Additional guidance and requirements will be necessary to support the development of a combined license applicant's O-RAP.

The staff concludes that the reliability and availability requirements in the Evolutionary Requirements Document are consistent with accepted industry practices and principles and do not conflict with existing regulatory require-ments and guidelines. If the information identified in the Evolutionary Requirements Document and the additional information identified in this report are provided by a plant designer, the D-RAP submittal should be acceptable.

Ξ,

# 7 CONSTRUCTION AND CONSTRUCTIBILITY

Section 7 of Chapter 1 of the Evolutionary Requirements Document specifies minimum requirements and goals proposed by EPRI with regard to the construction of an ALWR. EPRI states that these requirements support the overall objectives described in Volume I of the Requirements Document, particularly by reducing the utility's risk and plant capital costs through an improved construction schedule, greater productivity, and better construction techniques This section discusses construction schedule goals; integration of design, construction, and startup schedules; use of computer technology for design, construction, and startup activities; and integration of system completion, testing, and owner/operator acceptance.

Many of EPRI's requirements concerning construction and constructibility of the plant are beyond the regulatory purview of the NRC. The staff evaluated these requirements only from the aspect of how they may adversely affect the safe design, construction, and operation of the plant. However, it did identify items that are part of its regulatory responsibility.

### Inspection of Construction Activities

The NRC has the statutory responsibility, regardless of construction schedule, to verify that the plant is constructed in accordance with the design documents tendered with the application for an operating license. The owner/ builder must ensure that construction activities permit verification of the acceptability of the plant configuration in accordance with the requisite NRC Inspection Manual chapters. The staff will evaluate this matter during its review of an individual application for a combined license.

In Section 7.2.C.2 of the DSER for Chapter 1, the staff indicated that EPRI had committed to modify its data base program to include milestones for scheduling construction verification inspection points for inspection and enforcement personnel and startup tests. EPRI included this information in Section 7.2.8.1.2 of Chapter 1 of Revision 1 of the Evolutionary Requirements Document. Therefore, this DSER confirmatory issue is closed.

#### Schedular and Administrative Requirements

The schedular and administrative requirements proposed by EPRI in this section are outside the staff's regulatory purview. The staff notes that these measures, although outside the scope of the NRC's review, should assist the owner/builder's configuration control to facilitate preservation of the licensing basis.

#### Quality Assurance

In the DSER for Section 7 of Chapter 1, the staff identified an open issue that construction activities for which the owner/builder must provide a site organization should include quality assurance and quality control. This issue resulted from the staff's review of Revision 0 of the Evolutionary Requirements Document. In response to the staff's concern, EPRI revised Section 7.2.8.2 of Chapter 1 to make it clear that the site organization plan must include provisions for quality control and quality assurance. The staff concludes that this revision resolves the staff's original concern and, therefore, this DSER open issue is closed.

However, EPRI added additional information that was not clear in regard to its intent regarding quality assurance. For example, Section 7.2.8.2 of Chapter 1 could be interpreted as not requiring a commitment to apply pertinent quality assurance program requirements to facilities and systems that have some importance to safety ever though they ar not safety related (for example, see the new maintenance rule, 10 CFR 50.65(',2)). The staff concludes that pertinent quality assurance provisions should be applied to these activities and items. Applicants for FDA/DC or a combined license will have to acceptably describe a quality assurance program for these activities and items. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

### Physical Security

In Section 7.2.8.2 of Chapter 1, EPRI requires a plant security plan covering the construction phase. Although this is a utility requirement outside the scope of 10 CFR Part 73, the requirement is compatible with NRC requirements.

In Section 7.9.6 of Chapter 1, EPRI requires the utility to establish security boundaries as part of the startup testing program. In a letter dated May 13, 1991, EPRI agreed that the detailed construction and startup schedule will have to address NRC review and approval of the installed security system for the operating phase before the first fuel loading, but that this milestone was beyond the scope of the Evolutionary Requirements Document. The staff, therefore, will address this matter during its review of an individual application for a combined license.

The staff expects that at least 60 days before loading fuel, a combined license licensee will have confirmed that security systems and programs described in its physical security plan, safeguards contingency plan, and guard qualification and training plan have achieved operational status and are available for NRC inspection. Operational status means that the security systems and programs are functioning in their entirety as they would when the reactor is operating and will continue to do so. The licensee's determination that operational status has been achieved must be based on tests conducted under realistic operating conditions of sufficient duration that demonstrate that the equipment is properly operating and capable of long-term, reliable operation; that procedures have been developed, approved, and implemented; and that personnel responsible for security operations and maintenance have been appropriately trained and have demonstrated their capability to perform their assigned duties and responsibilities.

#### Reliability of Modular Construction

In Section 7.7 of Chapter 1, EPRI proposes the use of modular construction techniques for ALWRs. These techniques would allow construction and testing of portions of the plant in onsite fabrication shops or off site. These

modules would then be assembled in the laydown area of the plant. This technology offers the potential for much shorter construction schedules and reduced costs.

Specific licensing criteria addressing modular construction have not been developed for nuclear power plant construction. Structures, systems, and components that are assembled using modular construction techniques must possess, as a minimum, the same degree of structural strength and reliability as such items provided in currently licensed plants that were constructed using current onsite construction techniques. Items to be considered include segmented rebar cage connections and in-containment steel/concrete sandwichtype shear walls for which there are no modular construction design criteria and for which test information is limited. Other areas of concern include the integrity of joints (including strength and ductility), seismic damping values and stiffness degradation in structural modules, quality assurance and quality control requirements for transportation and installation of modules, and the scope of the verification testing after the modules are installed. The staff will address this matter during its review of an individual application for a combined license, should the applicant propose use of these techniques.

# 8 OPERABILITY AND MAINTAINABILITY

# 8.1 Introduction

In Section 8 of Chapter 1 of the Evolutionary Requirements Document, EPRI proposes requirements that are meant to enhance the operability and maintainability of the ALWR by incorporating experience gained at operating facilities. These requirements are intended to minimize the need for maintenance and surveillance, thereby minimizing the dose of radioactivity to plant personnel. Included are general design criteria for the control room and other control locations and criteria for designing the instrument and control systems and factoring in man-machine interface considerations.

# 8.2 Provisions To Enhance Operability and Maintainability

In Section 8.2 of Chapter 1 of the Evolutionary Requirements Document, EPRI proposes requirements regarding resolution of known operational and maintenance procedures and related training, and standardization of components and equipment. This section also gives human factors requirements that are related to the operation of the plant, including consideration of instrumentation and controls, control room design, and environmental conditions. In addition, EPRI provides requirements to address human factors that are related to maintenance of the plant, including support systems, accessways, and orientation of equipment. EPRI specifies requirements to integrate operations and maintenance requirements and addresses preventive maintenance and inspection concerns. Section 8.2 specifies requirements for the qualifications, organizations, and training of operations and maintenance personnel.

# Preventive Maintenance and Inspections

In Section 8.2.C.2 of the DSER for Chapter 1, the staff recommended that specific codes and standards relating to inservice inspection and test requirements be included in the Evolutionary Requirements Document. In response, in Section 8.2.6.3 of Chapter 1, EPRI requires that the preventive maintenance programs include the integration of equipment inspection and testing requirements imposed by the codes and standards that apply at the time of design. Also, EPRI specifies 10 CFR 50.55a in Table B.1-1 of Appendix B to Chapter 1, which requires using Section XI of the ASME Code and addenda for the inservice inspection requirements. The staff concludes that these requirements for the preventive maintenance program follow the ASME Code and are acceptable. Therefore, this DSER confirmatory issue is closed.

### Radiation Exposure

Section 8.3.2 of Chapter 1, to reduce worker radiation exposure, EPRI states that temperature and humidity in radiation areas will be controlled by the heating and ventilation system. All plant areas will be adequately illuminated to minimize the time and exposure associated with the installation of temporary lighting in work areas. To reduce worker doses, equipment will be oriented to facilitate maint nance operations. These provisions are intended to maintain worker doses as low as is reasonably achievable (ALARA) and, therefore, are acceptable.

# Personnel and Staffing

In the DSER for Section 8 of Chapter 1, the staff stated that EPRI had indicated that it would add a statement to the Evolutionary Requirements Document regarding the qualification of plant operating personnel. In its letter dated May 5, 1991, EPRI stated that qualifications and training requirements for the plant operating personnel were the responsibility of the plant owner and were outside the scope of the Evolutionary Requirements Document. EPRI added Section 8.2.6.5, which indicates that these personnel will meet the requirements of ANSI/ANS 3.1, "Selection and Training for Nuclear Power Plants." ANSI/ANS 3.1 excludes the selection and training of site security personnel, the staff concludes that there are no conflicts with the security training requirements of 10 CFR 73.55(b). The staff concludes that this is in accordance with Regulatory Guide 1.28, "Quality Assurance Program Requirements," and is acceptable. Therefore, this DSER confirmatory issue is closed.

#### Acoustical Monitoring

In Section 8.2.8 of the DSER for Chapter 1, the staff recommended that appropriate noise level requirements concerning acoustical monitoring requirements be added to Chapter 1 because noise is a major contributor to operator fatigue and ineffectiveness. In Revision 1 of Section 8.2.4.4.3 of Chapter 1, EPRI added noise level requirements that ensure that the designer of the plant will consider both reduction and attenuation of noise sources to reduce noise exposure of operators to levels specified by the Occupational Safety and Health Administration (OSHA). The staff did not find the revision acceptable because OSHA standards reflect maximum permissible exposures and not necessarily the sound levels required for operations and maintenance personnel to perform their jobs. By letter dated March 19, 1992, EPRI revised the statement to read:

The design of the plant shall consider both reduction and attenuation of noise sources to reduce exposure to operation and maintenance personnel. Personnel shall be provided with acoustical environments which will not cause personal injury, interfere with voice or other communications, cause fatigue, or degrade overall system effectiveness.

The staff concludes that the statement is acceptable. Therefore, this DSER confirmatory issue is closed.

#### Human Factors Considerations

In Section 8.2.8.4 of the DSER for Chapter 1, the staff recommended that IEEE P1023-1988, "Guide for the Application of Human Factors Engineering to Systems, Equipment, and Facilities of Nuclear Power Generating Stations," and EPRI-2360, "Human Factors Methods for Assessing and Enhancing Power Plant Maintainability," be referenced in this section. Although EPRI has not referenced these documents in the Evolutionary Requirements Document, the staff will evaluate an individual application for FDL/DC or a combined license to ensure that these documents have been considered. This DSER open issue is closed.

# 60-Year Life

Section 8.2.3.3 of Chapter 1 states that components will be designed for operating period of 60 years (minimum) or in accordance with Sect Chapter 2 of the Evolutionary Requirement: Document. As discuss tion 3.3 of this chapter, the staff will review the ALWR design life notwithstanding the fact that a 40-year limit is specified Energy Act and NRC's regulations.

#### 8.3 Minimizing Dose Levels to Personnel

Section 8.3 of Chapter 1 of the Evolutionary Requirements Document specifies the general requirements and design features that will be used to minimize radiation levels and to maintain personnel doses ALARA. Specific design criteria are addressed in other chapters of the Evolutionary Requirements Document, primarily Chapter 6.

To achieve its dose goal of 100 person-rem per year for the ALWR, EPRI established several basic requirements that must be met. These include the control of materials selection (including the elimination of cobalt to the extent possible); the use of adequate temporary and permanent shielding; the design of components to permit cleaning and chemical decontamination; the application of robotics for cleanup, maintenance, and inspection tasks; the packaging of equipment in modules for rapid disassembly for inspection and maintenance; and the design of heating and ventilation systems to control temperature and humidity in radiation work areas.

In Section 8.3.3 of Chapter 1, EPRI states that the plant designer will consider the use of electropolished surfaces for those areas of the plant (e.g., large-diameter reactor coolant system piping, steam generator channel heads) where this treatment will significantly reduce the dose to personnel during maintenance.

Section 8.3.4 of Chapter 1 states that the plant designer will perform an analysis to determine the effectiveness of using robotic applications in the ALWR. Inspection and surveillance functions will include reading of instruments and gauges, performing radiation surveys and measuring radiation levels, and taking smear surveys. Maintenance functions will include steam generator inspection and maintenance, control rod drive removal, radwaste drum handling, spent fuel consolidation, equipment decontamination, and routine surveillance and maintenance tasks. The Evolutionary Requirements Document specifies that the ALWR will include design features such as wider doors and aisles, ramps, and modular construction of equipment and systems (for ease of equipment removal and replacement) to facilitate the use of robotic devices. Table 1.8-4 includes "verify security locks" as one of several functions to be evaluated by the plant designer as a candidate for robotic inspection and surveillance. However, in its letter of May 17, 1991, EPRI stated that details of the security functions to be performed and replacement of a security officer were outside the scope of the Evolutionary Requirements Document. The staff, therefore, will address this matter during its review of an individual application for a combined license.

The staff concludes that the design features in this section are intended to minimize dose levels to plant personnel, are in compliance with Regulatory Guide 8.8, and are, therefore, acceptable.

EPRI Evolutionary Plant SER

# 8.4 Facility Requirements

Section 8.4 of Chapter 1 of the Evolutionary Requirements Document specifies that the plant will be designed to provide adequate support facilities for personnel and equipment. This section addresses general requirements for controlling personnel access to the plant and, particularly, to radioactive work areas. It provides requirements for plant services, contaminated and clean work shops, and spare parts control and includes requirements for the design of the plant's personnel access portal.

In its letter dated May 13, 1991, EPRI committed to change some of the security terms in Sections 8.4.1.1 and 8.4.1.2 of Chapter 1 to be consistent with the terminology in Section 5 of Chapter 9 of the Evolutionary Requirements Document. The staff has verified that these changes have been acceptably included in EPRI's June 11, 1991, markup.

EPRI states that the ALWR will be designed to allow controlled access to the plant for the number of personnel needed to perform the required activities. Changing rooms will be located away from radiation sources and will have lockers for at least 1,000 people. Radioactive work areas will be separated from clean work areas. To the extent possible, the ALWR design will incorporate ramps or steps instead of ladders for personnel movement between floors in areas where personnel are required to wear anticontamination clothing. Such measures are intended to facilitate the processing and movement of large numbers of personnel through the plant during major maintenance outages. The staff concludes that these features do not conflict with the Commission's regulations and guidance and are, therefore, acceptable.

### 8.5 Provisions for Replacement of Major Components

Section 8.5 of Chapter 1 of the Evolutionary Requirements Document specifies requirements to facilitate the replacement of all major components other than the reactor vessel and basic plant structures. The plant design will also include plans for the transportation and storage of major plant components that may be contaminated and contaminated special tools and equipment removed from the buildings. Designing the plant so that major components can be easily removed and transported out of the buildings without major structural modifications will shorten plant outage time and will, therefore, result in lower overall personnel exposures. The staff concludes that these features do not conflict with the Commission's regulations and guidance, and are, therefore, acceptable.

#### 8.6 Inspection and Testing

In Section 8.2.B of the DSER for Chapter 1, the staff requested that a qualifying statement be added to Chapter 1 specifying that the plant designer will select pumps and valves that can meet the requirements of Sections IWP and IWV of ASME Code, Section XI, which contain rules for inservice testing of pumps and valves. In response to this request, Revision 1 of Section 8.6.1 of Chapter 1 was added to provide this commitment. Therefore, this DSER confirmatory issue is closed. However, a more detailed discussion by the staff of inservice testing is contained in Section 12 of this chapter.

# Operation Problem Areas

In the section entitled "LWR Operation Problem Areas To Be Addressed in ALWR Design" of the DSER for Chapter 1, the staff requested that EPRI add the following item to Table 8-2 of the Evolutionary Requirements Document: "Insufficient structural integrity and mechanical reliability of pump components." This item was added to the renumbered Table 1.8-2 in Chapter 1. Therefore, this DSER confirmatory issue is closed.

#### 8.7 Hazardous and Toxic Chemicals

Section 8.7 of Chapter 1 of the Evolutionary Requirements Document specifies that the use of hazardous and toxic chemicals in the plant will be minimized to the extent practicable. This section excludes the use of hazardous chemicals in radiation-controlled areas, unless there is no practical alternative to their use. Requirements for the use of such substances are also provided. The staff concludes that these requirements do not conflict with the Commission's regulations and guidance and are, therefore, acceptable.

# 8.8 Conclusion

The staff concludes that the requirements of Section 8 of Chapter 1 of the Evolutionary Requirements Document do not conflict with the Commission's regulations and are, therefore, acceptable. However, certain details regarding human factors engineering and physical security were insufficient to enable a final determination regarding compliance with regulatory guidance. Therefore, the staff will review an individual application for FDA/DC to ensure that this guidance has been met.

### 9 QUALITY ASSURANCE

Section 9 of Chapter 1 of the Evolutionary Requirements Document identifies the major elements of an overall quality assurance (QA) program for an evolutionary ALWR plant. EPRI specifies the requirements for the supporting QA programs for the primary organizational entities involved in and supporting design, procurement, construction, and preoperational testing of an evolutionary ALWR plant. EPRI has also affirmed that this section is intended to be in accordance with all current regulatory requirements for QA at the level of detail provided.

In the DSER for Section 9 of Chapter 1, the staff stated that EPRI's responses to earlier staff comments had not been incorporated into the Evolutionary Requirements Document. EPRI has incorporated the changes in Sections 9.2.3 and 9.2.4 of Revision 1 of Chapter 1. The staff concludes that these sections are in accordance with Regulatory Guide 1.28, "Quality Assurance Program Requirements (Design and Construction)," and are, therefore, acceptable. Therefore, this DSER confirmatory issue is closed.

In the DSER for Section 9 of Chapter 1, the staff stated that two line items needed to be added to Table 9-1. EPRI has incorporated the requested changes in Table 1.9-1 of Revision 2 of Chapter 1. The staff concludes that this table provides an acceptable list of typical quality and quality assurance problems experienced during the design and construction of currently operating nuclear plants. Therefore, this DSER confirmatory issue is closed.

In the DSER for Section 10 of Chapter 1, the staff stated that the crossreferences in Table B.1-1 were incorrect. As discussed by the staff in Section 10 of this chapter, this table has been corrected. In the area of quality assurance, Table B.1-1 lists 10 CFR 50.34(f)(3)(ii), 50.34(f)(3)(iii), 50.55(e), 50.55(f), and 50.55a; General Design Criterion (GDC) 1 of Appendix A to 10 CFR Part 50; and Appendix B to 10 CFR Part 50. In each case, Table B.1-1 indicates that the ALWR will be designed to comply with these requirements and that the lead chapter for this subject is Chapter 1 of the Evolutionary Requirements Document. Table B.1-2 also lists Regulatory Guides 1.26 (Rev. 3), 1.28 (Rev. 3), 1.29 (Rev. 3), 1.30 (Rev. 0), 1.37 (Rev. 0), 1.38 (Rev. 2), 1.54 (Rev. 0), 1.94 (Rev. 1), and 1.116 (Rev. 0-R) and SRP Section 17.1 (Rev. 2). Revision 4 of Table B.1-2 indicates that the ALWR will be designed to comply with this guidance and that the lead chapter for this subject is Chapter 1 of the Evolutionary Requirements Document. The staff concludes that these references are acceptable. Therefore, this DSER confirmatory issue is closed.

In Section 6.1.2 of the DSER for Chapter 10, the staff raised a concern regarding the QA program for software. In its letter dated January 28, 1992, EPRI provided its response to this concern. However, EPRI limits its software QA program to safety-related software. Sillarly, as discussed in Section 7 of Chapter 1 of the Evolutionary Requirements Document, EPRI appears to limit its QA to safety-related items. The staff is concerned that the Evolutionary Requirements Document could be interpreted as not requiring a commitment to apply pertinent QA program requirements to software, facilities, structures, systems, and components that have some safety importance or have some importance to safety even though they are not safety related. The staff concludes that pertinent QA provisions should be applied to these activities and items

EPRI Evolutionary Plant SER

in accordance with the requirements of GDC 1 of Appendix A to 10 CFR Part 50. Applicants for FDA/DC or a combined license will have to acceptably describe a QA program for these activities and items. The staff will evaluate this matter during its review of an individual application for FDA/DC or a combined license.

0

The staff concludes that, with the exceptions noted above, the requirements of Section 9 of Chapter 1 of the Evolutionary Requirements Document do not conflict with the Commission's regulations and are, therefore, acceptable.

6

# 10 LICENSING

Section 10.1 of Chapter 1 of the Evolutionary Requirements Document specifies EPRI-proposed licensing r quirements for future ... WRs. EPRI states that Section 1 of Appendix B to Chapter 1 contains a list of NRC regulations and guidelines currently applicable to LWR design and identifies its position with respect to each. EPRI states that Section 1 of Appendix B to Chapter 1 will represent the regulatory requirements it believes are applicable to the ALWR design at the level of detail consistent with that of the Evolutionary Requirements Document when that document is completed.

Section 10.2 of Chapter 1 states that the ALWR will be designed to comply with the NRC regulatory requirements and guidance in effect on January 1, 1990, consistent with the commitments in Section 1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document. EPRI states that these requirements and guidance include applicable Commission regulations specified in 10 CFR, general design criteria, NRC policy statements, regulatory guides, the Standard Review Plan (NUREG-0800), and other documentation that resolves unresolved and generic safety issues. Although the staff understands EPRI's need to "freeze" the requirements it addresses to those in effect on January 1, 1990, the staff expects that the design certification applications will be in compliance with the Commission's regulations and guidance that are applicable and in effect at the time the certification is issued. The staff will evaluate this compliance during its review of an individual application for FDA/DC.

In addition, issue resolutions that are different from those arrived at during the staff's review of the Evolutionary Requirements Document may be developed as the staff completes its reviews of the detailed design information provided in the applications for FDA/DC and as these designs are subjected to the design certification rulemaking process. Therefore, the staff expects that the ALWR plant designers will comply with the issue resolutions adopted by the NRC staff during its reviews of applications for CDA/DC in accordance with the requirements of 10 CFR Part 52. The staff will evaluate this compliance during its review of an individual application for FDA/DC.

In the DSER for Section 10 of Chapter 1, the staff stated that the crossreferences in Table B.1-1 were incorrect. Because EPRI had committed to update the Table, the staff identified this as a confirmatory issue. The staff has verified that EPRI has corrected this table. Therefore, this DSER confirmatory issue is closed. Should EPRI make additional revisions to the Evolutionary Requirements Document, the staff will ensure that Table B.1-1 is upated appropriately.

Section 10.2.4 of Chapter 1 states that the plant designer will provide an ALWR design that is consistent with the disposition of the regulatory requirements and guidance identified in Section 1 of Appendix B. EPRI defines the commitment specified in Appendix B as follows:

Comply

The "comply" designation indicates that the ALWR design will comply fully with all regulatory requirements and guidance provided by the reference.

EPRI Evolutionary Plant SER

#### Optimization Subject

"Optimization subjects" are EPRI-initiated proposals to deviat 'OM regulatory requirements. EPRI proposes to resolve these issu." providing technically supportable alternatives to current regu requirements. EPRI specifies that the ALWR design will comply w at 911 regulatory requirements and guidance for a regulatory item associated with an optimization subject, except as described in Section 2 of Appendix B to Chapter 1 of the cvolutionary Requirements Document.

Section 10.2 of Chapter 1 states that the plant designer will provide a design that is consistent with Sections 2 and 3 of Appendix B to Chapter 1. These sections include EPRI-proposed resolutions for optimization subjects and generic safety issues, respectively.

In the DSER for Section 10 of Chapter 1, the staff indicated that NRC approval of the Requirements Document implies general agreement with the design criteria in the document, but such approval is not meant to imply that the document is a complete and adequate set of requirements for a nuclear power plant. The weaning of NRC's approval of the Requirements Document was listed as an open issue. As EPRI has stated, the Requirements Document is intended for use with companion documents, such as utility procurement specifications, that cover the remaining technical requirements for a specific plant. As discussed by the staff in Section 1 of Volume 1 of this report, the Requirements Document has no legal or regulatory status and is not intended to demonstrate complete compliance with the Commission's regulations, regulatory guidance, or policies. It is not intended to be used as a basis for supporting design certification for a specific design application, nor is it to be used to substitute for any portion of the staff's review of an individual application for FDA/DC. On the basis of the above discussion, this DSER open issue is closed.

### Inspections, Tests, Analyses, and Acceptance Criteria

As stated in 10 CFR 52.47(a)(1)(vi), applications for design certification must include proposed inspections, tests, analyses, and acceptance criteria (ITAAC) that are necessary and sufficient to provide reasonable assurance that, if the inspections, tests, and analyses are performed and the acceptance criteria met, a plant that references the design is built and will operate in accordance with the design certification. Section 3.3.5 of Volume I, "Policy and Top-Tier Design Requirements," of the Requirements Document requires the plant designer to prepare a set of ITAAC, with technical basis provided, that will demonstrate that the plant has been constructed and will be operated in conformity with Commission regulations, the combined license, and the Atomic Energy Act. The section also requires the nature and level of detail of acceptance criteria to be such as to allow the NRC staff to verify that the acceptance criteria have been met. Although Section 7.9 of Chapter 1 specifies certain requirements related to system completion and startup testing, the overall requirements in Chapter 1 do not provide guidance regarding the scope and content of ITAAC. In its letter dated May 13, 1991, EPRI indicated that it did not plan to provide ITAAC guidance, but was supporting the development by the Nuclear Management and Resources Council of industry guidance regarding ITAAC for design certification. Each plant designer will submit ITAAC for the specific design for NRC review using the industry's guidance.

EPRI Evolutionary Plant SER 1.10-2

The design detail to be developed by an applicant for design certification can be embcdied in three parts: (1) Tier 1 information that will be submitted in the application and certified by rulemaking, (2) Tier 2 information that will be submitted in an application but not certified, and (3) information not submitted but available for NRC audit. The two-tiered approach is to standardize design details to the maximum extent practicable, but allow the flexibility needed to finalize the design and construct the facility considering the procurement and design reconciliation process In SECY-90-377, "Requirements for Design Certification Under 10 CFR Part 52," dated November 8, 1990, the staff states that Tier 1 will include information developed during the conceptual phase, such as design criteria and bases, and certain information developed during the preliminary and detailed design phases, such as descriptions of systems and key components, functional and performance requirements for plant systems, simplified electrical single-line diagrams, simplified piping and instrumentation drawings, general arrangement drawings, and ITAAC.

The staff is developing additional guidelines for the scope and content of ITAAC and is even in pilot ITAAC submittals based on the General Electric advanced boiling sever reactor design. As described in SECY-91-178, "Inspections, Tests, Analyses, and Acceptance Criteria (ITAAC) for Design Certifications and Combined Licenses," dated June 12, 1991, Tier 1 ITAAC will be at a level of detail corresponding to the Tier 1 design information of the certified design rule. The staff expects that the Tier 1 verification requirements will be high level in nature and will address the design at a system functional performance level of detail. Numerical acceptance criteria values will only be specified when failure to meet the stated acceptance criteria would clearly indicate a failure to properly implement the design. Although including appropriate guidance on scope and content for ITAAC submittals the Evolutionary Requirements Document would ensure that each design submictal will include a complete and adequate ITAAC package for staff review, ITAAC is clearly the responsibility of the plant designer. The staff will evaluate the proposed ITAAC during its review of an individual application for FDA/DC or a combined license.

# 11 DESIGN PROCESS

### 11.1 Introduction

In Section 11 of Chapter 1 of the Evolutionary Requirements Document, EPRI proposes requirements for the design process for an ALWR. This process will include such activities as development, testing, analyses, preparation of specifications and drawings, modeling, documentation, and support of otners as re "ired to complete the licensing, construction, and startup of the plant, including turning the plant over to the operator. EPRI states that "licensing" is intended to include compliance with all Federal, State, and local regulations, including environmental impact reports.

#### 11.2 Technology Base

In Section 11.2 of Chapter 1 of this Evolutionary Requirements Document, EPRI states that ALWRs should be designed using systems, components, and equipment proven through several years of acceptable service in LWR plants. EPRI defines proven systems, components, and equipment to be those that have the same characteristics and that use materials proven under the same environmental and working conditions as those that have been successfully applied for at least several years in existing LWRs or similar operating environments and applications. Although the Evolutionary Requirements Document does not preclude the use of advanced technologies. EPRI states that unproven systems or equipment may be used only if sufficient justification is provided to support their use.

The staff concludes that these requirements do not conflict with the Commission's regulations and guidance. In the regulatory departure analysis in Appendix B to this chapter, the staff discusses the process it will use for determining the need for a prototype or other demonstration facility for the ALWR.

#### 11.3 Design Life

Section 11.3 of Chapter 1 of the Evolutionary Requirements Document provides requirements to be used to design for a plant life of 60 years. The staff concludes that these requirements do not conflict with the Commission's regulations and guidance. As discussed in Section 3.3 of this chapter, if requested by the applicant, the staff will review the ALWR designs for a 60-year life, notwithstanding the fact that a 40-year license term limitation is specified in the Atomic Energy Act and NRC's regulations.

In Section 8.2.C.3 of the DSER for Chapter 1, the staff recommended that EPRI require using the experience gained from two pilot plant studies (for Northern States Power and Virginia Electric Power plants) in the plant life extension program sponsored by EPRI and the Department of Energy. Section 11.3.4.5 of Chapter 1 of the Evolutionary Requirements Document appropriately references these studies. Therefore, this DSER confirmatory issue is closed.

# 11.4 Plant Simplification

Section 11.4 of Chapter 1 of the Evolutionary Requirements Document states that the ALWR design will be simpler than that of current operating plants because a minimum number of mechanical components (valves, pumps, heat exchangers, snubbers) and a minimum of instrumentation and controls will be used. EPRI specifies that the plant will be designed to simplify operations during all modes of operation, including operator actions to diagnose and manage abnormal and accident conditions.

The staff agrees with the overall objective to simplify systems and operations and will ensure that this objective has been implemented in accordance with the Commission's regulations and guidance during its review of an individual application for FDA/DC.

### 11.5 Standardization

Section 11.5 of Chapter 1 of the Evolutionary Requirements Document states that the ALWR design will be developed as a standard plant design, including, as a minimum, a standard design basis, standard site envelope, standard equipment, and standard technical documentation.

The staff supports the concept of standardization, as can be seen in the smulgation of 10 CFR Part 52, the Commission's standardization policy, and e staff's review of the EPRI Requirements Document. It, therefore, will sure that this concept has been implemented in accordance with the mmission's regulations and guidance during its review of an individual coplication for FDA/DC.

# 11.6 Specific ALWR Design Process Requirements

Section 11.6 of Chapter 1 of the Evolutionary Requirements Document specifies that the plant designer will use design methods required by applicable standards, codes, and regulations, as well as those based on the designer's own experience, methods, and design tools. EPRI specifies that the designer should use the special studies, evaluations, and design approaches that are provided throughout the Evolutionary Requirements Document.

In the regulatory departure analysis in Appendix B to this chapter, the staff discusses the applicable codes and standards to be used in the ALWR designs.

### 11.7 Project Information Network

Section 11.7 of Chapter 1 of the Evolutionary Requirements Document requires that a project information network (PIN) based on the guidelines and methodology of EPRI NP-5159, "Guidelines for Specifying Integrated Computer-Aided Engineering (CAE) Applications for Electric Power Plants," be established to define the activities, entities, attributes, and relationships for the total plant cycle. EPRI specifies that this network will be used by all ALWR plant design participants in organizing and identifying the products of the design process.

EPRI specifies that the PIN will segment the design into a number of systems and system groups, modeling, maintenance of the technical data base, systematic retrieval of information, standard identification of data, and

EPRI Evolutionary Plant SE
provision of a single source of technical data. The staff's evaluation of the PIN and its relationship to the information management system that EPRI specifies is provided in Section 11.12 of this chapter.

# 11.8 Design Development Plan

Section 11.8 of Chapter 1 of the Evolutionary Requirements Document specifies that the plant designer will develop a plan for the development and implementation of the plant design, including the responsibilities and authorities of the design organization, implementation plan, and schedules. These requirements do not conflict with the Commission's regulations and guidance, and are, therefore, acceptable.

# 11.9 Configuration Management

Seri on 11.9 of Chapter 1 of the Evolutionary Requirements Document states that the plant designer should develop a configuration management program to be used throughout all phases of the plant's life, including the design phase. The program is intended to ensure that documents used in defining the design basis, technical baselines, as-built configuration, and procedures will be available over the life of the plant.

This section originally included information about the plant security system. In its letter dated October 9, 1991, EPRI elected to remove documents and information that would require protection as safeguards informat from the lists of plant documentation specified in Attachments 1 and 2 to section 11. The staff concludes that this deletion addresses its concern raised in its letter dated August 19. 1991, and is, therefore, acceptable.

The requirements in this section do not conflict with the Commission's regulations and guidance and are, therefore, acceptable.

# 11.10 Design Integration

In Sections 2.2.F.3 and 2.2.F.4 of the DSER for Chapter 1, the staff recommended that the use of "living" probabilistic risk assessment (PRA) models in developing the overall design be required in the Evolutionary Requirements Document. The staff's evaluation of Appendix A to Chapter 1, which discusses EPRI's proposed PRA guidelines, is provided in the corresponding appendix of this report and supersedes this issue. Therefore, this DSER confirmatory issue is closed.

Section 11.10 of Chapter 1 of the Evolutionary Requirements Document specifies requirements to ensure that the design of systems and subsystems of an ALWR will be acceptably integrated to minimize the need for redesign and backfitting, to minimize failure rates, and to minimize planned and unplanned outage times. EPRI specifies the use of modeling, plant simulation, documentation, and evaluation of operating experience to achieve such minimization. These requirements do not conflict with the Commission's regulations and guidance and are, therefore, acceptable.

### 11.11 Interdisciplinary Design Reviews

Section 11.11 of Chapter 1 of the Evolutionary Requirements Document requires that an interdisciplinary design review be conducted by a number of highly skilled groups of distinct disciplines, in order that participation in the design reviews by a wide range of experts will help to uncover potential problems during the design stage.

In its letter dated March 1, 1991, the staff stated that although experts in many disciplines were specifically mentioned as being required to participate in the review group, experts in physical security were not included. In its letter dated May 13, 1991, EPRI noted that the makeup of the design review team was limited to the major engineering disciplines and activities because it was impractical to list all possible specialties that will be required for the review. Nevertheless, in its June 11, 1991, markup, EPRI modified Section 11.11.2 to specify that additional special disciplines will be included as appropriate for the subject under review. The staff concludes that this change resolves its concern and is acceptable.

The requirements in this section do not conflict with the Commission's regulations and guidance and are acceptable.

### 11.12 Information Management System

Section 11.12 of Chapter 1 of the Evolutionary Requirements Document specifies that the plant designer will use appropriate computer hardware and software to establish, manage, and operate an information management system (IMS) during the design process and will provide the IMS to the plant owner for use during construction and operation. EPRI states that the objectives of the IMS are the following:

- provide an effective means to acquire, store, retrieve, and manipulate the documents and data necessary to design, construct, start up, operate, and maintain the plant
- make effective use of computer-aided design and engineering during design, construction, and operation
- provide for implementation of a PIN as discussed by the staff in Section 11.7 of this chapter
- use the computer in configuration management
- ensure that information needed for construction and operations is available

Attachment 1 to Section 11 of Chapter 1 gives the functional and operational requirements for the IMS.

In its letter dated May 13, 1991, EPRI committed to revise the computer and network security provisions of Section 5.9 of Attachment 1 to clarify that these provisions apply to protection from unauthorized disclosure of sensitive information as well as data loss or contamination. The staff has verified that these changes were included in Revision 3 of this section. The staff discussed EPRI's proposed requirements for this system in SECY-91-226, "NRC Use of Computer and Computer Graphics-Aided Reviews for Advanced Reactor Designs and Nuclear Facilities," dated July 29, 1991. The industry is beginning to develop programs to define the requirements for information management systems for designing and operating new reactors and nuclear facilities. The staff is establishing an advanced technology group to develop strategies to ensure that the staff understands how the industry will use computer technologies to design, construct, and operate advanced reactors and nuclear facilities and to ensure that the NRC establishes an effective regulatory process to be consistent with the industry's approach to the use of these technologies.

### 11.13 Design/Construction Integration

Section 11.13 of Chapter 1 of the Evolutionary Requirements Document states that the construction schedule should be based on a standard plant design that is essentially complete, except for required site-specific engineering. The requirements of this section are intended to ensure that design documentation is completed on a schedule that minimizes its effect on the construction schedule.

Attachment 2 to Section 11 of Chapter 1 summarizes ALWR design activities. Item 1 in that attachment originally listed activities to be completed for design certification and safety determination. In its letter dated May 17, 1991, EPRI stated that because design certification and combined license rulemaking were beyond the scope of the Evolutionary Requirements Document, it would remove the reference to design certification. The staff has verified that this change was made in Revision 3 of this section and is acceptable. Item 2 in Attachment 2 summarizes in detail design engineering required to support the construction schedule. Some of the detailed design items in Item 2 have asterisks to indicate that parts of those items will be completed in the standard plant design, rather than by the plant owner. In its letter dated May 13, 1991, EPRI agreed to revise Item 2 to indicate that portions of the security plan (layout and detail drawings) will be part of the standard plant design. These changes were included in EPRI's June 11, 1991, markup; however, EPRI later elected to delete the security plan (layout and detail drawings) from Attachment 2 in response to the staff's concerns regarding protection of the information from unauthorized disclosure. The staff concludes that this is acceptable because it satisfies its concern regarding protection of safeguards information.

These requirements do not conflict with the Commission's regulations and guidance. However, as discussed by the staff in Section 7 of this chapter, the owner/builder must ensure that construction activities permit verification of the acceptability of the plant configuration in accordance with the requisite NRC Inspection Manual chapters. The staff will address this matter during its review of an individual application for a combined license.

### 11.14 Engineering Field Verification of As-Built Conditions

Section 11.14 of Chapter 1 of the Evolutionary Requirements Document states that the plant designer should identify and perform necessary engineering field verification activities to confirm the adequacy of the installation. The requirements in this section do not conflict with the Commission's regulations and guidance and are, therefore, acceptable.

# 11.15 Conclusion

The staff concludes that, with the exceptions noted above, the requirements in Section 11 of Chapter 1 of the Evolutionary Requirements Document do not conflict with the Commission's regulations and guidance and are, therefore, acceptable.

### 12 MECHANICAL EQUIPMENT DESIGN REQUIREMENTS

# 12.1 Introduction

B

Section 12 of Chapter 1 of the Evolutionary Requirements Document specifies requirements for the mechanical equipment used in many systems of a nuclear plant.

### 12.2 Inservice Testing of Pumps and Valves

In Section 2.2 of the DSER for Chapter 1, the staff stated that EPRI had agreed to include an appropriate reference to ASME Code, Section XI, in a future revision of Chapter 1. This reference has been acceptably added to Chapter 1. Therefore, this DSER confirmatory issue is closed. However, as a result of its review of the detailed information in Chapter 1 relative to inservice testing of pumps and valves, the staff requested that EPRI supply additional information, as discussed below. This discussion addresses issues identified by the staff in Chapters 5, 6, and 8 of this report.

Sections 12.2.7.1 and 12.4.3.2 of Chapter 1 of the Evolutionary Requirements Document state that system designs will include provisions for inservice testing of essential pumps and valves in accordance with American National Standards Institute/American Society of Mechanical Engineers (ANSI/ASME) OM-6, "Inservice Testing of Pumps," and OM-10, "Inservice Testing of Valves." These two standards are now referenced in ASME Code, Section XI, "Rules for Inservice Inspection of Nuclear Power Plant Components."

In SECY 90-016, the staff concluded that the requirements of Section XI of the ASME Code provide information on the operational readiness of the components, but, in general, do not necessarily provide for the verification of the capability of the components to perform their intended safety functions. It concluded that the ASME Code does not ensure the level of component operability that is desired for the evolutionary ALWR designs (see the staff's regulatory departure analyses in Appendix B to this c'apter).

Accordingly, in SECY-90-016, as supplemented by the staff's April 27, 1990, response to comments by the Advisory Committee on Reactor Safeguards (ACRS), the staff recommended criteria to the Commission to be used to supplement Section XI of the ASME Code. In its staff requirements memorandum on SECY-90-016, dated June 26, 1990, the Commission approved the staff's position for evolutionary LWRs. In addition, the staff agreed with the recommendation of the ACRS that the guidelines of Generic Letter 89-10, "Safety-Related Motor-Operated Valve Testing and Surveillance," be applied to evolutionary LWR designs. The ACRS also recommended that the staff resolve the check valve testing and surveillance issue and that consideration be given to industry-proposed alternative ways of meeting inservice testing and surveillance requirements. The staff's review of the inservice testing programs for pumps and valves in both the evolutionary and passive ALWR designs is based on these guidelines.

In its letter dated December 22, 1989, EPRI responded to the staff's questions on Chapter 8 of the Evolutionary Requirements Document. In these responses, EPRI committed to modify Section 12 of Chapter 1 to address the staff's concerns. However, in Section 3.2 of the DSER dated January 15, 1991, for

Chapter 8 of the Evolutionary Requirements Document, the staff concluded that EPRI's responses were not entirely acceptable. To satisfactorily implement the enhanced criteria discussed above, the staff concluded that additional modifications should be made to the Evolutionary Requirements Document. In its letter dated May 17, 1991, the staff requested additional information on the Evolutionary Requirements Document. The staff noted in that letter that, with certain exceptions, the request for additional information (RAI) was applicable to both evolutionary and passive plants. EPRI responded to that RAI in its letter dated August 1, 1991. The evaluation that follows is based on the information submitted in Revision 3 of the Evolutionary Requirements Document and on EPRI's August 1, 1991, response.

# 12.2.1 Scope of the Inservice Testing Program

In its response to the staff's request to clarify the term "essential" in regard to the scope of the inservice testing (IST) program for pumps and valves, EPRI stated that for its ALWR program, an "essential" component or system is viewed by the utilities as important for safety, reliability, and availability and for protecting their investment. This response is consistent with the staff's position that all safety-related equipment, including non-Code class safety-related pumps and valves, should be tested in accordance with ASME Codes, Section XI, and is, therefore, acceptable.

### Pump and Valve Reliability

In its letter dated May 17, 1991, the staff requested that EPRI provide a commitment to undertake a reliability assurance program to identify key components, to assess the applicability of existing data to these components, and to develop reliability estimates for these components. In its letter dated August 1, 1991, EPRI referred to several sections in Chapter 1 and one section in Chapter 5 of the Evolutionary Requirements Document. These sections generally require the plant designer to develop reliability estimates for key components and to perform sensitivity studies to better understand the impact of data uncertainties. This response is too broad to be completely acceptable. Specifically, with respect to existing data, the staff requires a commitment to assess the applicability of such data to safety-related components. Furthermore, where existing reliability data are not applicable, the expected reliability of those components will have to be developed through testing. Pending such a commitment, the staff will evaluate individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, this issue is closed.

In its letter dated May 17, 1991, the staff requested that the ALWR reliability program be used to determine actions necessary to improve component reliability and to maintain the desired component reliability level through the life of the plant. In its August 1, 1991, response, EPRI referred to several sections in Chapter 1 that were related to reliability and availability. In particular, it referred to Section 6.2.3 of Chapter 1, Revision 3, which states that the plant designer will prepare an analysis showing that the designs of the plant systems and supporting maintenance systems, recommended spare parts, surveillance tests, and test intervals will be adequate to meet the availability requirements. In the rationale portion of that section, EPRI stated that analyses relating plant system and component reliability to availability for the ALWR are necessary for making adequate decisions about plant system design. EPRI states that the sections referenced

in its August 1, 1991, response provide requirements to ensure that appropriate actions wil' be taken to improve reliability and that component reliability levels will be maintained. This response provides the requested information and is, therefore, acceptable.

In its letter dated May 17, 1991, the staff requested that testability of safety-related valves be provided for early in the design phase. The valve testability issue is discussed in Section 12.2.2 below.

### 12.2.2 Valves

During its review of Chapter 8 of the Evolutionary Requirements Document, the staff requested that EPRI propose a frequency of disassembly and inspection of safety-related check valves and motor-operated valves. EPRI stated that the frequency of valve inspections was beyond the scope of the Evolutionary Requirements Document. As discussed in Section 3.2 of the DSER for Chapter 8, the staff concluded that EPRI's response was not acceptable. In its May 17, 1991, letter, the staff again requested, as a minimum, a commitment to develop a program that will establish the frequency and extent of disassembly and inspection of safety-related valves, including the basis for the frequency and extent of each disassembly. In Revision 3 of Section 12.2.7.8 in Chapter 1 of the Evolutionary Requirements Document, EPRI provided a commitment that is consistent with the staff request.

In its letter dated August 1, 1991, responding to a concern raised on check valve testing requirements and testability, EPRI stated that the requirements in Section 12.2.7 of Chapter 1 of the Evolutionary Requirements Document were in compliance with the Commission's guidance in the staff requirements memorandum dated June 26, 1990, on SECY-90-016, and that a detailed design analysis considering such factors as component design, application, PRA insights, and design alternative was the appropriate method to determine the best valve for the application and the best testing method. EPRI stated further that for safety-related check valves that require testing, the plant designer will submit the details for NRC approval at the FDA/DC phase of each passive ALWR. The staff concludes that this response is not entirely acceptable. Although this issue is applicable to both the evolutionary and passive ALWRs, EPRI has not addressed it for the evolutionary plant design. The staff disagrees with EPRI's position that a commitment to check valve testing methods requires a detailed design analysis and concludes that a commitment to check valve testing methods should be part of the Evolutionary Requirements Document. In addition, a requirement should be added to the list of guidelines from EPRI NP-5479, "Application Guidelines for Check Valves in Nuclear Power Plants," in the requirement portion of Section 12.2.6.1 of Chapter 1 to state that, in the selection and application of valves, the plant designer should also consider parts clearance, disc stability, and wear relative to actual operational flow conditions. Pending such changes, the staff will review individual applications for FDA/DC in accordance with the above positions. On the basis of the above discussion, this issue is closed.

In its letter dated May 17, 1991, the staff requested that EPRI revise Section 12.2.7.2 in Chapter 1 to reflect the staff's position on full-flow testing of check valves as described in the letter. In its August 1, 1991, response, EPRI referred to its position as provided in its responses to RAI 210.39(b) and (f). For reasons similar to those discussed above, EPRI's response is not acceptable. The staff maintains that testing method and

testability are important parts of reliability assurance and that a commitment to the staff's position on full-flow testing of check valves should be part of the Evolutionary Requirements Document. Pending such a commitment, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, this issue is closed.

In its letter dated May 17, 1991, the staff requested additional information regarding the use of nonintrusive diagnostic techniques for check valves. In Revision 3 of Section 12.2.7.4 of Chapter 1 of the Evolutionary Requirements Document, EPRI states that check valve and system design will include provisions for the use of nonintrusive diagnostic techniques. This commitment is consistent with the staff's May 17, 1991, request and is, therefore, acceptable.

In its letter dated May 17, 1991, the staff requested additional information on motor-operated valve (MOV) testing requirements. Revision 3 of Sections 12.2.7.6 and 12.3.2.3.4 of Chapter 1 reflects the staff's comments relative to demonstrated design-basis capability. In particular, where in situ design-basis testing is not practical, MOVs with well-understood performance characteristics will be used. Also, requirements for preinstallation testing have been expanded to include full and partial designbasis conditions. Justification will be required to show that qualification tests apply to all other valves of the same type, size, and conditions. The staff concludes that the above commitments are consistent with staff positions relative to this issue and are, therefore, acceptable.

In its May 17, 1991, letter, the staff requested additional information on qualification testing of active and nonactive MOVs. In its August 1, 1991, response, EPRI stated that nonactive MOVs should be designed for potential mispositioning, but qualification testing was not required. This statement does not completely agree with the current staff guidelines on valve mispositioning. The staff concludes that nonactive MOVs in a safety-related system either should be designed to prevent mispositioning or should be required to be subjected to qualification testing to demonstrate their capability to recover from mispositioning. Mispositioning may occur through actions taken at any time locally (manual or electrical), at a motor control center or in the control room, and includes deliberate changes of valve position for performing surveillance testing. Therefore, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, this issue is closed.

Recent industry experience and the results of NRC inspections of MOV programs indicate several areas require attention in the EPRI document. Specifically, in addition to the technical information described in Section 12.2.2.5 of Chapter 1 of the Evolutionary Requirements Document that is to be provided with each valve, operator loads as a function of fluid temperature (subcooling) and seismic/dynamic effects, as well as precise internal dimensions of the valve, should be provided. In addition to consideration of stem leakage in establishing the proper globe valve orientation described in Section 12.2.2.6.2 of Chapter 1, any reliance on a globe valve to isolate flow or the use of the valve for throttling flow should also be considered in establishing proper orientation. In addition to ensuring that the valve bonret and disc will be designed to prevent pressurization due to heatup of fluid trapped in the bonnet as described in Section 12.2.2.8.2 of Chapter 1, the bonnet should be designed so that its internal pressurization is not

greater than that of both the upstream and downstream piping, or the motor operator should be designed to overcome such pressurization. EPRI should revise Section 12.3.2.3.3 of Chapter 1 to require that provisions be made for measuring both stem thrust and actuator torque because of the importance of information regarding the conversion of torque to thrust (i.e., stem factor). As a clarification of Section 6.2.2.1.4 of Chapter 5 regarding the capability of isolation valves to close against conditions that may exist during events requiring containment isolation, the isolation valves should be designed and test-qualified to be able to isolate flow resulting from a pipe break at the worst-case differential pressure (e.g., a condition resulting from a failure to scram the reactor in a timely manner), because the potential for a break in a line from the reactor vessel would likely be greatest when the reactor pressure was abnormally high. Penuing modification of these sections, the staff will review individual applications for FDA/DC in accordance with the above positions. On the basis of the above discussion, this issue is closed.

In it: letter dated May 17, 1991, the staff requested that EPRI address compliance with General Design Criterion (GDC) 54 of Appendix A to 10 CFR Part 50 with respect to containment isolation valve (CIV) design and to commit to leak test these valves to appropriate limits. GDC 54 includes requirements to design for the capability to leak test CIVs in all piping systems penetrating the containment. This requirement applies to the primary and secondary systems penetrating the containment. In its letter dated August 1, 1991, EPRI referred to Table B.1-1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, which states that the design will meet GDC 54.

In its May 17, 1991, letter, the staff also discussed some limitations of the Type C leak rate testing requirements of Appendix J to 10 CFR Part 50 for CIVs. In particular, Appendix J Type C leak rate testing adequately determines the containment leaktightness provided by the valves included in this test program. However, these tests do not require that individual valve leakage limits be defined, nor is corrective action required based on individual valve leakage rates. ASME/ANSI OM-10 only requires CIVs to be tested in accordance with Appendix J. Therefore, the staff requested that EPRI revise the Evolutionary Requirements Document to require analyses of leakage rates and corrective actions for CIVs. Acceptable requirements for the analysis of leakage rates and corrective action are contained in either Paragraph IWV 3421-3427(a) of Section XI of the ASME Code or Paragraph ISTC 4.3.3 of ASME/ANSI OM-10, 1990.

In its response dated August 1, 1991, EPRI explained that the requirement in Section 6.2.2.2 of Chapter 5 of the Evolutionary Requirements Document was intended to require the plant designer to minimize the number of valves that will be subjected to Type C testing in accordance with Appendix J to 10 CFR Part 50 rather than to set down the type of testing required for CIVs. EPRI's response also referred to Section 12.2.7.1 of Chapter 1 of the Evolutionary Requirements Document, which requires the plant designer to provide for testing of essential valves in accordance with ASME/ANSI OM-10. Furthermore, EPRI stated that the designation of specific inservice testing requirements was beyond the scope of the Evolutionary Requirements Document and properly belonged in FDA/DC documentation. The staff disagrees with EPRI's position that the designation of specific inservice testing requirements is beyond the scope of the Evolutionary Requirements. For the reasons discussed above, the staff has also determined that EPRI's response will not result in

individual CIV leak rate testing and is, therefore, not acceptable. The staff will review individual applications for FDA/DC in accordance with the above position.

# 12.2.3 Pumps

As stated in Section 3.2 of the DSER for Chapter 8 and the staff's letter dated May 17, 1991, it is the staff's position that the pumps for ALWR designs shoul be provided with instrumentation to verify that the net positive suction, head (NPSH) is greater than or equal to the NPSH required during all modes of pump operation. In Section 12.4.3.2 of Chapter 1 of the Evolutionary Requirements Document, EPRI indicates that its provision to test pumps in accordance with ASME/ANSI OM-6 will ensure that sufficient instrumentation and test connections are provided to monitor pump performance and trend degradation. However, OM-6 does not address the above staff position. Therefore, the staff requires a more explicit commitment to provide instrumentation to measure suction pressure. Since EPRI has not responded to this staff concern, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, this issue is closed.

In its letters dated December 22, 1989, and August 1, 1991, EPRI responded to the staff requests for additional information regarding the capability for pump flow testing. EPRI committed to modify Section 12 of Chapter 1 in accordance with the staff position that system design should provide the capability for pump flow testing at rates at least as large as the design flow rate. However, in the revised Section 12.4.3.1 of Chapter 1, EPRI states that system design will provide the capability for pump flow testing on a periodic basis at a flow rate that adequately verifies operability and will not result in pump degradation. It further states that, if practicable, the capability for system flow testing during normal operation will be provided and that, as a minimum, the capability will be approximately 60 percent of the design flow rate. However, 60 percent of the design flow rate is not acceptable. Therefore, the staff will review individual applications for FDA/DC in accordance with its position that the capability to test at 100 percent of the design flow rate should be provided. On the basis of the above discussion, this issue is closed.

In response to the staff's concern regarding the adequacy of the mini-flow system as discussed in Section 3.2 of the DSER for Chapter 8, EPRI deleted the phrase "25 percent of the pump design flow" from the rationale portion of Section 12.4.1.3 of Chapter 1 of the Evolutionary Requirements Document. In addition, EPRI states that system designs will be such that the pumps will not operate below the minimum flow required for pump protection for all operating modes. If minimum recirculation flow lines are required, these lines will be sized to ensure that degradation will not occur as a result of continuous operation on bypass. The design of the minimum flow lines, if installed, will permit periodic testing to verify the flow is in accordance with design. This commitment addresses the staff's concern, conforms to the applicable staff guideline, and is, therefore, acceptable.

In Section 3.2 of the DSER for Chapter 8 and in its letter dated May 17, 1991, the staff requested that EPRI provide a commitment to periodically disassemble and inspect all safety-related pumps. The staff requires, as a minimum, a commitment to develop a program that will establish the frequency and the

extent of disassembly and inspection of safety-related pumps, including the basis for the frequency and extent of each disassembly. EPRI has not responded to this request for the evolutionary plant. Pending such a commitment, the staff will review individual applications for FDA/DC in accordance with the above position. On the basis of the above discussion, this issue is closed.

# 12.2.4 Conclusion

The staff concludes that, with the exceptions noted above, Section 12 of Chapter 1 of the Evolutionary Requirements Document satisfies the staff guidelines discussed in Section 12.2 of this chapter relative to inservice testing of pumps and valves and is, therefore, acceptable.

### 12.3 Radiation Protection Considerations

### Pumps

Section 12.4 of Chapter 1 of the Evolutionary Requirements Document states that pumps will be designed with flanged connections to facilitate removal and replacement. Pumps in radiation areas will have long-life bearings and permanent-type lubrication where practical. Pumps in nuclear service will be provided with drain and flush connections to facilitate decontamination.

#### Tanks

Section 12.6 of Chapter 1 states that all tanks will be designed to prevent the unintended retention of particulate material by the incorporation of one or more of the following features: sloped or cone-shaped tank bottoms, grinding of internal welds to minimize crud traps, tank flushing capability, and lancing or chemical cleaning capability.

# Filters and Ion Exchangers

Section 12.9 of Chapter 1 states that process vessel manways will be sized to accommodate personnel wearing anticontamination clothing and remote or semiremote tools will be available to facilitate maintenance of vessel internals. Filters used will require minimal time and effort to keep in service. Cartridge filters that will have high radiation levels will be designed to be changed remotely. Standardized techniques for filter handling will be used throughout the plant to ensure that doses will be ALARA during filter maintenance and changeout. Bag filter housings will be designed for ease of removal. Such features are intended to minimize the personnel radiation exposure associated with changing filters and, therefore, are acceptable.

Ion exchangers will have downstream resin traps to prevent the escape of resins in the event of a failure of an internal screen in the ion exchanger. These resin traps will be located outside the ion exchanger enclosure (in a lower radiation area) and will be provided with a remote backwash capability. This remote backwash capability will permit the resin traps to be cleaned remotely from a low-dose-rate area, thereby eliminating the associated personnel radiation exposure normally accrued during cleaning of resin traps.

# Conclusion

The staff concludes that the design features of Section 12 of Chapter 1 of the Evolutionary Requirements Document do not conflict with the applicable guidance in Regulatory Guide 8.8 for minimizing personnel exposures, and SRP Sections 12.1, "Assuring That Occupational Radiation Exposures Are as Low as Is Reasonably Achievable," and 12.3-12.4, "Radiation Protection Design Features," and, therefore, are acceptable.

#### 13 CONCLUSION

# Chapter 1 of the Evolutionary Requirements Document

On the basis of its review, subject to resolution of the identified outstanding issues, the staff concludes that the requirements in Chapter 1 of the Evolutionary Requirements Document do not conflict with current regulatory guidelines and are acceptable. However, Chapter 1 contains requirements that are so general in nature that it is difficult for the staff to determine whether their implementation will meet the Commission's regulations, guidance, and practices. The staff's review of the remainder of the document is discussed in detail in the chapters that follow, providing a clearer picture of the extent to which the Evolutionary Requirements Document complies with the Commission's regulations. The conclusions regarding the staff's review of the entire Evolutionary Requirements Document are provided below. Applicants referencing the Evolutionary Requirements Document will be equired to demonstrate compliance with the additional guidance in the Standard Review Plan (NUREG-0800) or provide justification or alternative means of implementing the associated regulatory requirements.

#### Evolutionary Requirements Document.

On the basis of its review, subject to resolution of the identified outstanding issues listed in Section 1.4 of each chapter of this report, the staff concludes that the requirements in the Evolutionary Requirements Document (Volume II) do not conflict with current regulatory guidelines and are acceptable. However, by themselves, they do not provide sufficient information for the staff to determine if the evolutionary design will be adequate. Therefore, applicants referencing the Evolutionary Requirements Document will be required to demonstrate compliance with the additional guidance in the Standard Review Plan or provide justification or alternative means of implementing the associated regulatory requirements.

In staff requirements memoranda (SRM), the Commission instructed the staff to provide an analysis detailing where the staff proposes departure from current regulations or where the staff is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs. The staff considers uses to be policy issues. Appendix 8 to this chapter provides that analysis. The staff forwarded these issues to the Commission in SECY-90-016, SECY-91-078, and a draft Commission paper issued on February 27, 1992.

In its SRM dated June 26, 1990, and August 15, 1991, the Commission provided its decisions on SECY-90-016 and SECY-91-078, respectively.

After the staff issues the February 27, 1992, draft Commission paper in final form, the Commission will complete its review of the basis for the approach that the staff is proposing for those issues and, accordingly, may at some future point in the review determine that those issues involve policy questions that the Commission may wish to consider. Because the Commission has not reviewed the approaches to resolving these issues, they do not represent agency positions. In addition, certain technical issues still have to be resolved before the staff car complete its review. Therefore, the staff concludes that the Evolutionary Requirements Document (Volume II) specifies requirements that, subject to the resolution of the identified open issues and vendor- and utility-specific items, if properly translated into a design and constructed and operated in accordance with the NRC regulations in force at the time the design is submitted, should result in a nuclear power plant that will have all the attributes required to ensure that there is no undue risk to the health and safety of the public or to the environment. In addition to complying with existing regulations, such a facility would also be consistent with the Commission's policies on severeaccident protection.

#### CHAPTER 1, APPENDIX A, "PRA KEY ASSUMPTIONS AND GROUNDRULES"

# 1 INTRODUCTION

This appendix of the SER documents the NRC staff's review of Appendix A, "PRA Key Assumptions and Groundrules," to Chapter 1 of Evolutionary Requirements Document through Revision 3. Appendix A to Chapter 1 was prepared, under the project direction of EPRI and the ALWR Utility Steering Committee, by Duke Power Company; Jack R. Benjamin Associates; SAROS; TENERA, L.P.; and EPRI.

On June 30, 1989, EPRI submitted Appendix A to Chapter 1 for staff review. By letters dated April 10, July 13, and August 17 and 23, 1990, the NRC staff requested that EPRI supply additional information. EPRI provided the information in its response dated November 7, 1990.

On November 4, 1991, the staff issued its DSER for Appendix A to Chapter 1. On December 11, 1991, the staff and EPRI met with the Advisory Committee on Reactor Safeguards (ACRS) Subcommittee on Improved Light Water Reactors to discuss this appendix, the staff's corresponding DSER, the outstanding issues from the staff's review of the appendix, and EPRI's approach to resolving each issue.

On September 7, 1990, E<sup>---</sup> submitted Revision 1 of the Evolutionary Requirements Document. Revisio. 2, 3, and 4 were docketed on April 26 and November 15, 1991, and April 17, 1992, respectively. This appendix documents the staff's review of Appendix A to Chapter 1 of the Evolutionary Requirements Document through Revision 3, as well as its review of EPRI's responses to the staff's DSER that were docketed as pen-and-ink changes to the Evolutionary Requirements Document by letter dated January 24, 1992. EPRI has incorporated these changes as Revision 4.

# Use of the Probabilistic Risk Assessment in Future Designs

It is not the Commission's intent that probabilistic risk assessments (PRAs) for evolutionary plants merely be used to validate an already frozen design, but rather that the PRAs be used as a design tool to improve the design, enhance safety, and help provide valuable insights into specific plant vulnerabilities. Consistent with this philosophy, the Commission expects evolutionary plant vendors to make use of PRA insights to help in ensuring that designs have an appropriate balance of prevention and mitigation from a severe-accident standpoint and that the designs benefit from PRA safety insights. The staff intends that its reviews of evolutionary plant PRAs will not focus on checking only the quality of the PRA, but also how well the vendor or utility continues the use of the PRA tool to minimize the estimated core damage frequency and offsite consequences.

In its DSER for Appendix A to Chapter 1, the staff identified the use of PRA in design as an open issue. In response to the staff's DSER, EPRI modified the Evolutionary Requirements Document to indicate that the design-specific PRA will be integrated into the design process to enhance and improve the design (including meeting EPRI's core damage frequency and public risk objectives). Although the EPRI response is generally acceptable, FDA/DC applicants must also submit a description of the following:

- how PRA insights specifically influenced the design process
- what design features, if any, were added to or removed from the design as a result of PRA insights
- how plant operating experience was factored into the design-specific PRA
- how it was determined (including criteria) if there were any vulnerabilities in the plant design for internal and external events
- how the PRA was used to develop an appropriate balance of prevention and mitigation

The staff concludes that EPRI has provided guidance to vendors and designers on the use of a PRA during the design process and, therefore, this open issue is closed.

FDA/DC certification of a design will be based in part on a PRA of that design. Since the validity of a PRA is highly dependent on the assumed reliability of systems, structures, and components, the staff has determined that the plant designer should provide a reliability assurance program (RAP) as part of the FDA/DC application. This issue is discussed in this report in Section 6 of Chapter 1 and under Item II.M of the February 27, 1992, draft Commission paper on policy issues continued in Annex C of Appendix B to Chapter 1.

There will be three phases of the PRA for ALWRs:

- The development of the design-specific PRA by the ALWR vendor for FDA/DC.
- (2) The updating of the PRA (or the use of PRA techniques) at the combined license stage to include site-specific external events.
- (3) The updating of the PRA after the combined license has been issued to include as-built information such as actual equipment capacities, control room details, and updated human factors analysis. This phase is expected to be completed before startup of the plant and to continue through the life of the plant with the PRA being maintained as part of the RAP.

The PRA will be a u ful tool in developing appropriate plant maintenance activities. The NRC has recently promulgated the maintenance rule (10 CFR 50.65), which requires commercial power plant licensees to monitor the effectiveness of maintenance for safety-significant plant equipment in order to minimize the likelihood of failures and events caused by lack of effective maintenance. The rule, which will become effective in July 1996, requires licensees to monitor the performance or condition of certain systems, structures, and components (SSCs). Licensees are to compare monitoring results against licensee-established goals in a manner sufficient to provide reasonable assurance that those SSCs will be capable of performing their intended

functions. The goals are to be established commensurate with the safety significance of each SSC. Key assumptions in the plant-specific PRA and the PRA insights are to be considered with respect to how the assumptions and insights should affect plant maintenance. The rule encourages licensees to develop "living" PRAs.

The issue of a living PRA was identified as a confirmatory issue in the DSER for Chapter 1 because Appendix A had not yet been submitted when the staff prepared the DSER. The staff concludes that EPRI's key assumptions and groundrules rovide guidance to vendors on the conduct of a PRA. However, the staff will have to review the issue of a living PRA during its combined license review and during plant operation. Therefore, this confirmatory issue is closed.

#### 1.1 Review Criteria

Section 1 of Volume 1 of this report describes the approach and review criteria used by the staff during its review of Appendix A to Chapter 1 of the Evolutionary Requirements Document.

## 1.2 Scope and Structure of Appendix A to Chapter 1

Appendix A to Chapter 1 of the Evolutionary Requirements Document defines the ALWR Utility Steering Committee's overall requirements for conducting a PRA.

The key topics addressed in the staff's review of Anpendix A to Chapter 1 include

- scope of the PRA
- initiating events
- success criteria
- uncertainty analysis
- plant modeling
- human reliability
- containment analysis
- reliability data
- reliability assurance program
- seismic analyses
- fire analyses
- internal and external flood analyses
- use of PRA in design

# 1.3 Policy Issues

During its review of Appendix A to Chapter 1 of the Evolutionary Requirements Document, the staff did not identify issues that involve policy questions for the technical areas discussed in this appendix, other than those already identified in the Commission papers listed in Appendix B to Chapter 1 of this report.

# 1.4 Outstanding Issues

The DSER for Appendix A to Chapter 1 of the Evolutionary Requirements Document contained the following outstanding issues:

EPRI Evolutionary Plant SER 1A.1-3

### Open Issues

- (1) use of PRA in design (1)
- (2) events other than those initiated during full-power operation (1.6)
- (3) consideration of reactivity accidents (1.7)
- (4) definition of core damage (1.7)
- (5) uncertainty treatment (1.9)
- (6) form of the results (1.10)
- (7) initiating events (2.2)
- (8) truncation (2.5)
- (9) nested solution process (2.5)
- (10) mission time (2.10)
- (11) use of EPRI-proposed failure data (2.11)
- (12) instrumentation and controls reliability data (2.11)
- (13) tornadoes and extreme winds (3.2)
- (14) external flooding (3.2)
- (15) internal fires (3.2)
- (16) internal flooding (3.2)
- (17) quantitative uncertainty analysis for seismic events (3.3)
- (18) hazards analysis for ground response spectrum (3.3)
- (19) in-plant sequence assessment (4.5)
- (20) containment event analysis (4.6)
- (21) details of uncertainty analysis (4.6)
- (22) implementation of EPRI's public-safety requirements (5.1)
- (23) analysis of systems and sequences (6.1)
- (24) assessment of containment response (i 2)
- (25) analysis of source terms (6.3)
- (26) use of physically based source term (6.3)
- (27) human reliability analysis considerations (7.1)

EPRI Evolutionary Plant SER 1A.1-4

- (28) function, task, timeline, and link analyses (7.3)
- (29) performance shaping factors (7.3)
- (30) evaluation tools for performance shaping factors (7.3)
- (31) quantification methods for human reliability analysis (7.3)
- (32) reliability data base for consideration of loss of offsite power (Annex A)
- (33) failure rate for the main step-up transformer (Annex A)
- (34) ALWR reference site data (Annex B)

# Confirmatory Issue

)

generic data sources (7.3)

The final disposition of each of these issues is discussed in detail in the appropriate section of this appendix, as indicated by the parenthetical notation following each issue. All issues identified in the DSER for Appendix A to Chapter 1 have been resolved.

# 1.5 Vendor- or Utility-Specific Items

The vendor- or utility-specific items, with references to appropriate sections of this appendix given in parentheses, are listed below. The designators in front of each issue provide a unique identifier for each issue. The letter "E" indicates that the issue applies to evolutionary plant designs. The first number designates the chapter (or appendix) in which it is identified. The letter "V" designates that it is a vendor- or utility-specific item. The final number is the sequential number assigned to it in the appendix.

use of PRA in design E.1A.V-1 E.1A.V-2 modeling of a PRA (1.6) E.1A.V-3 shutdown and low-power events (1.6) E.1A.V-4 external events (1.6, 3.3, and 6.1) E.1A.V-5 core damage frequency (1.7) uncertainty treatment (1.9 and 6.1) E.1A.V-6 E.1A.V-7 documentation of method of truncation of accident sequences (1.10 and 2.5) E.1A.V-8 low-frequency accident initiators leading to core damage (2.2) E.1A.V-9 mission time (2.10) E.1A.V-10 failure rate for components (2.11) E.IA.V-11 tornadoes and extreme winds (3.2) E.1A.V-12 external river flooding (3.2) E.1A.V-13 hurricanes and storm surges (3.2) E.1A.V-14 tsunami (3.2) E.1A.V-15 internal fires (3.2) E.1A.V-16 site-specific external events E.1A.V-17 internal flooding (3.2) E.1A.V-18 seismic hazards analysis (3.3) E.1A.V-19 core-damage-sequence binning (4.1) E.1A.V-20 plant damage state definition (4.2)

EPRI Evolutionary Plant SER 1A.1-5

E.1A.V-21 containment isolation assumptions and criteria (4.3) E.1A.V-22 in-plant sequence assessment (4.5) E.1A.V-23 containment event analysis (4.6) E.1A.V-24 details of uncertainty analysis (4.6) E.1A.V-25 source term definition (4.7) E.1A.V-26 event tree binning (4.8) E.1A.V-27 risk measures related to containment performance (4.8) E.1A.V-28 use of mean values for characterization of risk results (5.1) E.1A.V-29 assessment of risk measures (5.2) E.1A.V-30 calculating of offsite consequences (5.2) E.1A.V-31 importance analysis for input to reliability assurance program (6.1) E.1A.V-32 assessment of containment response (6.2) E.1A.V-33 source term (6.3) E.1A.V-34 scope and objective of human reliability analysis HRA (7.1) E.1A.V-35 process and criteria to confirm adequacy of human reliability analysis (HRA) (7.2) E.1A.V-36 impact of advanced technologies on HRA (7.3) E.1A.V-37 function, task, timeline, and link analyses (7.3) E.1A.V-38 generic data sources (7.3) E.1A.V-39 performance shaping factors and their evaluation tools (7.3) E.1A.V-40 guantification methods for HRA (7.3) E.1A.V-4: luss of offsite power frequency (Annex A) E.1A.V-42 site data (Annex B)

# 1.6 Overall Scope and Methods of Appendix A to Chapter 1

Appendix A to Chapter 1 of the Evolutionary Pequirements Document defines the ALWR Utility Steering Committee's requirements for the PRA to be performed for evolutionary ALWRs. It defines the purpose and scope of the PRA, identifies previously developed methods to be used, identifies new or improved methods to be used, and defines procedures to be used if existing cocedures are incomplete or conflicting. Such a PRA is required to be performed to satisfy the regulatory requirements in 10 CFR Part 52. The staff concludes that the overall scope and methods described in Section 1.1 of Appendix A to Chapter 1 are acceptable, except as described below.

### Scope of Structures, Systems, and Components Modeled in PRA

It is unclear whether the Evolutionary Requirements Document requires the PRA modeler to include all important plant equipment. As clarification, the staff requires that ALWR vendors submit a PRA that models all equipment (1) that the NRC requires evolutionary plants to have available to prevent or mitigate severe accidents and (2) that an operator is likely to attempt to use to prevent or mitigate any initiating event that may lead to core damage. This must include consideration of accident management measures or strategies for preventing and mitigating severe accidents and for establishing and maintain-ing long-term cooling and containment heat removal.

### Modes of Operation Other Than Full Power

Revision 1 of Appendix A to Chapter 1 called for (at a minimum) a qualitative screening evaluation to be performed for events during modes other than full

power. In the DSER for Appendix A to Chapter 1, the staff concluded that a qualitative evaluation alone was not adequate and identified this as an open issue.

In its January 24, 1992, response to the DSER (which referenced EPRI's letter dated December 16, 1991, responding to the staff's request for additional information dated September 5, 1991), EPRI modified the Evolutionary Requirements Document to address shutdown risk in more detail. EPRI stated that the Evolutionary Requirements Document will evaluate "all operating conditions (including shutdown ones)." In addition, it will require that the plantspecific PRA performed for shutdown conditions evaluate core damage frequency (only Level 1 PRA required) with a simplified evaluation of release frequencies and magnitudes. The study will attempt to identify potential vulnerabilities so that evolutionary plant designers can remedy the most important ones. The staff concludes that the snutdown analysis guidelines proposed by EPRI are well thought out and generally acceptable with the following exceptions:

- Although it is necessary to develop assumptions that limit the scope of a shutdown PRA in order to make the analysis tractable, in reality it is impossible to predict all combinations and permutations of maintenance activities that might occur during outages. However, on the basis of inspections performed at operating plants, it is apparent that the risks associated with planned outages can be significantly reduced when plant personnel are aware of the risk potential of certain configurations. Use of the simplified shutdown PRA can help outage planners eliminate configurations that are particularly risk significant, sensitize shutdown planners to the need to minimize the time of exposure for risksignificant configurations that cannot be avoided, and provide the planners with safety insights that can be shared with those performing plant maintenance. Therefore, at each planned outage, combined license applicants should reevaluate the expected shutdown configurations using the simplified shutdown PRA to belt in configuration control, maintenance, and outage scheduling.
- Potential initiating events should not be excluded from consideration on the basis that safety functions are available and the window of risk is of short duration. For example, these criteria would screen out midloop operation if taken literally, even though mid-loop operation has been shown to be a potentially important contributor to shutdown risk.
- When determining the list of shutdown initiators to be considered, the analysts must consider existing (NRC) information notices that pertain to shutdown events and must consider internal flooding and fires.
- When making a functional assessment ine plant's response to a shutdown event, not only the effectiveness and types of alarms expected to be actuated should be analyzed, but also instrumentation pertaining to such things as vessel level and reactor coolant system temperature and all instrumentation specific to decay heat removal system performance (e.g., pump amperage). Quantitative analysis of the operator's ability to respond to a shutdown initiator requires an evaluation of the instrumentation available.

Consideration must be given to including insights from the shutdown PRA analysis in the accident management program guidance the vendors will develop for combined license applicants and in the technical specifications for modes other than full power.

Therefore, the staff concludes that the issue involving events other than those initiated during full-power operation is closed.

## External Events

At the FDA/DC stage, ALWR vendors must address seismic events, tornadoes, fires, and internal flooding. At the combined license stage, site-specific external events such as river flooding must be addressed. If a bounding analysis has not been performed by the ALWK vendor, combined license applicants must provide a site-specific analysis, using PRA techniques, that evaluates these external events. From the staff's viewpoint, the importance of these analyses comes not from bettom-line numbers, but rather from insights into plact design robustness, potential severe-accident vulnerabilities, and areas in which reliability and/or maintenance are particularly safety significant. In Section 3 of this appendix, the staff provides additional details regarding external events.

#### 1.7 Definition of Core Damage

Section 1.2 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that core damage from events involving loss of coolant inventory and/or loss of core heat removal will be assumed to have occurred f, and only if, both of the following have occurred:

- The collapsed level in the reactor has decreased so that active fuel in the core has been uncovered.
- A fuel cladding temperature of 2200 °F or higher is reached in any node of the core as defined in a best-estimate thermal-hydraulic calculation.

In its DSER for Appendix A to Chapter 1, the staff stated that EPRI's characterization of core damage was incomplete in that reactivity accidents were not considered and identified this as an open issue. In response to the open issue, EPRI added a criterion for defining core damage resulting from reactivity accidents. For reactivity accidents, core damage is assumed to occur if the axial peak, radial average energy exceeds 280 cal/g. The staff concludes that the definitions of core damage due to loss of coolant inventory and or loss of core heat removal and due to reactivity accidents are acceptable for purposes of the PRA; therefore, this DSER open issue is closed.

Section 1.2.2 of the original version of Appendix A to Chapter 1 stated that the plant design will be such that a realistic assessment of the mean core damage frequency will produce a best estimate no higher than 1E-5 event per reactor-year, including both internal and external events. In the DSER for Appendix A to Chapter 1, the staff noted that this requirement was inconsistent with the requirements of Chapter 1 that refer to a mean annual core damage frequency and identified this discrepancy as an open issue. In response to the DSER, EPRI modified Section 1.2.3 of Appendix A to require that the plant design be such that a realistic assessment of the core damage frequency will produce an estimate no higher than 1E-5 event per reactor-year

(including both internal and external events). The wording proposed here is still unclear, since there is no rigorous statistical definition of an "estimate" or "annual core damage frequency." These could be represented by the mean, median, mode, or some other parameter. In addition, point estimates do not permit investigation of the effect that uncertainties may have on the insights from a risk assessment.

Although the staff concludes that the goal of 1E-5 event per reactor-year is useful for designers and for potential customers of evolutionary plant designs, the Commission has stated that it does not intend to regulate advanced plants to a specific numerical core damage frequency goal. Although many measures of central tendency exist, the Commission has chosen "mean values" as the appropriate measure for safety goals and other Commission and staff risk targets. Therefore, when reporting core damage frequency and other risk results, "mean values" should be reported to the extent possible. In Section 1.9 of this appendix, the staff discusses further uncertainty analyses. A!WR vendors and combined license applicants may submit other measures of central tendency or statistical estimates, but the staff will use mean values (with their corresponding uncertainty distributions) for gaining insights and making decisions. Therefore, this DSER open issue is closed. The staff's further evaluation of EPRI's public safety goal is provided in Section 5.1 of this appendix.

### 1.8 Point-Estimate Quantification

Section 1.3 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that for each primary event input into the PRA model, a point estimate will be derived to represent that event in calculating the frequency of event sequences. Section 1.3 also states the mean value will be the point estimate used for this purpose. These mean values will be propagated through the PRA models, and point-estimate frequencies will be obtained for coredamage sequences and radionuclide release categories of interest.

### 1.9 Uncertainty Treatment

# Internal Event Uncertainties

Revision 1 of Section 1.4 of Appendix A to Chapter 1 of the Evolutionary Requirements Document stated that a qualitative uncertainty analysis will be performed as part of the PRA, and that this analysis will, as a minimum, involve the identification and description of the potentially important sources of uncertainty, and an assessment of the significance of these uncertainties with respect to the results and conclusions of the PRA. The document also stated that where necessary, the qualitative analysis will be supplemented by quantitative evaluations of the sensitivity of the results to key uncertainty issues to aid in investigating the significance of these sources of uncertainty. However, it did not distinguish between the uncertainty and sensitivity analyses that would be performed for the Level 1 versus the Level 2 and 3 portions of the analysis, or for internally versus externally initiated events.

In the DSER for Appendix A to Chapter 1, the staff concluded that uncertainty analyses should be such that the staff has reasonable assurance that the PRA reflects variability in (1) the significance of key actions, events, and

phenomena for the plant design; (2) the effectiveness of the accident mitigation systems and potential design improvements; and (3) the estimates of risk measures related to public health and safety. This was identified as an open issue in the DSER.

In response to this DSER open issue, EPRI proposed modifications to Section 1.4 of appendix A to Chapter 1 that provide further guidance on how uncertainty is to be considered in the PRA and that more clearly require that the qualitative uncertainty analysis be supplemented by a series of quantitative sensitivity studies. The revision also emphasizes the importance of analysts giving careful, systematic consideration to the sources of uncertainty that could be important, and the impact that each of these sources might have on the results.

The EPRI response represents an improvement in the Evolutionary Requirements Document but falls short of adequately addressing the full range of the staff's concerns. In particular, Appendix A still does not require a quantitative assessment of uncertainty or provide guidance on how uncertainties should be addressed for the different portions of the analysis (Levels 1, 2, and 3) and for internal versus external events. The staff recognizes the difficulty in performing uncertainty analyses and, in some areas such as Level 2, the limitations of such analyses. Nevertheless, given the inadequate state of knowledge of severe accidents and the considerable benefits of performing an uncertainty analysis (particularly for Level 1), the staff considers the present guidance on uncertainty analyses in the Evolutionary Requirements Document to be insufficient.

The staff requires that a full uncertainty analysis be performed for the Level 1 portion of the PRA, with uncertainties propagated from basic events, including initiating event frequencies, data, common cause/mode failure, success criteria, and human errors. The Level 1 uncertainty analysis is to be performed in such a manner that the most important uncertainties that contribute to core damage frequency are determined. Uncertainty analyses should be such that the staff has reasonable assurance that the PRA reflects variability in (1) the significance of key actions, events, assumptions (such as those related to success criteria), and phenomena for the plant design and (2) the effectiveness of the accident prevention systems and potential design improvements.

With regard to Level 2 uncertainties, the staff agrees with the rationale portion of Section 1.4 of Appendix A to Chapter 1 that many of the most important sources of uncertainty do not readily lend themselves to meaningful quantitative treatment and may be similar to what have been labeled "modeling" uncertainties. Furthermore, the staff recognizes that the evolutionary plant designs will incorporate features intended to minimize or eliminate the challenges posed by certain severe-accident phenomena, and that such features will reduce the significance of uncertainties in related severe-accident phenomena by reducing the frequency of occurrence and/or magnitude of the challenge. Examples of such features include reactor depressurization systems to reduce the frequency of reactor vessel failure at high pressure, inerting of the containment to minimize the potential for hydrogen combustion, and incorporation of reactor cavity flooding systems to minimize concerns regarding core-concrete interactions. Accordingly, a full, quantitative uncertainty analysis will not be required for the Level 2 portion of the PRA. Rather, the staff requires (1) the implementation of a systematic process for identifying

issues and phenomena of greatest risk significance for the advanced designs and (2) a more thorough treatment of those issues and phenomena and their associated uncertainties as part of the Level 2 analysis, for example, in the containment event trees (CETs).

The objectives of the Level 2 uncertainty treatment are to acknowledge and represent, within the context of the containment analysis, the full range of outcomes for those issues that ar highly uncertain. This is in contrast to a more simplified approach where uncertainty issues (perhaps bimodal in nature) are represented in the CET by a single, "best-estimate" outcome. It should be noted that this treatment of uncertainty is distinctly different from the type of uncertainty analysis described in NUREG-1150 ("Severe Accident Rates, An Assessment for Five U.S. Nuclear Power Plants") in which probability distributions are developed for key branch points and propagated through the CET to obtain statistical information such as 5 and 95 percent confidence limits.

If an ALWR vendor does not choose to perform a NUREG-1150-type Level 2 uncertainty analysis, the vendor's process for addressing Level 2 uncertainties should contain the following three major elements: (1) initial screening of issues for applicability to the ALWR design. (2) sensitivity analyses to further delineate issues of greatest risk significance, and (3) systematic analysis of issue uncertainty as part of the Level 2 analysis. The staff considers the EPRI guidance, which calls for a qualitative uncertainty analysis and supplementary quantitative sensitivity analyses, to be an acceptable starting point for this analysis. However, the evaluation called for by the Evolutionary Requirements Document must be structured in such a way as to identify and develop insights on key issues and phenomena, and then augmented by a more de' iled treatment of the risk-significant issues in the quantitative analysis.

The staff will address the acceptability of the Level 2 uncertainty treatment as part of its review of an application for FDA/DC. The adequacy of the analysis will be judged on the basis of the completeness of issues considered, reasonableness of parameter ranges and screening criteria for sensitivity analyses, and the methodology for decomposing and propagating the range of potential issue outcomes in the analysis. The treatment of uncertainties for Level 2 must provide the staff reasonable assurance that the PRA reflects the significance of key actions, events, and phenomena for the plant design, as well as the effectiveness of the accident mitigation features.

Therefore, the staff concludes that the DSER open issue concerning uncertainty treatment has been resolved. It discusses this issue in greater detail in Sections 4, 5, 6, and 7 of this appendix.

### External Event Uncertainties

External event uncertainty analysis requirements are discussed by the staff in Section 6.1 of this appendix. Either a limited sensitivity or a limited uncertainty analysis will be performed for all external events analyzed using PRA techniques. The staff does not expect ALWR vendors or combined license applicants to perform detailed and complete uncertainty analyses for external events.

# 1.10 Form of Results

In the DSER for Appendix A to Chapter 1, the staff identified an open issue concerning documentation of a PRA. The staff concluded that the documentation of the PRA must be adequate for its review. The Evolutionary Requirements Document should provide an analyst with guidance that ensures that the PRA documentation provides a traceable path and that enough information is presented so that a competent PRA analysis team can duplicate the results. The staff stated that it did not expect that a review would duplicate the analysis in all details. Review of the PRA could be facilitated by including, in addition to the written material, material on magnetic media, such as the fault trees, event trees, data, and dominant cutsets. In response to the DSER open issue, EPRI's guidance in Section 1.5 of Appendix A on presenting the results of the PRA has been significantly expanded, including models and quantification. The staff concludes that the guidance in the revised Evolutionary Requirements Document is adequate. Therefore, this open issue is closed. However, in addition to complying with the expanded guidance on documentation in Section 1.5, vendors must submit a description of the method of truncation used in the guantification process.

#### 2 PLANT MODELING

## 2.1 Model Structure

Section 2.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that the plant will be modeled in terms of a set of initiating events, event consequences composed of function or system success or failure states, and logic models that describe combinations of basic events that define the possible success and failure states. Containment functions whose failure could influence whether or not core damage occurred and/or the frequency of core damage will be included in the event trees. Each end state of each event tree will be designated either "success" or "core damage." The core-damage sequences, when combined with success or failure of systems needed to preserve containment integrity, will be categorized and grouped into plantdamage states for downstream modeling of the containment processes. The staff finds the proposed modeling structure to be adequate.

## 2.2 Initiating Events

Section 2.2 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that the analyst will develop a comprehensive list of potential initiating events for consideration in the PRA. EPRI states that the analyst should consider input from summaries of operating experience for current-generation plants, PRAs for plants with similar design characteristics, and review of the system designs. In the DSER for Appendix A to Chapter 1, the staff stated that EPRI did not indicate that initiating events that degrade mitigating or core-damage-prevention systems should be comprehensively identified and realistically quantified. An example of such an initiating event is failure of a dc bus, which may result in a reactor transient and, at the same time, fail one train of one or several safety systems. This was identified as an open issue in the DSER.

In response to the staff's DSER, EPRI modified Section 2.2 by providing guidance so that analysts will review each ALWR plant system to identify any failure that could both initiate a plant trip and degrade one or more systems that would otherwise be available to prevent core damage or mitigate consequences. The staff concludes that this revision is acceptable and this open issue is closed.

However, in addition, the claff will review applications for FDA/DC to ensure that ALWR vendors consider (1) low-frequency accident initiators (usually screened out in conventional PRAs) leading to core damage that could significantly challenge prevention and/or mitigation equipment and (2) low-frequency events with very high consequences. The search for these new initiating events must be conducted in such a way that events that are capable of affecting plant safety can be identified even at frequencies about a few percent of the core damage frequency. Since it is expected that evolutionary plant designs will systematically reduce the expected frequency of "traditional" initiating events that lead to core damage, it is important to look for initiators that have been ignored previously by analysts or were considered to have been bounded by "traditional" initiators. Two examples illustrate initiators not normally considered in PRAs but which may be important for ALWRs: (1) weltiple initiators (e.g., a loss of service water followed by an independent loss of offsite power before service water has been

EPRI Evolutionary Plant SER

recovered) and (2) low-frequency loss-of-coolant accidents (LOCAs) (e.g., multiple steam generator tube ruptures). The process used to make this search for initiators should be submitted.

# 2.3 <u>Success Criteria</u>

In Section 2.3 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI requires that a definition of success and failure for each function be provided based on realistic analysis of plant response. If conservative data are used to economize resources, the analysts are cautioned to identify conservative assumptions and to review the results to ensure that use of conservatisms does not obscure insights from the results. The analysts must also exercise care to ensure that any assumption or criterion considered to be conservative in one context does not introduce a nonconservatism in some other area (e.g., it is conservative to assume that containment pressure is high following a LOCA for containment analyses, but it is a nonconservative assumption for emergency core cooling system pump runout calculations). The staff agrees that these measures are prudent and believes the basis for these measures given by EPRI is valid.

### 2.4 Sequence Logical Identity

Section 2.4 of Appendix A to Chapter 1 of the Evolutionary Requireme Document states that the plant model and the solution and quantification techniques will retain the logical identity of the basic events that considtute each sequence. It is necessary to specifically identify which basicevent combinations contribute to the frequency of the dominant event sequences in order to understand and check the realism of the results. EPRI does not consider it sufficient to calculate sequence frequencies only and states that the specific equipment conditions must be known in order to determine whether recovery by the operations staff is possible and to judge how likely such recovery may be. This approach is logical and defensible. The NUREG-1150 studies, as well as some earlier studies, showed that recovery actions must be identified and quantified at the cutset level; this is in essential agreement with the EPRI approach.

# 2.5 Quantification

### Truncation

Section 2.5.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document discusses the truncation of accident sequence cutsets. The staff agrees with the statements in this section calling for retaining information regarding 'ow-frequency sequences. This may be important for identifying those sequences with a relatively higher potential for containment failure, as well as for preserving the ability to subsequently assess the effects of certain risk-sensitive areas. In the DSER for Appendix A to Chapter 1, however, the staff stated that the guidance did not provide assurance that a large number of cutsets, each of which is a small contributor but which in their aggregate form an appreciable contribution, would not be eliminated from further consideration. The staff further stated that there was no mention in Section 2.5.1 of whether the truncation would be performed before or after recovery actions (which are cutset dependent in many cases) are accounted for. In response to the staff's DSER, EPRI modified the requirements of Section 2.5 to provide assurance that large numbers of cutsets, each of which is a small contributor but which in their aggregate form an appreciable contribution, are not eliminated from further consideration. Section 2.5.1 was modified to state explicitly that truncation will be performed after recovery actions (which are cutset dependent in many cases) are accounted for. The staff concludes that this revision addresses its concern and is, therefore, acceptable. This open issue is closed.

#### Nested Solution Process

Section 2.5.2 of Appendix A to Chapter 1 discusses the nested solution process, which is a method of truncating certain cutsets of support systems before the support system is incorporated into the fault trees of systems depending on the support system. In the DSER for Appendix A to Chapter 1, the staff stated that the guidance given by EPRI was not sufficiently complete to determine whether the process was a source of significant error. The staff requires assurance that the truncation technique used does not result in underestimating common-mode failures between trains of the support system or between different systems. Also, the truncated cutsets must be demonstrated to be negligible, not only individually, but in their aggregate, even after recovery actions are taken into account on the dominant cutsets, to ensure that important insights are not lost.

In response to the staff's DSER, Ef'll modified the requirements of Section 2.5.2 to ensure that truncation limits at intermediate steps in the quantification process are selected to ensure that sufficient event combinations are retained, consistent with the truncation guidance in Section 2.5.1. This is acceptable to the staff; therefore, this open issue is closed.

In addition, the staff will review applications for FDA/DC to ensure that individual PRAs carefully document the method of truncation of the accident sequences, so that the method can be evaluated.

### 2.6 Modeling of Dependencies

Section 2.6 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that the potential for dependent failures be considered comprehensively and treated quantitatively using the best available methods. EPRI lists the following five types of dependencies that it requires be treated explicitly:

- sequence functional dependencies
- intersystem dependencies
- intercomponent dependencies
- dependencies due to human actions
- dependencies between core-cooling systems and containment systems

EPRI recognizes that dependencies have the potential to freat redundancy in the design and, therefore, deserve careful attention in FRA. Because of the greater degree of redundancy in the design requirements for an ALWR, it is particularly important to understand the potential effects of dependencies on an integrated level for the plant. The staff agrees that the proper modeling of dependencies is essential when conducting any PRA.

# 2.7 Interaction and Modeling of the Containment Systems

Section 2.7 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that the delineation of core-damage requences be coordinated with the assessment of containment response to ensure that any effects of containment phenomena on the availability of the systems needed to prevent core damage are appropriately reflected in the event trees. The staff agrees with EPRI that these steps are necessary in the performance of the PRA to avoid a possible source of erior.

### 2.8 Common-Cause Failures

Section 2.8 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that direct component-to-component and system-to-system functional dependencies be specifically addressed in the plant model. This section applies to root-caused events leading directly to multiple component outages from the shared cause. Common-cause in tiating events are explicitly addressed under external events and specific internal events in Sections 3 and 4 of Appendix A to Chapter 1 of the Evolutionary Requirements Document. The staff agrees that these dependencies must be included in the plant model if a realistic estimate of the core-damage sequence frequencies is to be obtained, and that neglect of these dependencies leads to an underestimate of the core damage frequency.

### 2.9 Human Interaction

Section 2.9 of Appendix A to Chapter 1 of the Evolutionary Requirements Document provides guidance on modeling human interactions in the PRA. The staff's evaluation of this aspect of the EPRI document is given in Section 7 of this appendix.

### 2.10 Mission Time

Section 2.10 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that a mission time of 24 hours will be used for equipment required to remain operable for successful core cooling after an initiating event. Recent PRA studies by the French indicate that long-term (up to 1 year) reliability of post-LOCA core cooling using recirculation from the containment sump can contribute appreciably to core melt frequency. On the basis of their findings that post-LOCA decay heat removal through emergency core cooling system operation can extend over a period of months, the French have identified system modifications and developed contingency procedures to improve the long-term reliability of recirculation cooling Although the risk significance of long-term core cooling may not be as great for ALWR designs in absolute terms, the French studies suggest that greater attention should be paid in the PRA to events outside the 24-hour time window traditionally assumed in PRA. In the DSER for Appendix A to Chapter 1, the staff concluded that mission times greater than 24 hours should be quantitatively considered for key systems required for long-term cooling. This should include consideration of accident management measures or strategies for establishing and maintaining long-term cooling. A quantitative bounding analysis may be adequate and may eliminate the need to consider in detail mission times greater than 24 hours for transients. This was identified as an open issue in the DSER.

Although EPRI made minor modifications to Section 2.10 in response to the staff's DSER, the staff continues to consider the EPRI guidance on mission time to be inadequate. Therefore, on the basis of the considerations discussed above, as well as in Section 1.6 of this appendix, the staff concludes that ALWR vendors must include the following in their PRAs:

- (1) The scope of the PRAs performed for ALWR designs must be extended to include treatment of the plant evolutions and system functions necessary to bring the reactor to cold shutdown and to maintain this condition for the long term.
- (2) Mission times are to be determined and justified by the ALWR plant vendor consistent with this expanded scope. This will necessitate consideration of mission times considerably longer than 24 hours, as well as explicit treatment of actions by onsite operating staff and offsite support organizations (i.e., all recovery actions) which may need to be accomplished within this timeframe.

Determination of an appropriate mission time should be based on consideration of system performance and reliability late in an event, availability and reliability of available backup systems and components, actions required to be taken by operating staff and offsite response organizations, and provisions that would be in place to ensure such actions could be taken in a timely manner. Mission times and success criteria will have to be justified by the plar designer and will be evaluated by the staff when it reviews the ALWR design-specific PRA as part of its review of an application for FDA/DC. This DSER open issue is closed.

# 2.11 Reliability Data

Section 2.11 of Appendix A to Chapter 1 of the Evolutionary Requirements Document discusses reliability data and suggests the use of the data base given in Annex A of Appendix A. In the DSER for Appendix A to Chapter 1, the staff concluded that Annex A of Appendix A to Chapter 1 should be modified to state that the analyst will have to review the failure data for applicability to each particular design, considering the operational environment of the equipment and the procedures applicable to the equipment. This applies to all the reliability data, including common-cause-failure parameters. This was identified as an open issue.

In response to the staff's DSER, EPRI modified Annex A of Appendix A of Chapter 1 as stated above. The staff finds this acceptable; therefore, this DSER open issue is closed.

In the DSER, the staff also stated that Annex A of Appendix A should be expanded to include data for systems anticipated in the ALWRs such as multiplexors, microprocessors, and cathode ray tube displays and identified this as an open issue.

In response to the staff's DSER, EPRI stated in a letter dated January 24, 1992, that generic data for the types of components cited were not widely available. In particular, failure rates for microprocessors varied substantially, depending on the complexity of the device and the specific application. In these cases, EPRI proposed to leave it to the analyst to identify the most applicable source of data. The staff concludes that EPRI's response

is acceptable and this open issue is closed. The staff will review FDA/DC applications to ensure that vendors provide detailed failure rate data for components that perform the essential functions of instrumentation and control systems.

At the combined license stage, the staff will review the site-specific lossof-offsite-power frequency and estimates of the site-specific probability of recovery of offsite power as a function of time after its loss. Loss-ofoffsite-power frequency and failure rates of high-voltage transformers are discussed by the staff in Annex A of this appendix.

The staff requires that test and maintenance intervals be considered when estimating the reliability of equipment. However, it may not be possible to test the full or integrated functions of some evolutionary design features. Some functions will never be tested in the as-built systems for reasons such as that their testing would result in the emptying of primary coolant into the containment. In some cases, safety-significant equipment will have long test intervals. As an example, since failures of motor-operated valves (MOVs) are well represented by a standby stress model, any MOV that has a long test interval is expected to have a high estimated failure probability. The testability of individual components is therefore an issue in addition to the testability of the whole function. On the other hand, testing at power of some components may lead to significant challenges to safety systems if additional random failures or human errors should occur.

Reliability data for non-safety-grade (but normally safety-grade in traditional PWRs and BWRs) components must be justified on the basis of the quality of the equipment purchased, test intervals, capability to perform its intended function in an adverse environment, experimental data, and applicable technical specifications.

# 3 EXTERNAL EVENIS

# 3.1 Identification of Initiating Events

Section 3.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that the list of potential external initiating events taken from American National Standards Institute/American Nuclear Society (ANSI/ANS) 2.12, "American Nuclear Society Guidelines for Combining Natural and Man-Made Hazards at Power Reactor Sites," be considered for an ALWR PRA. EPRI has selected a single source for methodology to ensure consistent treatment of external events in PRAs.

#### 3.2 Events That May Be Excluded Based on Qualitative Evaluation

EPRI has proposed that the following external events be excluded from a quantitative PRA analysis.

### Tornadoes and Extreme Winds

In Section 3.2.3 of Appendix A to Chapier 1 of the Evolutionary Requirements Document, EPRI states that, except for loss of offsite power, tornadoes and extreme winds may be excluded on the basis of a qualitative evaluation. The staff disagreed and identified this issue as open in its DSER for Appendix A to Chapter 1. In response to the staff's DSER, EPRI modified its guidance for evaluating tornadoes. The new guidance in Section 3.3.1 of Appendix A to Chapter 1 requires ALWR vendors to provide a simplified quantitative assessment of tornadoes. This assessment is to take into account the potential for a tornado to cause an extended loss (at least 24 hours) of non-safety-grade systems that could play a role in core cooling. In addition, the analysis will consider independent random failures of safety equipment if the failure on demand is 1E-3 or greater. The staff concludes that EPRI's proposed approach is acceptable and this DSER open issue is closed. The staff will review the site-specific hazard data at the combined stage. As this issue is addressed in the staff's February 27, 1992, draft policy paper, the staff's final position is contingent on approval by the Commission.

# External River Flooding

Section 3.2.17 of Appendix A to Chapter 1 states that external river flooding may be excluded on the basis of a qualitative evaluation. The staff disagreed and identified this issue as open in its DSER for Appendix A to Chapter 1. Subsequent submittals by EPRI were not sufficient to resolve the staff's concern. It is recognized that there are large uncertainties in the estimate of the river flood hazard, whether one is attempting to extrapolate from streamflow records or use runoff modeling methods. The National Research Council, in its publication, "Estimated Probabilities of Extreme Winds," dated 1988, states that "any method for estimating probabilities of floods greater than about the 100-year flood must include some form of extrapolation, a process that can, at best, introduce errors and, at worst, strain credulity." Therefore, the staff will review applications for a combined license to ensure that, at a minimum, the potential for large river floods is addressed explicitly in a bounding risk assessment. The staff will also accept a realistic assessment, using probabilistic techniques, of the chance of river flooding r using core damage at a particular site. Therefore, this DSER open issue is

closed. As this issue is addressed in the staff's February 27, 1992, draft policy paper, the staff's final position is contingent or approval by the Commission.

### Hurricanes and Storm Surges

At some coastal sites, it is possible that very large hurricanes could generate winds and/or storm surges that would damage sufficient plant equipment to lead to core damage. As a minimum, the potential for hurricanes and storm surges causing core damage at coastal sites will have to be analyzed in a bounding risk assessment. It is also acceptable to submit a realistic probabilistic risk analysis of hurricanes and storm surges. The analysis submitted will be evaluated by the staff at the combined license stage.

#### Tsunami

For some coastal sites, it is possible that tsunamis could generate waves of sufficient height that they would damage sufficient plant equipment to lead to core damage. As a minimum, the potential for tsunamis causing core damage at coastal sites must be analyzed in a bounding risk assessment. It is also acceptable to submit a realistic probabilistic risk analysis of tsunamis. The analysis submitted will be evaluated by the staff at the combined license stage.

# Internal Fire

Section 3.2.19 of Appendix A to Chapter 1 states that internal fires may be excluded on the basis of a qualitative evaluation. The staff disagreed and identified this issue as open in its DSER for Appendix A to Chapter 1. In response to the DSER, EPRI revised the Evolutionary Requirements Document to state that except for the main control room, a qualitative evaluation of severe-accident internal fires can exclude areas from consideration on the basis of a qualitative argument. The level of detail of a quantitative analysis of fires in the main control room would be a function of the estimated risk, based on first performing at least a bounding analysis. The staff concludes that the EPRI-proposed requirements in Sections 3.2.19 and 3.3.3 are still unacceptable. Therefore, it will review applications for FDA/DC to ensure that ALWR vendors perform either (1) a severe-accident fire analysis using the Fire- Induced Vulnerability Experiment methodology or (2) a fire PRA. These analyses are to be performed to gain insights into the adequacy of the design rather than as an exercise in determining the absolute value of estimated core damage frequency from internal fires. When performing the severe-accident fire analysis, ALWR vendors must address the following areas:

- Any fire area pinch point where multiple trains or divisions (including power or control cabling) are located in the same fire area must receive a quantitative assessment similar to that for the control room. In the fire analysis, the alternative shutdown capability must be analyzed and quantified from a human factors, systems function, and reliability standpoint.
- The fire analysis must evaluate whether safe shutdown can be achieved (and with what conditional probability) assuming that all equipment in any one fire area will be rendered inoperable by fire and that re-entry into the fire area for repairs and operator actions is not possible.

EPRI Evolutionary Plant SER 1A.3-2

Because of its physical configuration, the control room is excluded from this approach, provided (1) an independent alternative shutdown capability that is physically and electrically independent of the control room is included in the design and is reliably capable of shutting down the reactor and (2) the likelihood of hot shorts involving power and control wiring associated with the remote shutdown panel function is negligibly small.

- The fire analysis must evaluate the probability that given a fire, fire protection for redundant shutdown systems in the reactor containment building will ensure that at least one shutdown division will be free of fire damage.
- The fire analysis must evaluate the likelihood that smoke, hot gases, or fire suppressant will migrate into other fire areas to the extent that they could adversely affect safe shutdown capabilities, including operator actions.
- The underlying assumptions used in the fire analysis (especially those related to barriers between divisions) are to be documented and submitted to the NRC staff.
- The fire analysis must evaluate barrier penetration reliability, because barriers can be compromised or be deficient.
- In a fire analysis, spatial separation in accordance with the requirements of Appendix R to 10 CFR Part 50 will not be considered adequate to prevent damage to equipment in the same fire area.

The staff will confirm the validity of the assumptions in the fire portion of the PRA during its review of a specific application for FDA/DC and again when the plant is in its final stages of construction. Therefore, this DSER open issue is closed. As this issue is addressed in the staff's February 27, 1992, draft policy paper, the staff's final position is contingent on approval by the Commission.

## Other Site-Specific External Events

For a particular site, there may be specific external events that warrant severe-accident analysis such as accidents at nearby industrial or chemical plants, vulcanism, and transportation accidents. On a judicious basis at the combined license stage, applicants must address applicable site-specific external events either on the basis of a bounding risk analysis or a realistic probabilistic risk analysis.

### Internal Flooding

Section 3.2.20 of Appendix A to Chapter 1 states that internal flooding may be excluded on the basis of a qualitative evaluation. The staff concluded that this was unacceptable and identified this issue as open in its DSER for Appendix A to Chapter 1. Subsequent submittals by EPRI were not sufficient to resolve the staff's concern. Even if the ALWR is better designed to prevent the initiation and spread of internal flooding than present-day plants, internal flooding could still contribute an appreciable amount to the estimated core damage frequency. Therefore, the staff will review applications

for FDA/DC to ensure that ALWR vendors provide either a systematic PRA of internal flooding or some simplified risk assessment that is capable of identifying vulnerabilities to internal floods in the design. However, the staff is unaware of any acceptable simplified method for evaluating internal floods other than that performed in the Yankee Rowe individual plant examination. This DSER open issue is closed. As this issue is addressed in the staff's February 27, 1992, draft policy paper, the staff's final position is contingent on approval by the Commission.

# 3.3 Events That Will Require Quantitative Assessment for Each ALWR

#### Seismic Events

The seismic design certification is based on a free field vibratory ground motion characterized by a response spectrum with a peak ground acceleration of 0.3g as described in Regulatory Guide 1.60, "Design Response Spectra for Seismic Design of Nuclear Power Plants." At the combined license stage, the applicant is required to demonstrate that the site parameters are enveloped by the design certification values. If the site parameters exceed the design certification parameters, the combined license applicant will have to show sufficient seismic margin in design to meet the site-specific demand, in conformance with Section 2.5.2, "Vibratory Ground Motion," of the SRP.

Section 3.3.2 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that a seismic risk analysis must be performed as part of the PRA. It is the staff's position that a seismic PRA is not required.

The severe accident policy statement calls for the completion of a PRA and the consideration of severe-accident vulnerabilities exposed by the PRA in order to add to the assurance of public health and safety. The policy statement calls for the staff to review evolutionary designs for the purposes of concluding that the design is acceptable with regard to safety; it stresses the use of deterministic engineering analysis and judgment complemented by PRA. In addition, the policy statement calls for evolutionary plant vendors to use the PRA as a tool to consider a range of alternatives and combination of alternatives that address unresolved and generic issues and to search for cost-effective reductions in the risk from severe accidents.

In response to the policy statement, existing plant licensees were given the choice of using either a probabilistic risk or a margins methodology to search for potential vulnerabilities in the capability of the facility to deal with seismic events. A margins approach can provide a measure of the capability of the plant to withstand seismic events beyond the design basis or for some other specified external event.

Seismic PRAs and seismic margins assessments both have limitations and offer different insights about the capability of a plant to withstand severe accidents initiated by seismic events. PRAs can provide estimates of the frequency of severe accidents due to seismic events and can provide insights about the capability of the plant to survive seismic events. Two major advantages of PRA are that the response of all important systems is included in the analysis and failures in addition to seismic failures can be considered. The major disadvantage of a seismic PRA for a particular site is large uncertainties in the seismic hazard. The uncertainties would be even greater for a generic site. In light of these large uncertainties in seismic hazard,
the staff concludes that a seismic PRA for design certification should not be required. However, it believes insights about the capability of the plant to withstand a seismic event can be obtained from a margins analysis.

Seismic margins analyses typically calculate a high confidence/low probability of failure (HCLPF) for components or systems within the plant. This parameter represents an estimate of the seismic acceleration that the component could withstand and still perform its intended function. Past seismic margins analyses show that the HCLPF value for nuclear power plants is typically twice the design value. By determining the HCLPF for all the important components required to bring the plant to some stable state (typically hot 'hutdown for events initiated with the plant initially at power), the plant HCLFF, as well as the components or systems limiting the capability of a plant to survive a seismic event, can be determined. There are some limitations to a pure margins approach, some of which are magnified when there is a lack of design detail and no plant to physically examine. Limitations include the need to make assumptions about which systems and equipment would be required following a seismic event and the lack of accounting, in the methodology for non-seismic (i.e., random and human) failures. However, margins methods can be enhanced by using PRA insights to select success paths that likely would be used and to ensure that equipment reliability is considered in the analyses.

The staff intends to determine the adequacy (i.e., the robustness) of ALWR designs by using a modified margins analysis to identify seismic or other external event vulnerabilities. Insights about the seismic capability of the plant can be best obtained by merging PRA and margins approaches to take advantage of the strengths of each. This approach allows for a comprehensive and integrated treatment to understand the plant response to an earthquake. Plant logic models covering the various systems that could be used to prevent core damage are constructed typically by modifying internal events PRA models to include logic important to consideration of seismic failures. Neither site-specific nor generic seismic hazard curves are used as input for this approach. All significant operational sequences leading to safe shutdown (success paths) are identified using the event trees and fault trees based on fragility data for each component for each success path. The minimum HCLPF for each system determines the system and plant HCLPF. If the plant HCLPF is less than about twice the design ground motion zero period acceleration, a more detailed evaluation should be performed to determine if a vulnerability exists that should be strengthened. The HCLPF calculated in this manner provides a measure of the robustness of the plant in that it estimates with high confidence the earthquake ground motion the plant is expected to be ably to survive without core damage. HCLPF calculations also provide insights about which components and systems limit the plant seismic capability.

A seismic PRA rather than the PRA-based margins approach described above was the primary approach used to evaluate the advanced boiling water reactor (ABWR) seismic design. The staff compared the hazard estimates derived by Lawrence Livermore National Laboratory (LLNL) using the historical earthquake method (NUREG/CR-4885, "Seismic Hazard Characterization of the Eastern United States: Comparative Evaluation of the EPRI and LLNL Studies") and by EPRI with the results obtained using the ABWR bounding seismic hazard curve. The ABWR bounding hazard curve was exceeded by the LLNL mean hazard curves for the Pilgrim, Seabrook, and Watts Bar sites. These three sites in the Eastern United States were selected beca" 3 of their relatively high seismic hazard. The three LLNL curves were arithmetic mean hazard curves using input from five

ground motion experts. The staff used both the LLNL and EPRI seismic hazard estimates to quantify core damage frequency. The use of the LLNL hazard curves resulted in the prediction of much higher core damage frequencies than did the use of the EPRI hazard curve. However, the ABWR design was found to be capable of resisting earthquakes significantly larger than a safe shutdown earthquake of 0.3g based on an estimated plant HCLPF of about 0.6g using the seismic margins methodology.

Recent analyses of the seismic inputs to the LLNL hazard program suggest that uncertainties may be reduced in future LLNL results. As an example of reduced uncertainty in the LLNL estimates, the staff compared the LLNL hazard estimates of four (rather than five) ground motion experts. Results that included the estimates of the fifth ground motion expert were generally higher. Thus, using the estimates of only four ground motion experts might provide a means of showing the effect of reduced uncertainty. However, the LLNL results using the estimates of the four ground motion experts still exceeded the EPRI bounding curve. Thus, it is the staff's conclusion that the LLNL hazard results suggest the EPRI bounding curve for rock sites is not conservative.

In Section 3.3 of the DSER for Appendix A to Chapter 1, the staff identified quantitative uncertainty analysis for seismic events as an open issue. It also identified an open issue associated with the requirement in Section 3.3.2.4 to use a logarithmic standard deviation of 0.2 to reflect variability in the response spectrum. A seismic PRA is not required for seismic design certification. The staff will use the margins approach involving HCLPF values, as discussed above, in its review. Therefore, both of these DSER open issues are closed. However, when a specific site is selected for an evolu-tionary plant, the staff will review the combined license app: in to ensure that the plant-specific probabilistic seismic hazards anis in conformance with Section 2.5.2 of the SRP.

#### 4 CONTAINMENT ANALYSIS

Section 4 of Appendix A to Chapter 1 of the Evolutionary Requirements Document discusses the containment analysis. The staff's conclusions on how the guidance should be modified are given here. However, it will still be necessary for the staff to review the containment analysis portion of a PRA submitted for the ALWR designs in some detail during its review of an individual application for FDA/DC.

#### 4.1 Core-Damage-Sequence Binning

In Section 4.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI states that core-damage sequences are expected to be binned (grouped). If core-damage bins are used, they must be defined to that all sequences within a particular bin lead to similar effects with respect to containment sequence and source term phenomena. EPRI requires that the definition of bins provide a means to ensure that the delineation of coredamage sequences is discriminated sufficiently to afford the prover level of coordination with the containment analysis. This is intended to provide a means of managing the number of accident sequences and to provide an additional means of gaining information needed for the in-plant analysis. The binning of sequences is an acceptable procedure to limit the number or containment analyses performed. It is necessary that, as EPRI states, all sequences within a bin lead to similar effects with respect to containmen\* and source term phenomena. The staff will be evaluate sequence binning when it reviews an evolutionary design PRA as part of its review of an application for FDA/DC.

## 4.2 Containment System Analysis

Section 4.2 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that a containment systems analysis be developed to explicitly account for any common-mode failures between the core-damage-prevention systems and the containment systems. To reduce the number of deterministic analysis runs necessary to develop the containment event tree branch point probabilities, EPRI made a simplifying assumption. The frequency dominant accident sequence for each plant damage state will be used to define in-plant phenomenological analysis parameters for use in determining containment performance source terms. EPRI cautions that the plant damage states must be sufficiently and uniquely defined to ensure that they adequately reflect the characteristics important to the containment response and release magnitudes in order to avoid introducing uncertainties that could otherwise be avoided. This procedure is a generally accepted part of PRA methodology. The staff will evaluate the actual groupings of the accident sequences into plant damage states when it reviews an evolutionary design PRA as part of its review of an application for FDA/DC.

#### 4.3 Containment Isolation

Section 4.3 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that containment penetrations be accounted for in the evaluation of containment leak ge paths. EPRI requires that penetrations be analyzed for the following types of release pathways:

- Failure to isolate normally open lines or normally closed lines that might be open at the time of an accident (e.g., because of the mispositioning of valves)
- leakage through a penetration where no pathways should be present (leakage past the seat of a closed valve, through an electrical penetration, past the seal of a personnel airlock, etc.)

EPRI allows containment penetrations to be screened from the analysis if they can meet one of the following criteria:

- small conditional probability of leakage or of failure to be isolated (i.e., less than 1E-3 per event)
- low consequence (e.g., release that must take place through a line that will remain filled with water throughout the accident)
- closed loop inside or outside the containment
- small in size (e.g., instrumentation lines)

EPRI's rationale is that not all containment penetrations have the potential to be important rathways for releases from the containment and the use of screening criteria is appropriate to focus the PRA effort on the penetrations that are most likely to be important. These simplifications are acceptable, but the staff will evaluate the actual implementation of these assumptions when it reviews an evolutionary design PRA as part of its review of an application for FDA/DC.

#### 4.4 Containment Bypass

Section 4.4 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that containment bypass sequences be assessed and include all connections to the reactor coolant system. EPRI states that containment bypass sequences can result in significant releases from the containment and have the potential to be important risk contributors. Bypass sequences that have been identified as important in past PRAs include

- steam generator tube rupture (PWR only)
- residual heat removal isolation failure
- high-pressure coolant injection (BWR only)
- core spray (EWR only)
- feedwater and main steam (BWR only)
- suppression system bypass (BWR only)

In the NUREG-1150 studies for Surry and Sequoyah, bypass sequences dominate early fatality risk. The staff agrees that containment bypass sequences are important and, therefore, agrees with the EPRI requirement to assess such sequences.

#### 4.5 In-Plant Sequence Assessment

Section 4.5.2 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that the MAAP code will be the primary tool used to assess the thermal-hydraulic and other physical processes involved in the accident

EPRI Evolutionary Plant SER 1A.4-2

progression. While recognizing the value of MAAP as an integrated code for severe-accident analysis, the staff concludes that this requirement places an undue reliance on the MAAP code and fails to establish a requirement dealing with deficiencies of this code. More specifically, the staff concerned that for best-estimate calculations, the MAAP code (or other codes for that matter) will not adequately treat certain physically important severe-accident phenomena, and that the models and assumptions made in the MAAP code will not be generally accepted by the experts in the field. In the DSER for Appendix A to Chapter 1, the staff stated that the Evolutionary Requirements Document should require that best-estimate MAAP calculations be supplemented by additional analyses that take into account alternative outcomes of potentially risk-significant issues and phenomena recognized by the nuclear safety community as containing large uncertainties. This was identified as an open issue in the DSER.

In response to the staff's DSER, EPRI proposed a modification to the Evolutionary Requirements Document to further clarify, by example, the situations in which use of codes other than MAAP might be more appropriate. These examples are "to develop realistic success criteria for core cooling, or to investigate specific phenomena that are not addressed by MAAP or that may be subject to large uncertainties." This modification partially addresses the staff's concern regarding use of the MAAP code for accident analysis. However, the document still fails to recognize the limitations and deficienries of the MAAP code, such as those revealed through the recent industry/NRC MAAP 3.B familiarization effort.

In view of the remaining concerns regarding the use of codes for severeaccident analysis, the staff will review an application for FDA/DC against the codes most recently endorsed and approved by the NRC at that time, and will evaluate the acceptability of the plant designer's analyses on a case-by-case basis. This DSER open issue is closed.

The staff notes that the treatment of certain severe-accident phenomena within the scope of a PRA and the use of the MAAP code for this purpose is a related issue. This is discussed in Section 4.6 of Appendix A to Chapter 1 of the Evolutionary Requirements Document and in Sections 1.9 and 4.6 of this appendix.

#### 4.6 Containment Event Analysis

Section 4.6.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that the containment event tree (CET) will comprise the important phenomenological issues associated with containment loading and/or source term evolution. As supporting rationale indicates, the CET provides an excellent means to identify and quantify important phenomena. The document lists issues and phenomena that should be considered by the ALWR designer in the development of the CET. In the DSER for Appendix A to Chapter 1, the staff stated that the CET should be expanded so that the adequacy of accident mitigation systems and cf the estimation of risk measures related to public health and safety can be determined. This was identified as an open issue.

In a subsequent revision, EPRI added a notation that the issues to be considered are not limited to those listed. The staff concludes that the list of phenomena provides a reasonable characterization of the types of issues that should be addressed by a plant designer in the development of the CETs and

uncertainty analysis, but cautions that this set of phenomena may not necessarily be complete. For example, Section 6.6.2.6 of Chapter 5 of the Evolutionary Requirements Document states that a containment overpressure protection system (vent) system may be provided in ALWRs, yet no mention is made in Section 4.6.1 of Appendix A to Chapter 1 of the need to consider the impact of containment venting on containment success criteria or to treat containment venting in the CET. Therefore, the staff will evaluate how these issues were treated by the plant designer when it reviews an evolutionary design PRA as part of its review of an application for FDA/DC. This DSER open issue is closed.

Section 4.6.2 of Appendix A states that potentially important phenomena that arc not currently addressed in the MAAP code will also be addressed in the containment event chalysis and identifies five such phenomena in the rationale section. The staff agrees with this requirement, but noted in the DSER that the requirement should not apply only to phenomena that are not explicitly treated using the MAAP code. Rather, PRA analysis should consider al. potentially risk-significant phenomena for possible inclusion in the containment event analysis. In response to this concern, EPRI modified the wording in Section 4.6.2 of Appendix A so as to not recircit consideration to only those phenomena that are not currently addressed in the MAAP code. The staff concludes that the revised guidance is acceptable, but notes that the phenomena identified in this section are not complete; for example, core debris coolability should also be addressed, since MAAP models for this phenomenon are highly questionable. The staff discusses this matter further in Section 1.9 of this appendix.

Section 4.6.3 of Appendix A states that the CETs will be quantified using best-estimate point values. The staff indicated in the DSER that the guidance in this section was insufficient, given that risk estimates may be dominated by uncertainties in modeling and assumptions related to severe-accident progression and phenomenology.

In response to the staff concerns in the DSER, EPRI proposed modifications to Section 4.6.3 of Appendix A that clarify, to some extent, the development of best-estimate values. The revised guidance states that for phenomenological events that could be important to the potential for releases and are subject to large uncertainty, the best estimates should rerlect an appropriate assessment of expert opinion or other relevant input, in addition to the deterministic result obtained using MAAP. The document further states that the outcome of a deterministic code may not properly reflect the best estimate for some branch points, and that the deterministic outcome, together with other characterizations of the phenomena, should be used to arrive at a best estimate. An example provided in the document of one approach is "to develop a probability distribution for an event, reflecting the available inputs, with subjective weights assigned to each input. The mean value of such a distribution would then be used as the best estimate in the quantification process."

The staff considers that the proposed modifications partially address its concerns regarding Level 2 uncertainty analysis, but fall short in that they (1) fail to emphasize the importance of reflecting the range of possible issue outcomes in the containment analysis, particularly when an outcome may be bimodal in nature, such as debris coolability, and (2) fail to acknowledge that it may be appropriate to perform a full, explicit treatment of uncertainties in selected Level 2 issues and phenomena. (This might be done as a limited extension of the Level 1 uncertainty analysis.) The staff discusses this matter in greater detail in Section 1.9 of this appendix.

Consideration of a range of possible uncertainties is especially important in the design phase to assess the adequacy of proposed severe-accident mitigation features and to minimize the potential impact of particular uncertainties through design modifications and/or accident management strategies. Such consideration is also important when the uncertainty range can envelop important measures of containment performance, such as the probability of containment failure given the (postulated) occurrence of certain poorly understood phenomena. The staff concludes that this uncertainty analysis must be performed in sufficient detail so as to provide reasonable assurance that potential design improvements and/or accident management strategies to deal with key uncertainties can be identified and their effectiveness estimated.

The staff's recent work discussed in NUREG-1150 represents the most extensive uncertainty analysis performed to date. However, the staff does not require a classical statistical analysis of this magnitude and believes that the safety-significant insights from a Level 2 analysis can be gained with approaches that require fower resources than that of the NUREG-1150 approach. Toward this end, the staff supports use of NUREG-1150 data, modified where necessary, in vendor uncertainty analyses to reduce the number of parameters included in uncertainty analysis and/or to provide probability distributions, modified as recessary, for individual parameters. This was identified as an open issue in the staff's DSER for Appendix A to Chapter 1. Subsequent revisions of Appendix A by EPPI have not provided adequate guidance to vendors that addresses the staff's concerns. Therefore, the staff will review an individual application for FDA/DC as described above. This DSER open issue is closed.

# 4.7 Source Term Definition

Section 4.7 of Appendix A to Chapter 1 of the Evolutionary Requirements Document requires that the analyst use the most current version of the MAAP code available at the time for source term calculations. However, alternative codes may be used if justification is provided. In any case, EPRI states that an integrated model of the core melt and the containment is required to address generation, effects of steam inerting, containment geometry, and containment pressurization. The staff agrees with the use of integrated models, but has a number of technical concerns regarding the use of the MAAP code. (The staff's concerns regarding the use of the MAAP code are discussed in Section 4.5 of this appendix.) Accordingly, the staff will address source torm definition during its review of an individual application for FDA/DC.

#### 4.8 Plant Release Categories

Section 4.8 of Appendix A to Chapter 1 of the Evolutionary Requirements Document states that similar end points of the containment analysis may be grouped into release categories for use in the ex-plant consequence analysis. EPRI states that past PRAs have shown that CET end points may be grouped to simplify the analysis and reduce the number of ex-plant runs required. Elements to be considered in .... grouping process include

- time of release
- duration of release
- energy of release
- types and amounts of isotope fractions released

The staff agrees that CET end points may be grouped. The type of grouping suggested by EPRI is similar to those used in past PRAs and is acceptable to the staff. The staff intends to review the details of the binning process as part of its review of an individual application for FDA/DC to confirm the process has been properly implemented.

Although the binning of CET end states is acceptable, the staff notes that Appendix A fails to provide guidance to ALWR designers regarding the use of the CET end state/release category information to develop risk measures related to containment performance, such as conditional containment failure probability and frequency of large releases. The staff will require ALWR designers to provide this type of information for their designs and will address this aspect of the aralysis when it reviews an evolutionary ALWR PRA as part of its review of an application for FDA/DC. The staff's evaluation of EPRI's requirements regarding containment performance is provided in Chapter 5 of this report.

#### 5 OFFSITE CONSEQUENCES

Section 5 of Appendix A to Chapter 1 of the Evolutionary Requirements Document discusses the assumptions for the offsite consequence analysis portion of the PRA.

# 5.1 Implementation of the Public-Safety Requirement

In Section 5.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI requires that a complementary cumulative distribution function (CCDF) be developed for whole-body dose for a 1/2-mile radius. This must include all core-damage sequences with a frequency greater that 1E-8 per year from both internal and external initiators. EPRI proposes that the design will be considered to have met the EPRI risk goals if this CCDF falls outside the region bounded by a lower limit for frequency at 1E-6 per year and by a lower limit for the consequences of a 25-rem whole-body dose at i/2 mile. The staff agrees that the CCDF is a well-accepted method of visually displaying risk curves. However, it displays only a point-estimate CCDF intended to approximate the mean CCDF and provides an incomplete picture of risk. In the DSER for Appendix A to Chapter 1, the staff stated that uncertainties must be propagated through the analysis, and a family of CCDFs must be displayed, each associated with their confidence limit, in addition to the mean CCDF. This was identified as an open issue.

The staff concludes that uncertainties must be propagated through the analysis, as discussed in Section 1.9 of this appendix, and a family of CCDF curves developed, each with their degree of belief (in addition to the mean CCDF curve). This information will be evaluated by the staff as part of the design certification review of each evolutionary ALWR PRA.

Although the Commission finds the goal of 1E-5 per year useful for designers and for potential customers of ALWR plant designs, it has stated that it does not intend to regulate advanced plants to a specific numerical core damage frequency goal. Although many measures of central tendency exist, the Commission has chosen "mean values" as the appropriate measure for safety goals and other Commission and staff risk targets. Therefore, when reporting core damage frequency and other risk measures, the staff concludes that "mean values" must be reported to the extent possi'.'e. Therefore, the staff will review an individual application for FDA/DC using the guidance stated above. This DSER open issue is closed.

# 5.2 Method for Offsite Consequence Analysis

In Section 5.2.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI states that a reference site with the characteristics listed in Annex B will be used for calculating offsite consequences for the ALWR. Annex B contains meteorological data developed by EPRI to be representative of a bounding (80th percentile) site. This calculation of offsite consequences would be performed by each ALWR vendor to demonstrate that the plant design meets EPRI's goal for dose at the site boundary. The staff concludes that the use of a standardized set of meteorological site data is reasonable for the limited purpose of demonstrating that EPRI's de\_gn goal for dose at the site boundary has been met. The staff addresses the adequacy of the specific reference site data proposed by EPRI in Annex B of this appendix.

As part of FDA/DC for each evolutionary ALWR, the staff will require each ALWR vendor to provide an assessment of additional ris's measures (such as personrem and early and latent fatalities) to support the vendor's assessment of severe-accident mitigation alternatives for the ALWR design. Meteorological data alone are insufficient to calculate these additional risk measures and will need to be supplemented with bounding population data, such as those provided in Regulatory Guide 4.7, "General Site Suitability Criteria for Nuclear Power Stations."

Section 5.2.2 of Appendix A to Chapter 1 states that either MACCS or CRAC2, or another suitable code, will be used for calculating offsice consequences. Although the CRAC2 code provides an acceptable characterization of the consequences of severe accidents, the MACCS code represents an improvement over CRAC2 and is preferred by the staff fur cancer risk calculations. The present version of MACCS (Version 1.5) uses the results of the BEIR III study by the Committee on the Biological Effects of Ionizing Radiation, "The Effect on Population of Exposure to Low Levels of Ionizing Radiation," July 1980, in the calculation of health effects. However, the BLIR III study has been superseded by the BEIR V study. The BEIR V results indicate a higher cancer risk from low levels of ionizing radiation than did the earlier BE % III report. The results of the BEIR V study must be taken into account in the calculation of health effects. The taff is preparing an addendum to NUREG/CR-4214, "Health Effects Models for Nuclear Power Plant Consequence Analysis," Revision 1, Part II, to address the modification of models resulting from recent reports on the health effects of ionizing radiation. The risk coefficients for fatal cancers would be approximately doubled or tripled by the model modifications. Until these modifications are incorporated into MACCS, the staff concludes that use of CRAC2 is acceptable, but that the effect of model differences must be taken into account in interpreting risk results. The staff will address this issue during its review of an individual application for FDA/DC.

#### 6 UNCERTAINTY AND SENSITIVITY ANALYSES

ö

In Section 6 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI requires that sensitivity studies be performed for those issues or parameters that are judged to have relatively large associated uncertainty or that are particularly important to the PRA results. These sensitivity studies must include important aspects from each of the areas discussed below. EPRI indicates that the sensitivity studies may be qualitative or quantitative, depending on the nature of the issue being addressed, and are intended to provide adequate perspective with respect to uncertainty in the PRA results and the significance of potential contributors to risk.

In general, the staff accepts the scope defined for analysis (with a number of reservations stated below). However, it believes that the guidance in Section 6 is incomplete. For example, EPRI does not provide any details about the manner and range of variation for the parameters selected for back end sensitivity studies analysis. Moreover, the staff does not agree that sensitivity studies alone are in all cases an adequate substitute for uncertainty analysis. As discussed in Section 1.9 of this appendix, it is the staff's view that sensitivity analyses supplement uncertainty analysis by integrating areas of subjectivity in the analysis and by helping the analyst characterize major contributions to uncertainties in the results. The staff concludes that the uncertaint, and sensitivity analyses required by Section 6 of Appendix A to Chapter 1 of the Evolutionary Requirements Document should be augmented as described below.

# 6.1 Analysis of Systems and Sequences

In Section 6.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI lists areas that may be of particular importance with respect to the estimated frequency of core damage, including

- frequencies of rare initiating events that are important contributors to risk and any initiating events whose frequencies are assessed to be low relative to similar events for other nuclear power plants
- common-cause parameters, especially those whose assessment relied heavily on engineering judgment to compensate for a lack of experimental data
- human interactions following an initiating event
- seismic hazard

EPRI states that the list of areas represents those areas that may be most uncertain or are likely to be important to the frequency of core damage. In the DSER for Appendix A to Chapter 1, the staff stated that although it agreed that these areas should be considered, it believed there were other important areas of uncertainty that must explicitly be recognized. The staff concluded that the flood hazard (Section 3.2 of Appendix A to Chapter 1) must be included as well as failure rates for equipment items different from those used in current plants or subject to different operational conditions, such as advanced instrumentation and control components. This was identified as an open issue.

EPRI Evolutionary Plant SER 1A.6-1

EPRI responded to the staff's DSER by indicating that the list was not intended to be exhaustive and by modifying the list to also include failure rates for equipment different from that used in current plants. The staff has considered the adequacy of EPRI's new guidance in terms of its consistency with (1) the scope of analyses that the staff will consider during its review of an application for FDA/DC, as described in Section 1.9 of this appendix; (2) the requirements placed on the PRA by the reliability assurance program; and (3) the guidance for external events provided in Section 3 of Appendix A to Chapter 1. As discussed in the following three sections, the staff concludes that EPRI's response is acceptable and will review an application for FDA/DC to ensure that these issues are adequately addressed.

#### Consistency of Scope

As discussed in Section 1.9 of this appendix, the staff will review an application for FDA/DC to ensure that ALWR vendor PRAs include a full uncertainty analysis for the Level 1 portion of the PRA, with uncertainties propagated from basic events, including initiating event frequencies, data, common cause/mode failure, success criteria, and human error. The staff concludes that the areas identified in Section 6.1 of Appendix A to Chapter 1 of the Evolutionary Requirements Document are consistent with the staff positions described in Section 1.9 of this appendix regarding treatment of uncertainties, provided the following areas are also addressed by the ALWR vendor in the uncertainty analysis:

- failure rates for equipment subject to operational conditions different from those in conventional PWRs and BWRs
- initiating events that are unique to evolutionary designs
- success criteria

The staff will evaluate the adequacy of the vendor's treatment of uncertainty in each of the aforement aneas when it reviews a design-specific PRA as part of its review of an upplication for FDA/DC.

# Importance Analyses for Input to Reliability Assurance Program

The staff requires the ALWR vendors, in addition to systematically assessing the significance of uncertainties in risk results, to use the ALWR PRAs to provide insights into equipment that should be included in the design-specific reliability assurance program (RAP) and for which each vendor will formulate guidance. Such equipment will be identified using importance analysis of the PRA model.

EPRI has provided guidance in Section 6 of Chapter 1 of the Evolutionary Requirements Document on conducting a RAP; however, this guidance lacks specificity regarding how a utility is to determine what specific equipment should be included in the RAP. The staff will review applications for FDA/DC to ensure that ALWR venders provide a list of equipment they believe should be included in the RAP, and requires that the ALWR vendors provide guidance to combined license applicants on how the applicants should determine what additional equipment should be included in the program, on the basis of the as-built plant. Although the specific NRC review criteria for the RAP are still evolving, it is clear that the design-specific PRA should be used to

provide a technical basis for many aspects of the RAP and should be integrated into RAP development and application. Accordingly, the staff requires that ALWR vendors calculate and submit (for their RAP) the results of importance analyses for each of the basic events in the PRA, using "importance measures" to rank systems, structures, and components (SSCs). Appropriate "importance measures" should be chosen so that they, as a group, provide all the needed information on the SSCs' effect in controlling and reducing risk. The riskbased ranking of SSCs for the RAP can be achieved by using importance measures such the risk achievement (or risk increase worth), the risk reduction worth, or equivalent measures of importance such as Fussel-Vesely and Birnbaum. The risk achievement (or rist increase worth) importance measure is of special interest in RAP because it measures the relative importance of features that contribute to achieving the risk level assumed in the PRA and toward which reliability assurance activities should be directed. The risk reduction worth is of special interest in ranking activities aimed at reducing risk, such as modifications in plant operations or design.

The staff concludes that vendors must provide two sets of analyses as part of an application for FDA/DC - the first to reflect the relative importance of systems, components, and actions to core damage from internally initiated events, and the second to reflect the importance when all initiators are considered (i.e., internal and external events combined). The use of two sets is to prevent the possibility that insights from considering external initiators may swamp insights from considering internal initiators, since the staff believes that external initiators have larger uncertainties than many internal initiators and since the ALWR designs have significantly reduced the absolute value estimates of frequency of core damage from internal initiators. Importance analyses for external events analyzed with bounding or margins-type analyses will have to be evaluated in a qualitative manner.

The risk significance of equipment important to the level 2 analysis may not be readily identified through traditional importance analyses, which focus only on core damage frequency. Accordingly, in screening systems, structures, and components for inclusion in the RAP, the vendor (and subsequently the combined license applicant) must also consider equipment important to accident progression, containment performance, and releases. The staff would expect the vendor to develop a systematic method that uses the Level 2 portion of the PRA, and the associated sensitivity and uncertainty analyses, to perform this additional assessment.

In general, the ALWR vendor must (1) provide insight into what systems, structures, and components should be included in the RAP and (2) give guidance to combined license applicants on how to determine which systems, structures, and components should be included in the RAP on the basis of the as-built plant. Vendor guidance needs to address situations in which bounding or nonprobabilistic analyses are used by the ALWR vendor, since importance analyses may not accurately reflect the importance of equipment. Similarly, there may be areas and equipment that are in the plant or are site specific and that are safety significant, but are not modeled in the PRA (e.g., structures, passive components, a heater for a borated tank, or heat tracing for piping with highly borated water). Importance analysis would provide no insights in these areas. The staff will evaluate the adequacy of the verdor's importance analyses at the FDA/DC and combined license stages for each evolutionary design PRA.

# External Events

The staff intends to use high confidence/low probability of failure (HCLPF) values (both for the plant and by event sequence) obtained from the seismic margins analysis to help make regulatory and licensing decisions. Additional regulatory and licensing insights will be gained from evaluating sequences ranked by HCLPF values. Therefore, the staff will not require that hazard curve uncertainties be addressed in the design-specific PRA or its updates, as discussed in Section 3.3 of this appendix.

A severe-accident fire analysis must include at least a qualitative evaluation of uncertainties including fire initiation frequency and the chance that smoke, hot gases, and/or fire suppressant will cross 3-hour fire barriers that separate divisions.

A severe-accident internal flooding analysis must include at least a qualitative evaluation of the uncertainties, including flood initiation frequency, flow rates, detection failure and detection timing, and migration of water to other divisions or floors.

For external flooding, as discussed by the staff in Section 3.2 of this appendix, a detailed uncertainty analysis is not required because the return period uncertainties dwarf those of random equipment failure or equipment capacities. For external flooding, if the expected frequency with which river floods will exceed the probable maximum flood (PMF) is 1E-5 per year or greater, the combined license applicant should perform an analysis of the conditional probability of being unable to maintain core cooling, containment integrity, and reactivity control. It is expected that there would be adequate warning time in case of a flood so that the plant would be shut down before the flood exceeded the PMF.

## 6.2 Assessment of Containment Response

In Section 6.2 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI lists the following important parameters that could affect the frequency of serious releases that must be considered in the sensitivity studies:

- parameters related to hydrogen burn
- core debris coolability
- pressure capacity of the containment and the location and size of pressure-induced failure of the containment
- parameters that could affect high-pressure melt ejection
- parameters associated with the production of combustible gas outside the reactor vessel
- operator actions that could affect accident progression

EPRI states that these are among the most important uncertainties that could affect the assessment of conditional frequency of releases due to core-damage accidents. In the DSER for Appendix A to Chapter 1, the staff agreed that

uncertainties in these and other parameters were important but stated that sensitivity studies alone were not adequate, and must be supplemented by a full, quantitative uncertainty analysis.

The staff has reconsidered the issue of uncertainty analyses and concludes that a full, quantitative uncertainty analysis is not required for the Level 2 portion of the PRA analysis, which covers the aforementioned areas. Instead, the staff requires (1) the implementation of a systematic process for identifying issues and phenomena of greatest risk significance for the advanced designs and (2) a more thorough treatment of those issues and phenomena and their associated uncertainties as part of the Level 2 analysis, for example, in the containment event trees. (See Section 1.9 of this appendix for additional discussion.)

The staff concludes that the list of issues and phenomena in Section 6.2 of Appendix A is a reasonable set, but notes that this set is not necessarily complete. For example, containment bypass and aerosol plugging may be important for some ALWR designs, but these are not included on the list.

The staff will address the acceptability of the ALWR designer's treatment of severe-accident issues and phenomena as part of the design certification review for each ALWR design. The adequacy of the analysis will be judged on the basis of the completeness of issues considered, reasonableness of parameter ranges and screening criteria for sensitivity analyses, and the methodology for decomposing and propagating the range of potential issues outcomes in the analysis. The analysis must provide the staff reasonable assurance that the PRA reflects the significance of key actions, events, and phenomena for the plant design, as well as the effectiveness of the accident mitigation features. This DSER open issue closed.

## 6.3 Analysis of Source Terms

In Section 6.3 of Appendix A to Chapter 1 of the Evolutionary Requirements Document, EPRI lists the following parameters that could affect the magnitude of accidental radionuclide releases that must be investigated in appropriate sensitivity studies:

- the effectiveness of containment scrubbing mechanisms
- the form of iodine and the corresponding amount of iodine available for release

EPRI states that these are some of the most important parameters associated with the behavior of fission products in the reactor coolant system and in the containment that could affect the amount released to the atmosphere after containment failure. The staff indicated in the DSER that it was appropriate to consider these parameters in sensitivity studies, but stated that sensitivity studies alone were insufficient and must be supplemented by a quantitative uncertainty analysis.

The staff has reconsidered the issue of uncertainty analyses and concludes that a full, quantifative analysis of uncertainty is not required for the Levels 2 and 3 portions of the PRA.

As stated by the staff in Section 1.9 of this appendix, ALWR vendors must use a more systematic process for identifying risk-significant issues of relevance to the ALWR plant designs. This would include the conduct of sensitivity analysis to screen issues and identify those with significant impact on risk. Although the thrust of this systematic process is on Level 2 issues, it is also important that plant designers identify and assess those Level 3 issues that may result in a shift of accident frequencies from relatively begin accident release classes to the category of "large release," and consider the need to incorporate uncertainties in these issues in the context of the full uncertainty analysis.

In a February 7, 1991, response to an NRC question concerning plant certification issues contained in SECY-90-016, "Evolutionary Light Water Reactor Certification Issues and Their Relationship to Current Regulatory Requirements," EPRI stated that the plant designers must confirm as part of the PRA that the source terms for representative accident sequences for their actual standard plant designs are bounded by the physically based source term in the Evolutionary Requirements Document. However, no mention of the need for PRA analysts to perform this source term assessment is made in Appendix A to Chapter 1. This was identified as an open issue in the DSER.

Responding to this DSER open issue in a letter dated May 1, 1992, EPRI stated that it has modified Appendix A to Chapter 1 of the Evolutionary Requirements Document and has left the choice of using either the current source term specified in Atomic Energy Commission document TID 14344 or the updated physically based source term to the designers. At this time, both General Electric and ASEA Brown Bovlai/Combustion Engineering have indicated that their evolutionary designs can meet 10 CFF Part 100 dose limits with the TID 14844 source term. However, the vendors have also indicated that they may wish to apply the revised source term to their designs when it becomes available. Should a designer elect to use the physically based source term for licensing calculations, EPRI stated that it will develop and incorporate a requirement in the Evolutionary Requirements Document for the designer to compare the physically based source term with and the PRA source term at that time. The staff concludes that this is acceptable. Therefore, this DSER open issue is closed.

# 7 HUMAN RELIABILITY ANALYSIS

Guidance on the analysis of human interactions is provided in Section 2.9 of Appendix A to Chapter 1 of the Evolutionary Requirements Document. This section specifies that the EPRI Systematic Human Action Reliability Procedure (SHARP) analysis framework will be used for considering human interactions in the PRA and requires that the analysis deal explicitly with the following:

- (1) definition of human actions
- (2) screening for importance
- (3) task breakdown
- (4) representation in relation to systems logic models
- (5) iteration between human and hardware modeling
- (6) quantification
- (7) documentation

EPRI's SHARP analysis requires that human interactions be placed into the following categories:

interactions before an initiating event (Type A)

- actions related to the initiating events (Type B)
- interactions following an initiating event. including actions dictated by procedures (Type CP), and actions to recover equipment or systems to terminate a sequence (Type CR)

High-level guidance on the required scope of the human reliability analysis (HRA) and the review, classification, and quantification of each of these types of human interactions is provided in subsequent sections of Appendix A.

The staff, in its DSER for Appendix A to Chapter 1, identified several major shortcomings in the EPRI guidance. These included inadequacies in the document with regard to the following:

- incorporating HRA considerations throughout the PRA requirements process and providing mechanisms to ensure that credible and auditable results are obtained
- providing guidance and approaches for the conduct of function, task, and timeline analyses
- providing systematic guidelines for selecting and for evaluating personcentered, task-centered, and environment-centered performance shaping factors (PSFs)
- providing information on state-of-knowledge PSF evaluation tools to ensure that the PRA results are credible, auditable, and consistent across sequences and systems
- providing information on state-of-knowledge quantification methods

In a letter dated January 24, 1992, regarding these concerns, EPRI proposed modifications to the Evolutionary Requirements Document. The staff has reviewed Appendix A and EPRI's responses to the DSER in terms of the (1) scope and objective, of the required analysis, (2) framework or process for validating the HRA to reflect the detailed design, and (3) guidance ind requirements for conducting the specific elements of the HRG. The results of this evaluation are provided below.

## 7.1 Scope and Objectives of HRA

In its DSER for Appendix A to Chapter 1, the staff identified inadequacies in incorporating HRA considerations throughout the PRA requirements process and providing mechanisms to ensure that credible and auditable results are obtained as an open issue. EPRI's proposed revisions to Section 2.9 of Appendix A are primarily aimed at further clarifying the intended scope and objectives of the HRA. The thrust of the modifications is to more clearly reflect that the plant design will not be complete when the HRA is performed by the vendor and that because of the state of the plant design, there will be limitations on the analysis beyond those normally encountered in conducting an HRA. EPRI states that when the HRA is performed there will be no operating or emergency procedures for the plant. no control room, and no directly applicable experience with a like plant. As a result, EPRI's stated objectives of the HRA are "to obtain a nominal assessment of the human interactions to support the overall risk assessment, and to ensure that the assessment is accomplished in a consistent and traceable fashion." The revisions establish that the assessment of duman interactions will be sufficient to accomplish the following:

1

- identify the types of human interactions that may be in the nt to risk for the ALWR design
- provide a nominal quantification of the human interactions sufficient to support the overall assessment of core damage frequency and frequency of severe release
- provide a mechanism to investigate the potential effects of varying the reliability assessed for the human interactions
- establish a frimework for performing a more extensive HRA when the design progresses to the state of an actual plant, with operating procedures, iyout, etc.

Consistent with the reduced scope, EPRI has redefined those items that the analysis must explicitly address to consist of the following in lieu of Items (1) through (7) in Section 7 above:

- plant lugic model construction
- guantification
- analysis of recovery actions
- internal review

The staff recognizes that it is not possible to conduct a meaningful detailed HRA in the absence of such specific items as emergency operating procedures control room design layout and staffing, and plant simulators. Accordingly, the scope, objectives, and level of detail of an analysis performed at this

EPR. Liolutionary Plant SER

stage of plant assign would be distinctly different from, and more limited than, those associated with an HRA for a completed plant design. The staff considers the HRA objectives set forth by EPRI in the revised Evolutionary Requirements Document to be reasonable for a "scoping analysis" or "approximate quantification" for a partially completed plant design. However, the staff will review applications for FDA/DC or a combined license to ensure that vendors and applicants provide a more detailed and defensible treatment of human reliability after the design details have been developed and before plant operation. Such details include the design of the control room and manmachine interface, control room staffing, emergency procedures, and operator training. This process is discussed in greater detail in Sections 7.2 and 7.3 below. This DSER open issue is closed.

# 7.2 Process for Validating the HRA

Section 2.9 of Appendix A to Chapter 1 of the Evolutionary Requirements Document provides no specific requirements or g idance concerning refinement and validation of the scoping HRA to reflect the details of the final design, such as the use of advanced technology and automation, and its impact on human performance and the HRA.

The staff will review applications for FMAP to ensure that the vendor for each ALWR design establishes the process and criteria by which the combined license applicant will contirm the adequacy of the HRA treatment and, if appropriate, upgrade the analysis. It is the staff's expectation that this HRA implementation plan will establish an iterative process (involving a limited number of iterations) by which information developed as the control room design matures (and possibly other aspect: of the design) is assessed against the HRA assumptions, and conversely, HRA insights are factored into the control room design. Inputs to the HRA would come from several elements of the plant design process, such as the control room function analysis, task analysis, and man-in-the-loop testing. These inputs are expected to be qualitative in nature, but would provide a basis for judging certain quantitative aspects of the HRA. Outputs from the HRA would provide insights into other elements, such as the development of the man-machine interface and plant procedures. The staff will evaluate the HRA implementation plan or process when it reviews an ALWR design-specific PRA as part of its review of an individual application for FDA/DC.

#### 7.3 Guidance on Specific Elements of the HRA

#### Use of New Technologies

Increased automation (automation of tasks traditionally performed by an operator) and enhanced decision aids in ALWRs will resul in a shift of the operator's role from a direct manual controller to a sup visory controller and system monitor. The shift in role away from direct control is typically viewed as positive from a reliability standpoint, since the operator is considered one of the more unpredictable components in the system. It has been observed in other industries, however, that although some errors may be reduced or eliminated, such a change has frequently been associated with a shift of huma error to higher levels in the system that are more difficult to detect and quantify. Examples of such errors include (1) errors in attempting to recover from the failure of an automated system; (2) errors in the setup of automated systems, such as keying the wrong information or data into an

automated system; (3) actions taken in response to a false alarm (provided by an intelligent system and not adequately checked); and (4) failure to properly monitor the automated system (loss of vigilance because of overreliance on the automated system). The potential for these "new" types of errors that can occur in an advanced system must be considered in the human action modeling and HRA for ALWRs.

The staff has reviewed the degree to which the Evolutionary Requirements Document addresses, or at least recognizes, the potential for advanced technologies (e.g., digital displays, touch screens, automatic decision aids) anticipated in ALWR plant designs, and their implications for the designspecific PRA. These include the impact on human performance due to increased automation, changes in the human-system int. face and methods of operator interaction with the system, and advanced control room data management systems. The staff concludes that EPRI's guidance is deficient insofar as taking note of and providing contingencies for dealing with new technologies anticipated and currently being incorporated into ALWR designs. However, it is unclear if state-cf-knowledge HRA modeling, data gathering, and performance quantification tools are adequate to handle anticipated designs and the technologies they embrace, such as automated decision aids. This suggests the need for more empirical approaches to modeling and quantification, such as part-task prototyping and full-scale control room prototypes.

In the staff's view, the impact of advanced technologies on the HRA is an issue that is best addressed by the vendor and combined license applicant during the iterative process of designing the control room, rather than as part of the scoping HRA. In this regard, the staff requires that combined license applicants pay particular attention to the effects of system automation, decision aids, and artificial intelligence on human performance and errors as part of control room design process. A possible method for ascertaining the impact of the advanced technologies would be to perform accident simulations or walk-through, talk-through exercises with plant operators as part of the development of the control room prototype and of the detailed control room design. A final evaluation of the impact of advanced technology on the HRA must be made after the control room design and plant-specific simulator are available, as part of the subsequent effort to validate the HRA. In either case, the focus of these evaluations would be on observing the impact of the advanced features on operator performance and error potential, and confirming that potentially significant errors are included in the HRA.

The staff will review applications for FDA/DC to ensure that ALWR vendors define the process and acceptance criteria to be used by the combined license applicant to address the impact of new technologies on human performance. This process should ensure that potentially significant new human errors (i.e., not considered in the scoping HRA) are identified during the detailed development of the control room and addressed early in the design process either by adoption of alternative design approaches or by modification of the HRA to better reflect the potential for additional errors. The process should also provide guidance to the combined license applicant regarding the use of the plant-specific simulator in validating the assumptions and models in the HRA for the PRA. The staff will evaluate this process when it reviews an ALWR design-specific PRA as part of its review of an application for FDA/DC.

#### Function, Task, and Timeline Analyses

The staff has evaluated the degree to which Appendix A to Chapter 1 establishes requirements and provides guidance on the conduct of function, task, and timeline analyses. These analyses provide a mechanism for characterizing human actions and human action sequences and identifying potential problems in operator performance. This is especially important in areas where individual cognitive errors and combinations of human errors are possible during dynamic accident sequences. In its DSER for Appendix A to Chapter 1, the staff identified the lack of guidance in this area as a deficiency in Appendix A to Chapter 1. In response, by letter dated January 24, 1992, EPRI i dicated that function, task, timeline, and link analyses were inappropriate. given the state of the plant design at the design certific tion stage.

The staff concludes that Appendix A of Chapter 1 still does not provide adequate guidance in this area. Moreover, the document fails to establish a process for performing a more detailed assessment of human performance using appropriate analysis techniques once the plant design details have been established. The staff considers the EPRI guidance on he modeling of human interactions to be adequate for scoping-type analyses of human performance. However, the staff will review applications for a combined license to ensure that a reassessment of the HRA is performed befor plant operation that is based on function and task analyses that reflect the final design. (These analyses are typically performed as part of the detailed control room design process and the development of operating procedures.) The staff will review an application for FDA/DC to ensure that the ALWR vendor has developed guidance for a process and criteria to be used by the combined license applicant to perform this reassessment. The staff will evaluate the process and criteria when it reviews an ALWR design-specific PRA as part of its review of an application for FDA/DC. The DSER open issue related to function, task, timeline, and link analyses is closed.

## Generic Data Sources

The staff has reviewed the degree to which generic data sources are identified in Appendix A to Chapter 1. These data are needed as bounding and anchor values for quantifying individual human actions and more complex human action sequences involving several interdependent actions by individuals or groups of individuals. In the DSER for Appendix A to Chapter 1, the staff stated that EPRI had committed in a letter dated November 7, 1990, to add other sources of generic data in addition to the THERP (Technique for Human Error Rate Prediction) handbook (NUREG/CR-1278, "Handbook of Human Reliability Analysis With Emphasis on Nuclear Power Plant Applications"). This was identified as a confirmatory issue in the DSER.

The staff has confirmed that Appendix A has been revised to reference a number of additional sources of generic data, including the following:

- more recent work by Swain performed as part of the NRC Accident Sequence Evaluation Program (NUREG/CR-4772, "Accident Sequence Evaluation Program Human Reliability Analysis Procedure")
- the revised SHARP document (EPRI NP-7183-SL)

more recent work by Party (EPRI TR-100259) performed as part of the EPRI Operator Reliability Experiment project

()

The staff considers the data sources identified in Appendix A to Chapter 1 to be an acceptable but minimal set. It expects ALWR vendors to support the use of the data contained therein by comparisons with data from additional sources (e.g., other peer-reviewed HRAS). The NRC-sponsored Nuclear Computerized Library for Assessing Reactor Reliability provides one such source of both human and hardware component failure data that might be used by ALWR plant designers to support their PRA and HRA. The staff will assess this aspect of the HRA when it reviews an ALWK design-specific PRA as part of its review of an application for FDA/DC, in conjunction with the issues of quantification and performance shaping factors. This confirmatory issue is closed.

# Performance Shaping Factors and Evaluation Tools

The staff has evaluated the degree to which Appendix A to Chapter 1 provides (1) systematic guidelines for selecting person-centered, task-centered, and environment-centered performance shaping factors (PSFs) to ensure that the task actions are adequately characterized and (2) information on PSF evaluation tools to ensure that the HRA results are credible, auditable, and consistent across sequences and systems. In the DSER for Appendix A to Chapter 1, the lack of guidance in this area was identified as an open issue.

In response to staff concerns, EPRI modified Section 2.9 of Appendix A to Chapter 1 to refer to more recent work by Parry (EPRI TR-100259) for quantification of human interactions. The recent work provides two distinct methods of quantification, either of which can be used depending on the level of information available. One method uses a time-reliability correlation, and the other focuses on assessing PSFs such as the nature and content of procedures and level of training. These methods provide approximations of human performance derived from data that implicitly reflect the effects of PSFs without attempting to characterize individual contributing factors. EPRI also contended in its response that detailed treatment of PSFs would require information that could not be specified until after an actual plant was built, or at least a very good simulator was available.

The staff concludes that the guidance and approach in Appendix A to Chapter 1 of the Evolutionary Requirements Document regarding PSFs are adequate performing scoping analyses of human interactions, but insufficient with regard to the types of analyses that are needed after ALWR design details have been established and the plant-specific simulator is available. Explicitly identifying and evaluating specific PSFs is a key factor in any HRA. Without a clear understanding of the PSFs as they pertain to a particular plant, the credibility of the quantitative results is reduced, as is the potential to gain insights into remedial actions required for reducing human contributions to risk. Accordingly, the staff will review applications for FDA/DC to ensure that combined license applicants will perform a further assessment of PSFs and their impact on the HRA results after the ALWR control room design details have been established. The staff will also review applications for FDA/DC to ensure that ALWR vendors have developed guidance on the process and criteria to be used by the combined license applicants to assess the impact of PSFs on HRA results. Information on preparing this type of guidance is available from documentation developed by EPRI (Reports NP-309, "Human Factors Review of Nuclear Power Plant Control Room Design"; NP-2411, "Human Engineering Guide

for Enhancing Nuclear Control Room"; and NP-1982, "Evaluation of Proposed Control Room Improvements Through Analysis of Critical Operator Actions") and that developed under a variety of control room design, training, and operating procedures programs of the NRC. The staff will evaluate this process when it reviews an ALWR design-specific PRA as part of its review of an application for FDA/DC. The DSER open issues concerning PSFs and their evaluation tools are closed.

## Quantification Methods

The staff has reviewed the degree to which Appendix A to Chapter 1 provides guidance on state-of-knowledge quantification methods to ensure that the HRA results are credible, auditable, and consistent across sequences and systems.

In the DSER for Appendix A to Chapter 1, the staff indicated that the guidance in Appendix A did not adequately address the numerous quantification methods that have been developed and tested over the past 4 to 6 years by various domestic and international organizations involved in HRA. As mentioned in the two preceding subsections, EPRI has revised Appendix A to refer to more recent work performed by EPRI (Parry et al.). The more recent work by Swain (NUREG-/CR-4772) has also been referenced. Methods in either of these references could be used for assessing the nominal probabilities of human interactions for the ALWR PRAs and should produce results that can be readily traced. In either case, it would be necessary to document the timeline for the human interactions, as well as any other relevant features.

The staff concludes that the guidance in Appendix A still provides a less than complete picture of alternative quantification methods, some of which address dynamic modeling and cognitive error issues more comprehensively than the methods identified in the Evolutionary Requirements Document. However, the staff also recognizes that these alternative techniques are not a panacea for the difficulties in human performance quantification, because large uncertainty and variability are inherent in the application of any of these techniques. This has been demonstrated in EURATOM's Human Reliability Benchmarking study.

Most critical to the quality of an HRA is that all significant human actions are represented in the analysis and that the relative importance of human actions in the risk profile of the plant is determined. The particular numerical values assigned to human actions, in and of themselves, are of second order importance. With regard to the adequacy of the HRA quantification, ALWR vendors are required to further assess human error probability values as part of the sensitivity and uncertainty analyses called for in Sections 1.4 and 6.1 of Appendix A to Chapter 1. The staff will review applications for a combined license to ensure that applicants reassess the adequacy of the HRA quantification as part of the HRA validation process after the control room design details have been established. In this context, the staff would expect combined license applicants to make use of the control room prototype and plant-specific simulator to the extent necessary to confirm the adequacy of the HRA in terms of completeness of human interactions represented in the analysis and reasonableness of assumed human error probability values. With the design-specific considerations described above, th taff concludes that the EPRI guidance on quantification is adequate for the purpose of ALWR HRAs and this DSER open issue is closed.

# Documentation and Reporting

The staff has reviewed the degree to which Appendix A to Chapter I provides guidelines for developing and maintaining "running" documentation on HRA work in progress and for reporting the results of the HRA.

The SHARP framework discussed in Section 2.9 of Appendix A provides an outline for documenting the HRA. Although general, the outline does provide a mechanism for documenting the HRA to allow users and reviewers to judge the degree to which each of the following has been achieved:

- credible quantitative results (point estimates, uncertainty bounds, confidence intervals)
- quantitative and qualitative results that are fully auditable, that is, provide traceable relationships between the error probability estimates and human factors information on which the estimates are based
- qualitative results that can support causal analyses of problem areas and attendant remedial actions
- quantitative and qualitative data that are reproducible and lend themselves to issues that transcend individual sequences and plant systems
- state-of-knowledge computational methods, data, and procedures for conducting HRA within the context of a PRA
- contingencies for dealing with new technologies anticipated and currently being incorporated into ALWR designs

The staff concludes that the guidelines provided by EPRI for developing and maintaining documentation on HRA work in progress and for reporting the results of . . PRA are acceptable.

# 8 CONCLUSION

The staff concludes that the EPRI requirements established in Appendix A to Chapter 1 of the Evolutionary Requirements Document for conducting a PRA do not conflict with current regulatory guidelines and are acceptable. However, by themselves, they do not provide sufficient information for the NRC staff to determine that the design-specific PRAs will be adequate. Individual applicants for FDA/DC referencing the Evolutionary Requirements Document will be required to submit a design-specific PRA for review by the staff.

Therefore, the staff concludes that Appendix A to Chapter 1 of the Evolutionary Requirements Document specifies requirements that, subject to resolution of the identified vendor- and utility-specific items, if properly translated into a design and constructed and operaled in accordance with the NRC regulations in force at the time the design is submitted, should result in a facility that will have all the attributes required by the regulations to ensure that there is no undue risk to the health and safety of the public or to the environment. In addition to complying with existing regulations, such a facility would also be consistent with Commission policies on severeaccident protection and public safety goals.

#### ANNEX A - RELIABILITY DATA BASE FOR ALWR PRAS

Annex A of Appendix A to Chapter 1 of the Evolutionary Requirements Document describes the development of the initiating event frequencies and component reliability data to be used in developing the PRA. Section A1 of Annex A outlines the methods used in obtaining initiating event frequencies for lossof-coolant accidents and for most transient events. Section A2 of Annex A describes the treatment of the frequency and recovery of losses of offsite power in greater detail.

Section A3 of Annex A summarizes the sources of data used to arrive at the recommended hardware failure rates, maintenance unavailabilities, and common-cause factors.

Annex A, Part 2, contains the results of a survey of component failure rates for evolutionary ALWRs.

#### Section A2, "Loss of Offsite Power"

This section provides background information on how the loss-of-offsite-power frequencies were established for the reliability data base to be used in the ALWR PRAS. In the DSER for Appendix A to Chapter 1, the staff stated that the total loss-of-offsite-power frequency established by EPRI for the ALWR (0.0077 loss per site-year) is an order of magnitude less than the long-term historical average for total loss of offsite power at U.S. nuclear plants (0.07 loss per site-year) and is approximately 23 percent of the 3-year average (0.033 loss per site-year for 1986, 1987, and 1988) found in Nuclear Safety Analysis Center Report NSAC/147, "Losses of Off-site Power at U.S. Nuclear Power Plants: Through 1989." In the DSER, the staff recommended that EPRI work with the staff in resolving differences between NSAC/147 and the staff's data for loss-of-offsite-power events and identified this as an open issue.

In a letter dated January 26, 1992, EPRI stated that data related to loss of offsite power will be revised to include consideration of the NRC Office for Analysis and Evaluation of Operational Data (AEOD) data after the policy issue on alternative source of power for non-safety loads is resolved with the ctaff. The staff concludes that EPRI's response is acceptable and will review individual applications for FDA/DC to ensure that design-specific PRAs use data that accurately characterize the nature of the offsite power system, including a loss-of-offsite-power frequency of 0.033 per site-year. Therefore, this DSER open issue is closed.

#### Annex A, Part 2, "Survey of Component Failure Rates for Evolutionary ALWRs"

Item No. 65, Transformer (high voltage): fails to continue operating

Under this item, failure rates of high-voltage transformers taken from several sources are provided. The failure rate selected for ALWR PRAs is 1.2E-6 failure per hour. In the DSER for Appendix A to Chapter 1, the staff noted that the failure rate appeared to be low for main step-up transformers judging

from the number of reports the staff has received on main transformer failures. In response, EPRI modified Annex A to address main step-up transformers explicitly with a failure-to-continue-operating frequency of 5.4E-6 per hour. This value is acceptable to the staff. Therefore, this issue is closed.

#### ANNEX B - ALWR REFERENCE SITE

Annex B of Appendix A to Chapter 1 of the Evolutionary Requirements Document originally included reference site data on meteorology, population density and distribution, and evacuation and sheltering. When the DSER for Appendix A to Chapter 1 was issued, the staff understood that EPRI was evaluating whether to modify or delete the reference site data. In the DSER, the staff stated that it would review any revisions upon submittal, as appropriate, and identified this as an open issue. Consistent with the desire to decouple the plant design from siting considerations, all but the meteorological data were deleted in a subsequent revision. The staff concludes that this is acceptable because the population and evacuation data are not needed to estimate dose at the site boundary. The meteorological data were also updated to be more representative of a bounding (80th percentile) site.

By letter dated April 9, 1992, EPRI submitted a summary of the methods and assumptions used in the development of the ALWR 80th percentile meteorological data base. The staff has reviewed this information and concludes that the methods and assumptions used by EPRI in the development of the ALWR 80th percentile meteorological data base are consistent with current practice and methodology. Therefore, this DSER open issue is closed. As part of the FDA/DC review for each ALWR PRA, the staff will evaluate the site data used by the vendor in estimating dose at the site boundary to ensure that these data are representative of most potential ALWR sites.

## ANNEX C - COMPONENT SEISMIC CAPACITIES

Annex C of Appendix A to Chapter 1 of the Evolutionary Requirements Document describes the process by which generic seismic capacities were estimated for the evolutionary ALWR designs. These generic capacities, defined in terms of median spectral accelerations, were generated by reviewing available seismic PRAs, seismic fragility analyses, seismic margins analyses, and generic equipment ruggedness data.

EPRI states that the component capacities do not represent the highest or lowest capacities that can be identified in the literature. Rather, the values represent what EPRI considers reasonable capacities for the ALWR design. If the fragility values generated from the use of these data result in overly conservative results, a component-specific capacity may be developed that is based on design-specific or vendor-supplied information.

Annex C also describes the methodology and assumptions used in developing generic component fragilities.

The generic fragilities presented in this document may not be applicable to some of the advanced instrumentation and control components that may be used in ALWR reactors. The staff cautions ALWR vendors to this effect and requires that they justify the use of generic fragilities in their seismic analyses.

### CHAPTER 1, APPENDIX B, "LICENSING AND REGULATORY REQUIREMENTS AND GUIDANCE"

#### 1 INTRODUCTION

This appendix of the SER documents the NRC staff's review of Appendix B, "Licensing and Regulatory Requirements and Guidance," to Chapter 1 the Evolutionary Requirements Document through Revision 3. Appendix B to Chapter 1 was prepared, under the project direction of EPRI and the ALWR Utility Steering Committee, by Grove Engineering, Inc.; MPR Associates, Inc.; S. Levy Inc.; TENERA, L.P.; and EPRI.

Although Appendix B to Chapter 1 was not submitted as part of the original Evolutionary Requirements Document, it was included in Revision 1, which EPRI submitted on September 7, 1990. This appendix consolidates the resolutions proposed by EPRI for its proposed optimization subjects and certain unresolved and generic safety issues. Revisions 2, 3, and 4 of the Evolutionary Requirements Document were docketed on April 26 and November 15, 1991, and April 17, 1992, respectively. EPRI submitted additional information regarding this appendix by letters dated February 12 and December 6, 1991, and March 3, 1992.

Because EPRI had not originally submitted Appendix B to Chapter 1, the staff did not develop a DSER for it However, many of the topics now contained in Appendix B were discussed throughout the 13 chapters of the original document. Accordingly, the staff's original review of many of these topics was discussed in the corresponding chapters of the DSER.

#### 1.1 <u>Review Criteria</u>

Section 1 of Volume 1 of this report describes the approach and review criteria used by the staff during its review of Appendix B to Chapter 1 of the Evclutionary Requirements Document.

## 1.2 Scope and Structure of Appendix B to Chapter 1

Appendix B to Chapter 1 of the Evolutionary Requirements Document is a compilation of the regulatory requirements and guidance that EPRI believes are applicable to the design of evolutionary ALWRs. In addition, it gives EPRI's justification for deviations from current criteria and its proposed resolution of unresolved and generic safety issues that are considered applicable to the evolutionary ALWR design.

The key topics addressed in the Appendix B review include EPRI-proposed requirements for

- compliance with the Commission's regulations and regulatory guidance
- EPRI-proposed optimization subjects
- resolution of certain unresolved and generic safety issues
- resolution of policy issues

#### 1.3 Regulatory Requirements and Guidance

In Section 1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document lists all NRC regulations and guidance that EPRI believes are currently applicable to LWR designs. EPRI states that this section identifies the regulatory requirements (Table B.1-1) and guidelines (Table B.1-2) it believes are applicable to the design of the ALWR at the level of detail consistent with that of the Evolutionary Requirements Document when the document is completed. Tables B.1-3 and B.1-4 identify those requirements and guidelines that EPRI believes are not applicable to the ALWR design.

Section 10.2 of Chapter 1 of the Evolutionary Requirements Document states that the ALWR will comply with the NRC regulatory requirements and guidance in effect on January 1, 1990, consistent with the commitments in Section 1 of Appendix B to Chapter 1. EPRI states that these requirements and guidance include applicable Commission regulations in Title 10 of the Code of Federal Regulations (10 CFR), general design criteria, NRC policy statements, regulatory guides, NUREG-0800 ("Standard Review Plan [SRP] for the Review of Safety Aralysis Reports for Nuclear Power Plants") and other documentation that resolves unresolved and generic safety issues. Although the staff understands EPRI's need to "freeze" the requirements its addresses to those in effect on January 1, 1990, as stated in Section 10 of Chapter 1 of this report, the staff expects that the design certification applications will be in compliance with the Commission's regulations and guidance that are applicable and in effect at the time the certification is issued. In addition, issue resolutions that are different from those arrived at during the staff's review of the Evolutionary Requirements Document may be developed as the staff completes its reviews of the detailed design information provided in the applications for final design approval and design cartification (FDA/DC), and as these designs are litigated in the design certification hearings. Therefore, the staff concludes that the ALWR plant designers will comply with the issue resolutions adopted by the NRC staff during its reviews of applications for FDA/DC in accordance with the requirements of 10 CFR Part 52. The staff will evaluate this compliance during its review of an individual application for FDA/DC.

Sertion 10.2.4 of Chapter 1 states that the plant designer will provide an ALK. design that is consistent with the disposition of the regulatory requirements and guidance identified in Section 1 of Appendix B. EPRI defines the commitment specified in Appendix B as follows:

Comply

The "comply" designation indicates that the ALWR design will fully comply with all regulatory requirements and guidance provided by the reference.

Optimization Subject

"Optimization subjects" are proposals initiated by EPRI to deviate from regulatory requirements. EPRI proposes to resolve these issues by providing technically supportable alternatives to current regulatory requirements. EPRI specifies that the ALWR design will comply with all regulatory requirements and guidance for a regulatory item associated with an optimization subject, except as described in Section 2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

EPRI Evolutionary Plant SER 18.1-2

The staff concludes that, except as discussed throughout this report, Table B.1-1 through Table B.1-4 of Appendix B are generally acceptable for the evolutionary plant designs. The staff's review of the entire Evolutionary Requirements Document provides a clearer picture of the extent to which the document complies with the Commission's regulations.

# 1.4 Outstanding Issues

As stated previously, Appendix B to Chapter 1 was not submitted as part of the original Evolutionary Requirements Document package. Therefore, the staff did not develop a DSER for this appendix and no outstanding issues were associated with it. However, many of the topics now contained in Appendix B were discussed throughout the 13 chapters of the original document. Accordingly, the staff's original review of many of these topics was discussed in the corresponding chapters of the DSER.

On September 7, 1990, EPRI submitted Revision 1 of the Evolutionary Requirements Document, which included Appendix B to Chapter 1. During its review of Appendix B to Chapter 1, the staff identified several items that fall into the category of an outstanding policy issue on which the staff has taken a position, but for which the Commission has not had the opportunity to provide guidance.

The outstanding issues, with references to appropriate sections of this appendix given in parentheses, are listed below. The designators in front of each issue provide a unique identifier for each issue. The letter "E" indicates that the issue applies to the evolutionary prant design. The first number designates the chapter in which it is identified, and the letter that follows designates the appendix. The letter "O" designates that it is an open issue. The final number is the sequential number assigned to it.

#### Open Issues

- E.1B.0-1 impact of the elimination of the operating basis earthquake from the design process (2.1.1, Item IV.A of Annex A, and Item I.M of Annex C, Item C of Annex D)
- E.1B.0-2 applicability of industry codes and standards (2.1.1 and item II.A of Annex C)
- E.1B.0-3 tornado design basis (2.1.2 and Item II.F of Annex C)
- E.1B.0-4 main steamline classification (2.3.1.1 and Item II.E of Annex C)
- E.1B.0-5 simplification of postaccident sampling system (2.3.2 and Item II.I of Annex C)
- E.1B.0-6 containment leak rate testing (2.5.1 and Item II.H of Annex C)
- E.1B.0-7 source term (2.5.2.1, 2.5.2.2, Item I.B of Annex A, and Item I.A of Annex C)
- E.1B.0-8 seismic hazard curves (Item II.C of Annex C)
- E.1B.0-9 leak before break (Item II.D of Annex C)

EPRI Evolutionary Plant SER 1B.1-3

- E.1B.O-10 containment bypass (Item II.G of Annex C)
- E.1B.O-11 prototyping (Item II.K of Annex C)
- E.18.0-12 reliability assurance program (Item II.M of Annex C)
- E.1B.0-13 defense against common-mode failures in digital instrumentation and control systems (Item A of Annex D)
- E.1B.O-14 analysis of external events beyond the design basis (Item B of Annex D)
- E.1B.0-15 control room annunciator reliability (Item G of Annex D)

#### Confirmatory Issues

None

## 1.5 Vendor- or Utility-Specific Items

The vendor- or utility-specific itemr, with references to appropriate sections of this appendix given in parentheses, are listed below. The designators in front of each issue provide a unique identifier for each issue. The letter "E" indicates that the issue applies to the evolutionary plant designs. The first number designates the chapter in which it is identified, and the letter that follows designates the appendix. The letter "V" designates that it is a vendor- or utility-specific item. The final number is the sequential number assigned to it.

- E.1B.V-1 compliance of FDA/DC applications with Commission's regulations and guidance (1.3)
- E.1B.V-2 issue resolution for FDA/DC reviews (1.3)
- E.1B.V-3 elimination of missile provisions (2.1.2)
- E.18.V-4 dynamic seismic analysis of main steam piping and condenser (2.3.1.1 and Item II.E of Annex C)
- E.1B.V-5 main steamline classification (2.3.1.1 and Item II.E of Annex C)
- E.1B.V-6 seismic analysis and plant walkdown of turbine building (2.3.1.1 and Item II.E of Annex C)
- E.1B.V-7 plateout considerations for main steam piping and valves (2.3.1.2 and Item III.F of Annex C)
- E.1B.V-8 reactor pressure vessel level instrumentation system (2.4.1)
- F 18.V-9 source term (2.5.2.1, 2.5.2.2, Item I.B of Annex A, and Item I.A of Annex C)
- E.1B.V-10 compliance with Branch Technical Position MTEB 6.1 (2.5.2.2)
- E.18.V-11 fission product cleanup analysis (2.5.2.2)

- E.1B.V-12 deletion of charcoal adsorbers (2.5.2.2)
- E.1B.V-13 dedicated containment vent penetration (2.5.3 and Item I.K of Annex C)
- E.1B.V-14 decoupling of operating-basis earthquake (OBE) from safe shutdown earthquake (SSE) in seismic design of structures (Generic Safety Issue A-40) (3.2.7)
- E.1B.V-15 deletion of OBE damping values in Leismic design of structures (Generic Safety Issue A-40) (3.2.7)
- E.1B.V-16 use of algebraic sum method for modal combination of high-frequency modes for vibratory loads (Generic Safety Issue A-40) (2.2.7)
- E.1B.V-17 use of spectral peak hifting techniques in lieu of spectral broadening (Generic safety Issue A-40) (3.2.7)
- E.1B.V-18 plant-specific design and arrangement of control systems (Generic Safety Issue A-47) (3.2.9)
- E.1B.V-19 conformance to 10 CFR 50.34(f) hydrogen control requirements (Generic Safety Issues A-48 and 121) (3.2.10 and 3.2.45)
- E.1B.V-20 reliability of emergency diesel generators (Generic Safety Issue B-56) (3.2.14)
- E.1B.V-21 resolution of Generic Safety Issues 2 and 110 (3.2.18 and 3.2.42)
- E.1B.V-22 resolution of Generic Safety Issue 15 (3.7.1%)
- E.1B.V-23 independent reactor coolant pump seal cooling during station blackout (Generic Safety Issue 23) (3.2.20)
- E.1B.V 24 resolution of Generic Safety Issue 24 (3.2.21)
- E.1B.V-25 design details on threaded fasteners (Generic Safety Issue 29) (3.2.22)
- E.1B.V-26 reduction of biofouling in open-cycle service water and component cooling water systems (Generic Safety Issue 51) (3.2.23)
- E.1B.V-27 resolution of Generic Safety Issue 57 (3.2.24)
- E.18.V-28 resolution of Generic Safety Issue 73 (3.2.26)
- E.1B.V-29 equipment classification and vendor interface for reactor trip system components (Generic Safety Issue 75) (3.2.27)
- E.1B.V-30 2-week requirement for corrective maintenance (Generic Safety Issue 75) (3.2.27)
- E.1B.V-31 preventive maintenance and surveillance program for reactor trip breakers (Generic Safety Issue 75) (3.2.27)

EPRI Evolutionary Plant SER 18.1-5

- E.1B.V-32 resolution of Generic Safety Issue 76 (3.2.28)
- I.1B.V-33 cooldown rate in natural convection cooldown analysis (Generic Safety Issue 79) (3.2.29)
- E.1B.V-34 low-density storage racks in spent fuel pool for most recently discharged fuel (Generic Safety Issue 82) (3.2.30)
- E.1B.V-35 plant-specific design and arrangement for control room heating, ventilating, and air conditioning (HVAC) system (Generic Safety Issue 83) (3.2.31)
- 2.18.V-36 design of emergency filter units (Generic Safety Issue 83) (3.2.31)
- E.1B.V-37 design details for control room capacity following a design-basis accident (Generic Safety Issue 83) (3.2.31)
- E.1B.V-38 design details for control room HVAC systems in the smoke removal mode (Generic Safety Issue 83) (3.2.31)
- E.1B.V-39 resolution of Generic Safety Issue 87 (3.2.33)
- E.1B.V-40 adequacy of low-temperature overpressure protection design (Generic Safety Issue 94) (3.2.34)
- E.1B.V-41 adequacy of BWR water level redundancy (Generic Safety Issue ' ) (3.2.37)
- E.1B.V-42 interfacing system design details (Generic Safety Issue 105) (3.2.39)
- E.1B.V-43 inservice testing programs and technical specifications for appropriate pressure isolation valves (Generic Safety Issue 105) (3.2.39)
- E.1B.V-44 resolution of Generic Safety Issue 106 (3.2.40)
- E.1B.V-45 environmental qualification and inservice inspection and testing of large-bore hydraulic snubbers (Generic Safety Issue 113) (3.2.43)
- E.1B.V-46 use of prestressed concrete containments (Generic Safety Issue 118) (3.2.44)
- E.1B.V-47 reliability, operability, and on-line testability of protection system final actuation contacts (Generic Safety Issue 120) (3.2.45)
- E.18.V-48 operator training program and emergency operating procedures related to initiating feed-and-bleed cooling (Generic Safety Issue 122.2) (3.2.50)
- E.1B.V-49 auxiliary feedwater analyses (Generic Safety Issue 124) (3.2.52)

E.1B.V-50 operational aspects of electrical power reliability (Generic Safety Issue 128) (3.2.56)

E.13.V-51	resolution of Generic Safety Issue 130 (3.2.57)
Ē.1B.V-52	resolution of Generic Safety Issue 132 (3.2.58)
E.1B.V-53	resolution of Generic Safety Issue 135 (3.2.59)
E.1B.V-54	resolution of Generic Safety Issue 142 (3.2.60)
E.18.V-55	resolution of Generic Safety .ssue 143 (3.2.61)
E.1B.∛-56	resolution of Generic Safety Issue 151 (3.2.62)
E.1B.V-57	assessment of safety service water system failure modes and contri- butions to core damage frequency and identification of dominant accident sequences (Generic Safety Issue 153 (3.2.63)
E.18.V-58	resolution of Generic Safety Issue HF 4.4 (3.2.64)
E.1B.V-59	resolution of Generic Safety Issue HF 5.1 (3.2.65)
E.18.V-60	resolution of Generic Safety Issue HF 5.2 (3.2.66)
### 2 PLANT OPTIMIZATION SUBJECTS

Plant optimization subjects are proposals initiated by EPRI to deviate from regulatory requirements. EPRI proposes to resolve these issues by providing technically supportable alternatives to current regulatory requirements. Table 1.2 of Volume 1 of this report lists EPRI's proposed plant optimization subjects and their applicability to the evolutionary and passive plant designs. These issues are identified for the evolutionary plant design in Section 2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document. The staff's evaluation is given in this section under the corresponding sections in Appendix B to Chapter 1 of the Evolutionary Requirements Document. The regulatory departure analyses discussed in Section 4 and provided in Annexes A through C of this appendix provide additional information on these subjects.

### 2.1 Issues Related to Chapter 1 of the Evolutionary Requirements Document

#### 2.1.1 Operating-Basis Earthquake and Dynamic Analysis Methods

In Section 2.3.C of Revision 0 of Chapter 1 of the Evolutionary Requirements Document, EPRI proposed to define the peak ground acceleration of the operating-basis earthquake (OBE) to be one-third that of the safe shutdown earthquake (SSE). In Section 2.3.C of the DSER for Chapter 1, the staff stated that it had not completed its review of this matter. As a result of discussions with the staff and the Advisory Committee on Reactor Safeguards, EPRI modified its position. In Section 2.1.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI now proposes to eliminate the OBE as a design-basis event and to use alternative criteria for dynamic analysis methods. In SECY-90-016, "Evolutionary Light Water Reactor Certification Issues and Their Relationship to Current Regulatory Requirements," dated January 12, 1990, and the draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactor and Their Relationship to Current Regulatory Requirements," issued on February 27, 1992, the staff stated that it agreed that the OBE should not control the design of safety-related systems. As a result, the staff is in the process of rulemaking for Appendix A to 10 CFR Part 100 to decouple the OBE from the SSE in siting considerations. The staff is also evaluating the possibility of redefining the OBE in order to satisfy the OBE's function without explicitly analyzing responses. This change would diminish the role of the OBE in design by establishing a level that, if exceeded, would require that the plant be shut down for inspection activities. The staff agrees in principle with EPRI regarding the deletion of the OBE from plant design.

10

The implementation of this optimization subject will have a broad impact on many technical issues and will result in changes to many existing staff positions. For example, Section 2.1.1 of Appendix B to Chapter 1 suggests changes to Appendix A to 10 CFR Part 100, 13 regulatory guides, and 12 sections of the Standard Review Plan (SRP, NUREG-0800). Most of the technical issues pertaining to this optimization subject have been addressed by the staff as a part of its review of Section 4 of Chapter 1 of the Evolutionary Requirements Document. As discussed in various parts of Section 4 of Chapter 1 of this report, the staff is not in complete agreement with the Evolutionary Requirements Document on several of the dynamic analysis issues. In addition, in Section 4.4.3 of Chapter 1 of this report, the staff states that

it is still evaluating the impact of the elimination of the OBE from the design process. Therefore, because of the broad nature of this plant optimization subject and the lack of complete agreement between the staff and EPRI on many of the issues involved, the staff concludes that the information in Section 2.1.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document is not completely acceptable. Although the staff has identified its position on this matter in the draft Commission paper issued on February 27, 1992, its approach to resolving this issue has not been reviewed by the Commission and, therefore, does not represent an agency position. Therefore, the staff regards this is an open issue that must be satisfactorily resolved before it can complete its review of the Evolutionary Requirements Document. The staff's review of applicable issues is contained in Section 4 of Chapter 1 of this report. This optimization subject is discussed further in this report in Section 4.4.3 of Chapter 1 and under Item IV.A of SECY-90-016 contained in Annex A, Item I.M of the draft Commission paper issued on February 27, 1992, contained in Annex C, and Item C of the draft Commission paper issued on July 6, 1992, contained in Annex D of this appendix.

The concerns about the use of the OBE in the design that were raised by the staff in its letter dated April 24, 1991, are no longer applicable to its review of the Evolutionary Requirements Document because of its position regarding this optimization subject. However, as discussed in the DSER dated September 1987 for Chapter 1, certain issues related to the treatment of earthquake cycles for piping and equipment fatigue evaluations, seismic anchor motion effects, and concrete structure design also need to be resolved.

EP. I also proposes to reduce the conservatism in current seismic analysis techniques such as the damping values, algebraic combination of high-frequency modes, broadening of the response spectrum, spectral shifting analysis technique, multiple response spectrum analysis technique, and classification of piping stresses. Most of these issues are discussed in this report (see Chapter 1).

In its letter dated April 24, 1991, the staff questioned the appropriateness of classifying seismic stress as secondary stress. EPRI agreed that seismic stress was not self-limiting and was a load-controlled stress. In Revision 3 of the Evolutionary Requirements Document, EPRI deleted the sentence classifying seismic stress as secondary stress. The staff concludes that this modification is acceptable and this issue is closed.

Revision 3 of Section 2.1.1 of Appendix B to Chapter 1 indicates that ASME Code Cases N-451 and N-462 provide alternative rules on allowable seismic stresses for Class 1 and Class 2/3 piping. It also states that these code cases have not been accepted by the staff and that the ASME Code Committee is implementing changes to the code. However, until these code cases or the changes in the ASME Code are accepted by the staff, current rules acceptable to the staff should be followed. The staff's position on the applicability of industry codes and standards for the ALWR designs is provided under Item II.A of the draft Commission paper issued on February 27, 1992, contained in Annex C of this appendix. Basically, the issue of allowable seismic stresses for systems and components is not associated with civil structures. Although the staff has identified its position on this matter in the draft Commission papers of February 27 and July 6, 1992, its approach to resolving this issue has not been reviewed by the Commission and, therefore, does not represent an agency position. This is an open issue that must be satisfactorily resolved before the staff can complete its review of the Evolutionary Requirements Document

### 2.1.2 Tornado Design

Section 2.1.2 of Appendix B to Chapter 1 and Section 4.4.3.3.8 of Chapter 1 of the Evolutionary Requirements Document state that the plant designer will use American National Standards Institute/American Nuclear Society (ANSI/ANS) 2.3, "Standard for Estimating Tornado and Extreme Wind Characteristics at Nuclear Power Sites," to define tornado effects, using a probability of 10E-6 per year. This is an exception to Regulatory Guide 1.76, "Design Basis Tornado for Nuclear Power Plants." The staff's evaluation of this issue is given in this report in Section 4.5.2 of Chapter 1 and under Item II.F of the draft Commission paper contained in Annex C of appendix. Although the staff has identified its position on this matter in the draft Commission paper of February 27, '992, its approach to resolving this issue has not been reviewed by the Commission and, therefore, does not represent an agency position. This is an open issue that must be satisfactorily resolved before the staff can complete its review of the Evolutionary Requirements Document.

## Elimination of Missile Provisions

Section 2.1.2.3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that certain missile provisions, such as labyrinths in neating, ventilating, and air conditioning inlets and exhausts will be eliminated. EPRI states this elimination will lower construction costs and enhance maintenance.

Regulatory Guide (RG) 1.117, "Tornado Design Classification," and SRP Section 3.5.2, "Structures, Systems, and Components To Be Protected From Exter-nally Generated Missiles," describe a method acceptable to the staff for identifying those structures, systems, and components that should be protected from the effects of the design-basis tornado, including tornado missiles. Depending on the nature and source of the externally generated missiles. protection may be provided by missile barriers for individual components, by placing independent redundant systems in compartments located in a missileprotected structure, or by a subgrade location at a sufficient depth. The staff concludes that the EFRI-proposed elimination of certain missile provisions may be acceptable, provided sufficient justification is provided on a case-by-case basis regarding compliance with RG 1.117 and SRP Section 3.5.2. The staff concludes that EPRI's generic justification of reduction in cost and maintenance is not sufficient for eliminating missile-protection measures. The staff will evaluate this matter during its review of an individual application for FDA/DC.

# 2.2 Issues Related to Chapter 2 of the Evolutionary Requirements Document

There were no plant optimization subjects associated with Chapter 2 of the Evolutionary Requirements Document.

### 2.3 Issues Related to Chapter 3 of the Evolutionary Requirements Document

2.3.1 BWR Main Steamline Isolation Valves and Leakage Control and Classification of Main Steamline of Boiling-Water Reactor

In Section 2.3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI proposes to eliminate the BWR main steam isolation valve leakage control system and provide an alternative leakage pathway (i.e., the main steamline and the condenser) to the main condenser downstream of the isolation valves in the event of a loss-of-coolant accident (LOCA). This issue is also related to Section 3.3.2 in Appendix B to Chapter 1, Section 3.4.1.5 in Chapter 2, Sections 5.3.3 and 5.4.1.5 in Chapter 3, Section 1.2.3 in Chapter 5, and Section 3 in Chapter 13 of the Evolutionary Requirements Document. In its letters dated May 17 and August 29, 1991, the staff stated its positions on the quality group and seismic classifications of the BWR main/extraction steam system, main steamline, main turbine system, condenser, and related non-seismic structures that resulted from the proposed elimination of the main steam isolation valve leakage control system. The objective of these staff positions is to establish adequate measures to ensure that the above systems and structures will maintain their structural integrity during and following an SSE. Although this issue involves information in Chapters 1, 2, 3, 5, 6, and 13, for the sake of continuity, the basis for the staff's positions and the staff's evaluation of all of EPRI's responses are discussed below.

### Main Steamline Classification

### Background

The main steamlines in BWR plants contain dual quick-closing main steam isolation valves (MSIVs). These valves function to isolate the reactor system in the event of a break in a steamline outside the primary containment, a design-basis LOCA, or other events requiring containment isolation. Although the MSIVs are designed to provide a leaktight barrier, it is recognized that some leakage through the valves will occur. The current technical specification limi' for MSIV leakage is typically 11.5 standard cubic feet per hour (scfh) rar valve. Operating experience indicates that degradation has occasicnally occurred in the leaktightness of MSIVs and the specified low leakage has not always been maintained.

Because of recurring problems with excessive leakage of MSIVs, the staff recommended in RG 1.96, "Design of Main Steam Isolation Valve Leakage Control Systems for Boiling Water Reactor Nuclear Power Plants," the installation of a supplemental leakage control system (LCS) to ensure that the isolation function of the MSIVs is in accord with the specified limits.

In response to the MSIV leakage concerns, the BWR Owners Group (BWROG) commissioned a program of studies to determine the causes of high leakage rates and the means to eliminate them. The results of these studies were submitted to the NRC in General Electric proprietary reports NEDO-31643P (dated November 1988) and NEDO-31858P (dated February 1991), both entitled "Increasing Main Steam Isolation Valve Leakage Rate Limits and Elimination of Leakage Control Systems."

In addition, the NRC has recently reviewed the desirability of the MSIV LCS requirement. Because the LCS may not be effective for controlling MSIV leakage rates greatly in excess of technical specification limits because of limitations in its design, the staff established Generic Issue C-8, "Main Steam Line Valve Leakage Control Systems," to address these concerns for currently operating plants.

The staff's position on the EPRI-proposed resolution for both evolutionary and passive BWR designs follows. It is based on the assumed capability of the main steam piping (including its associated piping to the condenser) and the condenser to remain structurally intact so they can act as a holdup volume for fission products.

### Evaluation

The MSIVs do not provide a leaktight containment pressure boundary as intended in the plant design. Although substantial progress has been made in recent years to identify the causes of the leakage and to reduce the amount of leakage, the current typical technical specification limit of 11.5 scfh per valve is still difficult to achieve when the valve is rapidly closed against full-flow conditions at full reactor pressure and temperature.

The current assumption for operating plants in calculating the dose reference values specified in 10 CFR Part 100 is based on a conservative assumption that the leakage allowed by the technical specifications of 11.5 scfh per valve is released directly into the environment. No credit is currently taken for the pressure integrity of the main steam piping and condenser.

The proposed approach developed by EPRI and the BWROG would allow higher leakage limits and would take credit for the main steam piping and condenser to plate out the fission products following core damage. In this way, the main steam piping (and its associated piping) and the condenser would be used to mitigate the radiological consequences of an accident that could result in potential offsite exposures comparable to the dose reference values in 10 CFR Part 100.

Seismic Qualification: Appendix A to 10 CFR Part 100 requires that structures, systems, and components necessary to ensure the capability to mitigate the radiological consequences of accidents that could result in exposures comparable to the dose guideline exposures of Part 100 be designed to remain functional during and after a safe shutdown earthquake (SSE). Thus, the main steamline, portions of its associated piping, and the main condenser are required to remain functional if the SSE occurs. As such, these components are required to be analyzed to demonstrate their capability to maintain their structural integrity. Appendix A to 10 CFR Part 100 requires that the engineering method used to ensure that the safety functions are maintained during and after an SSE involve the use of either a suitable dynamic analysis or a suitable qualification test.

Non-Nuclear Experience: In Appendix D to NEDO-31858P, BWROG submitted a report to the staff entitled "Performance of Condensers and Main Steam Piping in Past Earthquakes," Report No. 50032.02-R-01 dated September 1990 by EQE Engineering. The report provided a summary of data on the performance of nonseismically designed main steam piping and condensers in past earthquakes

around the world. These data were used to evaluate the capability of main steam piping and condensers in typical General Electric (GE) Mark I, II, and III BWRs in the United for to withstand design-basis earthquakes.

The EQE report showed that piping and condensers exhibited substantial seismic ruggedness even when they were not designed for earthquake loadings. No failures of main steam piping were found. The report concluded that (1) the possibility of significant failure of GE BWR main steam piping or condensers in the event of a design-basis earthquake in the Eastern United States was highly unlikely and (2) any such failure would be contrary to the large body of historical earthquake experience data.

As a result of its review of the EQE report for condensers, the staff concludes that, in general, the experience data base is applicable to BWR condensers in U.S. plants. Overall, condensers in nuclear plants are similar to condensers in non-nuclear plants. However, because the pipe wall thickness is typically about 1 inch for main steam piping in BWRs, whereas the pipe wall thickness is typically much greater than 1 inch, and as much as 4 inches, for main steam piping in fossil plants, the staff concludes that the generic study for the main steam piping is not appropriate for BWRs.

Piping Failure Modes: The staff recognizes that welded piping systems exhibit seismic ruggedness even under loadings several times greater than a designbasis earthquake loading. The staff has also investigated the actual failure modes of piping due to earthquake loadings. In NUREG-1061, Volume 2 Addendum, "Summary and Evaluation of Historical Strong-Motion Earthquake Seismic Response and Damage to Above-Ground Industrial Piping," dated April 1985, the staff found that the observed failure of piping can be divided into three primary categories:

- (1) failure of pipe due to excessive displacement of attached equipment
- (2) failure of branch piping due to excessive displacement of attached piping mains
- (3) failure of piping associated with loss of pipe supports

Another observed category of pipe failure (beyond the scope of NUREG-1061) is that caused by the failure of the supporting or enclosing building which, in turn, causes the failure of the pice.

Classification of Main Steamline of BWRs: Because of the lack of evidence provided by the EQE report to support generic acceptability of the BWR main steam piping to withstand an SSE, the staff concludes that seismic analyses are required to ensure that the failure modes described above will not occur in BWR main steam piping. The staff also concludes that no undue burden exists for the ALWRs to require its main steam piping and branch lines to the first isolation valve to be classified as safety-related, seismic Category I piping with the appropriate quality assurance requirements imposed by Appendix B to 10 CFR Part 50 before the plant is constructed. Appendix A to SRP Section 3.2.2, "System Quality Group Classification," provides guidelines on the safety classification of BWR main steam piping. The SRP recommends that the main steamline from the second isolation valve up to, but not including,

the turbine stop valve including branch lines to the first valve be classified as Quality Group B (Safety Class 2). RG 1.29, "Seismic Design Classification," designates such piping as seismic Category I.

The staff's position is that the main steam piping from the outermost isolation valve up to the seismic interface restraint and branch lines up to the first closed valve must conform to Appendix A to SRP Section 3.2.2 and RG 1.29. The main steamline from the seismic interface restraint up to, but not including, the turbine stop valve (including branch lines to the first normally closed valve) must be classified as Quality Group B and inspected in accordance with applicable portions of ASME Code, Section XI, "Inservice Inspection," but may be classified as non-seismic Category I if it has been analyzed using a dynamic seismic analysis to demonstrate its structural integrity under SSE loading conditions. However, all pertinent quality assurance requirements of Appendix B to 10 CFR Part 50 are applicable to ensure that the quality of the piping material is commensurate with its importance to safety during both operational and accident conditions.

The seismic interface restraint must provide a structural barrier between the seismic Category I portion of the main steamline in the reactor building and the non-seismic Category I portions of the main steamline in the turbine building. The seismic interface restraint will be located inside the seismic Category I building. The main steamline in the turbine building must be classified as non-seismic Category I to be consistent wit, the classification of the turbine building. Therefore, the quality and safety requirements proposed by the staff for the main steamline from the outermost isolation valve up to the turbine stop valve are equivalent to the staff guidelines in Appendix A to SRP Section 3.2.2 and RG 1.29

<u>Classification of Piping Associated With the Main Steamline</u>: The BWROG has identified several interconnected alternative leakage pathways to allow MSIV leakage to drain to the condenser. Of these, the BWROG has proposed to use the drain line downstream of the MSIVs to provide a non-safety-grade leakage pathway to the condenser. In accordance with the BWROG recommendations, Section 2.3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document proposes the use of the main steamline drains for MSIV leakage control.

However, GE has indicated that the bypass piping of the main steamline is another appropriate leakage pathway that the designer may use for MSIV leakage control. The staff concludes that use of either pathway is an acceptable approach to resolving this issue, provided the chosen pathway is seismically analyzed to withstand SSE loading conditions, as discussed below. In addition, because these pathways are interconnected, the designer will have to seismically qualify all pathways up to the first normally closed valve to ensure this piping maintains its integrity during an SSE.

For ensuring the integrity of the MSIV leakage pathway (i.e., drain line or bypass line) from the first valve to the main condenser hotwell, the staff and EPRI both agree that preventing structural failure of the piping and hotwell would provide assurance that leakage from the MSIVs following a design-basis accident would not result in potential offsite exposures that exceed the dose guidelines of 10 CFR Part 100. The staff's position regarding the classification of the MSIV leakage pathway between the first normally closed valve and the condenser hotwell and the hotwell itself is as follows.

The staff proposes that (1) the MSIV leakage pathway from the first valve up to the condenser inlet and (2) the piping between the turbine stop valve and the turbine inlet not be classified as safety related or as seismic Category I, but should be analyzed using a dynamic seismic analysis to demonstrate their structural integrity under SSE loading conditions.

EPRI's proposal to use the seismic experience data base in lieu of a seismic analysis for the line from the first valve of the MSIV leakage pathway to the condenser is not consistent with the staff's position discussed above and is, therefore, not acceptable. Although this portion of the MSIV leakage pathway does not have to be seismic Category I, the experience data base does not provide adequate assurance of seismic integrity for such piping. In addition, EPRI's proposal to use the simplified seismic analysis procedure described in RG 1.143, "Design Guidance for Radioactive Waste Management Systems, Structures, and Components Installed in Light-Water-Cooled Nuclear Power Plants," is not acceptable for the first valve of the MSIV leakage pathway up to and including the condenser. To satisfy Item II.k of SRP Section 3.9.2, "Dynamic Testing and Analysis of Systems, Components, and Equipment," this piping and the condenser should be analyzed to the SSE loading condition using the same criteria as those that are applicable to seismic Category I items. The guidelines in RG 1.143 allow a simplified analysis using only the OBE loading conditions and are unacceptable. Therefore, the staff will evaluate this piping and the condenser during its review of an individual application for FDA/DC in accordance with the positions in Appendix A to SRP Section 3.2.2 and Regulatory Guide 1.29.

<u>Turbine Stop Valve and Associated Piping and Structures</u>: The staff's evaluations of EPRI's responses relative to the turbine stop valve, the piping between the stop valve and the turbine, and the structures associated with this issue are given in Section 3.1.1 of Chapter 13 of this report.

Condenser Design: Because the Evolutionary Requirements Document docs not address the condenser design details and no ALWR plant has been constructed, the staff cannot evaluate the applicability of the generic seismic experience data base for the ALWR condenser. Thus, it requires that the condenser be stismically analyzed by the combined license applicant to ensure that it is capable of maintaining its structural integrity during and anter an SSE. The condenser is not required to be classified as seismic Category I or as a safety related component. The staff concludes that the additional safetyrelated and seismic category I requirements (e.g., the quality assurance requirements in Appendix R to 10 CFR Part 50 or the inservice inspection requirements in the ASME Code) are excessive and unnecessary for ensuring the structural integrity of the condenser under seismic loadings with no significant increase in safety if they were to be imposed. However, applicants seeking to take advantage of the main steam piping, its associated piping, and the condenser as an alternative leakage path will be required to submit a request for an exemption from the quality assurance requirements of Appendix B to 10 CFR Part 50 for the MSIV leakage pathway and the main condenser.

<u>Turbine Building Design</u>: As discussed in Section 3.1.1 of Chapter 13 of this report, the turbine building should be seismically analyzed for the SSE to ensure that it will not adversely affect the structural integrity of the main steam piping, its associated piping, and the main condenser. The staff requires that plant-specific walkdowns be conducted before commercial operation to assess the potential failures of non-seismically designed systems,

structures, and components overhead, adjacent to, and attached to the main steam piping, its associated piping, and the condenser. For the walkdowns, an appropriate seismic experience data base should be used for identifying potential failure modes in order to provide high confidence that the main steam piping, its associated piping, and the main condenser in ALWRs can retain their structural integrity during and following an SSE.

### Conclusion

The approach described above for ALWRs to resolve the BWR main steamline issue provides reasonable assurance that the main steam piping from the outermoisolation valve up to the turbine stop valve, the MSIV leakage pathway (i.e., the drain line or bypass line) up to the condenser, and the main condenser will remain structurally intact, so that they can act as a holdup volume for fission products during and following an SSE. The final closure of the BWR main steamline issue will be verified through clant-specific walkdowrs during the combined license phase. Although the staff has identified its position on this matter in the draft Commission paper issued on February 27, 1992, its approach to resolving this issue has not been reviewed by the Commission and, therefore, does not represent an agency position. This is an open issue that must be satisfactorily resolved before the staff can complete its review of the Evolutionary Requirements Document. This issue is also discussed under Item II.E of the draft Commission paper of February 27, 1992, contained in Annex C of this appendix.

### Plateout

Plateout of radioactive iodine on the main steam pipe and condenser surfaces can result in significant dose mitigation. Several technical references indicate that particulate and elemental iodines would be expected to deposit on surfaces, and the rates of deposition would vary with temperature, pressure, gas composition, surface material, and particulate size. The staff is evaluating the extent to which credit for the fission product attenuation in the main steamlines and in the isolation condenser is appropriate and reasonable for BWRs even though the main steamlines downstream from the MSIV and its condenser are not designed to withstand the SSE as defined in Sect on III.C of Appendix A to 10 CFR Part 100. Since product attenuation in the main steamline and in the isolation condenser depends on the design of the plant, the staff will evaluate this issue during its review of an individual application for FDA/DC.

### 2.3.2 Simplification of Postaccident Sampling System

The regulations and guidance regarding the design of the postaccident sampling system (PASS) are given in 10 CFR 50.34(f)(2)(viii), RG 1.97 ("Instrumentation for Light-Water-Cooled Nuclear Power Plants To Assess Plant and Environs Conditions \_\_ring and Following an Accident"), and NUREG-0737 ("Clarification of TMI Action Plant Requirements"). Section 2.3.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document gives EPRI's proposed deviations from the design requirements for the PASS in several areas. The staff's evaluation of this issue is provided under Item II.1 of the draft Commission paper dated February 27, 1992, contained in Annex C of this appendix. Although the staff has identified its position on this matter in the draft Commission paper, its approach to resolving this issue has not been reviewed subcooling margin monitor and core exit thermocouples, meets the established requirements. The same cannot be said for reliance on pressurizer level indicat As discussed in early versions of Section 2.4.1 of Appendix B to Chapter if the Evolutionary Requirements Document, the pressurizer level increases for breaks in some locations and decreases for breaks in other locations. The positive indication provided by the RVLIS, rather than diverting the operator's attention, may provide the information for the operator to make a proper decision.

The RVLIS has already been shown to be a useful instrument in operating reactors. For example, at North Anna Unit 1, between June 17 and June 21, 1987, approximately 17,000 gallons of reactor coolant system (RCS) inventory was removed from the RCS. The pathway for the loss was through the pump backseat of the "A" reactor coolant pump, up the shaft, and out through the seals into the containment sump. The pump had been decoupled from the motor for stator replacement. The inventory loss was not detected because the pressurizer had been allowed to cool down, a vacuum had developed in the pressurizer, and a bubble had formed in the reactor vessel head. In this condition, pressurizer level indication did not provide an adequate representation of RCS inventory. Indication was available via the RVLIS portion of the integrated core cooling monitoring system. Even though the RVLIS had not been declared operational and operator action to terminate the condition was not taken on the basis of that information, it did alert the operators, and action was finally laken when the vacuum in the pressurizer was also detected. At that time, the RVLIS was used to track the vessel level during venting of the reactor head and refilling of the system.

The staff has also identified other event as ing which the RVLIS (if it had been installed) could have resulted in better operator response. These events included a natural circulation cooldown event at St. Lucie Unit 1, during which a bubble had formed in the reactor vessel head, even though there was continued indication of adequate subcooling margin.

In addition to the staff requirements for vessel level instrumentation as part of an ICC detection system, such instrumentation provides very useful information to be considered during accident-management activities. Although accident-management guidelines are being developed by vendor owners groups, the staff expects that indication of vessel level will be an important parameter for preventing core damage or possibly arresting core-melt progression before vessel failure. Therefore, the staff concludes that all ALWR resigns must have a RVLIS.

In Revision 4, EPRI modified Section 6.3.3.2 of hapter 4 of the Evolutionary Requirements Document to require a RVLIS and proposed general requirements for such a system. The staff concludes that these requirements are acceptable. However, because little specific information on the RVLIS system was provided, the staff requires that each PWR ALWR designer identify system design and performance criteria including the system's potential accident-management role and the resulting severe environment to which it may be subjected. The staff will evaluate the specific design during its review of an individual application for FDA/DC.

-61

# 2.5 Issues Related to Chapter 5 of the Evolutionary Requirements Document

### 2.5.1 Containment Leak Rate lesting

Section 2.5.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document proposes that the maximum interval between Type C leak rate tests be 30 months instead of the 24 months maximum currently required in Appendix J to 10 CFR Part 50. EPRI originally proposed this optimization subject in Section C.1 of Revision O of Chapter 5 of the Evolutionary Requirements Document.

EPRI states that ALWRs will have a refueling cycle of 24 months. Therefore, it believes that a Type C test interval of at least 30 months is needed to avoid the necessity of shutting down the facility solely to perform Type C local leak rate tests. Otherwise, it may be necessary to subject the facility to the increased risks, and p sonnel to the increased exposure, associated with testing at power. EPRI cites the economic benefits of a 24-month refueling cycle, the reduced occupational radiation exposure associated with longer test intervals, and the relatively minor overall effect on risk should the longer test interval result in increased containment leakage. In the latter regard, EPRI cites NUREG/CR-3539, "Impact of Containment Building Leakage on LWR Accident Risk," which states that an increase in containment leakage from 0.1 percent to 1.0 percent every 24 hours would increase the overall risk by 1.5 percent.

In addition, in its letter dated March 3, 1992. EPRI provided hand-marked page changes requesting additional changes to the containment leak rate testing requirements of Appendix J to 10 CFR Part 50 as contained in ANSI/ANS 56.8, "Containment System Leakage Testing Requirements," such as relief from the test interval for air locks. The staff's evaluation of these issues is given in this report under Item II.H of the draft Commission paper dated February 27, 1992, contained in Annex C of this appendix and Section 6.3 of Chapter 5. Although the staff has identified its position on the issue of the 24-month Type C test interval in the draft Commission paper, its approach to resolving this issue has not been reviewed by the Commission and, therefore, does not represent an agency position. This is an open issue that must be satisfactorily resolved before the staff can complete its review of the Evolutionary Requirements Document.

### 2.5.2 Physically Based Source Term

#### General Discussion

Section 2.4.1.2 of Chapter 1 of the Evolutionary Requirements Document requires the use of a more realistic severe reactor accident source term for the offsite radiological consequence assessment than the current source term specified in U.S. Atomic Energy Commission report TID-14844, "Calculation of Distance Factors for Power and Test Peactor Sites," March 23, 1962, and in RGs 1.3, "Assumptions Used for Evaluating the Potential Radiological Consequences of a Loss of Coolant Accident for Boiling Water Reactors," and 1.4, "Assumptions Used for Evaluating the Potential Radiological Consequences of a Loss of Coolant Accident for Potential Radiological Consequences of a Loss of Coolant Accident for Pressuried Water Reactors." Section 2.5.2 of Appendix B to Chapter 1 lists source term treatment for the evolutionary ALWR as a plant optimization subject. In addition to the discussion in these sections, EPRI submitted to the NRC two technical reports callied "Evolutionary Plant Source Term Report (October 1990)" and "Passive ALWR Source Term (February 1991)." In these sections and reports, EPRI points out that ALWR licensing-design-basis requirements as well as design enhancements related to severe accidents should be based on the full body of current knowledge remarding accident source terms. EPRI believes that the evolutionary designs should be evaluated on the basis of a realistic treatment of the fission product source term, including the extensive research that has been done on fission product behavior since TID-14844 was issued, and especially since the accident at Three Mile Island Unit 2 (TMI-2) in 1979. EPRI's view is that this approach will result in designs that are improved and provide enhanced safety protection.

In SECY-89-341, "Updated Light Water Reactor Source Term Methodology and Potential Regulatory Applications," dated November 6, 1989, SECY-90-016, and SECY-90-341, "Staff Study on Source Term Update and Decoupling Siting From Design," dated October 4, 1990, the staff discussed an integrated set of its activities involving regulatory development and implementation of updated source term information in connection with its review of the Evolutionary Requirements Document and the evolutionary ALWR designs. In these papers, the staff also discussed revisions of 10 CFR Parts 100 and 50 thr. will decounle reactor siting from plant design.

Both the NRC and the industry have expended a great deal of effort to obtain a better understanding of the fission product transport and release mechanism following a severe reactor accident. The accident source terms proposed by EPRI are based on a single, enveloping value for a bounding severe reactor accident sequence, using release data obtained from the severe fuel damage tests at the Power Surst Facility, source term measurements at the Loss of Fluid Test (LOFT) Facility, and data from the TMI-2 postaccident examination. The NRC source term developed by the staff is based primarily on source term calculations performed by the Source Term Code Package (STCP) for individual accident sequences selected in NUREG-1150, "Severe Accident Risks: An Assessment for Five U.S. Nuclear Power Plants." The STCP has been demonstrated and has emerged as an integral tool for the analysis of the fission product transport and release mechanism in the reactor coolant system and in the primary containment in ALWRs under severe reactor accident conditions.

The staff finds that there is good agreement between EPRI's proposed source terms for the passive plants and those of the staff in most areas, ex ept for the magnitude of nuclides with low volatility, where the staff's values are about an order of magnitude higher than those proposed by EPRI. However, there is less agreement between EPRI's proposed source term for the evolutionary plants and those of the staff, because EPRI's evolutionary plant source term is based on a partial core melt, whereas the staff's proposed source terms as well as those for EPRI's passive plants are based on a complete core melt. In either case, the staff concludes that the differences between the staff's and EPRI's estimates for the nuclides with low volatility will not have a major effect on offsite dose assessment. In SECY-92-127, "Revised Accident Source Terms for Light Water Nuclear Power Plants," the staff proposed a revised severe accident source term in a draft NUREG report, "In-Containment Accident Source Terms for Light Water Nuclear Power Plants." The draft NUREG report provided the types, timing, and quantities of fission products that would be released into the containment on the basis of a range of core-melt-accident scenarios, including failure of the reactor pressure

vessel and subsequent core-concrete interactions. After SECY-92-127 was written, the staff issued NUREG-1465, "Accident Source Terms for Light-Water Nuclear Power Plants," (June 1992) for a 90-day public comment period. The staff will evaluate the entire source term-related issue during its review of the Requirements Document and individual applications for FDA/DC. This is an open issue that must be satisfactorily resolved before the staff can complete its review of the Evolutionary Requirements Document. See Item 1.8 of SECY-90-016 contained in Annex A and Item I.A of the draft Commission paper dated February 27, 1992, contained in Annex C of this appendix for additional discussion of this issue.

### Specific Source Term Issues

In Section C.2 of the SSER for Chapter 5, the staff discussed several specific issues related to the source term. The following is the staff's evaluation of these issues.

#### Deletion of PWR Containment Spray Additive

In Section 1.2.3.1 of Chapter 5 of the Evolutionary Requirements Document, EPRI notes that previous regulatory practice called for a chemical additive in PWR containment spray solutions to raise the pH of the spray solution to enhance the solubility of elemental iodine in the spray droplets. EPRI also states that recent research shows that the vast majority of volatile fission products released will be in particulate form and that the containment spray additive is unnecessary and should be deleted. Also, EPRI believes that rates for fission product removal by the spray should be established on the basis of current research.

The review of the PWR spray system as a fission product removal system is covered in SRP Section 6.5.2, "Containment Spray as a Fission Product Cleanup System." The staff issued a revisior of SRP Section 6.5.2 in December 1988, in which it acknowledges that a chemical additive is not necessarily required during spray injection, but that pH control should be maintained for the sump solution during postaccident conditions. The deletion of a spray additive for the ALWR is consistent with the revised SRP and is acceptable. Revised SRP Section 6.5.2 provides fission product cleanup models that can be used with either RG 1.4 or current best-estimate fission product releases.

The Evolutionary Requirements Document does not specify chemical additives for long-term c stainment sump pH control. This feature is normally provided by baskets of trisodium phosphate suspended in the sump. Although the proposed design of the ALWR does not include a recirculation sump, chemicals could be readily located in other locations where they would be dissolved by containment spray. In Section C.2.1 of the DSER for Chapter 5, the staff stated that EPRI should address Branch Technical Position (BTP) MTEB 6-1 (SRP Section 6.1.1, "Emergency Safety Features Materials"), which requires containment sump pH control, and identified this as an open issue. The staff will evaluate this matter during its review of an individual application for FDA/DC to ensure that the designer acceptably addresses BTP MTEB 6-1. On the basis of the above discussion, this DSER open issue is closed. The staff's evaluation of this issue can be found in Section 8.2 of Chapter 5 of this report. Revised SRP Section 6.5.2 states that for sump solutions having a pH less than 7, molecular iodine should be conservatively assumed to evolve into the containment atmosphere. In Section C.2.1 of the DSER for Chapter 5, the staff stated that EPRI should include a requirement that the ALWR fission product cleanup analyses reflect the effect of reduced pH consistent with the methodology in revised SRP Section 6.5.2 and identified this as an open issue. The staff will evaluate this matter during its review of an individual application for FDA/DC to ensure that the effect of reduced pH is properly reflected in the cleanup analysis. On the basis of the above discussion, this DSER open issue is closed.

### Deletion of Charcoal Adsorbers

In Section 1.2.3.2 of Chapter 5 of the Evolutionary Requirements Document, EPRI states that activated charcoal filters in BWR standby gas treatment systems and other PWR ventilation systems are required for many areas in which fission products could be released. It also states that these filters are required solely for removing organic and elemental iodine, are complex, and have been difficult to operate and test. EPRI states that recent research shows that the amount of elemental iodine expected to be released in an accident is small enough that charcoal filtration is unnecessary, and that the reference to charcoal filters has been deleted in the Evolutionary Requirements Document with regard to fission product filtration systems.

In SECY-86-76, the staff identified air filtration systems as an area of p.tential change based on source term research. Recent research indicates that the fraction of fission product iodine present in elemental and organic forms may be lower than that specified in present regulatory guidance (RGs 1.3 and 1.4), but may still be at a level at which some charcoal filtration is warranted. Further, this research also indicates that the quantity of airborne material (radioactive and nonradioactive aerosols) expected to be produced in accident sequences of interest could be large in comparison to the retention capacity of the high-efficiency particulate air filters in present engineered safety features air filtration systems. Finally, many filter systems are also used to process activity released during normal operation during which charcoal may be useful. The staff intends to consider potential changes in air filtration systems based on the proposed revised source term discussed in SECY-92-127, but concludes that the complete removal of charcoal in air filtration systems, as stated in EPRI's position, is currently unjustified. In particular, the staff notes that there are large uncertainties with regard to gaseous iodine production due to pH changes in the in-containment refueling water storage tank, hydrogen burns, and irradiation of other chemical forms of iodine. Therefore, in Section C.2.2 of the DSER for Chapter 5, the staff concluded that the need for activated charcoal filters in appropriate ALWR ventilation systems must be evaluated as part of the overall development of the updated analytical methodology for the source term to be used for ALWRs and identified this as an open issue. The staff will evaluate this matter during its review of an individual application for FDA/DC. On the basis of the above discussion, this DSER open issue is closed.

#### BWR Suppression Pool Fission Product Scrubbing

In Sections 1.2.3.3 and 7.3.11 of Chapter 5, EPRI states that previous licensing practice did not recognize scrubbing of fission products by a BWR suppression pool and that recent research has shown such scrubbing to be

effective. EPRI originally took the position that a decontamination factor for radionuclide scrubbing by BWR suppression pools should be credited on the basis of evaluations of accident sequences using the MAAP code and decontamination factors determined by SUPRA, SPARC, or other technically defensible methods.

RG 1.3 (Position C.1.f) allows no credit for retention of iodine by a BWR suppression pool. However, as discussed in Section C.2.3 of the DSER for Chapter 5, the staff has issued a revised SRP Section 6.5.5, "Pressure Suppression Pools as Fission Product Cleanup Systems," dated December 1988. The essential feature of this revised section is that it recognizes that suppression pools are capable of scrubbing non-noble gas fission products and it would be an undue conservatism to ignore this capability. On the basis of the revised SRP Section 6.5.5, the staff concluded that credit may be given for suppression pool fission product removal, provided suppression pool decontamination factors are evaluated in accordance with the methodology prescribed in the revised SRP section and identified this as an open issue.

The staff concludes that this criterion is reflected in Table B.1-2 of Appendia B to Chapter 1 of the Evolutionary Requirements Document. In addition, Section 1.2.3.3 of Revision 1 of Chapter 5 has been modified to delete the reference to specific alternative methodologies. A statement to use SRP Section 6.5.5 methodologies was added to Section 7.3.11 of Chapter 5. These changes are consistent with the staff's position and are acceptable. Therefore, this OSER open issue is closed.

#### Timing of Fission Product Release

In current licensing practice, the assumption is that design-basis-accident fission product releases to the containment atmosphere occur virtually instantaneously. EPRI claims this assumption results in closure times of containment isolation valves that are shorter than necessary and capacities of ventilation fans that are larger than necessary, on the basis of analysis of sequences expected to dominate the likelihood of core damage. In Section 1.2.3.4 of Chapter 5 of the Evolutionary Requirements Document, EPRI takes the position that an accidental release of fission products into the containment should be assumed to occur no sooner than about 1 hour after reactor scram.

The basis for this assumption is given in EPRI's technical reports entitled "Evolutionary Plant Source Term Report (October 1990)" and "Passive ALWR Source Term (February 1991)." These reports state that the initial fission product release from the reactor coolant system to the containment atmosphere (gap activity) occurs more than 30 minutes after scram for existing plants. The reports further state that allowing for the enhancements provided by the ALWF design requirements, significant radioactive release (fuel activity) to the containment will occur no sooner than about 60 minutes after reactor scram for a plant designed to meet the Evolutionary Requirements Document.

For licensing purposes, timing of accident fission product releases into the containment is given by Positions C.1.a of RGs 1.3 and 1.4. These positions indicate that fission product releases should be assumed to be "immediately" available for leakage from the containment. In practice, the staff has typically taken this to mean "within 15 seconds" in order to allow for closure of containment isolation valves. The assumption of the release of the

quantity of fission products contemplated by RGs 1.3 and 1.4 within about 15 seconds is generally recognized as highly conservative for many accident sequences. This timing appears to be dependent on the accident sequence as well as the reactor type. Furthermore, the staff has not, at this time, decided whether certain sequences should be excluded from consideration because they are unlikely to dominate the overall likelihood of core damage. For these reasons, the staff believes that some relaxation in the assumed timing of large fission product releases into the containment associated with fuel degradation and melting may be warranted, but cannot support the specific value (1 hour after scram) proposed in the Evolutionary Requirements Document. In SECY-92-127, the staff indicated that the revised source term may make it possible to relax the closure time of certi n containment isolation valves on the basis of the longer time interval before fuel failure. It may also make it possible for plants with secondary containments (e.g., BWRs) to relax the time needed to attain a negative pressure within the secondary containment because initial fission product releases from the primary containment will be relatively small. The staff regards the issue of timing of the fission product release to be part of the overall open issue regarding the source term. The staff will evaluate this matter during its review of the Requirements Document and individual applications for FDA/DC.

2.5.3 Dedicated Containment Vent Penetration

In Section 2.5.3 of Appendix B to Chapter 1, EPRI states that it has provided a criterion for protecting the containment from long-term overpressurization solely through the containment design or, alternatively, the use of pressure relicf.

In SECY-90-016, the staff recommended that the Commission approve the use of an overpressure protection system that uses a dedicated containment vent for the Advanced BWR (ABWR) to avoid gross containment failure resulting from postulated slow-rising-overpressure scenarios that could result from postulated multiple safety system failures. In its staff requirements memorandum of June 26, 1990, the Commission approved the use of the containment overpressure protection system for the ABWR, subject to a comprehensive regulatory review to weigh the "downside" risks with the mitigation benefits of the system. In addition, the Commission directed the staff to ensure that full capability to maintain control over the venting process is provided in the design.

In Section 6.6.2.6 of Chapter 5 of the Evolutionary Requirements Document, EPRI provides requirements for protecting the containment from long-term overpressurization through the containment design or, alternatively, the use of pressure relief. The staff concludes that these requirements are consistent with the above guidelines and are acceptable. However, because the Evolutionary Requirements Document does not provide detailed design requirements, the staff will evaluate this matter during its review of an individual application for FDA/DC.

#### 2.5.4 Anticipated Transients Without Scram

In Section 2.5.4 of Appendix B to Chapter 1 of Revision 1 of the Evolutionary Requirements Document, EPRI stated that the Evolutionary Requirements Document fully addressed the regulatory requirements of 10 CFR 50.62 and General Design Criterion 26 in Appendix A to 10 CFR Fart 50 regarding reduction of the

probability of an anticipated-transient-withou -scram (ATWS) event and enhanceme of the mitigation capability should such an event occur in the evolution BWR. EP'I stated that, because of the increased margins, reduced reactor protection system, and the enhanced reliability of the demands on. control rod . 've system, automatic initiation of the standby liquid control system (SLCS) .s required by 10 CFR 50.62 was not necessary for the evolutionary BWR.

However, in its December 6, 1991 letter, EPRI stated that it had determined that automatic actuation of the SLCS was appropriate for the evolutionary BWR design. In Revision 4, EPRI deleted Section 2.5.4 from Appendix B to Chapter 1 of the Evolutionary Requirements Document because it no longer considers this to be a plant optimization subject for the evolutionary plant design. In addition, EPRI modified Section 4.6.3 of Chapter 5 to require automatic initiation of the SLCS.

The staff concludes that these changes are in accordance with 10 CFR 50.62 and are, therefore, acceptable. Therefore, this issue is closed. This issue is also discussed in this report under Item II.A of SECY-90-016 contained in Annex A of this appendix and Section 4.2 of Chapter 5.

### **3 GENERIC SAFETY ISSUES**

### 3.1 Introduction

In the DSER for the various chapters of the Evolutionary Requirements Document, the staff addressed 62 generic safety issues. Table 3B.1 in this appendix lists these issues and gives the status of each as provided in the appropriate DSER chapter. The table shows that 29 of these issues remained open as a result of the staff's review at the DSER stage and 3 were confirmatory issues. Eleven issues had vendor items. The staff's final evaluation of these issues is provided in Section 3.2.

In Section 3.1 of Appendix B to Chapter 1 of the current version of the Evolutionary Requirements Document, EPRI addresses 24 generic safety issues for which there was no generic resolution as of January 1990 in the NRC's generic issues management control system (GIMCS) and which EPRI determined were technically relevant to the ALWR evolutionary plant design using the criteria in NUREG-1197, "Advanced Light Water Reactor Program - Program Management and Staff Review Methodology." These issues are the following:

1.0.3	29	84	128
8-17	57	87	130
B-56	70	94	135
C-8	75	105	HF 4.4
15	79	113	HF 5.1
23	23	121	HF 5.2

Seven of these issues (I.D.3, B-56, 15, 57, 83, 87, and 135) were not addressed in the staff's DSERs; since January 1990, eight (29, 70, 75, 79, 84, 87, 94, and 135) have been technically resolved by the staff. The staff will evaluate the proposed elements for resolving these issues on the basis of the final agency issue resolution.

A note to Table B.3-1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that the screening criteria of NUREG-1197 were used to determine that Issues I.D.5(3), II.H.2, II.J.4.1, B-55, B-61, B-64, 106, and HF 4.4 were not technically relevant to the ALWR design. The Evolutionary Requirements Document, however, does not give the reasons why EPRI determined specific issues were not technically relevant to the ALWR design. The inclusion of Issue HF 4.4 in this note, for instance, appears to be in error in view of the fact that the Evolutionary Requirements Document proposes elements of resolution that include this issue. In addition, although the document does not propose elements for resolving Issue II.J.4.1, it includes commitments to comply with the revised regulations that form the basis for the staff's resolution of the issue.

To confirm and update EPRI's list, the staff used Appendix B, "Applicability of NUREG-0933 Issues to Operating and Future Plants," to NUREG-0933, "A Prioritization of Generic Safety Issues." The staff supplemented the information in Appendix B to NUREG-0933 with status information on various safety issues contained in the first-quarter fiscal year 1992 update of the

The issue titles and status are given in Section 3.2.

GIMCS report (memorandum for J. Taylor from E. Beckjord dated March 11, 1992). In confirming and updating EPRI's list, the staff identified 28 unresolved issues in Appendix B to NUREG-0933 as of December 31, 1991. Of these 28, for various reasons, five issues (I.D.5(3), II.J.4.1, P-55, B-61, and B-64) are not applicable to the ALWR Program. These five issues are also identified in the note to Table B.3-1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document as noted in the preceding paragraph and are discussed by the staff in Sections 3.2.2, 3.2.5, 3.2.13, 3.2.15, and 3.2.16 of this appendix. During its review, the staff noted that the current version of Appendix B to NUREG-0933 may not properly include all the generic safety issues applicable is inture plants. The staff is evaluating the applicability of these issues to future designs and may need to modify Appendix B to NUREG-0933 as a result of this review.

Two issues (C-8 and 84) identified in the Evolutionary Requirements Document are not included in Appendix B to NUREG-0933. These issues were included in Appendix B to NUREG-0933 as of December 31, 1989, but were later dropped. They are discussed in Sections 3.2.17 and 3.2.32, respectively, of this appen-dix. issue 135, which was resolved by the staff in March 1991, is not included in Appendix B to NUREG-0923. However, since proposed elements of resolution are provided in Appendix B to Chapter 1 of the Evolutionary Requirements Document, it is evaluated in Section 3.2.59.

In Section 3.2 of this appendix, the staff provides either an evaluation of each of these issues and the status of the staff's review or an explanation as to why the issue is not specifically evaluated. Since the generic agancy resolution has not yet been identified for many of the issues, the st ff may have to reevaluate the FDA/DC applicant's proposed resolution of individual issues once the generic resolution is determined. Where an issue's elements of resolution as proposed in the Evolutionary Requirements Document do not completely address the generic agency resolution of an issue or where the staff has postponed its evaluation of an issue until it reviews individual applications, the staff expects the applications for FDA/DC to address the final resolution in accordance with 10 CFR 52.47(a)(1)(iv) for the staff's review.

In its review, the staff identified 40 issues that are considered to be applicable to future plants (since they are listed in Appendix 8 to NUREG-0933), but that have not yet been prioritized. These are identified by number and title in Table 38.2 of this appendix. Although five of these issues (2, 76, 110, 118, and 132) have not been prioritized, the staff evaluated each in the DSER. In addition, as of December 31, 1991, the staff has identified six new issues, which are identified by number and title in Table 3B.3. However, it has not received specific information on these issues or whether it has been determined that they are applicable to future plants.

Although the staff is not considering the unprioritized and new issues at this time, it will monitor their safety status. The vendors will be required to address, in accordance with 10 CFR 52.47(a)(1)(iv), any that are applicable to the ALWR evolutionary plant designs and that are prioritized as high or medium. For the five unprioritized issues that the staff evaluated in the DSER and that contained open or confirmatory issues, the staff will defer their further evaluation to its review of applications for FDA/DC.

Irrespective of the specific issues addressed in this evaluation and in the Evolutionary Requirements Document, FDA/DC applications are required by 10 CFR 52.47(a)(1)(iv) to contain proposed technical resolutions of those unresolved safety issues and high- or medium-priority generic safety issues that are identified in the version of NUREG-0933 current 6 months before the application and that are relevant to the design.

### 3.2 Staff Evaluations of Generic Issues

### 3.2.1 I.D.3, Safety System Status Monitoring

<u>Issue</u>: This issue concerns the performance of a study to determine the need for all licensees and applicants not committed to Regulatory Guide (RG) 1.47, "Bypassed and Inoperable Status Indication for Nuclear Power Plant Safety Systems," to install a bypass and inoperable status indication system or similar system.

EPRI Proposal: Section 3.10.4.4 of Appendix B to Chapter 1 and Table B.1-2 in Appendix B to Chapter 1 of the Evolutionary Requirements Document show an ALWR Program commitment to comply with RG 1.47, Revision 0.

<u>Staff Evaluation</u>: The commitment to comply with RG 1.47 satisfies the staff's primary concern regarding this issue at this time. Issue I.D.3 has not yet been generically resolved by the staff and has a medium safety priority. Should final agency resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue I.D.3 for staff evaluation.

### 3.2.2 I.D.5(3), Improved Control Room Instrumentation Research - On-Line Reactor Surveillance System

<u>Issue</u>: The objective of this issue is to (1) perform research to determine the feasibility of detecting and diagnosing nuclear power plant operating anomalies using a continuous on-line noise surveillance system; (2) demonstrate such a system in an NRC-licensed commercial BWR plant (which has been done); and (3) prepare a report summarizing the conclusions and recommendations from the research.

<u>EPRI Proposal</u>: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document EPRI states that Issue I.D.5(3) was determined, through the NUREG-1197 screening process, to be not technically relevant to the ALWR design.

<u>Staff Evaluation</u>: The staff considers this issue to be a research item that has no apparent effect on ALWR evolutionary plant designs at this time. Appendix A to NUREG-1197 identifies issues that the staff determined were not applicable to the ALWR Program according to six categories. Although Issue I.D.5(3) is not specifically listed in any of the categories, the staff believes that it should be considered a Category (or Code) 2 item. Category 2 includes issues that address NRC training programs, policy developments, organizational changes, studies, code development and maintenance, and research activities not directly affecting plant design. On the basis of its ongoing programs, the staff has identified a technical resolution for this issue. Should final agency resolution of this issue dictate that this issue is otherwise applicable to ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue I.D.5(3) for staff evaluation.

#### 3.2.3 II.E.4.3, Containment Design - Integrity Check

<u>Issue</u>: Following an outage, a plant could inadvertently be returned to operation without all access openings being closed. This issue originated with the discovery at a nuclear power plant in 1979 that two 3-inch containment exhaust valves had been left open for approximately 1% years. To ensure containment integrity, an independent means should exist for directly verifying containment integrity during plant operation.

EPRI Proposal: The EPRI-proposed resolution of this issue is through the incorporation of design features to enhance administrative control of containment integrity. In addition to the leak rate testing specified to satisfy regulations, Chapter 5 of the Evolutionary Requirements Document states that specific design features will focus on reducing the probability of a containment penetration being inadvertently left open. By clearly identifying all isolation valves, ensuring easy access for verification, providing better position indication, and including fail-safe design features, the probability of a large penetration being left open is substantially reduced. In addition, Section 6.3.2.5 of Chapter 5 requires the means to enable the operator to perform a periodic check for gross leakage of containment atmosphere during normal operation. Instrumentation for other purposes (i.e., that used for Type A integrity leakage rate testing and air or gas addition flow monitors) will be used to the extent practicable for this function.

<u>Staff Evaluation</u>: In the DSER for Chapter 5, the staff reported that its assessment (NUREG-1273, "Technical Findings and Regulatory Analysis for GSI 11.E.4.3, 'Containment Integrity Check,'") of Issue II.E.4.3 generically resolved the issue for operating reactors and questioned the cost-benefit of a backfit for current operating plants that would require a continuous or shortterm periodic gross check of containment integrity. The staff indicated, however, that such a system would not be a backfit for an ALWR and that it was not evident, from the information in the Evolutionary Requirements Document, that evolutionary ALWRs would have a significantly reduced number of containment penetrations. The staff concluded that the next generation of LWRs should have an improved capability to detect inadvertant containment bypass and that the Evolutionary Requirements Document should address these concerns. EPRI has revised the Evolutionary Requirements Document stating that such improved detection capability will be available. Therefore, this open issue is resolved.

3.2.4 II.H.2, Obtain Technical Data on the Conditions Inside the Three Mile Island, Unit 2, Containment Structure

<u>Issue</u>: This issue entails the collection of certain technical information on the conditions of the facility at Three Mile Island, Unit 2, as cleanup operations proceed. The information is to be disseminated under another generic safety issue (II.H.3). The information obtained under this issue will be used in resolving other generic safety issues, such as Issues A-45 and A-40 (former unresolved safety issues) and issues II.B.5, II.B.7, II.B.8, and II.E.3.4.

EPRI Proposal: In the note to Table 8.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue II.H.2 was determined, through the NUREG-1197 screening process, to be not technically relevant to the ALWR d ign.

Staff Evaluation: This issue entails the collection of information that will be used in resolving other specific generic safety issues and, therefore, is not to be analyzed separately. Also, Appendix A to NUREG-1197 identifies issues that the staff determined were not applicable to the ALWR Program according to six categories. Issue II.H.2 is listed as a Code (or Category) 5 item, that is, an issue that is not applicable to plant design. In addition, according to Appendix B to NUREG-0933, Issue J7 K.2 is not applicable to future plants; therefore, an evaluation is not necessary.

### 3.2.5 II.J.4.1, Revise Deficiency Reporting Requirements

Issue: The event reporting requirements of 10 CFR Pact 21 and the deficiency reporting requirements of 10 CFR 50.55(e) are being revised to ensure that licensees report all reportable items promptly to the NRC and that the information submitted is complete. The reports received as a result of the anticipated rule changes are expected to (1) provide increased information on component failures that affect safety (so that prompt and effective corrective action can be taken) and (2) be used as input to an augmented role of the NRC's vendor and construction inspection programs.

EPRI Proposal: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue II.J.4.1 was determined, throuch the NUREG-1197 screening process, to be not technically relevant to the ALWR ign. This may be in error, however, since Table B.1-1 in Appendix B to Chap 1 shows an ALWR Program commitment to comply with 10 CFR Part 21 and 50.55(e).

Staff Evaluation: The staff has identified a resolution of Issue II.J.4.1, changes were published in the <u>Federal Register</u> (56 FR 36081) on July 31, 1991. EPRI's commitments to comply w.ch 10 CFR Part 21 and 50.55(e) satisfy the staff's primary concerns regarding this issue at this time.

The discussion in NUREG-0933 of Issue II.J.4.1 has not been updated to reflect the recent rule changes. Should the final agency resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address those actions for staff evaluation.

3.2.6 A-29, Nuclear Power Plant Design for the Reduction of Vulnerability to Industrial Sabotage

<u>Issue:</u> Reduction of the vulnerability of reactors to radiological sabotage is currently treated as a plant physical security function and not as a plant design requirement. Although present reactor designs provide a great deal of inherent protection against industrial sabotage, extensive physical security measures are still required to provide an acceptable level of protection. An

EPRI Evolutionary Plant SER 1B.3-5

1

alternative approach would be to consider more fully reactor vulnerabilities to sabotage during the preliminary design phase. Because emphasis is being placed on standardizing plants, it is especially important to consider measures that could reduce the vulnerability of reactors to sabotage. Any design features to enhance physical protection must be consistent with system safety requirements.

EPRI Proposal: Section 1.4.1 of Chapter 9 of the Evolutionary Requirements Document states that substantial inherent resistance to sabotage will be provided in an ALWR plant through rugged, reinforced-concrete external walls and internal barriers that restrict access. It also states that further resistance will be provided through physically separated, redundant safety systems that provide a high reliability for the prevention of cora damage and the mitigation of the consequences of initiating events, including sabotage. The access control features of the physical security system and the vital area barriers will provide the primary defense against the internal threat.

In addition, the Evolutionary Requirements Document indicates that the design will ensure that the necessary security functions are provided without unnecessarily constraining plant operability or impeding access to or egress from vital areas under emergency conditions.

The Evolutionary Requirements Document includes the following key requirements that are applicable to both BWRs and PWRs:

- Section 2.2.5 of Chapter 5 requires that the design features, configuration, and divisional separation of the engineered safety systems be integrated with the building design and the plant's security system to enhance the plant's capability for protection against sabotage.
- Section 2.3.2 of Chapter 5 requires that each division of safety system functions be totally independent and separated both mechanically and electrically except for areas in which it is physically impracticable or less safe. Requirements for mechanical separation that enhance protection from sabotage include 3-hour-fire barriers, physical protection of each division from flooding, and physical protection from pipe breaks both inside and outside the containment. Additionally, the ph ical barriers between divisions should be designed to enhance the plant's capability to resist sabotage.
- Section 1.5.2 of Chapter 6 summarizes EPRI's policy related to plant arrangement and the physical separation of engineered safety systems to reduce the vulnerability of ALWR plants to sabotage.
- Sections 2.3.5.4 and 2.3.8 of Chapter 6 provide specific requirements for plant arrangement that provide for protection from external threats and maximize security by enhancing sabotage resistance. EPRI states that the plant will be arranged to provide as much inherent protection as possible. Also, the plant designer will incorporate physical protection measures, such as hardening of vital walls, floors, and ceilings, into the design of the areas that house equipment performing vital functions. Utility ports penetrating vital area boundaries will be minimized and barriers will be designed to protect vital systems from sabotage.

 Section 5 of Chapter 9 provides the requirements for the site security system, including ensuring the compatibility of security system requirements with plant arrangement and safety system design, definition of vital systems, layout of vital components, security barriers, intrusion detection systems, isolation zone requirements, security alarms, access control, security communications, power supply, and data management.

Staff Evaluation: In its DSER for Chapter 9, the staff concluded that the Evolutionary Requirements Document had adequately addressed Issue A-29 with respect to external sabotage, but that it had not adequately addressed inside: sabotage. At that time, the discussion of this issue in Appendix B to Chapter 9 of the Evolutionary Requirements Document made no reference to the potential for sabotage that could occur during maintenance activities. In the DSER, the staff noted that in a letter to the NRC dated August 18, 1989, EPRI had committed to expand Section 5.2.2.1 of Chapter 9 to require that ALWR plant designers analyze the vulnerability of their designs to insider sabotage before finalizing the designs. The staff has confirmed that this section of the Evolutionary Requirements Document includes a requirement for the analysis to consider both insider and outsider threats. Therefore, this confirmatory issue is resolved.

3.2.7 A-40, Seismic Design Criteria (former unresolved safety issue)

<u>Issue</u>: Issue A-40 was formulated to identify and quantify conservatism or nonconservatism inherent in seismic design criteria for nuclear power flants. Several years of research and cooldination with the industry convinced the s\*aff that some changes in the regulatory guidance in the Standard Review Plan (SKP, NUREG-0800) for seismic design were warranted.

As the work on Issue A-40 progressed, the staff decid\_d to modify the regulatory guidance for seismic design in the SRP by revising the appropriate sections of the SRP to include the current state of knowledge in the licensing criteria. As a result of the resolution of Issue A-40, Revision 2 of SRP Sections 2.5.2. "Vibratory Ground Motion"; 3.7.1, "Seismic Design Parameters"; 3.7.2, "Seismic System Analysis"; 3.7.3, "Seismic Subsystem Analysis"; and 3.7.4, "Seismic Instrumentation," was issued in August 1989. All new applicants were required to comply with these revised SRP sections.

<u>EPRI Proposal</u>: Methods for operating-basis-earthquake (OBE) and dynamic analysis are addressed in Section 2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

<u>Staff Evaluation</u>: EPRI's proposal for certain aspects of Issue A-40 differs from the staff's acceptance criteria in the SRP sections referenced above, as follows:

 The Evolutionary Requirements Document states that the requirement in Appendix A to 10 CFR Part 100 that structures, systems, and components be designed to withstand an OBE should be eliminated, and that the OBE should be decoupled from the safe shutdown earthquake (SSE). Currently, 10 CFR Part 100, Appendix A requires that the magnitude of the OBE be at least one-half that of the SSE. Although Appendix A to 10 CFR Part 100 is being revised, decoupling the OBE from SSE is currently an exception to Appendix A to 10 CFR Part 100. If an ALWR applicant wishes to decouple the OBE from the SSE, it needs to specify the decoupling criteria, which the staff will evaluate and address as appropriate on a design-specific basis. This issue is discussed in this report in Section 4 of Chapter 1, Section 2.1.1 of Appendix B to Chapter 1, under Item IV.A of SECY-90-016 contained in Annex A of this appendix, under Item I.M of the draft Commission paper issued on February 27, 1992 in Annex C, and under Item C of the draft Commission paper issued on July 5, 1992, in Annex D.

Since the Evolutionary Requirements Document does not provide sufficient information to enable the staff to evaluate the design requirements in this area, the staff will evaluate this area during its review of individual applications for FDA/DC.

Regarding the damping values for seismic design for the OBE and SSE given in RG 1.61, "Dam ing Values for Seismic Design of Nuclear Power Plants," the Evolutionary Requirements Document states that since the SSE damping values are based on conservative lower bounds, higher damping values would be more realistic. Consequently, EPR1 proposes to use the damping criteria of American Society of Mechanicai Engineers Boiler and Pressure Vessel Code (ASME Code), Section III, Code Case N-411 and to delete the use of the OBE damping values.

The Code Case N-411 damping values are meant to be used for piping and equipment and not for seismic design of structures. The staff has accepted the Code Case N-411 damping values for piping and equipment supports provided the additional guidance of RG 1.84, "Design and Fabrication Code Case Acceptability - ASME Section III, Division 1," is followed. The use of the damping values of Code Case N-411 for structures for SSE loads and the deletion of the OBE damping values of RG 1.61 is a deviation from the established staff position. Justifications for such deviations must be provided and can be accepted by the staff only when the decoupling issue discussed above is resolved.

Since the Evolutionary Requirements Document does not provide sufficient information to enable the staff to evaluate the design requirements in this area, the staff will evaluate this area during its review of individual applications for FDA/DC.

The Evolutionary Requirements Document states that the modal combination of high-frequency modes for vibratory loads with significant highfrequency input (33 to 100 Hz) should be done using the algebraic sum method.

The algebraic sum method for modal combination of high-frequency modes for vibratory loads is a deviation from RG 1.92, "Combining Modal Responses and Spatial Components in Seismic Response Analysis," and Appendix A to SRP Section 3.7.2. Either Appendix A to SRP Section 3.7.2 should be used, or additional justification for this position must be provided for staff review. Since the Evolutionary Requirements Dcrument does not provide sufficient information to enable the staff to evaluate the design requirements in this area, the staff will evaluate this area during its review of individual applications for FDA/DC.

 The Evolutionary Requirements Document states that the use of the spectral peak shifting technique should be permitted as an alternative to spectrum broadering.

The use of the spectral peak shifting tech: use rather than spectrum broadening is a deviation from RGs 1.92 and 1.122, "Development of Floor Design Response Spectra for Seismic Design of Floor-Supported Equipment or Components." The staff has accepted the spectral peak shifting technique on a case-by-case basis after evaluating the frequency content and proximity of the peaks for plant-specific designs. Therefore, the staff will evaluate this exception during its review of individual applications for FDA/DC.

ø

3.2.8 A-46, Seismic Qualification of Equipment in Operating Plants (former unresolved safety issue)

<u>Issue</u>: The design criteria and methods for the seismic qualification of mechanical and electrical equipment in nuclear power plats have undergone significant change during the commercial nuclear power plats have undergone. the margins of safety in existing equipment to resist seismically induced loads and perform the intended safety functions may vary considerably. The seismic qualification of the equipment in operating plants must, therefore, be reassessed to ensure the capability to bring the plant to a safe shutdown condition after a seismic event. The objective of this issue was to establish an explicit set of guidelines that could be used to judge the adequacy of the seismic qualification of mechanical and electrical equipment at all operating plants in lieu of attempting to backfit current design criteria for new plants.

<u>EPR1 roposal</u>: Section 12.6.2 of Chapter 1 of the Evolutionary Requirements Document requires that anchor bolts for flat-bottomed tanks be extended at least 2 feet above the concrete foundation level and that the anchor bolt chairs be extended full height and continuously welded to the tank wall. These requirements are intended to distribute seismic forces and to develop ductility.

Staff Evaluation: In the DSER for Chapter 6, the staff stated that the Maine Yankee Seismic Margin Study and the resolution of Issue A-46 showed that the anchorage of the safety-related tanks was a weakness at operating plants and concluded that EPRI should emphasize tank anchorage design and installation. The staff has reviewed the requirements in Section 12.6.2 of Chapter 1 of the Evolutionary Requirements Document pertaining to the extension of anchor bolts and anchor bolt chairs for flat-bottomed tanks and the welding of anchor bolt chairs to tank walls and concludes that they adequately address the DSER open issue regarding tank anchorage design and installation. Therefore, this open issue is resolved. 3.2.9 A-47, Safety Implications of Control Systems (former unresolved safety issue)

<u>Issue</u>: Concerns have been raised regarding the potential for accidents or transients being made more severe as a result of control system failures, including control and instrumentation power supply faults. During the licensing review process, the staff performs an audit review of the nonsafety-grade control systems to ensure that an adequate degree of separation and independence is provided between these non-safety-grade systems and the safety systems. On this basis, it is generally believed that control system failures are not likely to result in loss of safety functions that could lead to serious events or result in conditions that the safety systems are not able to mitigate. However, indepth studies for all non-safety-grade systems have not been performed.

EPRI Proposal: Table B.1-2 in Appendix B to Chapter 1 of the Evolutionary Requirements Document shows an ALWR Program commitment to comply with Generic Letter (GL) 89-19, "Request for Action Related to Resolution of Unresolved Safety Issue A-47, 'Safety Implication of Control Systems in LWR Nuclear Power Plants.'"

Staff Evaluation: The staff has resolved Issue A-47 and has issued GL 89-19. GL 89-19 contains several recommendations regarding protection from certain control system failures and modification of selected emergency procedures to ensure that plant transients resulting from control system failures do not compromise public safety.

In its DSER for Chapter 1C, the staff concluded that EPRI's commitment (letter to the NRC dated December 21, 1990) to include a requirement that the ALWR designer comply with GL 89-19 was acceptable and that it would confirm that this commitment was acceptably incorporated into the Evolutionary Requirements Document. The staff has confirmed that Table B.1-2 in Appendix B to Chapter 1 indicates a commitment to comply with GL 89-19. Therefore, this confirmatory issue is resolved.

The staff also concluded that the Evolutionary Requirements Document did not provide the necessary design details to be able to determine if the plantspecific design and arrangement will be adequate to resolve this issue. Therefore, the staff will evaluate re olution of Issue A-47, including aspects of digital control and safety systems, during its review of individual applications for FDA/DC. See also the evaluation of Issue 76 in Section 3.2.28.

3.2.10 A-48, Hydrogen control Measures and Effects of Hydrogen Burns on Safety Equipment

<u>Issue</u>: As a result of the accident at Three Mile Island, Unit 2 (TMI-2), the Commission promulgated regulatory requirements on hydrogen control in 10 CFR 50.34 and 50.44. 10 CFR 50.34(f) requires a hydrogen control system based on a 100- percent fuel-cladding metal-water reaction and a hydrogen concentration limit of 10 percent on uniformly distributed hydrogen in the containment or a postaccident atmosphere that will not support hydrogen combustion. Plants covered by this rule included only those whose construction permits had not been issued at the time of the TMI-2 accident. The issue is whether this requirement applies to the evolutionary plants.

<u>EPRi Proposal</u>: In Section 3.5.3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI proposes that the evolutionary plant design have the capability to ensure that necessary accident prevention and mitigation functions can be performed during and after events in which hydrogen is produced. The Evolutionary Requirements Document requires the design to accommodate an amount of hydrogen equivalent to that generated by oxidation of 75 percent of the fuel cladding surrounding the active fuel. This is to be accomplished for a PWR dry containment by containment volume and mixing so that the uniformly distributed hydrogen gas concentration in the containment does not exceed 13 percent under dry conditions. For a BWR, the Evolutionary Requirements Document permits provision of a noncombustible containment atmosphere as an acceptable alternative approach.

<u>Staff Evaluation</u>: In Appendix B of its DSER for Chapter 5, the staff concluded that EPRI's proposed approach to resolving this issue was unacceptable. On the basis of the proposed resolution now in the Evolutionary Requirements Document, the staff agrees with EP<sup>P</sup>I's proposal on hydrogen control for BWRs, which indicates that inerting is an acceptable approach. This will meet the requirements of 10 CFR 50.44, the issuance of which resolved Issue A-48.

In SECY-90-016, the staff recommended to the Commission that the hydrogen control requirements for evolutionary plants be ident. all to those stated in 10 CFR 50.34(f). The Commission approved the staff's recommendation in a staff requirements memorandum dated June 26, 1990. In its letter to the NRC dated December 6, 1991, EPRI stated that the Evolutionary Requirements Cocument would be modified to fully comply with the staff position of 100-percent active fuel cladding and a maximum containment hydrogen gas concentration of 10 percent. This commitment is acceptable and resolves the open issue in the DSER. However, since the staff position in SECY-90-016 is that the plant-specific design must comply with 10 CFR 50.34(f) for combustble gas control, the staff will review relevant design features for conformance to 10 CFR 50.34(f) during its review of individual applications for FDA/DC. See also the evaluations in Sections 2.3 and 6.5.1 of Chapter 5 of this report and the evaluation of Issue 121 in Section 3.2.46 of this appendix.

3.2.11 B-17, Criteria for Safety-Related Operator Actions

<u>Issue</u>: Issue B-17 entails the development of criteria for safety-related operator action (SROA) during the response to or recovery from transients and accidents. The criteria would include a determination of actions that are to be automated in lieu of operator action and the development of a time criterion for SROA. Of particular concern is consideration of the need to automate emergency core cooling system (ECCS) realignment following a loss-of-coolant accident (LOCA), which currently requires manual operation in some PWRs. Such a realignment is dependent on break size and must be done before the borated water storage tank is depleted. Implementation of this criterion may result in changes and additions to the design of the engineered safety features control systems.

<u>EPRI Proposal</u>: Section 3.10.2.4.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that the ALWR plant design requires automation of the initiation of protect in and safety systems and that the automation of the operation of these systems precludes the need for operator

action for at least 30 minutes after the initiation of an event. The capability for manual initiation and operation is not precluded, and adequate information and control capabilities are required so that operators can effectively provide backup for automated actions.

Section 3.10.2.4.2 of Appendix B to Chapter 1 states:

- Sections 2.3.2.9, 2.3.3.5, and several places in Table 1.2-5 of Chapter 1 of the ALWR Requirements Document specify the ALWR overall requirements for the time after an event is initiated in which the operator must act as not less than 30 minutes, assuming a single failure.
- Section 2.1.1 of Chapter 10 states the objective of the ALWR design shall be to take full advantage of operator cayabilities, but not to challenge operator limitations.
- Section 2.2.9 of Chapter 10 states the policy for the design of the A!WR M-MIS [man-machine interface system] which requires that each monitoring, control, and protection function be evaluated as part of the design process to determine the appropriate level of automation. Consideration is required of such factors as operator workload, system response, operation complexity, level and duration of the operator's attention, and failure of the automatic features.
- Section 3.4.3 of Chapter 10 requires that the design choice on automatic versus manual control or monitoring is to be based on evaluations which specifically include consideration of operator workload, operator capabilicy, past experience with automatic or manual controls or monitoring in similar applications, operator vigilance and the need to keep the operator involved and knowledgeable as to the plant status, amount and complexity of M-MIS equipment (including software) and the resulting maintenance and testing burden, the consequences of and potential for malfunctions of the automatic equipment and for operating errors, and regulatory requirements.
- Section 8.2.3 of Chapter 10 defines the control and monitoring strategies which shall be used for protection and safety systems. This includes the requirement that startup or actuation of these systems shall be automatic but with an effective manual backup. It also requires that these systems operate automatically after actuation for at least 20 minutes and that manual operations be limited so that essentially continuous manning for extended periods of time, e.g., hours or days, is not required.
- The part of this issue concerning automatic versus manual EC's realignment does not apply to the ALWR PWR. The PWR safety injection system (SIS) may tains inventory by pumping water from an in-containment refueling water storage tank (IRWST) to the reactor vessel. The IRWST provides a continuous source of water to the SIS pumps because water collects in the IRWST and thus eliminates the need for realignment of the SIS pumps for long-term post-LOCA recirculation.

Section 3.10.2.4.3 of Appendix B to Chapter 1 states that "the ALWR design will satisfy NRC requirements concerning automation of safety-related operator

EPRI Evolutionary Plant SER 18.3-12

actions and operator response times. Since that part of the issue concerning automatic realignment of ECCS is not applicable to the ALWR by design, this issue is considered resolved for the ALWR."

Section 3.2.2 of Chapter 5 states that "the source of core coolant inventory makeup .nall be adequat. "or at least 36 hours without need for manually switching suction sources for non-LOCA events."

<u>Staff Evaluation</u>: In the DSER for Chapter 10, the staff concluded that EPRI had not provided the technical basis or specific requirements that support its assertion that the quantitative standards and the process requirements in the Evolutionary Requirements Document are significantly more limiting than those in American National Standards Institute/American Nuclear Society (ANSI/ANS) 58.8-1984, "Time Response Design Criteria for Safety Related Operator Actions at Nuclear Power Plants." In response to the DSER, EPRI has modified the Requirements Document to provide the necessary technical basis and specific requirements. In addition, EPRI has committed to satisfy NRC requirements concerring automation of safety-related operator actions and operator response times. Since the revision to ANSI/ANS-58.8 (now expected to be issued in April 1993) is not expected to alter the time response criteria of the 1984 edition of the standard, the staff concludes that EPRI has presented an appropriate resolution of this issue for ALWR evolutionary plant designs.

Issue B-17 has not yet been generically resolved by the staff and has a medium safety priority. Should the final agency resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC appl'cant to address resolution of Issue B-17 for staff evaluation.

3.2.12 B-32, Ice Effects on Safety-Related Water Supplies

<u>Issue</u>: The service water system (SWS) rejects heat to the ultimate heat sink, which, during an accident or transient, cools the component cooling water heat exchangers. This in turn cools the residual heat removal heat exchangers, as well as provides cooling for safety-related pumps and area cooling coils. Blockage of the SW' flow intake structure by ice has led to plant shutdowns, reduced power operation for repairs and modifications, and degraded modes of operation.

<u>EPRI Proposal</u>: EPRI states that the problems associated with unreliable water flow from ultimate heat sinks as a result of fouling, blockage, and pump failure are addressed by the design requirements and guidance in the Evolutionary Requirements Document. The plant designer is given specific direction throughout Chapter 8 of the Evolutionary Requirements Document to design the ultimate heat sink and related water systems for both safety-related and nonsafety-related functions to minimize or avoid these problems. The requirements in Chapter 8 that are intended to resolve the issue of ice blockage of the SWS flow intake include the following:

 Section 7.3.1.3 requires that the safety heat c <sup>+</sup> be capable of performing its intended function during freezing weather. System features will be provided to prevent the pond from freezing. The plant designer will provide the utility with supporting analyses showing that the safety heat sink will not freeze during the winter. Weather conditions from the site survey will be used for this analysis, and the effects of icing of the intake structure will be considered.

Section 7.3.2.2 requires the recirculation of warm water through the intake structures for anti-icing protection.

The Evolutionary Requirements Document also directs the plant designer to further investigate the problem of ice as a flow-blockage mechanism and to provide provisions for disposing of and/or dissolving such ice, as required, and to design the ultimate heat sinks and water flow systems to avoid or minimize, as achievable, this problem.

Staff Evaluation: This issue has been subsumed by Issue 153. See Section 3.2.63.

3.2.13 B-55, Improved Reliability of Target Rock Safety/Relief Valves

Issue: The majority of valves in BWR pressure relief systems are Target Rock safety/relief valves, and a significant number of failures of these valves have occurred. Failures include valves (1) failing to open properly on demand, (2) opening spuriously and then failing to reseat properly, and (3) opening properly and then failing to reseat properly.

In the late 1970s, the NRC staff concluded that the inadvertent blowdown events that had occurred as a result of malfunctions of pressure relief system valves had neither significantly affected the structural integrity or capability of the reactor vessel or its internals or the pressure-suppression containment system, nor resulted in any significant radiation releases to the environment. Even if such events were to occur more frequently, there would not likely be any significant effects. The performance of these valves, however, is under continual surveillance and the consequences of their failures are subject to review.

EPRI Proposal: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue B-55 was determined, through the NUREG-1197 screening process, to be not technically relevant to the ALWR design.

Staff Evaluation: Appendix A to NUREG-1197 identifies issues that the staff determined were not applicable to the ALWR Program according to six categories. Issue B-55 is listed as a Code (or Category) 3 item, that is, an issue that is applicable only to reactor types and design features not included in current standardized designs. It is the staff's understanding that Target Rock safety/relief valves are not to be used in any ALWR evolutionary plant designs.

Issue B-55 has not yet been generically resolved by the staff and has a medium safety priority. Should the final agency resolution of this issue dictate that it is otherwise applicable to ALWR evolutionary plant designs or indicate addition; actions, or should Target Rock safety/relief valves be used in any evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue B-55 for staff evaluation.

# 3.2.14 B-56, Diesel Reliability

Issue: If a loss of normally available (ac) power occurs at a nuclear power plant, redundant onsite emergency ac power sources provide power for necessary safety functions, which include reactor core decay heat removal (General Design Criterion (GDC) 34 of Appendix A to 10 CFR Part 50), emergency core cooling (GDC 35), and containment heat removal (GDC 38). The systems performing these functions are essential for preserving the integrity of the reactor core, reactor coolant system, and containment. Although reactor core decay heat can be removed for a limited time by systems that are independent of ac power, most licensees have selected Class 1E emergency diesel generators (EDGs) as the long-term ac power sources in most plants for meeting the requirements of 10 CFR 50.63. Therefore, the reliability of EDGs is a major factor in ensuring acceptable plant safety.

EPRI Proposal: Section 3.11.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states:

- The NRC is currently in the process of upgrading Regulatory Guide 1.9, Selection, Design, and Qualification of Diesel-Generator Units Used as Standby (Onsite) Electric Power Systems and [sic] Nuclear Power Plants, to incorporate improvements in diesel reliability programs. Regulatory Guide 1.108 will be withdrawn, and revisions to SRP sections and Technical Specifications are being prepared. NUMARC [Nuclear Management and Resources Council] plans to revise Appendix D or NUMARC-87-00 to assure consistency with the emergency diesel generator reliability program described in Regulatory Guide 1.9.
- The ALWR is committed to improving the reliability of their EDG units over current experience and have required a minimum target reliability of 0.98. This will be achieved by only using EDG units, including auxiliary and support and control systems, that have demonstrated this reliability or, for new designs, are qualified per the latest revisions of ANSI/IEEE Standard 387 and IEEE [Institute of Electrical and Electronics Engineers] Standard 323. Additionally the reliability of the EDGs will be improved by a design that allows longer start times, thus eliminating problems which adversely affect reliability. Finally, the EDGs will be backed up by combustion turbines which must also have a target reliability of 0.98. These requirements should resolve this issue for the ALWR Requirements Document.

Staff Evaluation: EPRI identifies a minimum EDG target reliability of 0.98 and a backup from combustion turbines having a target reliability of 0.98. However, targets alone do not ensure maintenance of acceptable reliability levels.

In the course of resolving Issue B-56, the Commission recently directed the staff to amend 10 CFR 50.63 to require monthly EDG surveillance testing and reporting of failures to start and load runs against criteria developed to identify potential degradation of EDG performance. This performance-based rule (when issued) would apply to LWRs. Comparable performance monitoring criteria would apply to ALWRs.

EPRI Evolutionary Plant SER 18.3-15

Since the Evolutionary Requirements Document does not provide sufficient information to enable the staff to fully evaluate the design requirements related to this generic issue, the staff will evaluate the design features for acceptability when it reviews individual applications for FDA/DC.

Issue B-56 has not yet been generically resolved by the staff and has a high safety priority. Should the final resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue B-56 for staff evaluation.

3.2.15 B-61, Allowable Emergency Core Cooling System Equipment Outage Periods

Issue: This issue entails the establishment of surveillance test intervals and allowable equipment outage periods for operating nuclear power plants, using analytically based criteria and methods for the technical specifications. The allowable outage and testing intervals in present technical specifications were determined primarily on the basis of engineering judgment.

EPRI Proposal: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue B-61 was determined, through the NUREG-1197 screening process, to be not technically relevant to the ALWR design.

Staff Evaluation: Appendix A to NUREG-1197 identifies issues that the staff determined were not applicable to the ALWR Program according to six categories. Issue B-61 is listed as a Code (or Category) 5 item, that is, an issue not applicable to plant design, such as plant operation and operating procedures, management of operations, accident management and emergency preparedness, operator training and qualifications, inspection and maintenance, and operating experience reporting. Since Issue B-61 does not appear to address plant design, the staff does not consider it to be applicable according to the definitions of Appendix A to NUREG-1197.

Issue B-61 has not yet been generically resolved by the stafi and has a medium safety priority. Should the final resolution of this issue dictate that it is otherwise applicable to ALWR evolutionary plant designs or indicate additional actions, the staff will expect the FDA/DC applicant to address the resolution of Issue B-61 for staff evaluation.

3.2.16 B-64. Decommissioning of Reactors

Issue: This issue entails the need for rulemaking to provide regulations for the orderly retirement of nuclear facilities from service and the safe disposition of the remaining radioactivity.

EPRI Proposal: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue B-64 was determined, through the NUREG-1197 screening process, to be not technically relevant to the ALWR design.

Staff Evaluation: According to Appendix B to NUREG-0933, Issue B-64 is not applicable to future plants; therefore, an evaluation is not necessary.

## 3.2.17 C-8, Main Steamline Isolation Valve Leakage Control Systems

<u>Issue</u>: This issue was initiated to investigate the desirability of the main steam isolation valve leakage control system (MSIVLCS) recommended in RG 1.96, "Design of Main Steam Isolation Valve Leakage Control Systems for Boiling Water Reactor Nuclear Power Plants," for BWRs. After it was identified, the staff determined that the issue was of little or no significance to plant risk. New concerns arose from subsequent operational experience that indicated a relatively high MSIVLCS failure rate and a variety of failure modes at some BWR plants. Additional data on the magnitude and frequency of MSIV leakage at BWRs renewed concerns about the viability of the MSIVLCS design. In oddition, the question of backfitting MSIVLCSs for BWRs was raised.

<u>PRI Proposal</u>: Section 3.3.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that Issue C-8 is addressed in Section 2.3.1 of Appendix B as an ALWR optimization subject. EPRI proposes to eliminate the safety-related leakage control system to allow higher leakage rates through the MSIV and to use an alternative MSIV leakage treatment method that takes credit for the large surface volume in the main steam piping and the condenser hotwell to plate out the fission products following core damage.

Staff Evaluation: As discussed in Section 3.1 of this appendix, Issue C-8 is not included in Appendix B to NUREG-0933 because the staff had determined that is was not applicable to future plants. In its DSER for Chapter 3, however, the staff evaluated Issue C-8 and indicated that it did not expect to reach a conclusion regarding EPRI's proposal until the issue is generically resolved. The staff's review of the MSIVLCS is given in Section 2.3.1 of this appendix. On the basis of that review, this issue is resolved.

#### 3.2.18 2, Failure of Protective Devices on Essential Equipment

<u>Issue</u>: Essential equipment has experienced a large number of failures or has been incapacitated as a result of either the failure or intentional bypass of protective devices intended to trip active engineered safety features (ESFs) for indications of equipment faults. The affected systems exist throughout the plant and include the plant control system, the plant protection system, and the ESFs. The staff is concerned that the reliability estimates for essential equipment may not properly account for failure of the protective devices. Because of the loss of redundant devices through failures of circuits intended to be independent, there is an increased probability of common-mode failure of redundant vital services. This issue needs to be studied further to determine if failure rate estimates for essential equipment have increased and if essential equipment could be made significantly more reliable by improving the reliability of protective devices.

<u>EPRI Proposal</u>: The Evolutionary Requirements Document includes a number of specific design requirements to address the safety concerns of this issue. Chapter 5 requires that

 plant systems embody sufficient robustness of design to tolerate a conservative number of spurious or inadvertent engineered safety system actuations without the need for followup tests or inspections to verify the systems' integrity or operability (Section 2.2.7)  each division of engineered safety systems be totally independent electrically unless it is otherwise physically impracticable or less safe (Section 2.3.2)

Chapter 10 requires that

- a robust system design, including segmentation of major functions, separation of redundant equipment within a segment, and fault-tolerant equipment, be developed to achieve nigh reliability and protection against the propagation of faults (Section 1.2.3)
- the man-machine interface system (M-MIS) equipment be designed and configured to readily support inservice testing by providing built-in test features, incorporating good human factors principles, and avoiding the use of undesirable features such as test jumpers or lifting of leads (Section 2.2.4)
- the M-MIS design process explicitly consider the potential for and the consequences of failures of plant and M-MIS components (Section 3.1.3.4)
- the capability for continuous on-line self-testing of hardware integrity be provided for as much of the M-MIS as is practicable (Section 3.6.1)
- on detection of a failure in the M-MIS, a system be designed so that it can be placed in such configuration that an additional single failure will not prevent system-level protection or safety action (Section 3.6.4)
- the mean time to detect and repair failures down to the lowest replaceable module, averaged across all types of M-MIS equipment for the entire design life, be less than 4 hours and the maximum time for any module be less than 8 hours (Section 3.7.4)

<u>Staff Evaluation</u>: Section 2.3.2 of Chapter 5 of the Evolutionary Requirements Document states that each division of the ESF systems needs to be electrically independent unless impracticable or less safe. In its DSER for Chapter 10, the staff agreed that the implementation of the design criteria in the Evolutionary Requirements Document would resolve this issue, with the exception of the staff's concern about lack of specific guidelines for exceptions to the independence criteria for electrical systems. Since it has not prioritized Issue 2, the staff will defer further evaluation of the open issue in the DSER to its review of individual applications for FDA/DC.

3.2.19 15. Radiation Effects on Reactor Vessel Support

<u>Issue</u>: Neutron damage of reactor vessel structural materials causes embrittlement that may increase the potential for propagation of flaws that may exist in the materials. As long as the operating environment in which the materials are used has a higher temperature than the materials' nil ductility transition temperature (i.e., the lowest temperature at which the materials would not be susceptible to failure by brittle fracture), no failure by brittle fracture is expected. When subjected to neutron irradiation, many materials become more susceptible to brittle fracture at the operating temperatures of interest, which should have been accounted for in the design and fabrication of the reactor vessel support structures. Research has indicated that the susceptibility of materials exposed to neutron irradiation has been underestimated. Loss of material fracture toughness may result in failure of the reactor vessel support structures and consequent reactor vessel movement as a result of transient stress or shock, such as would be experienced in a seismic event. Reactor vessel support structures in nuclear power plants using particular types of supports (long-column, shield-tank, short-column, suspension) are assumed to be susceptible to radiation damage.

<u>EPRI Proposal</u>: Proposed elements for resolving this issue are included in Section 3.4.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

<u>Staff Evaluation</u>: The elements of resolution proposed in the Evolutionary Requirements Document appear to adequately address the safety concerns of this issue. However, Issue 15 has not yet been generically resolved by the staff and has a high safety priority. Since it has not yet finalized its position on this issue for ALWRs, the staff will evaluate the applicant's proposed resolution of Issue 15 during its review of individual applications for FDA/DC.

3.2.20 23, Reactor Coolant Pump Seal Failures

<u>Issue</u>: This issue addresses the high rate of reactor coolant pump (RCP) seal failures that challenge the makeup capacity of the emergency core cooling systems in PWRs.

<u>EPRI Proposal</u>: Section 3.3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document provides the elements for resolving this issue. EPRI revised Chapter 3 of the Evolutionary Requirements Document to require that degradation of the shaft seal system be negligible following loss of both seal injection and pump cooling water for up to 1 hour. Also, in the event of loss of all ac power, reactor coolant system leakage through the RCP shaft will be limited so that the reactor core remains covered and natural circulation is maintained for at least 8 hours.

Staff Evaluation: In its DSER for Chapter 3, the staff concluded that some of the design requirements proposed by EPRI might not be practical unless adequate pump seals were developed and/or independent seal cooling was provided. In particular, the staff pointed out that Section 3.4.2.2.5 of Chapter 3 of the Evolutionary Requirements Document (at the time the DSER was prepared) stated that during station blackout conditions, the system isign leakage through the pump seals is to be limited to 8 gpm per pump to ensure that the reactor core remains covered for 8 hours. The staff believes that this requirement would be difficult to meet, given the present hydrostatic RCP seal. This seal requires an injection cooling supply of approximately 8 gpm and, if uncooled, the seal leakage flow could increase to 21 gpm. Hydrodynamic-type seals might meet the 8-gpm limit with the leakage uncooled; however, this could probably not be ensured for an extended station blackout. These leakage figures are for an intact seal system. Without seal cooling, the seal may fail ("pop open"), which could lead to leakage rates of several hundred gallons per minute.

EPRI Evolutionary Plant SER 18.3-19
Issue 23 has not yet been generically resolved by the staff and has a high safety priority. Unless the final agoncy resolution of this issue dictates other actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to submit a design that provides for independent RCP seal cooling during station blackout or to provide adequate testing of the proposed seal design to demonstrate integrity following extended loss of seal injection and cooling.

## 3.2.21 24, Automatic Emergency Core Cooling System Switch to Recirculation

<u>Issue</u>: The emergency core cooling system (ECCS) has two different phases, the injection phase and the recirculation phase. The injection phase involves initial cooling of the reactor core and replenishment of the primary coolant following a loss-of-coolant accident. The recirculation phase provides long-term cooling during the accident recovery period.

Switchover from the injection phase to the recirculation phase includes alignment of a number of valves to the recirculation position. Switchover can be achieved by a number of manual actions (manual option), by automating these actions (automatic option), or by automatic realignment of certain valves and manual completion of the switchover process (semiautomatic option). These three switchover options are vulnerable in varying degrees to human errors and hardware failures, as well as common cause failures. Although an automatic system designed to control the whole switchover process or a portion of it can reduce the impact of operator error in executing the switchover, automatic systems may a subject to spurious actuation. Spurious switchover of ECCS and containment spray pump suction to a dry containment sump can result in pump damage and possible loss of safety function, resulting in potentially unacceptable safety consequences. Review of past reactor experience indicates the existence of a significant number of ECCS spurious actuations, such as the iour that occurred in 1980 at the Davis-Besse Nuclear Plant, Unit 1.

EPRI Proposal: Not provided in Section 3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

<u>Staff Evaluation</u>: Section 3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that it specifically addresses compliance with high- and medium-priority and nearly resolved generic safety issues included as of January 1990 in the NRC's generic issue management control system (GIMCS). As of December 31, 1989, although Issue 24 was identified in Appendix B to NUREG-0S33 as being applicable to future plants, it had not yet been prioritized.

As of the first-quarter fiscal year 1992 update of the GIMCS report, Issue 24 has a medium safety priority. Since this issue is not addressed in the Evolutionary Requirements Document, the staff will evaluate its proposed resolution during its review of individual applications for FDA/DC.

3.2.22 29, Bolting Degradation or Failure in Nuclear Power Plants

<u>Issue</u>: Issue 29 was established in 1982 to address staff concerns about the number of events involving the degradation of threaded fasteners (bolt cracking, corrosion, failure, etc.) in operating plants from 1964 to the early 1980s. Many of the events were related to components of the reactor coolant pressure boundary (RCPB) and support structures of major components. This

EPR: Evolutionary Plant SER

raised questions regarding the integrity of the RCPB and the reliability of the component support structures following a loss-of-coolant accident or a seismic event.

Failures reported by licensees involved a variety of threaded fasteners and several causes. As a result, several different failure mechanisms had to be considered. Most frequent were wastage (corrosion) from boric acid attack and stress corrosion cracking (SCC). The former occurred more often at RCPB joints; the latter in structural bolting.

EPRI Proposal: Proposed elements for r a 'ving this in we are included in Section 3.1.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

Section 3...1.3 of Appendix 8 to Chapter 1 provides a discussion and regulatory status of Issue 29 and summarizes data on industry experience with bolting degradation completed by the NRC (NUREG-0943 and NUREG-1095) and the industry (Institute of Nuclear Power Operations SOER 84-5, "Bolting Degradat on or Failure in Nuclear Power Plants," and EPRI NP-3784, "A Survey of the Literature on Low-Alloy Steel Fastener Corrosion in PW? Power Plants"). Also mentioned is the cooperative industry effort of the Atomic Industrial Forum/Metal Properties Council Task Group on Bolting and ErRI to resolve the bolting issue (which resulted in EPRI NP-5769, "Degradation and Failure of Bolting in Nuclear Power Plants"). Results of EPRI NP-5769 and its NRC evaluation (NUREG-1339, "Resolution of Generic Safety Issue 29: Bolting Degradation or Failure in Nuclear Power Plants") are summarized. NRC's inter to revise the modard Review Plan to provide criteria for evaluating the bolting pract. proposed by applicants for new plants is also mentioned.

Section 3.1.1.4.2 of Appendix B to Chanter 1 provides requirements for all ALWRs regarding Issue 29. These requirements in Chapter 1 of the Evolutions Requirements Document are the following:

- 2.2.3 requires that the use of best-available materials and water stry be specified, based on the extensive LWR operating experience.
- Section 5.2.1.1 requires that the materials used be selected from materiz proven in service in LWR plants. Proven materials are those with the same nominal composition and subsequent processing steps and same exposure conditions as those used successfully for at least several years in existing LWRs.
- Section 5.2.4.3 requires that ALWR designs avoid very high strength fastener materials and make use of conventional materials applied well within the limits and for which successful experience has been obtained. In particular, the use of high-strength bolts or fasteners will be eliminated, where practicable, and sufficiently robus. designs will be used so that special, high-strength materials are not required.
- Section 5.3.3.1 requires that the materials for threaded fasteners used to maintain pressure boundary integrity in the reactor coolant and related systems and in the steam, feed, and condensate systems; threaded fasteners used inside those systems; and threaded fasteners used in pipe and component structural mountings for those systems be selected and specified on the basis of their previous satisfactory performance in

similar applications. Similarity will be based on comparable temperature and environment; comparable stresses; comparable service cycles; comparable fabrication and installation; and comparable inspection during fabrication and installation and while in service. The application of threaded fasteners will be in accordance with the requirements of Section 12.7 of Chapter 1 and with the guidelines of EPRI NP-6316, "Guidelines for Threaded-Fastener Application in Nuclear Power Plants."

- Section 5.3.3.2 requires that the lubricants used on all threaded fasteners that maintain pressure boundary integrity in the reactor coolant and related systems and in the steam, feed, and condensate systems; threaded fasteners used inside those systems; and threaded fasteners used in pipe and component structural supports for those systems be completely specified by the plant designer in appropriate drawings and specifications; that is, field selection of thread lubricants is not permitted. The thread lubricants will be selected on the basis of experience and test data that show they are effective, but do not cause or accelerate corrosion of the fastmer. Where leak sealants are used on threaded fasteners or can contact the fastener in service, their selection will be based on satisfactory experience or test data. Possible adverse interaction between sealants and lubricants should be considered
- Section 5.5.1 requires that materials selected for use in the ALWR be compatible with the full range of environmental conditions that may be encountered over the plant life. These environmental conditions include chemical (e.g., fluids that contact the material), temperature, and radiation factors. The plant designer will document the environmental conditions used as the basis for the selection of ALWR materials. These environmental conditions will be consistent with the specific requirements in other chapters of the Evolutionary Requirements Document.
- Section 12.7.1.1 requires that bolted joints and threaded fasteners be used only where a clear need for them has been established. The use of a bolted joint should include consideration of the initial cost, the subsequent maintenance effort, and the risk of service problems that such joints introduce. Where a bolted joint must be used to satisfy functional requirements, its adequacy should be achieved preferably by a simple, rugged design that does not depend on high-strength or unusual materials, precision parts, or specialized assembly and inspection methods.
- Section 12.7.1.4 requires that the configuration of bolted joints be based on their previous satisfactory performance in simil. applications and the current practice at the time the plant is design.d. Similarity will be based on comparable materials; comparable temperature and local environment; comparable stresses; comparable loading and service cycles; comparable fabrication and installation; comparable inspection during fabrication and installation and while in service; and comparable methods of locking or retention of tareaded faste.ars.
  - Section 12.7.2.1 requires that where the bolted joint provides sealing of a pre sure boundary (and especially where the contained material is radioactive, toxic, or corrosive), the joint configuration be designed to

18.3-22

minimize the potential for leakage, to minimize the consequences of such leakage (e.g., by using corrosion-resistant materials), and to facilitate the detection of any leakage.

- Section 12.7.2.1.1 requires that in those applications where gasket leakage can lead to corrosion and significant structural deviadation of the joint (e.g., borated water leakage on carbon steel), joint parts with adequate corrosion resistance be used. However, functional adequacy of the joint must not be compromised; that is, the most important attribute is that the joint be leaktight.
- Section 12.7.2.1.2 requires that if corrosion-resistant fasteners cannot be used in a particular joint and leakage could result in corrosion. inspection and maintenance capability be provided. Any sur' joints must be identified by the plant designer to the utility, in writing, along with the basis for their acceptability and how inspection and maintenance would be accomplished.
- Section 12.7.2.2 requires that the joint c. figuration, including threaded fasteners, facilitate inspection of the condition of the joint; for example, direct visual inspection of the condition of the loadcarrying portions of the joint should be practical without disassembly.
- Section 12.7.2.3 requires that the joint design specifically allow for tolerances in fabrication and variability in installed preload, as well as changes in preload that may occur in service (e.g., relaxation).
- Section 12.7.2.4 requires that where a boilted joint uses preloaded fasteners, the plant designer specify, in appropriate documents (drawings or procedures), the preload required and how it is to be achieved. including, for example, thread lubricants and torquing sequences. The setting of preloads and the selection of methods of preloading should not be left to the choice of shop or field activities.
- Section 12.7.3.1 requires the threaded fasteners be locked under the following conditions:
  - where the threaded fasteners are subjected to alternating loads that are a significant fraction of their minimum preload and especially wher the fastener's preload could be released by small deformations of the fastener or joint parts

-

- where the threaded fastener cannot be preloaded significantly even though the expected loads are small
- Section 3.1.1.4.3 states that the Evolutionary Requirements Document addresses all elements considered for the resolution of issue 29 and that Issue 29 is resolved for the ALWR.

Staff Evaluation: The staff technically resolved Issue 29 in October 1991 with the issuance of GL 91-17, "Generic Safety Issue 29, 'Bolting Degradation or Failure in Nuclear Power Plants,'" and NUREG-1339. The resolution is based on (1) operating experience with bolting in both nuclear and conventional power plants; (2) actions already taken through bulletins, generic letters,

EPRI Evolutionary Plant SER 18.3-23

and information notices since 1982; and (3) industry-proposed recommendations and actions, which are documented in EPRI NP-5769 and NP-5067, "Good Bolting Practices, A Reference Manual for Nuclear Power Plant Maintenance Personnel."

Even though the EPRI proposal discussed above appears to have met most of the recommendations in GL 91-17, the staff is not in total agreement with the proposal, as discussed below.

- Section 3.1.1.3 should reference GL 91-17 and discuss compliance with GL 91-17 as the basis for the technical resolution of Issue 29.
- The leaktightness of a bolted joint that provides sealing of a pressure boundary is significantly affected by whether the joint is properly assembled or not. If a bolted joint is to be successful, preloads developed at assembly must be sufficient to maintain joint integrity throughout the service life under external loads and service conditions. EPRI NP-5067, Volumes 1 and 2, provides, among other things, good bolting practices for assembling bolted joints and should be referenced in Section 3.1.1.3, as was done in GL 91-17.
- When Issue 29 was being resolved, the staff addressed several specific issues on threaded fasteners in bulletins, generic letters, and information notices (e.g., Pwn coolant pressure boundary builting and component degradation due to boric acid corrosion; stress corrosion cracking of internal bolting in certain types of check valves; traceability and material control of fasteners; and nonconforming, misrepresented, counterfeit, and/or fraudule t bolting), the details of which can be found in NUREG-1339. The pertinent information and requirements in these NRC bulletins, generic letters, and information notices issued to operating plant owners regarding Issue 29 should be factored into the discussion of the issue in the Evolutionary Requirements Document.

Since the Evolutionary Requirements Document does not provide sufficient commitments or information regarding these concerns and since it does not provide sufficient design details about the selection of bolting materials, lubricants, and leak sealants, the staff will evaluate resolution of these concerns during its review of individual applications for FDA/DC.

3.2.23 51. Improving the Reliability of Open-Cycle Service Water Systems

Issue: Issue 51 was established because fouling of safety-related service water systems (SWSs) by either mud, silt, corrosion products, or aquatic bivalves had led to plant shutdowns, reduced power operation for repairs and modifications, and degraded modes of operation in nuclear power plants.

EPRI Proposal: The only generic safety issue related to SWS reliability addressed in Appendix B to Chapter 1 of the Evolutionary Requirements Document is Issue 130, "Essential Service Water Pump Failures at Multiplant Sites" (see Section 3.8.1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document and Section 3.2.57 of this appendix). However, table B.1-2 in " B to Chapter 1 indicates a commitment to comply with GL 89-13, AD. "Service Water Systems Problems Affecting Safety Related Equipment," the closure document for the technical resolution of Issue 51 for existing plants. EPRI considers GL 89-13 to be appropriate for ALWRs.

EPRI Evolutionary Plant SER 18.3-24

The requirements in Chapter 8 of the Evolutionary Requirements Document that are intended to resolve this issue include the following:

- Section 1.3.3 states that the SWS will be an open system with its functions classified as both safety related and con-safety related.
- Section 1.3.5 states that the normal power heat sink will be the source for plant circulating water and non-safety service water. The ultimate heat sink (UHS) will be the source for safety service water and will act as the repository for heat rejected from safety-related and non-safetyrelated reactor auxiliary plant systems and components during all modes of plant operation. The UHS is assumed to be a cooling pond.
- Section 1.5.1 states that a goal of the designs referencing the Evolutionary Requirements Document will be to minimize the possible effects on the plant cooling water systems of mud, silt, organisms, and harsh chemistry of heat sink water.
- Section 1.5.4 requires that direct service water not be used for component cooling. A closed-loop component cooling water system (CCWS) will be used to transfer heat from the component heat loads via a heat exchanger to the SWS and the heat sink.
- Section 2.2.6 requires that the cooling water piping from the heat sink to the SWS be equipped with a filtration system. Provisions for periodic cleaning and backflushing, as necessary for the heat transfer and piping surfaces of the SWS, will also be included. Where required, chemical injection will be included for the removal of organisms.

.

- Section 2.2.7 requires that the systems' designs provide for periodic testing to ensure system adequacy and complete post-maintenance verification of operability.
- Section 2.2.11 requires that all systems be inspectable and testable for operability and cooling capability by using normal and/or periodic testing
- Section 3.3.2.1 specifies appropriate margin in the heat exchanger area to account for fouling and tube plugging. This margin permits a degradation in performance before tube cleaning or retubing is required.
- Section 3.3.2.4 requires that in the shell and tube SWS heat exchanger design, the SWS flow be routed through the tube side to facilitate the control and removal of fouling.
- Section 3.5.1 requires performance tests of key heat exchangers during preoperational testing if heat load is available. The plant designer is required to identify all heat exchangers that cannot be performance tested during preoperation tests and to a velop a test plan and schedule for testing wher heat load becomes available.

- Sections 3.6, 3.7.2, and 3.7.3 require the plant designer to select materials and water chemistry for cooling water systems to account for normal corrosion-erosion as well as those factors unique to cooling water systems, such as microbiologically induced corrosion, high purity water, or long layups. Additional requirements to address the issue of corrosion-erosion include the following:
  - Evaluate the addition of chemicals to the SWS to halt microbiological corrosion or to kill organisms in the water.
  - For the CCWS, provide for the addition of chemicals to the high purity water to reduce corrosion, and recommend the use of wet layup to reduce long-term corrosion of system heat exchangers.
  - Evaluate the types of service water generally available for plant cooling, and prepare a list of recommended system materials for piping, pumps, and heat exchangers operating under those conditions.

The Evolutionary Requirements Document states that eliminating the possible effects of fouling and blockage of the SWS and heat sinks is a design goal. The plant designer is given specific requirements and guidance for achieving this goal, including instructions to consider designs and new requirements that further mitigate the effects of fouling. The final design of the ALWR heat sinks and water flow systems will avoid or minimize, as achievable, the problems described in this section. EPRI's position is that implementation of these requirements should resolve Issue 51 for the ALWR.

<u>Staff Evaluation</u>: The staff's DSER for Chapter 8 contained two open issues related to biofouling. Although the Evolutionary Requirements Document includes a commitment to comply with GL 89-13, the concerns in the DSER, which have not yet been addressed by EPRI, are as follows:

Enclosure 1 of GL 89-13 recommends that an ongoing program of surveillance and control (biocide, chemical) be implemented and maintained to reduce folling problems in open-cycle SWSs. Section 5.4.3 of Chapter 8 of the Evolutionary Requirements Document discusses surveillance for reducing fouling, but not how to control it. Control of biofouling by the addition of biocides or cnemicals should be included in this section, together with surveillance, as the two main elements for reducing fouling. The addition of biocides and chemicals to raw service water is discussed in Section 3.7 of Chapter 8 of the Evolutionary Requirements Document, but that section should be referenced in Section 5.4.3, and vice versa. Biocide injection into raw service water before long wet layups is mentioned in Section 3.7.2. It should also be mentioned in Section 5.4.3 as an effective means of preventing microbiologically influenced corrosion.

Since the Evolutionary Requirements Document still does not provide sufficient commitments, information, or design details regarding this open issue, the staff will evaluate its resolution during its review of individual applications for FDA/DC.

Since GL 89-13 was issued, the staff became aware of the potential problems for plants in the Great Lakes area related to biofouling of SWSs and CCMSs caused by a recently identified biofouling agent, the zebra

EPRI Evolutionary Plant SE?

mussel. The staff issued NRC Information Notice (IN) 89-16, "Biofouling Agent: Zebra Mussel," to alert utilities to the tenacious nature and the unusually swift spread of those mussels in the Great Lakes area since they were first discovered there in 1988. Recently, a number of nuclear power plants in the Great Lakes area, including Fermi Unit 2 and D. C. Cook, have experienced biofouling problems due to zebra mussels in their cooling water systems and/or intake structures. The staff will review the potential for biofouling of SWSs and CCWSs for plants sited in the Great Lakes area on a case-by-case basis.

The DSER also contained an open issue related to heat exchanger testing. In the DSER, the staff concluded that the vendors should consider including the capability for performance testing of key heat exchangers during the planning and design stage. Section 3.5.1 of Chapter 8 of the Evolutionary Requirements Document contains provisions that address the staff's concerns. Therefore, this open issue is resolved.

In addition to the above, Table B.1-2 in Appendix B to Charter 1 of the Evolutionary Requirements Document should be corrected to snow that Chapter 8 is the lead chapter for GL 89-13 instead of Chapter 9. Also, "RESOLVED ISSUE (51)" should be added to the comments column.

3.2.24 57, Effects of Fire Protection System Actuation on Safety-Related Equipment

Issue: This issue addresses fire protection system (FPS) actuations that have resulted in adverse interactions with safety-related equipment at operating nuclear power plants. Events have shown that safety-related equipment subjected to FPS water spray could be rendered inoperable and that numerous spurious actuations of the FPS have been initiated by operator testing errors or by maintenance activities, steam, or high humidity in the vicinity of FPS detectors. The NRC issued Office of Inspection and Enforcement (IE) Informa-tion Notice 30-41, "Actuation of Fire Suppression System Causing Inoperability of Safety-Related Equipment," to alert licensees and provide recent examples in which FPS actuations caused damage or inoperability of systems important to safety. In addition, the staff is considering the need for modifying FPS requirements or licensing review procedures.

EPRI Proposal: Proposed elements for resolving this issue are included in Section 3.9.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

Staff Evaluation: The elements of resolution proposed in the Evolutionary Requirements Document that reference Chapter 9 are evaluated by the staff in Section 3 of Chapter 9 of this report. These elements and the others in Section 3.9.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document appear to adequately address the safety concerns of this issue. However, Issue 57 has not yet been generically resolved by the staff and has a medium safety priority. Since it has not yet finalized its position on this issue for ALWRs, the staff will evaluate the resolution of Issue 57 during its review of individual applications for FDA/DC.

## 3.2.25 70, Power-Operated Relief Valve and Block Valve Relia lity

<u>Issue</u>: Power-operated relief valves (PORVs) and block valves were originally designed as non-safety-grade components in the reactor pressure control system for use only when plants are in operation; the block valves were installed because of expected leakage from the PORVs. Neither valve type was required to safely shut down a plant or mitigate the consequences of accidents. In 1983 the staff determined that PORVs were relied on to mitigate design-basis steam generator tube rupture accidents and questioned the acceptability of relying on non-safety-grade components to mitigate design-basis accidents. This issue addressed assessing the need for improving the reliability of PORVs and block valves.

<u>EPRI Proposal</u>: Proposed elements for resolving this issue are included in Section 3.5.4 of Appendix B to Chapter 1 of the Evolutionary Requirements Document. Section 5.5 of Chapter 5 establishes requirements for a safetygrade safety depressurization and vent system (SDVS) for ALWR PWRs.

<u>Staff Evaluation</u>: In its DSER for Chapter 5 (February 1990), the staff concluded that EPRI's proposed resolution for Issue 70 was consistent with the staff's positions. However, the staff concluded that the Evolutionary Requirements Document should specify a criterion for the depressurization rate of the SDVS. In Section 6.6.4.3 of Chapter 5, EPRI specifies that the SDVS will be capable of depressurizing the reactor coolant system below 250 psig before reactor vessel melt-through as a means of precluding containment challenges through direct containment heating. In Section 5.4 of Chapter 5 of this report, the staff concludes that this is an acceptable design objective. Therefore, this open issue is resolved.

The staff has resolved Issue 70 and has issued GL 90-16, "Resolution of Generic Issue 70, 'Power-Operated Relief Valve and Block Valve Reliability,'" and Generic Issue 94, "Additional low Temperature Overpressure Protection for Light-Water Reactors" (requiring technical specifications revisions at PWRs with PORVs and block valves). It has also drafied revisions of the Standard Review Plan (Sections 3.2.2, "System Quality Group Classifications"; 5.2.2, "Overpressure Protection"; and 5.4.7, "Residual Heat Removal System").

The is -temperature overpressure protection (LTOP) function for the ALWR PWRs is to be provided by the residual heat removal pressure-relief system. In a letter to the NRC dated March 28, 1988, EPRI committed to revise appropriate chapters of the Evolutionary Requirements Document to provide information on LTOP. The staff has confirmed the incorporation of this information in Section 3.3.2 of Chapter 3 of the Evolutionary Requirements Document, and the staff evaluation is documented in Section 3.3 of Chapter 3 of this report. Therefore, this confirmatory issue is resolved.

3.2.26 73, Detached Thermal Sleeves

<u>Issue</u>: Fatigue failure problems associated with nozzle-thermal sleeve assemblies have been identified in piping systems of both BWRs and PWRs.

EPRI Proposal: None provided in Section 3 of Aprendix B to Chapter 1 of the Evolutionary Requirements Document.

Staff Evaluation: Section 3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that it specifically addresses compliance with the high- and medium-priority and nearly resolved generic safety issues included as of January 1990 in NRC's generic issue management control system (GIMCS). As of December 31, 1989, although Issue 73 was identified in Appendix B to NUREG-0933 as being applicable to future plants, it had not yet been prioritized.

As of the first-quarter fiscal year 1992 update of the GIMCS report, Issue 73 has a safety priority of nearly resolved. Consequently, the staff will evaluate the resolution of Issue 73 during its review of individual applications for FDA/DC.

3.2.27 75, Generic Implications of ATWS (Anticipated Transient Without Scram) Events at the Salem Nuclear Plant

Issue: On two occasions in 1983, Salem Unit 1 failed to scram automatically because both reactor trip breakers failed to open on receipt of an actuation signal. In both cases, the unit was successfully tripped by manual action. The failurs of the breakers was attributed to excessive wear from improper maintenance of the undervoltage relays that receive the trip signal from the protection system and result in the breakers opening mechanically. Three separate NRC actions were initiated to address this problem. One was plant specific and was addressed before the restart of Salem Unit 1. The second action was an investigation of the Salem events and the circumstances leading to them.

The third action was the formation of an NRC task force to study the overail generic implications of this event. The results of the task force's work were reported in NUREG-1000, "Generic Implications of ATWS Events at the Salem Nuclear Power Plant.' This action resulted in the issue being broken down into 16 parts. Four parts of the issue required actions by licensees and applicants and were resolved with the issuance of GL 83-28, "Required Actions Based on Generic Implications of Salem ATWS Events," and IF Bulletins 83-01, "Failure of Reactor Trip Breakers (Westinghouse DB-50) To Open on Automatic Trip Signal," and 83-04, "Failure of Undervoltage Trip Function of Reactor Trip Breakers." The remaining 12 parts of the issue were staff actions that have been resolved through licensing actions (7 parts), coverage in the Human Factors Program Plan (2 parts), issuance of IE Bulletin 83-08, "Electrical Circuit Breakers With an Undervoltage Trip Feature in Use in Safety-Related Applications Other Than the Reactor Trip System" (1 part), resolution of an unresolved safety issue (1 part), and staff plans to revise RG 1.33, "Quality Assurance Program Requirements Operation" (1 part).

EPRI Proposal: Proposed elements for resolving this issue are included in Section 3.10.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

Staff Evaluation: The staff has resolved Issue 75 and has issued GL 83-28 and IE Bulletins 83-01 and 83-04. Concerns regarding diversity issues resulting from the ATWS rule are discussed in the DSER for Chapter 5 (February 1990), SECY-20-016 (January 12, 1990), and the Commission's staff requirements memorandum concerning SECY-90-016 (June 26, 1990) and are not addressed here.

In the DSER for Chapter 10, the staff reported on its review of EPRI's proposals addressing this issue against the requirements for existing plants in GL 83-28.

With regard to GL 83-28, Item 1.1 ("Post-Trip Review - Program Description and Procedure"), the staff stated that it would address this matter during its review of individual applications for FDA/DC, since EPRI stated that the program and procedures used by the plant owner to assess unscheduled reactor shutdowns were not within the scope of the Evolutionary Requirements Document.

Regarding GL 83-28, Item 2.1 ("Equipment Classification and Vendor Interface -Reactor Trip System Components") and Item 2.2 ("Equipment Classification and Vendor Interface - Programs for All Safety-Related Components"), the staff concluded in the DSER that the design criteria in the Evolutionary Requirements Document were acceptable, but by themseives, did not provide sufficient information for the staff to complete its review of this issue. The staff stated that it would address these matters during its review of individual applications for FDA/DC.

Regarding GL 83-25, Item 3.1 ("Post-Maintenance Testing - Reactor Trip System Components") and Item 3.2 ("Post-Maintenance Testing - All Other Safety-Related Components"), the staff stated in the DSER that the EPRI requirements for post-installation validation tests and long-term surveillance and maintenance were acceptable for the definition-of-design phase. However, the staff asked EPRI to perform a quantitative analysis of plant risk to justify its 14-day design objective for corrective maintenance of the man-machine interface system equipment discussed in Section 3.5.2.3 of Chapter 10 of the Evolutionary Requirements Document, including the protection system, the plant control system, and the plant information and monitoring systems. EPRI stated (letter to the NRC dated January 28, 199°) that the 14-day requirement in Section 3.5.2.3 was based on the premise that the maintenance technicians will perform a corrective maintenance operation every 2 weeks on the entire population of these systems. Such a maintenance requirement would not be an unreasonable burden on the plant staff and did not subvert the stringent requirements on system reliability. In addition, EPRI changed the first part of the first sentence of Section 3.5.2.3 of Chapter 10 to read: "Corrective maintenance of any of the folicking major parts .... " This change is acceptable because it does not conflict with current regulatory guidance. Therefore, this open issue is closed. However, the staff will evaluate this item in detail during its review of individual applications for FDA/DC.

Regarding GL 83-28, Item 4.2 ("Reactor Trip System Reliability - Preventative Maintenance and Surveillance Program for Reactor Trip Breakers") and Item 4.5 ("Reactor Trip System Reliability - System Functional Testing"), the staff concluded in the DSER that EPRI's design criteria were acceptable (except as noted in Sections 3.6 and 8.3.4 of Chapter 10 of the DSER), but by themselves, did not provide sufficient information for the staff to complete its review of these items. The staff stated that it would address these items during its review of individual applications for FDA/DC.

3.2.28 76, Instrumentation and Control Power Interactions

Issue: Concerns have been raised regarding the potential for accidents or transferts being made more severe as a result of control system failures, including control and instrumentation power supply faults. During the

EPRI Evolutionary Plant SER

licensing review process, the NRC staff performs an audit review of the nonsafety-grade control systems to ensure that an adequate degree of separation and independence is provided between these non-safety-grade systems and the safety systems. On this brsis, it is generally believed that control system failures are not likely to result in loss of safety functions that could lead to serious events or result in conditions that the safety systems are not able to mitigate. However, indepth studies for all non-safety-grade systems had not been performed.

To address this issue, the staff performed an indepth evaluation of the control systems that are typically used during normal plant operation. The staff has completed part of this evaluation and published the results in NUREG/CR-3958, "Effects of Control System Failures on Transients, Accidents, and Core-Melt Frequencies at a Combustion Engineering Pressurized Water Reactor" (March 31, 1986), NUREG/CR-4385, "Effects of Control System Failures on Transients, Accidents, and Core-Melt Frequencies, and Core-Melt Frequencies at a Westinghouse Pressurized Water Reactor" (November 30, 1985), NUREG/CR-1386, "Effects of Control System Failures on Transients, Accidents, and Core-Melt Frequencies at a Westinghouse Pressurized Water Reactor" (November 30, 1985), NUREG/CR-1386, "Effects of Control System Failures on Transients, Accidents, and Core-Melt Frequencies at a Babcock and Wilcox Pressurized Water Reactor" (December 31, 1985), and NUREG/CR-4387, "Effects of Control System Failures on Transients, Accidents, and Core-Melt Frequencies at a General Electric Pressurized Water Reactor" (December 31, 1988). In these reports, the staff identified several specific concerns with some possible solutions, including reduction of the

- potential for steam generator overfill in PWRs by including an automatic high-water-level trip for the main and emergency feedwater systems
- potential for spillover of reactor coolant into the main steamlines of BWRs by providing a safety-grade high-water-level trip of feedwater
- probability of a small-break loss-of-coolant accident (LOCA) by including automatic actuation of pilet-operated relief valve (PURV) block valves to protect against inadvertent PORV lifts
- probability of a small-break LOCA by providing increased injection capability for the safety injection system

Concerns have also been raised regarding the dc power systems, including

- the potential for a critical challenge to the standby engineered safety features (ESFs) by an instrumentation and control power supply fault
- defeat of some of the ESFs called on to mitigate an initiating event caused by the same instrumentation and control power supply fault
- the potential for the complete or partial blinding of the operators to the status of the plan by the same instrumentation and control power supply fault

These concerns have not yet been evaluated by the staff.

EPRI Proposal: The Evolutionary Requirements Document requires a highly reliable safety-grade control system so that non-safety control system

failures will not pose an unacceptable risk. It concains a number of specific design features to address the NRC concerns in regard to this issue, including the following:

- Adverse system interactions identified during the design or review process will be eliminated (Section 2.3.1.5, Chapter 1).
- Safety systems will be provided with independent emergency power backup (Section 2.2.8, Chapter 5).
- At least two independent connections will be made to offsite power (Section 2.2.9, Chapter 5).
- Specified safety system functions will be ensured by redundant divisions (Section 2.3.1, Chapter 5).
- Each division will be independent and separated (Section 2.3.2, Chapter 5).
- Depressurization and cooldown will be possible using safety-grade equipment, assuming an initiating event and the most limiting single failure (Section 2.3.4, Chapter 5).
- Active containment systems will be safety grade and single-failure proof (Section 2.4.1.2, Chapter 5).
- Redundant safety-grade vessel water injection will be available (Section 3.2.1, Chapter 5).
- The decay heat removal system will be redundant and safety grade (Section 3.3.1, Chapter 5).
- The diverse reactivity controls will meet the applicable requirements of GDC 26 of Appendix A to 10 CFR Part 50 (Section 3.5, Chapter 5).
- The design will include very high reliability and low failure propagation characteristics (Section 2.1.4, Chapter 10).
- The design will possess sufficient segmentation and independence characteristics (Section 2.2.3, Chapter 10).
- Automatic control system failures will be addressed (Section 2.2.9, Chapter 10).
- Component failures will be evaluated (Section 3.1.3.4, Chapter 10).
- Appropriate failure modes will be considered in the design (Section 3.5 1.1, Chapter 10).
- Common sensing instrumentation will be minimized (Section 3.5.3.1.1, Chapter 10).
- Each function will use different processors (Section 3.5.3.1.2, Chapter 10).

- Segmented equipment will be physically separated (Section 3.5.3.1.3, Chapter 10).
- Multiplexers will use separate power supplies and be housed in separate enclosures (Section 3.5.3.1.4, Chapter 10).
- The design of the isolation devices will maintain Class 1E integrity in accordance with RG 1.75, "Physical Independence of Electric Systems" (Section 6.2.6.1, Chapter 10).
- Software interfaces between safety-related and non-safety-related systems will be discouraged (Section 6.3.2.2, Chapter 10).

In addition, EPRI specifies requirements for BWR designs that the design include three independent divisions of core coolant inventory control (CIC) and decay heat removal (DHR) systems, the CCIC and DHR divisions be separated, and the CCIC and DHR divisions be independently powered.

EPRI also specifies additional requirements for PWR designs that the design not include provisions for an automatic PORV, leak detection for each pressurizer safety valve be provided, the design include safecy-grade actuation for the emergency feedwater system, the design have two independent divisions for CCIC and DHR functions, the design include two independent residual heat removal divirions, the design provide small-break LOCA coverage with the safety injection system, and the safety depressurization and vent system have two valves in series and leak detection capability.

Staff Evaluation: In its DSER for Chapter 10, the staff stated that EPRI's commitment (letter to the NRC dated December 21, 1990) to comply with G' 89-19 was acceptable and that it would confirm that this commitment was acceptably incorporated into the Evolutionary Requirements Document. Revision 2 of the Evolutionary Requirements Document indicates a commitment to comply with GL 89-19 in Table B.1-2 in Appendix B to Chapter 1, which resolves the DSER confirmatory issue. The staff also stated that the Evolutionary Requirements Document did not provide the necessary design details to be able to determine if the plant-specific design and arrangement will be adequate to resolve this issue. In addition, Issue 76 has not yet been prioritized by the staff and final resolution of this issue may dictate actions for ALWR evolutionary plant designs. Therefore, the staff will evaluate the resolution of this issue and the design details, if necessary, during its review of individual applications for FDA/DC.

3.2.29 79, Unanalyzed Reactor Vessel Thermal Stress During Natural Convection Cooldown

<u>Issue</u>: The upper head region of a PWR reactor vessel (RV) is separated from the lower portion by the upper internals support plate. During power operation, "forced" reactor coolant flow is provided by the reactor coolant pumps (RCP3). For normal reactor shutdowns, RCPs are shut down sequentially as power is reduced, allowing the entire RV to cool down at a controlled rate without significant thermal gradients between the upper and lower portions of the RV. However, sudden loss of forced reactor coolant flow, such as that which occurs during a loss of offsite power, results in a natural convection cooldown (NCC) condition, with little or no cooling flow being supplied to the

EPRI Evolutionary Plant SER

upper head. Additionally, if the NCC proceeds at an unacceptably high cooling rate, substantial thermally induced stresses may occur in the RV closure flanges and studs, possibly exceeding ASME Code allowable stresses.

EPRI Proposal: EPRI recognizes the need to evaluate NCC events as part of ASME Code RV evaluations and states in Section 3.4.2.4 of Appendix B to Chapter 1 of the Evolutionary Requirements Document that an NCC event should be considered to occur at least 30 times over the 60-year design life of the RV.

Staff Evaluation: In the DSER for Chapter 4, Issue 79 was left open pending generic agency resolution. Issue 79 has since been resolved. The staff's efforts to resolve it have been based on a review of a Babcork and Wilcox (B&W) NCC analysis and the results of an NCC analysis by an NRC contractor, both of which were performed for the B&W 177 fuel assembly reactor vessel (B&W 177).

The staff noted limitations associated with the complicated analysis techniques and some calculated stresses as high as 98 percent of that allowed by the ASME Code. Therefore, no definitive conclusion could be made regarding compliance with applicable regulatory criteria for B&W 177s that might experience an NCU that was outside the bounds of the analysis assumptions or for vessels other than B&W 177s and other PWR vessels that might experience a significant NCC event in the future. However, on the basis of the analysis results and qualitative extrapolation of these results, the staff concluded the following:

- The B&W 177 is considered analyzed for NCC events that are bounded by the NCC transient profile shown in Figure 3 of NUREG-1374, "Technical Findings Related to Generic Issue 79."
- It is extremely unlikely that a single NCC event will cause the failure of any existing U.S. PWR reactor vessel, even if a cooldown rate of 100 °F/hour is exceeded.
- An NCC event that did not exceed a total cool 'own of 100 °F, independent of rate, would not be expected to compromise the safety of any existing U.S. PWR reactor vessel; wever, it might result in the reactor vessel being outside its documented design basis.

For the above reasons and since the Evolutionary Requirements Document does not provide sufficient commitments regarding the cooldown rate to be used in ALWR NCC analyses, the staff will evaluate this matter during its review of individual applications for FDA/DC. An applicant's proposed resolution of this issue will be acceptable if it includes a requirement to analyze an NCC event from 100-percent reactor power to cold shutdown as part of the reactor vessel ASME Code design requirements and the analysis (1) assumes 30 NCC events for the 60-year design life of the reactor vessel and (2) uses the maximum allowable cooldown rate specified h ALWR technical specifications.

In the DSER, the staff concluded that EPRI needed to recognize that the reactor pressure vessel should be capable of withstanding multiple natural circulation cooldowns. Section 3.4.2.4 of Appendix B to Chapter 1 of the Evolutionary Requirements Document adequately addresses this concern by

EPKI Evolutionary Plant SER 18.3-34

stating that NCC events should be considered to occur at least 30 times over the 60-year design life of the reactor vessel. Therefore, this open issue is resolved.

Issue 79 has not been generically resolved by the staff. The final agency resolution of this issue is included in Generic Letter 92-02, "Resolution of Generic Issue 79, 'Unanalyzed Reactor Vessel (PWR) Thermal Stress During Natural Convection Cooldown'." The GL did not impose any new requirements on licensers or applicants. Rather, it advised recipients (PWR licensees and applicants) of the reporting requirements delineated in 10 CFR 50.73(a)(2)(ii)(B). These reporting requirements are also applicable to ALWRs.

Specifically, for any PWR (including any PWR-type ALWR) that experiences a reportable NCC which may place the RV outside of its documented design basis, the affected licensee may be required to provide confirmation that no applicable regulatory design or fracture toughness criteria have been exceeded for the RV.

3.2.30 82, Beyond-Design-Basis Accidents in Spent Fuel Pools

<u>Issue</u>: A typical spent fuel storage pool with high-density storage racks can hold about five times the fuel in the core. If the pool were to be drained of water, the discharged fuel from the last two refuelings might still be "fresh" enough to melt under decay heat. Additionally, the zircaloy cladding of this fuel could be ignited during the heatup. The resulting fire, in a pool equipped with high-density storage racks, could spread to most or all of the fuel in the pool. This could cause a release of fission products from the fuel matrix.

<u>EPRI Proposal</u>: The Evolutionary Requirements Document contains the following requirements to decrease the likelihood of a loss of fuel pool cooling in the ALWR spent fuel pool:

- Section 4.3.2.1 of Chapter 1 requires that the spent fuel pool be a seismic Category I structure.
- Section 2.3.1.1.4.1 of Chapter 7 requires the spent fuel pool to be designed to withstand heavy load drops without pool leakage that would uncover the top of the fuel.
- Section 2.3.2.5.1 of Chapter 7 requires that the spent fuel pool be arranged to prevent cask movement over the pool.
- Section 9.3.1.2 of Chapter 8 does not permit any connections to the pool that could allow the pool to be drained below the minimum level over the spent fuel.

<u>Staff Evaluation</u>: In the DSER for Chapter 7, the staff concluded that the design requirements were consistent with the guidance in SRP Section 9.1.2, "Spent Fuel Storage," and were, therefore, acceptable. The staff stated, however, that although the likelihood of the complete draining of the spent fuel pool was low, the use of high-density storage racks increased the probability of a zircaloy-cladding fire as compared with the use of low-density or open-frame racks. The staff concluded that the use of low-density

EPRI Evolutionary Plant SER 18

storage racks was justified by a favorable value/impact ratio for new designs and recommended that EPRI make a commitment to use low-density storage racks, at least for the most recently discharged fuel. Therefore, the staff will expect the FDA/DC applicant to submit a design that uses low-density storage racks in the spent fuel pool for, as a minimum, the most recently discharged fuel.

## 3.2.3, 83, Control Room Habitability

<u>Issue</u>: This issue addresses the significant discrepancies found by the staff during a survey of existing plant control rooms. These discrepancies included the inconsistency of the design, construction, and operation of the control room habitability systems with the descriptions in the licensing-basis documentation. In addition, total system testing was inadequate and some testing of the control systems was not in accordance with the technical specifications.

EPRI Proposal: Proposed elements for resolving this issued are included in Section 3.5.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

<u>Staff Evaluation</u>: The staff has determined that the design requirements in Section 8.2.2.1 of Chapter 9 of the Evolutionary Requirements Document for the control room envelope heating, ventilating, and air conditioning (HVAC) system are consistent with the guidance in SRP Sections 6.4, "Control Room Habitability Systems," and 9.4.1, "Control Room Area Ventilation System," and are, therefore, acceptable. Section 8.2.2 of Chapter 9 of this report. However, the requirements by commended on ot provide sufficient information to be able to determine if the plant-specific design and arrangement will be adequate. Therefore, FDM/DC applicants will be required to demonstrate compliance with the eduitional guidance provided in the SRP or provide justification for the enalty means of implementing the associated regulatory requirements.

The open issue in the DSER for Chapter 9 concerning the design of air filtration systems is closed on the basis of the clarification in Section 8 ? 1.1.8 of Chapter 9 of the Evolutionary Requirements Document that the intent was to require application of the regulatory guides and reference standards to all nuclear air treatment filtration systems, not just the filters.

The open issue in the DSER concerning charcoal filters in air filtration systems is closed on the basis of EPRI's position that the use of charcoal filters in the filtration systems will be determined by analyses and evaluations on a plant-specific basis. The staff finds that Section 8.2.1.1.22 of Chapter 9 addresses the provisions for the use of charcoal filters in HVAC systems and provides design features to process activity from normal and offnormal operation. Section 8.2.1.1.8 of Chapter 9 requires that all safetyrelated nuclear air treatment filtration systems be designed, fabricated, installed, and tested in accordance with RG 1.52, "Design, Testing, and Maintenance Criteria for Postaccident Engineered-Safety-Feature Atmosphere Cleanup System Air Filtration and Ads option Units of Light-Water-Cocled Nuclear Power Plants," and ANSI/ASME w509, "Nuclear Power Plant Air Cleaning Units and Components," and NS10, "Testing of Nuclear Air Cleaning Systems." Also, Section 8.2.2.1.8 requires that emergency filter units (EFUs), at a minimum, consist of a prefilter, high-efficient particulate air (HEPA)

EFRI Evolutionary Plant SER

filter, and a supply fan in accordance with ANSI/ASME N509 and RG 1.52. However, RG 1.52 needs to be met by providing each EFU with the following sequential components: (1) demister, (2) prefilter, (3) HEPA filter, (4) iodine adsorber (impregnated activated carbon), (5) ducts and values, (6) fan, and (7) related instrumentation. Heaters or cooling coils used in conjunction with heaters should be used when the humidity is to be controlled before filtration. The downstream HEPA filter is needed to collect carbon fines from the upstream charcoal filter. The staff considers that it is incorrect to assume the regulatory guide efficiencies for the removal of elemental and organic iodine from the emergency outside air supply if there is no HEPA filter downstream of the charcoal adsorber for collecting carbon fines. The staff believes that an EFU, including a charcoal filter, should be safety related and the post-filter following a charcoal filter should be a HEPA filter to collect carbon fines. The staff will review individual applications for FDA/DC to the criteria in SRP Sections 6.4 and 9.4.1 to meet the requirements of GDC 19 of Appendix A to 10 CFR Part 50.

The open issue in the DSER concerning the control room capacity following a design-basis accident is closed because (1) Section 8.2.2.1.1 of Chapter 9 of the Evolutionary Requirements Document has been revised to state that the main control room HVAC system will be designed as a Safety Class 3, s. ismic Category T system; (2) the same section has been revised to change the occupancy requirer t from 5 days to at least 6 days; and (3) Section 8.2.2 of Chapter 9 has been revised to include the requirement of 1C CFR 73.55(c)(6) that the control room be bullet resistant. The staff will review individual applications for FDA/DC to the criteria in SRP Sections 5.4 and 9.4.1 to meet the requirements of GDC 19 during the entire duration of postulated accidents.

Section 8.2.2 of Chapter 9 states that the control complex will include the main control room envelope, computer room, Class 1E switchgear room, Class 1E battery rooms, and HVAC equipment room. The HVAC system for the main control room will be seismic fategory I, Safety Class J, and will conform to the requirements of GDC 15 with regard to the radiation exposure guidelines and guidance of SRP Sections 6.4 and 9.5.1; RGs 1.78, "Assumptions for Evaluating the Habitability of a Nuclear Power Plant Control Room During a Postulated Hazardous Chemical Release," and 1.95, "Protection of Nuclear Power Plant Control Room Operators Against an Accidental Chlorine Release"; and ANSI/ANS-59.2, "Safety Criteria for Nuclear Power Plant HVAC Systems Located Outside Primary Containment," for providing a safe working environment for control room operating personnel. It will provide protection against (1) any airborne hazardous chemicals above permissible concentrations, smoke in the control room, or accidental exposure of control room personnel to a high level of airborne radioactivity during emergency or isolation modes and (2) dust, smoke, and airborne radioactivity originating outside the pressure boundary Juring normal operation. The control room envelope will be slightly pressurized during normal operation and at least 1/8-inch water gauge will be maintained with respect to the surrounding areas during radiological conditions to prevent the entry of dust, smoke, and airborne radioactivity.

EPRI states that the control room HVAC system and other specific areas will be designed in accordance with National Fire Protection Association (NFPA) 90A, "Installation of Air Conditioning and Ventilating Systems." As stated in Section 3.3.1 of Chapter 9 of this report, by committing to NFPA 90A, EPRI has committed to use the HVAC systems for removing smoke from specific areas as a means of satisfying the smoke control provisions of NRC fire protection

EPRI Evolutionary Plant SER

guidelines. The staff will evaluate the design, installation, and operation of the HVAC systems functioning in the smoke removal mode during its review of individual applications for FDA/DC.

In Section 3.5.1.4 of Appendix B to Chapter I, EPRI states that the resolution of Issue 83, in general, is not design related. EPRI considers the concerns of this issue to be concerns about operating plant activities in the areas of maintenance, testing, installation, and training and, therefore, beyond the scope of the Evolutionary Requirements Document. However, Issue 83 addresses significant discrepancies found during a survey of existing plant control rooms. These discrepancies included inconsistencies in the design, construction, and operation of the control room habitability systems (compared with the descriptions in the licensing-basis documentation). In addition, the staff determined that total system testing was inadequate and that some testing of the control systems was not in accordance with the technical specifications.

The staff considers EPRI's proposed resolution of Issue 83 to be complete, subject to the staff's plant-specific reviews, as applicable, of individual applications for FDA/DC. During its reviews, the staff will ensure that the existing minimum requirements, including those mentioned above, have been met, by conforming to the guidance of SRP Sections 6.5 and 9.4.1 (to meet GDC 19). Appropriate requirements are identified in the Evolutionary Requirements Document that include conformance with the applicable regulatory guides (RGs 1.52, 1.78, and 1.95) and industry standards (ANSI/ANS-59.2, ANSI/ASME N509, and ANSI/ASME N510). However, during its plant-specific reviews, the staff will also verify conformance with industry standards (ASME Code AG-1. "Code on Nuclear Air and Gas Treatment," and American Society for Testing and Materials (ASIM) D3803, "Standard Test Methods for Radiological Testing of Nuclear-Grade Gas-Phase Absorbents") and evaluation methodology and verify that technical specifications and surveillance requirements for the control room habitability system as a whole to meet GDC 4, 5, and 19 have been proposed. This will demonstrate that the control room will adequately protect the control room operators and will remain habitable in accordance with TMI Task Action Plan Item III.D.3.4 (NUREG-C737, "Clarification of TMI Action Plan Requirements"). Additionally, the FDA/DC applicant will have to ensure that the control room habite flity design meets GDC 4, 5, and 19 and that operators are protected in accordance with TMI Action Plan Item III.D.3.4. The combined license applicant will have to verify (1) that the as-built design and the operating, maint since, and emergency procedures and training and the performance characteristics of the control room habitability system are consistent with the licensing-basis documentation and (2) that the technical specifications and surveillance procedures are consistent with the licensing-basis documentation. The combined license applicant will also have to provide adequate verification of system performance and integrity.

A possible resolution of Issue 83 has been identified. This resolution is documented in a staff report (H. Denton to W. Dircks, Executive Director for Operations) dated June 19, 1984, and a contractor report (T. Powers, Pacific Northwest Laboratory, to W. Milstead, NRC) dated December 3, 1984, both on control room habitability. Efforts included a survey of 11 plants and the issuance of several other contractor reports. Current efforts include preparation of a generic letter and SRP revisions. Should final agency resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue 85 for staff evaluation.

3.2.32 84, Combustion Engineering Power-Operated Relief Valves

<u>Issue</u>: Following the IMI-2 accident, the purpose and use of power-operated relief valves (PORVs) was the subject of considerable analyses and discussion. Although PORVs were originally installed to prevent challenges to the springoperated safety valves, plants designed by Westinghouse and Babcock and Wilcox (B&W) sometimes rely on PORVs for depressurization during certain design-basis events (e.g., steam generator tube rupture). Another use of PORVs at some plants is to provide low-temperature overpressure protection.

When this issue was being evaluated, all Westinghouse and B&W and older Combustion Engineering (CE) PWRs had PORVs. However, the newer CE designs did not include PORVs. The staff's review of this issue in 1984 showed that existing CE plants without PORVs met regulatory requirements, but that other considerations, primarily accident management for beyond-design-basis events and potential core-melt risk reduction, required further study. However, the staff considered events for which PORVs could prove to be of benefic to have a low probability of occurrence and was unaware of any immediate safety concerns associated with the absence of PORVs in CE-designed plants.

EPRI Proposal: Proposed elements for resolving this issue are included in Section 3.5.4 of Appendix B to Chapter 1 of the Evolutionary Requirements Document. PWR ALWRs will not have PORVs. Instead, the Evolutionary Requirements Document calls for a single-failure-proof safety depressurization and vent system (SDVS). The SDVS will be a dedicated safety-grade system that will be capable of performing those safety-related functions, with the exception of the low-temperature overpressure protection function, that can be performed by PORVs in the current generation of PWRs.

<u>Staff Evaluation</u>: In its DSER for Chapter 5, the staff concluded that the Evolutionary Requirements Document needed to specify a criterion for the depressurization rate of the SDVS. In Section 6.6.4.3 of Chapter 5, EPRI specifies that the SDVS will be capable of depressurizing the reactor coolant system below 250 psig before reactor vessel melt-through as a means of precluding containment challenges through direct containment heating. In Section 5.4 of Chapter 5 of this report, the staff concludes that this is an acceptable design objective. Therefore, this open issue is closed.

The staff has resolved Issue S4 and has established no new requirements. As discussed in Section 3.1 of this appendix, Issue 84 is set included in Appendix B to NUREG-0933.

3.2.33 87, Failure of High-Pressure Coolant Injection Steamline Without Isolation

<u>Issue</u>: This issue addresses a postulated break in the high-pressure coolant injection (HPCI) steam supply line and the uncertainty regarding the operability of the HPCI steam supply line isolation valves under those conditions. A similar situation can occur in the reactor water cleanup (RWCU) system. The HPCI steam supply line has two containment isolation valves in series (one inside the containment and one outside), both of which are normally open in most plants (two plants do operate with the outboard isolation valve normally closed). An HPCI supply valve located adjacent to the turbine an, the turbine stop valve are normally closed. The RWCU system also has two normally open containment isolation valves that must remain open if the system is to function.

At the valve manufacturers' facilities, only the opening characteristics are tested under operating conditions (because of flow limitations). Although the operation of the valves is tested periodically without steam, the capability of the valves to close against the forces created by the steam flow resulting from a downstream line break has not been demonstrate.

The valve type is not under General Electric's (bwk vendor) scope of control, but is selected by the plant architect-engineer. This results in a diversity of valves and valve types (Y-type globe valves and gate valves) and increases the difficultly of demonstrating valve operating capability.

EPRI Proposal: In Section 3.5.5.5 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI asserts that the motor-operated valve (MOV) testing and surveillance issue raised by Issue 87 should be considered resolved for the ALWR evolutionary plant designs. EPRI gives two arguments as a basis for this statement.

First, EPRI asserts that the NRC concerns about the performance of safetyrelated MOVs under design-basis conditions are addressed by the specific requirements sections of the Evolutionary Requirements Document. Second, EPRI states that it will take full advantage of its program to develop an NOV performance prediction methodology.

Staff Evaluation: The staff has resolved Issue 87; it issued a closeout memorandum (E. Beckjord to J. Taylor) on December 9, 1991.

The staff does not agree with EPRI's position that the MOY issue should be considered resolved for the ALKG evolutionary plant. For example, as discussed in Section 12.2.2 of Chapter 1 of this report, the specific requirements sections of the Evolutionary Requirements Document have not fully resolved NRC staff concerns regarding the performance of MOVs in the ALWR evolutionary plant. Also, EPRI's MOV performance prediction program is in its initial stages and cannot be relied on until successful results begin to be obtained. Finally, the ALWR evolutionary plant designs contain a few highly important MOVs (such as the RWCU system isolatic MOVs) that would need to function properly in the event of an abnormal plant condition.

The staff will evaluate this matter during its review of individual applications for FDA/DC in accordance with the above position and its positions on MOVs stated in Section 12.2.2 of Chapter 1 of this report.

3.2.34 94, Additional Low-Temperature Overpressure Protection for Light-Water Reactors

<u>Issue:</u> Low-temperature overpressurization was originally identified as a safety issue in November 1978. The resolution of this safety issue was given in a revision to SRP Section 5.2, "Overpressure Protection." However, from

EPRI Evolutionary Plant SER

1979 to July 1983, 12 pressure transients were reported to the NRC. Of these 12 pressure transients, 2 overpressurization transients suggested potential weaknesses in the existing overpressure protection criterion or in its implementation. Specifically, these two overpressurization transients suggested weaknesses relating to the operability and reliability of both trains of low-temperature overpressure protection when the plant was in a solid (water solid) condition.

Major overpressurization of the reactor coolant system, if combined with a critical-size crack, could result in a brittle failure of the reactor vessel. Failure of the reactor vessel could make it impossible to provide adequate coolant to the reactor and result in major core damage or a core-melt accident. This issue mainly affects the design and operation of PWRs.

EPRI Proposal: In Section 3.3.3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI discusses low-temperature overpressure protection (LTOP) for ALWRs. EPRI's design objective is to minimize the need for any special LTOP, while providing the capability for reliable LTOP should it be required. To accomplish this objective, EPRI states that primary system components are to be made of materials with a low nil ductility transition temperature. In addition, adequate relief capacity is to be provided for LTOP.

Section 2.2.2 of Chapter 4 of the Evolutionary Requirements Document specifies that materials, design, and fabrication methods for the reactor pressure vessel are to selected so that the materials will not behave in a brittle manner, be subject to rapidly propagating failure, or otherwise fail under normal operations and specified plant transients or events. Section 2.3.1.2 of Chapter 4 requires that the reactor pressure vessel base material and welding material chemistry be controlled so that material susceptibility to neutron embrittlement is as low as practicable. Section 3.3.2 of Chapter 3 provides requirements for LTOP in PWRs. Section 5.2.3.3.1 of Chapter 5 requires that relief valves be provided to ensure that the residual heat removal design pressure will not be exceeded as a result of the most limiting event. In addition, guidelines for relief capacity for LTOP are specified in Section 5.2.3.3 of Chapter 5.

Staff Evaluation: In the DSERs for Chapters 3 and 4, the staff noted that EPRI's additional requirements for LTOP for ALWRs appeared to be adequate and reasonable. However, the staff was unable to make a final judgment on the adequacy of EPRI's LTOP provisions because it was considering new requirements as part of the resolution of Issue 94. Therefore, this was identified as an open issue in both DSERs.

The staff has established guidance for resolving the LTOP concerns for operating reactors in NUREG-1326, "Legulatory Analysis for the Resolution of Generic Issue 94, 'Additional Low-Temperature Overpressure Protection for Light-Water Reactors,'" but has not yet finalized its position for ALWRs. Therefore, it will evaluate the adequacy of the specific design for LTOP during its review of individual applications for FDA/DC.

3.2.35 96, Residual Heat Removal Suction Valve Testing (integrated into Issue 105)

Issue 96 has been subsumed by Issue 105. See Section 3.2.39 for the staff's evaluation of Issue 105.

3.2.36 99, Reactor Coolant System/Residual Heat Removal Suction Line Valve Interlock in PWRs

According to Appendix B to NUREG-0933, Issue 99 is not applicable to future plants. See the staff's evaluation of Issue 105 in Section 3.2.39.

3.2.37 101, BWR Water Level Redundancy

Issue: This "ssue concerned the potential for a postulated break in an instrument line of a BWR in conjunction with the worst-case single failure. Should one of two reference columns break, a single failure associated with the other reference column could completely defeat mitigation systems for the resulting transients. BWRs typically have two reference columns in the water level instrumentation. Water level is measured by means of differential pressure sensors connected between the reactor vessel and the reference columns, which are full of water. Should a reference column break, the water in it would flash to steam and the water level indication in all channels connected to the broken column would give a false "high" reading. Should one of the reference columns break and should a worst-case single failure cause the other column to indicate high water level, the false signals could result in a low-water-level situation that could prevent mitigation systems, such as the emergency core cooling system, from initiating. In some plants, this specific scenario could lead to core uncovery. However, because there are several designs among the operating plants, there is not a single, generic solution to resolve the problem.

EPRI Proposal: Chapter 10 of the Evolutionary Requirements Document lists requirements for the design process and on reliability that should ensure that the designs complying with the Evolutionary Requirements Document will not be subjected to the postulated scenario. Section 3.1.3.1 of Chapter 10 requires that the man-machine interface system (M-MIS) designer review existing LWR plant designs to identify problems and develop their solutions. EPRI states that the requirements for availability and reliability of Section 3.5.1.2 will ensure that no single random failure of any M-MIS equipment will cause a forced outage. Sections 3.5.3 and 3.5.4.3 address multiple random failures of M-MIS equipment. In addition, Section 3.3.3.2.1 of Chapter 4 requires four divisions of wide-range level instruments with four spatially separated taps. EPRI considers that implementation of these requirements resolves this issue.

Staff Evaluation: In its DSER for Chapter 10, the staff concluded that EPRI's commitment (letter to the NRC dated December 21, 1990) to include compliance with the resolution of Issue 101 in accordance with GL 89-11, "Resolution of Generic Issue 101, 'Boiling Water Reactor Water Level Redundancy,'" in the Evolutionary Requirements Document was acceptable. The staff has confirmed that Table B.1-2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document indicates a commitment to comply with GL 89-11; therefore, this confirmatory issue is resolved. However, the Evolutionary Requirements

Document does not provide sufficient information for the staff to complete its review of this issue. The staff will evaluate resolution of Issue 101 during its review of individual applications for FDA/DC.

3.2.38 103, Design for Probable Maximum Precipitation

<u>Issue</u>: Lack of adequate drainage at reactor sites may render safety-related equipment inoperable. New procedures for estimating the probable maximum precipitation (MP) result in a more severe design-basis probable maximum flood than that determined using previous methodologies. In 1989, the staff revised SR. Sections 2.4.2, "Floods," and 2.4.3, "Probable Maximum Flood (PMF) on Streams and Rivers" (Revision 3) to incorporate the PMP procedures and criteria contained in the latest National Weather Service publications. In addition, on October 19, 1989, the staff issued GL 89-22, "Potential for Increased Roof Loads and Plant Area Flood Runoff Depth at Licensed Nuclear Plants Due to Recent Change in Probable Maximum Precipitation Criteria Developed by the National Weather Service," to inform the industry of these revisions.

<u>EPRI Proposal</u>: Table B.1-2 in the Evolutionary Requirements Document indicates that the ALWR design will meet SRP Sections 2.4.2 and 2.4.3 (Revision 3) and GL 89-22.

<u>Staff Evaluation</u>: Section 3.3.2 of Chapter 6 of the Evolutionary Requirements Document (Revision 0) re renced ANSI/ANS-2.8-1981 as part of the resolution of this issue at the time the DSER for Chapter 6 was issued. At that time, the staff had not reviewed this standard for consistency with regulatory requirements and staff guidance. Consequently, the staff did not complete its review of Issue 103 in the DSER and left review of this issue open.

On the basis of the commitment in the current version of the Evolutionary Requirements Document that the ALWR design will meet SRP Sections 2.4.2 and 2.4.3 (Revision 3) and GL 89-22, the staff concludes that EPRI has adequately addressed the resolution of Issue 103.

3.2.39 105, Interfacing Systems Loss-of-Coolant Accident at LWRs (subsumes Issue 96)

<u>Issue</u>: Either human errors, valve leakage, multiple valve failures, or a combination of these could permit overpressurization of low-pressure systems connected to high-pressure systems. Pipe or component rupture resulting from overpressurization could result in loss of decay heat removal if the rupture was outside the containment.

<u>EPRI Proposal</u>: Section 3.5.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that "the ALWR is being required to satisfy all of the current regulatory requirements which are intended to assure an acceptable level of isolation capability." This section also indicates that all ALWRs are required to satisfy the following additional requirements:

 All systems and subsystems connected to the reactor coolant system (RCS) that extend outside the primary containment boundary will be designed to the extent practicable to an ultimate rupture strength (URS) at least equal to full RCS pressure.

- For those interfacing systems or subsystems that do not meet the full reactor coolant system URS requirement, the plant designer will determine by evaluation if the degree and quality of isolation or reduced severity of the potential pressure challenges are low enough to preclude an intersystem loss-of-coolant accident (ISLOCA).
- Each of the high- to low-pressure interfaces (viz., those between the systems normally exposed to the RCS pressure and the interfacing lower pressure system) will include the following protective measures:
  - the call ility for leak testing of the pressure isolation valves
  - valve position indication that is available in the main control room even when isolation valve operators are deenergized
  - high-pressure alarms to warn control room operators when rising RCS pressure approaches the design pressure of the attached low-pressure systems and both isc ation valves are not closed

Additional recuirements for BWRs are the following:

ø

- An inboard testable check valve and an outboard motor-operated valve are to be used on the decay heat removal (DHR) injection lines. High-pressure piping and valves are to be used through the outboard isolation valve. Pressure interlocks are to protect low-pressure piping upstream of the outboard isolation valve.
- Check valves will be testable to verify free movement of the valve disk.
- Relief valves are to be provided in the low-pressure pump discharge piping to protect against overpressure due to backleakage from the RCS.
- The portions of the DHR system that connect directly to the RCS will be designed to a URS at least equal to full RCS pressure.
- Interlocks are to be provided as follows:
  - a two-way interlock so that it is not possible to have an open shutdown connection to the vessel in any given loop whenever the pool suction, pool discharge valve, or wetwell spray valves are open in the same loop
  - redundant interlocks to prevent the opening of the shutdown connections to and from the vessel whenever the pressure is above the shutdown range with increasing pressure causing closure of these valves
  - redundant interlocks preventing the opening or closing of the shutdown suction connections to the vessel in any given loop and discharge to radioactive waste whenever a low-reactor-water-level signal is present



1.4







25 1.4





PHOTOGRAPHIC SCIENCES CORPORATION

770 BASKET ROAD P.O. BOX 338 WEBSTER, NEW YORK 14580 (716) 265-1600



25

1.4







1.25

c1



150mm

6



Additional requirements for PWRs are the following:

- The residual heat removal (RHR) system will be designed for a pressure of 900 psig and a temperature of 400 °F and will use a minimum of Schedule 40 piping.
- Relief valves will be provided to ensure that RHR system design pressure will not be exceeded as a result of the most limiting event. The plant designer is to define the most limiting event on the basis of a review of experience will overpressure transients during RHR operation, including such events as inadvertent startup of the safety injection system and startup of reactor coolant pumps with coolant in the steam generators that is above average RCS temperature.
- Interlocks will be provided for the RHR suction isolation values to prevent the values from opening in the event RCS pressure exceeds RHR design pressure. An interlock that automatically closes the isolation values on high pressure is not to be provided.
- Portions of the RHR system that connect directly to the RCS will be designed to ensure that the URS will not be exceeded at full RCS pressure.

<u>Staff Evaluation</u>: The staff's DSER for Chapter 5 contained an open issue relating to low-pressure system: not designed to withstand full RCS pressure. The staff recommended that evolutionary ALWRs provide

- the capability for leak testing of the pressure isolation valves
- valve position indication that is available in the control room when isolation valve operators are deenergized
- high-pressure alarms to warn control room operators when rising RCS pressure approaches the design pressure of attached low-pressure systems and both isolation valves are not closed

This recommendation is reflected by requirements in Section 2.2.14.3 of Chapter 3 of the Evolutionary Requirements Document; therefore, this open issue is resolved.

SECY-90-016 discusses the ISLOCA, focusing on a UKS of systems and subsystems connected to the RCS that is at least equal to the full RCS pressure as a way of reducing the probability of an ISLOCA outside the containment. In SECY-90-016, the staff concludes that designing low-pressure systems to withstand full RCS pressure, to the extent practicable, is an acceptable means of resolving the ISLOCA issue. (Section 2.2.14.1 of Chapter 3 of the Evolutionary Requirements Document provides a requirement to this effect.) Nevertheless, the staff considers alternative design approaches, such as inclusion of pressure relief valves in low-pressure system piping, acceptable in principle, provided ; e consideration is given to 'elief capacity, effluent conservation, possible defeat of pressure isolation valve interlocks, and relief valve reliantity. Also in this regard, the staff noted in SECY-90-016 that for some low-pressure systems attached to the RCS, it might not be practical or necessary to provide a higher system ultimate pressure capability for the entire low-pressure system (this is also noted in the DSER). However, uniform criteria and design stresses allowable for ISLOCA conditions in low-

EPRI Evolutionary Plant SER

Sk

0

5

pressure piping systems have yet to be developed and approved by the staff for ALWRs, although ASME has been asked to consider developing Boiler and Pressure Vessel Code requirements for ISLOCA design in future reactors. In the meantime, the staff will evaluate exceptions on a case-by-case basis during its review of individual applications for FDA/DC.

In a staff requirements memorandum (SRM) dated June 26, 1990, the Commission approved the staff's position in SECY-90-016 on ISLOCA, provided all elements of the low-pressure system are considered (e.g., instrument lines, pump seals, heat exchanger tubes, and valve bonnets).

The staff concludes that the Evolutionary Requirements Document contains requirements that are sufficient to resolve this issue. However, since the document does not specifically identify interfacing systems other than the DHR or RHR systems, the staff will evaluate other interfacing systems designs during its review of individual applications for FDA/DC. It will ensure that all elements of low-pressure systems have been considered in accordance with the SRM of June 26, 1990, particularly those to be evaluated under Section 2.1 14.2 of Chapter 3 of the Evolutionary Requirements Document.

As noted in the DSER, for additional assurance that low-pressure systems will not be exposed to RCS pressure, the staff will review inservice testing programs and technical specifications for ALWR applications to ensure that the appropriate pressure isolation valves are identified and adequately tested.

Issue 105 has not yet been generically resolved by the staff and has a high safety priority. Should the final resolution of this issue indicate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue 105 for staff evaluation.

3.2.40 106, Piping and Use of Highly Combustible Gases in Vital Areas

<u>Issue</u>: Issue 106 addresses the risk associated with the use of hydrogen and other combustible gases, such as propane and acetylene, during normal plant operation. It does not cover the use of large quantities of liquid hydrogen at hydrogen water chemistry (HWC) installations at BWRs or liquified petroleum gases that are covered under Licensing Issue 136.

<u>EPRI Proposal</u>: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue 106 was determined, through the NUREG-1197 screening process, to be not technically relevent to the ALWR design.

Table B.1-2 in Appendix B to Chapter 1 of the Evolutionary Requirements Document shows an ALWR commitment to comply with SRP Section 9.5.1, Revision 3, "Fire Protection Program."

<u>Staff Evaluation</u>: In 1987, the staff accepted licensing topical report EPRI NP-5283-SR-A, "Guidelines for Permanent BWR Hydrogen Water Chemistry Installations," which provides guidelines for HWC installations. The guidelines give a number of system design features and administrative controls that are in addition to, or more restrictive than, those in SRP Section 9.5.1. These include

- new relations for separation distances for gaseous and liquid hydrogen storage facilities
- use of color-coded piping and warning signs (ANSI A13.1, "Scheme for the Identification of Piping Systems," and ANSI Z35.1, "Specification for Accident Prevention Signs")
- excess flow valves, system trips, and other design features (e.g., hydrogen detectors) to help mitigate the consequences of leaks or breaks and to perform the intended design function with or without normal ventilation, and as a minimum, a system trouble alarm and/or annunciator in the main control room
- periodic retesting to verify operability and functional performance of equipment

In its letter transmitting the safety evaluation report (SER) on Report NP-5283-SR-A to EPRI, the staff recommended that the report be modified to include hydrogen systems supplying hydrogen to the volume control tank at PWRs and for cooling electric generators at PWRs and BWRs. With the exception of information dealing specifically with the HWC application (e.g., certain trips, injection points, main steamline radiation), most of the EPRI guidelines dealing with hydrogen are applicable to these other uses. Although EPRI did not modify the report as suggested by the staff, the staff used the information in this report in evaluating Issue 106 for evolutionary ALWRs.

The Evolutionary Requirements Document does not address the reasons for determining that this issue is not technically relevant and does not provide sufficient information to enable the staff to fully evaluate EPRI's proposed requirements related to resolution of this issue. Consequently, the staff will evaluate the resolution of Issue 106 during its review of individual applications for FDA/DC. This issue may be resolved with respect to the volume control tank and generator applications by an applicant's commitment to comply with

- the current SRP Section 9.5.1, Branch Technical Position (BTP) CMEB 9.5-1, Part C.5.d(5), modified as shown below
- the staff's SER and the guidelines in EPRI NP-5283-SR-A pertaining to hydrogen that do not deal specifically with the HWC application
- the following general guidance:
  - BTP CMEB 9.5 1, Part C.5.d(5), modified as follows: "Hydrogen lines in safety-related areas should follow the guidance of Regulatory Guide 1.29, 'Seismic Design Classification,' Section C.2. The lines should (1) be equipped with an excess flow valve or equivalent protection located outside the building so that in case of a line break, the hydrogen concentration in affected areas does not exceed 2 percent volume or (2) be sleeved with the outer pipe vented directly to the outside."
  - The hydrogen system piping and components should be located to reduce risk (e.g., by reducing piping length and proximity to safety-related equipment).

EPRI Evolutionary Plant SER 18.3-47

Design features (e.g., removable spool pieces) and/or administrative controls should be provided to prevent inadvertent bypass of small or normally isolated hydrogen supplies, or flow-limiting devices should be used to limit hydrogen releases to a leak or break.

- Equipment and controls to mitigate the consequences of a hydrogen fire or explosion should be accessible and remain functional during an event.
- Design features and/or administrative controls should be provided to isolate the hydrogen supply if normal building ventilation is lost (e.g., as a result of building isolation caused by operator actions following a hydrogen fire or explosion in an auxiliary building).
- Backflow to a leak or break of hydrogen contained in components (e.g., generator) should be considered in evaluating the consequences of leaks or breaks, and measures should be taken to mitigate these consequences.
- Threaded joints in the hydrogen distribution lines within safetyrelated areas should be back welded.
- Safety-related equipment should not be located in the turbine building because of the hazards associated with hydrogen fires or explosions and large oil fires and the large uncertainties in estimating the consequences.

Issue 106 has not yet been generically resolved by the staff and has a medium safety priority. Should the final resolution of this issue indicate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue 106 for staff evaluation.

## 3.2.41 107, Generic Implications of Main Transformer Failures

Issue: As a result of main transformer faults at the North Anna Power Station, generic concerns were raised concerning the suppression of transformer fires and the impact of these fires on plant safety systems. A transformer fire of sufficient magnitude (location dependent) has the potential for degrading plant safety equipment and safety systems. The generic concerns arising from this issue involve (1) the proper maintenance, storage, and handling of transformers to prevent transformer failure and (2) the mitigation and containment of transformer fires. The staff has determined that there are four key areas in the prevention and control of transformer fires that should be addressed. These are the deluge system, drainage system, fire barriers, and firefighting and related procedures. Other generic concerns involve the layout and segregation of the transformer bay drains, use of fire barriers, and hindrances to firefighting related to access, communications, mobility, training, and procedures.

EPRI Proposal: Chapters 1, 6, and 9 of the Evolutionary Requirements Document provide design requirements for fire protection systems and include the following specific design criteria to address the safety concerns of this issue:

EPRI Evolutionary Plant SER 18.3-48

- Chapter 1 includes a general requirement that the evolutionary ALWR plant design provide for an integrated approach to fire prevention and mitigation of fire damage.
- Section 2.3.3.11 of Chapter & requires that outdoor oil-filled transformers ers have oil spill confinement features or drainage away from the buildings. Such transformers are to be located at least 50 feet from the building, or if building walls are within 50 feet of the oil-filled transformers, these walls will not have openings and will have a fireresistance rating of at least 3 hours. The outdoor oil-filled transformers are to be protected by deluge systems and drainage to accommodate the flow of oil and water as determined by the fire hazard analysis.
- Section 3.3.3.1 of Chapter 9 requires that the turbine-generator and associated areas described in NFPA 803, "Fire Protection for Light Water Nuclear Power Plants," be protected in accordance with NFPA 803.
- Section 3.3.3.3 of Chapter 9 specifies that automatic fixed water suppression over the fire area be provided for any equipment identified by the fire hazard analysis as containing a sufficient quantity of combustible material to warrant a fixed suppression system.

<u>Staff Evaluation</u>: In its DSER for Chapter 11, the staff stated that Section 2.3 of the DSER for Chapter 6 described an open issue regarding the location of oil-filled transformers in relation to exterior building walls. Section 2.3.3.11 of Chapter 6 of the Evolutionary Requirements Document provides requirements for controlling transformer fires that may damage the nearby buildings in an ALWR plant that the staff finds acceptable (see Section 2.3 of Chapter 6 of this report). Therefore, this open issue is resolved.

3.2.42 110, Equipment Protective Devices on Engineered Safety Features

<u>Issue</u>: Essential equipment has experienced a large number of failures or has been incapacitated as a result of either the failure or the intentional bypass of protective devices intended to trip active engineered safety features (ESFs) for indications of equipment faults. The affected systems exist throughout the plant and include the plant control system, the plant protection system, and the ESFs. The staff is concerned that the reliability estimates for essential equipment may not properly account for failure of the protective devices. Because of the loss of redundant devices through failures of circuits intended to be independent, there is an increased probability of common-mode failure of redundant vital services. This issue needs to be studied further to determine if failure rate estimates for essential equipment have increased and if essential equipment could be made significantly more reliable by improving the reliability of protective devices.

EPRI Proposal: The Evolutionary Requirements Document includes a number of specific design requirements to address the concerns of this issue. Chapter 5 requires that

 plant systems embody sufficient robustness of design to tolerate a conservative number of spurious or inadvertent engineered safety system actuations without the need for followup tests or inspections to verify systems' integrity or operability (Section 2.2.7)

EPRI Evolutionary Plant SER

 each division of engineered safety systems be totally independent electrically unless it is otherwise physically impracticable or less safe (Section 2.3.2)

Chapter 10 requires that:

- a robust system design, including segmentation of major functions, separation of redundant equipment within a segment, and fault-tolerant equipment, be developed to achieve high reliability and protection against the propagation of faults (Section 2.2.3)
- the man-machine interface system (M-MIS) equipment be designed and configured to readily support inservice testing by providing built-in test features, incorporating good human factors principles, and avoiding the use of undesirable features such as test jumpers or lifting of leads (Section 2.2.4)
- the M-MIS design process explicitly consider the potential for and the consequences of failures of plant and M-MIS components (Section 3.1.3.4)
- the capability for continuous on-line self-testing of hardware integrity be provided for as much of the M-MIS as is practicable (Section 3.6.1)
- on detection of a failure in the M-MIS, a system be designed so that it can be placed in such configuration that an additional single failure will not prevent system-level protection or safety action (Section 3.6.4).
- the mean time to detect and repair failures down to the lowest replaceable module, averaged across all types of M-MIS equipment for the entire design life, be less than 4 hours and the maximum time for a y module be less than 8 hours (Section 3.7.4)

<u>Staff Evaluation</u>: In its DSER for Chapter 10, the staff agreed that the implementation of the design criteria in the Evolutionary Requirements Document would resolve this issue, with the exception of the staff's concern about lack of specific guidelines for exceptions to the independence criteria for electrical systems. Since it has not prioritized Issue 110, the staff will defer further evaluation of the open issue in the DSER to its review of individual applications for FDA/DC.

3.2.43 113, Dynamic Qualification Testing of Large-Bore Hydraulic Snubbers

<u>Issue</u>: Large-bore hydraulic snubbers (LBHSs) are active mechanical devices used to restrain safety-related piping and equipment during seismic or other dynamic events (e.g., high-energy line break), yet they also allow sufficient piping and component flexibility to accommodate system expansion and contraction due to thermal transients, such as normal plant heatups and cooldowns. Dynamic qualification testing and periodic functional testing are important to verify that the LBHSs are properly designed and maintained for the life of the plant. The ALWR plant design is expected to be fully optimized with respect to the use of LBHSs; that is, it is expected that as few as possible will be used and that there will be no redundancy. EPRI Proposal: In Section 3.3.4.4.2 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI recognizes that the ALWR will not make as extensive use of LBHSs as do some current plants. Furthermore, this section states: "Section 4.2.3.5 of Chapter 6 of the ALWR Utility Requirements Document requires that test requirements for hydraulic snubbers shall be established by the plant designer for their qualification for use in the ALWR." (In the elements of resolution, EPRI only commits generally to develop dynamic tests for LBHSs.)

<u>Staff Evaluation</u>: In its DSER for Chapter 3, the staff concluded that EPRI's commitment (in its topic paper dated July 9, 1987) that dynamic test requirements for the LBHSs will be consistent with the staff's observations on dynamic qualification testing listed in the paper was adequate to consider Issue 113 resolved for the evolutionary plants. However, this specific commitment is not contained in the current version of the Evolutionary Requirements Document.

The staff's preliminary recommendations on the resolution of this issue at this time include numerous recommendations for improving dynamic qualification and testing in order to improve the reliability of LBHSs in both existing and future plants. Additionally, an adjunct issue regarding s ober single failures is also being evaluated for application to ALWRs where the use of snubbers is expected to be reduced as much as possible.

For the above reasons and since the Evolutionary Requirements Document does not provide sufficient detailed commitments regarding this the , if snubbers (including LBHSs) are used in ALWR evolutionary plant design. .he staff will evaluate the environmental (including dynamic) qualification and inservice inspection and testing requirements for conformance to NRC requirements that are current at the time of the application during its review of individual applications for FDA/DC.

Issue 113 has not yet been generically resolved by the staff and has a high safety priority. Should the final agency resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue 113 for staff evaluation.

3.2.44 118, Tendon Anchorage Failure

Issue: An inspection of a PWR prestressed concrete containment structure showed that three lower vertical tendon anchor heads were broken. The failures appeared to be caused by stress corrosion cracking. Quantities of water ranging from a few ounces to about 1.5 gallons were found in grease caps.

EPRI Proposal: Chapter 6 of the Evolutionary Requirements Document states that a freestanding steel containment for PWRs (Section 4.3.4.1) or a reinforced-concrete containment for BWRs (Section 4.3.3.2) will be used. Since a prestressed concrete containment is not specified, no specific requirements for tendon anchorage are provided.

Staff Evaluation: Since the Evolutionary Requirements Document indicates that prestressed concrete containments will not be used for evolutionary plant designs, this issue is not applicable. However, Issue 118 has not yet been

EPRI Evolutionary Plant SER 18.3-51
prioritized by the staff, and should prestressed concrete containments be used in any evolutionary plant designs, the staff will evaluate resolution of this issue, as applicable, during its review of individual applications for FDA/DC.

#### 3.2.45 120, On-Line Testability of Protection Systems

<u>Issue</u>: During its 1985 review of several plant technical specifications, the staff discovered that the design of protection systems of some plants did not provide as complete a degree of on-line (at-power) surveillance testing capability as other plants undergoing staff evaluation at that time. This raised questions about the on-line testability of protection systems and the possibility that some nuclear power plants might not provide complete testing capability. Issue 120 was established to examine these questions. Protection systems consist of the reactor protection system (RPS) and the engineered safety features actuation system (ESFAS). The main concern of Issue 120, however, is the on-line testability of the actuation subgroup (slave) relays in the ESFAS.

The requirements for at-power testability of components are included in GDC 21 of Appendix A to 10 CFR Part 50. RGs 1.22, "Periodic Testing of Protection System Actuation Functions," and 1.118, "Periodic Testing of Electric Power and Protection Systems," and IEEE 338-1977, "Criteria for the Periodic Testing of Nuclear Power Generating Station Safety Systems," provide -upplementary guidance. This guidance is intended to ensure that protection (including logic, actuation devices, and associated actuated equipment) will be designed to permit testing while a plant is operating at power without adversely affecting the plant's operation.

<u>EPRI Proposal</u>: Appendix B to Chapter 1 of the Evolutionary Requirements Document does not address Issue 120. In a letter to the NRC dated November 15, 1991, EPRI stated that Issue 120 was prioritized by the staff after the January 1, 1990, cutoff date for consideration in Appendix B to Chapter 1. EPRI further stated that Section 3.6 of Chapter 10 provides extensive requirements that appear to adequately address the technical concerns of this issue.

Chapter 10 of the Evolutionary Requirements Document addresses the testability requirements (Section 3.6) and the reactor protection and safety systems man-machine interface system (M-MIS) requirements (Section 8) as follows:

- Section 3.6.1 requires that the capability for continuous on-line selftesting of hardware integrity be provided for as much of the M-MIS as is practical. This testing should not affect the system functionality and should be performed on the module, as opposed to on-the-system basis. These tests may include, but are not limited to, RAM (random access memory) and ROM (read-only memory) failure checks, arithmetic processing unit failure checks, data-link buffer checks, and CPU (central processing unit) reset of watch-dog timers.
- Section 3.6.2 requires that the capability for periodic functional testing of the systems be provided. This periodic testing should be manually initiated, but automatically performed once initiated, and should meet the requirements of RGs 1.22 and 1.118 and IEEE 338. Automatic initiation of periodic testing may be provided where the testing does not degrade the system functionality.

- Section 3.6.7 regimes that built-in, automated test features be provided for periodic functional testing, as necessary, to eliminate physical reconfiguration of systems (e.g., adding jumpers, lifting leads, swapping cables) to perform the required tests. However, initiation of periodic, automated functional tests should meet the requirements in Section 3.6.2.
- Section 3.6.8 requires the safety-related sy' ems (e.g., reactor protection and safety systems) to have automatic t st features that are sufficient to meet the technical specification requirements for periodic surveillance of the system's functionality as defined by RGs 1.22 and 1.118 and IEEE 338.
- Section 3.6.9 requires that test features of the M-MIS be designed so that, to the degree practical, the tests can be performed with the plant at power without causing spurious actuation of reactor trip devices or safety system components. Where testing at power would upset plant operation or damage equipment, provisions should be made to test the equipment with the plant operating at reduced power or in a shutdown condition. All tests required to be performed to keep the plant at power or increase power should be capable of being performed without shutting down or reducing power.
- Section 8.2.3.4 requires that the M-MIS for the protection and safety systems normally provide for testing to be initiated at the direction of the operator, but the testing itself be largely automatic. For portions of the M-MIS of these systems that can be tested without activation of system components, the incorporation of automatic testing should be evaluated by the M-MIS designer. Where automatic testing is impractical, the M-MIS designer should provide operator aids or other features to minimize the potential for operator errors. In no case should the testing impair the capability to carry out the protection or safety function. Where practical, the system should automatically realign itself when its action is called for while it is being tested.
- Section 8.3.2.3 requires that the RPS provide for automatic self-testing of as much of the system as is practical; that is, the RPS should require initiation of testing by the operators only where automatic testing is not practical.

<u>Staff Evaluation</u>: The positions in Chapter 10 of the Evolutionary Requirements Document regarding the testability of protection systems appear, in general, to be consistent with the staff's interim position on the resolution of Issue 120. The staff, however, has a few concerns regarding the EPRI proposals.

In several places in Chapter 10 of the Evolutionary Requirements Document, EPRI uses the wording "as much as practical." Since "as much as practical" is subject to wide interpretation, more precise wording should be chosen to avoid co. ion and provide for extensive on-line testability.

On the basis of its review of Chapter 10 of the Evolutionary Requirements Document, the staff is under the impression that there is a high likelihood that the ALWR designer may choose to design the reactor protection and safety systems instrumentation and control circuits to contain sensors, all digital logic circuitry, and the combination of microprocessors and final actuation

contacts that take the place of subgroup (slave) relays (the final actuation contacts must change state in order to actuate the final actuated equipment). Should this be the case, the concern of Issue 120 regarding the on-line testability of the slave relays for the current plants is changed to the on-line testability of the microprocessors and the final actuation contacts for the ALWR.

Since the Evolutionary Requirements Document does not provide sufficient commitments, information, or design details regarding the above concerns, the staff will evaluate resolution of these concerns during its review of individual applications for FDA/DC.

See Section 3.2.39 for resolution of the open issue addressed in the staff's DSER for Chapter 5 (February 1990) for Issue 120.

Issue 120 has not yet been generically resolved by the staff and has a medium safety priority. Should the final agency resolution of this issue dictate additional actions for ALWR evolutionary plant deligns, the staff will expect the FDA/DC applicant to address resolution of Issue 120 for staff evaluation.

3.2.46 121, Hydrogen Control for Large, Dry PWR Containments

<u>Issue</u>: As a result of the TMI-2 accident, the Commission promulgated regulatory requirements on hydrogen control in 10 CFR 50.34 and 50.44. 10 CFR 50.34(f) requires a hydrogen control system based on a 100-percent fuelcladding metal-water reaction and a hydrogen concentration limit of 10 percent on uniformly distributed hydrogen in the containment or a postaccident atmosphere that will not support hydrogen combustion. Plants covered by this rule included only those whose construction percits had not been issued at the time of the TMI-2 accident. The issue is whether this requirement applies to the evolutionary ALWR.

<u>EPRI Prorosal</u>: In Section 3.5.3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI originally proposed that the evolutionary plant design have the capability to ensure that necessary accident prevention and mitigation functions can be performed during and after events in which hydrogen is produced. The Evolutionary Requirements Document originally required that the design accommodate an amount of hydrogen equivalent to that generated by oxidation of 75 percent of the fuel cladding surrounding the active fuel. This was to be accomplished for a PWR dry containment by containment volume and mixing so that the uniformly distributed hydrogen gas concentration in the containment does not exceed 13 percent under dry conditions. For a BWR, the Evolutionary Requirements Document permitted provision of a noncombustible containment atmosphere as an acceptable alternative approach.

<u>Staff Evaluation</u>: In Appendix B of its DSER for Chapter 5, the staff concluded that EPRI's proposed approach to resolving this issue was unacceptable. The staff agreed with EPRI's proposal on hydrogen control for BWRs, indicating that inerting was an acceptable approach and would meet the requirements of 10 CFR 50.44.

In SECY-90-016, the staff recommended to the Commission that the hydrogen control requirements for evolutionary plants be identical to those stated in 10 CFR 50.34(f). This regulation specifically requires a hydrogen control

system that can safely accommodate an amount of hydrogen equivalent to that generated by the reaction of 100 percent of the fuel-cladding metal and that can ensure that uniformly distributed hydrogen gas concentrations in the containment do not exceed 10 percent by volume. The Commission approved the staff's recommendation in a staff requirements memorandum dated June 26, 1990.

In a letter dated December 6, 1991, EPRI stated that it intends to fully comply with the staff's position. Revision 4 to the Evolutionary Requirements Document revised Section 3.5.3. of Appendix B to Chapter 1 of the Evolutionary Requirements Document to require that ALWR designs be capable of accommodating 100-percent oxidation of active fuel cladding and either maintaining the uniformly distributed maximum containment hydrogen gas concentration below 10 percent or ensuring that the atmosphere is incapable of supporting hydrogen combustion. This is acceptable and resolves the open issue in the DSER. However, since the staff position in SECY-90-016 is that the plant-specific design must comply with 10 CFR 50.34(f) for combustible gas control, the staff will review relevant design features for conformance to 10 CFR 50.34(f) during its review of applications for FDA/DC. See also the evaluations in Sections 2.3 and 6.5.1 of Chapter 5 of this report and the evaluation of Issue A-48 in Section 3.2.10 of this appendix.

- 3.2.47 122.1.a, Davis-Besse Loss-of-All-Feedwater Event of June 9, 1985: Short-Term Actions - Failure of Isolation Valves in Closed Position
- Issue 122.1.a was subsumed by Issue 124. See Section 3.2.52.
- 3.2.48 122.1.b, Davis-Besse Loss-of-All-Feedwater Event of June 9, 1985: Short-Term Actions - Recovery of Auxiliary Feedwater

Issue 122.1.b was subsumed by Issue 124. See Section 3.2.52.

3.2.49 122.1.c, Davis-Besse Loss-of-All-Feedwater Event of June 9, 1985: Short-Term Actions - Interruption of Auxiliary Feedwater Flow

Issue 122.1.c was subsumed by Issue 124. See Section 3.2.52.

3.2.50 122.2, Davis-Besse Loss-of-All-Feedwater Event of June 9, 1985: Short-Term Actions - Initiating Feed-and-Bleed

<u>Issue</u>: During the loss-of-feedwater event at Davis-Besse, the operators were reluctant to initiate feed-and-bleed cooling. The operators' actions raised concerns regarding the adequacy of emergency procedures, operator training, and available plant monitoring systems for determining the need to initiate feed-and-bleed cooling following the loss of the steam generator heat sink.

During the event at Davis-Besse, the steam generators became dry, meeting the criteria for the initiation of feed-and-bleed cooling. However, the operators did not initiate feed-and-bleed operations. The staff is concerned that procedures and/or training may not be adequate to ensure that operators initiate feed-and-bleed cooling when it is necessary to avert a core melt.

EPRI Proposal: Section 3 of Chapter 10 of the Evolutionary Requirements Document requires that the function and task analysis be used in the development of control room instrumentation and operating procedures and that it explicitly consider the actions of operators to control the plant.

<u>Staff Evaluation</u>: In its DSER for Chapter 10, the staff concluded that the design criteria in the Evolutionary Requirements Document were acceptable. However, it also concluded that since the Evolutionary Requirements Document does not provide the necessary details, the staff will evaluate the operator training program and emergency operating procedures during its review of individual applications for FDA/DC.

3.2.51 123, Deficiencies in the Regulations Governing Design-Basis-Accident and Single-Failure Criteria Suggested by the Davis-Besse Event of June 9, 1985

Issue 123 was mentioned in Appendix A of the staff's DSER for Chapter 3 (May 1988), but was not evaluated. In March 1992 Issue 123 was dropped as a generic issue; therefore, no evaluation is necessary.

3.2.52 124, Auxiliary Feedwater System Reliability

<u>Issue</u>: A function of the auxiliary feedwater (AFW) system in the majority of current plants is to supply water to the secondary side of the steam generators during system fill, normal plant heatup, normal plant hot standby, and normal plant cold shutdown. The AFW system also functions following loss of normal feedwater flow, including loss due to offsite power supply failure, and provides emergency feedwater (EFW) following such postulated accidents as a main feedwater line break or a main steamline break.

The loss of all feedwater at Davis-Besse resulted in an NRC investigation of the event that indicated that the potential inability to remove decay heat from the reactor core was due to the questionable reliability of the EFW system caused by any or all of the following:

- loss of all AFW as a result of the common-mode failure of the AFW pump discharge isolation valves to open
- excessive delay in recovering AFW because of difficulty in restarting AFW pump steam-driven turbines once they tripped
- interruption of AFW flow because of failures of features used to mitigate the effects of steamline and feedwater line break accidents

In addition, the investigation of the event indicated that (1) a two-train system with a steam turbine-driven AFW pump may not be able to achieve the desired level of reliability and (2) the provision to automatically isolate AFW from a steam generator affected by a main steam or feedwater line break may tend to increase the risk that adequate decay heat removal is not available rather than to decrease it.

<u>EPRI Proposal</u>: Section 5.3.1.2 of Chapter 5 of the Evolutionary Requirements Document states that the EFW system will be a dedicated safety-related system that will have no function for normal operation. The Evolutionary Requirements Document requires that

 A safety-grade supply of feedwater be provided of sufficient volume to permit safe cold shutdown, based on (1) a main feedwater line break without isolation of EFW flow to the affected steam generator for 30 minutes, (2) refill of the intact steam generators, (3) 8 hours of

EPRI Evolutionary Plant SER 18.3-56

operation at hot standby conditions, (4) subsequent cooldown of the reactor coolant system within 6 hours to conditions that permit operation of the residual heat removal system, and (5) continuous operation of one reactor coolant pump. (Section 5.3.2.3.1, Chapter 5)

- The EFW system use two noncondensing steam turbine-driven pumps and two electric motor-driven pumps. Any two pumps for four-steam-generator plants and any one pump for two-steam-generator plants must be capable of satisfying the flow requirements for licensing design-basis-accident conditions. Any single pump must be capable of satisfying the flow requirement for best-estimate evaluations of core damage frequency. (Section 5.3.3.1.2, Chapter 5)
- Each turbine-driven pump be supplied with steam only from the steam generator it supplies with feedwater. (Section 5.3.3.1.5.1, Chapter 5)
- A cavitating venturi be provided in the discharge line to each steam generator to prevent pump cavitation due to runout and also to minimize other potentially adverse effects of excessive EFW flow. (Section 5.3.3.1.8, Chapter 5)
- The rate of opening of the steam supply valves to the turbine-driven pumps be limited to the extent required to ensure reliable startup of the pumps. (Section 5.3.3.2.3, Chapter 5)
- Restart of the turbine-driven pumps following an overspeed trip be facilitated by ensuring ready access to pumps, by labeling the components required to reset the overspeed trip, and by requiring the manufacturer to provide a clear set of reset instructions to be posted adjacent to each pump. For designs that use an electronic trip that is set below the mechanical overspeed trip, reset capability from the control room must be provided. (Section 5.3.3.2.4, Chapter 5)
- Isolation and flow-regulating valves in the turbine-driven pump discharge lines be capable of performing their safety function independent of normal offsite and emergency onsite ac power. Manual capability must be provided to permit positioning of these valves in the event of a loss of power. (Section 5.3.3.4.2, Chapter 5)
- Automatic and manual initiation of EFW flow be provided. The EFW control system must have a safety-grade control system that will feed the steam generator at a maximum rate when the steam generator water level falls below the minimum steam generator water level assumed in the plant safety analysis. (Section 4.2.3.4, Chapter 3)
- The steam generator system be designed so that actuation of the EFW system is not required for at least 20 minutes following the point at which the low-level setpoint is reached. (Section 4.2.8.1, Chapter 3)
- The minimum flow delivered to the steam generators under licensing design-basis-accident conditions ensure adequate heat removal from the reactor coolant system. (Section 5.3.2.1.1, Chapter 5)

- The minimum flow delivered to the steam generators ensure adequate heat removal from the reactor coolant system for best-estimate evaluations of core damage frequency. (Section 5.3.2.1.3, Chapter 5)
- The plant designer define the mass and energy input to the containment resulting from flow of EFW to the affected steam generator following a main steamline break and ensure that this is accounted for in the containment design. Operator action to terminate EFW flow to the affected steam generator must not be assumed before 30 minutes. The system should not rely on automatic isolation of EFW to prevent containment overpressurization. (Section 5.3.2.2, Chapter 5)
- The plant designer perform an analysis of each automatic control loop of the EFW system to demonstrate capability for stable operation over the full range of operating conditions. (Section 5.3.3.5.1, Chapter 5.)

<u>Staff Evaluation</u>: In its DSER for Chapter 5 (February 1990), the staff concluded that EPRI's proposed resolution satisfactorily addressed the staff's AFW system reliability concerns. However, the staff will verify the acceptability of the analyses (discussed above) performed by the designer or applicant during its review of individual applications for FDA/DC.

3.2.53 125.11.7, Davis-Besse Loss-of-All-Feedwater Event of June 9, 1985: Long-Term Actions - Reevaluate Provision To Automatically Isolate Feedwater From Steam Generator During a Line Break

Automatic isolation of AFW from a steam generator is provided to mitigate the consequences of a steam or feedwater line break; however, it is needed only in a relatively rare event (steam or feedwater line break). In contrast, this isolation has many disadvantages. The safety significance of this issue arises from the fact that the negative aspects of automatic isolation involve accident sequences that have more frequent initiators and more significant consequences that the positive aspects. The staff concluded that removal of the AFW automatic isolation feature will neither result in a substantial safety improvement nor be cost effective and resolved the issue deciding to take no action. This does not preclude, however, an applicant from proposing the removal of the automatic isolation feature on the basis of plant-specific considerations.

See the staff's evaluation of Issue 124 in Section 3.2.52.

3.2.54 125.II.11, Davis-Besse Loss-of-All Feedwater Event of June 9, 1985: Long-Term Actions - Recovery of Main Feedwater as Alternative to Auxiliary Feedwater

This issue addressed alternative means of recovering feedwater should the AFW systems of PWRs fail. In resolving Issue 124, the staff evaluated the potential alternative recovery methods for both main and AFW systems for plants with two-train AFW systems. Contingent on implementation of the staff's recommendations proposed as the resolution of Issue 124, Issue 125.II.11 was dropped as a new and separate generic safety issue. See the staff's evaluation of Issue 124 in Section 3.2.52.

18.3-58

3.2.55 127, Maintenance and Testing of Manual Valves in Safety-Related Systems

<u>Issue</u>: Following a loss of power for the integrated control system at Rancho Seco, the operators attempted to close the auxiliary feedwater manual isolation valve. The isolation valve could not be moved, even with a valve wrench. The valve was seized because of inadequite lubrication as a result of a lack of preventive maintenance over the 10- to 12-year operational life of the plant. The staff informed licensees of this incident in IE Information Notice 86-61, "Failure of Auxiliary Feedwater Manual Isolation Valve." Because the estimated reduction in public risk would be minimal as a result of the resolution of this issue, Issue 127 was assigned a low safety priority.

<u>EPRI Proposal</u>: Section 5.6.3 of Chapter 10 of the Evolutionary Requirements Document states that the ALWR is expected to use a large amount of multiplexed data. This use will aid in plant operations, including maintenance planning and testing.

<u>Staff Evaluation</u>: In the DSER for Chapter 10, the staff agreed with EPRI that the implementation of the design criteria in the Evolutionary Requirements Document would resolve this issue. However, since a final resolution for this issue had not yet been developed, the staff concluded that any additional criteria resulting from the resolution of Issue 127 would have to be met by the ALWR designers.

Issue 127 has not yet been generically resolved by the staff and has a low safety priority. Unless the safety priority of this issue changes to medium or high, the FDA/DC applicant will not be expected to address the resolution of Issue 127.

#### 3.2.56 128, Electrical Power Reliability

<u>Issue</u>: Concerns have been raised regarding the dependence on Class 1E power, especially dc power, of the decay heat removal systems required for long-term heat removal. Failure of one division would generally result in a reactor scram, which would then require removal of decay heat. The frequency of reported failures of single dc divisions gives rise to the concern that the second dc division may not be available.

Two of the specific reasons for the concern that safety-related power may be unreliable are addressed by Issue 128. One reason is that some operating nuclear power plants do not have technical specifications or administrative controls governing operational restrictions for Class IE 120-V ac vital instrument buses and associated inverters. Without such restrictions, these power sources could be out of service indefinitely and might place certain safety systems in a situation where they could not meet the single-failure criterion. This is of particular concern during the period before the start and load of emergency diesel generators following a loss of offsite power. The other reason is that the design of some plants does not provide interlocks to prevent the inadvertent closure of the single tie breaker between Class IE buses of all voltages.

<u>EPRI Proposal</u>: Section 3.11.1.4.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that the concerns raised by this issue can be resolved by avoiding the use of bus tie breakers that could compromise

division independence and by providing a reliable dc power supply, especially when the failure of one dc power system leads to a reactor scram. Section 3.11.1.4.2 lists the following design criteria to address this issue:

- Section 2.2.8 of Chapter 5 requires that each division of the engineered safety systems requiring electric power be provided with an emergency onsite source of ac and/or dc power.
- Section 2.2.9 of Chapter 5 requires that at least two separate and independent connections be provided to offsite power sources capable of starting and running all Class IE loads required for safe shutdown.
- Section 2.3.1 of Chapter 5 requires that the specified functions of engineered safety systems be met by the use of redundant divisions.
- Section 2.3.2 of Chapter 5 requires that the divisions of engineered safety systems be totally independent and separated both mechanically and electrically, except for areas in which it is physically impractical or less safe.
- Section 3.3.1 of Chapter 5 requires the decay heat removal systems to be redundant and safety grade.
- Section 4.2.1 of Chapter 5 requires that BWR designs have three independent divisions for the core coolant inventory control (CCIC) and decay heat removal (DHR) systems.
- Section 4.2.6.1 of Chapter 5 requires that each division of the CCIC and DHR systems for BWR designs have its own independent emergency ac and dc power source.
- Section 5.1.2.1 of Chapter 5 requires that PWR designs have two independent divisions for the CCIC and DHR functions.
- Section 5.2.3.1.1 of Chapter 5 requires that the PWR designs have two independent divisions for the residual heat removal function.
- Section 2.3.6 of Chapter 11 requires that each division of engineered safety systems requiring electric power be provided with an independent emergency onsite source of ac and dc power
- Section 3.2.3 of Chapter 11 requires that at least two separate and independent connections for offsite power sources capable of starting and running all Class IE loads required for safe shutdown be included in the design.
- Sections 2.3.6, 2.3.7, and 2.3.10 of Chapter 11 require separation of electrical power systems to preclude interactions that could adversely affect the functioning of the dc power systems. Specifically, the use of bus ties between safety divisions is prohibited.
- Sections 2.3.9 and 2.3.11 of Chapter 11 require that non-safety-related loads be placed on power supplies that are completely separate from those on which safety-related loads are placed.

- Section 7.2.1 of Chapter 11 requires that the loss of any plant battery or dc bus concurrent with a single independent failure in any other system required for shutdown cooling not result in a total loss of reactor cooling capability.
- Section 7.3.2.4 of Chapter 11 requires that each reactor protection channel be normally powered from a dedicated Class 1E source that is normally independent of other dc sources.

In summary, the Evolutionary Requirements Document states that each division of the engineered safety systems will have an emergency onsite source of ac and dc power and at least two connections for offsite power, all of which will be separate and independent. Specifically, there will be three independent divisions of decay heat removal for the BWR design and two for the PWR design, each with its own emergency ac and dc power source.

<u>Staff Evaluation</u>: The design criteria EPRI proposes for the evolutionary design adequately address the design concerns of Issue 128. Additional related concerns regarding the failure of a dc bus causing an event and coincident failure of another dc bus are discussed further in Chapter 11 of this report.

However, some concerns of Issue 128 that involve operational aspects need to be addressed by implementation of appropriate procedures and controls (e.g., through technical specifications). Since the Evolutionary Requirements Document does not provide sufficient details regarding plant operations to fully resolve these concerns of Issue 128, the staff will evaluate these items (including operations, maintenance, and testing) during its review of individual applications for FDA/DC.

Issue 128 has not yet been generically resolved by the staff and has a high safety priority. Should the final resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue 128 for staff evaluation.

3.2.57 130, Essential Service Water Pump Failures at Multiplant Sites

<u>Issue</u>: This issue was identified as a result of the vulnerability of Byron Unit 1 to core-melt sequences with Unit 2 not operational. While Unit 2 was under construction, it was necessary to make a third service water pump available to Unit 1 via a crosstie with one of the two Unit 2 essential service water (ESW) pumps. This plant-specific issue raised concerns relative to multiplant units that have only two ESW pumps per plant but have crosstie capabilities. A limited survey of Westinghouse plants helped to identify the generic applicability of vulnerabilities of multiplant configurations with only two ESW pumps per plant. In the multiplant configurations identified (approximately 16 plants), all plants can share ESW pumps via a crosstie between plants. Efforts to resolve this issue are to include a survey of Babcock and Wilcox and Combustion Engineering plants to determine if similar multiplant configurations with two ESW pumps per plant and crosstie capabilities exist in these vendors' designs and a survey of single-unit plants to determine if similar ESW vulnerabilities exist. <u>EPRI Proposal</u>: Proposed elements for resolving for this issue are included in Section 3.8.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

<u>Staff Evaluation</u>: In the DSER for Chapter 8, the staff reported that Issue 130 raised concerns about multiplant units that have only two ESW pumps per plant with crosstie capabilities. Accordingly, the specific core-melt and radiological risks (consequences) determined by the evaluation of Issue 130 pertain only to the generic model multiplant configuration with two ESW pumps per plant. Sections 5.2.1 and 5.2.2 of Chapter 8 of the Evolutionary Requirements Document specify that the safety service water (SSW) system will have two pumps per division (with three independent divisions for the BWR and two independent divisions for the PWR). Section 3.8.1.4 of Appendix B to Chapter 1 states that a failure of a single pump will not prevent the SSW system from performing its intended safety-related function, and at multiplant sites, the reliability of the SSW system at the other unit or units will not i affected.

The staff indicated, however, that this design approach addressed only a portion of the reliability problems being considered under Issue 130. Both pumps are required in a given division to meet licensing-basis requirements, such as cold shutdown within 36 hours with a concurrent single failure, which defeats the redundant division. The Evolutionary Requirements Document also prohibits crossties between divisions and between units if more than one unit is placed at a site. Although elimination of the crosstie capability achieves simplicity and precludes certain operator errors, there are certain negative aspects, such as loss of flexibility during recovery actions.

The staf? stated that EPRI should further examine the reliability aspects associated with the ALWR ESW systems and propose enhancements to the designs, if warranted. It suggested that EPRI consider providing is separate and independent intake structures, as well as incorporating a cosstie capability between plants with the attendant flexibility in recovery a lions.

Regarding crosstie capability, EPRI responded that it had reviewed the SSW system configuration and its reliability and performed a probabilistic risk assessment for three different cooling water configurations (to compare relative reliabilities). From these analyses, EPRI concluded that crossties would be provided only if the plant designer demonstrated a compelling safety, operability, or availability need. Consequently, EPRI has decided not to include crossties to provide flexibility in recovery actions because the system design with crossties would increase in complexity.

Since a final resolution of Issue 130 has not yet been developed, the staff will evaluate the resolution of this issue during its review of individual applications for FDA/DC.

Issue 130 has a high safety priority.

3.2.58 132, Residual Heat Removal Pumps Inside Containment

<u>Issue</u>: Residual heat removal (RHR) pumps located inside the containment that have not been qualified for a harsh environment cannot be given credit in licensing analyses for providing long-term decay heat removal.

EPRI Evolutionary Plant SER 1B.3-62

<u>EPRI Proposal</u>: Section 5.2.3.1.4 of Chapter 5 of the Evolutionary Requirements Document specifies that all RHR pumps should be located outside the containment.

<u>Staff Evaluation</u>: In its DSER for Chapter 5 (February 1990), the staff concluded that the requirement that all RHR pumps be located outside the containment was a satisfactory resolution of this issue. However, Issue 132 has not yet been prioritized by the staff and a generic resolution has not been developed. The final resolution may dictate additional actions for ALWR evolutionary plant designs. Therefore, the staff will evaluate the resolution of this issue, as necessary, during its review of individual applications for FDA/DC.

3.2.59 135, Steam Generator and Steamline Overfill

<u>Issue</u>: This issue was initiated in 1986 to integrate all current generic issue activities on steam generator and steamline integrity. These activities started as a result of several steam generator tube rupture (SGTR) events that occurred in operating PWRs. At least one of the events led to overfilling of the secondary side of the steam generator. This overfilling resulted in water in the steamline, which in turn caused safety valves designed for steam service to pass water and resulted in unanalyzed loads on the steam piping.

The 14 subissues of Issue 67 regarding staff actions that are integrated under this issue are as follows:

(1)Improved A.cident Monitoring (Issue 67.3.3) (2) Reactor Vessel Inventory Measurement (Issue 67.3.4) (3) Reactor Coolant Pump Trip (Issue 67.4.1) Control Room Design Review (Issue 67.4.2) (4)(5) Emergency Operating Procedures (Issue 67.4.3) (6) Organizational Responses (Issue 67.6.0) (7)Reactor Coolant System Pressure Control (Issue 67.9.0) (8) Steam Generator Overfill (Issue 67.3.1) (9) Reassessment of Radiological Consequences (Issue 67.5.1) (10)Reevaluation of SGTR Design Basis (Issue 67.5.2) (11)Improved Eddy Current Tests (Issue 67.7.0) (11) Supplemental Tube Inspections (Issue 67.10.0) (13) Integrity of Steam Generator Tube Sleeves (Issue 67.2.1) (14)Denting Criteria (Issue 67.8.0)

Studies concentrating on these 14 subissues have provided a better understanding of steam generator and secondary system integrity, including water hammer and its effect on secondary system components and branch lines as well as radiological consequences. The conclusion resulting from the studies made under this issue was that Subissues 1-8 are resolved, Subissues 9-12 are being pursued independently, and Subissues 13 and 14 are of little safety significance and have been designated as regulatory impact issues.

The basis for the resolution of Subissues 1-7 was that the concerns were addressed in the Three Mile Island Action Plan and implemented by multiplant action (MPA) letters. For Subissue 8, all aspects of steam generator overfill (SGOF), except SGOF by SGTR, were considered under Issue A-47 (former unresolved safety issue). The staff investigated the risk associated with SGOF by SGTR and found it to be a small contributor to core damage frequency. Therefore, this subissue was closed without any new requirements.

<u>EPRI Proposal</u>: Section 3.3.5.5 of Appendix B to Chapter 1 of the Evolutionary Requirements Document (Revision 1) states that "total resolution of this issue is not applicable to design and thus is not applicable to the ALWR Utility Requirements Document." However, the Evolutionary Requirements Document includes requirements pertaining to the following:

- clear access in the vicinity of the primary side manways (Section 4.4.1.4, Chapter 3)
- space and arrangements inside the steam generator primary channel head to facilitate inspection and repair (Section 4.6.2, Chapter 3)
- water level indicator of reactor vessel inventory during normal operation and postaccident conditions (Section 4.5, Chapter 3)
- automatic control of steam generator water level (Section 4.2.3.2.1, Chapter 3)
- design to prevent secondary-side safety valve actuation for an SGTR (Section 4.2.5, Chapter 3)
- steam generator materials (Section 4.4.1.1, Chapter 3)
- tube bundle arrangement (Section 4.4.1.2.1, Chapter 3)
- prevention of water hammer in the steam generators (Section 4.4.1.8, Chapter 3)
- minimizing emergency feedwater flow (Section 5.3.3.1.8, Chapter 5)
- a safety depression ization and vent system to depressurize the reactor coolant system during an SGTR event (Section 5.5, Chapter 5)

The Evolutionary Requirements Document also states that "this issue has several elements, many of which are not yet defined or even assessed.... [I]t is not clear that the requirements directly address this issue."

<u>Staff Evaluation</u>: The staff agrees with the inclusion of the requirements listed above and the statement in the Evolutionary Requirements Document regarding the not yet defined and assessed elements of this issue. However, it is not clear why EPRI states that total resolution of this issue is not applicable to the evolutionary design and what is EPRI's position regarding the actions required by the MPA letters. Consequently, the staff will evaluate the resolution of Issue 135 during its review of individual applications for FDA/DC using the appropriate requirements and guidance current at that time.

### 3.2.60 142, Leakage Through Electrical Isolators in Instrumentation Circuits

Issue: This issue addresses electrical isolators used to maintain electrical separation between safety-related and non-safety-related electrical systems in nuclear power plants, thus preventing malfunctions in the mon-safety-related systems from degrading the performance of safety-related circuits. The primary concern is that the amount of energy that could pass through certain types of isolation devices (and be transmitted to safety-related circuitry) during certain electrical transients might damage or seriously degrade the performance of Class 1E components or cause the isolation devices to give a false output, or the electrically generated noise on the circuit might cause the isolation device to give a false output.

EPPI Propusal: Not provided in Section 3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

Statf Evaluation: Section 3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that it specifically addresses compliance with high- and medium-priority and nearly resolved generic safety issues included as of January 1990 in NRC's generic issue management control system (GIMCS). As of December 31, 1989, although Issue 142 was identified in Appendix B of NUREG-0933 as being applicable to future plants, it had not yet been prioritized.

As of the first-quarter fiscal year 1992 update of the GIMCS report, Issue 142 had a medium safety priority. Since this issue is not addressed in the Evolutionary Requirements Document, the staff will evaluate its resolution during its review of individual applications for FDA/DC.

3.2.61 143, Availability of Chilled Water Systems and Room Cooling

Issue: in recent years, several nuclear power plants have experienced problems with safety system components and control systems that were caused by a partial or total loss of heating, ventilating, and air conditioning (HVAC) systems. Many of these problems exist because of the desire to provide increased fire protection and the need to avoid severe temperature changes in equipment control circuits. Since the Browns Ferry fire, considerable effort has been expended to improve the fire protection of equiment required for safe shutdown. Generally, this improvement has been made by enclosing the affected equipment in small, isolated rooms. The result has been a significant increase in the impact of the loss of room cooling. Plant control and safety have improved with the introduction of electronic integrated circuits; however, these circuits are more susceptible to damage from severe changes in temperature caused by the loss of room cooling.

It is believed that failures of air cooling systems for areas housing key components, such as residual heat removal pumps, switchgear, and diesel generators, could contribute significantly to core-melt probability in certain plants. Because corrective measures are often taken at the affected plants once such failures occur, the impact of these failures on the proper functioning of air cooling systems has not been considered and plants with similar, inherent deficiencies may not be aware of these problems.

EPRI Proposal: Not provided in Section 3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

EPRI Evolutionary Plant SER 18.3-65

<u>Staff Evaluation</u>: Section 3.1 of Appendix B to Chapter 1 of the Evolutionary Requirement: Document states that it specifically addresses compliance with high- and medium-priority and nearly resolved generic safety issues included as of January 1990 in NRC's GIMCS. As of December 31, 1989, although Issue 143 was identified in Appendix B of NUREG-0933 as being applicable to future plants, it had not yet been prioritized.

As of the first-quarter fiscal year 1992 update of the GIMCS report, Issue 143 had a high safety priority. Since this issue is not addressed in the Evolutionary Requirements Document, the staff will evaluate its resolution during its review of individual applications for FDA/DC.

3.2.62 151, Reliability of Recirculation Pump Trip During an Anticipated Transient Without Scram

<u>Issue</u>: Issue summary was not published in NUREG-0933 when this appendix was prepared.

EPRI Proposal: Not provided in Section 3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

<u>Staff Evaluation</u>: Section 3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that it specifically addresses compliance with high- and medium-priority and nearly resolved generic safety issues included as of January 1990 in NRC's GIMCS. As of December 31, 1989, although Issue 151 was identified in Appendix B of NUREG-0933 as being applicable to future plants, it had not yet been prioritized.

As of the first-quarter fiscal year 1992 update of the GIMCS report, Issue 151 had a medium safety priority. Since this issue is not addressed in the Evolutionary Requirements Document, the staff will evaluate its resolution during its review of individual applications for FDA/DC.

3.2.63 153, Loss of Essential Service Water in Light-Water Reactors

<u>Issue</u>: This issue addresses the potential unavailability of the essential service water (ESW) system for all LWRs except those seven multiplant sites addressed under Issue 130. The ESW system at a nuclear power plant supplies cooling water to transfer heat from various safety-related and non-safetyrelated systems and equipment to the ultimate heat sink of the plant. It is known by different names at various types of plants. The design and operational characteristics of the ESW system are different for PWRs and BWRs. In addition, these characteristics may differ significantly in each of these reactor types.

Under Issue 153, the staff will examine all potential causes for ESW system unavailability, except those that are considered to be resolved by implementing the resolutions addressed in GL 89-13, such as biofouling, sediment, corrosion, and erosion (Issue 51). The safety concerns of this issue include partial or complete loss of ESW system functions resulting from common causes (such as icing of the intake structure), degradation of the ESW system, design deficiencies, and procedural or maintenance errors. A complete loss of the ESW system could lead to a core-melt accident, posing a significant risk to the public. EPRI Proposal: Not provided in Section 3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document.

The design features for the safety service water (SSW) system provided in Section 5.2 of Chapter 8 of the Evolutionary Requirements Document are summarized as follows:

- Performance Requirements
  - The SSW system will be designed to meet the required heat loads.
  - The SSW system will be provided with two pumps and two heat exchangeers per division.
  - The plant designer will provide analyses for all potential operating conditions that properly account for uncertainties.
- System Arrangement
  - The SSW system will be divided into approximately equal-sized divisions, two for the PWR and three for the BWR.
  - A division will be made up of independent piping systems, each with pumps, heat exchangers, strainers, controls and instrumentation, power supplies, and associated equipment required for regulating system flow.

<u>Staff Evaluation</u>: Section 3.1 of Appendix B to Chapter 1 of the Evolutionary Requirements Document states that it specifically addresses compliance with high- and medium-priority and nearly resolved generic safety issues included as of January 1990 in NRC's GIMCS. Issue 153 was identified after December 31, 1989, and, therefore, was not included in Appendix B to NUREG-0933 until the June 30, 1990, update as being applicable to future plants; it was assigned a medium safety priority in February 1991.

The performance requirements and system arrangement for the SSW system indicated above do not adequately address the safety concerns of Issue 153. These concerns include partial or complete loss of ESW system functions resulting from common causes, degradation of the SSW system, design deficiencies, and procedural or maintenance errors. The Evolutionary Requirements Document should provide an assessment of these potential failure modes and their associated contributions to the core damage frequency and should identify dominant accident sequences. The Evolutionary Requirements Document states that the plant designer will be responsible for addressing these concerns for the future plant-specific design.

Since the Evolutionary Requirements Document does not provide sufficient information, the staff will evaluate resolution of Issue 153 during its review of individual applications for FDA/DC.

3.2.64 HF 4.4, Procedures - Guidelines for Upgrading Other Procedures

<u>Issue</u>: This issue addresses whether the staff needs to develop technical guidance for industry use in upgrading both normal and abnormal operating

EPRI Evolutionary Plant SER 18.3-67

procedures. Future work in this area includes the development of guidelines for the review of programs to upgrade procedures at nuclear power plants.

<u>EPRI Proposal</u>: In the note to Table B.3-1 in Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that Issue HF 4.4 was determined, through the NUREG-1197 screening process, to be not technically relevant to the ALWR design. Nevertheless, the Evolutionary Requirements Document addresses this issue in Section 3.10.3 of Appendix B to Chapter 1.

<u>Staff Evaluation</u>: Appendix A to NUREG-1197 identifies issues that the staff determined were not applicable to the ALWR Program according to six categories. Issue HF 4.4 is listed as a Code (or Category) 5 item, that is, an issue not applicable to plant design, such as plant operation and operating procedures, management of operations, accident management and emergency preparedness, operator training and qualifications, inspection and maintenance, and operating experience reporting. By using the definition of Appendix A to NUREG-1197 for nonapplicability of generic safety issues to the ALWR Program, Issue HF 4.4 would be categorized as not applicable because it addresses procedures.

Despite this information, the Evolutionary Requirements Document addresses Issue HF 4.4 (in apparent conflict with the note to Table B.3-1 in Appendix B to Chapter 1 discussed above) stating in Section 3.10.3.4.3 of Appendix B to Chapter 1 that plant operating procedures will be an integral part of the plant design. As a result, the staff has evaluated the proposed elements for resolving this issue.

In its DSER for Chapter 10, the forf concluded that the information in the Evolutionary Requirements Docume as not sufficient to determine if the requirements for the design, detopment, and validation of plant procedures were acceptable. The staff stated that although the electronic display of procedures might enhance information display flexibility, the limitations and constraints associated with this technology, as well as the operability, maintainability, and reliability of this technology, should be fully evaluated in the context of the entire control room and other control station designs before committing to such an approach. EPRI also should evaluate issues concerning the use of mixed types of procedures from one control station to the next, as well as the requirement for the active simulator to be used for the validation of procedures. Since the Evolutionary Requirements Document still does not provide sufficient information regarding these concerns, the staff will evaluate the resolution of Issue HF 4.4 during its review of individual applications for FDA/DC.

Issue HF 4.4 has not yet been generically resolved by the staff and has a high safety priority. Should the final agency resolution of this issue dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issue HF 4.4 for staff evaluation.

3.2.65 HF 5.1, Man-Machine Interface - Local Control Stations, and

3.2.66 HF 5.2, Man-Machine Interface - Review Criteria for Human Factors Aspects of Advanced Controls and Instrumentation <u>Issue</u>: To ensure that the man-machine interface (M-MI) is adequate for the safe operation and maintenance of nuclear power plants, the staff concluded that the following should be developed: (1) human factors engineering guidelines for correcting M-MI problems and (2) regulatory guidance for integrating human factors engineering into new designs and into advanced technological improvements incorporated into existing designs.

Issue HF 5.1 addresses the expansion of efforts to provide guidance for the development of local control stations and auxiliary operator interfaces. Regulatory efforts on M-MIS have been limited to the control room and the remote shutdown panel. Additional guidance on improvements of existing annunciator systems may also result from these efforts. The staff expects to perform job and task analyses of control room crew activities to identify and describe communication and control links between the control room and the auxiliary control stations. In addition, the staff will analyze the functions of the auxiliary personnel from the task analyses to estimate the potential impact of errors by auxiliary personnel on the safety of the plant.

Issue HF 5.2 addresses the development of guidelines on the use and evaluation of improved annunciator systems that use advanced technologies. The staff is concerned that the human engineering guidelines used in current nuclear power plant control rooms, although adequate for current designs, may not be sufficient for the advanced and developing technologies that will be introduced into existing and future designs.

Solutions for these issues are expected to result in modifications of regulatory guides, the Standard Review Plan, and other guidance.

<u>EPRI Proposal</u>: In Section 3.10.3 of Appendix B to Chapter 1 of the Evolutionary Requirements Document, EPRI states that new and improved designs and systems, such as annunciator systems, computer systems, and other operational aids, will be required in a design that meets the design criteria of the Evolutionary Requirements Document to improve the M-MI. EPRI also states that a number of design features are required in the Evolutionary Requirements Document that address these human factors issues, including the following:

- Section 8.2.1.2 of Chapter 1 requires that human factors design principles be consistently applied throughout the design process for each operation workspace to reduce operation errors during all plant modes.
- Section 11.1.3.1 of Chapter 1 requires that the design process emphasize the human-machine interfaces and that an or bing analysis be conducted to ensure that these requirements will be met.
- Section 11.10.4 of Chapter 1 requires that the designer provide a plant simulator/performance model that can be used to study the human engineering aspects of the plant controls and control room design.
- Section 11.11 of Chapter 1 requires that the interdisciplinary design review group include one member knowledgeable about the principles of human factors.
- Section 2.1.1 of Chapter 10 requires that operator limitations not be challenged.

- Section 2.2.10 of Chapter 10 requires the application of advanced technology to the main control room to enhance the design.
- Section 4 of Chapter 10 requires the application of human factors engineering principles to all control stations.
- Section 4.3 of Chapter 10 requires annunciator a arms to be addressed.
- Sections 3.4.5 and 5.6 of Chapter 10 require operator aids to be addressed.
- Section 3 of Chapter 10 requires the specific identification of functions and tasks and their allocation among the operators and automatic systems.
- Section 4.1.5.2 of Chapter 10 requires that the M-MI system designer develop and verify human factors practices for advanced M-MI technology where there is limited published guidance.

EPRI considers that the implementation of these requirements will satisfy NRC concerns about local control stations and the re of new, advanced technology in the design of ALWRs.

#### Staff Evaluation:

(1) Local Control Stations

In the DSER for Chapter 10, the staff determined that the commitment in the Evolutionary Requirements Document to design local control stations (LCSs) to the same standards as control stations in the main control room was acceptable. However, it concluded that EPRI should

- Provide a clear definition of LCSs in the Evolutionary Requirements Document. Issue HF 5.1 currently includes single-component interfaces, such as manually operated valves, in its definition.
- Address the issue of functional centralization in the Evolutionary Requirements Document, since a key element in the handling of safety functions is the degree of centralization that exists, which is reflected in the number of different LCSs required to carry out safety functions. As the number of LCSs increases, so does the time required to execute procedures and the workload associated with crew coordination, communication, and verification of operating procedures. Each introduces potential sources of human error that can be expected to increase with the number of panels involved.

Since the Evolutionary Requirements Document still does not provide sufficient information regarding these concerns, the staff will evaluate the resolution of Issue HF 5.1 during its review of individual applications for FDA/DC.

(2) Annunciators

In the DSER for Chapter 10, the staff determined that the Evolutionary Requirements Document integrates human factors engineering for the alarm system into the design of the hardware and software and that it requires that certain human factors tests and evaluations of the alarm system be part of the overall test and evaluation program. However, EPRI should require additional human factors test and evaluation activities, including the development of

- a human factors verification and validation test plan
- a method to document test activities to provide traceability and ensure that all human factors requirements are addressed during tests and evaluations
- quantitative measures to assess human-system performance

Since the Evolutionary Requirements Document still does not provide sufficient information regarding these concerns, the staff will evaluate the resolution of Issue HF 5.2 during its review of individual applications for FDA/DC.

Issues HF 5.1 and HF 5.2 have not yet been generically resolved by the staff and both have a high safety priority. Should the final resolution of these issues dictate additional actions for ALWR evolutionary plant designs, the staff will expect the FDA/DC applicant to address the resolution of Issues HF 5.1 and HF 5.2 for staff evaluation.

Table 38	.1 Gener	ic Safety	Issues	Addressed	in the	DSERS
2 491 407 3 407 107 44				and the second second second second second		

Item/ Issue	Title	DSER Chap- ter	DSER Status
11.E 3	Containment Design - Integrity Check	5	Open issue (improved capabil- ity to detect inadvertent containment bypass)
A-29	Nuclear Power Plant Design for the Reduction of Vulnerability to Industrial Sabotage	9	Open issue (insider sabotage)
A-30	Adequacy of Safety-Related DC Power Supplies	11	Integrated into Issue 128
A-44	Station Blackout [former unresolved safety issue (USI)]	5	Resolved for Evolutionary Requirements Document
A-45	Shutdown Decav Heat Removal Requirements (former USI)	5	Resolved for Evolutionary Requirements Document
A-46	Seismic Qualification of Equipment in Operating Plants (former USI)	6	Open issue (emphasis on tank anchorage and installation in the Evolutionary Requirements Document)
A-47	Safety Implications of Control Systems (former USI)	10	Confirmatory issue (commit- ment to Generic Letter 89-19 in Evolutionary Require- ments Document); vendor item (acceptability of design details); see also Issue 76
A-48	Hydrogen Control Measures and and Effects of Hydrogen Burns on Safety Equipment	5	Open issue (zirconium-water reaction and hydrogen generation limits and control); see also Issue 121
B-17	Criteria for Safety-Related Operator Actions	10	Open issue (justification for not complying with proposed revisions to ANSI/ANS 58.8)
B-22	LWR Fuel	4	Resolved for Evolutionary Requirements Document
B-29	Effectiveness of Ultimate Heat Sinks	8	Resolved for Evolutionary Requirements Document
B-32	Ice Effects on Safe*Related Water Supplies	8	Not evaluated: see also Issues 51 and 130

EPRI Evolutionary Plant SER 18.3-72

۵

# Table 3B.1 (Continued)

÷

•

0

Item/ Issue	Title	DSER Chap- ter	DSER Status
C-8	Main Steamline Isolation Valve Leakage Control Systems	3	Open issue (pending final NRC resolution)
D-?	Emergency Core Cooling System Capability for Future Plants	5	Resolved for Evolutionary Requirements Document on the basis of commitment to satisfy the requirements of the Commission's Severe Accident Policy Statement
2	Failure of Protective Devices on Essential Equipment	10	Open issue (specific guide- lines on exceptions to independence criteria for electrical systems); see also Issue 110
23	Reactor Coolant Pump Seal Failures	3, 8	Open issue (independent reactor coolant pump seal cooling during station blackout)
29	Bolting Degradation or Failure in Nuclear Power Plants	3	Open issue
51	Improving the Reliability of Open-Cycle Service Water Systems	8	Open issue (guidance in Generic Letter 39-13 and Information Notice 89-16); see also Issues B-32 and 130
67.7.0	Improvec Eddy Current Tests	3	Resolved for Evolutionary Requirements Document
70	Power-Operated Relief Valve and Block Valve Reliability	5	Open issue (specification of criterion for depressuriza- tion rate of safety depres- surization and vent system)
75	Generic Implications of ATWS (Anticipated Transient Without Scram) Events at the Salem Nuclear Plant	10	Auguirements Document except for: Open issue (justification of 14-day design Objective for corrective maintenance of man-machine interface system equipment);

Item/ Issue	Title	DSER Chap- ter	DSER Status
			Vendor items (post-trip roview program and proce- dures; equipment classifica- tion and vendor interface for reactor trip system compo- nents and all preventative maintenance and safety- related equipment; details of surveillance program for reactor trip breakers)
76	Instrumentation and Control Power Interactions	10	Confirmatory issue (commit- ment to Generic Letter 89-19 in Evolutionary Requirements Document); Vendor item (acceptability of design details); see also Issue A-47
79	Unanalyzed Reactor Vessel Thermal Stress During Natural Convection Cooldown	4	Resolved for Evolutionary Requirements Document
82	Beyond-Design-Basis Accidents in Spent Fuel Pools	7	Open issue (low-density storage racks for most recently discharged fuel)
84	Combustion Engineering Power- Operated Relief Valves	5	Open issue (specification of criterica for depressuriza- tion rate of safety depres- surization and vent system)
91	Main Crankshaft Failures in Transamerica Delaval Emergency Diesel Generators	11	Resolved for Evolutionary Requirements Document
93	Steam Binding of Auxiliary Feedwater Pumps	5	Resolved for Evolutionary Requirements Document
94	Additional Low-Temperature Overpressure Protection for Light-Water Reactors	3, 4	Open issue (pending final NRC resolution)
96	Residual Heat Removal Suction Valve Testing	5	Open issue (pending NRC's completion of intersystem

Table 3B.1 (Conti "ed)

Item/		DSER Chap-	
Issue	Title	ter	DSER Status
			resolution program for BWRs and PWRs); see also Issues 99, 105, and 120
99	Reactor Coolant System/Residual Heat Removal Suction Line Valve Interlock on PWRs	5	Open issue (pending NRC's completion of intersystem loss-of-coolant accident resolution program for BWRs and PWRs); see also Issues 96, 105, and 120
101	BWR Water Level Redundancy	10	Confirmatory issue (commit- ment to comply with Generic Letter 89-11); vendor item (review of specific designs for water level instrumentation)
103	Design for Probable Maximum Precipitation	6, 8	Open issue (reference to SRP Section 2.4.10 and Generic Letter 89-22 in Evolutionary Requirements Document)
105	Interfacing Systems Loss-of- Coolant Accident at LWRs	5	Open issue (pending NRC's completion of intersystem loss-of coolant accident resolution program for BWRs and PWRs); see also Issues 96, 99, and 120
107	Main Transformer Failures	11	Open issue (pending resolu- tion of open issue on trans- former location in Chapter 6)
110	Equipment Protective Devices on Engineered Safety Features	10	Open issue (specific guide- lines on exceptions to independence criteria for electrical systems); see also Issue 2
113	Dynamic Qualification Testing of Large-Bore Hydraulic Snubber	3 S	Resolved for Evolutionary Requirements Document
115	Enhancement of the Reliability of Westinghouse Solid State Protection System	10	Resolved for Evolutionary Requirements Document

Table 3B.1 (Continued)

Tabl	0 28	1 3	1Conti	nund)
1001	6 98	8 - 8	Court	inuca)

1

Item/ Issue	Title	DSER Chap- ter	DSER Status
116	Accident Management for Future Plants	10	Issue subsumed into staff's efforts on accident manage- ment; to be handled under open issue in Chapter 5 on severe accident management program
117	Allowable Time for Diverse Simultanecus Equipment Outages	5	Resolved for Evolutionary Requirements Document
118	Tendon Anchorage Failure	6	Not applicable
120	On-Line Testability of Protection Systems	5	Open issue (pending NRC's completion of intersystem loss-of-coolant accident resolution program for BWRs and PWRs); see also Issues 96, 99, and 105
121	Hydrogen Control for Large, Dry PWR Containments	5	Open issue (zirconium-water reaction and hydrogen gen- eration limits and control); see also Issue A-48
122.1.a	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985: Short-Term Actions - Failure of Isolation Valves in Closed Position	5	Resolved for Evolutionary Requirements Document; Vendor item (acceptability of designers' analyses); see also Issues 122.1.b, 122.1.c, 124, 125.II.7, and 125.II.11
122.1.b	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Short-Term Actions - Recovery o Auxiliary Feedwater	5 f	Resolved for Evolutionary Requirements Document; Vendor item (acceptability of designers' analyses); see also Issues 122.1.a, 122.1.c 124, 125.II.7, and 125.II.11
122.1.c	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Short-ferm Actions - Interrupti of Auxiliary Feedwater Flow	5 : on	Resolved for Evolutionary Requirements Document; Vendor item (acceptability or designer's analyses); see also Issues 122.1.a, 122.1.b 124, 125.11.7, and 125.11.11

Ver

Item/ Issue	Title	DSER Chap- ter	DSER Status
122.2	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Short-Term Actions - Initiating Feed-and-Bleed	10	Resolved for Evolutionary Requirements Document; Vendor item (operator training program)
123	Deficiencies in the Regulations Governing Design-Basis-Accident and Single-Failure Criteria Suggested by the Davis-Besse Event of June 9, 1985	3	Reassigned to Chaptor 1 of Evolutionary Requirements Document; to be addressed in the final safety evaluation report
124	Auxiliary Feedwater System Reliability	5	Resolved for Evolutionary Requirements Document; Vendor item (acceptability of designers' analyses); see also Issues 122.1.a, 122.1.b, 122.1.c, 125.11.7, and 125.11.11
125.1.3	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Long-Term Actions - Safety Parameter Display System Availability	10	Resolved for Evolutionary Requirements Document
125.I.4	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Long-Term Actions - Plant- Specific Simulator	10	Resolved for Evolutionary Requirements Document
125.1.5	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Long-Term Actions - Safety Systems Tested in All Conditions Required by Design-Basis Accident	10	Resolved for Evolutionary Requirements Document
125.1.6	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985 Long-Term Actions - Valve Torque Limit and Bypass Switch Setting:	10	Dropped as a generic issue; no evaluation needed

Table 3B.1 (Continued)

# Table 38.1 (Continued)

L. HERE SER	70±°s	DSER Chap- ter	DSER Status
125.117	Davis sse Loss-of-All- See erm Actions - Reevaluate Provision To Automatically Isolate Feedwater From Steam Generator During a Line Break	5	Resolved for Evolutionary Requirements Document; Vendor item (acceptability of designers' analyses); see also Issues 122.1.a, 122.1.b, 122.1.c, 124, and 125.II.11
125.11.11	Davis-Besse Loss-of-All- Feedwater Event of June 9, 1985: Long-Term Actions - Recovery of Main Feedwater as Alternative to Auxiliary Feedwater	5	Resolved for Evolutionary Requirements Document; Vendor item (acceptability of designers' analyses); see also Issues 122.1.a, 122.1.b, 122.1.c, 124, and 125.11.7
126	Reliability of PWR Main Steam Safety Valves	3	Resolved for Evolutionary Requirements Document
127	Maintenance and Testing of Manual Valves in Safety-Related Systems	10	Resolved for the Evolutionary Requirements Document
128	Electrical Power Reliability	11	Open issue (listing of operational, maintenance, and testing issues of Issue 128 in Evolutionary Requirements Document)
130	Essential Service Water Pump Failures at Multiplant Sites	8	Open issue (needs to be addressed); see also Issues B-32 and 51
132	Residual Heat Removal Pumps Inside Containment	5	Resolved for Evolutionary Requirements Document
HF 4.4	Procedures - Guidelines for Upgrading Other Procedures	10	<b>Open issue</b> (electronic display of procedures and mixed types of procedures)
IF 5.1	Man-Machine Interface - Local Control Stations	10	Open issue (definition of local control stations and centralization); see also Issue HF 5.2
HF 5.2	Man-Machine Interface - Review Criteria for Human Factors Aspects of Advanced Controls and Instrumentation	10	<b>Open issue</b> (definition of local control stations and centralization); see also Issue HF 5.1

EPRI Evolutionary Plant SER 1B.3-78

Issue	Title		
2*	Failure of Protective Devices on Essential Equipment		
76*	Instrumentation and Control Power Interactions		
78	Monitoring of Fatigue Transient Limits for Reactor Coolant System		
89	Stiff Pipe Clamps		
110*	Equipment Protective Devices on Engineered Safety Features		
118*	Tendon Anchorage Failure		
132*	Residual Heat Removal Pumps Inside Containment		
144	Scram Without a Turbine-Generator Trip		
145	amprove Surveillance and Startup Testing Programs		
146	Support Flexibility of Equipment and Components		
147	Fire-Induced Alternate Shutdown Control Room Panel Interactions		
148	Smoke Control and Manual Firefighting Effectiveness		
149	Adequacy of Fire Barriers		
152	Design Basis for Valve That Might Be Subjected to Significant Blowdown Loads		
154	Adequacy of Emergency and Essential Lighting		
155.1	Generic Concerns Arising From TMI-2 Cleanup - More Realistic Source Term Assumptions		
155.2	Generic Concerns Arising From TMI-2 Cleanup - Establish Licensing Requirements for Nonoperating Facilities		
155.3	Generic Concerns Arising From TMI-2 Cleanup - Improve Design Re- quirements for Nuclear Facilities		
155.4	Generic Concerns Arising From TMI-2 Cleanup - Improve Criticality Calculations		
155.5	Generic Concerns Arising From TMI-2 Cleanup - More Realistic Severe Reactor Accident Scenario		
155.6	Generic Concerns Arising From TMI-2 Cleanup - Improve Decontamina- tion Regulations		
155.7	Generic Concerns Arising From TMI-2 Cleanup - Improve Decommission- ing Regulations		
156.1.1	Systematic Evaluation Program Issues - Settlement of Foundations and Buried Equipment		

#### Table 3B.2 Safety Issues Applicable to Future Plants That Have Not Been Prioritized

\*Although not prioritized, the evolutionary DSER chapters addressed Issues 2, 76, 110 318, and 132. See Table 3B.1.

EPRI Evolutionary Plant SER 18.3-79

7

1

# Table 3B.2 (Continued)

0

lssue	Title
156.1.2	Systematic Evaluation Program Issues - Dam Integrity and Site Flooding
156.1.3	Systematic Evaluation Program Issues - Site Hydrology and Ability To Withstand Floods
156.1.4	Systematic Evaluation Program Issues - Industrial Hazards
156.1.5	Systematic Evaluation Program Issues - Tornado Missile
156 2.1	Systematic Evaluation Program Issues - Severe Weather Effects on Structures
156.2.2	Systematic Evaluation Program Issues - Design Codes, Criteria, and Load Combinations
156.2.3	Systematic Evaluation Program Issues - Containment Design and Inspection
156.2.4	Systematic Evaluation Program Issues - Seismic Design of Struc- tures, Systems, and Components
156.3.1.1	Systematic Evaluation Program Issues - Shutdown Systems
156.3.1.2	Systematic Evaluation Program Issues - Electrical Instrumentation and Controls
156.3.2	Systematic Evaluation Program Issues - Service and Cooling Water Systems
156.3.3	Systematic Evaluation Program Issues - Ventilation Systems
156.3.6.1	Systematic Evaluation Program Issues - Emergency AC Power
156.3.6.2	Systematic Evaluation Program Issues - Emergency DC Power
156.3.8	Systematic Evaluation Program Issues - Shared Systems
156.4.2	Systematic Evaluation Program Issues - Testing of the Reactor Protection System and Engineered Safety Features
156.6.1	Systematic Evaluation Program Issues - Pipe Break Effects on Systems and Components

EPRI Evolutionary Plant SER 18.3-80

t

.

### Table 3B.3 Newly Identified Safety Issues

lssue	Title
157	Containment Performance Improvement Program
158	Performance of Power-Operated Valves Under Design-Basis Conditions
159	Qualification of Safety-Related Pumps While Running on Minimum Flow
160	Spurious Actions of Instrumentation Upon Restoration of Power
161	Use of Non-Safety-Related Power Supplies in Safety-Related Circuits
162	Inadequate Technical Specifications for Shared Systems at Multi- plant Sites When One Unit Is Shut Down

#### **4 REGULATORY DEPARTURE ANALYSES**

In the staff requirements memorandum (SRM) dated August 24, 1989, the Commission instructed the staff to provide an analysis detailing where the staff proposes departure from current regulations or where it is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs. The staff considers these to be policy issues. In this section, the staff discusses the regulatory departure analyses of those issues identified for the evolutionary plant designs.

During its review of the EPRI Requirements Document, the staff identified a number of issues significant to reactor safety as it considered operating experience, probabilistic risk assessment studies, and evaluations of the evolutionary and passive ALWR designs. In Commission papers SECY-90-016 and SECY-91-078, the staff proposed resolutions for some of these issues for evolutionary designs. In addition, the staff developed draft Commission papers, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," and "Design Certification and Licensing Policy Issues Pertaining to Passive and Evolutionary Advanced Light Water Reactor Designs," that were issued on February 27 and July 6, 1992, respectively. These draft Commission papers address the status of those issues identified previously as well as new issues that pertain to both the evolutionary and passive LWR designs. The staff considers the discussions in these papers to be its regulatory dc, ture analyses of the policy issues identified to date for the evolutionary plant designs.

The staff also forwarded other policy papers to the Commission that address the status of the staff's review of other issues related to the reviews of the ALWR. A list of these papers is provided in Appendix E to Volume 1 of this report.

Some of the issues discussed in these papers represent matters specific to the design of these facilities; others address the implementation of the design certification process of 10 CFR Part 52. In the Commission papers, the staff identified those instances in which its positions differ from current regulatory requirements or in which the staff is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs. A discussion of the nature of the current regulatory requirements or their interpretation, the positions of the ALWR vendors and of EPRI, and, where available, the resolution that the staff is proposing, including the departure, if any, from current regulatory requirements and practice, and he basis for the staff's position are given for each issue. To aid in identifying its positions, the staff underlined those for which it requested Commission approval.

These issues are considered fundamental to agency decisions on the acceptability of the ALWR designs. For easy identification, Table 4B-1 lists the issues that are applicable to the Evolutionary Requirements Document with a crossreference to the chapters and sections of this report in which they are discussed. The Commission addressed SECY-90-016 and SECY-91-078 in its SRM of June 26, 1990, and August 15, 1991, respectively. The Advisory Committee on Reactor Safeguards (ACRS) forwarded its comments on these papers to the Commission by letters dated April 26, 1990, and April 23, 1991. The staff's responses to the comments of the ACRS were issued on April 27, 1990, and May 15, 1991.

The February 27 and July 6, 1992, draft Commission papers have been forwarded to the ACRS. The ACRS forwarded its comments on the February 27, 1992, draft policy paper by letters dated May 13 and August 17, 1992. The staff's response to the May 13, 1992, comments of the ACRS was issued on June 12, 1992.

The staff will include the views of the ACRS in the final papers and document its final positions before seeking Commission approval. Since the Commission has not reviewed the approaches to resolving these issues, they do not represent agency positions. Therefore, the staff regards these as open issues that must be satisfactorily resolved before it can complete its review of the Evolutionary Requirements Document. These issues will be closed once the Commission approves their resolution or provides alternative guidance.

Copies of SECY-90-016 and SECY-91-078, related ACRS reports and the staff's responses to those reports, and related Convission SRM are provided in Annexes A and B of this appendix. Annex C contains a copy of the February 27, 1992, draft Commission paper related ACRS reports, and the staff's response to one of those reports. Annex D contains a copy of the July 5, 1992, draft Commission paper.

Policy Issues	Chapter	Section
Use of physically based source term	1 1A 18 18 18 5	2.4.1.2 4.7 2.5.2 Annex A, Item I.B Annex C, Item I.A 2.7
Anticipated transients without scram	18 18 18 5	3.2.16 Annex A, Item II.A Annex C, Item I.B 4.2, 4.3
Mid-loop operation	1B 1B 3 5	Annex A, Item II.B Annex C, Item I.C 9 5.2
Station blackout	1B 1B 5 11	Annex A, Item II.C Annex C, Item I.D 2.2 5
Fire protection	18 18 5	Annex A, Item <i>i</i> I.D Annex C, ltem I.D 2.5
Intersystem loss-of-coolant-accident	18 18 18 3 5	3.2.22 Annex A, Item II.E Annex C, Item I.E 9 5.2
Hydrogen control	1 1B 18 1B 5	2.5.3 3.2.26 Annex A, Item III.A Annex C, Item I.F 2.3, 6.5, 6.6
Core-concrete interaction - capability to cool core debris	1B 1B 5 6	Annex A, Item III.E Annex C, Item I.G 6.6.2 4.3.2
High-pressure core melt ejection	18 18 5	Annex A, Item III.C Annex C, Item I.1 7.2
Containment performance	1B 1B 5	Annex A, Item III.D Annex C, Item I.J 2.1

# Table 4B.1 Policy Issues for the Evolutionary Plant Designs

ŝ.

•

Policy Issues	Chapter	Section
Dedicated containment vent penetration	18 18 18 5	2.5.3 Annex A, Item III.E Annex C, Item I.K 6.6.3
Equipment survivability	1 18 18 5	4.8.2 Annex A, Item III.F Annex C, Item I.L 6.6.6
Elimination of operating-basis earthquake	1 1B 1B 1B 1B	4 2.1.1, 3.3.1 Annex A, Item IV.A Annex C, Item I.M Annex D, Item C
Inservice testing of pumps and valves	1 18 18 5	12 Annex A, Item IV.B Annex C, Item I.N 3.1
Industry codes and standards	18 18	2.1.1 Annex C, Item II.A
Electrical distribution	18 18 11	Annex B Annex C, Item II.B 4
Seismic hazard curves	1A 1B	3.3 Annex C, Item II.C
Leak before break	1 18	4.5.5 Annex C, Item II.D
Classification of main steamline of boiling-water reactor	1B 1B 1B 2 3 13	2.3.1.1 3.2.8 Annex C, Item II.E 3.4.1.5 5.3.3, 5.4 3.1.1
Tornado design basis	1 18 18	4.5.2.5 2.1.2 Annex C, Item II.F
Containment bypass	1A 1B 5	4.4 Annex C, Item II.G 4.5
Containment leak rate testing	1B 1B 5	2.5.1 Annex C, Item II.H 2.1.3, 6.3

# Table 4B.1 (Continued)

Policy Issues	Chapter	Section
Postaccident sampling system	1B 18 3	2.3.2 Annex C, Item II.I 7
Level of detail	1B	Annex C, Item II.J
Prototyping	18	Annex C, Item II.K
Inspections, tests, analyses, and acceptance criteria	1 18	10 Anncx C, Item II.L
Reliability assurance program	1 1B	6 Annex C, Item II.M
Probabilistic risk assessment beyond design certification	1A 1B 1B	All Annex C, Item II.N Annex D, Item E
Severe-accident mitigation design alternatives	18	Annex C, Item II.O
Generic rulemaking related to FDA/DC	18	Annex C, Item II.P
Defense against common-mode failures in digital instrumentation and control systems	18	Annex D, Item A
Analysis of external events beyond the design basis	1A 1B	3 Annex D, Item B
Control room annunciator reliability	18	Annex D, Item G

### Table 48.1 (continued)

#### ANNEX A

DOCUMENTS RELATED TO SECY-90-016 "EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS" JANUARY 12, 1990
# CONTENTS

		Page
ECY-90-016, dated January 12, 1990	ł.	18.A-3
Report by the Advisory Committee on Reactor Safeguards on SECY-90-016, dated April 26, 1990		1B.A-29
Staff Response to ACRS Conclusions Regarding Evolutionary Light Water Reactor Certification Issues (Regarding SECY-90-01C), dated April 27, 1990	×	1B.A-43
Staff Requirements Memorandum, "SECY-90-016 - Evolutionary Light Water Reactor Certification Issues and Their Relationship to Curre Regulatory Requirements," dated June 26, 1990	ent	1B.A-55

3



(Notation Vote)

January 12, 1990

SECY-90-016

For: The Commissioners

From: James H. Taylor Executive Director for Operations

Subject: EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS

Purpose: To present the staff's recommendations concerning proposed departures from current regulations for the evolutionary ALWRs. The staff requests Commission approval of the positions as described in this paper.

Background: In the April 21, July 31, and August 24, 1989 staff requirements memoranda (SRMs), the Commission asked the staff to identify the issues and acceptance criteria used to judge the acceptability of future designs; to identify where the staff proposes to go beyond the regulations or to be less restrictive; and to identify if the Advanced Boiling Water Reactor (ABWR) would meet the Commission's Safety Goal with or without a vent. The Commission asked that these issues be discussed in the context of certification of the ABWR and the other evolutionary advanced light water reactor (ALWR) designs as well as the staff's review of the evolutionary Electric Power Research Institute (EPRI) Requirements Document.

Discussion: Operating experience as well as a number of studies (e.g. PRAs) have identified a number of issues significant to reactor safety. Based on this background the staff has identified the following list of issues as fundamental to agency decisions on the acceptability of evolutionary ALWR designs.

evolutionary LWR public safety goals
 source term

Contact: D.C. Scaletti, NRR/DRSP 2-1104

C.L. Hiller, NRR/DRSP 2-1118

The Commissioners

Ű.

anticipated transients without scram (3)4 mid-loop operation 5 station blackout 6 fire protection 1 intersystem LOCA 8 hydrogen generation and control 19 core-concrete interaction - ability to cool core tebris (10) high pressure core melt ejection containment performance (11)(12) ABWR containment vent design (13)equipment survivability (14) (15) operating basis earthquake/sai shutdown earthquake inservice testing of pumps and valves

- 2 -

The resolutions proposed by EPRI and the LWR vendors, and the staff positions and recommendations regarding tach of these issues are discussed in detail in the enclosure. In addition to these issues, each application for a Design Certification will have to propose technical resolutions for those Unresolved Safety Issues and medium- and highpriorty Generic Safety Issues which are identified in KUREG-D933 and technically relevant to the design in accordance with 10 CFR 52.47 and the Severe Accident Policy Statement.

The Commission's approval of, or alternate guidance on, the proposed resolution of these issues is necessary for the staff's continued review of EPRI's ALWR Requirements Document, General Electric's (GE's) ABWR, Westinghouse's RESAR SP/90 and Combustion Engineering's (CE's) System 80+ designs. Approval or guidance is particularly important to the staff's evaluations of the GL ABWR and the EPRI ALWR Requirements Document since these reviews have progressed the furthest. The certification review for CE's System 80+ is just beginning. Westinghouse has indicated that they do not intend to pursue an FLY/certification for the RESAR 1.2/90 at this time. Additions 1 Commission approval or guidance on significant issues related to certification of CE's System 80+ and other future designs will be discussed with the Commission as part of the development of the licensing review bases (LRB) for these designs. This approach is consistent with recent Commission guidance in an SRM dated December 15, 1989. It should be noted that some of the issues presented in the enclosure are issues proposed by EPRI which they refer to as plant optimization subjects. MRC approval of a plant optimization subject

Ģ

The Conmissioners

would result in a resolution that is less restrictive than present regulations, Commission policy, or past licensing practices. For these reasons, the optimization subjects such as hydrogen control, source term, and the relationship between the operating bases earthquake and the safe shutdown earthquete could have a major schedular impact on the evolutionary LWR certification proces. The staff has provided the respective applicant's proposed solutions as well as the staff's positions on these issues in the enclosure to provide a comparison, and to provide an indication of the diversity of proposed solutions under consideration by the staff. The staff recommendations identified in this paper have been developed as a result of (1) the staff's reviews of current generation reactor designs and evolutionary ALWRs, (2) consideration of operating experience, including the TMI-2 accident, (3) results of the probabilistic risk assessments (PRAs) of current-generation reactor designs and the evolutionary LWRs. (4) early efforts conducted in support of severe accident rulemaking, and (5) research conducted to address previously identified safety issues. Information related to the issues and staff positions discussed in this paper have previously been provided to the Commission in SECY-89-013. SECY-89-153, SECY-89-228 and SECY-89-341 and have been underlined in the enclosure.

The staff positions recommended in this paper are consistent with those previously taken in the staff's review of the ABWn LRB and in several ABWR-related safety evaluation reports issued to date. The staff believes that, pending detailed staff review, there is a high degree of confidence that the ABWR would meet the positions identified in the enclosure. Therefore, Commission approval of the staff recommendations would close these policy issues for the ABWR and would permit staff review to continue. The recommended positions are also consistent with those identified in the staff's draft safety evaluations related to certain chapters of the EPRI-ALWR Requirements Document. The staff is reviewing severe accident and certification issues addressed in the EPRI-ALWR Requirements Document and the staff's final conclusions are awaiting Commission approval of the positions described in this paper. Approval of the staff recommendations would allow for continuation of the staff review of the EPRI ALWR Requirements Document in accordance with recent Commission guidance.

Conclusions:

The staff believes its conclusions and recommendations regarding these matters are in keeping with the commission's

The Commissioners

policy expectation that future designs for nuclear plants will achieve a higher standard of severe accident safety performance.

The staff will inform the Commission during its reviews if additional enhancements to existing requirements, beyond those identified in the enclosure, are determined to be necessary for evolutionary ALWR designs.

Coordination:

The Office of General Counsel has reviewed this paper and has no legal objection, but notes that any program for review of new reactor designs which authorizes the NRC to impose requirements beyond those needed to meet current Commission regulations raises the issue that, if the NRC staff can pose additional requirements for certification, other parties should be able to do so as well. The traintional way to avoid such problems is through rulemaking, but as indicated in SECY 89-311, the staff believes that design certification process is a more effective method of vesolving severe accident issues than a generic severe accident rule or several individual changes. Further, OGC notes that questions regarding the desirability of additional severe actident mitigation measures still need to be addressed under NEPA, either in the design certification rulemaking and hearing or in some preliminary rulemaking.

A copy of this paper has been provided to the Advisory Committee on Reactor Safeguards.

Recommendations:

That the Commission

- (1) Approve the staff positions detailed in the enclosure, and
- (2) Note that if the staff determines other issues need to be addressed in accordance with Commission guidance, the staff will inform the Commission of its positions on these matters in a timely manner.

EPRI Evolutionary Plant SER 1B.A-6

The Commissioners

.

- (3) Note that the staff, in accordance with the Staff Requirements Memorandum dated August 24, 1989, plans to issue the draft SER on Chapter 5 of the EPRI ALWR Requirements Document.
- (4) Note that following Commission resolution of the policy issues discussed in this paper, the staff plans to finalize and reissue the above draft SEF.

James M. Tayley Executive Director for Operations

Enclosure: As stated

Commissioners' comments or consent should be provided directly to the Office of the Secretary by COB Tuesday, January 30, 1990.

. 5 .

Commission Staff Office comments, if any, should be submitted to the Commissioners NLT <u>Tuesday</u>, January 23, 1990, with an information copy to the Office of the Secretary. If the paper is of such a nature that it requires additional time for analytical review and comment, the Commissioners and the Secretariat should be apprised of when comments may be expected.

DISTRIBUTION: Commissioners OGC OIG GPA REGIONAL OFFICES EDO ACRS ACNW ASLBP ASLAP SECY

#### ENCLOSURE

#### EVOLUTIONARY ALWR CERTIFICATION ISSUES

### 1. General Issues

A. ALWR Public Safety Goal

The EPRI Requirements Document proposes that the evolutionary ALWRs comply with the following public safety goals:

- The frequency of core damage will be less than
  1.0 X 10E-5 events/per reactor-year, and
  (Note: EPRI refers to this as a "quantitative investment protection goal")
- (2) Whole body dose at an assumed 0.5 miles site boundary must be less than 25 rem for events whose cumulative frequency exceeds 1 X 10E-6 per reactoryear.

In the Licensing Review Bases (LRE) for the ABWR, GE has committed to meet the following goals:

- Demonstrate that the likelihood of core damage will have a mean value of less than one in one hundred thousand reactor years (i.e., 1.0 X 10E-5).
- (2) The expected mean frequency of occurrence of offsite doses in excess of 25 rem beyond a half mile radius from the reactor is to be less than once per million reactor years (i.e., 1.0 X 10E-6), considering both internal and external events.
- (3) The containment design is to assure that the containment conditional failure probability is less than one in ten when weighted over credible core damage sequences.

The staff is presently reviewing the LRB for the System 80+ design in which CE has proposed goals which are similar to the goals developed in the ABWR LRB. Since Westinghouse has no immediate plans of pursuing certification of RESAR SP/90, work on the LRB is presently not planned. However, similar to GE and CE. Westinghouse has stated, in meetings with the starf and the Advisory Committee on Reactor Safaguards, they are committed to meeting the ALWR public safety goals as well as the goals in the Commission's Safety Goal Policy Statement.

The staff is reviewing the proposed ALHR public safety goals to ensure they are consistent with the Commission's Safety Goal Policy Statement, which proposed both qualitative as well as quantitative safety goals for future reactor designs. The current regulations do not specify requirements in numerical terms of frequency of core damage or large release events. However, the Commission in its Safety Goal Folicy Statement, has proposed that the staff examine a general performance guideline that "the overall mean frequency of a large release of radioactivity to the environment from a reactor accident should be less than 1 in 1,000,000 per year of reactor operation." SECY-89-102 recommended approval by the Commission of the use of the following quantitative objectives in its implementation of the Safety Goal Policy for future standardized plants:

- The mean core damage frequency target for each design should be less than 1. 1.0 X 10E-5 event per reactor-year, and
- The overall mean frequency of a large release of radioactive muterials to 2. the environment from a reactor accident should be less than 1 in 1,000,000 per year of reactor operation where a large release is defined as one that has a potential for causing an offsity early fatality.

The staff concludes that the staff-proposed quantitative safety goals submitted in SECY-89-102 are consistent with the Commission's Safety Goal Policy Statement. Additional Commission guidance on the establishment of quantitative goals and im-plementation of safety goal policy will assist the staff in its continuing assessment of the evolutionary ALWRs. Although the staff considers the goals defined in SECY-BP-102 to be acceptable for evolutionary ALWRs, it should be noted that both the EPRI public safety goal and the ABWR public safety goal are considerably more stringers then the large related considers the goals defined in SECYstringent than the large release guideline defined in SECY-89-102.

Although the staff has indicated it believes the ALWR Public Safety Goal contains meritable goals for the industry to adopt, the staff has not completed its review of this issue and is in the process of reviewing how EPRI implements these goals.

#### B. Source Term

As noted in SECY 89-341, the staff's methodology for determining compliance with the siting requirements of 10 CFR Part 100 has been based on the 1962 "TID-14844" source term. This methodology, which involves calculation of offsite dose for comparison against Part 100 dose criteria (i.e. criteria for establishing the size of the exclusion area and the low population zone), is widely acknowledged to utilize conservative assumptions. At the time this asproach was developed, these conservatisms were considered appropriate and were based on uncertainties associated with accident sequences and equipment performance; and as a means to assure that future plant sites would be essentially equivalent to sites approved up until that time. The conservatisms initially included in the methodology have been essentially retained up to this time.

EPRI has stated that the evolutionary ALWR licensing design-basis requirements as well as design enhancements related to severe accidents should be based on the full body of current knowledge regarding accident source terms. They believe that the evolutionary designs should be evaluated based on a realistic treatment of fission product source terms, including the extensive research that has been done on fission product behavior since TID-14844 was issued, and especially since the Three Mile Island accident in 1979. EPRI's view is that this approach will result in designs which are improved and provide enhanced safety protection. EPRI has identified this as a plant optimization issue.

GE has indicated that the ABWR will meet the offsite dose criteria established in 10 CFR Part 100. However, they propose to utilize updated information such as system performance and reliability information, developed since promulgation of Part 100, to justify some departure from the current methodology for calculating the offsite dose. The ABWR's current design includes a single stand-by gas treatment system (SGTS) charcoal filter bed, and no main steam isolation valve (MSIV) leakage control system (LCS). Previous BWR designs utilized redundant SGTS charcoal filter beds and, since 1976, most have been equipped with a HSIV-LCS. The staff's interpretation of the General Design Criteria (GDCs) would classify filters as active components and require redundancy to permit any dose reduction credit in calculating a Part 100 dose. Since 1976, MSIV-1CSs have been required in most BWRs to meet 10 CFR Port 100 for design bases acc." Ints. Part 100 requires equipment used to mitigate consequences of design basis accidents to be seismically designed (it identifies equipment necessary to mitigate the consequences of accidents whose offsite consequences are comparable to the Part 100 dose guidelines as designed to withstand the vibratory motion of an SSE). Since non-safety grade equipment such as piping downstream of the MSIVs and the condenser are not seismically designed for SSE, credit for these systems has not been accorded in calculating offsite doses for Part 100 purposes. The staff is considering these deviations from the current methodology for demonstrating compliance with Part 100. The staff has concluded, based on current information and experience, that some deviation from current practice, or exemptions from the regulations identified above, may be warranted in the review of evolutionary designs. Presently, the staff believes that no other deviations would be necessary to demonstrate ABWR compliance with Part 100.

The other evolutionary ALWR vendors (Westinghouse and Combustion Engineering) have indicated that their evolutionary designs will comply with 10 CFR Part 100 and that they will work with the NRC and EPRI to utilize more realistic source term information to assess design enhancements related to severe accidents.

As stated in SECY 89-341, the staff is undertaking an examination of the implications of decoupling siting from plant design for future reactors. Under this plan, reactor site characteristics would be reviewed separately from the reactor without utilizing source terms or dose calculations. This would require revision to Part 100 and other regulatory staff practices. The results of such a study will establish appropriate guidelines for any future plant license applications. In the interim, however, the staff recommends that the Commission approve the following approach for evolutionary ALWRS:

Assure that evolutiona	y desi;	ns meet	the requi	irements of	f 10 CFk 100
------------------------	---------	---------	-----------	-------------	--------------

Conside	r devi	lotic	ons	to	çur	ren	t ene	ethod	010	QY I	1111	ized	to	alculate.	Part
100 dos	es on	0 01	ase-	by-	Cas	e bi	0515	s uti	112	ing	eng	Ineer	ing	Judgement	Carlo and a stranger server
includi	ng upo	dated	d in	for	mat	ion	07	SOUT	ce	terr	n an	d equ	IDM	ent relieb	ilit

Such deviations could impact plant design features, therefore, these deviations will be identified in the SERs that will be forwarded to the Commission for its information well in advance of issuance, as directed the SRM pertaining to SECY 89-311 dated December 15, 1989.

- Do not modify current siting practice, even though it is recognized that such deviations could result in calculated low population zones and exclusion areas which are smaller than those that have been approved for currently operating reactors.
- Continue to interact with EPRI and the evolutionary ALWR vendors to reach agreement on the appropriate use of updated source term information for severe accident performance considerations.

11. Preventative Feature Issues

A. Anticipated Transfent Without Scram (ATWS)

The ATWS rule 10CFR 50.62 was promulgated to reduce the probability of an ATWS event and to enhance mitigation capability if such an event occurred.

EPRI has indicated that its approach to resolving the ATWS issue is compliance with the ATWS rule. Design requirements beyond those which would be required to meet the rule have not been proposed.

The ABWE design includes a number of features that reduce the risk from an ATWS event. These features include a diverse scram system with both hydraulic and electric run-in capabilities on the control rods, a manually operated standby liquid control system (SLCS), and a recirculation pump trip capability. In addition, the scram discharge volume has been removed from the ABWR, eliminating some of the potential ATWS problems associated with the older BWR designs. While the ATWS rule requires an automatically initiated SLCS, GE has concluded that the diverse scram system and enhanced reliability of the reactor protection system negates the need for an automatic SLCS. GE has agreed to provide a reliability analysis in order to support this position. The staff will review the analysis to determine if an exemption from 10 CFR 50.62, to approve manual SLCS, is justified. The staff analysis will be provided in a future safety evaluation report for the ABWR.

Westinghouse has concluded that a diverse scram system is unnecessary for the RESAR SP/90 design due to 1) high reliability of the integrated reactor protection system (IPS), 2) a turbine trip and emergency feedwater actuation that is independent of the IPS, 3) Ability to manually rip the rod contro<sup>3</sup> motor generators from the main control board, and 4) a highly negative moderator temperature coefficient. Westinghouse has committed to provide a detailed analysis to demonstrate that the consequences of an ATWS are acceptable at the time an FDA application is submitted.

The CE System 80+ design includes a control-grade Alternate Protection System which provides an alternate reactor trip signal and an alternate feedwater actuation signal separate and diverse from the safety-grade reactor trip system.

EPRI Evolutionary Plant SER

## B. Mid-Loop Operation

The staff is concerned that decay heat removal capability could be lost when a PWR is shut down for refueling or maintenance and drained to a reduced reactor coolant system (RCS) or "mid-loop" level. For example, a significant problem has been the loss of residual heat removal (RHR) suction due to air-binding of the RHR pumps. This is usually caused by an uncontrolled low loop level and consequent air ingestion into the pump suction.

The EFEJ Requirements Document specifies requirements consistent with measures applicable to cperating reactors as described by the administrative procedures identified in Generic Letter 88-17, but does not specify design modifications to address the root cause of this event.

Westinghouse has committed to install a vortex breaker at the RHR hot leg connections to significantly reduce air entrainment during mid-loop operation. This feature, in conjunction with other design features of the plant, should greatly reduce concerns over mid-loop operation. CE has indicated that it will address this issue through analysis, consideration of specific design features, and/or operational restrictions. Specific design resolutions for the System 80+ have not been provided. Mid-loop operation is not an issue with the ABWR.

The staff expects improvements in instrumentation in many existing PWRs, but does not require specific modifications to the nuclear steam supply system (NSSS) to correct mid-loop problems. However, the staff believes that physical modifications such as those proposed by Westinghouse, may be necessary to essentially eliminate any concerns with mid-loop operation for future evolutionary pressurized ALWRs. Mid-loop operation is not explicitly covered by current regulations, however imposition of such requirements would exceed current staff licensing practices. Therefore, the staff recommends that the <u>Commission approve the staff position that evolutionary PWR vendors propose</u> design features to ensure high reliability of the shutdown decay heat removal system.

hi

## C. Station Blackout

The station blackout rule (10 CFR 50.63) allows utilities several design alternatives to ensure that an operating plant can safely shut down in the event that all ac power (offsite and onsite) is lost.

The EPRI Requirements Document provides for improvements in offsite power reliability, onsite power reliability and capacity, and station blackout coping capability. EPRI is also proposing that a large capacity, diverse alternate ac power source (combustion turbine generator) with the capability to power one complete set of normal safe shutdown loads be included in evolutionary ALWR designs.

The RESAR SP/90 emergency feedwater system includes two ac-independent and two dc-independent turbine-driven pumps. The electrical design includes two full capacity emergency diesel generators. In addition, it includes a backup seal injection pump powered by a small dedicated diesel generator which has enough capacity to also charge the station batteries. Westinghouse believes that this design will provide a 24-hour coping time which is sufficient to eliminate the need for the addition of an installed spare (full capacity) alternate ac power source.

The System EO+ design includes two turbine-driven emergency feedwater pumps and two nutor-driven emergency feedwater pumps. The electrical design includes two full capacity emergency diesel generators and a diverse alternate ac power source. This alternate source of ac power is expected to be a control-grade combustion turbine with sufficient capacity and capability to power either one of the electrical divisions. In addition, the plant design has full load rejection capability and the capability to subsequently provide electrical power from the turbine generator. Each of the four safety-related instrument channels has a dedicated battery backup. Class IE electrical Divisions I and II, which include the two emergency diesel generators are each provided dc power by an assigned pair of these batteries.

The ABWR design includes three independent electrical divisions, each with high-pressure and low-pressure water injection capability, each powered by a full capacity emergency diesel generator, and each division capable of independently shutting down the reactor. Additionally, the ABWR design includes an alternate ac combustion turbine to back up the diesel generators. The design has a capability to survive a 10-hour blackout period utilizing the reactor core isolation cooling (RCIC) turbine and station betteries. Extended blackout capabilities are also provided by the ac-independent water addition system. This system allows for makeup to the reactor vessel following RCS depressurization by connecting a direct drive diesel fire pump or by connecting an external pumping source, such as a fire truck, to a yard standpipe.

The staff believes that the preferred method of demonstrating compliance with 10 CFR 50.63 is through the installation of a spare (full capacity) alternate ac power source of diverse design that is consistent with the guidance in Reculatory Guide 1.155, and is capable of powering at least one complete set of

normal safe shutdown loads. Although an alternate ac power source is provided as an acceptable resolution to this issue in 10 CFR 50.63, staff imposition would exceed current Commission regulations. Therefore, the staff recommends that the Commission approve imposition of an alternate ac source for evolutionary ALWRS.

- 7 -

## D. Fire Protection

The staff has concluded that fire protection issues that have been raised through operating experience and through the External Events Program must be resolved for evolutionary ALWRs. To minimize fire as a significant contributor to the likelihood of severe accidents for advanced plants, the staff concludes that current NRC guidance must be enhanced. Therefore, the evolutionary ALWP designers must ensure that safe shutdown can be achieved, assuming that all equipment in any one fire area will be rendered inoperable by fire and that re-entry into the fire area for repairs and operator actions is not possible. Because of its physical configuration, the control room is excluded from this approach, provided an independent alternative shutdown capability that is physically and electrically independent of the control room is included in the design. Evolutionary ALWRs must provide fire protection for redundant shutdown systems in the reactor containment building that will ensure, to the extent practicable, that the shutdown division will be free of fire damage. Additionally, the evolutionary ALWR designers must ensure that smoke, hot gases, or the fire suppressant will not migrate into other fire areas to the extent that they could adversely affect safe-shutdown capabilities, including operator actions. Because the layout of a nuclear plant is designspecific, plant-specific design details will be reviewed by the staff on an individual basis. The staff will require a description of safety-grade provisions for the fire-protection systems to ensure that the remaining shutdown capabilities are protected, as well as demonstration that the design complies with the migration criteria discussed above.

The ALWR Requirements Document indicates that fire protection will be as specified in 10 CFR 50.48 and Appendix R. It states that for equipment in the same general area, a 3-hour fire barrier will be utilized in lieu of physical separation unless it is "impractical or less safe." However, no guidelines are provided in the Requirements Document as to the application of these criteria.

The evolutionary ALWR designers have indicated that their fire protection designs are consistent with the staff's proposed enhancements. GE has provided its ABWR fire protection analysis which is currently under review by the staff.

Appendix R to 10 CFR Part 50 was promulgated for plants that were in operation prior to January 1, 1979. Subsequently, PRAs performed on more than a dozen plants have showed that fire is a significant contributor to core damage. The staff believes that in keeping with the Commission's desire for enhanced safety for evolutionary ALWRs, fire protection requirements should reflect experience from operating reactors and the greater understanding of sev. re accidents that has been acquired since Appendix R was promulgated. Therefore, the staff recommends the Commission approve the use of the enhanced fire protection position underlined above for evolutionary ALWRS.

## E. Intersystem LOCA

161

Future evolutionary ALWR designs can reduce the possibility of a loss-of-coolant accident (LOCA) outside containment by designing (to the extent practicable) all systems and subsystems connected to the reactor coolant system (RCS) to an ultimate rupture strength at least equal to the full RCS pressure.

For both BWRs and PWRs, EPRI states that low-pressure systems which could be overpressurized by the RCS should be designed with sufficient margin to withstand full RCS pressure without structural failure.

For BWRs, pressure isolation valve instrumentation and controls should be provided to (1) prevent opening shutdown cooling connections to the vessel in any loop when the pool suction valve, discharge valve, or spray valves are open in the same loop, (2) prevent opening the shutdown connections to and from the vessel whenever the RCS pressure is about the shutdown range, (3) automatically close shutdown connections when RCS pressure rises above the shutdown range, and (4) prevent operation of shutdown suction valves in the event of a signal that the water level in the reactor is low.

For PWRs, relief values sized to protect against overpressure transients, should be provided on the RHR system. RHR suction values should be provided with permissive interlocks to prevent opening if RCS pressure exceeds RMR design pressure.

Westinghouse has indicated that, should the isolation valve, of the RESAR SP/90 fail, the design pressure of the piping outside of the containment will be sufficient to withstand primary side pressure or will be vented to the Emergency water storage tank (EWST).

CE has eliminated the low-pressure safety injection system and increased the design pressure of the shutdown cooling system piping in the System 80+ design. With this higher design pressure, the shutdown cooling system is expected to maintain its integrity even when exposed to full reactor coolant system pressure.

The ABWR has been designed to minimize the possibility of an interfacing system LOCA in the following ways. The low pressure systems directly interfacing with the RCS are designed with 500 psig piping which provides for a rupture pressure of approximately 1000 psig. In addition, the high/low-pressure motor-operated isolation valves have safety-grade, redundant pressure interlocks. Also, the motor-operated emergency core cooling system (ECCS) valves will only be tested when the reactor is at low pressure. All inboard check valves on the ECCS will be testable and have position indication. Additionally, design criteria used by GE require that all pipe designed to 1/3 or greater of reactor pressure reactor system messure. The pipe designed to less than 1/3 reactor pressure requires at loss three malfunctions before the pipe would be subjected to reactor system pressure.

The staff concludes	that designing,	to the extent	practicable,	low-pressure
systems to withstand	d full KCS pressu	ire is an accer	ptable means	for resolving
this is: . However	r, the staff beli	leves that for	those system	s that have
not been "signed t	c withstand full	RCS pressure,	evolutionary	ALWRs should

EPRI Evolu onary Plant SER

provide (1) the capability for leak testing of the pressure isolation valves, (2) valve position indication that is available in the control room when isolation valve operators are deenergized and (3) high-pressure alarms to warn control room operators when rising RCS pressure approaches the design pressure of attached low-pressure systems and both isolation valves are not closed. Imposition of these recurrements exceed Commission regulations and guidance; therefore, the staff ommends that the Commission approve these positions for evolutionary ALWRs.

The staff notes that for some low-pressure systems attached to the RCS, it may not be practical or necessary to provide a higher system ultimate pressure capability for the entire low-pressure connected system. The staff will evaluate these exceptions on a case-by-case basis during specific design certification reviews.

III. Mitigative Feature Issues

A. Hydrogen Generation and Control

The Commission's Severe Accident and Standa dization Policy Statements provide that future designs should address the provisions of 10 CFR 50.34(f). The Commission's stated policy has been codified in 10 CFR Part 57 to require the technically relevant provisions of 10 CFR 50.34(f) be met. Specifically, in order that containment integrity be maintained, 10 CFR 50.34(f)(2)(ix) requires future designs to provide a system for hydrogen control that can safely accommodate hydrogen generated by the equivalent of a 100 percent fuel-clad metal-water reaction. In addition, the regulation requires this system to be capable of precluding uniform concentrations of hydrogen from exceeding 10 percent (by volume), or an inerted atmosphere within the containment must be provided.

The ALWR Requirements Document specifies that containment and combustible gas control systems should be designed to accommodate 75 percent in-vessel zirconium-water reaction of the active fuel cladding, and 13 percent containment uniform hydrogen concentration. It states that 75 percent cladding raidation is believed to be a conservative upper limit on the amount of hydrogen generated in a degraded-core situation including recovery. EPRI has identified this as an optimization issue.

The RESAR SP/90 design proposes to mitigate the effects of a 100 percent metalwater reaction and to preclude uniform hydrogen concentration from exceeding 10 percent (by volume) through the use of hydrogen igniter and hydrogen recombiner systems.

The System 80+ design proposes to be consistent with the recommendations of the ALWR Requirements Document resulting from staff review. The information will include justifications for the assumed extent of metal-water reaction and the allowable uniform hydrogen concentrations.

The ABWR design meets the requirements of 10 CFR 50.34(f)(2)(ix) by utilizing a nitrogen-inerted atmosphere within its containment. Also, a hydrogen recombiner for design-basis accidents will be provided in the ABWR design.

Aside from the issue of regulatory compliance and applicability, and due to the uncertainties in the phenomenological knowledge of hydrogen generation and combustion, the staff concludes that compliance with the criteria of 20 CFR 50.34(f) remains appropriate for combustible gas control design in ALWRs. Research (discussed in NUREG/CR-4551) indicates that in-vessel hydrogen generation associated with core-damage accidents may range from approximately 40-95 percent active cladding oxidation equivalent. The amount of cladding oxidation is dependent on a variety of parameters related to sequence progression: reactor coolant system pressure, reflood timing and flow rates, as well as core-melt progression phenomena. Thus, a 75-percent-equivalent cladding reaction continues to be viewed as a reasonable design basis for hydrogen generation for severe accidents in which the reactor pressure vessel (RFV) remains intact. However, it is the staff's view that ALWRs should provide protection for hydrogen generation resulting from a wider spectrum of accidents, i.e., full core-melt accidents with RPV failure. In that context, it is also necessary to consider ex-vessel hydrogen generation as a result of core debris reacting with available water or core-concrete interactions. Calculations using the CORCON models indicate that if the core debris is cooled in relatively rapid fashion (1-2 hours), additional hydrogen generation will be less than that equivalent to a 25-percent cladding oxidation reaction. This relatively limited ex-vessel reaction is conditional on the existence of a coolable debris bed and the availability of sufficient water. If extensive core-concrete interaction occurs due to the absence of cavity flooding, more hydrogen generation should be considered. Considering the effects discussed above, the staff concludes that an equivalent 100 percent cladding oxidation reaction is an appropriate deterministic design criteria and a reasonable surrogate for the combination of both in-vessel and ex-vessel hydrogen generation.

Due to the uncertainties in the phenomenological knowledge of hydrogen generation and combustion, it is still the staff's position that, as a minimum, evolutionary ALWRs should be designed to (1) accommodate hydrogen equivalent to 100-percent metal-water reaction of the fuel cladding and (2) limit containment hydrogen concentration to no greater than 10 percent. Furthermore, because hydrogen control is necessary to preclude local concentrations of hydrogen below detonable limits, and given uncertainties in present analytical capabilities, the staff concludes evolutionary ALWRs should provide containment-wide hydrogen control (e.g., igniters, inurting) for severe accidents. Additional advantages of providing hydrogen control mitigation features (rather than reliance on random ignition of richer mixtures) includes the lessening of pressure and temperature loadings on the containment and essential equipment. The staff recommends that the Commission approve the staff's position that the requirements of 10 CFR 50.34(f)(2)(ix) remain unchanged for evolutionary ALWRs.

B. Core-Concrete Interaction - Ability To Cool Core Debris

In the unlikely event of a severe accident in which the core has melted through the reactor vessel, it is possible that containment integrity could be breached it is emplted core is not sufficiently cooled. In addition, interactions between the core debris and concrete can generate large quantities of additional hydrogen and other non-condensible gases, which could contribute to eventual overpressure failure of the containment.

The EPRI Requirements Document contains a number of design features that are intended to mitigate the effects of a molten core. To promote long-term debris coolability, the Requirements Document states that the cavity floor should be sized to provide 0.02 m<sup>2</sup>/MWt. The Requirements Document also specifies that the containment should be designed to ensure adequate water supply to the floor and that an alternate means of introducing water into the containment, independent of normal and emergency ac power, should be provided. Passive schemes for providing flooding of the floor areas beneath the vessel are proposed and described in general terms for both BWRs and PWRs. The Requirements Document also states that the steel shell or liner of the containment should be protected from core debris by at least 3 feet of concrete.

Westinghouse indicated that they will comply with the EPRI core-debris dispersal criteria of 0.02 m<sup>2</sup>/MWt and that the RESAR SP/90 design will include some method (not yet defined) that would ensure automatic flooding of the lower cavity, using the in-containment refueling water storage tank (RWST), in the event of a severe accident. Westinghouse is currently evaluating alternative designs to ensure compliance with that commitment.

CE has also indicated that the System 80± design will comply with the EPRI core-debris dispersal criteria of 0.02 m<sup>2</sup>/Mwt. Also, the in-containment refueling water storage tank will provide a source of water for lower cavity flooding.

The ABWR design has a number of features that the staff generally agrees would mitigate the effects of a molten core. It is designed with a lower drywell flooder and a cavity space sufficient to be able to disperse core debris at an energy level of 0.02 m<sup>2</sup>/MWt. The flooder consists of a number of temperature-sensitive fusible plugs that allow suppression pool water to enter the drywell cavity when high temperature resulting from core debris occurs in the lower drywell. The horizontal vents to the suppression pool will remain covered in the event of lower drywell flooding, ensuring that releases continue to be scrubbed through the suppression pool water. GE anticipates that any core-concrete interaction will be stopped when the suppression pool water quenches the molten core debris. By providing sufficient area to allow the core debris to spread to a shallow bed and by flooding the core debris, it is expected that the potential for extensive core-concrete interactions will be significantly reduced. In addition, even if limited core-concrete interactions continue, the overlying pool of water will mitigate the consequences of these interactions by scrubbing the fission products and cooling the gases released from the core-concrete interaction.

The staff believes that an acceptable resolution to this issue can be provided by the evolutionary ALWR vendors if their designs

- provide sufficient reactor cavity floor space to enhance debris spreading, and
- provide for quenching debris in the reactor cavity.

### Use of these criteria exceed current regulatory practice.

It should be noted that the specific cavity sizing criteria (0.02 m<sup>2</sup>/MWt) proposed in the Requirements Document is still under evaluation by the staff. The issue of debris coolability is an area in which there is active ongoing experimental research including relatively large scale testing jointly sponsored by EPRI and NRC. Additionally, without assurance of core debris coolability, the level of protection afforded by a 3-foot thickness of concrete and the issue of vessel pedestal attack (ablation of concrete supporting the reactor vessel by the molten core debris) require further evaluation. The staff will continue to evaluate the issue of core debris coolability and the specific cavity sizing criteria (0.02m<sup>2</sup>/MWt) proposed by EPRI as more data and information becomes available. The staff intends to assess the debris flooding schemes proposed by EPRI on a design-specific basis.

The staff recommends the Commission approve exceeding past regulatory practice in resolving this issue. The staff recommends approval of the general criteria, stated above, that evolutionary ALWR designs; 1) provide sufficient reactor cavity floor space to enhance debris spreading, and 2) provide for quenching debris in the reactor cavity. Design specific approaches to resolve this issue will be evaluated by the staff on a case-by-case basis to ensure compliance with these criteria.

C. High Pressure Core Melt Ejection

One potential effect of a severe accident that could potentially result in containment failure is the phenomenon of direct containment heating (DCH). The staff is concerned that this phenomenon might occur from the ejection of molten core debris under high pressure from the reactor vessel resulting in wide dispersal of core debris and extremely rapid addition of energy to the containment atmosphere.

To limit direct containment heating, the ALWR Requirements Document states that the cavity/pedestal/drywell configuration should be designed to preclude entrainment of core debris by gases ejected from a failed reactor vessel. It also states that a safety-grade RCS safety depressurization and vent system (SDVS) will be provided. The staff review has concluded that reactor vessel depressurization capability and cavity design features to entrap ejected core debris constitute an acceptable approach to the issue of high-pressure core melt ejection.

Westinchouse has indicated that the configuration of the cavity of the RESAR SP/90 containment will prevent core debris from entering the upper containment. In addition, their ac independent depressurization system will reduce the probability of a high-pressure molten-core ejection from the reactor vessel.

CE has indicated the System 80+ design includes an indirect cavity vent path, including a debris collection chamber, (which is configured to de-entrain solid core debris and minimize direct containment heating) and a large floor area to enhance core debris coolability. In addition, the design includes a safety grade depressurization system which minimizes the possibility of high-pressure molten-core ejection.

The ABWR design incorporates a safety grade depressurization system and a suppression pool that surrounds the lower drywell cavity and thereby reduce the risk of high pressure core ejection and would prevent core debris from reaching the containment boundary and breaching its integrity.

The staff concludes that, during a high-pressure core-melt scenario, a depressurization system should provide a rate of RCS depressurization to preclude molten-core ejection and to reduce RCS pressure sufficiently to preclude cree; rupture of steam generator tubes. Primary systems of evolutionary ALWRs should have the capability to be depressurized after loss of decay heat removal. In addition, the staff concludes that the ALWR Requirements Document should include a requirement that reactor cavities be arranged in such a manner that high-pressure core debris ejection resulting from vessel failure will not impinge on the containment boundary. The staff concludes that ALWR designs should include a depressurization system and cavity design features to contain ejected core debris. Imposition of these requirements exceed current Commission regulations. The staff recommends that the Commission approve this position for evolutionary ALWRs.

D. Containment Performance

The containment function, i.e., maintenance of a strong leak tight barrier against radioactive release, is faced with distinct challenges as a result of a severe accident. These challenges may be roughly divided into two categories, energetic or rapid energy releases and slower, gradually evolving releases to the closed containment system. Examples of containment loadings that fall into the first category include high-pressure core melt ejection with direct containment heating, hydrogen combustion, and the initial release of stored energy from the reactor coolant system. Slow energy releases to the containment are typified by decay heat and noncondensible gas generation. Engineering practice in containment design calls for passive capability in dealing with energetic energy releases where practicable while long-term energy releases may be controlled by both passive means as well as through active intervention.

In view of the low probability of accidents that would challenge the integrity of the containment, the staff concludes that the probability of failure of the mitigation systems (those systems which can reduce the consequences of a core damage accident), from the onset of core damage to loss of containment integrity resulting in an uncontrollable leakage substantially greater than the design basis leakage, should not exceed approximately 0.1. However, the staff intends

EPRI Evolutionary Plant SER

to ensure that the containment can deal with all credible challenges and does not intend to apply this conditional containment failure probability (CCFP) guideline in a manner that could be interpreted to potentially detract from overall safety. The staff will accept a CCFP of 0.1 or a deterministic containment performance goal that offers comparable protection. For this reason, the staff concludes that the following general criterion for containment performance ouring a severe-accident challenge would be appropriate for the evolutionary ALWRs in place of a CCFP.

The containment should maintain its role as a reliable leck tight barrier by ensuring that containment stresses do not exceed ASME service level C limits for a minimum period of 24 hours following the onset of core damage and that following this 24 hour period the containment should continue to provide a barrier against the uncontrolled release of fission products.

Maintaining containment integrity for a minimum period (e.g. 24 hrs) is based on providing sufficient time for the remaining airborne activity in the containment (principally noble gases and iodine) to decay to a level that would not exceed 10 CFR Part 100 dose guideline values when analyzed realistically. if controlled venting were to occur after that time. During this period, containment integrity should be provided, to the extent practicable, by the passive capability of the containment itself and any related passive design features (e.g., suppression pool). The staff further believes that following this period, the containment should contine to provide a barrier against the uncontrolled release of fission products. However, in keeping with the concept of allowing for intervention in coping with long-term or gradual energy release, the staff believes that after this minimum period, the containment design may utilize controlled, elevated venting to reduce the probability of a catastrophic failure of the containment. Alternatively, a design may utilize diverse containment heat removal systems or rely on the restoration of normal containment heat removal capability if sufficient time is available for major recovery actions (e.g., 48 hours).

EPRI has indicated that the ALWR public safety criteria do not contain explicit criteria for conditional probability of containment failure or other mitigation features since the ALWR Steering Committee believes that such criteria could potentially distort the balance in safety design and inhibit innovative improvements in core protection features. However, EPRI has not yet indicated their position on an alternate containment performance goal.

Westinghouse has not yet committed to a specific containment performance goal for RESAR SP/90 although it is expected that the mitigation features discussed by Westinghouse would lead to a CCFP of less than 0.1 for all credible accident scenarios.

CE expects that the System 80+ design will meet the CCFP goal of 0.1 when weighted over credible core damage sequences given the following assumptions.

-Credible core damage sequences are defined as all core damage event sequences with a frequency of greater than 1.0 X 10E-6 per reactor year. External events which would cause both core damage and concurrently fail the containment and which have a frequency of less that 1.0 X 10E-5 per reactor year will not be considered.

-Containment failure is defined as a post core damage release resulting in a dose greater than 25 rem beyond one-half mile from the reactor.

The ABWR design currently includes a hardened wetwell vent for containment over pressure protection and is committed to a CCFP that is less than 0.1 when weighted over credible core damage sequences. In meetings with the staff, EPRI has stated that they consider a containment vent to be philosophically and institutionally undesirable and potentially unworkable. For additional information see related discussions under ALWR Public Safety Goal and ABWR Containment Vent Design.

Defense in depth, a long standing fundamental principle of reactor safety, results in the concept that multiple barriers should be provided to ensure against any significant release of radioactivity. In its Severe Accident Policy Statement, the Commission indicated that it "... fully expects that verdors engaged in designing new (or custom) plants will achieve a higher standard of severe accident safety performance than their prior designs." Commission reaffirmed this policy in an SRM dated December 15, 1989 relating to SECY-89-311. A defense-in-depth approach reflects an awareness of the need to make conservative safety judgements in the face of uncertainties; in effect, not putting all the eggs in one basket. In that regard, the reactor containment boundary should serve as a rollable barrier against fission product release for credible sovere-accident phenomena/challenges. Special effort should be made to eliminate or further reduce the likelihood of a sequence that could bypass the containment. The continued reliance on the traditional principle of containment of fission products following an accident is seen as a logical and prudent approach to addressing reasonable questions which will persist regarding the ability to accurately predict certain aspects of severe accident behavior. In order to ensure balance between prevention and mitigation, some criteria on containment performance are appropriate. Accordingly, a general goal of limiting the conditional containment failure probability to less than 1 in 10 when weighted over credible core-damage sequences would constitute appropriate attention to the defense-in-depth philosophy. Alternatively, a deterministic containment performance goal that provides comparable protection would be appropriate.

Probabilistic risk assessment (PRA) is a very powerful tool that permits systematic integrated assessment of design strengths and weaknesses. However, because very low frequency scenarios (approximately 1.0 X 10E-6 per reactor-year) are being addressed, it is important to recognize the large uncertainties in the quantification of these scenarios. The overall uncertainties in severe accident behavior are driven largely by insufficient data for assessing common-cause failures, difficulty in quantification of the potential for human

EPRI Evolutionary Plant SER

errors, and questions about completeness of analyses and uncertainties in phenomenological behavior. For this reason, the staff considers it acceptable to utilize a deterministic containment performance criterion that would provide a level of containment performance comparable to that which could be demonstrated using a probabilistic containment failure goal of 0.1, given a severe accident.

It is recommended that the Commission approve the staff's position to use a CCFP of 0.1 or a deterministic containment performance goal that offers comparable protection in the evaluation of evolutionary ALWRs.

E. ABWR Containment Vent Design

In Amendment 8 of the ABWR SSAR submittal (July 28, 1989) GE submitted a sensitivity analysis of the ABWR PRA to determine the net risk benefit of a vent system. Basically, this system is a containment overpressure relief system and is designed to avoid gross containment failure resulting from postulated slow rising overpressure scenarios that could result from postulated multiple safety system failures. These sensitivity analyses indicate that, with cr without a vent system, the ABWR design meets the quantitative health objectives of the Commission's safety goal with a wide margin.

The staff's detailed review of the ABWR risk analyses, including the sensitivity analyses on the vent system, is currently underway. Based on the review to date, the staff believes that the scope of methods and data used in the AEWR PRA are sufficient and do not expect that the ABWR risk to exceed the Commission's quantitative health objectives with or without a vent system.

The staff's safety goal implementation plan also recommended that a subsidiary target related to plant performance be used. This target states that, for future plants, a mean core damage frequency due to internal events and external events be less than 1.0 X 10E-5 per reactor year of operation. The staff's review of Amendment 8 of the ABWR SSAR indicates that the overall core damage frequency from internal events (transients, ATWS events, and postulated LOCAs) and external events (primarily from beyond design basis seismic events and postulated fires) is about 6 X 10E-6 per reactor year. GE has determined that the proposed vent system has negligible impact on the core damage frequency. The staff notes that GE has provided an additional means of decay heat removal (a third train of RHR and an ac-independent water makeup system which relies on the fire water system to supply water to the core and cuntainment sprays in emergency situations) for the ABWR design to reduce the frequency of the sequences involving loss of containment heat ramoval function, thus reducing the benefit (on core damage frequency only) of the ABWR vent system for these types of accident sequences.

The desirability of venting a BWR containment to mitigate multiple-failure accidents far beyond the design basis has been accepted for some time. Since 1981 the BWR Emergency Procedure Guidelines (EPGs), developed by the BWR Owners Group and approved by the NRC for existing BWRs, have called for venting the containment wetwell air space. GE believes containment

EPRI Evolutionary Plant SER

overpressure protection represents a practical and beneficial feature to incorporate in the ABWR. The overpressure protection feature is essentially passive, relatively inexpensive in a new plant, provides insurance against the consequences and financial risks associated with end-of-spectrum accident scenarios, is consistent with the BWR EPGs, and appears to be consistent with the ALWR philosophy of robustness.

GE has established two severe accident goals in the risk analyses submitted to the staff. These goals were defined in the ABWR LRB. The first goal states that the frequency of a severe accident release resulting in a whole body dose of 25 rem beyond one-half mile from the reactor should not exceed 1 X 10E-6 per reactor-year. This design goal is basically the same as the EPRI ALWR design goal. The second goal defined in the ABWR LRB states that the conditional containment failure probability should be less than one in ten (CCFP 0.1) when weighted over credible core damage sequences. The staff and GE agree that the definition of containment failure is an uncontrollable leakage substantially greater than the design basis resulting from loss of containment integrity following the onset of severe core damage. The ABWR design with the vent system is expected to meet the above goals; however, staff review in this area is not yet complete.

GE has performed an analysis utilizing this definition of containment failure to determine if the ABWR meets the CCFP goal of 0.1. The analysis indicates that the CCFP for the ABWR design, without a vent system, is equal to approximately 0.5 and does not meet the 0.1 goal, however with a vent system, the CCFP equals approximately 0.06.

Based upon the preliminary review of the ABWR severe accident design, the staff has determined that, as far as overall risk impact is concerned, the GE ABWR public safety goal is significantly more stringent than the Commission's quantitative health objectives. Also, the staff concludes that the public safety goal proposed by GE for the ABWR design is more stringent than the "large release guideline" as defined in the staff's proposed safety goal implementation plan. Therefore, based on the apparent enhanced level of safety provided by the ABWR's severe accident design features, which include the over pressure protec-tion system, the staff recommends the Commission approve its use in the ABWR design certification process.

## F. Equipment Survivability

with regard to the Commission's request concerning "The measures to ensure that systems and equipment required only to mitigate severe accidents are available to perform their intended function (e.g., environmental qualifications)," the staff believes that features provided for severe-accident protection (prevention and mitigation) only (not required for design basis accidents) need not be subject to (a) the 10 CFR 50.49 environmental qualification requirements. (b) all aspects of 10 CFR Part 50. Appendix B quality assurance requirements, or (c) 10 CFR Part 50, Appendix A redundancy/diversity requirements. The reason for this judgment is that the staff does not believe that severe core damage accidents should be design basis accidents (DBA) in the traditional sense that DEAs have been treated in the past.

Notwithstanding that judgment, however, mitigation features must be designed so there is reasonable assurance that they will operate in the severe-accident environment for which they are intended and over the time span for which they are needed. In instances where safety related equipment, (which is provided for design bases accidents) is relied upon to cope with severe accidents situations; there should also be a high confidence that this equipment will survive severe accident conditions for the period that is needed to perform its intended function. However it is not necessary for redundant trains to be qualified to meet this goal.

During the review of a specific ALWR design the credible severe accident scenarios, the equipment needed to perform mitigative functions, and the conditions under which the mitigative systems must function, will be identified. Equipment survivability expectations under severe accident conditions should include consideration of the circumstances of applicable initiating events (e.g., station blackout, earthquakes) and the environment (e.g., pressure, temperature, radiation) in which the equipment is relied upon to function. The required system performance criteria will be based on the result. of these design-specific reviews. In addition, the staff concludes that severe-accident mitigation equipment for evolutionary ALWRs should be capable of being powered from an alternate power supply as well as from the normal Class IE onsite systems. Appendices A and B to Regulatory Guide 1.155, "Station Blackout," provide guidance on the type of quality assurance activities and specifications which the staff concludes are appropriate for equipment utilized to prevent and mitigate the consequences of severe accidents.

The steff requests that the Commission approve the staff position that features provided only for severe-accident protection need not be subject to the 10 CFR 50.49 environmental qualification requirements, 10 CFR Part 50, Appendix B quality assurance requirements, and 10 CFR Part 50, Appendix A redundancy/ diversity requirements.

IV. Non-Severe Accident Issues

The following issues, which are not normally considered through PRA analysis or not considered as severe accident issues for the evolutionary ALWRs, are brought to the Commission's attention because either the staff's positions or the vendor requests differ from past practices.

A. Operating Bases Earthquake(OBE)/Safe Shutdown Earthquake(SSE)

Presently, 10 CFR Part 100 requires that the magnitude of the OBE be at least ore-half that of the SSE. It has been an industry wide experience that such a requirement leads to a design that is governed by the OBE requirements and produces unnecessary and inconsistent margins for the SSE loading. This requirement was included in the regulation when the staff did not have substantial experience with the seismic resistance of plants that incorporated OBE design at half the SSE value. Since then a number of research programs have been conducted including a large industry effort on testing and observation of actual earthquake experience of industrial facilities; consequently, the NRC funded Piping Review Committee has concluded that the OBE at existing plants are too high, therefore, controlling the design of some safety systems, and recommended that the OBE be decoupled from the SSE. Certain interim measures,

such as allowing somewhat higher damping values for piping analysis, have been taken to partially implement the Piping Review Committee recommendations (NUREG 1061, 1984). But the complete implementation of the recommendations would involve a revision of 10 CFR Part 100, Appendix A. Because of higher priority work, the effort on revision of this regulation has been postponed. It should be noted that the Commission has, in certain site specific cases, previously approved OBEs of less than one-half the SSE.

EPRI has requested that NRC regulations be changed to reduce the magnitude of the OBE relative to the SSE as a basis for the design. All evolutionary ALWR vendors agree with the request. GE has stated that they agree with EPRI in principle, however, the ABWR design uses an OBE that is one-half the SSE; therefore, this is a non-issue for the ABWR. EPRI has identified this as an optimization issue.

The staff agrees that the OBE should not control the design of safety systems. However, a staff position on this issue to be applied generically to all future issigns has not yet been fully developed. For the evolutionary reactors, the stelf will consider requests to decouple the OBE from the SSE on a designspecific tasis. Such a decoupling would require an exemption to the Commission's regulations, therefore the staff recommends that the Commission approve this design-specific relief approach.

E. Inservice Testing of Farms and Valves

The ASME Code, Section XI, "Rule: for Inservice Inspection of Nuclear Power Plant Components" has been used to establish past testing requirements for ASME Code Class 1, 2, and 3 safety-relited pumps and valves. These requirements provide certain information on the operational readiness of the components, but in general, do not necessarily verify the capability of the components to perform their intended safety function. It is the staff's judgement that the Code does not assure the necessary level of component operability that is desired for the evolutionary LWR designs. The staff believes that the following aspects of pump and valve testing and inspection are necessary to provide an adequate level of assurance of operability. The following provisions should be applied to all safety related pumps and valves and not limited to ASME Code Class 1, 2, and 3 components.

-Piping design should incorporate provisions for full flow testing (maximum design flow) of pumps, and check valves.

-Designs should incorporate provisions to test motor operated valves under design basis differential pressure.

-Check valve testing should incorporate the use of advanced non-intrusive techniques to address degradation and performance characteristics.

-A program should be established to determine the frequency necessary for disassembly and inspection of pumps and valves to detect unacceptable degradation which cannot be detected through the use of advanced non-intrusive techniques.

The staff has informed EPR1 and the evolutionary ALWR designers of its concerns. No position has yet been expressed by these groups. Imposition of these procedures would exceed past licensing guidance; therefore, the staff recommends that the Commission approve these provisions for evolutionary ALWRs.

- 20 -

REPORT BY THE ADVISORY COMMITTEE ON REACTOR SAFEGUARDS ON SECY-90-016 "EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS" APRIL 26, 1990



UNITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARDS WASHINGTON, D. C. 20055

April 26, 1990

The Honorable Kenneth M. Carr Chairman U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Chairman Carr:

## SUBJECT: EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS

During the 358th, 359th, and 360th meetings of the Advisory Committee on Reactor Safeguards, February 8-10, March 8-10, and April 5-7 and 18-19, 1990, we discussed with representatives of the NRC staff the staff's positions and recommendations concerning the evolutionary light water reactor (ELWR) certification issues contained in SECY-90-016 (Ref. 1). During some of these meetings, we had the benefit of discussions with representatives of the Electric Power Research Institute (EPRI) and the General Electric Company. We also had the benefit of the documents referenced.

We were told by the staff that the positions for which they are seeking Commission approval are described in the underlined portions of the enclosure to SECY-90-016, entitled "Evolutionary ALWR Certification Issues." Unless indicated otherwise, our comments relate to these staff positions. Our comments and recommendations on the staff positions are given below.

I. GENERAL ISSUES

## 1. Evolutionary LWR Public Safety Goals

The NRC staff has concluded that the quantitative goals submitted for Commission consideration in draft SECY-89-102 (Ref. 2) are acceptable for ELWRS. The staff notes that both public safety goals in the EPRI ALWR Requirements Document (Ref. 3) and the ABWR Licensing Review Basis Document (Ref. 4) are considerably more restrictive than the large-release guideline defined in draft SECY-89-102. The staff further notes that additional Commission guidance on quantitative safety goals will assist the staff in its continuing assessment of ELWRS.

We believe, as stated in our previous reports (e.g., ACRS report on Key Licensing Issues Associated With DOE Sponsored Reactor Designs, dated July 20, 1988), that the Commission's Safety Goal Policy is appropriate guidance for regulatory decisions relating to ELWRs, other advanced reactors, and the operating plants. We regard it as not inappropriate that applicants should work to tighter standards when it serves their purposes, but we do not believe it is appropriate that the NEC should require such standards. In its Safety Goal Policy the Commission, in effect, said it would regulate to a level of safety that is adequate, not to the highest level that is possible.

#### 2. Source Term

This issue is dealt with by a proposal to assure that evolutionary designs meet the requirements of 10 CFR 100 (Reactor Site Criteria). The requirements of this regulation include a limit on doses experienced by an individual at the exclusion area boundary, and at the boundary of the low population zone during the course of an accident. In calculating these doses, the instructions in 10 CFR 100 prescribe that the fission products released to the containment must be those which would be expected from accidents which "result in substantial meltdown of the core with subsequent release of appreciable quantities of fission products." For plants currently operating, regulatory guides have delineated specific, but somewhat arbitrary, quantities of fission products that are acceptable to the staff in calculating the leakage from containment and the resultant doses at the specified boundaries.

In contrast, for the ELWRs, the staff proposes to explore the specification of a source term on a case-by-case basis, rather than using the arbitrary source term prescribed in the past. Since the issue of siting of these plants is not yet resolved, and since revisions to 10 CFR 100 are being considered, there may be no alternative to proceeding as the staff proposes, however awkward it may seem.

However, we can make no informed judgment concerning the appropriateness of the procedure until we know more about the criteria to be used in the selection of a source term, and the results of its application.

#### **II. PREVENTATIVE FEATURE ISSUES**

#### 3. Anticipated Transients Without Scram (ATWS)

The staff recommends that the Commission approve the staff's position that diverse scram systems be required for the ELWRs.

It appears to us that a design that can ride out an ATWS without serious damage is feasible for PWRs and is preferable to a scram

## The Honorable Kenneth M. Carr 3 April 26, 1990

-

0

.

system with diverse logic, which has a reliability calculable, at best, with large uncertainty. We recommend that the staff permit demonstration that the consequences of an ATWS are acceptable as an alternative to a diverse scram logic. The uncertainty in such a demonstration is probably considerably less than that in demonstrating that the contribution of an ATWS to risk is made acceptable by installation of a diverse scram logic system.

#### 4 . Mid-Loop Operation

We have been told previously of evidence that events initiated during mid-loop operations may be major contributors to risk in PWRs. However, shutdown operations are generally not accounted for in PRA studies, such as those reported in NUREG-1150 (Ref. 5), so the risks are not well quantified. For the operating plants, this issue has been dealt with through resolution of Generic Issue 99 (Improved Reliability of RHR Capability in PWRs). For the ELWRs, the staff recommends that PWR applicants propose design features to ensure high reliability of the shutdown decay heat removal system.

We agree with the staff's proposal, but recommend that more specific requirements be considered for mid-loop operation:

- Design provisions to help ensure continuity of flow through the core and residual heat removal system with low-liquid levels at the junction of the DHR system suction lines and the RCS
- Provisions to ensure availability of reliable systems for decay heat removal
- Instrumentation for reliable measurements of liquid levels in the reactor vessel and at the junction of the DHR system suction lines and the RCS
- Provisions for maintaining containment closure or for lapid closure of containment openings

## 5. Station Blackout

The Station Blackout Rule (10 CFR 50.63) requires that each lightwater nuclear power plant licensed to operate must be able to withstand for a specified duration, and then recover 'from, a station blackout as defined in 10 CFR 50.2. This rule permits the utilities to submit alternative methods for coping with station blackout. This rule also states that a method based on an alternate ac power source, as defined in 10 CFR 50.2, will constitute an acceptable capability.

For the ELWRs, the staff recommends that the Commission require the

installation of an alternate ac power source as the only basis taken to demonstrate compliance with 10 CFR 50.63. The staff recommends that the alternate ac source have capacity to supply power for one safety train, including one complete set of normal safe shutdown loads, and that it be of diverse design. The alternate ac power supply must be designed to serve any safety train when needed, thereby serving as an additional backup power supply for the Class IE power supplies. The staff has stated that the diversity requirement will not preclude use of diesel generators, even though diesel generators are used for the Class IE emergency power supplies.

Although taken by itself this may seem to be desirable, it has not been demonstrated that it is required to conform to the safety goal. Nevertheless, we endorse the staff's recommendation.

### 6. Fire Protection

The staff concluded that the fire protection issues raised through operating experience and the Individual Plant Examination for External Events (IPEEE) Program (Ref. 6) must be resolved for the ELWRS. To accomplish this, the staff is proposing that the current NRC guidance for fire protection be enhanced as described by the staff during the March 27, 1990, meeting of our Subcommittees on Extreme External Phenomena and Severe Accidents. The enhancements proposed by the staff when combined with the requirements of 10 CFR 50.48 (Fire Protection) without exception and the guidance provided by the Standard Review Plan Section 9.5.1 (Fire Protection Program) should constitute an acceptable basis for prescribing fire protection features for the ELWRS.

The proposed enhancements represent a significant improvement in physical separation requirements and in the need to consider the effects of smoke, heat, and fire suppressant migration into other areas. In particular, redundant train separation is likely to be the most significant feature leading to reduced fire risk. We recommend that the proposed enhancements include separation of environmental control systems.

The fire-risk issues that were examined in the Fire Risk Scoping Study (Ref. 7), however, are not fully addressed in SECY-90-016. They should be.

We agree with the staff's recommendation for resolution of this issue with the above caveats.

## 7. Intersystem LOCA

The staff's position is that designing low-pressure systems to withstand full RCS pressure (to the extent practicable) is an acceptable means for resolving this issue. For those systems that

have not been designed to withstand full RCS pressure, the staff indicates that other measures will be required. We recommend approval of the proposed staff resolution, provided consideration is given to all elements of the low pressure piping system (e.g., instrument lines, pump seals, heat exchanger tubes, and valve bonnets).

#### **III. MITIGATIVE FEATURE ISSUES**

#### Hydrogen Generation and Control 8.

The staff recommends that the ELWR designs provide a system for hydrogen control that can safely accommodate hydrogen generated by the reaction of water with 100% of the fuel cladding surrounding the active fuel. (Note: This is not 100% of the fuel rod cladding, nor does it include other metal in the core which could produce hydrogen if it were heated to a red heat in the presence of steam.) There is substantial uncertainty in establishing the amount of hydrogen that might be formed in a severe accident. We support the staff's recommendation.

The staff also recommends that the system be capable of precluding uniform concentrations of hydrogen greater than 10%. The EPRI ALWR Requirements Document specifies 13%. We are not aware of any experimental or analytical work that demonstrates that the detonation of hydrogen at the 10%, 13%, or some other level could damage the integrity of the containment and essential components. It is our impression that the effect, if any, is schething that experts dealing with gas explosions can calculate with reasonable confidence. We suggest that the staff seek further technical information on possible effects, including stratification, before establishing a limit for the average hydrogen concentration.

#### Core-Concrete Interaction - Ability To Cool Core Debris 9.

The staff proposes that the ELWR designs provide suff cient reactor cavity floor space to enhance debris spreading, and provide for quenching of the debris in the reactor cavity. Quantification of what constitutes sufficient reactor cavity floor space is still an open question, as is the means by which one quenches the core debris. The resolution of this issue will require engineering judgment as many of the physical processes are not fully understood. We agree with the staff's recommendation.

## 10. High-Pressure Core Melt Ejection

To cope with the possible effects of direct containment heating (DCH), the staff concludes ". . . that ALWR design should include a depressurization system and cavity design "eatures to contain ejected core debris."

This is an extremely improbable event, and we see no need to require two modes of coping with the possibility. Either depressurization or cavity design provisions alone should be adequate. Because of possible safety benefits for other events, reliable depressurization is probably the preferred approach.

#### 11. Containment Performance

The staff recommends that a containment performance guideline, expressed as a conditional containment-failure probability (CCFP) of 0.1, be used in evaluation of the Laws Fasigns. As an alternative, the staff proposes a deterministic performance goal that it believes would offer comparable protection.

We have previously recommended (ACRS Comments on An Implementation Plan For The Safety Goal Policy, dated May 13, 1987) such a quantitative guideline for containment performance as a part of the implementation of the Safety Goal Policy. However, this should be regarded as guidance to the NRC staff in its development of requirements for applicants. Merely passing on this guidance to applicants is not enough because the definition of CCFP is too imprecise. The deterministic performance criterion for containment systems suggested by the staff is also difficult to interpret.

We have undertaken an effort (ACRS report on Containment Design Criteria, dated March 15, 1989) to propose containment design criteria for future plants. But, as we said at the beginning of our study, we did not expect that it would directly affect the certification of the ELWR designs. This was, to some extent, because we recognized that our study would take some time to complete, but principally because the ELWR designs are now essentially complete and have been for some time.

We understand that the staff, assisted by the Brookhaven National Laboratory, is developing a regulatory guide that would serve as a basis for review of ELWR containment performance. We believe that the staff proposal will be adequate for ELWR review if it is supported by an appropriate regulatory guide developed on a timely schedule, and if it can be reasonably demonstrated that a containment that meets this guidance has a CCFP of not more than 0.1.

#### 12. ABWR Containment Vent Design

During our April 5-7, 1990 meeting, we heard presentations from the staff and the General Electric Company regarding the staff's proposal that the Commission approve the use of severe accident design features that include a containment overpressure protection system in the ABWR design. We recommend that use of a containment overprotection system be approved subject to the results of the regulatory review.

# The Honorable Kenneth M. Carr 7 April 26, 1990

## 13. Equipment Survivability

The staff recommends that features provided in the ELWR designs that are intended only for severe accident protection (prevention and mitigation) need not be subject to 10 CFR 50.49 (Environmental Qualification Requirements), 10 CFR 50, Appendix A (Redundancy and Diversity Requirements), and 10 CFR 50, Appendix B (Quality Assurance Requirements). However, the staff will require that mitigation features must be designed so there is "reasonable assurance" that they will perform their intended function in the severe accident environment and over the time span for which they are needed. Further, the staff proposes that at least one train of features provided for design basis accident protection, but also relied upon for severe accident protection, must be able to survive severe accident conditions for the time period that is needed to perform its intended function with "high confidence." In addition, the staff proposes to require that severe accident mitigation equipment be capable of being powered from an alternate power supply, as well as from the normal Class IE on-site systems.

To accomplish "reasonable assurance" and "high confidence," the staff will require that severe accident protective features use high quality industrial grade components which will be selected for the service intended and qualified by analysis or tests.

We endorse the staff's position. We note, however, that in this instance the staff's position includes much more than the underlined portions of the enclosure to SECY-90-016.

## IV. NON-SEVERE ACCIDENT ISSUES

# 14. Operating Basis Earthquake (OBE) /Safe Shutdown Earthquake (SSE)

The staff states that it has not yet developed a position on this issue that can be applied generically to all future designs and recommends that the Commission approve a design-specific approach. We have no objection to the staff considering exemptions to the requirement that the OBE be at least one-half the SSE, where this can be justified. We note that this has been done in the past for 14 plants at 9 sites, but in each case using site-specific data. Other bases for justification may have to be provided for un-sited standard plant designs.

In the longer term, we recommend that the staff and the industry attempt to develop a position that can be defined ger rically. One approach worthy of study would be to abandon the use of two earthquake levels for the design of structures, systems, and components. Instead, the design could be based only on the SSE, with appropriate load factors and limit states, and a smaller but pore likely earthquake could be established as a threshold for plant shutdown and inspection.

April 26, 1990

#### Inservice Testing of Pumps and Valves 15.

The staff proposes that certain aspects of the testing and inspection of pumps and valves be enhanced to ensure the necessary level of component operability for the ELWR designs. We endorse the staff's proposal with the following clarification and additions:

- Although not stated explicitly, we were told during the March 7, 1990 meeting of our Subcommittee on Mechanical Components that the staff intends to apply the requirements of Generic Letter 89-10 (Ref. 8) to the ELWR plants as well. We endorse this intention.
- We recommend that the staff's requirement for full-flow . testing capability be extended to other safety-related valves (e.g., MOVs) not just check valves. The requirement for flow testing of MOVs is included in Generic Letter 89-10.
- We recommend that the staff resolve the issue of che valve testing and surveillance requirements for existing LWR plants and indicate how it is to be applied to the ELWRs prior to issuing the FDAs.
- We recommend that the staff be encouraged to entertain proposals from the FDA applicants regarding alternative ways of meeting the in-service testing and surveillance requirements.

Additional comments by ACRS Members Harold W. Lewis and James C. Carroll and ACRS Members William Kerr, David A. Ward, and James C. Carroll are presented below.

Sincerely,

Carlyle Michelson

Carlyle Michelson Chairman

Additional Comments by ACRS Members Harold W. Lewis and James C. Carroll

Apart from one paragraph submerged as part of Item 1, this letter endorses the scattershot approach the staff has taken to the important question of regulation of new reactors. It therefore deserves to be called Camel II, in deference to the Committee's similar letter of January 15, 1987. The differences are that this list has in fact had more careful consideration, and that its elements originated with specific staff proposals. Indeed, in many

cases the genealogy can be traced to industry initiatives, and the staff is simply proposing to make Landatory those things that the industry has previously proposed to do on its own. None of this pays the slightest attention to the Commission's Safety Goal Policy, nor is there any hint of an effort to seize this opportunity to move the regulatory process in the direction of coherence and consistency. This is a pudding without a theme.

Let us then try to provide some perspective, since the Committee has chosen not to do so.

The Committee has often commented on the central role of the Safety Goals in providing a focus and objective for the body of regulation. Since this list sets the tone for the licensing of the next generation of light-water reactors, it is particularly important that its relation to Commission policies, especially the Safety Goal Policy, be clear. At the risk of repetition, we, and we believe the Committee, have never urged that specific regulatory decisions (such as these) be judged individually in the context of the Safety Goals, but only that the body of regulation be judged in that light. Individual decisions must still be made deterministically, with expertise and good judgment, but as part of a coherent overall body of regulation. Still, fifteen items come close to being a "body", and it is informative to see the role of the Safety Goals in the formulation of the staff recommendations. The Safety Goal policy, and other commission policies, are supposed to provide the glue that binds the whole structure together.

In effect, the staff says that it has proposed to the Commission a set of new safety goals (SECY-89-102), the Commission has not acted on them, either way, and therefore the staff will use them as if the Commission had approve. While we sympathize with the staff predicament, we think that ... entirely inappropriate. The staff proposals include such things as a core-damage probability of 1E-5 per reactor-year, a "large accident" probability of 1E-6 per reactor year (with a bizarre definition of large accident), and a so-called conditional containment-failure probability. Not one of these has been approved by the Commission, yet the staff has used them in formulating its proposed policies on these items. It has rationalized this usurpation of power by asking for Commission action on SECY-89-102, and by stating that its own safety goals are "consistent" with those of the Commission. Of course any set of goals more stringent than yours will be consistent with your own, and acceptance of this argument will mean that the staff can regulate beyond your policies, more or less at will. That is precisely the situation your original goals were intended to foreclose. The Committee has often recommended that your Safety Goals be used as a final statement of "how safe is safe enough", not as a rigid minimum level of safety, beyond which the sky is the limit. Of course the industry may well have good reason to go further, but that is another matter.

EPRI Evolutionary Plant SER 18.A-39
The Honorable Kenneth M. Carr 10

In addition, as your own OGC has pointed out in SECY-90-016, this has the potential to open a Pandora's box, in which each party to a licensing proceeding may be able to claim the rights the staff claims -- to insist on improvements beyond the rules. You will have to face this problem at some time, and the sooner the better.

We do not wish to understate the difficulty involved in translating a safety-goal policy into a workable body of regulation. The Committee has written you of its own recommendations for an organized approach to that problem, but we believe it can and Nuclear safety is not helped by letting that should do more. problem fester -- the fact that it is difficult is no excuse for inattention. It is too much to expect regulation to be coherent and rational in the absence of an objective for that regulation.

We do think it was useful for the Committee to respond to your specific request for technical help on the fifteen questions posed, but you should recognize that this was done in one absence of a measuring rod. Each item was therefore judged on its own, and the Committee has turned its back on the opportunity to respond in a structured and coherent way. Any one of these items might have come out ifferently if it had been measured against an underlying rationale. In our view, the Committee has forfeited a chance to be of real service to both you and the public.

# Additional Comments by ACRS Members William Kerr, David A, Ward, and James C. Carroll

By the "rulemaking" approach to design certification the Commission has sidestepped the development of revisions to regulations that would reflect knowledge gained from experience and research over the last ten or more years. As a result, important new requirements are being imposed on applicants through a variety of staff actions and reactions. This is a loosely controlled process in which major policy decisions are made without an appropriate intensity of review. Contributing to the lack of discipline is what we believe to be a serious ambiguity in the Commission's policy on advanced reactors. The Commission has said it expects future reactors to be safer. But, whether this is a mandate or simply an expectation that a maturing industry will produce safer plants is not clear. The staff has interpreted it as a mandate and has translated this into an unauthorized extension of the safety goals. This is despite the statement in NUREG-1226, "Development and Utilization of the NRC Policy Statement on the Regulation of Advanced Nuclear Power Plants," published June 1988 (p. 4-1) that "the Commission expects but does not require enhanced safety margins other than those that may be required by the Safety Goal Policy." The Commission should not indefinitely postpone the development of a modern set of regulations. Only in this way will a proper balance be struck between adequate protection of the

The Honorable Kenneth M. Carr 11

....

public health and safety and the advantages to the public that can come from efficient development of the nuclear power option.

# References:

- SECY-90-016, memorandum dated January 12, 1990, from J. 1. Taylor, Executive Director for Operations, NRC, to the Commissioners, Subject: Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements
- Draft SECY-89-102, memorandum dated March 30, 1989, from V. 2. Stello, Jr., Executive Director for Operations, NRC, to the Commissioners, Subject: Implementation of Safety Goal Policy
- Electric Power Research Institute (EPRI), Advanced Light Water 3. Reactor Requirements Document (Chapters 1 through 13), issued December 1987
- General Electric Company, Advanced Boiling Water Reactor 4. Licensing Review Basis Document, issued August 1987
- U. S. Nuclear Regulatory Commission, NUREG-1150, "Severe 5. Accident Risks: An Assessment for Five U. S. Nuclear Power Plants," Volumes 1 and 2, dated June 1989
- Memorandum dated March 8, 1990 from W. Minners, Office of 6. Nuclear Regulatory Research, NRC, to R. Fraley, ACRS, Subject: Proposed Generic Letter on Individual Plant Examination for Severe Accident Vulnerabilities Due to External Events (IPEEE) and Supporting Documents (Predecisional)
- Sandia National Laboratories, "Fire Risk Scoping Study: 7. Investigation of Nuclear Power Plant Fire Risk, Including Previously Unaddressed Issues," NUREG/CR-5088, published January 1989
- Generic Letter 89-10, "Safety-Related Motor-Operated Valve 8. Testing and Surveillance," issued on June 28, 1989 to licensees for all power reactors, BWRs, PWRs, and vendors in addition to General Codes applicable to generic letters.

EPRI Evolutionary Plant SER 18.A-41

STAFF RESPONSE TO ACRS CONCLUSIONS REGARDING EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES (REGARDING SECY-90-016) APRIL 27, 1990

-

.

۲

APR 27 1990

MEMORANDUM FOR: Chairman Carr Commissioner Roberts Commissioner Rogers Commissioner Curtiss Commissioner Remick

FROM: James M. Taylor Executive Director for Operations

SUBJECT: STAFF RESPONSE TO ACRS CONCLUSIONS REGARDING EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES

The ACRS sent a letter to the Commission addressing the subject issues. In general, the ACRS supported the staff positions, with additional comments and recommendations. The staff has evaluated these ACRS comments and recommendations and its evaluation is provided in the enclosure.

> Original Signed By: James M. Taylor

James M. Taylor Executive Director for Operations

Enclosure: As stated

Enclosure A

# EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQU'REMENTS

# Evolutionary LWR Public Safety Goals

The Advisory Committee on Reactor Safeguards (ACRS) stated, "We regard it as not inappropriate that applicants should work to tighter standards when it serves their purposes, but we do not believe it is appropriate that the NRC should require such standards."

The staff is accepting EPRI's proposed core damage frequency goal and their large release safety goal. We are not proposing to push them for more conservative safety goals, but the staff expects ELWR applicants to demonstrate that they have taken reasonable steps to achieve their stated goals.

# 2. Source Term

The ACRS stated that, "Since the issue of siting of these plants is not yet resolved, and since revisions to 10 CFR 100 are being considered, there may be no alternative to proceeding as the staff proposes, however awkward it may seem. However, we can make no informed judgment concerning the appropriateness of the procedure until we know more about the criteria to be used in the selection of a source term, and the results of its application."

The staff will keep the ACRS informed of our source term approach for ALWRs as we develop the details of 'ts application during our reviews.

# 3. Anticipated Transients Without Scram (ATWS)

The ACRS stated that "It appears to us that a design that can ride out an ATWS without serious damage is feasible for PWRs and is preferable to a scram system with diverse logic, which has a reliability calculable, at best, with large uncertainty. We recommend that the staff permit demonstration that the consequences of an ATWS are acceptable as an alternative to a diverse scram logic."

The staff agrees with the ACRS comment that resolution of ATWS through a design that can ride out an ATWS is preferable to reliance on a diverse scram logic. This option was offered in the staff Draft Safety Evaluation Report for Chapter 5 of the EPRI Requirements Document and has been proposed by EPRI in Chapter 10. Whether the currently proposed diverse systems could be deleted would be based on the adequacy of this design analysis to demonstrate that an ATWS would not lead to serious damage.

# a. Mid-Loop Operation

The ACRS stated that, "We agree with the staff's proposal, but recommend that more specific requirements be considered for mid-loop operation:

- Design provisions to help ensure continuity of flow through the core and residual heat removal system with low liquid levels at the junction of the DHR system suction lines and the RCS.
- Provisions to ensure availability of reliable systems for decay heat removal.
- Instrumentation for reliable measurements of liquid levels in the reactor vessel and at the junction of the DHR system suction lines and the RCS.
- Provisions for maintaining containment closure or for rapid closure of containment openings.

the staff agrees with the ACRS comments. In fact, past staff actions on this issue for operating reactors focused on most of these items. The staff plans to ensure that these factors are addressed in the ELWR designs.

## 5. Station Blackout

The ACRS stated that the staff recommended that an alternate diverse ac power source capable of powering one complete safety train of safe shutdown loads be installed in the ELWRs as the means used to meet the requirements of the Station Blackout Rule (10 CFR 50.63). The ACRS noted that the rule permits alternative methods for coping with station blackout, and cited 10 CFR 50.2 as defining the acceptable capability for the alternate AC source.

The ACRS also stated, "Although taken by itself, this may seem to be desirable, it has not been demonstrated that it is required to conform to the safety goal. Nevertheless, we endorse the staff's recommendation."

The alternate ac power source was identified as the preferred approach for meeting the Station Blackout Rule only in ELWRs because is offers several advantages over other approaches. It has the advantage of powering a complete complement of safe shutdown equipment, it can be utilized in other events in addition to station blackout (such as during mid-loop operation when one of the normal Class IE standby power sources is out for maintenance or otherwise unavailable), it is essentially not time limited while providing power during a station blackout, and it provides for a uniform hardware based approach requiring little or no coping analyses and fewer specialized operating procedures.

# 6. Fire Protection

The ACRS supports the enhanced fire protection requirements recommended by the staff. Additionally, they recommend further enhancements including:

- Required separation of Environmental Control System (HVAC systems) so that independent fire areas containing redundant safe shutdown trains are not serviced by the same ventilation train and.
- That the 6 items identified in the fire risk scoping study br fully addressed for ELWRs.

The staff proposed requirement to consider the effects of smoke, heat, and fire suppressant migration may result in separate HVAC systems. However, other options may be available to the designer.

In reference to the ACRS recommendation that the 6 fire risk scoping study issues be fully addressed for ELWRS, it is the staff opinion that this is being done through the proposed enhancements and current regulatory efforts as indicated below.

#### 1. Improved Analytical Fire Codes

The enhanced guidance eliminates redundant safety systems in the same fire area so it is not necessary to have a code to show that the redundant train in the same area is protected. This issue is most for ELWRs with the enhanced criteria.

#### 2. Control System Interactions

Current regulations require an independent safe shutdown capability in the event of loss of the control room. This is reiterated in the enhanced guidance for ELWRs.

#### 3. "otal Environment Equipment Survival

GDC-3 requires fire protection systems to be designed not to impair safe shutdown systems. Although in some instances fire system actuation has disabled a safety system, in no case have redundant systems been disabled.

# 4. Fire Barrier Qualification, and

#### 5. Manual Fire Fighting Effectiveness

Both of these issues (4 and 5) are addressed in the current regulations. The ELWR designs must include information to demonstrate that appropriate maintenance exists for fire barriers and training exists for fire brigades.

# 6. Seismic/Fire Interactions

The fire risk scoping study concluded that this could be addressed by a walkdown. The staff will evaluate this issue as part of the PRA review and expects that a walkdown will be required as part of ELWR design certification.

# 7. Intersystem LOCA

The ACRS stated that, "We recommend approval of the staff resolution, provided consideration is given to all elements of the low pressure system, (e.g., instrument lines, pump seals, heat exchanger tubes, and valve bonnets.)"

The staff agrees with the ACRS comment and will consider all elements of the low pressure system to ensure that the system, to the extent practicable, will withstand full RCS pressure.

# B. Hydrogen Generation and Control

The ACRS stated that the, "Staff recommends that the ELWR designs provide for hydrogen control that can safely accommodate hydrogen generated by the reaction of water with 100% of the fuel cladding surrounding the active fuel." The ACRS stated that, "We support the staff's recommendation."

The ACRS also stated, "We are not aware of any experimental or analytical work that demonstrates that the detonation of hydrogen at the 10%, 13%, or some other level could damage the integrity of containment and essential components. It is our impression that the effect, if any, is something that experts dealing with gas explosions can calculate with reasonable confidence. We suggest that the staff seek further technical information on possible effects, including stratification, before establishing a limit for the average concentration."

This staff recommendation is consistent with the regirements set forth in 10 CFR 50.34(f). Although this section of the regulations was originally written for a select group of plants, 10 CFR Part 52 established these requirements to be a minimum standard for future plants. In this context, the requirement of 100% has become a surrogate to address both in-vessel and ex-vessel hydrogen production. It was not intended to be mechanistically based, but rather a design goal that appears to be reasonable in view of all the analyses performed to date. The staff has not uncovered any new evidence in this regard that would cause them to alter their view. While it is true that there are no analyses that demonstrate that detonation loads are a threat to the containment or components within the containment, the staff believes that such an endeavor would be extremely complex and coupled with large degrees of uncertainty. The question of wave reflection and whether they could possibly reinforce within the containment is but one unknown affect. Therefore, the staff believes a prudent course of action is to assure that the likelihood of both global and local detonations is minimized.

To this end the staff has reviewed the available detonation test results and concludes that 10% represents a reasonable limit to assure non-detonable mixtures. It is felt that the true lower limit lies somewhere between 10% and 13%. Therefore, establishing a 10% criteria provides a degree of margin which is to address the issue of local detonations and the potential for stratification resulting in concentration gradients within the containment.

# 9. Core-Concrete Interaction--Ability to Cool Core Debris

The SSE agreed with the staff's recommendation that each ELWR design provide sufficient reactor cavity floor space to enhance debris spreading and provide for quenching of the debris in the reactor cavity. The staff had indicated that considerable uncertainty remains on debris coolability and that they would continue to evaluate the issue as more experimental data and information became available. ACRS concurred in this evaluation and stated that "...resolution of this issue will require engineering judgment as many of the physical processes are not fully understood."

Experiments to assess coolability of water covered debris are underway at Argonne National Laboratory under joint EPRI/Industry/NRC support. The results of these experiments are expected to be available within one year.

# 10. High-Pressure Core Melt Ejection

The ACRS stated, "This is an extremely improbable event, and we see no need to require two modes of coping with the possibility. Either depressurization or cavity design provisions alone should be adequate. Because of possible safety benefits for other events, reliable depressurization is probably the preferred approach."

High pressure melt scenarios are dominant for some designs. The staff proposal is to provide a design concept with a degree of mitigation along with a certain amount of prevention. The RCS depressurization capability retains a degree of uncertainty. Such questions as the rate of depressurization and the cut-off pressure need to be addressed. As a result, we believe that a design can be developed with little or no added expense that can be effective in potentially eliminating the direct path to the containment. Since there are no major down sides, it appears to make sense to pursue this parallel approach of both depressurization and cavity design.

# 11. Containment Performance

The ACRS stated that, "We believe that the staff proposal will be adequate for ELWR review if it is supported by an appropriate regulatory guide developed on a timely schedule, and if it can be reasonably demonstrated that a containment that meets this guidance has a CCFP of not more than 0.1."

The staff believes that a containment performance guideline, either a conditional containment failure probability of 0.1 or r deterministic performance goal that offers comparable protection, is necessary to give the ALWR designers guidance on what is acceptable for containment performance under severe accident conditions. The alternative is to have a containment designed for loss-of-coolant accident (LOCA) and other design basis accident (DBA) conditions and then to evaluate its severe accident failure probability after it is designed.

The staff will keep the ACRS informed of our development of a regulatory guide in this area and of our progress in evaluating ALWR containments using the containment performance guideline as our review progresses.

#### 12. ABWR Containment Vent Design

The ACRS stated that "We recommend that use of a containment overprotection system be approved subject to the results of the regulatory review."

The staff's review of the vent system design is u derway. The purpose of this review is to insure that any downside of ve ting is minimized and that capability is provided to maintain control over the venting process.

# 13. Equipment Survivability

The ACRS states that, "We endorse the staff's position. We note, however, that in this instance the staff's position includes much more than the underlined portions of the enclosure to SECY-90-016."

The ACRS supports the staff position that equipment for severe accident mitigation does not need to meet the same quality standards as equipment for design basis accidents. Instead of 10 CFR 50.49, 10 CFR 50, Appendix A, and 10 CFR Appendix B, severe accident equipment need only be high quality industrial grade, selected and oualified by analysis or test for the intended service.

# 14. Operating Basis Earthquake (OBE)/Sufe Shutdown Earthquake (SSE)

ACRS has no objection to the staff considering exemptions to the requirement that OBE be at least one half the SSE, where this can be justified. ACRS suggests: 1) other bases for justification may have to be developed for unsited standard plant designs, 2) design be based only on the SSE with appropriate load factors and limit states, and 3) a smaller but more likely earthquake be established as a threshold for plant shutdown and inspection.

The staff accepts the ACRS suggestion which is very close to the design specific approach proposed by the staff.

# 15. Inservice Testing of Pumps and Valves

The ACRS endorses the staff's proposed approach while emphasizing the need to apply the requirements of Generic Letter 89-10 to ELWR plants. The ACRS also recommends that the staff resolve the check valve testing and surveillance issue and indicate how it is to be applied to ELWRs including consideration of industry proposed alternative ways of meeting inservice and surveillance requirements.

The staff agrees with the ACRS recommendations and will keep the ACRS informed of any new developments in this area.

STAFF REQUIREMENTS MEMORANDUM "SECY-90-016 - EVOLUTIONARY LIGHT WATER REACTOR CERTIFICATION ISSUES AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS" JUNE 26, 1990

141

.



UNITED STATES NUCLEAR REGULATORY COMMISSION WAEHINGTON, D.C. 20555

June 26, 1990

for Operations/

DEFICE OF THE SECRETARY

MEMORANDUM FOR:

FROM:

SUBJECT:

SECY-90-16 - EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES AND THEIR RELATIONSHIPS TO CURRENT REGULATORY REQUIREMENTS

James M. Taylor, Executive Director

This is to advise you that the Commission as detailed below has approved in part and disapproved in part the staff's recommendations in SECY-90-16.

- I. General Issues.
  - A. ALWR Public Safety Goal,

The Commission (with Chairman Carr and Commissioners Roberts, Curtiss and Remick agreeing) has disapproved the use of 10° per year of reactor operation as a core damage frequency for advanced designs. As noted in the SRM on SECY-89-102 (dated June 15,41990), the Commission supports the use of 10° per year of reactor operation as a core damage frequency goal. Although the Commission strongly supports the use of the information and experience gained from the current generation of reactors as a basis for improving the safety performance of new designs, the NRC should not adopt industry objectives as a basis for establishing new requirements. However, if the staff in applying the criteria of 10 CFR Part 52 (and in view of the uncertainties associated with PRA's) concludes that additional requirements are needed, based on our experiences with prior designs, in order to provide

NOTE: THIS SRM AND THE SUBJECT SECY PAPER WILL BE MADE PUBLICLY AVAILABLE 10 WORKING DAYS AFTER ISSUANCE OF THE SRM. assurance that future designs will meet the Safety Goal Policy Statement, then the staff should provide those additional requirements to the Commission for consideration as they are identified.

Commissioner Rogers approved the staff's use of 10"5 as an expected design target for ELWR designers and endorsed a requirement that applicants be able to demonstrate that they have taken reasonable steps to reach these targets. However, he does not endorse those goals as an absolute requirement for approval of any specific design.

consistent with the Commission's decision on SECY-89-102, the Commission approved the overall mean frequency of a large release of radioactive material to the environment from a reactor accident as less than one in one million per year of reactor operation. The Commission has not agreed on a definition of a large release and has requested a paper from the staff (See SRM from SECY-89-102). (RES) (Suspense: 9/28/90) 9000136

#### Source Term. в.

The Commission (with all Commissioners agreeing) has approved the staff's approach to the source term with the addition of the following element:

On an expedited basis, incorporate appropriate changes to regulations, regulatory practices, and the review process resulting from source term research. (RES/NRR) - As appropriate - Pending Source Term Study Results. SECY paper due imminently. 5-01208

#### II. Freventative Feature Issues.

#### λ. ATWS.

The Commission (with all Commissioners agreeing) has approved the staff position. However, if the applicant can demonstrate that the consequences of an ATWS are acceptable the staff should accept the demonstration as an alternative to the diverse scram system. Commissioner Curtiss further believes that the staff should retain the flexibility to accept designs with non-diverse scram logic in those instances where it is demonstrated to the staff's satisfaction that the reliability of the scram function is such that the risk from ATWS is insignificant. (NRR)

#### Mid-Loop Operation. B .

The Commission (with all Commissioners agreeing) has approved the staff's proposed position, with the ACRS recommendation of April 26, 1990, that four additional specific requirements be considered for mid-loop operation. (NRR)

# C. Station Blackout.

The Commission (with all Commissioners agreeing) has approved the staff's position that the evolutionary ALWR's have an alternate ac power source of diverse design capable of powering at least one complete set of normal shutdown loads. The staff should provide a clear definition of "diversity" so as to provide guidance on whether it means different types, different manufacturers, different models, etc. Commissioner Curtiss noted that, in his view, the clarification should focus on limiting common mode failure potential but need not go so far as to require completely different generator driver technologies (e.g. should not necessarily require both diesel and gas turbine driven generators). (NRR)

# D. Fire Protection.

The Commission (with all Commissioners agreeing) has approved the staff's position on fire protection as presented in SECY-90-16 and supplemented by the staff's April 27, 1990, response to the ACRS comments. (NRR)

### E. Intersystem LOCA.

The Commission (with all Commissioners agreeing) has approved the staff's position on intersystem LOCA provided that, as recommended by the ACRS, all elements of the low pressure system are considered (e.g instrument lines, pump seals, heat exchanger tubes, and valve bonnets.) (NRR)

#### III. Mitigative Feature Issues.

#### A. Hydrogen Generation and Control.

The Commission (with all Commissioners agreeing) has approved the staff's position that the requirements of 10 CFR 50.34(f)(2)(ix) should remain unchanged for evolutionary plants. The staff should seek additional technical information, as suggested by the ACRS, and if reconsideration is warranted the Commission should be advised. (RES/NRR)

B. Core-Concrete Int raction -- Ability to Cool Core Debris.

The Commission (with all Commissioners agreeing) has approved the staff's position. (NRR/RES)

C. High Pressure Core Melt Election,

The Commission (with all Commissioners agreeing) has approved the staff's position that the ELWR designs include a depressurization system and cavity design to contain core debris. The cavity design, as a

EPRI Evolutionary Plant SER 18.A-59

mitigating feature, should not unduly interfere with operations including refueling, maintenance, or surveillance activities. (NRR/RES)

## D. Containment Performance.

The Commission (with all Commissioners agreeing) has approved, consistent with SECY-89-102, the use of a 0.1 CCFP as a basis for establishing regulatory guidance for the ELWRS. This objective should not be imposed as a requirement in and of itself. The use of the CCFP should not discourage accident prevention and the staff should review suitable alternative, deterministicallyestablished, containment performance objectives providing comparable mitigation capability if submitted by applicants. Any such alternatives should be submitted to the Commission following staff review. (NRR/RES)

## E. ABWR Containment Vent Design.

The Commission (with all Commissioners agreeing) has approved the staff's recommended use of the containment overpressure protection system on the ABWR, subject to the results of the comprehensive regulatory review which should fully weigh the potential "downside" risks with the mitigation benefits of the system. Staff should ensure that full capability to maintain control over the venting process is provided. (NRR)

F. Equipment Survivability.

The Commission (with all Commissioners agreeing) has approved the staff's position. (NRR)

# IV. Non-Severe Accident Issue.

- A. <u>Operating Basis Earthquake (OBE)/Safe Shutdown</u> <u>Earthquake (SSE).</u> The Commission (with all Commissioners agreeing) has approved the staff's position. (NRR)
- B. Inservice Testing of Pumps and Valves.

The Commission (with all Commissioners agreeing) has approved the staff's position as supplemented in their April 27, 1990, response to the ACRS comments. The Commission notes that due consideration should be given to the practicality of designing testing capability, particularly for large pumps and valves. (NRR/RES)

The Commission also agreed that in those cases where the staff proposed requirements depart from current regulations, consideration should be given to incorporating these requirements into the regulations. (See SRM dated May 27, 1990, M90053A). Finally, the staff is encouraged to strive to sustain the level of attention and resources that have been devoted recently to the review process for the EPRI requirements document. The recent comments of the EPRI representatives at the June 4, 1990 Commission briefing suggest that such a commitment, if sustained, can be most beneficial in assisting EPRI and the NRC staff in our respective efforts to reach a common understanding on the key technical issues. (NRR)

cc: Chairman Carr Commissioner Roberts Commissioner Rogers Commissioner Curtiss Commissioner Remick OGC ACRS IG ASLBP ASLAP ANNEX B

DOCUMENTS RELATED TO SECY-91-078 "CHAPTER 11 OF THE ELECTRIC POWER RESEARCH INSTITUTE'S (EPRI's) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES" MARCH 25, 1991

# CONTENTS

	Page
SECY-91-078, dated March 25, 1991	1B.B-3
Report by the Advisory Committee on Reactor Safeguards on SECY-91-078, dated April 23, 1990	1B.B-11
Staff Response to ACRS Conclusions Regarding SECY-91-078, dated May 15, 1991	1B.B-17
Staff Requirements Memorandum, "SECY-91-078 - Chapter 11 of the Electric Power Research Institute's (EPRI's) Requirements Document and Additional Evolutionary Light Water Reactor (LWR) Certification Issues," dated August 15, 1991	18.8-23



SECY-91-078

March 25, 1991

FOF:

The Commissioners

From: James M. Taylor Executive Director for Operations

SUBJECT: CHAPTER 11 OF THE ELECTRIC POWER RESEARCH INSTITUTE'S (EPRI'S) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES

Purpose:

To inform the Commission of the staff's intent to issue the draft safety evaluation report (DSER) for Chapter 11 of the EPRI Requirements Document. Additionally to request Commission approval of staff recommendations concerning additional proposed changes to regulatory practice for the evolutionary advanced light water reactors (ALWRs).

Background:

In the staff requirements memorandum (SRM) of December 15, 1989 pertaining to SECY-89-334, "Recommended Priorities for Review of Standard Plant Design," the Commission provided the following guidance to the staff:

> The SERs on the EPRI Requirements Document for both the evolutionary and the passive plant designs should be submitted to the ACRS for review and to the Commission for information and for review and approval of policy issues for which the Commission has not previously decided.

Further, in the SRM of June 22, 1990 pertaining to SECY-90-146, "Process, Schedule, and Resources For the Review of Evolutionary and Passive Advanced Light Water Reactors," the Commission directed the staff to follow the process presented in SECY-90-065, "Evolutionary and Passive Advanced Light Water Reactor Resources and Schedules." The staff included

CONTACT: J. H. Wilson, NRR 2-1325

NOTE: TO BE MADE PUBLICLY AVAILABLE WHEN THE FINAL SRM IS MADE AVAILABLE.

C. L. Miller, NRR 2-1118

1B.B-3

the aforementioned guidance to SECY-89-334 as an element of the process described in SECY-90-065. In SECY-90-401, "Draft SER for Chapters 6, 7, 8, 9, 12, and 13 of EPRI's ALWR Documents for Evolutionary Reactor Plant Designs," the staff. following this guidance, informed the Commission that it would issue DSERs for Chapters 6, 7, 8, 9, 12, and 13 of the EPRI Requirements Document for evolutionary reactor plant designs. The staff issued these documents on January 15, 1991. This paper provides the DSER for Chapter 11, "Electric Power Systems." The remaining DSERs for the evolutionary design criteria, Appendix A to Chapter 1 and Chapter 10, are currently in preparation by the staff.

Discussion:

Operating experience and a number of studies (e.g., probabilistic risk assessments (PRAs)) have identified a number of issues significant to reactor safety. In addition, in SECY-90-016, "Evolutionary Light Water Reactor Certification Issues and Their Relationship to Current Regulatory Requirements," the staff identified several policy issues that apply to future evolutionary ALWR designs and for which the Commission provides guidance in its SRM of June 26, 1990. The staff, in its continued review of the EPRI ALWR Pequirements Document, has identified the following two additional issues:

 alternate source of power for non-safety loads
connection of safety bus offsite power sources through non-safety buses

Enclosure 1 contains a detailed discussion of each of these issues and addresses those instances in which the staff positions differ from current regulatory requirements or in which the staff is substantially supplementing or revising interpretive guidance applied to currently licensed light water reactors (LWRs). In this enclosure, the staff also discusses the nature of the current regulatory requirement or interpretation, the positions of the ALWR vendors and of EPRI, the departure that the staff is proposing, and the basis for the proposed departure. To aid in identifying the staff's positions, the staff has underlined its positions and has cross-referenced them with the sections in the Chapter 11 DSER where they are discussed.

The staff developed the recommendations identified in this paper after (1) reviewing current generation reactor designs and evolutionary ALWRs, (2) considering operating experience, and (3) evaluating the results of the PRAs of LWRs. In addition, these positions are consistent with current design practices at recently licensed operating reactors.

To follow the process outlined in SECY-90-065, the staff will need to identify policy issues and bring them to the Commission for guidance before completing and distributing the DSERs.

EPRI Evolutionary Plant SER

1B.B-4

However, the staff identified the issues discussed in this paper as it developed the DSER on Chapter 11 of the EPRI ALWR Requirements Document. Accordingly, the staff has enclosed the completed DSER (Enclosure 2) to provide the Commission additional information regarding these matters and to put the identified issues into their proper technical context. Additionally the staff believes that it would be beneficial to provide EPRI with the DSER in parallel with Commission review. The DSER would indicate that the previously identified policy issues are before the Commission for consideration. Distribution of the documents would expedite the review schedule by providing EPRI with a listing of staff identified open issues. Resolution of these issues will be addressed in the final SER on the EPRI Requirements Document.

Conclusions:

The staff believes its conclusions and recommendations regarding these matters are in keeping with the Commission's pulicy expectation that future designs for nuclear plants will achieve a higher standard of safety performance.

The staff requests the Commission's approval of, or elternate guidance on, the proposed resolution of these issues in order to continue to review Chapter 11 of EPRI's ALWR Requirements Document for evolutionary plants and to perform the design certification of General Electric's ABWR, and Combustion Engineering's System 80+ designs.

By permitting the issuance of the DSER for Chapter 11 to EPRI, the Commission could expedite the review schedule while it considers these policy issues. In the DSER, the staff states that these policy issues are before the Commission for consideration. The staff will provide a regulatory departure analysis, based on Enclosure 1 of this paper, as Appendix C to the DSER on Chapter 11.

The staff will promptly inform the Commission during its reviews if it determines that additional enhancements to existing requirements, beyond those already identified, are necessary for evolutionary ALWR designs.

<u>Coordination:</u> The Office of General Counsel has reviewed this paper and has no legal objection. This paper is being forwarded to the ACRS for their review and comments.

Recommandations: That the Commission

- (1) Approve the staff positions detailed in Enclosure 1
- (2) Note that if the staff identifies other policy issues, the staff will inform the Commission of its positions in a timely manner

The Commissioners

- \* 4 \*
- (3) Note that absent alternative Commission guidance, the staff will frue the enclosed DSER on Chapter 11 of the EPRI ALWR R, irements Document for evolutionary plant designs 10 is king days after the date of this paper. The DSER will identify the two instances in which the staff is proposing to depart from current regulatory requirements and will state that the Commission is reviewing the basis for the approach that the staff is proposing and, accordingly, may determine that such issues involve policy questions that the Commission may wish to consider.

ecutive Director for Operations

Enclosures: 1. Policy Issues Analysis and Recommendations

 Draft Safety Evaluation Report on Chapter 11

Commissioners' comments or consent should be provided directly to the Office of the Secretary by COB Tuesday, April 9, 1991.

Commission Staff Office comments, if any, should be submitted to the Commissioners <u>NLT Tuesday, April 2, 1991</u>, with an information copy to the Office of the Secretary. If the paper is of such a nature that it requires additional review and comment, the Commissioners and the Secretariat should be apprised of when comments may be expected.

DISTRIBUTION: Commissioners OGC IG GPA EDO ACRS ASLBP ASLAP SECY

EPRI Evolutionary Plant SER

1B.B-6

ENCLOSURE 1

# POLICY ISSUES ANALYSIS AND RECOMMENDATIONS

# 1. Alternate Source of Power for Non-Safety Loads

General Design Criterion (GDC) 17, "Electric Power Systems," requires that an onsite electric power system and the offsite electric power system be provided to permit functioning of structures, systems, and components important to safety. The offsite electric power system must have two physically independent circuits from the transmission network to the onsite electric distribution system.

Although the NRC has not established regulatory requirements on the number of power sources to the following non-safety related loads, the licensees for almost ell nuclear power plants in the United States have provided two power sources to nonsafety-related loads such as reactor coolant pumps, reactor recirculation pumps, main feedwater pumps, condensate pumps, and circulating water pumps. During unit operation, a fast transfer of the nonsafety loads is usually provided to the startup transformer when imminent loss of the unit auxiliary transformer is sensed, such as following a main generator trip or a fault of the unit auxiliary transformer circuit. This process maintains power to the identified nonsafety loads and allows the plant to be shut down under these circumstances without a loss of normal feedwater systems or of forced circulation to the reactor coolant system.

The Electric Power Research Institute (EPRI) has specified requirements in the Requirements Document is only provide for a single source of power from the unit auxiliary transformer to nonsafety loads such as reactor coolant pumps, reactor recirculation pumps, main feedwater pumps, condensate pumps, and circulating water pumps. This one power source provides power to these loads during power operation, startup, and shutdown. A generator circuit breaker will isolate the main generator from the unit auxiliary transformer circuit during startup and shutdown when the main generator is unavailable. In existing plant designs, the main generator circuit breaker/unit auxiliary transformer configuration provide this isolation. However, an alternate source of power to this group of nonsafety loads is also provided at recently licensed operating plants, but not in the criteria of the EPRI Require ints Document.

The General Electric (GE), Combustion Engineering (CE), and Westinghouse standard plant designs all use the main circuit breaker/unit auxiliary transformer configuration as the primary power source to the subject nonsafety loads. However, the Westinghouse SP/90 design also provides an alternate source of power to the nonsafety loads, and the GE ABWR design provides an alternate source of power to a portion of the nonsafety loads (one of four mair nonsafety buses). The CE System 80+ design uses the EPRI approach (it does not provide for an alternate power source).

An additional source of power would significantly reduce the number of plant trips that involve a loss of power to the nonsafety loads and require that the plant be shut down under natural circulation. Such an additional source of power would improve plant safety, because these events continue to be identified as more severe than the turbine-trip-only event in standard plant safety analysis reports. The staff concludes that EPRI should enhance the ALWR design criteria in this area because they are less conservative than those that have been provided in existing, recently licensed plant designs. Therefore, the staff's position is that an evolutionary ALWR design should include an alternate power source to the nonsafety loads unless the design can demonstrate that the design margins in the evolutionary ALWR will result in transients for a loss of nonsafety power event that are no more severe than those associated with the turbine-trip-only event in current existing plant designs.

The staff addressed this issue in Section 4.2.1 of the Chapter 11 DSER (Enclosure 2 of this paper).

2. Connection of Safety Bus Offsite Power Sources Through Nonsafety Buses

GDC 17 requires that an onsite electric power system and offsite electric power system be provided to permit functioning of structures, systems, and components important to safety. The offsite electric power system must have two physically independent circuits from the transmission networks to the onsite electric distribution system. Although GDC 17 specifies that two offsite power circuits are required, and although it provides further additional design criteria on these circuits, it does not specify whether the circuits should directly connect the safety buses to the offsite power sources or whether this connection could be made through intervening nonsafety buses. The Institute of Electrical and Electronic Engineers (IEEE) Standard 308-1974, which is endorsed by Regulatory Guide 1.32, Rev. 2, "Criteria for Safety-Related Electric Power Systems for Nuclear Power Plants," provides a figure that shows the Class 1E safety buses are directly connected to an offsite power transformer (startup transformer). However, the figure is for illustration only, and the Standard does not require the direct connection. IEEE Standard 765-1983 allows the Class IE safety buses to be connected through nonsafety buses to the offsite transformers, but it also states that direct connection of the two circuits to each redundant Class 1E bus may further improve availability. However, the staff has not endorsed IEEE Standard 765, with a regulatory guide. Therefore, no regulatory requirements or guidance address the connection of safety bus offsite power sources through nonsafety buses.

EPRI's position is that many current designs of U.S. and foreign plants feed safety buses through nonsafety buses or from common transformer windings, and operating experience with these designs has not indicated any particular shortcomings. EPRI stated that there are real benefits in not connecting the safety buses directly to the offsite power supply, such as better protection of Class 1E systems against voltage surges affecting the offsite source. EPRI also states that the ALWR design makes provisions for a direct connection between safety buses and the offsite source in the event of problems with the nonsafety buses through which the safety buses are fed. EPRI indicated that some circuits are manually actuated and directly connect the safety buses to the reserve offsite transformer. The staff accepts such circuits. However, the EPRI Requirements Document does not require such a feature. In the GE ABWR standard plant design, all the offsite power sources are directly connected to the Class 1E safety bises with no intervening nonsafety buses. In the Westinghouse SP/90 design, one circuit provides a direct connection between the Class 1E safety buses and an offsite power source while the remaining circuits connecting the safety buses to the offsite sources are all routed through intervening nonsafety buses. The CE System 80+ design in this area is identical to the EPRI approach in that it provides for a direct connection between safety buses and the offsite source in the event of problems with the nonsafety buses through which the safety buses are fed.

The staff concludes that feeding the safety buses from the offsite power sources through nonsafety buses or from a common transformer winding with nonsafety loads is not the most reliable configuration. Such an arrangement increases the difficulty in properly regulating voltage at the safety buses, subjects the safety loads to transients caused by the nonsafety loads, and adds edditional failure points between the offsite power sources and safety loads. Therefore, it is the staff's position that at least one offsite circuit to each redundant safety division should be supplied diractly from one of the offsite power sources with no intervening nonsafety buses, in such a manner that the offsite source can power the safety buses upon a failure of any

The staff discusses this issue in Section 4.2.2 of the Chapter 11 DSER (Enclosure 2 to this paper).

ENCLOSURE 2 HAS BEEN SUPERSEDED BY THIS REPORT AND, THEREFORE, IS NOT INCLUDED IN THIS ANNEX

÷

0

REPORT BY THE ADVISORY COMMITTEE ON REACTOR SAFEGUARDS ON SECY-91-078 "PROPOSED POLICY ISSUES IDENTIFIED IN SECY-91-078, "CHAPTER 11 OF THE ELECTRIC POWER RESEARCH INSTITUTE'S (EPRI'S) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES'" APRIL 23, 1991

-



UNITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARDS WASHINGTON, D. C. 20055

April 23, 1991

The Honorable Kenneth M. Carr Chairman U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Chairman Carr:

SUBJECT: PROPOSED POLICY ISSUES IDENTIFIED IN SECY-91-078, "CHAPTER 11 OF THE ELECTRIC FOWER RESEARCH INSTITUTE'S (EPRI'S) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTION-ARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES"

During the 372nd meeting of the Advisory Committee on Reactor Safeguards, April 11-13, 1991, we discussed the two Policy Issues identified in SECY-91-078 related to the certification of the Evolutionary Light Water Reactors. Our Subcommittee on Improved Light Water Reactors also discussed these issues on April 9-10, 1991 in its continuing review of the EPRI Advanced Light Water Reactors (ALWR) Requirements Document. During these meetings, we had the benefit of discussions with representatives of the NRC staff and EPRI. We also had the benefit of the documents referenced.

The staff's position regarding the first Policy Issue is that "an evolutionary ALWR design should include an alternate power source to the non-safety loads unless the design can demonstrate that the design margins in the evolutionary ALWR will result in transients for a loss of non-safety power event that are no more severe than those associated with the turbine-trip-only event in current existing plant designs." The staff's major concern is that the ALWR designs are departures from past practice and may result in an increased frequency of shutdowns that require cooling by natural circulation. Presently licensed plants have electrical systems that provide an alternate power source to non-safety load: on shutdown. However, the staff did not substantiate its concerns with respect to the proposed EPRI design requirements.

EPRI claims that the ALWR is designed to safely accommodate shutdown with natural circulation and that the increased frequency of such events is small with this design. The EPRI requirements for the ALWR electrical system design fully meet General Design Criterion (GDC) 17, "Electric Power Systems," and the staff guidance contained in Regulatory Guide 1.32, Revision 2, "Criteria for Safety-Related Electric Power Systems for Nuclear Power Plants." The ALWR electrical power system design is arranged to The Honorable Kenneth M. Carr 2

supply electric power to the plant's safety loads from the main generator, the plant switchyard, an independent transmission line, a gas turbine generator, and the diesel generators. The design uses a generator circuit breaker between the main generator and the step-up transformer and has an improved full turbine load rejection capability. EPRI claims high reliability of electric power to the unit auxiliary transformers and has provided data to support its claim that the benefits derived from adding ry alternate power source to the non-safety loads are small and not cost effective. We concur with the EPRI position.

The staff's position regarding the second Policy Issue is based on a misunderstanding of the text of the EPRI requirements. As a result, the staff proposes an additional requirement that "at least one offsite circuit to each redundant safety division should be supplied directly from one of the offsite power sources with no intervening non-safety buses, in such a manner that the offsite source can power the safety buses upon a failure of any non-safety bus." The staff's concern is that routing offsite power to the safety buses through non-safety buses may subject safety equipment to undesirable disturbances on the non-safety buses. Therefore, the staff's position would require the capability to supply safety the staff's position would require the capability to supply safety buses directly from offsite power. The staff did not substantiate its concern. However, the EPRI requirements for ALWR electrical power system design already provide one alternate circuit to each of the redundant safety divisions directly from offsite power. This meets the staff's position. EPRI agreed to clarify the text to document this requirement. EPRI's position is that the direct circuit from offsite to each of the redundant safety divisions should be the backup power supply and the norm. 2 supply should be from the plant's auxiliary electric system. We concur with EPRI's position, but do not believe that this should become a regulatory position, but do not believe that this should become a regulatory re. . ement.

Sincerely,

David A. Ward Chairman

- SECY-91-078, Memorandum dated March 25, 1991 for the 1. Commissioners from James M. Taylor, Executive Director for Operations, Subject: Chapter 11 of the Electric Power Research Institute's (EPRI's) Requirements Document and Additional Evolutionary Light Water Reactor (LKR) Certification Issues (Predecisional)
- U.S. Nuclear Regulatory Commission, Office of Nuclear Reactor 2. Regulation, Draft Safety Evaluation Report on Chapter 11 of

60

The Monorable Kenneth M. Carr 3 April 23, 1981

the Advanced Light Water Reactor Requirements Document for Evolutionary Plant Designs, March 1991 Electric Power Research Institute, "Advanced Light Water Reactor Requirements Document, Chapter 11 - Electric Power Systems," Issued April 11, 1989 3.

STAFF RESPONSE TO ACRS CONCLUSIONS REGARDING SECY-91-078 MAY 15, 1991

.



UNITED STATES NUCLEAR REGULE DRY COMMISSION

May 15, 1991

Mr. David A. Ward Chairman Advis ry Committee on Reactor Safeguards U.S. Nuclear Regulatory Commission Washington, DC 20555

Dear Mr. Ward:

This letter is in response to your letter to Chairman Carr of April 23, 1991. in which you addressed the two policy issues that were identified in SECY-91-078 and that were also identified in the staff's draft safety evaluation report (DSER) on Chapter 11 of the Electric Power Research Institute (EPRI) Requirements Document for evolutionary plant designs.

The basis for the staff positions on these issues is derived from the lessons learned from operating experience coupled with the philosophy of defense in depth. As part of its review the staff must ensure that proposed changes in the design do not lead to increased likelihood of abnormal occurrences as compared to today's designs. The staff's positions on these issues are taken to ensure that, as a minimum, the levels of safety in these areas are no less than that found in many existing operating plant designs.

Regarding the first policy issue, the loss-of-power-to-nor safety-buses event continues to be identified as more severe than the turbine-trip-only event for current reactors. Also, most recently licensed plants in the United States have provided two power sources to non-safety loads. It should be noted that the loss of normal power to the non-safety buses is more complicated and places greater demands on hardware and personnel than simply a loss of feedwater event as shown in EPRI's presentation to the ACRS. The plant response to the event is significantly impacted by the unavailability of multiple major components. There is a loss of forced circulation by the reactor coolant pumps (recirculation pumps in BWRs), a loss of heat sink by the circulating water pumps, a loss of heat removal capability by the non-essential service water pumps, a loss of condensate feed by the condensate pumps, and a loss of heat removal capability by the component cooling water pumps. The staff believes that if the alternate power supply feature is not provided, the likelihood of this type of challenge per reactor year is increased by up to an order of magnitude. The addition of an alternat source of power to the non-safety buses is an important feature that signif cantly reduces the challenges to the plant presented by this complicated event.

#### David A. Ward

The EPRI presentation also included a cost estimate ( $\$15M \pm 5M$ ) for taplementation of the recommended design features. This estimate appears to be high and the staff will pursue this matter with EPRI in order to attain a more precise correlation between the specific design features involved and the estimated costs. It is the staff's opinion that a reasonable, simple design change providing all the recommended functions can be implemented at a fraction of the cost of the overall plant electrical system, and that this design change would constitute an important improvement in safety.

- 2 -

Regarding the second policy issue, the staff has taken the position that at least one offsite circui\* to each redundant safety division should be supplied directly from one of the offsite power sources with no intervening non-safety buses, in such a manner that the offsite source can power the safety buser bon a failure of any non-safety bus.

As was stated in the DSER for Chapter 11, the text of the EPRI Requirements Document does not require this feature and the staff's position is that the EPRI Requirements Document should be amended to require it. EPRI did not disagree with this position. The DSER further states that it has been the staff's experience that the benefits to safety of not connecting safety bures through non-safety buses or to common transformer windings usually outwein ever safety benefits may be achieved by doing so.

These positions constitute policy issues for the Commission's consideration in accordance with the staff requirements memorandum of August 24, 1989, pertaining to SECY-89-228, "Draft Safety Evaluation Report on Chapter 5 of the ALWR Requirements Document," in which the Commission instructed the staff to:

Identify these instances where the staff is proposing to depart from current regulations or where the staff is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs.

In its staff requirements memorandum of December 15, 1989, pertaining to SECY-89-334, "Recommended Priorities for Review of Standard Plant Designs," the Commission also provided the following guidance to the staff:

The SERs on the EPRI Requirements Document for both the evolutionary and the passive plant designs should be submitted to the ACRS for review and to the Commission for information and for review and approval of policy issues for which the Commission has not previously decided.

The staff's positions are not embodied in any current regulatory guidance, and in accordance with the Commission's desire to review such issues, the staff identified these issues to the immission for their review and approval. David A. Ward

.

The staff believes that its recommendations regarding both of these matters vill ensure that the likelihood of complex plant challenges is not increased and that the levels of safety in these areas are not reduced from that found in many existing operating plant designs. Therefore, the staff continues to recommend the Commission's approval of the staff's positions on these issues.

Sincerely,

Original Signed By. James M. Taylor

James M. Taylor Executive Director for Operations

cc: Chairman Carr Commissioner Rogers Commissioner Curtiss Commissioner Remick SECY

STAFF REQUIREMENTS MEMORANDUM "SECY-91-078 - CHAPTER 11 OF THE ELECTRIC POWER RESEARCH INSTITUTE'S (EPRI'S) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES" AUGUST 15, 1991


UNITED STATES NUCL PREGULATORY COMMISSION WASHINGTON D.C. 20555

August 15, 1991

OFFICE OF THE SECRETARY

> James M. Taylor, Executive Director MEMORANDUM FOR: for Operations Samuel J. Chilk, Secretar FROM: SECY-91-078 - CHAPTER 11 OF THE ELECTRIC SUBJECT: POWER RESEARCH INSTITUTE S/(EPRI'S) REQUIREMENTS DOCUMENT AND ADDITIONAL EVOLUTIONARY LIGHT WATER REACTOR (LWR) CERTIFICATION ISSUES

This is to advise you that the Commission (with all Commissioners agreeing) has approved the staff's positions in SECY-91-078. The staff should proceed with the issuance of the DSER for Chapter 11 of the EPRI requirements document with the requirements that; 1) evolutionary ALWR plants include an alternate power source to the nonsafety loads to provide additional margin for abnormal events based on operational experience, and 2) a direct connection will be provided between the offsite power source and the safety buses that will function upon a failure of any nonsafety bus.

The Commission also has agreed that the staff should raise policy issues to the . CRS and Commission early in the process, so that the Commission may make a decision on the approach to be taken on such issues. This does not require the staff to wait for the completion of ready-to-issue DSERs or SERs before coming to the Commission with policy issues. Policy issues should be presented for Commission consideration at the earliest possible opportunity, following sufficient discussions with the applicant to flesh out the parameters of such issues for Commission resolution. In this regard, while the staff should engage in early and open discussions with industry in order to identify and clarify the scope of, and industry positions on, such policy issues, the Commission should be apprised of such issues and provide guidance and approval before the staff takes a position on such issues with the applicant or the industry.

SECY NOTE: This SRM, the subject SECY paper and the vote sheets of the Chairman and Commissioners Rogers, Curtiss and Remick will be made publicly available 10 working days from the date of this SRM.

On a related matter, the direction to provide draft SERs to the Commission for review is intended to permit the Commission an opportunity to review, at its discretion, such SERs prior to issuance and, at the same time, to permit issuance of draft SERs while the Commission proceeds in parallel with the resolution of any <u>newly</u> identified policy issues. The earlier the Commission is apprised of policy issues, the less likely it will be that the staff will have to issue draft SERs at a time when policy issues remain unresolved.

Also, the staff should consider incorporating any modifications that arise from Commission decisions that depart from current regulations or substantially supplement or revise interpretive guidance into the regulations and regulatory guidance.

cc: The Chairman Commissioner Rogers Commissioner Curtiss Commissioner Remick OGC GPA ACRS

### ANNEX C

DOCUMENTS RELATED TO THE DRAFT COMMISSION PAPER "ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS" FEBRUARY 27, 1992

## CONTENTS

Draft Commission Paper "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," dated February 27, 1992	1B.C-3
ACRS Letter Related to Draft Commission Paper of February 27, 1992, dated May 13, 1992	18.C-63
Staff Response to ACRS Letter of May 15, 1992, dated June 12 1992	1B.C-69
ACRS Letter Related to Draft Commission Paper of February 27, 1992, dated August 17, 1992.	1B.C-73

.

Page

DRAFT COMMISSION PAPER "ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS" FEBRUARY 27, 1992

# DRAFT

12

.

For: The Commissioners

James M. Taylor

Executive Director for Operations

Subject:

From:

0

63

ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY RE-QUIREMENTS

Pyrpose:

To identify issues pertaining to evolutionary and passive light water reactors (LWRs) and the staff's recommendations concerning resolution of those issues for which the staff has completed its review. The staff requests that the Commission approve the positions described in this paper, and will inform the Commission of its proposed resolutions for the remaining issues upon completing its review of these items.

Summary:

The staff has identified a number of policy and significant technical issues pertaining to the evolutionary and passive LWRs. This paper provides a brief description and status of issues previously identified to the Commission and new issues. The staff has underlined the positions for which it is requesting Commission approval. Issues for which no resolution has been proposed yet will be addressed in future communications with the Commission as the staff's review progresses.

CONTACT: T. J. Kenyon, NRR 504-1120

Background:

The Commission instructed the staff to identify policy issues to the Advisory Committee on Reactor Safeguards (ACRS) and the Commission early in the review process, so that the Commission can make a decision on the approach to be taken for resolution of such issues. The staff was further instructed to provide an analysis detailing where it proposes departure from current regulations or where the staff is supplementing or revising interpretive guidance applied to currently licensed light water reactors.

The staff has forwarded several policy papers' to the Commission that identify proposed resolutions for policy matters and major technical issues concerning both the evolutionary (1300 MWe) and passive (600 MWe) LWR designs. SRMs have been issued to the staff that provide the Cummission's decisions regarding resolution of certain policy and major issues for evolutionary LWR designs.

The staff has identified a number of issues significant to Discussion: reactor safety as it considered operating experience, probabilistic risk assessment studies, and evaluations of the evolutionary and passive LWR designs. In Commission papers SECY-90-016, "Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements," and SECY-91-078, "Chapter 11 of the Electric Power Research Institute's (EPRI's) Requirements Document and Additional Evolutionary Light Water Reactor (LWR) Certification Issues," the staff proposed resolutions for some of these issues for evolutionary designs, which the Commission addressed in their SRMs of June 26, 1990, and August 15, 1991, respectively. The staff also forwarded policy papers to the Commission that identify and address the status of the staff's review of other issues. For completeness, this paper provides a brief description and status of these issues as well as new issues that pertain to the Advanced Light Water Reactor (ALWR) designs.

> In reviewing the EPRI's ALWR Requirements Document and the vendors' ALWR designs, the staff has identified the following policy and major technical issues for both the evolutionary and passive LWR designs. Certain issues represent matters specific to the design of these facilities while

- 2 -

Enclosure 3 to this document lists the papers that the staff has forwarded to the Commission regarding policy issues that have been identified to date for both evolutionary and passive advanced light water reactors (ALWRs). The staff will reference applicable documents throughout this paper, as appropriate.

others address the implementation of the design certification process of 10 CFR Part 52.

These issues are addressed in Sections 1, 11, and 111 of Enclosure 1. as listed below:

### I. SECY-90-016 Issues

A. use of physically-based source term

- 3 -

- B. anticipated transients without scram (ATWS)
- C. mid-loop operation
- D. station blackout
- E. fire protection
- F. intersystem loss-of-coolant-accident
- G. hydrogen control
- H. core concrete interaction ability to cool core debris
- I. high pressure core melt ejection
- J. containment performance
- K. dedicated containment vent penetration
- L. equipment survivability
- M. elimination of operating basis earthquake (OBE)
- N. in-service testing of pumps and valves

## II. Other Evolutionary and Passive Design Issues

- A. industry codes and standards
- B. electrical distribution
- C. seismic hazard curves and design parameters
- D. leak-before-break
- E. classification of main steamlines of boiling water reactors (BWRs)
- F. tornado design basis
- G. containment bypass
- H. containment leak rate testing
- post-accident sampling system
  level of detail
- K. prototyping
- L. inspections, tests, analyses, and acceptance criteria (ITAAC)
- M. reliability assurance program (RAP)
- N. site specific probabilistic risk assessments
- O. severe accident mitigation design alternatives (SAMDAs)
- P. generic rulemaking related to design certification

### III. Passive Design Issues Only

- A. regulatory treatment of non-safety systems
- B. definition of passive failure
- C. thermal-hydraulic stability of the simplified boiling water reactor (SBWR)
- D. safe shutdown requirements

.

- E. control room habitability
- F. radionuclide attenuation
- G. simplification of off-site emergency planning

Enclosure 1 provides a discussion of the nature of the current regulatory requirement or interpretation, the positions of the ALWR vendors and of EPRI, and, where available, the resolution that the staff is proposing, including the departure, if any, from current regulatory requirements and practice, and the basis for the staff's position. To aid in identifying the staff's positions, the staff has underlined the positions for which it is requesting Commission approval. Issues for which no resolution has been proposed will be addressed in future communications with the Commission as the staff's review progresses.

Enclosure 2 is a list of Commission papers, sorted by issue, in which the staff addressed issues for the evolutionary, passive, and/or advanced designs. The staff is providing this list for reference while reviewing Enclosure 1.

The staff developed the recommendations identified in this paper after (1) reviewing current generation reactor designs, evolutionary designs, and limited passive ALWR design information, (2) considering operating experience, (3) evaluating the results of the PRAs of LWRs and ALWRs, and (4) considering the Commission's guidance on issues resolved for the evolutionary ALWRs. The staff corcludes that these issues are fundamental to the agency's decisions on the acceptability of the evolutionary and passive designs. As discussed in Section II.P of Enclosure 1 and SECY-91-262, "Resolution of Selected Technical and Severe Accident Issues for Evolutionary Light Water Reactor Designs," the staff proposes to implement final positions on these matters as approved by the Commission through individual design certifications and generic rulemaking, as appropriate, for the evolutionary and passive designs.

The staff has reviewed only conceptual design information from the passive plant vendors in conjunction with the EPRI Requirements Document for passive designs, so its positions regarding passive designs may change as detailed design information becomes available, and testing on the passive systems is completed.

The staff concludes that it would be beneficial to provide EPRI and the ALWR vendors with the staff's conclusions presented herein as expeditiously as possible in order to help bring to closure those matters concerning both the evolutionary and passive reactor designs, and to facilitate

further discussion of the issues for the passive designs. This would allow the vendors to further their designs.

Therefore, the staff proposes to issue Enclosures 1-3 of this paper to the industry after 3 working days from the date of this paper to support meetings between the NRC staff, EPR!, the vendors, and the ACRS. The staff will indicate that the proposed resolutions identified to the indu ...y are before the Commission for consideration, and, therefore, may not represent final positions.

### Schedule

In SECY-91-161, "Schedules for the Advanced Reactor Reviews and Regulatory Guidance Revisions," the staff included a separate milestone for identifying policy issues for both the evolutionary and passive LWRs to the Commission and ACRS. The staff is submitting this paper to complete this milestone for both the evolutionary and passive LWR designs.

The staff does not expect to identify additional policy matters for the evolutionary designs. However, as the staff proceeds with its review of the EPRI ALWR Requirements Document for passive plants and the passive LWR design applications, it may identify other issues not identified herein. The staff allowed time for identifying and addressing such issues in the schedule estimates for the passive reactor design reviews provided in SECY-91-161. The staff will notify the Commission of new policy matters when they are identified.

<u>Conclusions</u>: The staff requests approval of the proposed resolution of those issues pertaining to evolutionary LWR designs in order to complete its review of EPRI's ALWR Requirements Document for evolutionary plant designs and to perform the final design approval/design certification review of General Electric's Advanced Boiling Water Reactor and Asea Brown Boveri (ABB) Combustion Engineering's System 80+ designs.

> The staff also requests the Commission to approve the proposed resolution of those issues pertaining to the passive designs. This will enable the staff to proceed with its review and to provide guidance to the reactor vendors in these areas early in the development of their passive designs. Issues for which the staff has not reached its final conclusions will be addressed in future papers to the Commission.

EPRI Evolutionary Plant SER

The Commissioners

<u>Coordination</u>: The Office of General Counsel has reviewed this paper and has no legal objection. OGC notes that Commission approval would be <u>tentative</u>, subject to further review in design certification rulemakings, and communications of the Commission's positions to the ALWR vendors and EPRI should make this point. This paper is being forwarded to the ACRS for its review and comments.

Recommendations: That the Commission:

- Approve the staff's positions underlined in Enclosure 1.
- (2) Note that the staff will seek commission approval of its positions for those issues still under staff review before taking a final position with the applicant or industry.
- (3) Note that the staff will provide Enclosures 1-3 to the ALWR vendors and EPRI after 3 working days from the date of this paper. The staff will indicate that the proposed resolutions are before the Commission for consideration, and, therefore, may not represent final positions.
- (4) <u>Note</u> that if the staff identifies other policy issues, the staff will seek Commission approval of its positions in a timely manner.

James M. Taylor Executive Director for Operations

Enclosures:

- Policy Issues Analysis and Recommendations
- Commission Papers Applicable to ALWR Issues
- 3. References

EPRI Evolutionary Plant SER

1B.C-10

- 6 -

Enclosure 1

# STAFT

### POLICY ISSUES ANALYSIS AND RECOMMENDATIONS

The staff has identified a number of issues significant to reactor safety as it considered operating experience, probabilistic risk assessment studies, and evaluations of the evolutionary (1300 MWe) and passive (600 MWe) LWR designs. In Commission papers SECY-90-016 and SECY-91-078, the staff proposed resolutions fc. some of these issues for evolutionary designs, which the Commission addressed in their SRMs of June 26, 1990 and August 15, 1991, respectively. The staff also forwarded policy papers to the Commission that identify and address the status of the staff's review of other issues. For completeness, this enclosure provides a brief description and status of these issues as well as new issues that pertain to both the evolutionary and passive LWR designs.

In reviewing the EPRI ALWR Requirements Document and the vendors' ALWR designs, the staff has identified the following policy and major technical issues for both the evolutionary and passive LWR designs. Certain issues represent matters specific to the design of these facilities while others address the implementation of the design certification process of 10 CFR Part 52. The following is a discussion of each of these issues. The staff has identified those instances in which its positions differ from current regulatory requirements or in which the staff is substantially supplementing or revising interpretive guidance applied to currently licensed LWRs. A discussion of the nature of the current regulatory requirement or interpretation, the positions of the ALWR vendors and of EPRI, and, where available, the resolution that the staff is proposing, including the departure, if any, from current regulatory requirements and practice, and the basis for the staff's position is provided for each issue. To aid in identifying the staff's positions, the staff has underlined the positions for which it is requesting Commission approval. Issues for which no resolution has been proposed will be addressed in future communications with the Commission as the staff's review progresses. Plant optimization subjects, which are items proposed by EPRI that do not meet regulatory requirements, are also identified.

This enclosure is divided into three sections. Section I addresses the staff's current positions regarding the applicability and resolution of issues identified in SECY-90-016 (for evolutionary designs only) for the passive LWR plant designs. Section II addresses the status and, where available, proposed resolution of other issues that are applicable to both the evolutionary and passive ALWR designs. Section III addresses issues that the staff has identified while reviewing the conceptual design information from the passive plant vendors in conjunction with the EPRI Requirements Document for passive designs. Positions on the passive designs may change as detailed design information becomes available and testing on the passive systems is completed.

Enclosure 3 lists the papers that the staff has forwarded to the Commission regarding policy issues that have been identified to date for both evolutionary and passive advanced light water reactors (ALWRs). The staff will reference applicable documents throughout this paper, as appropriate.

Enclosure 2 is a list of Commission papers, sorted by issue, in which the staff addressed issues for the evolutionary, passive, and advanced designs. This enclosure is included to provide a reference list for each issue. These papers will be referenced throughout this enclosure as appropriate.

### I. SECY-90-016 ISSUES

The following is a discussion of the staff's current positions regarding the applicability and resolution of issues identified in SECY-90-016 (for evolutionary designs only) for the passive LWR plant designs. Unless specified otherwise, the implementation guidance provided in SECY-90-016 is considered applicable to the discussions below.

### A. Use of Physically-Based Source Term

As discussed in SECY-90-016, the staff's methodology for determining compliance with the siting requirements of 10 CFR Part 100 has been based on the source term provided in Technical Information Document (TID)-14844, issued in 1962. This methodology, which involves calculation of offsite dose for comparison against Part 100 dose criteria (i.e., criteria for establishing the size of the exclusion area and the low population zone), is widely acknowledged to utilize conservative assumptions. At the time this approach was developed, these conservatisms were considered appropriate and were based on uncertainties associated with accident sequences and equipment performance, and a means to assure that future plant sites would be essentially equivalent to sites approved up until that time. The conservatisms initially included in the methodology have been essentially retained up to this time.

EPRI has proposed a physically-based source term to be used for the licensing design basis fission product release based on a bounding severe reactor accident to be used for both the evolutionary and passive reactor designs. Technical justification is provided in their October 18, 1990 and February 12, 1991 submittals entitled "Licensing Design Basis Source Term Update for the Evolutionary ALWR," and "Passive ALWR Source Term," respectively. EPRI has identified this as a plant optimization subject.

The EPRI-proposed source terms are based on a single, enveloping value for a bounding severe reactor accident sequence, using release data obtained from the Severe Fuel Damage Tests at the Power Burst Facility, the LOFT source term measurements, and data from the TMI-2 post accident examination. For the evolutionary designs, EPRI proposes changes in the assumptions concerning the fission product fuel release magnitude, the fission product release timing, the chemical form of iodine, the retention of aerosol in the reactor coolant, and the use of the suppression pool and containment sprays for removal of aerosol and soluble gases. For the passive designs, EPRI proposes that the source term be also based on consideration of passive mitigation functions and systems such as steam condensation-driven aerosol removal, main steam isolation valve leakage control, and secondary building fission product leakage control.

In SECY-90-016, the staff recommended that the Commission approve the following approach for evolutionary ALWRs:

- Assure that evolutionary designs meet the requirements of 10 CFR 100,
- Consider deviations to current methodology used to calculate 10 CFR Part 100 doses on a case-by-case basis using engineering judgement including updated information on source term and equipment reliability,
- 3. Do not modify current siting practice, and
- Continue to interact with EPRI and the evolutionary ALWR vendors to reach agreement on the appropriate use of updated source term information for severe accident performance considerations.

In its June 26, 1990 SRM, the Commission approved the staff's approach to the source term for the evolutionary designs with the addition that the staff incorporate appropriate changes to regulations, regulatory practices, and the review process resulting from source term research.

As a result of this guidance, the NRC staff is developing a revised source term based on source term calculations performed by the source term code package for individual accident sequences selected in NUREG-1150, "Severe Accident Risks: An Assessment for Five U.S. Nuclear Power Plants," December 1990. The staff concludes that the fission product release source terms proposed by EPRI and those being developed by the staff are very close for all radionuclide groups, except tellurium and low volatile elements. The reasons for and the impact of the differences between Brookhaven National Laboratories' (BNL's) and EPRI's estimates for tellurium and the low volatile elements are still under review.

General Electric (GE) and Combustion Engineering, Inc. (CE) have indicated that their evolutionary designs (the Advanced Boiling Water Reactor (ABWR) and the System 80+, respectively) can meet the offsite dose criteria established in 10 CFR Part 100 using the current TID-14844 source term. However, the vendors have indicated that they may wish to apply the forthcoming updated TID source term to their designs when it becomes available. The staff concludes that this approach is acceptable provided the vendor addresses the updated TID source term package in its entirety. If the evolutionary vendors choose to adopt this source term, the FDA/DC application must include complete analyses of the radiological consequences of design basis accidents using all of the guidelines provided in the updated TID source term. However, this may impact the review schedules of the evolutionary LWRs, because the updated TID source term is currently scheduled to be published for a 90 day public comment period in April 1992, and the staff's technical position on the fission product removal mechanisms in the containment are not finalized.

The passive ALWR vendors (Westinghouse and GE) have indicated that their designs will comply with the source term developed as a result of the staff's

evaluation of this issue with EPRI. The effort to revise the TID source term is expected to be completed in a time period sufficient to support the FDA/DC review of the passive plant designs.

The staff is in the final stage of completing its proposed update of the TID-14844 source term, including fission product removal mechanisms within the containment. The revised source term and its applicability to the ALWRs will be addressed in a separate Commission paper scheduled in the first quarter of 1992. Therefore, the staff does not request any Commission action on this issue at this time.

B. Anticipated Transients Without Scram (ATWS)

As discussed in SECY-90-016, the ATWS rule (10 CFR 50.62) was promulgated to reduce the probability of an ATWS event and to enhance mitigation capability if such an event occurred. The staff recommended that the Commission approve its position that diverse scram systems should be provided for evolutionary ALWRs. In addition, the staff indicated that GE would perform a reliability analysis to determine whether they could justify the manual operation of the standby liquid control system (SLCS) in the event of an ATWS in lieu of automatic operation as required by 10 CFR 50.62.

In its June 26, 1990 SRM, the Commission approved the staff's position, but directed that, if the applicant can demonstrate that the consequences of an ATWS are acceptable, the staff should accept the demonstration as an alternative to the diverse scram system.

EPRI has indicated that its approach to resolving the ATWS issue is compliance with the ATWS rule. Design requirements beyond those which would be required to meet the rule have not been proposed. In its December 6, 1991 letter, EPRI stated it has determined that automatic actuation of the SLCS was appropriate, and that it was modifying the Requirements Document for evolutionary designs to reflect that position. EPRI no longer considers this to be a plant optimization subject.

The ABWE design includes a number of features that reduce the risk from an ATWS event. These features include a diverse scram system with both hydraulic and electric run-in capabilities on the control rods, a standby liquid control system (SLCS), and a recirculation pump trip capability. In addition, the scram discharge volume has been removed from the ABWR, eliminating some of the potential ATWS problems associated with the older BWR designs. In its letter dated October 9, 1991, GE indicated it will include an automatic SLCS.

The CE System 80+ design includes a control-grade alternate protection system which provides an alternate reactor trip signal and an alternate feedwater actuation signal separate and diverse from the safety-grade reactor trip system.

The staff is in the process of evaluating the evolutionary plant submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related policy matters to result from its review.

Although EPRI has not submitted its position regarding the automatic actuation of the SLCS for passive designs, the staff expects EPRI to provide design requirements for the passive designs that are consistent with those for the evolutionary designs. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs. The staff will evaluate their designs upon submittal of their final design approval/design certification application.

Because the policy aspects of this issue are resolved (pending review of the design application for the passive designs), the staff does not request any Commission action on this issue at this time. Should a policy question be raised as a result of the staff's review of this matter, the staff will inform the Commission at the earliest opportunity.

### C. Mid-Lop Operation

In SECY-schold, the staff stated that it was concerned that decay heat removal capability could be lost when a PWR is shut down for refueling or maintenance and drained to a reduced reactor coolant system (RCS) or "mid-loop" level. The staff recommended that the Commission approve its position that evolution-ary PWR vendors propose design features to ensure high reliability of the shutdown decay heat removal system. In its April 26, 1990 letter, the Advisory Committee on Reactor Safeguards (ACRS) recommended that additional requirements be considered to resolve this issue. In its April 27, 1990 memorandum, the staff indicated that it would ensure that these recommendations were addressed in the evolutionary LWRs. In its June 26, 1990 SRM, the Commission approved the staff's position in conjunction with those of the ACRS.

The EPRI Requirements Document and evolutionary PWR designers have provided features to address the specific issue of mid-loop operation. The staff is in the process of evaluating the evolutionary plant submittals to ensure acceptable implementation of the Commission's directive.

However, the staff is concerned that the overall question regarding the vulnerability of the ALWRs during shutdown and low power operation has not been adequately evaluated by the vendors and EPRI. This issue has been discussed for evolutionary designs in a memorandum to the Commission dated September 5, 1990. The ALWR vendors and EPRI have been requested to assess shutdown and low power risk, identifying design specific vulnerabilities and weaknesses and documenting their consideration and incorporation of design features which minimize such vulnerabilities.

In its December 16, 1991 letter, EPRI subm. ted proposed changes to the Requirements Document to address this issue for the evolutionary designs. The staff has not yet received a response from  $t_1 = evolutionary vendors$ .

The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs. The staff expects that EPRI will specify requirements for the passive designs similar to those of the evolutionary.

- 6 -

The staff concludes that passive plants must also have a reliable means of maintaining decay heat removal capability during all phases of shutdown activities, including refueling and maintenance, and will evaluate the adequacy of the passive Requirements Document and designs during its review. The staff does not consider this issue to be a policy matter, but rather an element of its normal review. Therefore, the staff does not request any Commission action on this issue at this time. Should a policy question be raised as a result of the staff's review of this matter, the staff will inform the Commission at the earliest opportunity.

### D. Station Blackout

As discussed in SECY-90-016, the station blackout rule (10 CFR 50.63) allows utilities several design alternatives to ensure that an operating plant can safely shut down in the event that all ac power (offsite and onsite) is lost. The staff concluded that the preferred method of demonstrating compliance with 10 CFR 50.63 is through the installation of a spare (full capacity) alternate ac power source of diverse design that is consistent with the guidance in Regulatory Guide 1.155, and is capable of powering at least one complete set of normal shutdown loads. The staff recommended that the Commission approve imposition of an ac source for evolutionary ALWRS. In its June 26, 1990 SRM, the Commission approved the staff's position.

In addition to other design features to address the issue of station blackout, the EPRI Requirements Document for evolutionary designs and the evolutionary vendors have provided for a large capacity, diverse alfornate ac power source (combustion turbine generator) with the capability to power one complete set of normal safe shutdown loads. The staff is in the process of evaluating their submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related policy matters to result from its review.

Because the passive designs do not rely on active systems for safe shutdown following an event, EPRI and the passive plant designers do not believe safety-related diesel generators are necessary to address station blackout concerns. In addition, they believe an alternate ac power source should not be required. However, the staff concludes that the non-safety-related diesel generators may require some regulatory oversight. This issue is enveloped for the passive designs under the issue on regulatory treatment of non-safety systems (see paragraph III.A of this enclosure).

The staff is still evaluating this issue for the passive plant designs. It is identifying this issue in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest

opportunity. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

### E. Fire Protection

As discussed in SECY-90-016, the staff recommended that current NRC guidance to resolve fire protection issues should be enhanced to minimize fire as a significant contributor to the likelihood of severe accidents for advanced plants. In its April 26, 1990 letter, the ACRS recommended that the staff should consider additional matters in its evaluation of the fire protection designs. In its June 26, 1990 SRM, the Commission approved the staff's position as supplemented by the staff's April 27, 1990 response to the ACRS.

The EPRI Requirements Document and the evolutionary ALWR designers have indicated that their fire protection designs are consistent with the staff's proposed enhancements. The staff is in the process of evaluating their submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related policy matters to result from its review.

The staff expects that EPRI will specify requirements for the passive designs similar to those of the evolutionary. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive plant designs, it has not identified any unique features of the designs that would preclude the staff's conclusion that these designs should also be evaluated against the enhanced fire protection criteria. <u>Therefore</u>, the staff recommends that the Commission approve the staff's position that the passive plants should also be reviewed against the enhanced fire protection criteria approved in the Commission's June 26, 1990 SRM.

### F. Intersystem Loss-of-Coolant Accident

As discussed in SECY-90-016, the staff recommended that future evolutionary ALWR designs reduce the possibility of a loss-of-coolant accident (LOCA) outside containment by designing (to the extent practicable) all systems and subsystems connected to the reactor coolant system (RCS) to withstand the full RCS pressure. However, for both the passive and evolutionary reactors, design stress allowables for intersystem LOCA conditions in low pressure piping systems and uniform criteria must yet be developed and approved by the staff. The staff further recommended that systems that have not been designed to withstand full RCS pressure should include

- 1. the capability for leak testing of the pressure isolation valves,
- valve position indication that is available in the control room when isolation valve operators are deenergized, and

 high-pressure alarms to warn control room operators when rising RCS pressure approaches the design pressure of attached low-pressure systems and both isolation valves are not closed.

In its June 26, 1990 SRM, the Commission approved the staff's position on intersystem LOCA provided that all elements of the low pressure system are considered (e.g., instrument lines, pump seals, heat exchanger tubes, and valve bonnets). The staff has been applying items 1 - 3 to systems that have not been designed to operate at full RCS pressure.

The EPRI Requirements Document and the evolutionary ALWR designers have indicated that their submittals are consistent with the approved resolution to this issue. The staff is in the process of evaluating their submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related policy matters to result from its review.

The staff expects that EPRI will specify requirements for the passive designs similar to those of the evolutionary. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive plant designs, it has not identified any unique features of the designs that would preclude the staff's conclusion that these designs should also be evaluated against the staff's previous recommendation. <u>Therefore</u>, the staff recommends that the Commission approve the staff's position that the passive plants should also be reviewed against the position for intersystem LOCA as approved in the Commission's June 26, 1990 SRM.

### G. Hydrogen Control

Containments are required to be designed for control of hydrogen generation following an accident. 10 CFR 52.47(a)(ii) requires all applicants for design certification to demonstrate compliance with any technically relevant portions of the Three Mile Island requirements set forth in 10 CFR 50.34(f). 10 CFR 50.34(f) requires a system for hydrogen control that can safely accommodate hydrogen generated by the equivalent of a 100 percent fuel-clad metal water reaction, and to assure that uniformly distributed hydrogen concentrations in the containment do not exceed 10 percent (by volume) or that the post-accident atmosphere will not support hydrogen combustion.

In SECY-90-016, the staff recommended that, because of the uncertainties in the phenomenological knowledge of hydrogen generation and combustion, evolutionary ALWRs should be designed, as a minimum, to

- accommodate hydrogen equivalent to 100-percent metal-water reaction of the fuel cladding.
- limit containment hydrogen corcentration to no greater than 10 percent, and

 provide containment-wide hydrogen control (e.g., igniters, inerting) for severe accidents.

The staff recommended that the Commission approve the staff's position that the requirements of 10 CFR 50.34(f)(2)(ix) remain unchanged for evolutionary ALWRs. In its June 26, 1990 SRM, the Commission approved the staff's position.

ø

In its December 6, 1991 letter, EPRI stated that the EPRI Requirements Document for evolutionary designs will be modified to fully comply with the above positions. The ABWR design meets the requirements of 10 CFR 50.34(f)(2)(ix) by using, in conjunction with other systems, a nitrogeninerted atmosphere within its containment. The System 80+ design proposes to be consistent with the recommendations of the ALWR Requirements Document resulting from staff review. The staff is in the process of evaluating these submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related policy matters to result from its review.

The staff expects that EPRI will specify requirements for the passive designs similar to those of the evolutionary. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive riant designs, it has not identified any unique features of the designs that would preclude the staff's conclusion that these designs should also be evaluated against the staff's previous recommendation. <u>Therefore, the staff recommends that the Commission approve the staff's position that the passive plants should also be designed, as a minimum, to start the staff recommendation.</u>

- accommodate hydrogen equivalent to 100-percent metal-water reaction of the fuel cladding.
- limit containment hydrogen concentration to no greater than 10 percent. and
- provide containment-wide hydroger control (e.g., igniters, inerting) for severe accidents.

### H. Core Debris Coolability

In the unlikely event of a severe accident in which the core has melted through the reactor vessel, it is possible that containment integrity could be breached if the molten core is not sufficiently cooled. In addition, interactions between the core debris and concrete can generate large quantities of additional hydrogen and or ar non-condensible gases, which could contribute to eventual overpressure failure of the containment. Therefore, the staff concluded that plant designs should include features to enhance core debris coolability. As discussed in SECY-90-016, the staff recommended that the Commission approve the general criteria that evolutionary ALWR designs

1. provide sufficient reactor cavity floor space to enhance debris spreading. and

2. provide for quenching debris in the reactor cavity.

In its June 26, 1990 SRM, the Commission approved the staff's position.

In addition, the staff indicated in SECY-90-016 that it was evaluating the level of protection afforded by covering the containment liner and other structural members with concrete. The staff concludes that it may be necessary to protect these structural components with concrete.

The EPRI Requirements Document and the evolutionary ALWR designs provide a number of design features that are intended to mitigate the effects of a molten core. Among other features, the evolutionary designs are proposing floor sizing criteria of 0.02 m2/MWt and provisions to flood the lower drywell or reactor cavity. The staff does not support or dispute the EPRI floor sizing criteria of 0.02 m<sup>2</sup>/MWt. Instead, it concludes that it is appropriate to review the specific vendor designs to determine how they have addressed the three items discussed above to provide an increased level of protection relative to core debris coolability. The staff concludes that the "core-onthe-floor" accident will not be considered as a new design basis accident. However, the staff expects the vendors to consider the effects of an unmitigated core-concrete interaction on the production of non-condensible gases, the release of additional fission products from the core-concrete interaction, and additional heat and hydrogen generation in the new designs. Further, the staff will evaluate how the vendors have addressed core debris interactions with water in the cavity to account for steam and hydrogen generation.

The three criteria discussed above are intended to ensure that the ALWR vendors provide measures to the extent practical to mitigate severe accidents while avoiding turning severe accidents into traditional design basis accidents. As the staff neither supports nor disputes particular floor sizing criteria, the vendors should ensure that the containment can withstand the pressure increases caused by core-concrete interactions. For the range of severe accidents of concern, the vendors should realistically estimate the amount of core-concrete interaction that will occur, and ensure that the containment will accommodate the resultant conditions for at least 24 hours. where insufficient data exists to develop realistic estimates, the vendor may propose alternatives, such as additional tests or the use of other methodologies for determining the degree of corr concrete interaction. The ALWR vendors should also perform parametric studies to determine how sensitive the containment response is to variations in the amount of core-debris which is available to interact with the concrete. The staff concludes that incorporation of the mitigative measures to the extent practical and assurance of containment integrity for a 24 hour period will provide defense in depth as well as an appropriate degree of robustness in the containment design.

In its December 6, 1991 submittal, EPR1 indicated that it will specify requirements for the passive designs similar to those of the evolutionary designs. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive plant designs, it has not identified any unique features of the designs that would preclude the staff's conclusions that these designs should also be evaluated against the staff's previous recommendation. Therefore, the staff recommends the Commission approve the staff's position that both the evolutionary and passive LWR designs

- 1. provide reactor cavity floor space to enhance debris spreading.
- 2. provide a means to flood the reactor cavity 1, assist in the cooling process.
- 3. protect the containment liner and other structural members with concrete. if necessary, and
- 4. ensure that the containment can accommodate the pressure increases resulting from core-concrete interactions involving a range of scenarios which release core debris into the containment for 24 hours following the start of a severe accident.

### 1. High Pressure Core Melt Ejection

In SECY-90-016, the staff recommended that the Commission approve the staff's position that evolutionary ALWR designs should include a depressurization system and cavity design features to contain ejected core debris to reduce the potential for containment failure by direct containment heating (DCH). The staff is concerned that this phenomenon might occur from the ejection of molten core debris under high pressure from the reactor vessel resulting in wide dispersal of core debris, rapid oxidation, and extremely rapid addition of energy to the containment atmosphere.

In its June 26, 1990 SRM, the Commission approved the staff's position with the directive that the cavity design, as a mitigating feature, should not unduly interfere with operations, including refueling, maintenance, or surveillance activities. Examples of cavity design features that will decrease the amount of ejected core debris that reaches the upper containment include ledges or walls that would deflect core debris and an indirect path from the lower reactor cavity to the upper containment. The staff will review the LWR designs relative to the above criteria.

Notwithstanding the goal of providing a retentive cavity design, the inherent uncertainties in the industry's understanding of core dispersal phenomena following the failure of the reactor coolant system (RCS) lead to the recommendation that defense-in-depth be provided through a highly reliable RCS

depressurization system. Such a system would provide additional assurance that DCH would not occur by preventing reactor vessel failure at high pressure, and the rapid dispersal of core debris.

The EPRI Requirements Document and plant designers have provided features to address this issue for evolutionary designs. The staff is in the process of evaluating their submittals to ensure acceptable implementation of the Commission's directive. The staff's preliminary review of the passive ALWRs has also identified the importance of depressurization of the reactor coolant system (RCS) to the safe shown of the plant during transients or accidents. Depressurization of the RCS crucial to the operation of the passive safety features that limit the likelihood of core damage, as well as to reducing the potential for containment failure by direct containment heating from the ejection of core debris at high pressure. Therefore, the staff has determined that the passive ALWR designs must include a highly reliable depressurization system. The staff will also be evaluating the capability of non-safety systems to provide coolant injection under high pressure conditions should the depressurization system fail (see paragraph III.A of this enclosure).

Although the staff is still evaluating this issue for the passive plant designs, it has not identified any unique features of the designs that would preclude the staff's conclusions that these designs should also be coaluated against the staff's previous recommendations. <u>Therefore</u>, the staff recommends the Commission approve the general criteria that the passive LWR c signs

- 1. provide a reliable depressurization system, and
- provide cavity design features to decrease the amount of elected core debris that reaches the upper containment.

#### J. Containment Performance

As discussed in SECY-90-016, the staff recommended that the Commission approve the staff's position to use a conditional containment failure probability (CCFP) of 0.1 or a deterministic containment performance goal that offers comparable protection in the evaluation of evolutionary ALWRs. The staff concluded that the following general criterion for containment performance during a severe-accident challenge would be appropriate for the evolutionary ALWRs in place of a CCFP.

The containment should maintain its role as a reliable leak tight barrier by ensuring that containment stresses do not exceed ASME service level C limits for a minimum period of 24 hours following the onset of core damage and that following this 24 hou period the containment should continue to provide a barrier against the uncontrolled release of fission products.

The containment performance goal was proposed to ensure that the containment will perform its function in the face of most credible severe accident challenges.

In its June 26, 1990 SRM, the Commission approved the use of a 0.1 CCFP as a basis for establishing regulatory guidance for the evolutionary LWRs. The Commission directed that this objective should not be im osed as a requirement, and that the use of the CCFP should not discourage accident prevention. The staff was directed to review suitable alternative, deterministically-established containment performance objectives providing comparable mitigation capability that may be submitted by the applicants. The Commission directed that any such alternatives be submitted to the Commission.

While the staff has identified, both in this paper and in SECY-90-016, the major challenges to the containment (e.g., hydrogen burns, corium interactions with water and containment structures), and the need to provide the means for mitigation of these challenges, the containment performance goal acts as a final check to ensure that the design (including its mitigation features) would be adequate if called upon to mitigate a severe accident. Although not explicitly identified in SECY-90-D16, the staff will also evaluate the impact of molten fuel-coolant interaction and the resulting steam and hydrogen generation (and any dynamic forces due to ex-vessel fuel-coolant interactions) on the integrity of the containment, consistent with the containment performance goal. The intent of both the CCFP and the alternative deterministic performance criteria discussed above is to provide this final check, as well as defense-in-depth. The philosophy behind the use of the proposed deterministic goal is that adequate time must be provided for fission product decay before allowing a release from the containment to the environment. Since Service Level C is applicable only to metal containments, a comparable criterion is needed for the concrete containments, such as used in the ABWR and SBWR designs. The staff is evaluating options for this criterion.

In its letter dated December 16, 1991, EPRI stated that its overall containment performance requirements address the assurance of a robust containment for ALWRs capable of accommodating risk-significant severe accident loads without failure. The staff concludes that EPRI's position is consistent with the deterministic containment performance goal. EPRI has also identified a comprehensive list of containment challenges resulting from core damage sequences. Based on its containment performance studies for the evolutionary and passive plant designs, EPRI has concluded that the described design features limit the l:kelihooo and magnitude of these challenges, and that they ensure the capability of the containment to accommodate all challenges which are potentially risk significant. The staff is in the process of evaluating the adequacy of the containment performance of the individual ALWR reactor designs to ensure that these and any other potential sequences that may be identified during the staff's review are adequately addressed. The staff will evaluate the criteria for the vendor's determination of the challenges to the containment. The evaluation of containment bypass sequences will be addressed on a vendor-specific basis during the staff's design reviews. The staff does not expect any related policy matters to result from its review of the evolutionary designs.

EPRI Evolutionary Plant SER

- 13 -

The staff expects tha EPRI will specify requirements for the passive designs similar to those of t evolutionary. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive plant designs, the staff concludes that it is appropriate to apply the same containment performance goal to the passive designs as a basis for establishing regulatory guidance. Therefore, the staff recommends the Commission approve the staff's position to use a conditional containment failure probability (CCFP) of 0.1 or a deterministic containment performance goal that offers comparable protection in the evaluation of the passive ALWRs. The staff will consider any suitable alternative, deterministically-established containment performance objectives providing comparable mitigation capability. Applicants using the deterministic approach will be required to clearly define the challenges considered in this evaluation.

### K. Dedicated Containment Vent Penetration

As discussed in SECY-90-016, the staff recommended that the Commission approve the use of an overpressure protection system that sed a dedicated containment vent for the ABWR. This system is designed to avoid gross containment failure resulting from postulated slow rising overpressure scenarios itst could result from postulated multiple safety system failures.

'n its June 26, 1990 SRM, the Commission approved the staff's recommended use of the containment overpressure protection system for the ABWR, subject to a comprehensive regulatory review to weigh the "downside" risks with the mitigation benefits of the system. In addition, the Commission directed the staff to ensure that full capability to maintain control over the venting process is provided in the design.

In its December 6, 1991 letter, EPRI stated that the EPRI Requirements Document for evolutionary designs will be modified to require containment overpressure protection through the containment design (considering size and strength of the containment) or through the use of pressure relief. EPRI proposes that the need for a containment vent be determined on a designspecific basis. EPRI considers this to be a plant optimization subject. The ABWR includes a dedicated vent system in its containment design.

The staff is in the process of evaluating these submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related policy matters to result from its review. The staff expects that EPR! will specify requirements for the passive designs similar to those of the evolutionary. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Because of the stage of design development, the staff has insufficient information to determine whether a containment vent is necessary for passive plant designs at this time. The containment performance riteria proposed in Section 1.3 of this enclosure will serve as the basis for the staff's review of containment integrity and the need for a containment vent. <u>Therefore, the staff recommends that the Commission Approve the staff's position that the need for a containment vent for the passive plant designs be evaluated on a design specific basis.</u>

### L. Equipment Survivability

As discussed in SECY-90-016, the staff recommended that the Commission approve the staff's position that 'eatures provided only for severe-accident protection need not be subject to the 10 CFR 50 49 environmental qualification requirements, 10 CFR Part 50, Appendix B quality assurance requirements, and 10 CFR Part 50, Appendix A redundancy/diversity requirements. The c sons for this judgement is that the staff does not balieve that severe core damage accidents should be treated in the same manner that design basis accidents (DBAs) have traditionally been treated, due to the la ge differences in their likelihood of occurrence. However, SECY-90-016 further stated that mitigation features must be designed so there is reasonable assurance that they will operate in the severe-accident environment for which they are intended and over the time span for which they are needed. In instances where safety related equipment which is provided for design bases accidents is relied upon to cope with severe accident situations, there should also be a high confidence that this equipment will survive severe accident conditions for the period that it is meded to perform its intended function. During the review of the credible severe accident scenarios for specific ALWR designs, the equipment needed to perform mitigative functions, and the conditions under which the mitigat'. " systems must function, will be identified. Equipment survivability ex; stations under severe accident conditions should include consideration of the circumstances of applicable initiating events (e.g., station blackout, earthquakes) and the environment (e.g., pressure, tempera-ture, radiation) in which the equipment is relied upon to function. The required system performance criteria will be based on the results of these design-specific reviews. In its June 26, 1990 SRM, the Commission approved the staff's position.

In its letter dated May 6, 1991, the staff clarified its position that this criteria would be applied to those features provided only for severe accident mitigation.

The EPRI Requirements Document and the evolutionary ALWR designers have indicated that their submittals are consistent with this criteria. The staff is in the process of evaluating their submittals to ensure acceptable implementation of the Commission's directive, but does not expect an related policy matters to result from its review. The passive ALWR vencers have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive plant designs, it has not identified any unique features of the designs that would preclude the staff's conclusion that these designs should also be evaluated against this criteria. Therefore, the staff recommends that the Commission approve the staff's position that features provided only for severe-accident mitigation for the passive plant designs need not be subject to the 10 CFR 50.49 environmental qualification requirements. 10 CFR Part 50. Appendix B quality assurance requirements, and 10 CFR Part 50. Appendix A redundancy/ diversity requirements. As discussed in SECY-90-016, the staff concludes that guidance such a: that found in Appendices A and B of Regulatory Guide 1.155, "Station Blackout," is appropriate for equipment used to mitigate the consequences of severe accidents.

#### M. Elimination of Operating Bases Earthquake

In SECY-90-016, the staff discussed its proposal to decouple the operating basis earthquake (OBE) from the safe-shutdown earthquake (SSE). The regulations in Appendix A to 10 CFR Part 100 establish the OBE at one-half of the SSE. The staff stated that the OBE should not control the design of safety systems and was evaluating possible changes to the regulations that would reduce the magnitude of the OBE relative to the SSE. The staff recommended that the Commission approve the review approach to consider requests to decouple the OBE from the SSE on a design-specific basis for evolutionary designs. In its June 26, 1990 SRM, the Commission approved the staff's recommendation.

In a plant optimization subject, EPRI requested that the staff evaluate the elimination of the OBE altogether from design of systems, structures, and components in nuclear power plants. In its April 26, 1990 letter, the ACRS also recommended this approach. The NRC staff, in evaluating the decoupling of the OBF from the SSE, is also evaluating the possibility of redefining the OBE in . er to satisfy its function without an explicit response analysis. This change would diminish the role of the OBE in design by establishing a level which, if it is exceeded, would require that the plant be shut down for inspection activities.

EPRI's position on seismic design is that it is unnecessary to perform two complete sets of seismic analyses -- one for the OBE and one for the SSE. The NRC staff agrees, in principle, with this position but finds that extant design practices for piping and structures do not result in designs that are significantly controlled by the OBE. As stated in SECY-90-016, certain interim measures, such as allowing higher damping values for piping analyses, have been already implemented to alleviate the situation of having the OBE significantly controlling the design.

The elimination of the OBE response analysis would require all current OBE design-related checks to be performed for a fraction of the SSE. The staff is currently developing various alternatives with the industry to revise the codes and standards when design-related checks are based on the OBE. For example, in piping design, the ASME Boiler and Pressure Vessel Code currently establishes rules for the evaluation of earthquake cycles on fatigue and relative seismic anchor motion effects that are based on the OBE. In addition, the NRC requirements for postulating the number and location of pipe

ruptures are also derived from the OBE. If the OBE is eliminated from design, then these loadings might need to be performed by using the SSE and establishing new appropriate allowable limits.

Therefore, the staff recommends that the Commission approve the staff's approach to eliminate the QBE from design of systems, structures, and compo-nents. Until the final rulemaking to Appendix A to 10 CFR Part 100 is completed, the elimination of the QBE from design of either the evolutionary or passive designs will require an exemption from the current regulations with acceptable supporting justification from the designer. The details of how current OBE-related design checks will be performed using the SSE will be resolved between the industry and the NRC staff through the appropriate coderelated activities or supplemental regulatory guidance.

N. In-Service Testing of Pumps and Valves

As discussed in SECY-90-016, the staff recommended that the Commission approve the staff's position that the following provisions be applied to all safety related pumps and valves and not limited to ASME Code Class 1, 2, and 3 components.

- 1. Piping design should incorporate provisions for full flow testing (maximum design flow) of pumps, and check valves.
- 2. Designs should incorporate provisions to test motor operated valves under design basis differential pressure.
- 3. Check valve testing should incorporate the use of advanced non-intrusive techniques to address degradation and performance characteristics.
- 4. A program should be established to determine the frequency necessary for disassembly and inspaction of pumps and valves to detect unacceptable degradation which cannot be detected through the use of advanced non-intrusive techniques.

The staff concluded that these requirements were necessary to provide an adequate level of assurance of operability.

In its June 26, 1990 SRM, the Commission approved the staff's position as supplemented in the staff's April 27, 1990 response to ACRS comments. The Commission further noted that due consideration should be given to the practicality of designing testing capability, particularly for large pumps and valves.

The EPRI Requirements Document and the evolutionary ALWR designers have indicated that their submittals are consistent with this criteria. The staff is in the process of evaluating their submittals to ensure acceptable implementation of the Commission's directive, but does not expect any related

policy matters to result from its review. The passive ALWR vendors have indicated that their designs will comply with the Requirements Document for passive designs.

Although the staff is still evaluating this issue for the passive plant designs, it has concluded that this is an issue for passive designs because the passive safety systems rely on the proper operation of this equipment (i.e., check valves, depressurization valves) to mitigate the effects of transients. This is further discussed under the issue regarding the definition of passive failures (see paragraph III.B of this enclosure). <u>Therefore</u>. the staff recommends that the Commission approve the staff's position that these requirements also be imposed on passive ALWRS.

### II. Other Evolutionary and Passive Design Issues

The following is a discussion of the status and, where available, proposed resolution of issues that are applicable to both the evolutionary and passive ALWR designs.

### A. Industry Codes and Standards

In SECY-91-273, the staff raised the concern that a number of design codes and industry standards dealing with new plant construction have been recently developed or modified, and that the NRC has not yet determined their acceptability. EPRI and the ALWR vendors are using codes and standards in their applications that the staff has not endorsed.

The staff recommends that the Commission approve the staff's position that, consistent with past practice, it use the newest codes and standards that have been endorsed by the NRC in its reviews of both the evolutionary and passive plant design applications. Unapproved revisions to codes and standards will be reviewed on a case-by-case basis.

B. Electrical Distr. ution

In SECY-91-078, the staff recommended that the Commission approve its position that an evolutionary plant design should include

- an alternate power source to the non-safety loads unless the design can demonstrate that the design margins will result in transients for a loss of non-safety power event that are no more severe than those a sociated with the turbine-trip-only event in current existing plant designs, and
- at least one offsite circuit to each redundant safety division supplied directly from one of the offsite power sources with no intervening nonsafety buses in such a manner that the offsite source can power the sifety buses upon a failure of any non-safety bus.

### In its August 15, 1991 SRM, the Commission approved the staff's positions.

In its December 6, 1991 letter, EPRI stated that it was modifying the EPRI Requirements Document for evolutionary plants, and would be submitting its proposed resolution in a future response. The staff will evaluate EPRI's submittal and those of the evolutionary final design approval/design certification applicants to ensure acceptable implementation of the Commission's directive, but it does not expect any related policy matters to result from its review.

Because the passive designs do not rely on active systems for safe shutdown following an event, the staff has not determined the applicability of this issue to the passive designs. This issue is enveloped for the passive designs under the issue on regulatory treatment of non-safety systems (see paragraph III.B of this enclosure).

The staff is still evaluating this issue for the passive plant designs. It is identifying this issue in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest opportunity. Should this issue be identified as applicable to the passive plant designs, the staff will provide a proposed resolution to the Commission later. Therefore, the staff jues not request any Commission action on this issue at this time.

### C. Seismic Hazard Curves and Design Parameters

To assess the seismic risk associated with an ALWR design, EPRI has proposed the use of generic bounding seismic hazard curves for sites in the central and eastern United States. EPRI proposes that these curves be used in the seismic probabilistic risk assessment (PRA). The regulations do not require and the staff does not intend to require that a seismic PRA be performed to determine if a site is acceptable.

To assess the EPRI ALWR seismic hazard bounding curve for rock sites, the staff compared the EPRI curve to results derived by Lawrence Livermore National Laboratories (LLNL) using the historical earthquake method discussed in NUREG/CR-4885, "Seismic Hazard Characterization of the Eastern United States: Comparative Evaluation of the LLNL and EPRI Studies," 1987. The staff also compared the EPRI bounding curve to hazard curves generated by EPRI using the historical method for the Seabrook site (see letter dated October 17, 1991). The historical hazard curves below 0.1g reflect the past few hundred years of historical earthquake data. The historical hazard curves at higher accelerations are estimates based on the historical earthquake data. Both the LLNL and EPRI hazard curves, which were derived using the historical method, exceed the EPRI bounding curve at acceleratic below about 0.1g. Because the EPRI bounding curve is exceeded at low peak accelerations the adequacy of the EPRI bounding curve at higher peak accelerations.

Hazard curves generated for the Seabrook Station Probabilistic Safety Assessment (1983) by the licensee exceed the EPRI bounding hazard curve. The Seabrook SSE has a peak acceleration of 0.25g, whereas a higher SSE of 0.3g is proposed for ALWR sites. On the basis of the deterministic design basis of 0.3g, the EPRI-proposed criteria can be assumed to be suitable for the Seabrook site. However, based on the probabilistic assessment, the EPRI bounding hazard curve would underestimate the core damage frequency. Thus, the EPRI bounding hazard curve is non-conservative when compared to a licensee submittal.

The LLNL hazard curves are used in the staff's reviews of seismic hazard and are generally higher than the EPRI results for the same sites. Some LLNL hazard curves for sites in the Eastern United States (discussed in NUREG/CR-5250, "Seismic Hazar, Characterization of 69 Nuclear Plant Sites East of the Rocky Mountains," 1989) exceed the EPRI bounding hazard curve.

During the staff's review of the ABWR PRA, PRA results using both the LLNL and EPRI hazard estimates were compared with results using the ABWR bounding seismic hazard curve. The ABWR bounding hazard curve was exceeded by the LLNL mean hazard curves for the Pilgrim, Seabrook, and Watts Bar sites. These three sites in the eastern United States were selected because of their relatively high seismic hazard. The staff used both the LLNL and EPRI seismic hazard estimates to quantify core damage frequency. The PRA using the LLNL hazard curves predicted much higher core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the EPRI hazard curve. However, and WR core damage frequencies than the PRA using the LLNL for the PRA using the formula to be capable of resisting earthquakes signifies the provide formula to be capable to be cap

The evolutionary and passive ALWA is the provision of the stated that their applications will be consistent with the training is rise. However, based on the staff's review of historical seismicity and the LLNL hazard estimates, the staff has concluded that the EPRI seismic hazard bounding curve is not sufficiently conservative. The staff is svaluating the seismicity and ground motion inputs used in the LLNL and EPRI studies to determine if the uncertainties in the curves can be reduced.

Based on the deterministic process used by the staff to judge the seismic capability of the GE and CE designs, the staff concludes that, with few exceptions, almost all areas east of the Rocky Mountains would be candidate sites for these designs. As part of the COL process, the arglicant will have to demonstrate that the site-specific seismic parameters meet the certified design parameters to assure issue preclusion at the COL hearing. Should an actual site value exceed the design envelope in a certain area, a specific analysis will have to be performed to verify that the design is still acceptable for that site.

The discussion on seismic hazard curves provided in this section is for information only, and is provided to identify a complete list of issues applicable to the passive designs. If a policy question is identified as a result of its review, the staff will inform the Commission of the issue at the earliest opportunity. Therefore, the staff does not request any Commission action on this issue at this time. General Design Criterion (GDC) 4 states, in part, that, "...dynamic effects associated with postulated pipe ruptures in nuclear power units may be excluded from the design basis when analyses reviewed and approved by the Commission demonstrate that the probability of fluid system piping rupture is extremely low under conditions consistent with the design basis for the piping."

Under the broad scope revision to GDC 4 (52 FR 41288, October 27, 1987), the NRC allows the use of advanced technology to exclude from structural design consideration the dynamic effects of pipe ruptures in nuclear power plants provided it is demonstrated that the probability of pipe rupture is extremely low under conditions consistent with the design bases for the piping. Demonstration of low probability pipe rupture requires a deterministic fracture mechanics analysis that evaluates the stability of postulated small, throughwall flaws in piping and the ability to detect leakage through the flaws long before the flaw could grow to unstable sizes. The concept underlying such analyses is referred to as "leak-before-break (LBB)."

To date, the LBB approach has been approved by the NRC staff for currently operating and near-term operating licensed nuclear power plants based on a case-by-case review of plant-specific analyses. The NRC staff has approved the use of LBB for PWR primary coolant loop piping in all but five units in the United States. In addition, the pressurizer surge, accumulator, and residual heat removal piping have been approved for 11 units. In all cases, the LBB approvals have been granted for piping inside primary containment and for piping at least six inches nominal diameter. The piping includes both austenitic and carbon steel material. However, the LBB-approved carbon steel piping have all been clad with stainless steel material. To date, no BWRs have requested LBB approval.

EPRI and GE are proposing to adopt the LBB approach for ALWRs when certain details of the piping design, materials properties, and stress conditions are known. As discussed in SECY-89-013, the staff will evaluate the acceptability of the use of leak-before-break considerations in the ALWR designs when it can be justified. The staff has evaluated the EPRI and GE proposal for LBB application to ALWRs, as discussed below.

### LBB Acceptance Criteria

The staff concludes that the analyses referred to in GDC-4 should be based on specific data, such as piping geometry, materials, and piping loads. The staff must review the LBB analyses for specific piping designs before the applicant can exclude the dynamic effects from the design basis. For ALWRs seeking design certification under 10 CFR Part 52, the analyses may be allowed to incorporate preliminary stress analysis results provided bounding limits (both upper- and lower-bound) are determined in order to establish assurance that adequate margins are available for leakage. loads, and flaw sizes.

A margin of 10 on leakage is required so that leakage from the postulated flaw size is assured of detection when the pipe is subjected to normal operational loads. A margin of 2.4 on loads is required to ensure that leakage-size flaws are stable at normal plus accident loads (e.g., safe-shutdown earthquake and safety-relief valve discharge loads). A factor of 2 between the leakage-size flaw and the critical-size flaw is required to ensure an adequate stability margin for the leakage-size flaw.

In addition, for ALWRs that seek approval of LBB during the design certification phase, certain information will be required for LBB analyses to establish through-wall flaw sizes and flaw stability. For through-wall flaw sizes, a lower-bound, normal-operational stress limit must be established for dead weight, pressure, and thermal loadings. The mean or best-estimate stressstrain curve should be used. For flaw stability, an upper-bound stress limit should be established for normal loadings plus safe shutdown earthquake and "uppression pool hydrodynamic loadings. A lower-bound stress-strain curve for pase metal should be used regardless of whether the weld or base metal is limiting. In addition, a lower-bound toughness for the weld metal should be used.

A deterministic fracture mechanics evaluation accounting for material toughness is required. Applicants may propose any fracture mechanics evaluation method for NRC staff review. However, the applicants will have to demonstrate the accuracy of the method by comparing it with other acceptable methods or with experimental data.

Using the above approach, an initial set of bounding values and a preliminary LBB analysis can be established for ALWRs during the design certification phase. These bounding values and preliminary analyses can be verified when as-built and as-procured information become available during the combined license phase. Prior to fuel-loading, verification of the preliminary LBB analyses should be completed and based on actual material properties and final, as-built piping analysis as part of inspection, tests, analyses, and acceptance criteria (ITAAC) associated with 10 CFR Part 52.

### LBB Limitations

Because of the dependency of the LBB analyses to accurately predict the flaw stability, the LBB approach has established certain limitations for excluding piping that is likely to be susceptible to failure from various degradation mechanisms during service. A significant portion of the LBB review involves the evaluation of the susceptibility of the candidate piping in various degradation mechanisms to demonstrate that the candidate piping is not susceptible to failure from these degradation mechanisms. The operating history and measures to prevent or mitigate these mechanisms are reviewed by the NRC staff.

The LBB approach cannot be applied to piping that can fail in service from such effects as water hammer, creep, erosion, corrosion, erosion-corrosion, fatigue, thermal stratification, and environmental conditions. The rationale is that these degradation mechanisms challengs the assumptions in the LBB acceptance criteria. For example, (1) water hammer may introduce excessive dynamic loads which are not accounted for in the LBB analyses, and (2) corrosion and fatigue may introduce flaws whose geometry may not be bounded by the postulated through-wall flaw in the LBB analyses. Adhering to the "defense-in-depth" principle, piping susceptible to failure from these potential degradation mechanisms is excluded from LBB applications. Alternatively, features to mitigate the possibility of certain degradation mechanisms may b: proposed to ensure that LBB assumptions are not invalidated. For example, LBB might be considered for carbon steel piping for which the effects of erosion-corrosion have been eliminated through the use of high chromium steels with proven resistance to erosion-corrosion or through the use of carbon steel piping that is clad on the fluid surface with erosion-corrosion

A detailed discussion of the limitations and acceptance criteria for LBB used by the NRC staff is provided in NUREG-1061, Volume 3, "Evaluation of Potential for Pipe Breaks, Report of the U.S. Nuclear Regulatory Commission Piping Review Committee," dated November 1984.

### Design Basis with LBB

The broad scope rule introduced an acknowledged inconsistency into the design basis by excluding the dynamic effects of postulated pipe ruptures while retaining non-mechanistic pipe rupture for containments, emergency core cooling systems (ECCSs), and environmental qualification (EQ) of safetyrelated electrical and mechanical equipment. The NRC staff subsequently clarified its intended treatment of the containment, ECCS, and EQ in the context of LBB application in a request for public comments on this issue that was published on April 6, 1988 (53 FR 11311).

Effects resulting from postulated pipe ruptures can be generally divided into local dynamic effects and global effects. Local dynamic effects of a pipe rupture are uniquely associated with that of a particular pipe rupture. These specific effects are not caused by any other source or even a postulated pipe rupture at a different location. Examples of local dynamic effects are pipe whip, jet impingement, missiles, local pressurization, pipe break reaction forces, and decompression waves in the intact portions of that piping or communicating piping. Global effects of a pipe rupture need not be associated with a particular pipe rupture. Similar effects can be caused by failures from sources such as pump seals, leaking valve packings, flanged connections, bellows, manways, rupture disks, and pipe ruptures of other piping. Examples of global effects are gross pressurization, temperatures, humidity, flooding, loss of fluid inventory, radiation, and chemical condition. For the ABWR, global effects also include suppression pool hydrodynamic loads (e.g., safetyrelief valve discharges, pool-swell/fallback, condensation oscillation, and chugging loads). Although the suppression pool hydrodynamic loads caused by a main steam or feedwater pipe rupture might be excluded for the design of piping, equipment, and internal containment structures (other than-those serving a containment function), the possibility of such dynamic effects being caused by a reactor internal pump ejection, failures of flanged connections,

and blowdowns from rupture disks or squib-actuated valves have not been addressed at this time. The option does exist to establish a postulated pipe break of a high-energy line smaller than the main stram of feedwater line break to envelope the possible global dynamic effects described above. This approach would be required to be submitted to the NRC staff for review and approval prior to its use. Until then, the use of a postulated sipe rupture of a main steam or feedwater line should be assumed for suppression pool hydrodynamic loads.

The application of LBB technology eliminates the local dynamic effects of postulated pipe ruptures from the design basis. Because the global effects from the postulated pipe rupture provide a convenient and conservative design envelope, the NRC staff will continue to require the consideration of global effects for various aspects of the plant design, such as environmental qualification of equipment, design of containments, design of subcompartment enclosures, etc.

#### Recommendations

The elimination of dynamic effects of postulated high energy pipe ruptures from the design basis of ALWRs using advanced fracture mechanics analyses (leak-before-break approach) is permitted in the revised GDC 4 of Appendix A to 10 CFR Part 50. The limitations and acceptance criteria for LBB applications in ALWRs are the same as those established for currently operating nuclear power plants. <u>Therefore, the staff recommends that the Commission</u> approve the application of the LBB approach to both evolutionary and passive ALWRs seeking design certification under 10 CFR Part 52 when appropriate bounding limits are established during the design certification phase using preliminary analyses results and verified during the combined license phase by performing the appropriate ITAAC discussed herein. However, the specific details will need to be developed as the process is implemented during the first trial application.

E. Classification of Main Steamline of Boiling Water Reactors (BWRs)

### Background

The main steamlines in boiling water reactor (BWR) plants contain dual quickclosing main steam isolation valves (MSIVs). These valves function to isolate the reactor system in the event of a break in a steamline outside of the primary containment, a design basis LOCA, or other events requiring containment isolation. Although the MSIVs are designed to provide a leak-tight barrier, it is recognized that some leakage through the valves will occur.

The current procedure for determining the acceptability of MSIV leakage involves calculating the dose in accordance with 10 CFR Part 100 based on a conservative assumption that the leakage limit allowed by the technical specification (normally 11.5 scfh per valve) is released directly into the environment. No credit is currently taken for the pressure integrity of the main steam piping and condenser. Because of recurring problems with excessive leakage of MSIVs, Regulatory Guide 1.96, "Design of Main Steam Isolation Valve Leakage Control Systems for Boiling Water Reactor Nuclear Power Plants," recommended the installation of a supplemental leakage control system (LCS) to ensure that the isolation function of the MSIVs complies with the specified limits. Operating experience with the LCS has required substantial maintenance and worker exposure. Additionally, the NRC has generic concerns with the effectiveness of the LCS to perform its intended function under conditions of high MSIV leakage (Generic Issue C-8, Main Steam Line Valve Leakage Control Systems).

These concerns led EPRI to propose an alternative approach to ensure that doses associated with MSIV leakage would be acceptably low. EPRI identified this issue as a plant optimization subject. The resolution proposed by EPRI would eliminate the safety-related leakage control system, allow higher leakage limits through the MSIVs, and use an alternate MSIV leakage treatment method that takes credit for the large surface volume in the main steam piping and condenser hotwell to plate-out the fission products following core damage. In this way, the main steam piping (and by-pass line) and the condenser are used to mitigate the consequences of an accident. Appendix A to 10 CFR Part 100 requires that structures, systems, and components necessary to assure the capability to mitigate the consequences of accidents remain functional during and after a safe shutdown earthquake (SSE). These components are classified as safety-related and Seismic Category I. In addition, Appendix B to 10 CFR Part 50 establishes quality assurance requirements for safety-related, seismic Category I systems, structures, and components.

### Discussion

Section 3.2.2 of the SRP recommends that the main steamline from the outermost isolation valve up to, but not including, the turbine stop valve including branch lines up to the first valve, be classified as Quality Group B (Safety Class 2). Regulatory Guide 1.29 designates such piping as Seismic Category I. The staff concludes that the main steam piping from the outermost isolation valve up to the seismic interface restraint and branch lines up to the first closed valve should conform to Appendix A of Section 3.2.2 of the SRP and Regulatory Guide 1.29. The main steamline from the seismic interface restraint up to but not including the turbine stop valve should be classified as Quality Group B but may be classified as non-seismic Category I. However, all pertinent quality assurance requirements of Appendix B to 10 CFR Part 50 are applicable to this portion of the main steamline from the seismic interface restraint to the turbine stop valve. These requirements are needed to assure that the quality of the piping material is commensurate with its importance to safety during both operational and accident conditions.

The seismic interface restraint must provide a structural barrier between the Seismic Category I portion of the main steamline in the reactor building and the non-seismic Category I portions of the main steamline in the turbine building. The seismic interface restraint must be located inside the Seismic Category I building. The classification of the main steamline in the turbine building as non-seismic Category I is needed for consistency with the classification of the turbine building. On this basis, the quality and safety
requirements imposed on the main steamline from the outermost isolation valve up to the turbine stop valve are equivalent to the staff guidelines in Appendix A to Section 3.2.2 of the SRP and Regulatory Guide 1.29.

To ensure the integrity of the by-pass piping from the first valve to the main condenser hotwell, the staff and EPRI both agree that preventing gross structural failure of the piping and hotwall would provide assurance that leakage from the MSIVs following a design basis accident would not exceed the 10 CFR Part 100 guideline. The issue remaining is the classification of the main steam by-pass piping between the first normally-closed valve and the condenser hotwell as well as the hotwell itself. The staff proposes that the main steam by-pass line from the first valve up to the condenser inlet, and the piping between the turbine stop valve and the turbine inlet should not be classified as safety-related nor as seismic Category I, but should be analyzed using a dynamic seismic analysis to demonstrate its structural integrity under SSE loading conditions.

The staff proposes that the condenser be seismically analyzed to ensure that it is capable of maintaining its structural integrity during and after the SSE. The dose analysis considers that the condenser is open to atmosphere. Thus, it is only necessary to ensure no gross structural failure of the condenser. Similarly, it is only necessary to ensure that failure of nonsafety related systems, structures, and components due to a seismic event do not cause failure of the main steam piping, by-pass line, or the condenser. The staff is developing design acceptance criteria for seismic analyses of piping systems which would become part of the inspections, tests, analyses, and acceptance criteria (ITAAC) for the COL holder to perform (see Section II.L of this enclosure). The staff proposes a similar approach for the above non-safety related systems, structures, and components.

#### Recommendations

The staff concludes that the above-described approach for both evolutionary and passive ALWRs to resolve the BWR main steam line issue provides reasonable assurance that the main steam piping from the outermost isolation valve up to the turbine stop valve, the main steam by-pass line up to the condenser, and the main condenser will retain their pressure and structural integrity during and following a safe shutdown earthquake. The proposed credit for non-safety related components (e.g., main steam by-pass piping and condenser) to mitigate the consequences of a design basis accident might require an exemption from the regulations. The staff recommends that the Commission approve the abovedescribed approach for the resolution of the main steam line classification for both the evolutionary and passive ALWRs.

#### F. Tornado Design Basis

The current NRC regulatory position with regard to design basis tornados is contained in two documents written in 1974, WASH-1300, "Technical Basis for Interim Regional Tornado Criteria," and Regulatory Guide 1.76, "Design Basis Tornado for Nuclear Power Plants." WASH-1300 states that the probability of occurrence of a tornado that exceeds the Design Basis Tornado (DBT) should be on the order of 10E-7 per year per nuclear power plant and the regulatory guide delineates the maximum wind speeds of 240 to 360 per Pair (pending on the regions.

As a result of EPRI's earlier efforts on this EPRI-proposed plant optimization subject, the regulatory positions in Regulatory Guide 1.76 were reevaluated by an NRC contractor using the considerable quantity of tornado data which is now available but was not when the regulatory guide was developed. The contractor completed its reevaluation as discussed in NUREG/CR-4661 (PNL-9697), "Tornado Climatology of the Contiguous United States," dated May 1986. At the heart of this study is the tornado data tape prepared by the National Severe Storm Forecast Center (NSSFC) with 30 years of data, 1954 through 1983. This data tape contains the data for the approximately 30,000 tornados that occurred during the period.

The contractor found that the tornado strike probabilities range from near 10E-7 per year for much of the western United States to about 10E-3 per year in the central United States. Based on discussions between the contractor and the staff, wind speed values associated with a tornado having a mean recurrence interval of 10E-7 per year were estimated to be about 200 mph for the United States west of the Rocky Mountains and to be 300 mph for the United States east of the Rocky Mountains.

In its December 6, 1991 letter, EPRI proposed that a maximum tornado wind speed of 300 mph and a tornado recurrence interval of 10E-7 per year tornado strike probability be used for the design basis tornado to be used in the design of the evolutionary ALWRS. During a January 30, 1992 meeting with the staff, EPRI indicated that it would delete the reference to the tornado recurrence interval from the Requirements Document. The evolutionary and passive ALWR designers have indicated that their applications will be consistent with the EPRI Requirements Document.

The tornado design basis requirements have been used in establishing structural requirements (e.g., minimum concrete wall thicknesses) for the protection of nuclear plant safety-related structures, systems, and components against the effects not covered explicitly in review guidance such as Regulatory Guides or the SRP. Specifically, some aviation (general aviation light aircraft) crashes, nearby explosions, and explosion debris or missiles have been reviewed and evaluated routinely by the staff by taking into account the existence of the tornado protection requirements. The staff's acceptance of EPRI's proposal will also necessitate a concurrent review and evaluation of their effect on the protection criteria for some external impact hazards, such as general aviation c. nearby explosions. Therefore, external impact hazards will be reviewed on a site-specific basis.

Based on the updated tornado data and the analysis provided in NUREG/L-4661, the staff concludes that it is acceptable to reduce the tornado destgn basis wind speeds to 200 mph for the United States west of the Rocky Mountains, and to 300 mph for the United States east of the Rocky Mountains. <u>Therefore, the</u> staff recommends that the Commission approve the staff's position that a

1B.C-37

- 27 -

maximum tornado wind speed of 300 mph be used for the design basis tornado to be used in the design of evolutionary and passive ALWR designs. As part of the COL process, the applicant will have to demonstrate that a design capable of withstanding a 300 mph tornado will also be sufficient to withstand other site hazards.

The staff expects that the use of this criteria will not preclude siting the ALWR plant designs on most sites in the United States. However, should an actual site hazard exceed the design envelope in a certain area, a specific analysis will have to be performed to verify that the design is still acceptable for that site.

#### G. Containment Bypass

The phenomenon of containment bypass is associated with the failure of the containment system to channe) fission product releases through the suppression pool, or the failure of passive containment cooling system heat exchanger tubes in the large pools of water outside the containment. The fundamental characteristic of a BWR pressure-suppression containment is that steam that is released from the reactor coolant system will be condensed and scrubbed of radionuclides in a pool of water (the suppression pool) and the pressure rise in the containment will thereby be limited. This is accomplished by directing the steam from the reactor coolant system to the suppression pool through a vent system, but leakage paths could exist in this pathway between the drywell and the wetwell airspace that could allow steam to bypass the suppression pool and might over-pressurize the containment. Potential sources of steam bypass include leakage through the vacuum relief valves, cracking of the drywell structure, and penetrations through the drywell structure. In addition, a containment design which uses an external heat exchanger carries with it the potential for containment bypass from a leak in the heat exchanger. High temperatures associated with severe accidents or core debris carried from the reactor vessel could threaten the integrity of the heat exchanger tubes, and therefore provide a pathway for the release of fission products. Bypass of internal suppression pools could lead to overpressurization of the containment, and threaten its integrity. The staff believes that vendors should make reasonable efforts to minimize the possibility of bypass leakage, and should also allow for a certain amount of leakage in the containment design.

The provision of containment sprays in the drywell and/or wetwell would also reduce the impact of suppression pool bypass leakage on containment performance. These systems spray water into the containment and lower its temperature and pressure. They also scrub the containment atmosphere of fission products, and mitigate the effects of bypass on fission product distribution. In view of the contribution they can make to accident management, the staff is evaluating the need for containment spray systems for all ALWRs.

The discussion on steam bypass of the suppression pool is for information only and is provided to identify a complete list of issues applicable to the advanced designs. If a policy question is identified as a result of the

10

staffs' review, the Commission will be informed of the issue at the earliest opportunity. Therefore, the staff does not request any Commission action on this issue at this time.

#### H. Containment Leak Rate Testing

EPRI proposed that the maximum interval between Type C leakage rate tests should be 30 months rather that the 24 month maximum interval currently required in Appendix J to 10 CFR Part 50 for both evolutionary and passive plant designs. This plant optimization subject was proposed by EPRI to allow some margin between the nominal 24 month refueling interval and the Type C test interval to ensure that plant shutdowns will not be required solely to perform Type C tests. Other issues (air lock testing, Type C leak testing methods) have also been raised, but have not been raised to the Commission as a policy question.

In parallel with the staff's review of this issue on the ALWRs, the staff has developed proposed changes to Appendix J of 10 CFR Part 50 for all reactors, which were sent to the Commission in SECY-91-348. This document proposes modification to the regulation that would allow the increased interval as well as address other issues raised by EPR1. The staff's justification for these modifications is provided in SECY-91-348.

In addition, the staff has extended the time interval for performing Type C leakage rate testing on currently operating plants on a case-by-case basis (by as much as 1 year).

Therefore, the staff recommends that the Commission approve the staff's position that, until the rule change proceedings for Appendix J of 10 CFR 50 are completed, the maximum interval between Type C leakage rate tests for both evolutionary and passive plant designs be 30 months rather than the 24 month maximum interval currently required in Appendix J to 10 CFR Part 50.

3

1. Post-Accident Sampling System

10 CFR 50.34(f)(2)(viii) requires the designer to provide the

...capability to promptly obtain and analyze samples from the reactor coolant system and containment that may contain TID-14844 source term radioactive materials without radiation exposures to any individual exceeding 5 rem to the whole body or 75 rem to the extremities. Materials to be analyzed and quantified include certain radionuclides that are indicators of the degree of core damage(e.g., noble gases, iodines and cesiums, and non-volatile isotopes), hydrogen in the containment atmosphere, dissolved gases, chloride, and boron concentrations.

Regulatory Guide 1.97 and NUREG-0737, "Clarification of TMI Action Plan Requirements," provide guidance regarding the design of the post-accident sampling system (PASS) used to implement this regulation. EPRI has proposed deviation from the design requirements for the PASS in several areas, as discussed below. EPRI has identified this issue as an optimization subject.

#### Elimination of the Hydrogen Analysis of Containment Atmosphere Samples

EPRI has stated that the hydrogen analysis of the containment atmosphere can be accomplished by the safety-grade containment hydrogen monitor required by 10 CFR 50.34(f)(2)(xvii) and 11.F.1 of NUREG-0737. The staff concludes that adequate capability for monitoring post-accident hydrogen is provided by the safety-grade instrumentation, and that this is acceptable justification for an ALWR vendor to use in requesting this deviation. Because this exemption has been granted previously on currently operating plants, the staff does not consider this request to be a policy matter.

#### Elimination of Dissolved Gas and Chloride Analyses of Reactor Coolant Samples

EPRI considers the analyses of the reactor coolant for dissolved gas and chloride to be unnecessary because gases accumulated in the reactor vessel (mainly hydrogen) will be removed by venting, and corrosion due to the presence of chloride and oxygen will be minimized by prompt depressurization and cooling. Additionally, the amount of dissolved hydrogen in the reactor coolant can be determined based upon the hydrogen concentration measured in the containment atmosphere. 10 CFR 50.34(f)(2)(vii) and Item 11.8.3 of NUREG-0737 specify that the PASS should have the capability to analyze dissolved hydrogen, oxygen, and chloride. This requirement was formulated before reactor vessels were required to have vents. With vented reactor vessels, the information on hydrogen concentration became less important.

However, the staff concludes that even with vented reactor vessels, there are some postulated accident sequences in which the reactor coolant system (RCS) is intact at reduced pressure, and heat is being removed by subcooled decay heat removal, e.g., similar to the TMI-2 accident. For these cases, it will not be possible to evaluate concentrations of the dissolved gases in the reactor coolant from their concentrations measured in the containment. The information on the amounts of dissolved hydrogen, chloride and oxygen in the reactor coolant is an important factor in evaluating post-accident conditions existing in the reactor vessel. The presence of hydrogen can affect flow of coolant in the core, and chloride and oxygen can cause corrosion of reactor components that may still be significant even at a low reactor coolant tempersture. Therefore, the requirement for PASS sampling of coolant should not be eliminated. However, the staff agrees that sampling Z4 hours after the end of power operation would be adequate to assure long term decay heat removal.

Therefore, the staff recommends that the Commission approve the staff's position that the capability to analyze dissolved hydrogen, exycen. and chloride in accordance with the requirements of 10 CFR 50.34(f)(2)(viii) and 11.B.3 of NUREG-0737 be required of the post-accident sampling system for the evolutionary and passive ALWRs. However, the time for taking these samples can be extended to 24 hours after the accident.

44

#### Relaxation in the Time Requirement for Sampling Activity Measurements

EPRI states that if boron solution has been added to permit plant shutdown, reactor water samples can be taken for boron analyses starting eight hours after the end of power operation. EPRI states the samples for activity measurements will not be required for 24 hours after the accident.

- 31 -

Item II.B.3 of NUREG-0737 specifies that the PASS should have the capability to obtain coolant and containment atmosphere sampling results within three hours from the time after the accident. The purpose of this requirement is to ensure that the capability exists for sampling in cases when samples are needed while the accident is still in progress. EPRI has committed that the neutron flux monitoring instrumentation will comply with the Category criteria of Regulatory Guide 1.97. Therefore, this instrumentation will have fully qualified, redundant channels that monitor over the power range of 10E-6 percent to full power. Based on this commitment, the staff concurs with EPRI that boron sampling will not be required for the first 8 hours after an accident. Samples for activity measurements are required for evaluating the condition of the core. During the accident management phase, this information will be provided by the containment high range area radiation monitor, the containment hydrogen monitor, the reactor vessel water level indicator for BWRs, and the core exit thermocouples for PWRs. These data will be sufficient to meet the needs of the plant operators for the first 24 hours after an accident. The need for PASS activity measurements will occur during the accident recovery phase when the degree of core damage and general plant contamination will have to be evaluated. Based on this justification, the staff concludes that the requested extension of time for sampling activity measurements 24 hours after an accident is acceptable.

Therefore, the staff recommends that the Commission approve the staff's position that the requirements for sampling reactor coolant for boron and for activity measurements using the post-accident sampling system in the evolutionary and passive ALWRs deviate from the requirements of Item II.B.3 of NUREG-0737. The modified requirement would require the capability to take samples for boron and for activity measurements B hours and 24 hours, respectively, after the end of power operation.

#### J. Level of Detail

In its February 15, 1991 SRM on SECY-90-377, the Commission provided guidance regarding the level of detail of information required to determine the adequacy of design certification applications under 10 CFR Part 52. Although the level of detail issue is applicable to all design certification applications, the staff has been reviewing the ABWR as the lead plant in resolving this issue.

The staff identified several areas in the ABWR application where additional information is needed in order to resolve its safety concerns. The design detail resulting from the resolution of all of the staff's safety concerns will constitute the level of detail needed to support design certification in

1B.C-41

accordance with the SRM on SECY-90-377. The level of design detail ultimately required by the staff could affect the schedules for all of the standard plant applications that the NRC has received to date.

In a meeting with GE, senior NRC managers and the vendor discussed certain areas of the review for which the designer has not provided final design details. These areas include piping design, leak-before-break, control room design, radiation protection, and advanced instrumentation and controls. The staff and GE agreed to pursue the development of design acceptance criteria with associated NRC "check points" as a substitute for detailed design information for a few limited areas of the design. To accomplish this, the NRC safety determination at design certification would be based on acceptance criteria that are general in nature, lut measurable and observable. The "check points" would serve as milestones to confirm compliance with system requirements and the acceptance criteria. These issues would be documented in the Safety Analysis Report and the inspections, tests, analyses, and acceptance criteria (ITAAC), as appropriate.

The staff concludes that the level of detail issue is applicable to all design certification applications, but expects it to be resolved in the context of the ABWR review. The discussion provided in this section is for information only, and is provided to identify a complete list of issues applicable to the passive designs. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

#### K. Prototyping

SECY-91-074 discussed the process of the staff will use for determining the need for a prototype or other demonstration facility for the advanced reactor designs. The staff stated it will follow the procedure outlined in the paper to determine the various types of testing, up to and including a prototype facility, that may be needed to demonstrate that the advanced reactor designs are sufficiently mature to be certified.

Because the need for prototype testing is a design specific issue, this issue cannot be resolved during the EPRI review. The staff is in the process of evaluating the submittals of the evolutionary ALWRs to determine the need for prototype testing, but has not identified any areas that may require such testing yet.

As discussed in SECY-91-273, the necessity for separate effects and scaled integral testing for passive designs is under consideration.

The discussion on prototyping provided in this section is for information only, and is provided to identify a complete list of issues applicable to the passive designs. If a policy puestion is identified as a result of its review, the staff will inform the (Immission of the issue at the earliest opportunity. Therefore, the staff does not request any Commission action on this issue at this time. Inspections, Test, Analyses, and Acceptance Criteria (ITAAC)

#### 10 CFR 52.79(c) requires that

The application for a combined license must include the proposed test, inspections, and analyses which the licensee shall perform and the acceptance criteria therefor which are necessary and sufficient to provide reasonable assurance that, if the tests, inspections, and analyses are performed and the acceptance criteria met, the facility has been constructed and will operate in conformity with the combined license. Where the application references a certified standard design, the tests, inspection, analyses, and acceptance criteria contained in the certified design must apply to those portions of the facility design which are covered by the design certification.

SECY-91-178 provided the staff's recommendations concerning the form and contert of the inspections, test, analyses, and acceptance criteria (ITAAC) for a design certification role and a combined license as required by 10 CFR Part 52. SECY-91-210 requested Commission guidance on an industry proposal that would allow the NRC staff to issue standardized plant final design approvals (FDAs) before the final staff approval of the proposed ITAAC. In its September 24, 1991 SRM, the Commission provided guidance regarding development of ITAACs for final design approval/design certification applications.

GE has been identified by the Nuclear Management and Resources Council (NUMARC) as the industry lead for developing ITAAC submittals. The staff will review the ITAAC submittals when they are received from the vendors, currently expected from GE in January 1992 for the ABWR. CE plans to submit the ITAAC for the System 80+ following GE's submittal, currently scheduled for May 1992.

The staff concludes that the ITAAC issue is applicable to all final design approval/design certification applications, but expects it to be resolved in the context of the ABWR review. The discussion provided in this section is for information only, and is provided to identify a complete list of issues applicable to the passive designs. The staff's response to the September 24, 1991 SRM is scheduled for completion in March 1992. The staff will continue to interact with the industry on this matter. Should additional policy questions be raised as a result of the staff's review, the staff will inform the Commission at the earliest opportunity. Therefore, the staff does not request any new Commission action on this issue at this time.

#### M. Reliability Assurance Plan (RAP)

In SECY-89-013, the staff stated that a program to ensure that the design reliability of safety significant systems, structures, and components is maintained over the life of a plant, referred to as the reliability-assurance program (RAP), would be required for design certification. The ALWR vendors and EPRI were informed that the staff was considering this matter in November 1988. The need for a safety-oriented reliability effort for the nuclear industry was initially expressed by the NRC in Section II.C.4 of NUREG-0600, "NRC Action Plan Developed as a Result of the TMI-2 Accident," dated August 1980. Initial research in this area focused on enhancing the reliability of safety systems and their supporting auxiliary systems. Follow-on research examined the feasibility and cost effectiveness of applying an aerospace-proven reliability concept to commercial nuclear reactors as a regulatory option to address the issue concerning anticipated transients without scram (ATWS). Nork on this project continued until March 1985 when the research on operational safety reliability was consolidated into the research efforts on risk-based technical specifications.

Other research was performed for the NRC in support of operating and new reactor licensing review activities. NASA performed research that involved the application of NASA's Systems Assurance Analysis, which addressed reliability, safety, and quality concerns, to selected nuclear plant systems. In addition, as part of the license application review process for the Clinch River Breeder Reactor Program (CRBR), the NRC reviewed the CRBR RAP that was proposed by the Department of Energy.

The staff is working on the development of a detailed guidance document for the development of a reliability assurance program for ALWRs as well as reviewing the ALWR vendor's final design approval/design certification applications and the EPRI Requirements Document for evolutionary and passive plant designs. The staff views the RAP for ALWRs as a program that exists at two distinct levels: the first level applies to vendor submittals for final design approval/design certification; and the second level is applicable to a referencing applicant for a construction and operative ficense. The first level involves a top-level program that defines the scope, conceptual framework, and essential elements of an effective RAP. The second level fully develops and implements the program based on the plant-specific design information.

The staff is working with EPRI and the ALWR vendors on the development of the first level RAP for ALWRs. Should policy questions be raised as a result of the staff's review, the staff will inform the Commission at the earliest opportunity. The discussion provided in this section is for information only, and is provided to identify a complete list of significant issues applicable to the ALWR designs. In accordance with the Commission's November 12, 1991 SRM, the staff will include a discussion of issues concerning the reliability assurance program in its periodic updates on the advanced reactor program. Therefore, the staff does not request any Commission action this issue at this time.

N. Site-Specific Probabilistic Risk Assessments (PRAs)

10 CFR 52.47 requires all applicants seeking standard design certification to provide a probabilistic risk assessment (PRA). However, details of the specific site characteristics where a plant would be sited are not required until the combined operating license (CDL) licensing stage. Rather than

1B.C-44

provide a PRA that envelopes all potential sites for all hazards (e.g., volcanism or external flooding on a river site), the staff has sought enveloping analyses for seismic events and tornadoes from the final design approval/design certification applicants, and concludes that COL applicants should provide site-specific PRA information that deals with other potential hazards specific to the particular site. Therefore, the staff recommends that the <u>Commission approve the staff's position that site-specific PRA information be</u> <u>submitted at the COL stage that addresses applicable site-specific hazards</u> <u>such as river flooding, storm surge, tsunami, vulcanism, and hurricanes, and</u> <u>that enveloping analyses for seismic events and tornadoes be required from the</u> <u>final design approval/design certification applicants</u>.

O. Severe Accident Mitigation Design Alternatives (SAMDAs)

As discussed in SECY-91-229, the National Environmental Policy Act (NEPA), Section 102(C)(iii) requires, in part, that

...all agencies of the Federal Government shall...(C) include in every recommendation or report on proposals for legislation and other major Federal actions significantly affecting the quality of the human environment, a detailed statement by the responsible official on...(iii) alternatives to the proposed action.

The U.S. Court of Appeals, in <u>Limerick Ecology Action v. NRC</u>, 869 F.2d 719 (3rd Cir. 1989), effectively required the NRC to include consideration of certain severe accident mitigation design alternatives (SAMDAs) in the environmental impact review performed as part of the operating license application for the Limerick Generation Listion. The staff has concluded that a NEPA evaluation in the form of an environmental impact statement that considered SAMDAs would be an essential element of an application for a combined license under Subpart C of 10 CFR Part 52, for those applications that reference a design certified under Subpart B.

Therefore, in SECY-91-229, the staff recommended that the Commission approve the staff's recommendations to

- 1. address SAMDAs for certified designs in a single rulemaking,
- approve the staff's approach for considering the costs and benefits of the review of SAMDAs for standard plant design certification, and
- approve the staff's proposal to advise applicants for a final design approval/design certification that they must assess SAMDAs and provide the rationale supporting their decisions.

In its October 25, 1991 SRM, the Commission approved the staff's recommendations and requested they be kept informed on the staff's progress in its evaluation of the SAMDAs for final design approval/design certification applications. Consistent with the third recommendation of SECY-91-229, the staff requested the ALWR vendors to assess SAMDAs for their designs and provide their rationale for determining whether the SAMDAs would improve the safety of those designs. The staff will evaluate the applicants' responses to these inquiries upon receipt of their submittals.

The staff concludes that the SAMDAs issue is applicable to all final design approval/design certification applications, but expects it to be resolved in the context of the ABWR and System 80+ reviews. The discussion provided in this section is for information only, and is provided to identify a complete list of issues applicable to the passive designs. The staff will continue to interact with the industry on this matter. Should additional policy questions be raised as a result of the staff's review, the staff will inform the Commission at the earliest opportunity. Therefore, the staff does not request any Commission action on this issue at this time.

#### P. Generic Rulemaking Related to Design Certification

n

SECY-91-262 provided the Commission with the staff's recommendations regarding generic rulemaking related to design certification. The staff recommended that the Commission

- approve the staff's proposal to proceed with the design-specific rulemakings through individual design certifications to resolve selected technics; and severe accident issues for the ABWR and System 80+ designs, and
- note the staff's intent to proceed with generic rulemaking where appropriate for evolutionary and passive designs, as information becomes available from ongoing efforts on these issues, independent of the design review and certification processes.

The staff has not yet received Commission guidance on SECY-91-262.

As discussed in SECY-91-262, the staff concludes that consideration of generic rulemaking in lieu of design-specific rulemaking is applicable to all final design approval/design certification applications. However, the design of the passive plants is not sufficiently developed at this time for the staff to determine whether generic rulemaking should be initiated for the passive plant designs. Certain generic rulemaking activities related to the evaluations of source terms during postulated severe accidents are ongoing and the results of these rulemakings may be used during design certification of the passive reactor designs. Currently, this work is focused on updating 10 CFR Part 100 to separate siting criteria from reactor design criteria. The staff plans to incorporate the revised source term criteria in 10 CFR Part 50, and, in addition, to later consider the incorporation of generic severe accident criteria. The discussion provided in this section is for information only, and is provided to identify a complete list of issues applicable to the passive designs. Should additional policy questions be raised as a result of the staff's review, the staff will inform the Commission at the earliest opportunity. Therefore, the staff does not request any new Commission action on this issue at this time.

#### III. Passive Design Issues Only

The following is a discussion of the issues that the staff has identified while reviewing the conceptual design information from the passive plant vendors in conjunction with the EPRI Requirements Document for passive designs. These issues are unique to the passive plant designs.

#### A. Regulatory Treatment of Non-Safety Systems

In contrast to both the current generation of light water reactors and the evolutionary ALWRs, the passive ALWR designs rely on safety systems which use the driving forces of buoyancy, gravity, and stored energy sources. These passive systems supply safety injection water, provide core and containment cooling, and perform other functions. There are no pumps in these passive safety systems, and all valves are powered by dc electric power from batteries, are air-operated, or use check valves actuated by the pressure differential across the valve. EPRI and the passive reactor vendors propose that these designs do not include safety-grade ac electric power.

The passive ALWR designs also include non-safety-grade active systems to provide defense-in-depth capabilities for reactor coolant makeup and decay heat removal. These systems serve as the first line of defense in the event of transients or plant upsets to reduce challenges to the passive systems. These active systems include: (1) the chemical and volume control system and control rod drive system, which provide reactor coolant makeup for the AP-600 and SBWR, respectively, (2) the reactor shutdown cooling system and backup feedwater system for PWR decay heat removal, and the reactor water cleanup system for BWR decay heat removal, (3) the fuel pool cooling and cleanup system for spent fuel decay heat removal, and (4) the associated systems and structures which are needed to support these functions, including non-safety standby diesel generators. In addition, the passive ALWR designs also include non-safety-grade active systems, such as the control room heating, ventilating, and air conditioning system, for mitigation of the radiological conse-quences of an accident. Many of these systems traditionally have been safety-grade systems, but in the passive plants, they are not designed to meet safety-grade criteria, and credit is not taken for them in the Chapter 15 licensing design basis accident analyses. In SECY 90-406, the staff identified the role of these non-safety systems in the passive designs as an emerging technical issue.

Associated with the new, passive design approach, the licensing design basis analysis relies solely on the passive safety systems to demonstrate compliance with the acceptance criteria for various design basis transients and accidents. However, uncertainties remain concerning the performance of the unique passive features and overall performance of core and containment heat removal because of lack of a proven operational performance history. For example, there are uncertainties about the performance of check valves in the passive safety systems which operate at low differential pressures provided by natural circulation or gravity injection. These low pressures may not provide sufficient force to fully open sticking check valves (i.e., pumped emergency core cooling systems (ECCS) systems are more likely to overcome stuck valves). These uncertainties enhance the importance of the active non-safety systems in providing the defense-in-depth to prevent and mitigate accidents and core damage. Therefore, the staff's review of the passive designs requires a review of not only the passive safety systems, but also the functional capability and availability of the active non-safety systems to provide significant defense-in-depth and accident and core damage prevention capability.

For those active systems that perform defense-in-depth functions, the EPRI Requirements Document for passive designs specifies performance and systems and equipment design requirements. These include radiation shielding requirements to permit access following an accident, redundancy, the availability of non-safety-grade electric power, and protection against internal hazards, as well as safety analysis and testing to demonstrate system capability to satisfy defense-in-depth considerations. Although it does not provide specific requirements for the reliability of these systems, EPRI, in response to staff questions, has indicated that it is evaluating specific reliability targets and other measures to provide confidence that the passive plants will meet performance requirements, and that EPRI will address both passive safely and active non-safety systems.

In addition, technical specification development is a subset of the overall regulatory treatment of the passive designs. The staff is evaluating the need to establish reliability-based Technical Specifications for passive designs to determine which systems and components (including certain non-safety systems) will require the imposition of technical specifications, and the parameters of the technical specifications (length, surveillance, etc.). The Reliability Assurance Program is expected to strongly influence the technical specifications.

Since the passive ALWR design philosophy departs from current licensing practices, new regulatory and review guidance is necessary so that the staff can appropriately review the AP-600 and SBWR submittals expected in mid-1992. Significant decisions need to be made concerning the scope of staff review of the non-safety systems and reliance on the passive safety systems. The staff may not require that the active systems meet all the safety-grade criteria, but there should be a high level of confidence that these active systems are available when needed in their defense-in-depth roles. The staff is conducting a sample evaluation of the passive designs to identify the safety importance of the passive safety and active non-safety systems. Using the results of the studies and interaction with EPRI, the staff will develop a review scope for these active non-safety systems, and will establish functional performance requirements and acceptance criteria to ensure they have adequate capability and availability when called upon. The staff is still evaluating this issue for the passive plant designs. The discussion provided in this section is to provide a status of the issue for the Commission, and is provided to identify a complete list of issues applicable to the passive designs. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

#### B. Definition of Passive Failure

Appendix A to 10 CFR 50 states that any applicant must design against single failure of passive components in fluid systems important to safety, where a single failure is defined as an occurrence which results in the loss of capability of a component to perform its intended safety functions. Fluid and electric systems are considered to be designed against an assumed single failure if neither (1) a single failure of any active component (assuming passive components function properly) nor (2) a single failure of a passive component (assuming active components function properly) results in a loss of the capability of the system to perform its safety functions. However, the introduction to Appendix A to 10 CFR Part 50 notes that the conditions under which a single failure of a passive component in a fluid system should be considered in designing the system against a single failure are under development.

SECY-77-439 describes how the staff was using the single failure criteria in its reactor safety review process. As discussed in that paper, an active failure in a fluid system means (1) the failure of a component which relies on mechanical movement for its operation to complete its intended function on demand, or (2) an unintended movement of the component. Examples include the failure of a motor- or air-operated valve to move or to assume its correct position on demand; spurious opening or closing of a motor- or air-operated valve; or the failure of a pump to start or to stop on demand. In some instances, such failures can be induced by operator error.

A passive failure in a fluid system means a breach in the fluid pressure boundary or a mechanical failure which adversely affects a flow path. Examples include the failure of a simple check valve to move to its correct position when required; the leakage of fluid from failed components, such as pipes and valves - particularly through a failed seal at a valve or pump; or line blockage. Motor-operated valves which have the source of power locked out are allowed to be treated as passive components.

In past licensing reviews, the staff has been inconsistent in its treatment of passive failures in fluid systems in that in certain cases, it imposed a passive failure in addition to the initiating event while in others it did not. The staff has determined that in most instances, the probability of most types of passive failures in fluid systems is sufficiently small that they need not be assumed in addition to the initiating failure in application of the single failure criterion to assure the safety of a nuclear power plant.

In particular, staff practice has normally been to treat check valves, except for containment isolation systems, as passive devices rather than active devices during transients or design basis accidents. However the staff is considering redefining failure of check valves to that of an tive failure because safety-related check valves in the passive designs a to perate under different conditions (low flow and pressure without pump pres. to open valves) than current generation reactors and evolutionary designs, they have increased safety significance to the operation of the passive safely systems, and operating experience has shown that they have a lower reliability than originally anticipated. This would cause these valves to be evaluated in a more stringent manner than that of previous licensing reviews.

The staff is still evaluating this issue for the passive plant designs. It is identifying this issue in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest opportunity. The staff's proposed resolution 7 this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

#### C. SBWR Stability

In BWRs, thermal-hydraulic instabilities can cause oscillations that can result in violation of the minimum critical power ratio (MCPR) safety limits. The staff has concluded that GE's analytical codes have been sufficiently validated to demonstrate the stability of the ABWR design. However, the codes that GE is using have not yet been adequately validated for the passive BWR design.

As discussed in SECY-91-273, the staff has determined that an early NRC assessment is needed of the vendor's analytical and experimental basis for demonstrating nuclear/thermal-hydraulic stability, and to identify any tests or analyses that may be needed to support staff technical evaluations of the issue. The NRC staff and its consultant, Oak Ridge National Laboratories (ORNL), have performed a review of the thermal-hydraulic stability characteristics of the SBWR based on preliminary design information provided by GE. This assessment included calculations with the LAPUR computer code developed by NRC/ORNL which showed that, while the system appears to be very stable under normal operating conditions, there are abnormal conditions of operation that might be reached under credible transient sequences that can result in the onset of density-wave power and flow oscillations. In addition, a low flow and low power instability due to a geysering effect between parallel channels has been identified as a concern during normal operating transients such as start-up and shutdown.

On December 6, 1991, the staff mel with EPRI and GE to discuss the EPRI/GE resonse to the staff's conclusions that more extensive SBWR stability studies are needed and that codes which have been validated against thermal-hydraulic tests representative of the SBWR design, including the large open chimney, would be needed to perform these studies. EPRI and GE informed the staff that the chimney design has been changed and that existing experiments are repre-

sentative of the divided chimney now employed. GE indicated it will validate its codes for density-wave instability studies against these experiments and will provide the results of this work for NRC review. The geysering instability is being studied using small-scale experiments performed by a Japanese partner to GE. The Japanese SAFAR code will be validated against these experiments and used for analytical prediction of stable operating boundaries. GE plans to recommend start-up/shutdown procedures similar to those used in the Dutch Dodewaard reactor to avoid geysering instability. EPRI and GE believe that the SBWR is not vulnerable to a loop-type instability also reported by the Japanese; rather, it was characteristic of the experimental apparatus used.

GE has modified the SBWR conceptual design and has identified existing experimental data which they believe to be appropriate validation of codes to be used for stability studies. EPRI and GE have indicated agreement with the staff that such studies will be needed to confirm the stability behavior of SBWR during verious transient cenarios (including ATWS). However, GE has not provided sufficient information to permit NRC evaluation of the applicability and fficiency of the foreign experiments they have identified for use during code validation. The vendor has agreed to make this information available to the NRC as soon as it obtains permission from the foreign sources, possibly in early 1992. Until these experiments can be reviewed by NRC, the potential need for additional experiments to support stability evaluations for design certification remains open. The staff is identifying this issue in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest opportunity. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

#### D. Safe Shutdown Requirements

General Design Criterion (GDC) 34 requires that a residual heat removal (RHR) system be provided to remove residual heat from the reactor core so that specified acceptable fuel design limits (SAFDLs) are not exceeded. Regulatory Guide (RG) 1.139 and Branch Technical Position (BTP) 5-1 implement this requirement and set forth conditions to cold shutdown (200°F for a PWR and 212°F for a BWR) using only safety grade systems within 36 hours. The RG presents the basis for this requirement, as follows:

...even though it may generally be considered safe to maintain a reactor in a hot standby condition for a long time, experience shows that there have been events that required eventual cooldown and long-term cooling until the reactor coolant system was cold enough to perform inspection and repairs. It is therefore obvious that the ability to transfer heat from the reactor to the environment after a shutdown is an important safety function for both PWRs and BWRs. Consequently, it is essentia: that a power plant have the capability to go from hot-standby to cold-shutdown conditions...under any accident conditions. Passive ALWR designs use passive heat removal systems for decay heat removal. They are restricted by the inherent ability of the passive heat removal processes and cannot reduce the temperature of the reactor coolant system below the boiling point of water for the heat to be transferred to the incontainment refueling water storage tank (IRWST) of the AP600 or the isolation condenser of the SBWR. Even though active shutdown cooling systems are available to bring the reactor down to cold shutdown or refueling conditions, these active RHR systems are not safety-grade, and do not comply with the guidance of RG 1.139 or BTP 5-1.

EPRI states that it is not necessary for the passive safety systems to be capable of achieving cold shutdown because they believe that the passive decay heat removal (DHR) systems have an inherent high long-term reliability. The EPRI Requirements Document for passive plant designs states that the passive ALWR designs will employ a redundant safety system, for both the hot standby and long-term cooling modes. In addition, it defines safe shutdown as 400°F. EPRI has indicated that it meets GDC 34 requirements because redundant passive decay heat removal systems can operate at full RCS pressure and place the reactor in the long-term cooling mode immediately after shutdown. Additionally, EPRI requires that operation of the plant in the ling-term cooling mode be automatic, eliminating operator actions to cool down the plant. Operation of the passive DHR system also does not require any ac power or pumps to operate. EPRI further states that the non-safety systems that will take the plant to cold shutdown conditions "...are highly reliable in their own right...and a failure in these systems would not prevent the plant from achieving cold shutdown."

The staff is currently evaluating the EPRI position with respect to this issue to assess the acceptability of their proposed alternative approach for meeting GDC 34. The long-term DHR capability of the proposed passive systems offers potential advantages over current active systems. However, there are several issues that the staff must resolve before reaching a final position on this matter, including reliability criteria for the non-safety systems which have the capability to bring the plant to cold shutdown and the acceptability of 400'F as a safe long-term state. The staff is identifying this issue in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest opportunity. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

#### E. Control Room Habitability

GDC 19 of Appendix A to 10 CFR Part 50 states that adequate radiation protection shall be provided to permit access and occupancy of the control room under accident conditions without personnel receiving radiation exposures in excess of 5 rem whole body, or its equivalent to any part of the body, for the duration of the accident. In Section 6.4 of the SRP, the staff defines this dose criterion in terms of specific whole body and organ doses (5 rem to whole body, and 30 rem each to thyroid and skin). Recently, the NRC has embraced the principal recumendations of Publication No. 26 of the International

18.0-52

Commission on Radiological Protection in the promulgation of a major revision of 10 CFR Part 20. The adoption of these recommendations, which includes use of the effective dose equivaler:, did not result in a conforming change to the dose criteria for control room operators that is described in GDC 15. In current plants, safety-grade filtered control room HVAC systems with charcoal absorbers are use. to ensure that radiation doses to operators could be maintained within the GDC 19 criteria in the event of an accident.

EPRI has proposed the exposure limit for control room operators to be 5 rem whole body, 75 rem beta skin dose, and 300 rem thyroid dose. For thyroid and beta skin doses, EPRI states that individual breathing apparatus and protective clothing will be provided if required to meet the regulatory limits. The staff has determined that EPRI's requirements for thyroid and beta skin dose (with no breathing apparatus and protective clothing) have not been adequately justified at this time. The permanent credit for the use of breathing apparatus in design basis accidents has nover been allowed. The staff concludes that taking this credit is not acceptable because the long term use of breathing apparatus will degrade control room operator performance during and following an accident.

In addition, EPRI states that the control room shall be designed such that it can be maintained during the 72-hour design basis period as the primary location from which personnel can safely operate in the case of an assumed accident and other postulated design conditions. Depending on the accident sequences, the duration of the accident can be much longer than 72 hours in design basis accidents. GDC 19 states that "adequate radiation protection shall be provided to permit access and occupancy of the control room under accident conditions...for the duration of the accident." Therefore, the staff. concludes that the analyses of control room habitability should consider the duration of the accident, and not just the EPRI-proposed 72-hour design basis period.

The EPRI Requirements Document states the control room ventilation system for passive plants will be designed when any opriate redundancy and filtration (including charcoal), and will be powered both from on-site and off-site ac power sources. When ac power is available, control room habitability is provided by the non-safety grade heating, ventilation, and air conditioning (HVAC) system. However, EPRI states that ALWR designs should not rely up the HVAC system to meet the current dose criteria described in Section the SRP. Instead, EPRI has proposed the less conservative dose criteria and the 72-lour design basis period.

The staff disagrees with EPRI on the above proposal. In order to resolve this ritter, the staff may require that EPRI and the vendors provide a high level of assurance that the control room ventilation system will be available when needed. Although the system may not need to meet all of the safety-grade criteria, it may be appropriate to allow some credit for non-safety ventilation and filtration systems based on reliability considerations. The extent of this credit will be determined as part of the staff's review of the regulatory treatment of non-safety systems as discussed in Section III.A of this enclosure. It should be noted that unlike the case of core-cooling, there is no passive safety-grade system to fall back on for control room habitability.

During a January 30, 1992 meeting with the staff, EPRI proposed an alternative approach to resolving this issue that conceptually uses a safety-grade pre-surization system able to be recharged remotely after 72 hours to provide adequate ventilation for the control room operators during transients. The staff will evaluate EPRI's alternative proposal upon receipt.

As discussed in Section I.A of this enclosure, the staff is currently reviewing the new severe accident source term proposed by EPRI in conjunction with the staff's technical update effort on the TID-14844 source term. The estimated potential radiological consequences to the pastive plant contain room operators during a severe accident will depend on the outcome of the forthcoming resolution of severe accident source term, namely chemical for of fission products, release fractions, and release timing. In addition, the control room habitability assessment is further dependent upon the fission product removal processes inside, as well as outside, of the primary containment before it reaches the control room air intake and/or the control building that houses the control room. Therefore, the staff is unable to complete its control room habitability assessment until the source term and its behavior mechanism are satisfactorily resolved.

However, the acceptable dose criteria for control room operators are not affected by the outcome of the source term review. Regardless of the outcome of the source term review, the staff concludes that the thyroid dose limit of 300 rem and the skin dose of 75 rem (with no protective clothing) specified in the Requirements Document without further technical justifications, does not neet the regulatory requirements and therefore, is not acceptable.

Therefore, the staff requests that the Commission approve the staff's positions that

- the analyses of control room habitability be based on the dose criterion defined in GDC 19 of A. endix A to 10 CFR Part 50 and Section 6.4 of the SRP (5 rem to whole body, and 30 rem each to thyroid and skin), and
- the analyses of control room habitability should be based on the duration of the accident in accordance with GDE 19 of Appendix A to 10 CFR Part 50.

In addition, the staff notes that the issue of control room habitability is closely tied with those concerning the revised source term and the regulatory treatment of non-safety systems. Should related policy questions be raised as a result of the staff's review of these matters, the staff will inform the Commission at the earliest opportunity.

#### F. Radionuclide Attenuation

EPRI and the passive ALWR designers rely on assumptions involving fission product removal processes inside containment by natural removal effects and holdup by the secondary building and piping systems. A containment spray system is not included in the EPRI Requirements Document for passive plant designs. The staff is concerned that use of the auxiliary building for holdup may require additional restrictions be placed on the auxiliary building during normal operation that the licensee may have difficulty complying with.

This issue also affects the control room habitability issue discussed in Section III.E of this report, as the industry indicates that fission products will be removed before they reach the control room air intake and/or the control building that houses the control room.

The staff is still evaluating this issue as well as the need for a containment spray system for the passive plant designs. The staff is also evaluating whether credit for the fission product attenuation in the main steamlines and condenser is appropriate for the passive BWR design because the main steamlines downstream of the main steam isolation valves and associated condenser are not designed to withstand the safe shutdown earthquake (SSE) as defined in Section 111.c of 10 CFR Part. The staff concludes that plateout of radioactive iodine on the main sceam pipe and condenser surfaces following a severe accident can realistically provide significant dose mitigation. Several technical references indicate that particulate and elemental iodines would be expected to deposit on surfaces with rates of deposition varying with temperature, pressure, gas composition, surface material, and particulate size.

These issues are identified in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest opportunity. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

### G. Simplification of Off-Site Emergency Planning

EPRI has proposed to significantly simplify off-site emergency planning for passive designs because of EPRI's estimated low probability of core damage and, in the event of a core damage accident, the assurance of containment integrity and low off-site dose. EPRI's proposal would eliminate requirements for early notification of the public, detailed evacuation planning, and provisions for exercising the off-site plan. The on-site emergency plan and limited off-site actions would be retained. EPRI has identified this matter as a plant optimization subject.

During a January 30, 1992 meeting with the staff, EPRI proposed to work with the staff to define a process for addressing simplification of emergency planning, including developing technical criteria and methods that, if met,

1B.C-55

would justify such action, and defining the process for implementation of this approach. The results of this effort would be used as input to a generic rulemaking proposal to be initiated by NUMARC.

The staff concludes that certain modifications from the emergency planning requirements of 10 CFR Part 50 and from the siting criteria in 10 CFR Part 100 may be appropriate for the passive designs based on their unique characteristics. However, an agency determination on these issues will require evaluation of detailed design information. The staff concludes that the unique characteristics of these designs should be taken into account in determining the extent of emergency planning requirements in the plume exposure pathway emergency planning zone. A plr t's ability to prevent the significant release of racioactive material or to provide very long delay times prior to a release for all but the most unlikely events should be reflected in any decision on emergency planning requirements for the passive design. However, the staff will require a high degree of assurance that all potential containment bypass accident sequences have a very low likelihood before relaxing emergency planning requirements. The lack of information concerning source term and risk precludes further staff evaluation of the emergency preparedness requirements for the passive reactor designs at this time. Moreover, the issue is complicated by the fact that the promulgation of emergency planning requirements following the TMI-2 accident was not r amised on any specific assump-tions about severe accident probability. I nce it may be, as a policy matter, that even very low calculated probability values should not be a sufficient basis for changes to emergency planning requirements.

The staff will evaluate this issue for the passive plant designs when sufficient supporting information is available. It is identifying this issue in this paper to comply with the Commission's directives to present policy matters for Commission consideration at the earliest opportunity. The staff's proposed resolution of this issue will be provided to the Commission later. Therefore, the staff does not request any Commission action on this issue at this time.

2/11/92

¥. 2. 1. 16 4

1.27

		-	Issue	
Category		Ne.	Title	Commi. 100 Papers
Ι.	<ol> <li>SECY-90-016 Issues</li> </ol>	Α.	Use of Physically-Based Source Term	86-228 88-203 89-013 89-153 89-228 89-341 90-016 90-307 90-329 90-341 90-353
		Β.	AT₩S .	89-153 89-228 90-016 90-353
		с.	Mid-Loop Operation	89-228 90-016 97 353
		D.	Station Blackout	89-013 89-153 89-228 90-016 90-329 90-353
		Ε.	Fire Protection	89-013 89-228 90-016 90-353
		F.	Intersystem LOCA	89-153 89-228 90-016 90-353

## COMMISSION PAPERS APPLICABLE TO ALWR ISSUES

EPRI Evolutionary Plant SER 1B.C-57

# COMMISSION PAPERS APPLICABLE TO ALWR ISSUES (CONTINUED)

Category		maarmader	15576	
		No. Title		Commission Papers
	SECY-90-016 Issues (cont.)	G.	Hydrogen Control	89-013 89-153 89-228 90-016 90-329 90-353
		Η.	Core-Concrete Interaction- Ability to Cool Core Debris	89-153 89-228 90-016 90-353
		1.	High Pressure Core Melt Ejection	89-228 90-016 90-353
		J.	Containment Performance	89-228 90-016 90-353 91-273
		Κ.	Dedicated Containment Vent Penetration	89-153 89-228 90-016 90-329 90-353
		٤.	Equipment Survivability	89-228 90-016 90-353
		К.	Elimination of OBE	89-013 90-016 90-329 90-353 91-135
		Ν.	In-Service Testing of Pumps and Valves	89-228 90-016 90-353 91-273

# COMMISSION PAPERS APPLICABLE TO ALWR ISSUES (CONTINUED)

 -	-		-	
 •			- 25	
 ×.	-	2.2		
		-		

******	Category	No.	Title	Commission Papers
11.	Other Evolu- tionary and Passive Design Issues	Α.	Industry Codes and Standards	91-273
		Β.	Electrical Distribution	91-078
		С.	Seismic Hazard Curves and Design Parameters	91-135
		D.	Leak-Before-Break	89-013
		Ε.	Classification of Main Steamline of BWRs	none
		F.	Tornado Design Basis	none
		G.	Containment Bypass	none
		Н,	Containment Leak Rate Testing	89-013 89-228 91-273
		1.	Post-Accident Sampling System	none
		э.	Level of Detail	90-241 90-377
		К.	Prototyping	91-074 91-273
		L.	JAATI	91-178 91-210
		Μ.	Reliability Assurance Program	89-013
		Ν.	Site-Specific PRAs	89-013
		0.	SAMDAs	91-229
		Ρ.	Generic Rulemaking Related to Design Certification	91-262

#### COMMISSION PAPERS APPLICABLE TO ALWR ISSUES (CONTINUED)

		Issue		Commission Papers
Category		No.	Title	
111.	Passive Design Issues Only	Α.	Regulatory Treatment of Non- Safety Systems	89-013 90-406
		Β.	Definition of Passive Failure	77-439
		С.	SBWR stability	89-153 91-273
		D.	Safe Shutdown Requirements	none
		Ε.	Control Room Habitability	none
		F.	Radionuclide Attenuation	none
		G.	Simplification of Off-Site Emergency Planning	88-203

#### REFERENCES

SECY-77-439, "Single Failure Criterion," August 17, 1977.

SECY-86-228, "Introduction of Realistic Source Term Estimates into Licensing," August 6, 1986.

SECY-88-203, "Key Licensing Issues Associated with DOE Sponsored Advanced Reactor Designs," July 15, 1988.

SECY-89-013, "Design Requirements Related to the Evolutionary Advanced Light Water Reactors (ALWRs)," January 19, 1989.

SECY-89-153, "Severe Accident Design Features of the Advanced Boiling Water Reactor (ABWR), \* May 10, 1989.

SECY-89-228, "Draft Safety Evaluation Report on Chapter 5 of the Advanced Light Water Reactor Requirements Document, " July 28, 1989.

SECY-89-341, "Updated Light Water Reactor (LWR) Source Term Methodology and Potential Regulatory Applications," November 6, 1989.

SECY-90-016, "Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements," January 12, 1990.

SECY-90-241, "Level of Detail Required for Design Certification Under Part 52," July 11, 1990.

SECY-90-307, "Impacts of Source Term Timing on NRC Regulatory Positions," August 30, 1990.

SECY-90-329, "Comparison of the General Electric Advanced Boiling Water Reactor (ABWR) Design and the Electric Power Research Institute's (EPRI's) Advanced Light Water Reactor (ALWR) Requirements Document, " September 20, 1990.

SECY-90-341, "Staff Study on Source Term Update and Decoupling Siting from Design," October 4, 1990.

SECY-90-353, "Licensing Review Basis for the Combustion Engineering, Inc. System 80+ Evolutionary Light Water Reactor, \* October 12, 1990.

SECY-90-377, "Requirements for Design Certification Under 10 C7R Part 52," November 8, 1990.

SECY-90-406, "Quarterly Report on Emerging Technical Concerns," December 17, 1990.

#### REFERENCES (continued)

SECY-91-074, "Prototype Decisions for Advanced Reactor Designs," March 19, 1991.

SECY-91-078, "Chapter 11 of the Electric Power Research Institute's (EPRI's) Requirements Document and Additional Evolutionary Light Water Reactor (LWR) Certification Issues," March 25, 1991.

SECY-91-135, "Conclusions of the Probabilistic Seismic Hazard Studies Conducted for Nuclear Power Plants in the Eastern United States," May 14, 1991.SECY-91-178, "Inspections, Tests, Analyses, and Acceptance Criteria (ITAAC) for Design Certifications and Combined Licenses," June 12, 1991.

SECY-91-210, "Inspections, Tests, Analyses, and Acceptance Criteria (ITAAC) Requirements for Design Review and Issuance of a Final Design Approval, " July 16, 1991.

SECY-91-229, "Severe Accident Mitigation Design Alternatives for Certified Standard Designs," July 31, 1991.

SECY-91-262. "Resolution of Selected Technical and Severe Accident Issues for Evolutionary Light Water Reactor (LKR) Designs, \* August 16, 1991.

SECY-91-272, "Role of Personnel and Advanced Control Rooms in Future Nuclear Power Plants, \* August 27, 1991.

SECY-91-273, "Review of the Vendor's Test Programs to Support the Design Certification of Passive Light Water Reactors," August 27, 1991.

SECY-91-292, "Digital Computer Systems for Advanced Light Water Reactors," September 16, 1991.

SECY-91-348, "Issuance of Final Revision to Appendix J to 10 CFR 50, and Reslated Final Regulatory Guide 1.XXX (MS 021-5), \* October 25, 1991.

ACRS LETTER RELATED TO DRAFT COMMISSION PAPER OF FEBRUARY 27, 1992 MAY 13, 1992



INITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARDS WASHINGTON, D. C. 20555

May 13, 1992

1

Mr. James M. Taylor Executive Director for Operations J.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. Taylor:

SUBJECT: ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS

During the 383rd, 384th, and 385th meetings of the Advisory Committee on Reactor Safeguards, March 5-7, April 2-4, and May 6-9, 1992, we discussed with representatives of the NRC staff the staff's positions, recommendations, and resolution schedules concerning the certification issues for evolutionary and passive light water reactors contained in the draft SECY paper dated February 7, 1992. We also had the benefit of the documents referenced. The staff requested ACRS comments on the draft SECY paper. Our comments and recommendations on some of the staff's positions are given below.

I. SECY-90-016 Issues

Item M. Elimination of Operating Basis Earthquake

Appendix A to 10 CFR Part 100 currently establishes the Operating Basis Earthquake (OBE) at a level one-half of the Safe Shutdown Earthquake (SSE). With this specification, the OBE exerts undue influence over the seismic design and requires a full spectrum analysis in addition to that of the SSE. The staff's proposal is to effectively decouple the OBE from design. We agree with the staff's recommendation.

II. Other Evolutionary and Passive Design Issues

### Item A. Industry Codes and Standards

We agree with the staff's recommendation to use the newest codes and standards that have been endorsed by the NRC in its reviews of both the evolutionary and passive plant design applications, and its

> EDO --- 007766 92-03358-A-00

EPRI Evolutionary Plant SER

1B.C-65

recommendation that unapproved revisions to codes and standards be reviewed on a case-by-case basis.

Item D. Leak Before Break

2

We agree with the staff's recommendation to extend the application of the leak-before-break approach for both evolutionary and passive advanced light water reactors.

Item E. <u>Classification of Main Steamlines of Boiling Water</u> Reactors (BWRs)

> We agree with the staff's recommendation for resolution of the main steamline classification for both evolutionary and passive BWRs.

Item F. Tornado Design Basis

Based on a study (NUREG/CR-4661) that compiled a considerable quantity of tornado data, the staff recommends that the maximum tornado wind speed of 300 mph (compared with the present 360 mph) be used for the design-basis tornado. We agree that the best available data should be used, but caution that design-basis specifications have sometimes been established conservatively to provide margins to deal with events not specifically addressed in the design basis. We recommend that the staff's position be approved with a qualification that the staff require assurance that other potential loads that may have been previously subsumed within the tornado design basis be taken into account if necessary.

Item H. Containment Leakage Rate Testing

The staff recommends that the maximum interval between Type C leakage rate tests for both evolutionary and passive designs be increased to a 30month interval from the 24-month interval now required in 10 CFR Part 50, Appendix J. No significant safety penalty caused by this change has been identified. We agree with the proposed staff position.

Item I. Post-Accident Sampling System (PASS)

The staff is requesting approval of changes in requirements for the PASS currently found in 10 CFR 50.35(f)(2)(viii). These requirements, and the

guidance contained in Regulatory Guide 1.79 and in NUREG-0737, resulted from consideration of the TMI-2 accident.

3

We agree with the staff's proposal but have the following comments:

- The requirements as contained in the above referenced regulation refer to "the reactor coolant system and containment that may contain TID-14844 source term redicactive materials" and to measurement of these and other materials. In light of source terms now considered in severe accident analysis, it is advisable to revise this obsolete description.
- The proposal for "Elimination of the Hydrogen Analysis of Containment Atmosphere Samples" is appropriate, given that safety grade hydrogen monitoring instrumentation will be installed.
- 3. The Electric Power Research Institute (EPRI) proposed elimination of an existing requirement for the capability to sample the reactor coolant at operating pressure in order to measure the dissolved gas and chloride in the coolant. EPRI claims that maintaining the systems on existing plants produces significant exposure of operating personnel, and that given a severe accident, no useful information, not otherwise available, is provided by this capability. The staff proposes to retain the requirement, but to change the time after accident onset at which the capability must be available from 8 to 24 hours. During our discussion with the staff, we were unable to elicit any reason for this requirement other than that it was established following the TMI-2 accident. We cannot endorse continuation of the requirement for high pressure sampling on the basis of information available to us.
- 4. The staff proposes approval of a position that "would require the capability to take samples for boron and for activity measurements 8 hours and 24 hours, respectively, after the end of power operation." The intent appears appropriate, however, we suggest that it might be better to specify a time at which the information from measurements becomes avail-

EPRI Evolutionary Plant SER

1B.C-67

able to the operator rather than the time at which samples can be taken. Further, we assume that what is required is boron concentration rather than the presence or absence of boron. Finally, we suggest that the phrase "after the end of power operation" be made more specific.

#### Item N. Site-Specific Probabilistic Risk Assessment

If, as concluded by the staff, enveloping analyses are practical for both seismic events and tornadoes, it is appropriate that these be part of the submittal at the time of certification. However, enveloping analyses are not as practical for other external events such as river flooding, storm surge, tsunamis, hurricanes, and volcanism. Therefore, the staff recommends that these other types of site-specific PRA information be submitted at the combined operating license (COL) stage. We agree with this recommendation but would like to hear more about how the staff proposes to deal with any unacceptable findings at the COL stage.

Sincerely,

hhad

David A. Ward Chairman

References:

- Draft SECY paper dated February 7, 1952, for the Commissioners, from James M. Taylor, NRC Executive Director for Operations, Subject: Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements (Draft Predecisional)
- SECY-90-010 dated January 12, 1990 for the Commissioners from James M. Taylor, NRC Executive Director for Operations, Subject: Evolutionary Light Water Reactor (LWR) Certification Issues and their Relationship to Current Regulatory Requirements
- U.S. Nuclear Regulatory Commission, NUREG/CR-4661, Subject: Tornado Climatology of the Contiguous United States, dated May 1986

STAFF RESPONSE TO ACRS LETTER OF MAY 15, 1992 JUNE 12, 1992

.



#### UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D.C. 20666

June 12, 1992

Mr. David A. Ward, Chairman Advisory Committee on Reactor Safeguards U. S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. Ward:

SUBJECT: ADVISORY COMMITTEE ON REACTOR SAFEGUARDS (ACRS) COMMENTS REGARDING THE DRAFT COMMISSION PAPER, "ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS"

In your letter of May 13, 1992, you provided the U.S. Nuclear Regulatory Commission (NRC) staff with ACRS comments regarding the NRC staff's positions on several of the issues discussed in the draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements."

The NRC staff notes that the ACRS agrees with the staff's proposed positions on several of the issues discussed in the draft paper. We also note the ACRS' comments and recommendations on the tornado design basis (Item II.F), postaccident sampling system (PASS) (Item II.I), and site-specific probabilistic risk assessment (PRA) (Item II.N) issues. We will consider your comments as we revise and finalize the subject Commission paper.

In Item II.F of the draft Commission paper, the staff stated that it will ensure that other loads that may have been enveloped within the tornado design basis will be identified and, if necessary, considered in the design basis. The staff will ensure that, if an actual site hazard exceeds the design envelope in a certain area, the applicant will perform an analysis to verify that the design remains acceptable for the site.

The staff has reviewed its basis for requiring that the PASS have high pressure sampling capability in order to measure the concentrations of dissolved gas and chloride in the reactor coolant. The staff believes that the need for high pressure sampling stems from the possibility of partially mitigated severe accidents which do not involve early reactor depressurization. This is fundamentally a PWR concern since, for a PWR, there appears to be a substantially greater probability of a severe accident being arrested while the reactor coolant system remained at pressure. In fact, the TMI-2 accident is just such an example. In addition, there is a significantly greater concern for maintaining reliable natural circulation and decay heat removal in the presence of the gases which would evolve as a PWR was depressurized. The evolution of a large amount of non-condensable gas in a passive PWR would appear to be the most sensitive scenario since the passive plant decay heat removal systems are so dependent on natural circulation. It should also be noted that the passive safety systems on a passive PWR would not take the plant to a cold shutdown condition following a partially mitigated severe

#### Mr. David A. Ward

accident. The active non-safety systems would need to be used by the operators to perform the final cooldown and depressurization. When such actions are called for, the staff believes that the operators need to have a full appreciation of the consequences of depressurizing the plant and possibly introducing non-condensable gases which could complicate or interfere with the successful termination of the event. The operator can then make informed decisions relating to issues such as: reactor coolant pump start-up and operation; instrumentation reliability; and the use of active or passive decay heat removal systems.

For the longer term, determining the oxygen and chloride concentrations would help ensure that plant personnel take appropriate actions to minimize the likelihood of accelerated primary system corrosion following the accident. This is a secondary consideration since long term samples could likely be taken at low pressure. However, once the need to take the high pressure sample has been established, because of the non-condensable gas concern, it does appear appropriate to test such a sample for oxygen and chlorides in order to get an early indication of potential corrosion problems. Therefore, the staff has concluded that PASS high pressure sampling capability is necessary. The staff has reviewed the proposed requirement for boron sampling and will clarify the requirement to specify that the analyses include determination of the boron concentration. In addition, the staff will clarify the term "end of power operations" to help ensure that the sampling time requirements are clearly understood.

The staff is developing a position on the treatment of external events in site-specific PRAs for advanced light water reactor designs. The staff will incorporate this position into a draft Commission paper which is scheduled to be forwarded to the ACRS by July 1992. The staff plans to develop a separate Commission paper discussing the form and content of the combined license (COL). This Commission paper should help address any questions the ACRS may have on resolving findings made during the COL review process.

Sincerely. James M. Laylor **Executive** Director for Operations

cc: The Chairman Commissioner Rogers Commissioner Curtiss Commissioner Remick Commissioner de Planque SECY

DISTRIBUTION: See next page

EPRI Evolutionary Plant SER

1B.C-72

ACRS LETTER RELATED TO DRAFT COMMISSION PAPER OF FEBRUARY 27, 1992 AUGUST 17, 1992


UNITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARDS WASHINGTON, D. C. 20555

August 17, 1992

Mr. James M. Taylor Executive Director for Operations U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. Taylor:

SUBJECT: ISSUES PERTAINING TO EVOLUTIONARY AND PASSIVE LIGHT WATER REACTORS AND THEIR RELATIONSHIP TO CURRENT REGULATORY REQUIREMENTS

During the 386th, 387th, and 388th meetings of the Advisory Committee on Reactor Safeguards, June 4-5, July 9-11, and August 6-8, 1992, we discussed with representatives of the NRC staff the staff's positions, recommendations, and resolution schedules concerning the certification issues for evolutionary and passive light water reactors contained in the draft SECY paper dated February 7, 1992. This supplements our letter or May 13, 1992, and provides our comments and recommendations on some of the staff's positions for the passive light water reactors. The section titles and letter designations correspond to those in the draft SECY paper.

I. <u>SECY-90-016 Issues</u> (For Passive Plants)

E. Fire Protection

The NRC staff is seeking Commission approval to use the enhanced fire protection criteria previously approved for evolutionary Advanced Light Water Reactor (ALWR) plants by the Commission's Staff Requirements Memorandum (SRM) of June 26, This SRM approved the staff's position on fire 1990. protection as presented in SECY-90-016 and supplemented by the staff's April 27, 1990 response to our report on the SECY. We recommended separate Heating, Ventilating, and Air Conditioning (HVAC) systems for each division as an important step toward ensuring adequate environmental separation of safety systems. The staff agreed that consideration of smoke, heat, and fire suppressant migration may result in separate HVAC systems, but other options may be available to the designer. Our report to the Commission of April 13, 1992, on the Draft Safety Evaluation Report for the ABWR identified the adequacy of physical separation as a continuing issue for the

ABWR, due in part to the use of a shared HVAC system for multiple trains of redundant safety systems during normal plant operation.

Our concern with shared HVAC systems is related to the need for adequate isolation of such systems during certain disruptive events (e.g., fires, floods, or pipe breaks). If the isolation is not adequate, the HVAC arrangement may become a pathway whereby effluents from the event are conducted to locations where required safe shutdown equipment is located. This is not a concern if either (1) the HVAC isolation provisions are able to withstand the event consequences (e.g., pipe whip, jet impingement, static and dynamic pressure, and elevated temperature) during and after closure with consideration of single active component failures and acceptable leakage, or (2) the safe shutdown equipment is qualified for the environmental exposure resulting from a release of the adverse environment at any credible location along the HVAC pathway such as duct openings or blowout locations.

Except for the concern with shared HVAC, we support the staff recommendation that the passive plants should be reviewed against the enhanced fire protection criteria approved in the Commission's SRM.

#### F. Intersystem Loss-of-Coolant-Accident

The staff's position is that designing these low-pressure fluid systems that interface the reactor coolant system (RCS) to withstand full RCS pressure (to the extent practicable) is an acceptable means for resolving this issue. For those systems that have not been designed to withstand full RCS pressure, the staff indicates that other measures will be required. We recommend approval of the proposed staff resolution, provided consideration is given to all elements of the low pressure piping system (e.g., instrument lines, pump seals, heat exchanger tubes, and valve bonnets).

#### G. Hydrogen Control

The staff recommends that the evolutionary LWR designs provide a system for hydrogen control that can safely accommodate hydrogen generated by the reaction of steam with 100 percent of the fuel cladding surrounding the active fuel. (Note: This is not 100 percent of the reactive metal in the core.) We support the staff's recommendation.

The staff also recommends that the system be capable of precluding uniform containment concentrations of hydrogen greater than 10 percent. We are aware of analytical work in

#### Mr. James M. Taylor

support . the resolution of Generic Issue 106, "Piping and the Use of Highly Combustible Gases in Vital Areas," that suggests the possibility of transition to detonation at average concentrations as low as 12 percent. We recommend that the staff do a similar analysis of the impact of hydrogen combustion, and possible detonation including stratification, before establishing a limit for the average hydrogen concentration. This is of particular importance to steelshell containments.

3

#### Ι. High Pressure Core Melt Ejection

To cope with the possible effects of direct containment heating (DCH), the staff concludes, ". . . that ALWR design should include a depressurization system and cavity Jesign features to contain ejected core debris."

DCH is an extremely improbable event, and we see no need to require two modes of coping with the possibility. Either depressurization or cavity design provisions alone should be adequate. Because of possible safety benefits for other events, reliable depressurization is the preferred approach.

#### J. Containment Performance

The staff has not yet developed an adequate technical position relating to requirements for containment performance in passive LWRs. We agree that the proposed value of 0.1 for a conditional containment-failure probability (CCFP) is reasonable but, as we stated in our letter of April 26, 1990, segarding "Evolutionary Light Water Reactor Certification 1.sues and Their Relationship to Current Regulatory Requirements," this value is defined only within the context of a family of initiating events. It should be used by the staff in the development of its requirements and not merely passed on to applicants.

The deterministic criterion proposed by the staff is not a simple alternative to the CCFP. It could be used more logically as a complement. Using ASME Code Service Level C stress limits is not unreasonable given a known loading for which the containment is to be designed. However, determination of the appropriate loading is the hard part of the problem and the suggested deterministic criterion is essentially meaningless without it. The staff states that "applicants using the deterministic approach will be required to define the challenges considered in this evaluation." The staff takes no position on what those challenges should be or how they are to be quantified. Apparently the intent is to default to a "design specific review." This approach leaves

#### Mr. James M. Taylor

the applicant without any real guidance from the Commission on this important topic.

- 4

We acknowledge that it is a very difficult task to establish containment performance criteria but is important. We suggested what we believe to be the best approach in our letter of May 17, 1991, "Proposed Criteria to Accommodate Severe Accidents in Containment Design."

### K. Dedicated Containment Vent Penetration

The staff proposes that the decision on the need for a containment vent for passive designs should not be made at this time but should wait until specific plant designs are evaluated. We believe that the Commission should make a generic judgment about the acceptability of containment vents for LWRs. This should be a part of establishing general criteria for containment design as proposed in our letter of May 17, 1991.

#### L. Equipment Survivability

We agree with the staff's recommendation that features provided only for severe-accident mitigation for the passive plant designs not be subject to the environmental qualification requirements of 10 CFR 50.49, quality assurance requirement of 10 CFR 50, Appendix B, and redundancy/ diversity requirements of 10 CFR 50, Appendix A.

#### N. In-Service Testing of Pumps and Valves

We support the staff recommendation that the special pump and valve design, testing, and inspection provisions be imposed on all safety-related pumps and valves for the passive ALWRs.

#### III.E - Control Room Habitability

There were several significant differences between the staff and EPRI at the time the staff drafted this policy issue. EPRI has subsequently made a proposal to modify its Utility Requirements Document to include a requirement for a passive, safety grade, control room pressurization system that would use a bottled air supply to maintain operator doses within regulatory limits for the first 72 hours following an accident. (The regulations require that operator doses be so limited for the <u>duration of the accident</u>.) The pressurization system proposed by EPRI would be designed to be replenished by off-site portable supplies after 72 hours if needed. Accordingly, EPRI has recommended that the staff close this issue.

#### Mr. James M. Taylor

August 17, 1992

We discussed this matter with the staff and EPRI during our June 4-5, 1992 meeting. The staff told us that it is currently evaluating the EPRI proposal and is not prepared to close this issue. ACRS had several comments regarding design features of the passive control room pressurization system proposed by EPRI. We believe that the staff should take these comments into account in its evaluation. We may provide additional recommendations after the staff has completed its evaluation.

Sincerely,

David A. Ward Chairman

References:

- Draft SECY Paper dated February 7, 1992, from James M. Taylor, Executive Director for Operations, NRC, for the Commissioners, Subject: Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements
- SECY-90-016 dated January 12, 1990, from James M. Taylor, Executive Director for Operations, for the Commissioners, Subject: Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements
- 3. Memorandum dated April 27, 1990, from James M. Taylor, Executive Director for Operations, NRC, for NRC Commission, Subject: Staff Response to ACRS Conclusions Regarding Evolutionary Light Water Reactor Certification Issues

### ANNEX D

# DRAFT COMMISSION PAPER "DESIGN CERTIFICATION AND LICENSING POLICY ISSUES PERTAINING TO PASSIVE AND EVOLUTIONARY ADVANCED LIGHT WATER REACTOR DESIGNS" JULY 6, 1992

For: The Commissioners

From: James M. Taylor

Executive Director for Operations

Subject: DESIGN CERTIFICATION AND LICENSING POLICY ISSUES PERTAINING TO PASSIVE AND EVOLUTIONARY ADVANCED LIGHT WATER REACTOR DESIGNS

Purpose: To present the Commission with several additional issues that the staff has identified pertaining to passive and evolutionary light water reactors (LWRs) and to request that the Commission approve the positions described in this paper. The staff will review those issues for which it has not recommended a position, and will provide the Commission with a recommendation when the review is complete.

Summary: The staff has discussed seven additional tachnical and policy issues pertaining to either evolutionary LWRs, passive LWRs or both. In addition, the staff has provided the results of its review concerning the regulatory treatment of nonsafety-related systems in passive plant designs. The staff has underlined the positions for which it is requesting Commission's approval.

CCNTACTS: T. Hiltz, NRR 504-1105

R. Perch, NRR 504-2844

### DRAFT

EPRI Evolutionary Plant SER

18.D-3

The Commissioners

Background:The staff continues to inform the Advisory Committee on<br/>Reactor Safeguards (ACRS) and the Commission of new policy<br/>issues early in the review process, so that the Commission<br/>can determine the approach for resolving them. The staff is<br/>providing an analysis of the areas in which it proposes to<br/>depart from current regulations or in which it is supple-<br/>menting or revising interpretive uidance applied to cur-<br/>rently licensed LWRs.The staff has forwarded several policy papers' to the<br/>Commission proposing resolutions for policy matters and

- 2-

Commission proposing resolutions for policy matters and major technical issues concerning both the evolutionary and passive LWR designs. The draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and their Relationship to Current Regulatory Requirements," was released to the public 7 days after it was forwarded to the Commission to support dialogue between the NRC staff, the ACRS, the EPRI, the vendors, and other industry representatives.

Similarly, the staff proposes to release this draft Commission paper and Enclosures 1 through 3 to the public 3 days after forwarding it to the Commission.

The discussions contained in this draft Commission paper may be combined with the discussions contained in the draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and their Relationship to Current Regulatory Requirements," and forwarded to the Commission as a single Commission paper.

Discussion: The staff has identified significant issues regarding the safety of evolutionary and passive advanced light water reactor (ALWR) designs. The staff proposed resolutions for some of these issues in Commission papers SECY-90-016, "Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements," SECY-91-078, "Chapter 11 of the Electric Power Research Institute's (EPRI's) Requirements Document and Additional Evolutionary Light Water (LWR) Certification Issues," and the draft Commission paper which was forwarded to the Commission on February 20, 1992, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements." The Commission

Enclosure 3 is a list of the papers that the staff has forwarded to the Commission regarding policy issues that have been identified for evolutionary and passive ALWRs. The staff will reference applicable documents throughout this paper.

EPRI Evolutionary Plant SER

The Commissioners

addressed SECY-90-016 and SECY-91-078 in its SRM of June 26. 1990, and August 15, 1991, respectively.

The issues discussed in Enclosure 1 of this paper include

- A. defense against common mode failures in digital instrumentation and control (I&C) systems B. analysis of external events beyond the design basis
- C. limination of the operating basis earthquake from seismic design
- D. multiple steam generator tube ruptures

-3-

- E. probabilistic risk assessment (PRA) beyond design certification
- F. role of a passive plant control room operator
- G. control room annunciator reliability
- H. regulatory treatment of nonsafety systems in passive plant designs

The issues concerning the role of the operator in a passive plant control room and the regulatory treatment of nonsafity systems apply only to passive plant designs. The six remaining issues apply to evolutionary and passive ALWR designs. Background information for some of these issues can be found in the previous Commission papers that are referenced in Enclosures 1 and 2.

Enclosure 1 provides a description of the eight policy issues including a discussion of the current regulatory requirement or interpretation, a discussion of the industry position and, where available, a discussion of the staff's proposed position. The staff has included a detailed discussion of the basis for the staff's position on each issue, as available. The staff has underlined the positions for which it is requesting the Commission's approval. Issues for which no resolution has been proposed will be addressed in the future.

Enclosure 2 is a list of issues pertaining to evolutionary, passive, and advanced reactor designs. Adjacent to each issue is a list of Commission papers in which the staff has addressed the issue. Enclosure 3 is a list of Commission papers which concern ALWR designs.

The staff developed the recommendations in this paper after (1) reviewing current operating reactor designs, evolutionary designs, and limited passive ALWR design information; (2) considering operating experience; (3) evaluating the results of the PRAs of LWRs and ALWRs; (4) considering the Commission's guidance on issues resolved for the evolutionary ALWRs; (5) completing the draft safety evaluation

DELAFT

EPRI Evolutionary Plant SER

1B.D-5

D

ø

١.

The Commissioners

reports for the EPRI Utility Requirements Document for passive and evolutionary ALWR designs; and (6) considering EPRI and vendor comments on these issues.

-4-

The staff considered EPRI comments regarding aspects of these additional policy issues discussed in letters from EPRI of April 9, May 5, and May 8, 1992. In its letter of April 9, 1992, EPRI provided a detailed discussion on defense against common mode failure in digital instrumentation and control systems. In its letter of May 18, 1992, Asea Brown Boveri/Combustion Engineering discussed diversity for the digital instrumentation system for its System 80+ desi,

The staff concludes that these issues are fundamental to the Agency's decisions on the acceptability of the evolutionary and passive designs. As discussed in SECY-91-262, "Resolution of Selected Technical and Severe Accident Issues for Evolutionary Light Water Reactor (LWR' Designs," the staff proposes to implement final positions on these matters as approved by the Commission through individual design certifications and generic rulemaking, as appropriate, for the evolutionary and passive designs.

The staff concludes that it would be beneficial to provide EPR] and the ALWR vendors with the staff's conclusions presented herein as expeditiously as possible in order to help resolve those matters concerning both the evolutionary and passive reactor designs, and to facilitate further discussion of the issues for the passive designs.

Therefore, the staff proposes to issue Enclosures 1 through 3 to the industry after 3 working days from the date this paper is forwarded to the Commission to support meetings between the NRC staff, EPRI, the vendors, and the ACRS. The staff will indicate that the proposed resolutions are before the Commission for consideration, and, therefore, may not be final positions.

The staff doer int expect to identify additional policy matters for t - volutionary designs. However, as the staff proceeds with its review of the EPRI ALWR Requirements Document for passive plants and the passive LWR design applications, it may find other issues not discussed herein. The staff allowed time for identifying and addressing such issues in the schedule estimates for the passice reactor design reviews provided in SECY-91-161, "Schedules for the Advanced Reactor Reviews and Regulatory Guidance Revisions." The staff will notify the Commission of new policy matters when they are found.

L'ENVIENT

EPRI Evolutionary Plant SER 1B.D-6

The Commissioners

<u>Conclusions</u>: The staff requests approval of the staff's proposed positions for those issues pertaining to evolutionary LWR designs in order to perform the final design approval and the design certification review of General Electric's Advanced Boiling Water Reactor and Asea Brown Boveri (ABB) Combustion Engineering's System BO+ LWR designs.

-5-

The staff also requests that the Commission approve the staff's proposed positions for those issues pertaining to the passive designs. This will enable the staff to proceed more efficiently with its schedule review of Westinghouse's Advanced Passive AP600 and General Electric's Simplified Boiling Water Reactor ALWR designs.

DRAFT

<u>Coordination</u>: The Office of General Counsel (OGC) has reviewed this paper and has no legal objection. OGC notes that Commission approval would be <u>tentative</u>, subject to further review in design certification rulemakings and that communications with ALWR vendors and FPRI regarding these Cowmission positions should state this fact. The staff is forwarding this paper to the ACRS for its review and comments.

Recommendations: That the Commission

- <u>Approve</u> the staff's positions underlined in Enclosure 1.
- (2) Note that the staff will seek the Commission's approval of its positions for those issues that the staff is reviewing before taking a first position with the applicant or industry.
- (3) Note that the staff will provide Enclosures 1 through 3 to the ALWR vendors and EPRI after 3 working days from the date this paper is forwarded to the Commission. The staff will indicate that the proposed resolutions are before the Commission for consideration, and, cherefore, may not be final positions.

Disk Enge

EPRI Evolutionary Plant SER

1B.D-7

The Commissioners

(4) Note that if the staff finds other policy iss.es, it will seek the Commission's approval of its positions in a timely manner.

-6-

James M. Taylor Executive Director for Operations

Enclosures:

- Policy Issues Analysis and Recommendations
- 2. ALWR Issue Cross-reference Matrix 3. Commission Papers Applicable to ALWRS

LIS C. PER

EPRI Evolutionary Plant SER

18.D-8

Enclosure 1

DRAG

#### POLICY ISSUES ARALYSIS AND RECOMMENDATIONS

The following is a discussion of eight design certification and licensing issues associated with the review of evolutionary and advanced light water reactor (ALWR) designs. The staff has included a detailed discussion of the pasis for each of its proposed positions that depart from current regulatory requirements. The staff has underlined the positions for which it is requesting the commission's approval. Issues for which no resolution has been proposed will be add ... ed in the future.

Δ., Defense Again Cummon Mode Failures in Digital Instrumentation and Control Systems

Instrumentation and control (I&C) systems help to ensure that the plant operates safely and reliably by monitoring, controlling, and protecting critical plant equipment and processes. The digital isC systems for ALWRs differ significantly from the analog systems used in operating nuclear power plants. The digital I&C system shares more data transmission functions and process equipment than an analog system would share.

Digital I&C systems share databases (software) and process equipment (haraware). Therefore, a hardware design error, a software design error, or a software programming error may result in a common mode failure of redundant equipment. The staff is concerned that the use of digital computer technology in 1&C systems could result in safety-significant common mode failures. The staff developed these concerns more fully in SECY-91-292, "Digital Computer Systems for Advanced Light Water Reactors." Some of the major points contained in SECY-91-292 are summarized as follows:

- Common mode failures could defeat not only the redundancy achieved by the hardware architectural structure but also could result in the loss of more than one echelon of defense in depth provided by the monitoring, control, reactor protection, and engineered safety functions performed by the digital I&C systems.
- The two principal factors for defense against common mode lures are quality and diversity. Maintaining high quality will increase the reliability of both individual components and systems. Diversity in assigned functions, for both equipment and human activities, and diversity in nardware and software can reduce the probability that a common mode failure will propagate.
- The staff intends to require some level of diversity, such as a reliable analog backup.

Current regulations applicable to analog I&C systems also apply to digital I&C systems. In addition, the scaff has developed limited guidance for digital IAC systems in Regulatory Guide 1.152, "Criteria for Programmable Digital Computer Software Systems in Safety Related Systems of Nuclear Power Plants."

EPRI Evolutionary Plant SER 16.D-9

-2-

However, as discussed in SECY-91-292, there are currently no regulatory requirements which adequately address the potential safety concerns associated with digital I&C systems. Quality and diversity are important defenses against common mode failures. However, there are no consensus standards for certifying the disign of digital I&C systems for application in nuclear power plant designs. In Enclosure 2 to SECY-91-292, the staff discussed regulatory requirements that it is considering to help ensure defense against common mode failures. The staff is considering new requirements in the following areas:

- assessment of diversity
- engineering activities
- design implementation
- safety classification of I&C systems

The staff has made significant progress in establishing regulatory guidance that could be used for assessing diversity. The staff, with the support of Lawrence Livermore National Laboratory (LLNL), has performed a study of the General Electric Company (GE) advanced boiling water reactor (ABWR) design to assess the adequacy of its defense in depth and diversity. This assessment was performed using the method described in NUREG-0493. "A Defense-in-Depth and Diversity Assessment of the RESAR-414 Integrated Protection System" for each transient and accident evaluated in Chapter 15 of the safety analysis report. The staff will use the results of this assessment to help determine if additional diversity is necessary to defend against postulated common mode software and hardware failures.

EPRI discussed requirements for engineering activities and design implementation for digital I&C systems in Chapter 10 of the EPRI Advanced Light Water Reactor Utility Requirements Documents (URDs) for both the evolutionary and the passive plants (Volumes II and III, respectively). The staff has discussed the issue of diversity in digital control systems with EPRI. The staff concludes that the EPRI requirements do not contain sufficient detail to help ensure adequate diversity. In SECY-91-292, the staff discussed regulatory requirements that could be developed for engineering activities and design implementation.

As discussed in SECY-91-292, the staff is continuing to develop safety classification criteria for I&C systems in ALWR designs. As discussed in SECY-91-292, the international technical community is considering this topic in the draft International Electrotechnical Commission (IEC) standard, "The Classification of Instrumentation and Control Systems Important to Safety for Nuclear Power Plants." EPRI proposed certain classification standards in its "ALWR Position Paper for Passive System Classification and Requirements," submitted by letter dated March 19, 1992. The staff will consider these positions before reaching a final position on safety classification of digital 1&C systems relates to diversity through the defense-in-depth assessment of safety credit for systems that have previously been classified as nonsafety systems.



EPRI Evolutionary Plant SER

1B.D-10

-3-

Recently, increased attention has been given to detailed assessments of the integrity of software for safety-critical functions. These assessments have covered a broad range of applications which include computer-based medical treatment facilities, computer-based fly-by-wire aircraft control systems, and protection systems for nuclear power plants. The staff found a consensus among computer science and software engineering experts that applications critical to safety should be backed-up by some system not based on software because the quantitative estimate for the reliability of I&C systems based on high integrity software cannot yet be determined. The type of this backup and the extent of the functions it should perform depend on the level of confidence in the computer systems.

As previously discussed, the staff has made sig ificant progress in establishing potential regulatory guidance to ensure adequate diversity for digital I&C system applications. As a result of its review, the staff rec. mends that the Commission approve the following approach for assessing diversity and the following requirements for a backup system which is not based on software and which is to be used for systems-level actuation and displays:

- The applicant shall assess the defense in depth and diversity of the proposed instrumentation and control system to demonstrate that vulnerabilities to common mode failures have been adequately addressed. The staff considers software design errors to be credible common mode failures that must be specifically included in the evaluation. An acceptable method of performing analyses is described in NUREG-0493. Other methods proposed by an applicant will be reviewed individually.
- In performing the assessment of defense in depth and diversity, the vendor or applicant shall analyze each postulated common mode failure for each event that is evaluated in the accident analysis section of the safe's analysis roppert (SAR). The vendor or applicant shall demonstrate adequate diversity within the design for each of these events.
- 3. If a postulated common mode failure could disable a safety function. <u>iten a diverse means</u>, with a documented basis that the diverse means is <u>unlikely to be subject to the same common mode failure, shall be</u> <u>required to perform either the same function or a different safety</u> <u>function that provides equivalent protection</u>. The diverse or different safety function may be performed by a nonsafety system if the system is of sufficient quality to perform the necessary function under the associated event conditions. Diverse digital or non-digital systems are considered to be accepted means. Manual actions from the control room are acceptable if time and information are available to the operators. The amount and type of diversity may vary among designs and will be evaluated individually.
- 4. A set of safety-grade displays and controls, independent of the computer system(s) and located in the main control room, shall be provided for system-level actuation and monitoring of critical safety functions and

### DRAFT

EPRI Evolutionary Plant SER

18.D-11

#### -4-

parameters. The displays and controls shall be provided for those system-level actuations and critical safety functions and parameters which are required by control room operators to place the reactor plant in a hot shutdown condition. The displays and controls shall be conventionally hardwired in the system architecture to the lowest level practicable. Each set of equipment required will be evaluated individually.

The hardwired system-level controls and displays provide the plant operators with unambiguous information and control capabilities. These hardwired controls and displays are required to be in the main control room to enable the operators to expeditiously mitigate the effects of the postulated common mode failure of the digital I&C system. The control room would be the center of activities to safely cope with the event which could also involve the initiation and implementation of the plant emergency plan. The design of the plant should not require operators to leave the control room for such an event. For the longer term recovery operations, credit may be taken for actions from outside the main control room, when the emergency response organization is fully briefed and in place to take such actions.

B. Analyses of External Events Beyond the Design Basis

In the Commission policy statement, "Policy Statement on Severe Accidents Regarding Future Designs and Existing Plants," issued on August 8, 1985 (50 FR 32138), the Commission stated that applicants for future evolutionary reactor plant design reviews should complete a probabilistic risk assessment (PRA) and consider improving the means to avoid or mitigate severe accident vulnerabilities exposed by the PRA in order to help ensure the public health and safety. The Commission also stated that evolutionary plant vendors should use the PRA in considering a range of alternatives and a combination of alternatives that address unresolved and generic issues and to search for costeffective means to reduce the risk from severe accidents. In the policy statement, the Commission stated that the staff should review evolutionary ALWR designs to determine the safety acceptability of the design, stressing deterministic engineering analysis and judgment, complimented by PRA. After issuing this policy statement, the Commission incorporated 10 CFR Part 52 into the regulations. Section 52.47 of 10 CFR Part 52 requires that an application for design certification contain a design-specific PRA.

In Generic Letter 88-20, "Individual Plant Examinations for Severe Accident Vulnerabilities - 10 CFR 50.54(f)," and its supplements, the staff states that construction permit holders and power reactor licensees should consider the safety implications of both internal and external events by performing separate individual plant examinations (IPEs) and individual plant examinations for external events (IPEEE). PRAs and IPEs that have evaluated both internal and external events generally estimate the risks from external events to be the same order of magnitude as internal events. Therefore, the staff concludes that the design-specific PRAs required in 10 CFR Section 52.47 should include an assessment of both internal and external events.

> DRAFT 18.D-12

EPRI Evolutionary Plant SER

- 5-

Lessons from past studies indicate that fire, internal floods, and seismic events can be important contributors to core damage. However, the estimates of core damage frequencies for the fire and seismic events continue to include considerable uncertainty. Therefore, the staff concludes that fire and seismic events can best be evaluated using the margins methods developed for existing plants, supported by insights from internal events PRAs, to find the vulnerabilities of these new designs that could be important to safety. Following this method, the designer would focus on the capacity of the design to sustain the effect of the external events, rather than relying on the highly uncertain estimates. The designer could use traditional probabilistic techniques to study internal floods.

The staff intends to determine the adequacy ("robustness") of ALWR designs by having the designer perform a modified margins analysis to determine the vulnerabilities to seismic events. The designer can best determine the seismic capability of the plant by merging PRA and margins approaches to take advantage of the strengths of each. This approach allows for a comprehensive and integrated treatment of the plant's response to an earthquake. Plant logic models covering the various systems that could be used to prevent cory damage are constructed, typically, by modifying PRA models for internal events to include logic important in considering seismic failures. The models would not include data from site-specific or generic seismic hazard curves. The designer would determine all significant operational sequences leading to safe shutdown (success paths) using the event trees and fault trees based on fragility data for each component for ... h success path. The designer would determine the value of the minimum high confidence. low probability of failure (HCLPF) for the plant by determining the HCLPF value lar each system. The HCLPF values calculated in this manner is a measure of the cobustness of the plant, being an accurate estimate of the earthquake ground motion for which the plant is expected to be able to survive without core damage. If the value of the plant HCLPF is less than about twice the design ground motion zero period acceleration, the designer should perform a more detailed evaluation to find any vulnerability against which to strongthen protection. HCLPF calculations also indicate which components and systems limit the seismic capability of the plant.

ALWR evolutionary designs that include physical separation between safety divisions appear to respond better to internal and external events than do traditional designs. This physical separation reduces the effects of the events and enables the designer to use more deterministic screening methods to assess these effects. Events such as tornadoes and extreme wind may be enveloped using bounding analyses to show that the hazard is insignificant. Bounding analyses of a site-specific external event should either (1) demonstrate that the frequency of occurrence is sufficiently low that it would not significantly contribute to risk at the site or (2) demonstrate that the design would be robust even if the external event occurred. At the design certification, the staff will evaluate fires and internal floods and other external event; that are not site dependent.

The Advisory Committee on Reactor Safeguards (ACRS) has concluded that, if enveloping analyses are practical for seismic events and tornadoes, these analyses should be part of the submittal for design certification. The ACRS

EPRI Evolutionary Plant SER

1B.D-13

DRAFT

#### -6-

agrees that the applicants should submit analyses of other external events that may be site-specific (i.e., storm surge, tsunamis, hurricanes, river flooding and volcanism) for the combined construction/operating license (COL) review. However, the ACRS stated that they are unsure how the staff will resolve any unacceptable findings at the COL stage.

In its letter of May B, 1992, EPRI stated that the seismic margins assessment is the only suitable margins approach that it knows for evaluating external events. EPRI recommended performing PRA evaluations for those external events that are not eliminated from detailed evaluation.

The staff proposes that ALWR vendors submit analyses of external events beyond the design basis that are separate from analyses required for COL applicants. Different requirements are needed because the site-specific characteristics are not known at the design certification state, and the as-built design details necessary to confirm assumptions made in the vendor's analyses are not all available. Therefore, at the design certification stage, the staff can state its conclusion about the existence of potential vulnerabilities relative to the overall design only. External events analyses sometimes reveal subtle vulnerabilities that can be found only by analyzing detailed designs and by physically walking down the plant.

At the COL stage, the staff will review the site-specific characteristics to ensure that events enveloped by bounding analyses at the design stage have been properly addressed. The staff plans to conduct walkdown inspections to confirm that design commitments have been met.

The staff recommends that the Commission approve the following staff positions regarding analyses of external events beyond the design basis: (1) The staff will require that the analyses submitted in accordance with 10 CFR Section 52.47 include an assessment of internal and external events. PRA insights will be used to support assessment of fire and seismic events. Traditional probabilistic techniques can be used to evaluate internal floods. Site-specific events such as tornadoes and extreme wind may be enveloped using bounding analysis to show that the events are insignificant: and (2) In performing the design certification review, the staff will evaluate external events that are not site-dependent such as fires and internal floods and appropriate bounding analysis. In performing the COL review, the staff will review the site-specific characteristics to ensure that events enveloped by bounding analyses have been properly addressed.

Elimination of Operating-Basis Earthquake from Seismic Design C. .

In SECY-90-016, "Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements," the staff requested the Commission's approval to separate the magnitude of the ground motion of the operating-basis earthquake (OBE) from that of the safe shutdown earthquake (SSE). The Commission approved the staff's position in its SRM of June 26, 1990. In the draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," the staff further requested the Commission to

-7-

approve eliminating the OBE from the design of systems, structures, and components in both evolutionary and passive advanced reactor designs. The proposed amondment to 10 CFR Part 100 would allow, as an option, that the OBE be eliminated from design certification when the OBE is established at less than or equal one third the SSE. In this manner, the OBE serves the function as an 'inspection level earthquake' below which the effect on the health and safety of the public would be insignificant and above which the licensee would be required to shut down the plant and inspect for damage.

The OBE level that the staff will propose in its revision of Appendix A to 10 CFR Part 100 is one-third of the SSE. This proposed revision states, as an option, that for an OBE equal to one-third or less of the SSE, the OBE can be eliminated from design considerations. The elimination of the OBE from design was requested by EPRI and also recommended by the ACRS in its letter of April 26, 1990.

The staff is assessing the safety margins of several areas of nuclear plant design when the OBE is eliminated from consideration. The industry and staff recognize that earlier seismic criteria resulted in certain aspects of the plant design such as the piping systems being controlled by the OBE. The industry and the staff view the 'controlling' nature of the OBE design as additional margin above the safety margins established by the design bases. Therefore, eliminating the OBE would not result in a significant decrease in the overall plant safety margin. The staff is currently performing a detailed evaluation of the extent to which the OBE controls the design and the effect on the design of structures, systems, and components when the OBE is eliminated from design consideration.

The overall design of reactor site structures is generally conservative, and the structural responses for all combinations of loads, including those from earthquakes, are kept at or below the material yield stresses to preclude plastic deformation.

The staff has examined the structural load combinations and the corresponding acceptance criteria and has determined, on the basis of analyses, tests, and engineering judgment, that the structural design produced by using SSE load combinations envelope those load combinations produced by the OBE. These conclusions are consistent with the staff's licensing experience accumulated during its review of many seismic calculations for individual plants and test data from NRC-sponsored research.

The designer of piping systems considers the effects of primary and secondary stresses and evaluates fatigue caused by repeated cycles of loading. Primary stresses are induced by the inertial effects of vibratory motion. The relative motion of anchor points induces secondary stresses. The repeating seismic stress cycles induce cyclic effects (fatigue).

The staff reviewed these areas and concludes that, for primary stresses, if the OBE is established at one-third of the SSE, the load combinations with the SSE control the piping design when the earthquake contribution dominates the load combination. Therefore, the staff concludes that eliminating the OBE

EPRI Evolutionary Plant SER

18.D-15

DRAFT

-8-

piping stress load combinations for primary stresses in piping systems will not significantly reduce the existing safety margins because the load combination with the SSE loading is controlling.

Eliminating the OBE will directly affect the current methods used to evaluate the adequacy of cyclic and secondary stress effects in the piping design. Eliminating the OBE from the load combination could cause uncertainty in evaluating the cyclic (fatigue) effects of earthquake-induced motions in piping systems and the relative motion effects of piping anchored to equipment and structures at various elevations because both of the effects are currently evaluated only for OBE loadings.

Accordingly, to account for earthquake cycles in the fatigue analysis of piping systems, the staff proposes to develop guidelines for selecting a number of SSE cycles at a fraction of the peak amplitude of the SSE. These guidelines will provide a level of fatigue design for the piping equivalent to that currently provided in the Standard Review Plan (SRP)(NUREG=OBOO). Currently, the staff's guidelines in SRP Section 3.9.2 recommend an equivalent of 50 OBE peak cycles for fatigue evaluation. The staff will develop new guidelines after conducting regulatory research and, as necessary, will incorporate the guidelines into an SRP revision or into a regulatory guide. To account for earthquake cycles in the fatigue analyses of piping systems performed until the new guidance is issued, the staff proposes using one half of the peak amplitude of 75 SSE cycles to evaluate the fatigue of piping. The 75 SSE cycles at one-half its peak amplitude will provide a level of fatigue design for the piping that is equivalent to that currently provided in SRP Section 3.9.2 when the differences in the structural damping between the OBE and SSE are taken into procunt.

The staff will ensure, as an interim position, that the effects of anchor displacements in the piping caused by an SSE are considered together with the Level D Service limit. The American Society of Mechanical Engineers (ASME) Boiler and Pressure Vessel Code (ASME Code), Section III, Paragraph NC/ND-3655 specifies that seismic anchor displacement effects do not need to be considered for Service Level D. The ASME Code requires that seismic anchor motion stresses be considered for Service Level B for which the OBE has been traditionally the designated seismic loading. If the OBE was eliminated from the piping design, the ASME Code, Section III evaluation would have no requirement for considering the effects of seismic anchor motion. The staff's recommendation will correct this deficiency and will require an evaluation of seismic anchor motion effects for the SSE alone. Its effects would be evaluated to the Service Level D limit for which the SSE has been traditionally the designated seismic loading. The staff will continue to develop this approach and consider national consensus codes and standards activities.

Pipe rupture is a rare event that can be caused by errors in the design, construction, or operation; unanticipated loads; or unanticipated corrosive environments. The staff notes that piping failures generally occur at high stress and fatigue locations, such as at the ends of a piping system where it connects to component nozzles.

DRAFT

1B.D-16

#### -9-

Recent dynamic pipe tests conducted by EPRI and the NRC demonstrated that the butt-welded piping can withstand seismic inertial loadings higher than an SSE without rupturing. Thus, the staff concludes that the likelihood of a doubleended pipe rupture caused by an OBE level earthquake is remote. Operating experience has shown that pipe failures (splits, through wall cracks and double-ended pipe ruptures) are more likely to occur under conditions caused by normal operation. These conditions include erosion-corrosion, thermal constraint, fatigue, and operational transients.

The staff recommends that, when the OBE is eliminated from the design, no replacement earthquake loading be used to establish the postulated pipe rupture locations. The staff recommends that the criteria for postulating pipe ruptures in high energy piping systems be based on factors attributed to normal and operational transients alone. Further reduction in the number of postulated pipe rupture locations can be considered when compensatory measures are established to minimize the potential for pipe ruptures during normal operating and transient conditions (e.g., control of erosion/commonsion or use of upgrade piping materials). However, the guidelines for the environmental qualification and compartment pressurization are currently based on the mechanistic break locations. Therefore, the staff proposes to revise its criteria for these areas to ensure that each area containing a high energy line be provided with appropriate guidelines to account for the environmental effects of a mass and energy release equivalent to that of a pipe rupture or crack without considering the dynamic effects. The staff plans to develop the environmental guidelines and will issue the guidelines for public comment.

Eliminating the OBE from explicit design consideration affects several aspects of the seismic qualification of safety-related mechanical and electrical equipment. When equipment is qualified by analysis, the acceptance criteria are derived from the ASME Code. Therefore, the conclusions drawn for piping stresses apply equally to this mechanical and electrical equipment and eliminating the OBE should have a negligible effect on the results of analyses performed to qualify equipment. When equipment is qualified by testing, vibration tests are performed for five OBE events with various orientations and then a test for the SSE is conducted. If the OBE is not considered, the equipment may need to be tested first with several fractional SSE loadings and then with one full SSE loading, or the duration of the vibratory motion of the SSE should be changed. Eliminating the OBE will not reduce the margin below the margin provided by current practice. The staff recommends testing equipment to ensure it can withstand fatigue to the degree provided in existing regulatory guidance: 50 cycles at one-half of the SSE peak amplitude or 150 cycles at one-third of the SSE peak amplitude.

The staff will conduct research to determine the number of SSE cycles and the fraction of their peak amplitude for which the equipment is to be tested. The staff will also review the results of research conducted by the industry standards group, as appropriate.

EPRI agrees that the OBE should be eliminated from design and analysis requirements for ALWR designs. The EPRI Requirements Document will require a seismic margins assessment which demonstrates a margin for an earthquake substantially larger than the SSE.

EPRI Evolutionary Plant SER

-10-

The staff has evaluated the effect on safety of eliminating the CBE from the design load combinations for selected systems, structures, and components. The staff is continuing to review this issue and is developing appropriate criteria for an analysis using only the SSE. The staff is not requesting the Commission to approve the interim positions discussed above. The staff will keep the Commission informed as the review progresses and will note in case-specific safety evaluations, instances in which the applicant proposes to use an SSE-only analysis. The staff will include its recommendations in these reports.

#### D. Multiple Steam Generator Tube Ruptures

### Multiple Steam Generator Tube Ruptures for Passive PWRs

The design-basis accident involving steam generator tube rupture (SGTR) in the current generation of pressurized water reactors (PWRs) is the Eupture of one steam generator (SG) tube, with the rate of discharge of primary coolant through the SG tube break greater than the normal charging capacity of the reactor coolant inventory control system. The staff is considering whether multiple SGTRs should be included in the plant design basis for advanced PWR designs. The SRP Section 15.6.3 requires the applicant to analyze for a single SG tube rupture. Currently, there is no requirement to analyze for multiple SG tube ruptures.

In NUREG-0844, "NRC Integrated Program for the Resolution of Unresolved Safety Issues A-3, A-4, and A-5 Regarding Steam Generator Tube Integrity," of September 1988, the staff estimated the probabilities of single and multiple tube ruptures. When the staff prepared these estimates in 1986, four single SGTRs had occurred in PWRs. All occurred in the United States and all the affected plants were Westinghouse plants: Point Beach Unit 1 (February 1975); Surry Unit 2 (September 1976); Prairie Island Unit 1 (October 1979), and R. E. Ginna (January 1982). Since that time, two more single SGTRs have occurred in the U.S., one at North Anna Unit 1 (July 1987) and another at McGuire Unit 1 (March 1989).

In NUREG-0844, the staff estimated the frequency of a single SGTR to be 1.5 x  $10^{-2}$  per reactor year (RY). The staff based this estimate on the four events that occurred in approximately 300 "mature" reactor-years (RYs) of operation of Westinghouse plants in the U.S. ("mature" RYs are accumulated after the first 2 years of plant operation). Combustion Engineering and Babcock and Wilcox plants, which had accumulated 77 and 66 mature reactor-years, respectively, at that time without experiencing any SGTRs, were assumed to have the same probability of SGTRs as Westinghouse plants. In the same report, the staff estimated the probability of a multiple tube rupture event, using binomial statistics, as 1.6 x  $10^{-3}$  per RY. The staff based this estimate on a 50 percent level of confidence (probability) for an event that had never occurred in the approximately 440 mature RY accumulated among all U.S. PWRs at that time.

DRAFT

18.D-18

EPRI Evolutionary Plant SER

#### -11-

Since the staff issued NUREG-0844, the total number of mature RY of operation for both Westinghouse PWRs and all U.S. PWRs has approximately doubled. Westinghouse plants have now accumulated approximately 535 mature RYs, and all U.S. PWRs have accumulated about 827 mature RYs. The experience with Westinghouse plants (6 SGTRs in 535 RYs) indicates that the frequency of a single SGTR is approximately  $1.1 \times 10^{-2}$  per RY. Combustion Engineering and Babcock and Wilcox plants appear to have lower SGTR frequencies, since with a failure rate of 1.1 x 10" about 3 SGTRs would have been expected in nearly 300 RY of operating experience, and none have occurred. The NRC has not received any report of a multiple SGTR in any U.S. or foreign plant. For consistency with the NUREG-0844 estimate of the probability of multiple SGTR events, a new estimate has been derived based on a 50 percent confidence level for an event that has not occurred in approximately 827 RY of U.S. PWR operation to date. This estimated frequency for a multiple SGTR event is approximately 8.4 x 10"\*/RY.

The causes of SGTRs are generally grouped into two categories: those which occur as initiating events, and those which occur as a consequence of other events that increase the stress on the SG tubes. The probability estimates given above are for SGTRs caused as initiating events. These events include random SGTRs caused by degradation of the tube over time and SGTRs caused by or associated with damage from foreign objects that may be present in the steam generator. Of the 4 SGTRs reported in NUREG-D844, two (Ginna and Prairie Island) are believed to have been caused at least in part by the impact of foreign objects on the steam generator tubes. The SG tubes in other plants have also leaked because of damage from foreign objects, although this leakage did not exceed the makeup capacity of reactor coolant inventory control systems. This issue is of concern in the context of determining the credibility of multiple SGTR events. While it would seem highly improbable that two random SGTR failures would occur simultaneously (as indicated in NUREG-0844), damage or tube failure caused by a foreign object could be a more likely initiator of multiple SGTR. In the Ginna event, the licensee examined the SG tubes after the event and found that although only 1 SG tube had ruptured, more than 20 SG tubes had been severely damaged.

The staff is reviewing the issue of whether to consider a single SGTR or a multiple SGTR as the design basis event for the AP600. The staff is concerned that an AP600 plant could respond in substantially different ways to the two accidents and that a multiple SGTR event could pose substantial challenges to the plant's passive safety systems.

In dealing with a SGTR in a conventional plant, operators isolate the faulted steam generator and reduce the primary system pressure to help stop primaryto-secondary leakage. The operators use the safety-related pressure- and inventory-control systems in these plants (pressurizer spray, high-pressure safety injection) to carry out these procedures. While no multiple tube ruptures have occurred, leak rate would likely increase with the number of tubes ruptured, and the operators would act to mitigate the consequences of the event as quickly as possible. However, the basic procedures to be employed by plant operators in such an event would be similar to those used for single tube ruptures, and the plant conditions would probably be similar

EPRI Evolutionary Plant SER 18.D-19

DRAFT

-12-

during the transient to those in a single SGTR. Analyses and tests of multiple tube rupture at the SEMISCALE facility have confirmed that the basic plant response is similar to that for a single SGTR event.

The AP600 plant includes no active safety-related inventory or pressure control systems. The core makeup tanks (CMTs) add high-pressure inventory by providing a gravity-driven injection of borated water. A natural circulation passive residual heat removal (RHR) system provides safety-related decay heat removal. The AP600 also uses an automatic depressurization system (ADS) to reduce the primary system pressure in the event of a loss-of-coolant accident (LOCA), which permits injection of a large amount of low-pressure makeup water from the in-containment refueling water storage tank (IRWST). The first stage of the ADS is triggered upon reducing the CMT level to a predetermined point, with subsequent stages actuated as CMT level reaches successively lower values.

Primary system depressurization below that of the secondary system appears undesirable during an SGTR. While the pressure of the primary system should be reduced to about that of the secondary system to inhibit primary-to-second-ary leakage, using the ADS will likely further lower the RCS pressure to such an extent that unborated water could flow from the secondary side of the steam generators back into the primary system. This could cause reactivity to increase in the core, with possible detrimental results. Westinghouse representatives assert that the AP600 has been designed with sufficient margin to the ADS initiation setpoint to allow at least 30 minutes of CMT injection after a single SGTR without triggering the ADS. Westinghouse claims that this should be sufficient time for the operators to employ both safety-related and available non safety-related systems to reduce 'S pressure, isolate the faulted steam generator, and terminate the event.

However, if a multiple SGTR occurred, with a substantially greater leak rate of primary coolant, the AP600 may not be able to accommodate the accident without actuating of th ADS. The operators will have substantially less time to bring the event under control before the CMT level is reduced to the ADS setpoint. ADS actuation might result in secondary-to-primary leakage of unborated water. This water could flash to steam as it enters the RCS if the steam generator water is above the saturation temperature at the primary system pressure. Since the AP600's passive safety systems rely on small differential pressures to circulate and inject emergency core coolant (ECC), introducing a large amount of steam into the RCS from flashing secondary water could pressurize the primary system and disrupt or degrade ECC injection. Therefore, contrary to the response of current plants, the plant may respond to a multiple SGTR event in a manner considerably different from that for a single SGTR. The consequences may also differ significantly.

The designer could provide a number of methods for minimizing the consequences of multiple SGTRs, especially to retard or prevent secondary-to-primary leakage or to lessen the amount of reactivity added as a result. These methods include (1) depressurizing the secondary system to maintain the RCS pressure at a value greater than the secondary pressure and prevent back leakage; (2) providing a system to borate the secondary water automatically if it leaks into the RCS from the steam generator; (3) or providing procedures

EPRI Evolutionary Plant SER 1B.D-20

DRAFT

#### -13-

that inhibit ADS actuation if the primary-to-secondary barrier is breached. The staff is not aware that Westinghouse is considering any of these SGTR mitigation methods for the AP600.

Design basis accidents, such as large-break and small-break LOCAs, have estimated frequencies of occurrence of the order of 10<sup>-5</sup> per RY to 10<sup>-5</sup> per RY. These events generally provide the most rigorous test of the plant's safety systems. Recognition that the multiple tube rupture frequency could be in the range of 10<sup>-5</sup> per RY to 10<sup>-6</sup> per RY and that the passive plant response for a multiple tube rupture could be significantly different from that for a single tube rupture has led the staff to conclude that rupture of more than a single tube should be considered within the design basis of the plant.

The staff is continuing to evaluate the appropriate number of ruptured SG tubes that should be included within the AP600's design basis. As a minimum, the plant designer or applicant should analyze the multiple SGTRs event to determine, to the extent possible, the quantitative differences in the plant's response to these events. Therefore, as an interim step, the staff intends to require that Westinghouse include analyses of ruptures of 2-5 tubes in its AP600 safety analysis. The staff will report to the Commission when it determines the number of tube ruptures to be incorporated into the design basis of the AP600.

EPRI stated that their Utility Requirements Documents require that future pressurized water reactors have substantially improved capability to handle SGTRs. EPRI also stated that these requirements address improvements in materials, design, and operation to prevent SGTRs and address design features to improve the performance and response of the plant after an SGTR. With regard to multiple SGTR, the EPRI has concluded that the passive plants are not unique with regard to multiple SGTRs, and accordingly, multiple SGTRs should not be included in the design basis for passive LWRs.

The staff recommends that the Commission approve the staff's positions to require that the analysis of multiple SGTRs of 2 to 5 SG tubes be included in the application for design certification for passive PWRs. The staff will evaluate these analyses during the final design approval and design certification review process to help determine the number of SG tube ruptures that will be incorporated into the passive PWR design basis. The staff may require an applicant to supmit additional analysis for ruptures of more than five SG tubes.

#### Containment Bypass Potential Resulting From Multiple SGTRs

The staff has identified an additional containment performance issue that has not been adequately addressed. Specifically a rupture of one or more SG tubes could lead to a bypass of the containment. During a SGTR event, the SG safety or relief valves could be actuated, discharging primary system radioactive inventory outside the containment. The staff concludes that the applicant for design certification should consider providing means to mitigate this containment challenge. This issue applies to both evolutionary and passive PWR designs.

1B.D-21

-14-

In discussing SECY 9C-016 issue III.D on containment performance, the starf emphasized the importance of maintaining containment integrity following a postulated severe accident. The Commission endorsed the staff's goal of reducing the probability for conditional containment failure through use of quantitative guidelines or alternative deterministic objectives. The EPRI Requirements Document states that PWR containments should be designed to provide a leak-tight barrier to prevent uncontrolled release of radioactivity in the event of a postulated (design basis) accident. Containment bypass due to SGTRs could violate containment integrity and hamper meeting the severe accident goals discussed in SECY-90-016.

The staff concludes that containment bypass resulting from multiple SGTRs can be a significant challenge to containment integrity. Therefore, the staff concludes an applicant for design certification or plant designer should consider design features that would reduce or eliminate containment bypass leakage from such a scenaric. Features that could mitigate the releases from a tube rupture include:

- A highly reliable (closed loop) steam generator shell side heat removal system that relies on natural circulation and stored water sources.
- A system which returns some of the discharge from the steam generator relief valve back to the primary containment.
- Increased pressure capacity on the steam generator shell side with a corresponding increase in the safety valve setpoints.

The staff recommends that the Commission approve the staff's positions to require that the applicant for design certification for a passive or evolutionary PWR assess design features to mitigate the amount of containment bypass leakage that could result from SG tube ruptures. The applicant or plant designer should consider the mitigation features that would likely be available following a postulated severe accident. The staff concludes that PWR designers should assess such features and address the desirability of this mitigation function. The staff will review this issue when it performs the design certification review.

E. PRA Beyond Design Certification

A plant-specific PRA is an excellent method for assessing overall safety and integrating plant systems and human interactions. Evaluation of a PRA can also reveal important engineering evaluations, assumptions and uncertainties. In the advanced design certification process, PRA insights are used to select among design options, to strengthen the design against previously known vulnerabilities, to characterize the design, and to evaluate the balance between event prevention and mitigation in the design.

# DRAFT

EPRI Evolutionary Plant SER

18.D-22

#### -15-

At the COL stage, the applicant may be able to provide site-specific information and detailed design information which was not available during the certification process. During the construction stage, the applicant can also consider as-built information. The information from both of these stages could affect the PRA insights previously developed. Experience has shown that subtle design interfaces involving support system systems interactions, or man-machine interfaces can significantly affect is risk profile of a plant. The staff concludes that updating the PRA insights can affect implementation of programs in activities such as training, emergency operating procedure development, reliability assurance, maintenance activities, and 10 CFR 50.59 evaluations. Therefore, the PRA should be revised to account for the more detailed information and refinements in the level of design detail. Currently, there are no regulatory requirements that address revising the PRA after it has been completed.

EPRI considers the PRA to be a valuable tool to support plant operation. and recommends that the PRA should be revised once the plant is built and maintained for the duration of the operating license.

The staff recommends that the Commission approve the staff's position that, throughout the duration of the combined or operating license, the PRA should be revised to address significant plant modifications, operating experience. and other developments that may affect previous PRA insights. The PRA for each plant could be revised as part of the process for revising the final safety analysis report (FSAR) required in the regulations.

#### F. Role of the Operator in a Passive Plant Control Room

In SECY 91-272, "Role of Personnel and Advanced Control Rooms in Future Nuclear Power Plants," the staff discussed an issue regarding the role of the operator in a passive plant control room. Specifically, operators in a passive plant control room may use nonsafety-related systems and active "investment protection" systems as the primary means to mitigate transients and accidents (regulatory treatment of these active nonsafety systems is discussed in Section H). Operators will use these systems, before safetyrelated passive systems are initiated, when responding to transients and accidents. The design of safety-related systems in the passive plant differs significantly from the design of safety-related systems in current operating plants and in evolutionary plant designs.

To operate the passive plant safely, the operator must understand the opera-tion of the "investment protection" systems and their interfaces with the safety-related passive systems. Passive plant operators will be required to perform new functions and tasks unlike those for the evolutionary plants. These new functions and tasks will be due to the new approach in operational philosophy noted above, the increase in automation, and the greater use of advanced technology in the passive plant designs. These new functions and tasks will likely involve greater reliance on monitoring and decision making rather than performing actions directed in procedures. Thus, the design must carefully define the operator's role to ensure that it properly develops the man/machine interface design to facilitate these functions and tasks.

.

-16-

EPRI stated that the ALWR program has provided for "man-in-the-loop" testing during first time engineering specified within Chapter 10 of the EPRI Utility Requirements Document. EPRI indicated that this requirement should adequately ensure that the human component in the man-machine interface system is explicitly included to correct the problem of insufficient focus on the operator in previous designs.

The staff concludes that an extensive man-in-the-loop test and evaluation program will be necessary for the passive plant control room designs. Such testing would likely require a fully functional integrated control room prototype to demonstrate that the passive designs properly consider the operator's role for ensuring plant safety.

Therefore, the staff recommends that the Commission approve the staff' position that sufficient man-in-the-loop testing and evaluation be performed and that a fully functional integrated control room prototype is necessary for passive plant control room designs to demonstrate that functions and tasks are integrated properly into the man/machine interface design. These requirements will be incorporated into the design acceptance criteria (DAC). Each applicant may demonstrate that a control room prototype of reduced scope is sufficient to demonstrate that functions and tasks are integrated properly in the man/machine interface design.

#### G. . Control Room Annunciator (Alarm)<sup>1</sup> Reliability

The annunciator system in a nuclear power plant provides a "first alert" to the control room operator of an abnormal state in the plant, usually over a full spectrum of transients from the malfunctioning of a single piece of equipment to the development of an abnormal state of one or more critical process parameters. The annunciator system also focuses the operator attention on the location and the nature of the malfunction or disturbance. The extent to which this is done is dependent upon the design features of the annunciator system.

Recent events at operating U.S. nullear plants involving the loss of the plant annunciator system have revealed the vulnerability of the power supply of these systems to single failures. At present, the NRC has no requirements specific to the annunciator system. The acceptance criteria and guidelines for I&C systems important to safety (Appendix A to SRP, Section 7.1) developed from the general design criteria for the I&C, the control room, and the protection and reactivity control systems, do not include the annunciator system by name. IEEE Standard 279, "Criteria for Protection Systems for Nuclear Power Generating Stations," states that the protection systems design should provide the operator with information pertinent to its own status and to the generating station's safety. In a few special cases, specific alarms are required to comply with regulatory requirements because they are essential for the manual initiation of protective actions.

The annunciator system consists of sets of alarm panels and sound equipment.

#### -17-

When the operating U.S. plants were being designed, the international community observed the same requirements as those in the U.S., with few exceptions such as International Electrotechnical Commission Publication 231A, Supplement to Publication 231, "General Principles of Nuclear Reactor Instrumentation," 1969. This publication gives specific requirements for the design of safety alarms but does not list their functional requirements.

The international requirements changed in 1984 when the International Atomic Energy Agency (IAEA) published Safety Guide D8, "Safety-Related Instrumentation and Control Systems for Nuclear Power Plants." This safety guide discusses top-level requirements for those I&C systems, including the control room annunciator system, that perform functions important to safety but are part of the traditional nonsafety systems.

Safety Guide DB recommends a method for determining the relative importance to safety and the general principles for developing graded requirements for design features that determine the reliability and availability of these I&C systems. The staff discussed the need for such classification in some detail in SECY-92-292, "Digital Computer Systems for Advanced Light Water Reactors."

The EPRI Utility Requirements Document for both the evolutionary and passive ALWR plants states that

The main control room (MCR) shall contain compact, redundant operator work stations with multiple display and control devices that provide organized hierarchical access to alarms, displays and controls. Each work station shall have the full capability to perform MCR functions as well as support division of task between two operators.

The display and control features shall be designed to satisfy existing regulations, for example: separation and independence requirements for Class IE circuits (IEEE Standard 384); criteria for protection systems (IEEE Standard 279); and requirements for manual initiation of protective actions at the systems level (Regulatory Guide 1.62). The M-MIS designer shall use existing defensive measures (e.g., segmentation, fault tolerance, signal validation, self-testing, error checking, supervisory watchdog programs) as appropriate to ensure that alarm, display, and control functions provided by the redundant work stations meet these standards.

Thus, EPRI requires compact workstations with full capability to perform control room functions with fully organized alarms, displays and controls. These workstations, including the alarms, are to be redundant and meet the requirements for independence and separation of Class IE and associated circuits described in IEEE 384. "Criteria for Independence of Class IE Equipment and Circuits." The requirements for redundancy also apply to the power supplies associated with these workstations. If the design of redundant stations including the annunciator system does not meet the traditional interpretation of independence (i.e., a master/slave configuration) a single

18.0 - 25

EPRI Evolutionary Plant SER

#### -18-

failure shall not result in a loss of the annunciator system. These requirements form a set of graded requirements for the alarm control, and indication functions that implement the classification approach discussed in SECY-91-292.

The staff concludes that additional requirements for ALWR alarm systems are necessary to minimize the problems experienced by operating nuclear power plants, such as the total loss of power to annunciators because of problems with their power supply. In SECY-91-292, the staff stated that additional requirements could be developed, as needed, from the EPRI Requirements document. The EPRI requirements for redundant control room workstations and displays that include the alarm functions are adequate for these stations.

The staff recommends that the Commission approve the staff's position that the alarm systems for ALWRs meet the EPRI requirements, as discussed above, for redundancy, independence, and separation. In addition, alarms that are provided for manually controlled actions, for which no automatic control is provided, and that are required for the safety systems to accomplish their safety functions shall meet the applicable requirements for Class JE equipment and circuits.

#### H. Regulatory Treatment of Nonsafety Systems

Unlike the current generation of LWRs or the evolutionary ALWR designs, the passive ALWR designs rely exclusively on natural forces, such as density differences, gravity, and stored energy for their safety systems to supply safety injection water and provide core and containment cooling. These passive systems do not include pumps, and all valves either require only DC electric power by means of batteries or are air-operated. Check valves will operate by means of pressure differential across the valve. The passive plants will not have safety-related ac electric power. The designers will likely designate all the active systems as nonsafety systems.

Although the passive ALWR designs rely on the passive safety systems to perform the design basis safety functions of replacing reactor coolant and removing decay heat, some of the active nonsafety systems are required to provide defense-in-depth and provide long term recovery capabilities. These active nonsafety systems are the first line of defense during a transient or plant upset, thus reducing challenges to the passive safety systems. These active systems include (1) the chemical and volume control system and control rod drive system, which provide reactor coolant makeup for the passive PWR and passive boiling water reactor (BWR), respectively, (2) the reactor shutdown cooling system and backup feedwater system for removing decay heat in the PWR, and the reactor water cleanup system for removing decay heat in the BWR, (3) the fuel pool cooling and cleanup system for removing decay heat from the spent fuel, and (4) the associated systems and structures to support these functions, including nonsafety-related standby diesel generators. Since these active systems are not designed to meet safety-related criteria, the designers in their Chapter 15 Ficensing design basis accident (DBA) analyses will not consider these active systems.

### DRAFT

EPRI Evolutionary Plant SER

16.D-26

#### -19-

Since the passive ALWR design philosophy departs from the current licer practice, new regulatory and review guidance is needed regarding the staff review on the nonsafety systems. In the draft Commission Paper, Pertaining to Evolutionary and Passive Light Water Reactors and Their K tionship to Current Regulatory Requirements," the staff listed the regutreatment of nonsafety systems for the passive design as a significant technical issue and indicated that the staff will propose a resolution of this issue to the Commission. The discussions provided herein are intended to inform the Commission of the staff's continuing effort, outline the staff's proposed approach for reviewing the functional performance of the active nonsafety systems, and seek the Commission's approval of the proposed approach.

In a position paper submitted by a letter of March 19, 1992, EPRI addressed its position on the regulatory treatment of nonsafety systems for passive ALWRs. EPRI's position can be summarized as follows:

- 1. The safety-related systems, structures, and components (SSCs) and a small group of additional safety features ensure public health and safety consistent with the Commission's Safety Goal Policy Statement. The additional features, which may be nonsafety grade, include those features needed to deal with the anticipated transient without scram (ATWS) and station blackout (codified transients). Requirements imposed by regulation must be addressed by these SSCs and features. PRAs are performed to demonstrate that the specific designs are consistent with these criteria.
- 2. Augmented protection functions, such as active nonsafety systems, increase the protection for the investor and minimize challenges to safety systems, and therefore are not subject to NRC regulations but satisfy owner requirements established by the Utility Requirements Document (URD). The owners maintain these systems to enhance flexibility for operation and maintenance.

The staff does not agree fully with the EPRI position that nonsafety systems are not subject to NRC requirements. First, although EPRI stated that the passive safety systems (and a small group of additional features) alone will meet the Commission's safety goals, the staff finds that it is very difficult to quantify measure the reliability of the passive systems because of the large uncertainty of passive system performance. The PRA results for the passive plants would likely take significant credit for nonsafety systems. The limited operational experience is not sufficient to address uncertainties with the performance of the passive features and the overall performance of systems to make up reactor coolant inventory and to remove heat from the core and containment.

Although uncertainties associated with the passive system performance can be minimized by carefully planning and performing soparate effects and integral system tests, the staff remains concerned about the characteristics of the phenomena associated with the passive systems. For example, the low differential pressures during natural circulation or gravity injection may not provide sufficient means of forcing sticking check valves to operate, unlike the

EPRI Evolutionary Plant SER

DRAFT

#### -20-

pumped ECCS which can overcome such stuck valves. The AP600 design relies on the core makeup tanks to provide high pressure passive safety injection. However, the driving head may not be able to inject the CMT fluid into the vessel under all conditions because it may exert less force than the back pressure and flow impedance. The staff is also uncertain about the thermalhydraulic characteristics of a gravity injection following a depressurization and blowdown to low pressure. Other thermal-hydraulic uncertainties include the effects of fouling on the heat transfer surfaces of heat exchangers and the effects of noncondensibles on natural circulation. These concerns are complicated by the fact that full functional testing of passive systems may not be possible, since core and containment heat removal require that the RCS be depressurized and vented to the containment. It is difficult to confirm the claimed high level of reliability based solely on the passive systems because of their large uncertainties, and therefore, it is important for the active systems to provide diversity and redundancy to the passive systems to prevent core damage.

Second, proposed ALWR requirements specify that the passive systems will be capable of performing their safety functions of cooling the core and containment without operator action or offsite support for 72 hours after the initiating events, and therefore, the design bases of the passive safety systems are centered on the 72-hour capability. After 72 hours, the licensee may need to rely on active systems to mitigate an accident. This increases the need to rely on nonsafety systems for long periods, unlike the practice at current plants, which use active safety systems designed for long-term operation. For example, in SRP 9.5.4, the NRC requires the licensee to store a minimum of 7 days' supply of diesel fuel for each redundant diesel generator system following a loss of offsite power and a design basis accident (DBA). In Regulatory Guide 1.27, the staff specifies that the licensee should provide a safety-related ultimate heat sink with sufficient cooling capacity for at least 30 days to safely shut down the reactor.

Third, the passive DHR designs are restricted by their heat removal rate and the ability of the passive heat removal process, which cannot reduce the temperature of the reactor coolant system below the boiling point of the water, to transfer heat to the in-containment refueling water storage tank (passive PWR design) or the isolation condenser (passive BWR design). The passive DHR systems can only be relied upon to lower primary system temperature to 400°F (after the industry proposed 72 hour accident duration), which would be classified as safe shutdown conditions for the passive ALWRs. The active DHR systems mut: be relied upon to bring the reactor to cold shutdown. The staff has received limited information on the operation of passive safety systems during reactor shutdown conditions. When a reactor is shut down, the steam generators may be unavailable, and the passive safety injection and heat removal systems may be isolated. Systems that may be isolated when the reactor is shutdown include the accumulators, core makeup tanks, passive safety injection system, and passive RHR system. Therefore, the active

<sup>4</sup> The concept of long term operations is contained in the SRP and in the Regulatory Guides. DRAFT

EPRI Evolutionary Plant SER 1B.D-28

#### -21-

systems may be the only available means of removing heat and making up core coolant. This further emphasizes the importance of active nonsafety systems.

Fourth, the active systems are the first line of defense to prevent challenges to the passive systems. Since traditional safety systems such as auxiliary feedwater and emergency ac power are now proposed to be nonsafety systems, the likelihood of total loss of feedwater or ac power could be higher than it is the current plants. For example, if the nonsafety backup feedwater systems failed in a loss-of-main-feedwater event, the feedwater could be unavailable for an extended period. If the nonsafety standby diesel generators failed during a loss of offsite power, a station blackout could occur. Although the passive safety systems should be designed to mitigate the effects of these events, such events are highly stressful to the operating staff.

Finally, active nonsafety systems and equipment may be important to prevent and mitigate damage to the core and to recover the plant after a severe accident.

For these reasons, the staff concludes that the active nonsafety systems are significant to safety in performing their defense-in-depth functions of preventing and mitigating accidents and core damage. These systems must be considered in evaluating the overall safety of the passive designs. There-fore, the staff does not agree with the EPRI position that these active nonsafety systems are not subject to NRC requirements. Rather, in reviewing the passive ALWR designs, the staff will evaluate both the performance of passive safety systems and the functional capability and availability of the active nonsafety systems to ensure that the plant has a robust defense-in-depth capability to prevent accidents and core damage. The staff desires a high level of assurance that these active systems are available when needed to perform defense-in-depth functions.

The EPRI passive plant requirement document specifies the design requirements for those active systems and equipment that perform defense-in-depth functions. EPRI includes requirements for radiation shielding, redundancy and single active failure considerations, electric power availability, protection against internal flooding and hazards, and system testing capability. For example, the requirements document requires the consafety auxiliary systems and components to (1) be arranged and shielded to permit access for operation and maintenance under conditions anticipated during and after nonaccident events leading to operation of the passive safety systems, and meet the requirements for equipment survivability if listed for use as part of severe accident management program; (2) have redundancy to ensure the defense in depth of each system, assuming either (a) a single active failure of equipment which must change state or position to perform the defense-in-depth function or (b) unavailability of this equipment because of maintenance; (3) have available electric power from normal station ac and the redundant equipment (or trains) to be separated to the extent practicable such as by receiving power from separate buses; (4) have redundant components protected against internal flooding, and other requirements related to have included in the design for any hazards in the plant; (5) be demonstrated by testing to be

DRAFT

1B.D-29

EPRI Evolutionary Plant SER

-22-

capable to satisfy the defense-in-depth requirements in system design analyses; and (6) have included in the design for any equipment necessary for plant recovery the capability to operate for the assumed 72-hour accident duration for the expected environment. The EPRI Requirements Document also requires at least one nonsafety system to provide means to maintain the reactor coolant inventory, remove decay heat, and ensure compliance with the specified design limits for infrequent and moderate frequency events without relying on safety systems except for the reactor protection system.

The EPRI requirements are noteworthy and provide important high-level goals for the active nonsafety systems. In rarticular, EPRI's goal of providing at least one active system that can respond to moderate and infrequent events without actuating the passive safety systems, is an important goal. The staff will review this capability carefully in the passive designs. However, the specific EPRI guidance falls short in specifying requirements to ensure capability and availability of safety-important systems and components. For example, the guidance does not include the scope and degree of required redundancy and separation in sufficient detail and does not specify the methodology for determining the limiting single active failure. EPRI did not specify the requirements for seismic category, wind or external flooding protection, or quality group classification criteria for these systems and components. EPRI also did not specify design criteria (such as IEEE Standard-279) for electrical systems that initiate active systems in the event of a reactor accident and did not specify the inservice inspection and testing requirements. EPRI proposed selection criteria for the technical specification (TS) controls for certain important instrumentation and process variables. EPRI also proposed these criteria for the safety SSCs that are part of the primary success paths and that function or actuate to mitigate a DBA or transient. The active nonsafety systems are not included in the TS control, but can be considered in the justification for continued operation if the passive systems fail to meet the required limiting conditions for operation.

While the active nonsafety systems will not be required to meet strict safetyrelated requirements, the staff believes that a certain subset or modified subset of the safety-related criteria is required for those important active systems. The performance requiremen's for an active system should be influenced by the importance of the system to safety, and the designer should consider the safety importance of the active systems during the design process. The staff believes that the active systems should conform to graded requirements based on their importance to safety. Thus, safety important active systems may require more strict criteria than are required for less important systems to ensure appropriate capability and availability. The vendor or applicant should also establish a graded safety classification system for I&C systems as discussed in SECY-91-292. This safety classification system accounts for I&C systems that perform functions that are important to safety, but are not part of the safety system.

The applicant should maintain the operational capability and availability of the important active systems and components by establishing a reliability assurance program that ensures adequate predictive and preventive maintenance

EPRI Evolutionary Plant SER

.

#### -22-

nd proper inservice inspection and testing, and TS control that provides appropriate configuration control and system operability requirements during power operation and shutdown conditions.

While formal equipment qualification may not be necessary, it should be domonstrated that these systems will function in the conditions they will experience during plant upset. The staff will evaluate the applicant's proposed system requirements based on safety importance and functional performance requirements.

The staff believes that safety importance of various systems can only be evaluated on a design-specific basis. This is because the relative importance of a system is influenced by the specific system design and arrangements, the test intervals of pumps and valves, and the capability and availability of active systems. The safety importance of systems is highly design dependent, and therefore, the staff requires detailed system design information to perform its evaluation.

To compare the importance of passive safety and active nonsafety systems, the staff, with contractor assistance, conducted an evaluation of a preliminary AP600 design. This was a scoping study because complete design information and success criteria were not available. The staff utilized event sequence diagrams, equivalent to event trees, to define success and failure paths and to find any alternatives to each system included in the design for each event scenario. The staff also used PRA-related methods to measure the importance of the systems in order to compare the importance of the passive systems with that of the active systems. The staff concludes strict criteria are appropriate for those systems found to be of high importance. The staff found that the active nonsafety RHR system had a very high worth, and therefore, may need strict criteria to ensure its functional capability and availability. This system was very important for both LOCA response and decay heat removal during shutdown modes. The staff also found that, while the depressurization system ranked very high in importance, the AP500 included no alternative to this system. Therefore, the CVCS system should meet very strict criteria and there should be a very reliable depressurization system.

The staff concludes that the plant designer should demonstrate that the capability and availability of each system is commensurate with its safety importance. Therefore, to perform the review for design certification, the staff will require the following information on the active nonsafety systems:

1. The plant designer or applicant should perform an evaluation to determine which active systems are important. The safety analysis report (SAR) should include (1) an evaluation of relative safety importance of both passive and active systems (including the methods of analysis, importance measures, accidents and transients scenarios analyzed, and success criteria of the nonsafety systems), (2) an evaluation to determine the limiting single failures, and (3) the preventive and mitigative capabilities of the active systems for severe accidents. The plant designer or applicant should evaluate the required system capability for initiating events from conditions encompassing both power operation and shutdown conditions.

-24-

The plant designer or applicant should follow the guidance in the general design criteria (GDC) and current safety standards, such as IEEE 279, to identify important aspects of the design on those active nonsafety systems determined to be important to safety. For example, the GDC require that SSCs important to safety should be designed to meet the quality standards commensurate with the importance of the safety functions to be performed (GDC 1); to withstand the effects of natural phenomena (GDC 2); to minimize the probability and effect of fires and explosion (GDC 3); to accommodate the effects of normal operation, maintenance, testing, and postulated accidents, including LOCA, and to be compatible with the environmental conditions associated with each of these operating conditions (GDC 4); to not to be shared among nuclear power units (GDC 5); and to be provided with an onsite and an offsite electric power system to permit functioning (GDC 17). GDC 34 and 35 require suitable redundancy in components and features, and suitable interconnections, leak detection, and isolation capabilities be provided for RHR and emergency core cooling to ensure that the system safety function can be accomplished with onsite or offsite power, assuming single failure. The staff concludes that the plant designer or applicant must, as a minimum, address these GDC and perform an evaluation to show capability and margin. These evaluations should demonstrate that the intent of the GDC have been addressed and provide an explanation if the design deviates from the GDC.

- 2. The plant designer or applicant should perform thermal-hydraulic analyses to demonstrate adequate capability of the active nonsafety systems to respond to plant upset conditions. The SAR submittal should include (1) a list of those moderate frequency and infrequent events for which the active systems are intended to provide defens: in depth by performing preventive and mitigative functions, (2) safety analyses of these events to demonstrate that the nonsafety systems alone meet the ALWR safety design criteria of not challenging the passive safety systems for moderate and infrequent events, and (3) analysis to demonstrate any additional capability to respond to DBA events.
- 3. The plant designer or applicant should determine the performance and reliability assurance requirements for the important active nonsafety systems to ensure that these systems are operable and available. These requirements may include specific design features for redundancy, separation, quality group, seismic category, protection against fire and flood, the quality assurance (QA) and quality control (QC) program, the reliability assurance program, TS control, maintenance and surveillance, inservice inspection (ISI) and inservice testing (IST), and configuration management activities.

The staff concludes that, with appropriate analysis, active nonsafety systems can be relied upon as a first line of defense-in-depth functions complementing the passive safety systems.

# DRAFT

EPRI Evolutionary Plant SER

H ..

1B.D-32
-25-

The staff recommends that the Commission approve the staff's approach for resolving the regulatory treatment of the active nonsafety systems. Specifically, a graded approach based on system importance to safety will be used to establish specific requirements of the active nonsafety systems to ensure their capability and availability. This graded approach will require, as a minimum, the plant designer or applicant to submit the evaluations and analyses described above to help verify that the capability and availability of each systems is commensurate with its safety importance. The importance of systems should be evaluated for each design because the specific design and arrangement of systems, and the required test intervals of pumps and valves, and other components will affect the relative importance of various systems. The staff will evaluate each vendor's methodology and criteria used to establish the relative importance of active nonsafety systems, and will evaluate the applicant's proposed system requirements.

## DRAFT

EPRI Evolutionary Plant SER

1B.D-33

Enclosure 2

#### ALWR ISSUES CROSS-REFERENCE MATRIX

		-	Issue		
	Category	No.	Title	Commission Papers	
Ι.	SECY-90-016 Issues	Α.	Use of Physically-Based Source Term	86-228 88-203 89-013 89-153 89-228 89-341 90-016 90-307 90-307 90-329 90-341 90-353 92-127	
		β.	ATWS	89-153 89-2/2 90-016 90-353	
		с.	Mid-Loop Operation	89-228 90-016 90-353	
		D.	Station Blackout	89-013 89-153 89-228 90-016 90-329 90-353	
		E.	Fire Protection	89-013 89-228 90-016 90-353	
		F.	Intersystem LOCA	89-153 89-228 90-016 90-353	

DRAFT

EPRI Evolutionary Plant SER 18.D-34

#### ALUR ISSUES CROSS-REFERENCE MATRIX (CONTINUED)

	-	-			
			* *		
_				_	

	Category	No.	Title	Commission Papers
Ι.	SECY-90-016 Issues (cont.)	G.	Hydrogen Control	89-013 89-153 89-228 90-016 90-329 90-353
		Н,	Core-Concrete Interaction: Ability to Cool Core Debris	89-153 89-228 90-016 90-353
		1.	High Pressure Core Melt Ejection	89-228 90-016 90-353
		J.	Containment Performance	89-228 90-016 90-353 91-273
		К,	Dedicated Containment Vent Penetration	89-153 89-228 90-016 90-329 90-353
		L.	Equipment Survivability	89-228 90-016 90-353
		Μ.	Elimination of OBE	89-013 90-015 90-329 90-353 91-135

<sup>1</sup>"Draft" refers to the draft Commission paper, "Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," which was forwarded to the Commission on February 20, 1992, and made available to the public on February 27, 1992.

-2-

EPRI Evolutionary Plant SER 18.D-35

.

Also

# ALWR ISSUES CROSS-REFERENCE MATRIX (CONTINUED)

	18.8.4.8					
-	Category	No.	Title	Commission Papars		
1.	SECY-90-016 Issues (cont.)	۴.	In-Service Testing of Pumps and Valves	89-228 90-016 90-353 91-273		

DRAFT

EPRI Evolutionary Plant SER 18.D-36

-3-

# ALWR ISSUES CROSS-REFERENCE MATRIX (CONTINUED)

Teena

			an ann an a' mananan ann an Ard B. R. Karanan an			
1000 - 100-	Category	No.	Title	Commission Papers		
11.	Other Evolu- tionary and Passive Design Issues	Α.	Industry Codes and Standards	91-273		
		8.	Electrical Distribution	91-078		
		с.	Seismic Hazard Curves and Design Parameters	91-135		
		D.	Leak-Before-Break	89-013		
		ξ.	Classification of Main Steamline of BWRs	draft		
		F,	Tornado Design Basis	Graft		
		۵.	Containment Bypass	draft		
		Η.	Containment Leak Rate Testing	85-013 89-228 91-273		
		Ι.	Post-Accident Sampling System	draft		
		J.	Level of Detail	90-241 90-377		
		К.	Prototyping	91-074 91-273		
		L.	ITAAC	91-178 91-210		
		M.	Reliability Assurance Program	89-013		
		Ν.	Site-Specific PRAs	89-013		
		0.	SAMDAs	91-229		
		₽.	Generic Rulemaking Related to Design Certification	91-262		
		Q.	Common Mode Failures in Digital I&C Control Systems	91-292		

EFRI Evolutionary Plant SER 18.0-37

all's

-4-

DRAFT

# ALWR ISSUES CROSS-REFERENCE MATRIX (CONTINUED)

		-	Issue		
	Category	Ne	Title	Commission Papers	
11.	Other Evolu- tionary and	R.	Beyond Design Basis Analysis of External Events	89-013 Graft	
	Issues (cont.)	S .	Multiple SG Tube Ruptures	none	
		τ.	PRA Beyond Design Certification	none	
		U.	Control Room Annunciator Reliability	none	

EPRI Evolutionary Plant SER 18.0-38

14 12

-5-

DRAFT

## ALWR ISSUES CROSS-REFERENCE MATRIX (CONTINUED)

-					
π.,				6.	
3.4	C 8	111	14	α.	

	Category	Nou	Title	Commission Pépers
11.	II. Passive Design Issues	Α.	Regulatory Treatment of Nonrafely Systems	89-010 90-406
	Β.	Definition of Passive Failure	77-439	
		С.	SBWR stability	89-153 91-273
		D.	Safe Shutdown Requirements	draft
	Ε.	Control Room Habitability	drafi	
		F.	Radionuclide Attenuation	draft
		G.	Simplification of Offsite Emergency Planning	88-203
		н.	Role of Control Room Operator	91-272

DRAFT

, the second

-6-

Enclosure 3

#### COMMISSION PAPERS APPLICABLE TO ALWRS

SECY-77-439, "Single Failure Criterion," August 17, 1977.

SECY-86-228, "Introduction of Realistic Source Term Estimates into Licensing," August 6, 1986.

SECY-88-147, "Integration Plan for Closure of Severe Accident Issues," May 25, 1988.

SECY-88-202, "Standardization of Advanced Reactor Designs," July 15, 1988.

SECY-88-203, "Key Licensing Issues Associated with DOE Spor red Advanced Reactor Designs," July 15, 1988.

SECY-89-012, "Staff Plans for Accident Management Regulatory and Research Programs," January 18, 1989.

SECY-89-013, "Design Requirements Related to the Evolutionary Advanced Light Water Reactors (ALWRs)," January 19, 1989.

SECY-89-153, "Severe Accident Design Features of the Advanced Boiling Water Reactor (ABWR)," May 10, 1989.

SECY-89-178, "Policy Statement Integration," June 9, 1989.

SECY-89-228, "Draft Safety Evaluation Report on Chapter 5 of the Advanced Light Water Reactor Requirements Document," July 28, 1989.

SECY-89-341, "Updated Light Water Reactor (LWR) Source Term Methodology and Potential Regulatory Applications," November 6, 1989.

SECY-90-016, "Evolutionary Light Water Reactor (LWR) Certification Issues and Their Relationship to Current Regulatory Requirements," January 12, 1990.

SECY-90-065, "Evolutionary and Passive Advanced Light Water Reactor Resources and Schedules," March 7, 1990.

SECY-90-146, "Process, Schedule, and Resources for the Review of Evolutionary and Passive Advanced Light-Water Reactors," April 20, 1990.

SECY-90-241, "Level of Detail Required for Design Certification Under Part 52," July 11, 1990.

SECY-90-307, "Impacts of Source Term Timing on NRC Regulatory Positions," August 30, 1990.

SECY-90-313, "Status of Accident Management Program and Plans for Implementation," September 5, 1990.

A Sector M

1B.D-40

DRAFT

SECY-90-329, "Comparison of the General Electric Advanced Boiling Water Reactor (ABWR) Design and the Electric Power Research Institute's (EPRI's) Advanced Light Water Reactor (ALWR) Requirements Document," September 20, 1990.

SECY-90-341, "Staff Study on Source Term Update and Decoupling Siting from Design," October 4, 1990.

SECY-90-353, "Licensing Review Basis for the Combustion Engineering, Inc. System 80+ Evolutionary Light Water Reactor," October 12, 1990.

SECY-90-377, "Requirements for Design Certification Under 10 CFR Part 52," November 8, 1990.

SECY-90-406, "Quarterly Report on Emerging Technical Concerns," December 17, 1990.

SECY-91-074, "Prototype Decisions for Advanced Reactor Designs," March 19, 1991.

SECY-91-078, "Chapter 11 of the Electric Power Research Institute's (EPRI's) Requirements Document and Additional Evolutionary Light Water Reactor (LWR) Certification Issues," March 25, 1991.

SECY-91-135, "Conclusions of the Probabilistic Seismic Hazard Studies Conducted for Nuclear Power Plants in the Eastern United States," May 14, 1991.

SECY-91-161, "Schedules for the Advanced Reactor Reviews and Regulatory Guidance Revisions," May 31, 1991.

SECY-91-178, "Inspections, Tests, Analyses, and Acceptance Criteria (ITAAC) for Design Certifications and Combined Licenses," June 12, 1991.

SECY-91-210, "Inspections, Tests, Analyses, and Acceptance Criteria (ITAAC) Requirements for Design Review and Issuance of a Final Design Approval," July 16, 1991.

SECY-91-229, "Severe Accident Mitigation Design Alternatives for Certified Standard Designs," July 31, 1991.

SECY-91-239, "Prez ,lication Reviews of Advanced LWR Designs," August 5, 1991.

SECY-91-262, "Resolution of Selected Technical and Severe Accident Issues for Evolutionary Light Water Reactor (LWR) Designs," August 16, 1991.

SECY-91-272, "Role of Personnel and Advanced Control Rooms in Future Nuclear Power Plants," August 27, 1991.

SECY-91-273, "Review of Vendors' Test Programs To Support the Dusign Certification of Passive Light Water Reactors," August 27, 1991.

EPRI Evolutionary Plant SER

....

.

-

1B.D-41

-2-

SECY-91-292, "Digital Computer Systems for Advanced Light Water Reactors," September 16, 1991.

SECY-91-348, "Issuance of Final Revision to Appendix J to 10 CFR 50, and Related Final Regulatory Guide 1.XXX (MS 021-5)," October 25, 1991.

SECY-92-030, "Integral System Testing Requirements for Westinghouse's AP600 Plant," January 27, 1992.

SECY-92-037, "Need for NRC-Sponsored Confirmatory Integral System Testing of the Westinghouse AP600 Design," January 31, 1992.

SECY-92-053, "Use of Design Acceptance Criteria During 10 CFR Part 52 Design Certification Reviews," February 19, 1992.

SECY-92-092, "The Containment Performance Goal, External Event Sequences, and the Definition of Containment Failure for Advanced Light Water Reactors," March 17, 1992.

SECY-92-120, "NRC Staff Review Schedules for the Westinghouse AP60C and the General Electric (GE) Simplified Boiling Water Reactor (SBWR) Designs," April 7, 1992.

SECY-92-127, "Revised Accident Source Terms for Light Water Nuclear Power Plants," April 10, 1992.

SECY-92-133, "Draft Safety Evaluation Report for Volume I and Volume III of the Electric Power Research Institute's Advanced Light Water Reactor Requirements Document," April 14, 1992.

SECY-92-134, "NRC Construction Inspection Program for Evolutionary and Advanced Reactors Under 10 CFR Part 52," April 15, 1992.

SECY-92-170, "Rulemaking Procedures for Design Certification," May 8, 1992.

SECY-92-196, "Development of Design Acceptance Criteria (DAC) for the Advanced Boiling Water Reactor (ABWR)," May 28, 1992.

"Issues Pertaining to Evolutionary and Passive Light Water Reactors and Their Relationship to Current Regulatory Requirements," draft Commission paper which was forwarded to the Commission on February 20, 1992, and made available to the public on February 27, 1992.

DRAFT

EPRI Evolutionary Plant SER

1B.D-42

-3-

NRC FORM 336 U.S. 1 (2-89) NRCM 1102 3201, 3202 BIBLIOGRAPHIC DATA SH (See instructions on the reverse)	NUCLEAR REGULATORY COMMISSION 1	REPORT NUMBER (Assigned by NRC, Add Vol., Supp., Rev., and Addendum Numbers, If eny.)					
2. TITLE AND SUBTITLE	NEAR DE DIVERSION DE CARLES E CARLES A DE L'ANDRE OR DE L'ANDRE DE CARLES DE LA COMPANY DE LA COMPANY DE LA COMP	Vol. 2, Pt. 1					
NRC Review of Electric Power Research Inst Advanced Light Water Reactor Utility Requi	'Electric Power Research Institute's it Water Reactor Utility Requirements Document						
Evolutionary Plant Designs Chapter 1	4	August 1992 FIN OR GRANT NUMBER					
5. AUTHOR(S)		TYPE OF REPORT					
		Safety Evaluation Report					
		July 1986-August 1992					
B. PERFORMING ORGANIZATION – NAME AND ADDRESS /// NRC. provide Division name and mailing address.	sn, Office or Region, U.S. Nuclear Regulatory Commi	saion, and mailing address, if contractor, provide					
Associate Directorate for Advanced Reactor Office of Nuclear Reactor Regulation U.S. Nuclear Regulatory Commission Washington, D.C. 20555	s and License Renewal						
9. SPONSORING ORGANIZATION - NAME AND ADDRESS (II NRC. type "Same as and mailing address.)	above"; If contractor, provide NRC Division, Office o	r Region, U.S. Nuclear Regulatory Commission.					
Same as above.							
10. SUPPLEMENTARY NOTES							
Project Number 669							
11. ABSTRACT (200 words or head)							
The staff of the U.S. Nuclear Regulatory Commission has prepared Volume 2 (Parts 1 and 2) of a safety evaluation report (SER), "NRC Review of Electric Power Research Institute's Advanced Light Water Reactor Utility Requirements Document - Evolutionary Plant Designs," to document the results of its review of the Electric Power Research Institute's "Advanced Light Water Reactor Utility Requirements Document." This SER gives the results of the staff's review of Volume II of the Requirements Document for evolutionary plant designs, which consists of 13 chapters and contains utility design requirements for an evolutionary nuclear power plant (approximately 1300 megawatts-electric).							
12. KEY WORDS/DESCRIPTORS (Litt words or opinions that will state measure in fourth	to the pages 1						
advanced light water reactor (ALWR) Utility Requirements Document (URD)	policy issues optimization subjects	Unlimited					
final design approval (FDA)	licensability	(This Page)					
design certification (DC)	Utility Steering Commit	ttee Unclassified					
evolutionary plants standardization		Unclassified 15. NUMBER OF PAGES					
regulatory stabilization Advanced Reactor Service Accident Program	(ARSAP)	16. PRICE					
NRC FORM 335 (2-89)							



Federal Recycling Program

#### UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D.C. 20555-0001

OFFICIAL RUSINESS PENALTY FOR PRIVATE USE, \$300

÷

ŝ

.

SPECIAL FOURTH-CLASS RATE POSTAGE AND FEES PAID USNRC PERMIT NO. G-67